

Single Technology Appraisal

Obeticholic acid for treating primary biliary cholangitis [ID785]

Committee papers

Any information supplied to NICE which has been marked as confidential, has been redacted. All personal information has also been redacted.



NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

SINGLE TECHNOLOGY APPRAISAL

Obeticholic acid for treating primary biliary cholangitis [ID785]

Contents:

- 1. <u>Pre-Meeting Briefing</u>
- 2. Final Scope and Final Matrix of Consultees and Commentators
- 3. <u>Company evidence submission with list price from Intercept</u> <u>3a. Erratum to the company submission (list price)</u>

<u>3b. Company PAS price submission</u> <u>3c. Erratum to the company PAS submission (PAS price)</u>

4. Clarification:

4a. NICE request to the company for clarification on their submission
4b. Company's response to NICE's request for clarification
4c. 4ci. Further request for clarification and responses
4d. Response to questions regarding Markov model

5. Patient group, professional group and NHS organisation submission from:

5a. British Association for the study of the liver (BASL) 5b. British Society of Gastroenterology (BSG) joint with Royal College of Physicians (RCP) 5c. The PBC Foundation

6. Expert personal perspectives from:

<u>6a. Clinical expert, nominated by British Society of Gastroenterology and Royal college of Physicians</u>
<u>6b. Patient expert, nominated by The PBC Foundation</u>
<u>6c. Clinical expert, nominated by British Associated for the study of the liver (BASL)</u>

7. Evidence Review Group report prepared by Kleijnen Systematic Reviews

7a. 7ai ERG PAS Appendix and ERG correction page 7b. ERG erratum

8. Evidence Review Group report factual accuracy check

Any information supplied to NICE which has been marked as confidential, has been redacted. All personal information has also been redacted.

Pre-meeting briefing Obeticholic acid for treating primary biliary cirrhosis

This slide set is the pre-meeting briefing for this appraisal. It has been prepared by the technical team with input from the committee lead team and the committee chair. It is sent to the appraisal committee before the committee meeting as part of the committee papers. It summarises:

- the key evidence and views submitted by the company, the consultees and their nominated clinical experts and patient experts and
- the Evidence Review Group (ERG) report.

It highlights key issues for discussion at the first appraisal committee meeting and should be read with the full supporting documents for this appraisal.

Please note that this document includes information from the ERG before the company has checked the ERG report for factual inaccuracies.

The lead team may use, or amend, some of these slides for their presentation at the Committee meeting.

Key decision points

- Is there high unmet medical need? What is standard of care for people intolerant to/ have inadequate response to UCDA?
- Are fibrates relevant comparators?
- What is the natural history of people with low ALP? Can ALP be maintained at this level and will it prevent progression of PBC?
- How generalisable is the clinical trial in terms of
 - the proportion of people with moderate and severe PBC?
 - the proportion of people on OCA monotherapy (n=11)
 - the value of the composite primary outcome in clinical practice (is there a strong enough relationship between the surrogate and improved long term liver outcomes)?
- How is OCA used in clinical practice (is it titrated)?
- How severe is the pruritus associated with OCA?
- Utility values for PMB/Liver health states appropriate?

Primary biliary cirrhosis (PBC)

- Also known as primary biliary cholangitis is rare, progressive, autoimmune, non-viral disease of the liver that gradually destroys the interlobular bile ducts. This causes accumulation of cytotoxic bile acids in the liver, which leads to inflammation, liver fibrosis, cirrhosis, and ultimately liver failure.
- Estimated prevalence in the UK is ~ 3.9 per 10,000 population, equating to ~ 19,175 people in England.
- Incidence is 0.58 per 10,000 population.
- ~ 90% of people with the condition are women, and age of diagnosis is typically between 30 and 65 years
- ~ 60–80% of patients are asymptomatic at diagnosis. The diagnosis in asymptomatic patients is usually established after the chance finding of an elevated ALP level during the course of an unrelated illness.

PBC progression and association with alkaline phosphatase and total bilirubin level

- The presence of AMA and elevated ALP levels are two early characteristic markers of PBC. It is important to treat patients when their ALP levels >1.67x ULN, equivalent to 200 U/L.
- An increase in serum bilirubin is detected only when significant liver damage has occurred, with a sharp increase occurring in the terminal phase. A rise in bilirubin to as low as 0.5 x ULN, is a significant change in the course of the disease



Source: CS figure 6

Elevated bilirubin

ALP and bilirubin level correlation with death and liver transplantation

 The company stated that serum ALP along with bilirubin are used to manage patients, test the efficacy of novel therapies in clinical studies, and gauge the risk of long-term adverse clinical outcomes.





Source: CS figure 7, p.35

Current management

- First choice treatment: ursodeoxycholic acid (UDCA)
- Inadequate response to UDCA in up to 70% of people
- For people who are unable to tolerate, or whose disease responds inadequately to, UDCA no licensed or effective treatments
- Immunosuppressive therapy no longer used for primary biliary cirrhosis. No benefit, severe side effects, not recommended in guidelines.
- Liver transplant is an option only for people with end stage liver disease or decompensated cirrhosis (10% of these people have disease recurrence)

Obeticholic acid (OCA)

	Granted on 12 December 2016
Conditional marketing authorisation	"primary biliary cholangitis (also known as primary biliary cirrhosis) in combination with ursodeoxycholic acid (UDCA) in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA"
	Conditional on follow up studies to demonstrate clinical benefits of reduced alkaline phosphatase and bilirubin levels e.g. delayed liver fibrosis, cirrhosis, liver transplant, death
Administration & dosage	Oral tablet (5 mg or 10 mg) Start on 5 mg/day, increase to 10 mg/day after 6 months
Duration of treatment	As long as person continues to benefit (dose reduced/interrupted for severe intolerability due to pruritis)
Cost	£2,384.04 for 30 tablets (list price) \rightarrow £29,005.78/year A confidential patient access scheme has been approved

OCA mechanism of action

- OCA is farnesoid X receptor (FXR) agonist which:
 - \downarrow bile acid synthesis
 - ↓ uptake and ↑ secretion of bile acids in and out of liver cells
 - mediates antiinflammatory and anti-fibrotic pathways



FXR regulation of biological pathways



Source: CS Figure 2

Patient perspective - the PBC Foundation

- Living with PBC has been described as "Living on an emotional and physical rollercoaster".
- Symptoms vary from week to week and day to day, patients (and carers) never know what PBC will bring on a day to day basis.
- The main symptoms include: chronic fatigue (with the linked symptom of cognitive impairment), pruritus, joint/muscle/bone pain, and nausea
- The symptoms, just themselves, can have an enormously detrimental effect on quality of life which leads to social isolation and reinforces negative quality of life issues.
- The diagnosis of an incurable, progressive disease, particularly one with such a foreboding outline can lead to an insurmountable emotional burden.
- Patients experience a lack of knowledge and understanding within medical communities, particularly in primary care and district hospital levels. It is widely anticipated that a successful novel therapy for PBC will provide opportunity for much needed education of medics about the PBC.

Decision problem (final scope)

Population	People with primary biliary cirrhosis whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid (UDCA)						
Intervention	OCA alone or in combination with UDCA						
Comparators	 For people whose disease has an inadequate response to UDCA: UDCA alone or in combination with fibrates For people who are unable to tolerate UDCA: Fibrates No additional treatment 						
Outcomes	 Mortality Liver function based on markers of liver biochemistry Symptoms, including pruritus, fatigue and abdominal pain Time to liver transplantation Primary biliary cirrhosis related events, including ascites, varices, encephalopathy and HCC Adverse effects of treatment Health-related quality of life 						

Company deviations from the scope (1): comparators

Company submission did not include fibrates, with the rationale that fibrates:

- do not have a marketing authorisation for primary biliary cirrhosis in the UK
- not standard of care
- contraindicated in primary biliary cirrhosis
- rarely used: % of patients in the UK-PBC cohort (% /2,245) have ever taken fibrates for any condition (not necessarily for PBC)
- do not have proven efficacy; there only a limited number of studies of fibrates in primary biliary cirrhosis, with the following challenges:
 - small patient numbers
 - all except 1 study was conducted in Japanese patients
 - poor trial quality and high risk of bias
 - significant safety concerns

Clinical effectiveness evidence

POISE trial

- Phase 3, randomised double-blind, placebo-controlled, parallel group trial. Stratified randomisation according to:
 - higher risk of developing clinical outcomes
 - intolerance to UDCA
 - presence or absence of biochemical response to UDCA treatment
- 59 sites in 13 countries including 7 sites in England and 2 sites in Scotland
- Intervention: OCA 10 mg, OCA titration, placebo with or without UDCA
- Primary outcomes:
 - percentage of participants at 12 month achieving the composite endpoint: ALP
 <1.67x ULN, and total bilirubin ≤ULN, and ALP decrease ≥15% from baseline
- Secondary outcomes:
 - percentage of patients having the primary endpoint in the OCA titration group at Month 12,
 - percentage of patients reaching the endpoint at Week 2, Month 3, Month 6 and Month 9, and comparing the 10 mg OCA fixed dose group with the OCA titration group at Month 6.
- The primary variable for the long-term safety extension phase was the percentage of patients having the composite endpoint based on previous treatment in the double-blind phase

POISE trial

- Patients were stratified by the presence or absence of the following biochemical response criteria and tolerance to UDCA treatment:
 - ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, and intolerant to UDCA or currently taking UDCA
 - ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, and currently taking UDCA or currently taking UDCA
- A request for titration to OCA 10 mg for the remainder of the double-blind phase (Months 6-12) could be made by the investigator for patients who met any of the following criteria at the Month 6 assessment:
 - ALP ≥1.67x ULN, and/or
 - Total bilirubin >ULN, or
 - <15% ALP reduction at month versus the mean pre-treatment value, and</p>
- Provided AEs did not limit the administration of the higher dose of OCA.

Design of the POISE trial

12-Month, Double-Blind Treatment Period (ITT Population, N = 216)

If on UDCA: continue UDCA

Placebo (n = 73)





Source: CS p.49 BL – baseline, UDCA – ursodeoxycolic acid

Overview of POISE trial



Source: CS Figure 11

POISE baseline characteristics

- 81% of patients were younger than 65 years old.
- 91% female
- 11 patients were UDCA intolerant

	Placebo	OCA titration	OCA 10 mg			
	(n=73)	(n=70)	(n=73)			
Mean age, years (SD)	55.5 (10.0)	55.8 (10.5)	56.2 (11.0)			
BMI<30 kg/m ²	79%	83%	84%			
Pre-treatment liver biopsy	10%	19%	12%			
UDCA use at baseline	93%	93%	92%			
Baseline liver parameters, ITT po	pulation					
ALP >3x ULN	32%	27%	27%			
Total bilirubin >ULN	10%	6%	10%			
ALP, alkaline phosphatase; BMI, body mass index; OCA, obeticholic acid; SD, standard deviation; UDCA, ursodeoxycholic acid; ULN, upper limit of normal						

POISE results

S	Responders at Month	12
me	Placebo (n=73)	10%
00	10 mg OCA (n=73)	47%
out	Titration OCA (n=70)	46%
Ž	Titration subgroup	
nar	Remained at 5 mg OCA for 12 months (n=36)	53%
Prir	Titrated to 10 mg OCA at Month 6 (n=33)	39%
S	ALP reduction from baseline ≥40°	% at Month 12
ame	Placebo (n=73)	1%
utco	10 mg OCA (n=73)	34%
y 01	Titration OCA (n=70)	30%
dar	Mean total bilirubin levels at	Month 12
one	Placebo (n=73)	13.2 (SE 1.0)
Sec	10 mg OCA (n=73)	9.7 (SE 0,6)
0,	Titration OCA (n=70)	9.9 (SE 0.6)

Source: CS table 23 and ERG report p. 49

Adverse events

Participants, n (%)	Placebo n=73	OCA titration	OCA 10 mg n=73
Any TEAE	66 (90)	65 (93)	69 (95)
Any treatment-related AE	38 (52)	42 (60)	54 (74)
Any SAEs	3 (4)	11 (16)	8 (11)
Mild TEAEs	29 (40)	16 (23)	19 (26)
Moderate TEAEs	28 (38)	27 (39)	29 (40)
Severe TEAEs	9 (12)	22 (31)	21 (29)
Any TEAE leading to discontinuation	2 (3)	5 (7)	8 (11)
Pruritus	27 (37)	35 (50)	48 (66)
Fatigue	8 (11)	6 (9)	6 (8)
Nausea	4 (5)	3 (4)	4 (5)
Discontinuation due to pruritus	0 (0)	1 (1)	7 (10)
Number of deaths	0 (0)	1 (1)	0 (0)
AE, adverse event; SAE, serious adverse event;	TEAE, treatmer	nt-emergent adverse	e event

ERG comments

- No trials comparing OCA with fibrates were identified.
- The group receiving OCA as monotherapy is underrepresented in the POISE trial (11 patients).
- The majority of patients in POISE appeared to be at an earlier stage of disease so the effects on those with more advanced disease are less clear.
- The POISE trial only examined surrogate outcomes; OCA showed positive effects on surrogate endpoints, and there is some evidence that surrogate endpoints are related to relevant outcomes. However, the size of the relationship is unclear.
- Two extension studies to POISE trial were not similar enough to be pooled with POISE trial. Clinical outcomes await the publication of the COBALT trial which is estimated to be in 2022.

Cost effectiveness evidence

Model structure



Source: CS Figure 24

Model details

- De novo model, lifetime horizon, 3.5% discount for utilities and costs;
 3 month cycle length
- Model comprises 2 parts:
 - biomarker component based on surrogate outcomes of ALP and bilirubin biomarkers in three different health states based on the expected risk of disease progression:
 - low risk (ALP \leq 1.67 x ULN [ALP \leq 200 units/L]);
 - moderate risk (ALP > 1.67 x ULN and total bilirubin; TB \leq 1.0 x ULN)
 - severe risk (TB > 1.0 x ULN [TB > 20 µmol/L] or compensated cirrhosis);
 - liver disease component based on clinical endpoints:
 - pre-liver transplant,
 - · decompensated cirrhosis,
 - hepatocellular carcinoma (HCC),
 - liver transplant,
 - post-liver transplant state,
 - potential PBC re-emergence and death.

Clinical data used in the model

- Data from the POISE trial and literature was used to estimate transition probabilities between the health states in the biomarker component of the model.
- The literature was used to estimate and extrapolate the transition probabilities for the health states in the liver disease component of the model.

Transition probabilities

TPs between health states in the PBC-specific component of the model

- In its base case, the company used different methods to derive TPs for each treatment arm in the PBC-specific component of the model: trial data for OCA and published literature for comparators
 - OCA monotherapy and combination with UDCA: patient-level data from POISE (TPs for OCA monotherapy were based on data from patients who received OCA with UDCA)
 - Rationale: insufficient data from POISE for OCA monotherapy
 - No treatment (for people who cannot tolerate UDCA): TPs from published literature (Corpechot 2000 study of UDCA vs. no active treatment in PBC)
 - Rationale: insufficient data from POISE (n=5) and not appropriate to assume the same TPs for UDCA-intolerant population and UDCA inadequate-responders (who continue to benefit from UDCA)
 - UDCA (for people whose disease responded inadequately to UDCA): TPs based on patient-level POISE data, adjusted using published literature (GLOBE and UK risk scores)
 - Rationale: The company expected UDCA inadequate responders to get worse after the end of the trial but TPs were not available for the full time horizon.

TPs from the PBC-specific health state ("abnormal bilirubin and rising, or compensated cirrhosis") to the more severe liver-specific health states

• Derived from published literature (see table 54 of the company submission)

Use of HRQoL data in the model

- HRQoL is assumed to be constant in each of the biochemistry states, (the utility for patients in a certain health state does not change over time).
- HRQoL worsens as patients proceed from the biochemistry component to the liver disease component of the model.
- Utility values used in the cost-effectiveness analysis have been adjusted.
- The following health states in the liver disease component of the model had their corresponding utility values decreased by to simulate the worsened HRQoL experienced by PBC patients in comparison to HBV/HCV patients, as an interpretation of KOL feedback by the company:
 - decompensated cirrhosis,
 - pre-transplant at listing, 3 and 6 months after listing health states
 - 3, 6, 12 and 24 months post liver transplant health states
- Clinical experts verified company's assumptions that PBC patients are likely to have worse utility values than HCV/HBV patients despite being in the same health state except of patients with hepatocellular carcinoma.

Utilities

State	Utility	Primary source				
Low risk	0.84	Younossi et al. 2001				
Moderate risk	0.84	Younossi et al. 2001				
Severe risk	0.55	Wright et al. 2006				
Decompensated cirrhosis		Wright et al. 2006				
Hepatocellular carcinoma	0.45	Wright et al. 2006				
Pre-transplant: utility at listing		Wright et al. 2006				
Pre-transplant: 3 months after listing		Wright et al. 2006				
Pre-transplant: 6 months after listing		Wright et al. 2006				
Liver transplant: 3 months post-transplant		Wright et al. 2006				
Liver transplant: 6 months post-transplant		Wright et al. 2006				
Liver transplant: 12 months post-transplant		Wright et al. 2006				
Liver transplant: 24 months post-transplant		Wright et al. 2006				
Re-emergence of PBC Wright et al. 2006						
ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; PBC, primary biliary cholangitis/cirrhosis						

Health-state costs and resource use

Health states	Value
Low risk	Staff: £221 (1x Outpatient appointment, 1x outpatient follow-up)
ALP: \leq 200 U/L and Bili:	Hospital costs: £27 (3 blood tests, 3 times per year, at a cost of £3)
Normal	Total: £248
Moderate risk	Staff: £345 (1x Outpatient appointment, 2x outpatient follow-up
ALP: > 200 U/L and Bili:	appointments)
Normal	Hospital costs: £27 (3 blood tests, 3 times per year, at a cost of £3)
	Total: £496
Severe risk	Total: £6,254
Decompensated	Total: £12,509
cirrhosis	
Hepatocellular	Total: £11,147
carcinoma	
Pre-transplant (end	Total: £18,217
stage)	
Re-emergence of PBC	Total: £248
Liver transplant	Total: £65,029
Follow-up 1 year after	Total for 2 years divided by 2: £18,166
liver transplantation	
Follow-up 2 years after	Total for 2 years divided by 2: £18,166
liver transplantation	

Company's base case deterministic results

UDCA intolerant population, using the PAS price of OCA

	Costs	LYG	QALYs	ICER
No treatment (placebo)	£103,233	11.30	6.61	_
OCA titration	£251,671	16.68	13.56	£21,351

UDCA inadequate responder population, using the PAS price of OCA

	Costs LYG QAL						
UDCA + placebo	£96,977	12.35	7.85	_			
OCA titration + UDCA	£261,791	£28,281					
ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.							

Source: CS tables 23, 24 (erratum PAS price)

Company's deterministic sensitivity analyses

- The following parameters was investigated through one way deterministic sensitivity analysis:
 - Discontinuation probabilities
 - Probabilities of experiencing an adverse event
 - Utility weights
 - Transition probabilities
 - Discounting
 - The probability of death for the general population
 - Adverse events costs and disease management costs (i.e. health state costs)
- In patients whose disease has inadequate response to UDCA and UDCA intolerant patients, the most influential parameters on the cost effectiveness results were the health states utility values for the health states of the biomarker component of the model and the transition probabilities between these health states.

Company scenario 1 Use of original HCV utility values

Scenario 1 analysis: UDCA intolerant patients

Technologies	Total Incremental						
	Costs	LYG	QALYs	Costs	LYG	QALYs	ICER
No treatment (placebo)	£103,233	11.30	6.91	_	_	_	_
OCA titration	£251,671	16.68	13.61	£148,439	5.38	6.70	£22,160

Scenario 1 analysis: UDCA inadequate responders

Technologies	Total			Incremental					
	Costs	LYG	QALYs	Costs	LYG	QALYs	ICER		
UDCA +	£06 077	12 35	Q 11						
placebo	290,977	12.55	0.11	_		-	-		
OCA titration	0061 701	16 70	10 70	C1C4 014	4 4 2	E 61	COO 274		
+ UDCA	£201,791	10.78	13.72	104,014	4.43	10.0	£29,374		
ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALY, quality-adjusted life									
i vear UDCA ursode	oxycholic acid								

Source: Table 35 and 36, CS erratum PAS price

Company scenario 2 Use of alternative transition probabilities

Scenario 2 analysis: UDCA intolerant patients

Technologies		Total		Incremental			ICEP
recimologies	Costs	LYG	QALYs	Costs	LYG	QALYs	ICEN
No treatment (placebo)	£94,717	10.89	6.39	-	Ι	Ι	_
OCA titration	£250,303	16.61	13.52	£155,586	5.73	7.13	£21,824

Scenario 2 analysis: UDCA inadequate patients

Technologies	Total			Incremental			
	Costs	LYG	QALYs	Costs	LYG	QALYs	ICER
UDCA +	£89,666	12.00	7.67	_	_	-	_
placebo							
OCA titration	£260,540	16.72	13.65	£170,874	4.72	5.98	£28,596
+ UDCA							
ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALY, quality-adjusted life							

year; UDCA, ursodeoxycholic acid.

Source: Table 37 and 38, CS erratum PAS price



- The decision to exclude fibrates as a comparator is inconsistent with the scope and may not be appropriate.
- There is an ongoing study on PBC patients treated with fibrates and that treatment with fibrates may present a viable comparator in the future.
- Improvements in surrogate outcomes were not reflected in the disease specific quality of life tool (PBC-40) over the 12 month period.
ERG comments

- Concerns regarding the model structure:
 - Biomarker component:
 - Aggregation of two different health states into one (compensated cirrhosis and abnormal total billirubin count)
 - Patients receiving OCA treatment and who are in the low and moderate risk biomarker component health states at the end of the first year remain there for the remainder of their lives. This holds for UDCA but may not hold for OCA. The ERG conducted the exploratory analysis (please see more in the next slides)
 - Liver disease component:
 - Company's model diverges from those used in other liver diseases in that an additional pre-liver transplant health state was introduced (compared to, for instance, TA 330). It groups patients together that came from different health states (HCC, DCC, severe risk) and who may experience different HRQoL.

ERG comments

- Population
 - Patients enter the biomarker component of the mode in the moderate (76.85%) and severe risk (23.15%) health states which is not reflective of the proportion of patients entering the POISE study in the severe risk health state (8.42%). This leads to more patients in the model remaining in the severe risk health state and moving to more severe disease (the liver disease component in the model). This in turn would potentially bias model outcomes in favour of OCA
 - The ERG considers that the proportions of patients entering the model in the moderate and severe risk health states should be based on data from POISE. Although the company state that the proportions entering the model are derived from POISE, it is not clear how this was done.

ERG comment on transition probabilities

- The ERG's concerns with estimating transition probabilities in the biomarker component are:
 - 1. Discrepancy between the transition probabilities reported in company submission Table 49 and those used in the economic model.
 - 2. Assumption of no discontinuation beyond 12 months
 - Usage of unclear calibration methods based on the literature instead of the POISE trial for the non-OCA regimen for inadequate responders to UDCA;

ERG exploratory analyses

- 1. Used transition probabilities in the company submission for biomarker component because of a discrepancy with the numbers used in the model
- 2. Used unadjusted transition probabilities from POISE for the non-OCA regimen (for biomarker component of the model)
- 3. Proportions in the starting health states from POISE
- 4. NHS reference costs for outpatient visits
- Health state costs of £1,561 for compensated cirrhosis consistent with TA330 for the severe risk health state in the biomarker component) (instead of £6,254)
- 6. Used age-dependent utilities (from the UK general population) for the low and moderate risk health states in the biomarker component of the model
- 7. Removed the HRQoL decrements (in the liver disease component of the model)

ERG base case

UDCA inadequate responders – OCA PAS price

	Incremental results		
	ΔQALY	∆Costs	ICER
Company base-case (deterministic)*	5.79	£164,551	£28,425
1. Fixing discrepancies between transition probabilities ¹	5.83	£164,806	£28,280
2. Use transition probabilities from POISE for the non-OCA regimen	5.20	£171,036	£32,897
3. POISE trial proportion in starting health states	5.55	£170,482	£30,736
4. Use NHS reference costs for outpatient visits	5.83	£165,453	£28,394
5. Use health state costs consistent with TA330	5.83	£180,737	£31,017
6. Use UK age-dependent utility values	4.93	£164,808	£33,458
7. Remove utility decrement	5.61	£164,808	£29,377
ERG base-case (deterministic)	4.17	£189,968	£45,541
ERG base-case (probabilistic)	4.22	£189,706	£44,945
ICER, incremental cost-effectiveness ratio; NHS, National Health Service adjusted life years; UDCA, ursodeoxycholic acid	ces; OCA, obe	ticholic acid; QA	ALYs, quality-

ERG base case

UDCA intolerant patients - OCA PAS price

	Incr	emental re	sults
	ΔQALY	ΔCosts	ICER
Company base-case (deterministic)	6.91	£148,210	£21,438
1. Fixing discrepancies between transition probabilities ¹	6.95	£148,438	£21,351
2. Use transition probabilities from POISE for the non-OCA regimen	6.56	£151,875	£23,152
3. POISE trial proportion in starting health states	6.89	£152,275	£22,111
4. Use NHS reference costs for outpatient visits	6.95	£149,461	£21,500
5. Use health state costs consistent with TA330	6.95	£166,622	£23,969
6. Use UK age-dependent utility values	5.92	£148,441	£25,085
7. Remove utility decrement	6.70	£148,441	£22,162
ERG base-case (deterministic)	5.38	£173,399	£32,217
ERG base-case (probabilistic)	5.46	£173,001	£31,682
ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic ac	S, National He	ealth Services; C	DCA,

ERG exploratory analyses

UDCA inadequate responders – OCA PAS price

	Incremental results		
	ΔQALY	ΔCosts	ICER
ERG base-case (deterministic)	4.17	£189,968	£45,541
ERG base-case (probabilistic)	4.22	£189,706	£44,945
1. Use transition probabilities and model	3.80	£221,832	£58,412
structure from TA330			
2. Using transition probabilities based on	2.59	£206,182	£79,668
the POISE trial after 12 months for the non-			
OCA treatment arms			
3. Assume that transition probabilities	3.75	£185,078	£49,294
between biomarkers health states of the			
OCA arm are >0%			
4. Use alternative costs for liver transplant	4.17	£191,025	£45,794
ICER, incremental cost-effectiveness ratio; OCA, obeticholic a ursodeoxycholic acid	cid; QALYs, q	uality-adjusted I	ife years; UDCA,

Source: ERG report PAS appendix, table 9

ERG exploratory analyses

UDCA intolerant patients – OCA PAS price

	Inc	remental res	sults
	ΔQALY	∆Costs	ICER
ERG base-case (deterministic)	5.38	£173,399	£32,217
ERG base-case (probabilistic)	5.46	£173,001	£31,682
1. Use transition probabilities and model structure	4.91	£214,417	£43,686
from TA330			
2. Using transition probabilities based on the	2.61	£202,848	£77,715
POISE trial after 12 months for the non-OCA			
treatment arms			
3. Assume that transition probabilities between	4.97	£168,979	£34,031
biomarkers health states of the OCA arm are >0%			
4. Use alternative costs for liver transplant	5.38	£174,703	£32,459
ICER, incremental cost-effectiveness ratio; OCA, obeticholic acid; C ursodeoxycholic acid	ALYs, qualit	y-adjusted life y	ears; UDCA,

Innovation

- OCA is a rationally designed FXR agonist that is the first to provide a novel, innovative therapy for patients with PBC.
- OCA has the potential to make a substantial and meaningful improvement in the quality and quantity of life for patients with PBC by providing an alternative or additional efficacious treatment option that will reduce the risk of, delay, or prevent the need for liver transplant.
- OCA offers a unique therapeutic modality, providing both hepatoprotective effects and potent and selective FXR-mediated effects, to patients who are currently at continued risk of hepatocellular carcinoma, fibrosis, cirrhosis and progression to liver transplantation or death.
- Anti-inflammatory effects of OCA may also contribute to the prevention of bile duct loss
- It would the only effective drug for patients who do not have an adequate response to, or who are intolerant to, UDCA.

Equality issues

- People with PBC face stigma in society because of the negative connotations of the term 'cirrhosis' and the association with alcoholism and drug abuse. This is one of the reasons that PBC has recently undergone a name change to primary biliary cholangitis
- PBC is a rare disease and it is essential that patients have the same opportunities to gain access to new treatments
- PBC mainly affects women, which itself presents a challenge with diagnosis since the early symptoms of PBC are often wrongly dismissed as menopausal symptoms or depression

Key issues for consideration: cost effectiveness

- Are primary outcome used in the POISE trial appropriate for modelling lifetime impact of OCA on liver outcomes?
- Do the health states in the model for low, medium and high risk PBC reflect how patients are categorised in clinical practice?
- Is it plausible to assume that patients remain in the same state while on OCA?
- Is it appropriate to apply a relative **second** reduction in utility for some liver health states for people with PBC compared with Hepatitis B/C?
- Is OCA an innovative transformative treatment for PBC in people who don't respond or an intolerant to UDCA?
- Does the committee consider structure and transition probabilities used in the company model or ERG exploratory analysis to best represent the progression of PMB?

- Use the format painter to highlight academic in confidence information <u>like this</u>
- Use the format painter to highlight commercial in confidence information like this

Confidential

Authors

- Irina Voicechovskaja
 Technical Lead
- Eleanor Donegan

Technical Adviser with input from the Lead Team (Ian Bernstein, Stephen Sharp, Pamela Rees)

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal

Obeticholic acid for treating primary biliary cirrhosis

Final scope

Remit/appraisal objective

To appraise the clinical and cost effectiveness of obeticholic acid within its marketing authorisation for treating primary biliary cirrhosis.

Background

Primary biliary cirrhosis (PBC), sometimes known as primary biliary cholangitis, is a progressive autoimmune disease that affects the liver and biliary system and destroys the small interlobular bile ducts. This prevents bile flowing from the liver to the small intestine (cholestasis) and leads to a build-up of bile in the liver cells, which damages the liver. PBC causes the formation of excess fibrous connective tissue (fibrosis) and eventually may lead to scarring of the liver (cirrhosis). The exact cause of PBC is not known, although it is thought a combination of environmental and genetic factors may play a part. The most common symptoms of PBC are itchy skin (pruritus) and fatigue, however, up to half of people with PBC do not have any symptoms until extensive liver damage occurs.

The estimated prevalence of PBC in England is approximately 18,900 people based on 35 people per 100,000 being diagnosed with PBC^{1, 2}. In 2013 there were 131 deaths from PBC in England and Wales³. Most people with PBC are aged between 30 and 65 years, and around 90% of people with the condition are women.

Treatment for PBC aims to alleviate symptoms and slow disease progression. Ursodeoxycholic acid is the only current treatment available for primary biliary cirrhosis. The estimated proportion of people whose disease have an inadequate response to ursodeoxycholic acid ranges between 20% and 70%^{1,4}. Fibrates have been used in clinical practice alone and in combination with ursodeoxycholic acid for people whose disease has an inadequate response to, or are unable to tolerate ursodeoxycholic acid. Symptomatic treatment of pruritus related to PBC includes the use of cholestyramine, rifampicin or naltrexone. Liver transplantation is the only treatment that can improve prognosis for people with PBC who have end-stage liver disease, however, the disease can recur following transplantation.

The technology

Obeticholic acid (brand name unknown, Intercept Pharmaceuticals) is a farnesoid-X receptor agonist and modified bile acid derived from the endogenous human bile acid chenodeoxycholic acid. It is administered orally.

Obeticholic acid does not currently have a marketing authorisation in the UK for primary biliary cirrhosis. It is being studied in clinical trials alone and in combination with ursodeoxycholic acid compared with placebo alone or in combination with ursodeoxycholic acid in adults whose disease had an inadequate response to ursodeoxycholic acid or who were unable to tolerate ursodeoxycholic acid.

Intervention(s)	Obeticholic acid alone or in combination with ursodeoxycholic acid	
Population(s)	People with primary biliary cirrhosis whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid.	
Comparators	For people whose disease has an inadequate response to ursodeoxycholic acid:	
	 Ursodeoxycholic acid alone or in combination with fibrates 	
	For people who are unable to tolerate ursodeoxycholic acid:	
	Fibrates	
	No additional treatment	
Outcomes	The outcome measures to be considered include:	
	mortality	
	 liver function based on markers of liver biochemistry 	
	 symptoms, including pruritus, fatigue and abdominal pain 	
	 time to liver transplantation 	
	 primary biliary cirrhosis related events, including ascites, varices, encephalopathy and hepatic cell carcinoma 	
	 adverse effects of treatment 	
	 health-related quality of life. 	

Economic analysis	The reference case stipulates that the cost effectiveness of treatments should be expressed in terms of incremental cost per quality-adjusted life year. The reference case stipulates that the time horizon for estimating clinical and cost effectiveness should be sufficiently long to reflect any differences in costs or outcomes between the technologies being compared. Costs will be considered from an NHS and Personal Social Services perspective.
Other considerations	Guidance will only be issued in accordance with the marketing authorisation. Where the wording of the therapeutic indication does not include specific treatment combinations, guidance will be issued only in the context of the evidence that has underpinned the marketing authorisation granted by the regulator.
Related NICE recommendations and NICE Pathways	Related Guidelines: Clinical Guideline in Preparation, 'Assessment and management of cirrhosis'. Earliest anticipated date of publication May 2016. Related NICE Pathways: NICE Pathway: Liver conditions, Pathway created: Mar 2014. <u>http://pathways.nice.org.uk/pathways/liver- conditions</u>
Related National Policy	NHS England commissions specialist services for Primary Biliary Cirrhosis under its policy for Liver transplantation services in adults and children. Source: <u>Manual for prescribed specialised services</u> Page 161 Department of Health (2013) <u>NHS Outcomes</u> <u>Framework 2014-2015</u> , Domains 1, 2, 4 and 5

References

- 1. UK-PBC http://www.uk-pbc.com/about/aboutpbc/. Accessed July 2015
- Office for National statistics (2013) Population Estimates for UK, England and Wales, Scotland and Northern Ireland, Mid-2013. Accessed July 2015
- 3. Office for National Statistics (2013) <u>Mortality Statistics: Deaths</u> <u>Registered in England and Wales</u>. Accessed February 2015
- 4. Expert opinion at the scoping workshop

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Matrix of consultees and commentators

Consultees	Commentators (no right to submit or		
	appeal)		
Company	General		
Lotercent Pharmaceuticals (obeticholic	Allied Health Professionals Federation		
acid)	 Board of Community Health Councils in 		
	Wales		
Patient/carer group	British National Formulary		
Addenbrookes Liver Transplant	Care Quality Commission		
Association	 Department of Health, Social Services 		
Black Health Agency	and Public Safety for Northern Ireland		
British Liver Trust	Healthcare Improvement Scotland		
Liver4Life	Medicines and Healthcare Products		
Muslim Council of Britain	Regulatory Agency		
South Asian Health Foundation	National Association of Primary Care		
Specialised Healthcare Alliance	 National Pharmacy Association 		
The PBC Foundation	NHS Alliance		
	 NHS Commercial Medicines Unit 		
Protessional groups	 NHS Confederation 		
 British Association for the Study of the Liver 	 Scottish Medicines Consortium 		
Liver	Welsh Association of Gastroenterology		
British Liver Nurses Forum	and Endoscopy (WAGE)		
 British Society of Castroontorology 	Comparator companies		
 Difficility of Gastroenterology CORE – Digestive Disorders 	AAH Pharmaceuticals (bezafibrate, aiprofibrate, comfibrazil)		
Foundation	cipronorate, germorozii)		
ESPRIT	Actavis OK (bezalibrate, Allianae Heeltheere (bezefibrete)		
 Roval College of Anaesthetists 	 Alliance Healthcare (bezanbrate, ciprofibrate, gemfibrozil) 		
 Roval College of General Practitioners 	 DE Pharmaceuticals (bezafibrate 		
 Roval College of Nursing 	ciprofibrate, gemfibrozil)		
 Royal College of Pathologists 	Dr Falk Pharma UK (ursodeoxycholic		
Royal College of Physicians	acid)		
Royal College of Surgeons	Mawdsley Brooks& Company		
Royal Pharmaceutical Society	(bezafibrate, gemfibrozil)		
Royal Society of Medicine	Mylan (bezafibrate)		
UK Clinical Pharmacy Association	Norgine (ursodeoxycholic acid)		
<u>Others</u>	Pfizer (gemfibrozil)		
 Department of Health 	Phoenix Healthcare Distribution		
NHS England	(bezafibrate, ciprofibrate, gemfibrozil)		
 NHS Herts Vallevs CCG (West) 	 Primegen (ciprofibrate) 		

National Institute for Health and Care Excellence

Matrix for the single technology appraisal of Obeticholic acid for treating primary biliary cirrhosis [ID785] Issue date: August 2016 Page 1 of 3

Сс	onsultees	Commentators (no right to submit or appeal)
•	NHS North Norfolk CCG Welsh Government	 Sandoz (bezafibrate) Sigma Pharmaceuticals (bezafibrate, ciprofibrate, gemfibrozil) Teva UK (bezafibrate, gemfibrozil) Tillomed Laboratories (gemfibrozil) Waymade Healthcare (bezafibrate, gemfibrozil) Wockhardt UK (ursodeoxycholic acid) <u>Relevant research groups</u> Cochrane Hepato-Biliary Group Foundation for Liver Research MRC Clinical Trials Unit National Institute for Health Research Public Health England Public Health Wales

NICE is committed to promoting equality, eliminating unlawful discrimination and fostering good relations between people who share a protected characteristic and those who do share it. Please let us know if we have missed any important organisations from the lists in the matrix, and which organisations we should include that have a particular focus on relevant equality issues.

PTO FOR DEFINITIONS OF CONSULTEES AND COMMENTATORS

Definitions:

Consultees

Organisations that accept an invitation to participate in the appraisal; the company that markets the technology; national professional organisations; national patient organisations; the Department of Health and the Welsh Government and relevant NHS organisations in England.

The company that markets the technology is invited to make an evidence submission, respond to consultations, nominate clinical specialists and has the right to appeal against the Final Appraisal Determination (FAD).

All non-company consultees are invited to submit a statement¹, respond to consultations, nominate clinical specialists or patient experts and have the right to appeal against the Final Appraisal Determination (FAD).

Commentators

Organisations that engage in the appraisal process but that are not asked to prepare an evidence submission or statement, are able to respond to consultations and they receive the FAD for information only, without right of appeal. These organisations are: companies that markets comparator technologies; Healthcare Improvement Scotland; other related research groups where appropriate (for example, the Medical Research Council [MRC], National Cancer Research Institute); other groups (for example, the NHS Confederation, NHS Alliance and NHS Commercial Medicines Unit, and the *British National Formulary*.

All non-company commentators are invited to nominate clinical specialists or patient experts.

National Institute for Health and Care Excellence

¹Non-company consultees are invited to submit statements relevant to the group they are representing.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single technology appraisal (STA)

Biliary cirrhosis (primary) – obeticholic acid [ID785]

Company evidence submission

26th October 2016

Contents

С	Contents2			
Lis	List of Tables4			
Lis	st of I	Figures	7	
At	brev	iations	9	
1		Executive summary	13	
	1.1	Statement of the decision problem	15	
	1.2	Description of the technology being appraised	18	
	1.3	Summary of the clinical effectiveness analysis	18	
	1.4	Summary of the cost-effectiveness analysis	20	
2		The technology	22	
	2.1	Description of the technology	22	
	2.2	Marketing authorisation/CE marking and health technology assessment	25	
	2.3	Administration and costs of the technology	27	
	2.4	Changes in service provision and management	29	
	2.5	Innovation	30	
3		Health condition and position of the technology in the treatment pathway	32	
	3.1	Disease overview	32	
	3.2	Burden to patients, carers and society	36	
	3.3	Clinical pathway of care	38	
	3.4	Life expectancy	38	
	3.5	Relevant NICE guidance, pathways or commissioning guides	39	
	3.6	Clinical guidelines	39	
	3.7	Issues relating to current clinical practice	42	
	3.8	Equality	42	
4		Clinical effectiveness	43	
	4.1	Identification and selection of relevant studies	43	
	4.2	List of relevant randomised controlled trials	46	
	4.3	Summary of methodology of the relevant randomised controlled trials	48	
	4.4	Statistical analysis and definition of study groups in POISE	60	
	4.5	Participant flow in the relevant randomised controlled trials	64	
	4.6	Quality assessment of the relevant randomised controlled trials	71	
	4.7	Clinical effectiveness results	71	
	4.8	Subgroup analysis	. 102	
	4.9	Meta-analysis	. 108	
	4.10	Indirect and mixed treatment comparisons	. 108	
	4.11	Non-randomised and non-controlled evidence	. 108	

	4.12	Adverse reactions	108
	4.13	Interpretation of clinical effectiveness and safety evidence	114
	4.14	Other ongoing studies	118
5		Cost effectiveness	121
	5.1	Published cost-effectiveness studies	121
	5.2	De novo analysis	135
	5.3	Clinical parameters and variables	139
	5.4	Measurement and valuation of health effects	149
	5.5	Cost and healthcare resource use identification, measurement and valuation	າ 158
	5.6	Summary of base-case de novo analysis inputs and assumptions	168
	5.7	Base-case results	169
	5.8	Sensitivity analyses	175
	5.9	Subgroup analysis	196
	5.10	Validation	196
	5.11	Interpretation and conclusions of economic evidence	197
6		Assessment of factors relevant to the NHS and other parties	198
	6.1	Population: people eligible for treatment	198
	6.2	Costs included	199
	6.3	Resource savings	199
	6.4	Budget impact	200
	6.5	Additional factors not included in analysis	200
	6.6	Limitations of the analysis	200
7		References	202
8		Appendices	211

List of Tables

Table 1: The decision problem	. 15
Table 2: Technology being appraised	. 18
Table 3: Incremental cost-effectiveness results: UDCA-intolerant patients	. 21
Table 4: Incremental cost-effectiveness results: UDCA inadequate responder patients	. 21
Table 5: Costs of the technology being appraised	. 27
Table 6: SF-36 PCS scores in the PBC population and the general population	. 36
Table 7: Duration of different phases of PBC	. 39
Table 8: Biochemical treatment response criteria	. 40
Table 9: Databases searched and interfaces used in the systematic reviews for evider relating to interventions for primary biliary cirrhosis [†]	nce . 44
Table 10: List of relevant Phase 3 RCTs	. 46
Table 11: Included articles relating to the Phase 2 study 747-201	. 46
Table 12: Included articles relating to the Phase 2 study 747-202	. 47
Table 13: Included articles relating to the Phase 3 study POISE	. 47
Table 14: Included articles with integrated analyses	. 48
Table 15: Key inclusion and exclusion criteria	. 52
Table 16: Biochemical treatment response criteria	. 56
Table 17: Statistical analyses	. 62
Table 18: Subject stratification [†]	. 64
Table 19: Analysis populations	. 65
Table 20: Baseline patient demographics	. 66
Table 21: Baseline PBC disease characteristics: ITT population	. 67
Table 22: Baseline liver parameters: ITT population	. 69
Table 23: Summary of primary efficacy endpoint analyses	.71
Table 24: Odds ratio of percentage reduction in ALP from baseline: ITT population	.74
Table 25: Percentage of responders and odds ratios based on biochemical treatment response criteria: ITT population	. 76
Table 26: Change in ALP from baseline at Month 6 and Month 12: ITT population	. 79
Table 27: Change in total bilirubin from baseline at Month 6 and Month 12: ITT population	. 80
Table 28: Change in conjugated bilirubin from baseline at Month 6 and Month 12: ITT population	. 81
Table 29: Change in GGT, ALT, and AST from baseline at Month 6 and 12: ITT population	. 84
Table 30: ELF score using observed data: ITT population	. 87
Table 31: Study 747-201 (double-blind phase)	. 92
Table 32: Study 747-201 (interim results from the long-term safety extension phase)	. 95

Table 33: Study 747-202 (double-blind phase)
Table 34: Study 747-202 (long-term safety extension phase)
Table 35: Demographic and baseline characteristics by OCA titration subgroups: subset of ITT population
Table 36: Results for subjects not taking UDCA 107
Table 37: Overview of adverse events: safety population (POISE)
Table 38: Summary of TEAEs, severity of AEs, and treatment-related AEs occurring in ≥5% of subjects in either OCA treatment group
Table 39: Details of the ongoing trial 747-302 (COBALT)
Table 40: Details of the ongoing trial 747-205 120
Table 41: Databases searched and interfaces used in the cost effectiveness systematic review
Table 42: Characteristics of PBC patients at start of UDCA- or non-UDCA treatment . 126
Table 43: Incidence (per 100 person-years) of major events over time among Norwegian PBC patients ($n = 182$) treated with UDCA during the first 5 years
Table 44: Demographic details of patients assessed for liver transplantation and patientswho underwent transplantation127
Table 45: Patient characteristics in Mayo and Canadian UDCA-PBC trials
Table 46: Summary list of published cost-effectiveness evaluations
Table 47: Features of the de novo analysis
Table 48: Biochemical criteria of optimal response to UDCA in PBC
Table 49: Biochemical transitions during the first year – OCA titration (POISE trial) 142
Table 50: PBC component: Transition probabilities to "abnormal bilirubin and rising, or compensated cirrhosis" and decompensated cirrhosis
Table 51: Simplified model - progression parameters (151)
Table 52: PBC components (UDCA intolerant): Transition probabilities to severe PBC and decompensated cirrhosis
Table 53: PBC component: Transition probabilities to severe PBC (OCA)
Table 54: Liver disease component: transition probabilities 147
Table 55: Mean time to event for liver-related death
Table 56: Databases searched and interfaces used in the quality of life systematic review
Table 57: Utility values extrapolated from Longworth et al (2003)
Table 58: Mean HRQoL for each disease stage as reported in Wright et al 2006 (155) 153
Table 59: SF-36 dimensions scores at baseline (N=165) (747-202 CSR)
Table 60: Included HRQoL studies
Table 61: Summary of utility values for cost-effectiveness analysis
Table 62: Databases searched and interfaces used in the quality of life systematic review

Table 63: Studies reporting resource data (original SLR and hand-searched publication	ons) 161
Table 64: Studies reporting cost data (SLR update)	165
Table 65: Unit costs associated with the technology in the economic model	166
Table 66: List of health states and associated costs in the economic model	166
Table 67: List of adverse reactions and summary of costs included in the economic model	167
Table 68: Summary of variables applied in the economic model	168
Table 69: Base-case results for UDCA-intolerant patients, using the list price of OCA	171
Table 70: Base-case results for UDCA inadequate responders, using the list price of OCA	171
Table 71 Summary of model outcomes for UDCA-intolerant population	172
Table 72: Summary of model outcomes for UDCA inadequate responder population	172
Table 73: Summary of QALY gain by health state for UDCA-intolerant population	173
Table 74: Summary of QALY gain by health state for UDCA inadequate responders	173
Table 75: Summary of costs by health state for UDCA-intolerant population	174
Table 76: Summary of costs by health state for UDCA inadequate responders	174
Table 77: Generic PSA inputs: UDCA-intolerant population	176
Table 78: Generic PSA inputs: UDCA inadequate responder population	179
Table 79: Incremental cost effectiveness results of PSA for the UDCA intolerantpopulation, using the list price of OCA	182
Table 80: Incremental cost effectiveness results of PSA for the UDCA inadequate responder population, using the list price of OCA	182
Table 81: Generic DSA inputs: UDCA-intolerant population	186
Table 82: Generic DSA inputs: UDCA inadequate responder population	189
Table 83: Parameters changed for scenario 1	194
Table 84: Scenario 1 results for UDCA intolerant patients, using the list price of OCA	195
Table 85: Scenario 1 results for UDCA inadequate responders, using the list price of OCA	195
Table 86: Parameters changed for scenario 2	195
Table 87: Scenario 2 results for UDCA intolerant patients, using the list price of OCA	195
Table 88: Scenario 2 results for UDCA inadequate responders, using the list price of OCA	195
Table 89: Estimation of number of patients eligible for treatment with OCA	199
Table 90: Budget impact of OCA	200

List of Figures

Figure 1: FXR is a nuclear receptor expressed in the liver	. 23
Figure 2: FXR regulation of biological pathways	. 24
Figure 3: The molecular structure of OCA and CDCA	. 25
Figure 4: The destructive cycle of PBC	. 33
Figure 5: Pattern of ductopenia and cholestasis in PBC	. 34
Figure 6: The progression of PBC and associated changes in biochemical markers	. 34
Figure 7: Bilirubin and ALP at 1-year follow-up	. 35
Figure 8: PRISMA diagram for the systematic review of clinical evidence	. 45
Figure 9: Study design	. 49
Figure 10: LTSE phase study design	. 50
Figure 11: CONSORT diagram for POISE trial	. 65
Figure 12: Secondary analyses: percentage of subjects achieving the primary endpoir at other time points	nt . 72
Figure 13 Sensitivity analysis of primary endpoint using observed data only	.73
Figure 14: Percentage reduction from baseline in ALP	.73
Figure 15: Percentage of responders based on Paris I ⁺ , Toronto II, Mayo II and Paris I biochemical treatment response criteria: ITT population	l . 75
Figure 16 Percentage responders at baseline to Paris I criteria at Month 6 and Month	12 . 78
Figure 17: Change in ALP from baseline over time: ITT population	. 79
Figure 18: Summary of change in total and conjugated bilirubin	. 82
Figure 19 GGT, ALT, and AST change from baseline over time	. 83
Figure 20: LS mean change from baseline in PBC-40 itch score: ITT population	. 86
Figure 21: The progression of PBC and associated changes in biochemical markers	. 90
Figure 22: ALP levels over time by treatment in the double-blind phase	. 91
Figure 23: PRISMA diagram illustrating the flow of the cost and healthcare resource u systematic review	se 123
Figure 24: Overview of model structure	138
Figure 25: Calibration process diagram	143
Figure 26: Two-stage Markov model (151)	145
Figure 27: Biochemical responders vs. non-responders (153)	147
Figure 28: PRISMA diagram illustrating the flow of the HRQoL systematic review	151
Figure 29: Mean EQ-5D scores of transplantation patients before and after transplantation, including UK population norm EQ-5D scores(141)	152
Figure 30: PRISMA diagram illustrating the flow of the cost and healthcare resource u systematic review	se 160
Figure 31: Scatter plot for PSA (UDCA-intolerant population), using the OCA list price	183

Figure 32: Cost-effectiveness acceptability curve for OCA titration versus placebo (UDCA-intolerant population), using the OCA list price
Figure 33: Scatter plot for PSA (UDCA inadequate responders population), using the OCA list price
Figure 34: Cost-effectiveness acceptability curve for OCA + UDCA titration versus UDCA + placebo (UDCA inadequate responders population), using the OCA list price
Figure 35: Tornado diagram for OCA titration versus no treatment (placebo) in the UDCA intolerant population, using the list price of OCA
Figure 36: Tornado diagram for OCA + UDCA titration versus UDCA + placebo in the UDCA inadequate responder population, using the list price of OCA

Abbreviations

AASLD	American Association for the Study of Liver Disease
AB	Abnormal bilirubin
AE	Adverse event
ALD	Alcoholic liver disease
ALP	Alkaline phosphatase
ALT	Alanine aminotransferase
AMA	Anti-mitochondrial antibody
ANCOVA	Analysis of covariance
AST	Aspartate aminotransferase
ATC	Anatomical Therapeutic Chemical
BAS	Bile acid sequestrants
ВМІ	Body mass index
BNF	British National Formulary
BP	Bodily pain
BSG	British Society of Gastroenterology
СА	Cholic acid
СС	Compensated cirrhosis
CDCA	Chenodeoxycholic acid
CETP	Cholesterol ester transfer protein
СНВ	Chronic hepatitis B
СНС	Chronic hepatitis C
СНМР	Committee for Medicinal Products for Human Use
CI	Confidence interval
CK-18	Cytokeratin-18
CLD	Chronic cholestatic liver disease
CLDQ	Chronic liver disease questionnaire
СМН	Cochran-Mantel-Haenszel
CONSORT	Consolidated Standards of Reporting Trials
CRP	C-reactive protein
CSR	Clinical study report
DARE	Database of abstracts of reviews and effects
DC	Decompensated cirrhosis
DCA	Deoxycholic acid

DCC	Decompensated cirrhosis
DEXA	Dual-emission x-ray absorptiometry
DM	Disease management
DSA	Deterministic sensitivity analysis
DSU	Decision Support Unit
EASL	European Association for the Study of Liver
ECDCA	Ethyl-chenodeoxycholic acid
eCRF	Electronic case report form
EE	Efficacy evaluable
ELF	Enhanced liver fibrosis
EMA	European Medicines Agency
EMEA	European Medicines Evaluation Agency
EOT	End of treatment
EU	European Union
FDA	Food and Drug Administration
FGF-19	Fibroblast growth factor-19
FXR	Farnesoid X receptor
GGT	Gamma-glutamyl transferase
GH	General health
GP	General practitioner
НА	Hyaluronic acid
HBV	Hepatitis B virus
HCC	Hepatocellular carcinoma
HCHS	Hospital & Community Health Services
HCV	Hepatitis C virus
HDL	High-density lipoprotein
HDLc	High-density lipoprotein cholesterol
HIV	Human immunodeficiency virus
HRQoL	Health-related quality of life
HUI	Health utility index
ICER	Incremental cost-effectiveness ratio
ICU	Intensive care unit
IL	Interleukin
INR	International standardised ration
IQR	Interquartile range

ІТТ	Intention-to-treat
IWRS	Interactive web response system
KOL	Key opinion leader
LA	Lysophosphatidic acid
LCA	Lithocholic acid
LCAT	Lecithin-cholesterol acyltransferase
LDL	Low density lipoprotein
LDLc	Low-density lipoprotein cholesterol
LDLT	Living donor liver transplantation
LLN	Lower limit of normal
LP	lipoprotein
LS	Least squares
LT	Liver transplantation
LTSE	Long-term safety extension
LYG	Life years gained
MCID	Minimal clinically important difference
MCS	Mental component summary
MELD	Model for End Stage Liver Disease
МН	Mental health
MRS	Mayo risk score
NB	Normal bilirubin
NCPE	National Centre for Pharmacoeconomics
NHS	National Health Service
OCA	Obeticholic acid
OD	Once daily
OPTN	Organ procurement and transplantation network
PAS	Patient access scheme
PASLU	Patient access schemes liaison unit
PBC	Primary biliary cholangitis/cirrhosis
PCS	Physical component summary
PF	Physical functioning
РК	Pharmacokinetics
PSA	Probabilistic sensitivity analysis
PSC	Primary sclerosing cholangitis
PT	Prothrombin time

PYE	Patient years of exposure
QALY(s)	Quality adjusted life year(s)
RCT	Randomised controlled trial
RE	Role emotional
REML	Restricted maximum likelihood
RMSE	Root mean square error
RP	Role physical
SAE	Serious adverse event
SD	Standard deviation
SE	Standard error
SF	Social functioning
SLR	Systematic literature review
SMC	Scottish Medicines Consortium
SOC	System organ class
SmPC	Summary of product characteristics
SVR	Sustained viral response
ТВ	Total bilirubin
TE	Transient elastography
TEAE	Treatment-emergent adverse event
TGF	Transforming growth factor
TIMP	Tissue inhibitor of metalloproteinase
TNF	Tumour necrosis factor
ТР	Transition probability
UDCA	Ursodeoxycholic acid
ULN	Upper limit of normal
VAS	Visual analogue scale
VLDL	Very low density lipoprotein
VT	Vitality
WADD	Weighted average daily dose

1 Executive summary

Primary biliary cholangitis (PBC), previously known as primary biliary cirrhosis, is a rare, progressive, debilitating autoimmune liver disease that follows an unpredictable course and that, if inadequately controlled, leads to complications (including ascites, varices and portal hypertension), liver transplant, and death. The course of the disease involves bile duct loss, which leads to the accumulation of toxic bile acids in the liver, resulting in inflammation, fibrosis and cirrhosis. There is currently no predictor to indicate which patients will progress slowly or rapidly, although patients with earlier age of onset and/or male sex often have more aggressive disease that is refractory to existing treatment (1).

There are considerable healthcare costs associated with PBC. In 2014/15, there were 707 hospital admissions in England for PBC, 963 consultant episodes, 3,767 bed days, and at least 45 liver transplants (2, 3). The cost of PBC increases with the stage of the disease, with a liver transplant itself reported to cost £64,452 in 2014 (4). Whilst the demand for liver transplantation is increasing, supply is stable. PBC is one of the most frequent indications for liver transplantation in Europe (5, 6), although, due to the rapid decline of PBC patients, once they qualify for inclusion on the transplant waiting list they are more likely to die before transplant than patients with other liver diseases (for example hepatitis C, alcoholic liver disease, and hepatitis B) (7).

PBC is significantly more prevalent in women, with age of diagnosis typically between 30 and 65 years (8). At the point of diagnosis patients will have lived with symptoms of the disease for a median of 5.6 years. In POISE (the pivotal Phase 3 trial for obeticholic acid [OCA]), 91% of patients were female with a mean age of 55.8 years and mean age at diagnosis of 47.3 years. These patients will have spent additional time uncontrolled on ursodeoxycholic acid (UDCA) experiencing disease progression before being eligible for the trial. The majority of patients are of working age in the UK (81% of patients in POISE were <65 years old. Therefore, PBC is also likely to have an impact on societal costs, due to lower productivity and the inability to work. In addition, patients diagnosed at a young age are more likely to have more rapidly progressing disease, which can have a devastating effect on their families.

The current management of PBC focuses on reducing biochemical markers, initially alkaline phosphatase (ALP) levels, and in the more severe stages, bilirubin, in order to minimise the risk of long-term progression (9). Currently, the only licensed treatment for PBC is UDCA, which has been shown to be effective in increasing transplant-free survival (10). However, up to 74% of patients have an inadequate response to, or are unable to tolerate, UDCA (1, 11-14), and there are currently no available treatment options for these patients. Therefore, they are at significantly increased risk of clinical complications, the requirement of liver transplant, hepatocellular carcinoma (HCC), or death (1, 11-13, 15-19).

OCA is a novel, rationally designed drug, targeting the farnesoid X receptor (FXR), which represents a step-change in the treatment of PBC. It is the first treatment to provide a novel, innovative mechanism of action for patients with PBC, and is the first drug to be developed for patients with PBC in nearly 20 years. It is used in combination with UDCA as adjunct therapy in those patients with PBC who have an incomplete response to UDCA, and where ALP is already significantly elevated, to prevent disease progression.

It is also intended to be used as monotherapy in those who are intolerant to UDCA. OCA received orphan drug designation by the EMA for the treatment of PBC in July 2010 (20). FXR activation leads to a reduction in toxic bile acids in the liver by decreasing their synthesis and facilitating their transportation from the liver (21). OCA is a once-daily tablet that has been shown to clinically and statistically significantly reduce the clinically relevant biomarkers (alkaline phosphatase [ALP] and bilirubin), the underlying immune response as indicated by TNF alpha and IgG and downstream indicators of liver damage (hepatocellular transaminases) in patients who have an inadequate response to, or who are intolerant to, UDCA. It has been shown that achieving ALP <1.67x ULN, which equates to 200 U/L, and bilirubin ≤1 mg/dL is the most discriminating algorithm for predicting clinical outcomes such as varices, ascites, death or liver transplantation (17). Therefore, POISE assessed response by looking to demonstrate clinically and statistically meaningful reductions in ALP and bilirubin:

- ALP <1.67x ULN, which equates to 200 U/L, and
- Total bilirubin <ULN, which equates to 20 µmol/L (1.17 mg/dL), and
- A 15% decrease from baseline in ALP so that patients who had only a small change from slightly above 200 U/L at baseline were excluded.

This aggressive responder criteria will mean that patients who experience a clinically relevant improvement may not be classified as a responder as their ALP does not fall below 200 U/L.

In conclusion, there are both high clinical and patient unmet needs in PBC, with no treatment options for patients who have an inadequate response to, or are intolerant to, UDCA. These patients are at increased risk of complications, the requirement of a liver transplant, HCC, and death (1, 11-13, 15-19). OCA provides a novel and innovative treatment option for these patients, showing clinically and statistically significant benefits in terms of reductions in inflammatory markers compared with placebo (with or without UDCA, depending on tolerability). Treatment with OCA should therefore prevent or slow the progression of the disease, providing benefit to the patient as well as avoiding costly downstream outcomes, e.g. ascites, and liver transplantation. OCA provides a long-awaited treatment option for PBC, and is a cost-effective use of NHS resources (see Section 1.4).

1.1 Statement of the decision problem

Table 1: The decision problem

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
Population	People with primary biliary cirrhosis [†] whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid	As per scope	Not applicable
Intervention	OCA alone or in combination with UDCA	As per scope	OCA is taken in combination with UDCA for people whose disease has an inadequate response to UDCA, and as monotherapy in people who are unable to tolerate UDCA
Comparator(s)	 For people whose disease has an inadequate response to UDCA: UDCA alone or in combination with fibrates For people who are unable to tolerate UDCA: Fibrates No additional treatment 	 For people whose disease has an inadequate response to UDCA, the following comparators were considered: UDCA For people who are unable to tolerate UDCA, the following interventions were considered: Placebo 	 Fibrates are not licensed in the UK, nor are they standard of care, and they are contraindicated in PBC (22, 23). They are rarely used, with only of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC) (24). Their efficacy is yet to be proven, with only a limited number of studies reporting results for the use of fibrates in PBC (25-29), with the following challenges: The studies were investigator-initiated and only had small patient numbers All but one study were conducted in Japanese patients In addition, there are significant

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
			safety concerns with the use of fibrates in PBC, with one study (28) reporting three deaths in 13 patients in the UDCA + fibrates arm compared with no deaths in 14 patients in the UDCA monotherapy arm. In addition, one patient developed HCC in the fibrates + UDCA arm, compared with none in the UDCA monotherapy arm (28).
Outcomes	 The outcome measures to be considered include: Mortality Liver function based on markers of liver biochemistry Symptoms, including pruritus, fatigue and abdominal pain Time to liver transplantation PBC-related events, including ascites, varices, encephalopathy and HCC Adverse effects of treatment HRQoL 	 Surrogate efficacy outcomes are included in POISE: Liver function biomarkers (ALP and bilirubin) Other biomarkers relevant to PBC (GGT, AST, ALT, FGF-19, CK-18 and bile acids) Inflammation biomarkers (CRP, TNF-α, TGF-β and IL-6) Non-invasive evaluations of fibrosis (ELF and FibroScan[®] TE) 	Due to the rare and chronic nature of PBC and the slow progression in most patients, a long-term trial is required to capture clinical outcomes such as mortality, transplant-free survival, and the incidence of complications. The primary outcome measured in POISE related to combined ALP and bilirubin levels, which have both been shown to be strongly correlated with disease prognosis (15, 17, 18). Other biomarkers relevant to PBC, inflammation biomarkers, and non- invasive evaluations of fibrosis have been included to further support changes in disease progression. There is currently a long-term Phase 3b trial ongoing, COBALT (see Section 4.14), that aims to capture clinical outcomes and should support

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
			the longer-term impact of OCA on PBC already shown in POISE.
Economic analysis	The reference case stipulates that the cost effectiveness of treatments should be expressed in terms of incremental cost per quality-adjusted life year. The reference case stipulates that the time horizon for estimating clinical and cost effectiveness should be sufficiently long to reflect any differences in costs or outcomes between the technologies being compared. Costs will be considered from an NHS and personal social services perspective.	As per scope	Not applicable
Subgroups to be considered	Not applicable	Not applicable	Not applicable
Special considerations including issues related to equity or equality	Guidance will only be issued in accordance with the marketing authorisation. Where the wording of the therapeutic indication does not include specific treatment combinations, guidance will be issued only in the context of the evidence that has underpinned the marketing authorisation granted by the regulator.	Equality document	People with PBC face stigma in society because of the negative connotations of the term 'cirrhosis' and the association with alcoholism and drug abuse (30). In addition, PBC is a rare disease affecting mainly women, and it is essential that patients have the same opportunities to gain access to new treatments.

Abbreviations: ALP, alkaline phosphatase; ALT, alanine transaminase; AST, aspartate transaminase; CK-18, cytokeratin-18; CRP, C-reactive protein; ELF, enhanced liver fibrosis; FGF-19, fibroblast growth factor-19; GGT, gamma-glutamyl transpeptidase; HRQol, health-related quality of life; HCC, hepatocellular carcinoma; IL-6, interleukin-6; NHS, National Health Service; NICE, National Institute for Health and Care Excellence; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; TE, transient elastography; TGF; transforming growth factor; TNF, tumour necrosis factor; UDCA, ursodeoxycholic acid.

[†]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cholangitis. At the time of consultation with NICE, the official name was primary biliary cirrhosis and, as such, this is reflected in this table.
1.2 Description of the technology being appraised

UK approved name and brand name	Obeticholic acid (OCALIVA®)
Marketing authorisation/CE mark status	Awaiting European marketing approval. CHMP positive opinion is expected on 14 th October 2016, with EMEA marketing authorisation expected mid-December 2016.
Indications and any	Indication:
restriction(s) as described in the summary of product characteristics	The treatment of PBC in combination with UDCA in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA.
	Contraindications:
	Complete biliary obstruction
	• Hypersensitivity to the active substance or to any of the following excipients: microcrystalline cellulose; sodium starch glycolate (Type A); magnesium stearate; poly(vinyl alcohol), partially hydrolysed (E1203); titanium dioxide (E171); macrogol 3350 (E1521); talc (E553b); iron oxide yellow (E172).
Method of administration and dosage	Provided as a film-coated tablet containing 5 mg or 10 mg OCA. The recommended starting dose is 5 mg taken orally, once daily. Based on the assessment of tolerability after 6 months, the dose should be increased to 10 mg once daily to achieve optimal response.

Table 2: Technology being appraised

Abbreviations: CHMP, Committee for Medicinal Products for Human Use; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid.

1.3 Summary of the clinical effectiveness analysis

The clinical study programme for OCA includes one pivotal Phase 3 study (POISE), which forms the main efficacy evidence for OCA, and two phase 2 studies (and extensions) providing supportive evidence.

The double-blind phase of POISE was a 12-month, international, multicentre, Phase 3, placebo-controlled study with a double-blind, randomised, parallel-group design in subjects aged ≥18 years with PBC who had previously failed treatment with UDCA or are intolerant to UDCA. The study is the largest PBC clinical trial to date, and was designed with input from the UK and global PBC group to assess the efficacy, safety, and tolerability of OCA. The primary efficacy endpoint was achieved.

Treatment with OCA resulted in clinically and statistically significant improvements from placebo as assessed by a composite ALP and total bilirubin endpoint (ALP <1.67x ULN [200 U/L], bilirubin \leq ULN [20 µmol/L] and \geq 15% decrease from baseline in ALP). ALP is a key biochemical marker for the progress of PBC, and elevated bilirubin is more indicative of end-stage disease. The reduction of both has been shown to translate into reduced risks of complications, liver transplantation, and death (17). Furthermore, as secondary endpoints, the effect of OCA on several other independent biochemical response criteria (i.e. Paris I, Paris II, Toronto II, and Mayo II), all of which are shown

to correlate with improved prognostic outcomes, were supportive of the potential effect of OCA in inhibiting the progression of PBC and improving clinical outcomes.

Other secondary endpoints, including several biochemical markers of cholestasis, inflammation, hepatobiliary injury, fibrosis, and apoptosis, demonstrated clinically and statistically significant improvements compared with placebo that were sustained during the 12-month period. These improvements in IgM, CRP, and CK-18 values, paralleled by the improvement in liver biochemistry due to FXR-mediated effects on bile acid homeostasis, are further supportive of a beneficial disease-modifying effect of FXR activation over at least a 12-month period.

Subgroup analyses demonstrated that the effect of OCA on achieving the primary composite endpoint and changes in ALP and total bilirubin were independent of age at diagnosis, duration of PBC, and baseline ALP. In general, the baseline and demographic subgroup analyses were consistent with the observed effect in the overall population, in that OCA-treated subjects had more favourable outcomes than subjects receiving placebo. For subjects receiving BAS, efficacy was modestly attenuated in subjects receiving OCA 5 mg but was not affected in subjects receiving OCA 10 mg.

Based on the clinical response and adverse event profile, initiating subjects on OCA 5 mg and titrating to 10 mg is the optimal dosing strategy. For some subjects, the composite endpoint was achieved with 5 mg OCA, however, an additional incremental benefit was gained by titrating to 10 mg OCA in those that failed to achieve an optimal response within 6 months of initiating treatment, and this is therefore the recommended dosing strategy for all patients according to the summary of product characteristics (SmPC). It is important to note that the benefit of up-titrating is not fully captured in POISE, since patients in the titration arm who reached the primary endpoint criteria did not up-titrate at 6 months and therefore did not receive any additional benefit from a higher (10 mg) dose for the second 6 months.

An ongoing long-term safety extension (LTSE) of POISE has shown continuing efficacy of OCA in terms of ALP and bilirubin levels. Evidence from two Phase 2 studies and their LTSEs provide further support for the efficacy of OCA observed in POISE.

OCA was generally well tolerated, with pruritus being the most commonly reported adverse reaction. However, it is important to consider that pruritus is a common side effect of PBC; 63% of patients in POISE had a history of pruritus. Therefore, patients and their clinicians are typically familiar with the condition and its management.

In conclusion, OCA for the treatment of adults with PBC who have had an inadequate response with UDCA or are intolerant to UDCA is an effective and a generally well tolerated treatment option. OCA (5 mg and 10 mg) resulted in clinically and statistically significant improvements in a range of evidence-based disease-related prognostic factors, which is expected to result in a significant reduction in the need for liver transplant and/or death from PBC.

1.4 Summary of the cost-effectiveness analysis

Results in this submission are based on the list price of OCA. However, OCA will be offered under a patient access scheme (PAS) and so the results presented in this submission are for guidance only. Results of the economic analyses using the PAS price are presented in the accompanying PAS template, as requested by NICE, and are more reflective of the true cost-effectiveness of OCA.

The model presented for OCA is a decision-analytic model that reports costeffectiveness in terms of incremental cost per QALY. The model evaluates the economic consequences of OCA 5-10 mg (titrated dose) alone (in the case of UDCA intolerant PBC patients) and adding OCA 5-10 mg OD to 15.4 mg/kg UDCA (in the case of UDCA inadequate responders). In other words, the model examines the cost-effectiveness of OCA titration versus placebo in UDCA intolerant patients, and OCA + UDCA titration versus UDCA monotherapy in UDCA tolerant patients.

Model inputs for health utilities are based on utility scores derived from literature. Where PBC-specific utility values were not available, HBV/HCV utility values were used instead; following key opinion leader interviews, utility values for the more severe health states (i.e. decompensated cirrhosis, pre-transplant and post-transplant) had a decrement applied to them to reflect the worse outcomes and accelerated disease progression of PBC patients compared to HBV/HCV patients.

The base case results for UDCA-intolerant patients using the list price are shown in Table 3, and for UDCA inadequate responders in Table 4. Results using the PAS are presented in the accompanying PAS template. In addition, patients treated with OCA were found to have an 84% lower chance of undergoing liver transplant compared with patients treated with UDCA.

Table 3: Incremental cost-effectiveness results: UDCA-intolerant patients

Technologies		Total			Incremental		ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment (Placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; QALYs, quality-adjusted life years; OCA, obeticholic acid.

Table 4: Incremental cost-effectiveness results: UDCA inadequate responder patients

Technologies	Total		Incremental			ICER (£)	ICER (£)	
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA + Placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.79		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; QALYs, quality-adjusted life years; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

2 The technology

Summary

- OCA is a novel, rationally designed drug targeting the FXR receptor
- It is the first treatment to provide a novel, innovative mechanism of action for patients with PBC, and is the first drug to be developed for PBC in nearly 20 years. It provides the first effective therapy for patients who have an inadequate response to UDCA who would otherwise face progression of their disease
- FXR activation leads to a reduction of up to 60% in the levels of toxic bile acids in the liver by decreasing their synthesis and facilitating their transportation from the liver
- OCA demonstrates clinically and statistically meaningful reductions in ALP and bilirubin, the main accepted markers of response in PBC
- In addition, FXR mediates other anti-inflammatory and anti-fibrotic pathways, which are expected to prevent injury to hepatocytes, improve liver function, and reduce the risk of, delay, and/or prevent the need for liver transplant in a cholestatic liver disease such as PBC
- OCA is a once-daily tablet that has been shown to clinically and statistically significantly reduce the most important biomarkers of PBC (alkaline phosphatase [ALP] and bilirubin)
- OCA received orphan drug designation by the EMA in July 2010

2.1 Description of the technology

Brand name: OCALIVA®

UK approved name: Obeticholic acid

Therapeutic class: Bile acid preparations (ATC code (A05AA04)

Mechanism of action: OCA is a novel selective and potent agonist for the farnesoid X receptor (FXR), a nuclear receptor expressed at high levels in the liver and intestine. FXR is a key regulator of bile acid, inflammatory, fibrotic, and metabolic pathways (see Figure 1 and Figure 2). OCA reduces the accumulation of bile acids in the liver and protects hepatocytes from bile acid toxicity by activating FXR. This decreases the concentration of bile acids by reducing their synthesis from cholesterol and by facilitating their transport out of hepatocytes (21). In addition, FXR mediates other anti-inflammatory and anti-fibrotic pathways (31).



Figure 1: FXR is a nuclear receptor expressed in the liver

Abbreviations: FXR, farnesoid X receptor.





Abbreviations: FXR, farnesoid X receptor.

Chenodeoxycholic acid (CDCA) is a natural bile acid that was identified as the most active physiological ligand for FXR. In structure-activity studies, it was found that the introduction of an alkyl group at the 6α position increased the activity and specificity of CDCA to FXR (32). A series of alkylated bile acid analogues were subsequently designed and studied, and it was found that the most highly potent and specific FXR agonist was OCA, which includes an ethyl group at the C6 α position (Figure 3) and has approximately 100 times the potency of CDCA.

Figure 3: The molecular structure of OCA and CDCA



Abbreviations: CDCA, chenodeoxycholic acid; EC₅₀, half-maximal effective concentration; FXR, farnesoid X receptor; OCA, obeticholic acid.

In addition to its effects on bile acids, OCA has hepatoprotective effects, including its choleretic, anti-inflammatory, and anti-fibrotic properties. These are expected to prevent injury to hepatocytes, improve liver function, and reduce, delay, and/or prevent the need for liver transplant in a cholestatic liver disease such as PBC. OCA is the first new and effective treatment for PBC in almost 20 years.

2.2 Marketing authorisation/CE marking and health technology assessment

2.2.1 Marketing authorisation/CE marking

For the indication covered in this submission:

- Regulatory submission to EMA: June 2015
- CHMP positive opinion: expected 14th October 2016
- Marketing authorisation: expected mid-December 2016

2.2.2 (Anticipated) indication(s) in the UK

OCA is anticipated to be indicated for the treatment of PBC in combination with UDCA in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA.

2.2.3 (Anticipated) restrictions or contraindications

2.2.3.1 Contraindications

The contraindications listed in the draft SmPC are:

- Complete biliary obstruction
- Hypersensitivity to the active substance(s) or to any of the following excipients: microcrystalline cellulose; sodium starch glycolate (Type A); magnesium stearate; poly(vinyl alcohol), partially hydrolysed (E1203); titanium dioxide (E171); macrogol 3350 (E1521); talc (E553b); iron oxide yellow (E172).

2.2.3.2 Special warnings and precautions

Liver-related adverse events

Elevations in alanine aminotransferase (ALT) and aspartate aminotransferase (AST) have been observed in patients taking OCA. Clinical signs and symptoms of hepatic decompensation have also been observed. These events have occurred as early as within the first month of treatment. Liver-related adverse events have primarily been observed at doses higher than the maximum recommended dose of 10 mg once daily. Patients should be monitored during treatment with OCA for elevations in liver biochemical tests and for the development of liver-related adverse reactions.

Severe pruritus

Severe pruritus was reported in 23% of patients whose starting dose was OCA 10 mg once daily, 19% of patients on OCA titration (starting dose of 5 mg with up-titration to 10 mg) and in 7% of placebo patients. The median time to onset of severe pruritus was 11 days for patients on OCA 10 mg, 158 days for patients who were titrated at 6 months, and 75 days for patients in the arm that did not receive OCA. Management strategies include the addition of bile acid sequestrants or antihistamines, dose reduction, reduced dosing frequency, and/or temporary dose interruption (see Section 2.3).

2.2.4 SmPC

The draft SmPC is provided in Appendix 1.

2.2.5 (Draft) assessment report

The draft assessment report is currently not available.

2.2.6 Main issues discussed by regulatory authorities

The draft assessment report is currently not available.

2.2.7 Anticipated date of availability in the UK

OCA will be available in the UK from December 2016 and launched in January 2017.

2.2.8 Regulatory approval outside the UK

The US Food and Drug Administration (FDA) granted accelerated approval of OCA for the treatment of PBC in combination with UDCA in adults with an inadequate response to UDCA, or as a single therapy in adults unable to tolerate UDCA on 27 May 2016.

2.2.9 Ongoing HTAs in the rest of the UK

Submission to the SMC and the NCPE are anticipated to be made in Q4, 2016.

2.3 Administration and costs of the technology

The administration and costs of the technology are summarised in Table 5.

	Information	Source
Pharmaceutical formulation	Film-coated tablet, each containing 5 mg (round tablet) or 10 mg (triangular tablet) of OCA. Yellow tablet debossed with INT on one side and 5 or 10 on the opposite side	SmPC (21)
	(representing 5 mg and 10 mg variations, respectively).	
Acquisition cost (excluding VAT) [†]	 List price: OCA 5 mg, pack of 30 tablets: £2,384.04 	
	 OCA 10 mg, pack of 30 tablets: £2,384.04 	
	 This corresponds to a price per year of £29,005.78 (with either 5 mg or 10 mg being taken once daily) 	
Method of administration	The tablet should be taken orally with or without food.	SmPC (21)
Doses	Each film-coated tablet contains either 5 mg or 10 mg of OCA.	SmPC (21)
Dosing frequency	The recommended starting dose is 5 mg OD. Based on the assessment of tolerability after 6 months, the dose should be increased to 10 mg OD to achieve optimal response.	SmPC (21)
Average length of a course of treatment	Patients should continue to take OCA for as long as the patient continues to benefit from treatment.	
Average cost of a course of treatment	Not applicable – patients are expected to take the treatment continuously	
Anticipated average interval between courses of treatments	Not applicable – the treatment is a once daily dose.	

Table 5: Costs of the technology being appraised

	Information	Source
Anticipated number of repeat courses of treatments	Not applicable – patients are expected to take the treatment continuously until they have stopped responding.	
Dose adjustments	The recommended starting dose is 5 mg OD. Based on the assessment of tolerability after 6 months, the dose should be increased to 10 mg OD to achieve optimal response. For patients experiencing severe intolerability due to pruritus, consider one of the following:	SmPC (21)
	Reducing the dosage of OCA to: 5 mg overy other day, for	
	patients intolerant to 5 mg OD	
	 5 mg OD, for patients intolerant to 10 mg OD 	
	• Temporarily interrupting OCA dosing for up to 2 weeks followed by restarting at a reduced dosage	
	 Continue to increase the dosage to 10 mg once daily, as tolerated, to achieve optimal response 	
	 Consider discontinuing treatment with OCA for patients who continue to experience persistent intolerable pruritus. 	
	For patients with hepatic impairment:	
	 The recommended starting dosage for moderate (Child-Pugh Class B) and severe (Child-Pugh Class C) hepatic impairment is 5 mg once weekly 	
	 If an adequate reduction in alkaline phosphatase and/or total bilirubin has not been achieved after 3 months of OCA 5 mg once weekly, and the patient is tolerating the medicinal product, the dose of OCA should be increased to 5 mg twice weekly (at least 3 days apart) and subsequently to 10 mg twice weekly (at least 3 days apart) depending on response and tolerability 	
	 No dose adjustment is needed for mild hepatic impairment (Child-Pugh Class A). 	
Anticipated care setting	Secondary care.	

Abbreviations: OCA, obeticholic acid; OD, once daily; SmPC, summary of product characteristics; UDCA, ursodeoxycholic acid.

2.3.1 Patient access scheme

For the purpose of this submission, Intercept have submitted an accompanying PAS template, as requested by NICE, with a proposed PAS price for OCA. As such, the results of the economic analysis presented in this submission use the list price for OCA, and are provided for illustrative purposes only to provide contextual results for the cost-effectiveness. The economic analyses using the PAS price are more reflective of the true cost-effectiveness of OCA.

2.4 Changes in service provision and management

2.4.1 Additional tests or investigations

No additional tests or investigations are required for OCA treatment.

2.4.2 Main resource use to the NHS associated with the technology

OCA treatment will be initiated by specialists in the treatment of PBC.

2.4.3 Additional infrastructure requirements

No additional infrastructure will be required to facilitate treatment with OCA.

2.4.4 Patient monitoring requirements

Patients will be required to undergo a consultation 6 months after treatment initiation to assess tolerability and determine if the dose should be increased to 10 mg to achieve optimal response.

2.4.5 Concomitant therapies

OCA is indicated for the treatment of PBC in combination with UDCA in adult patients with an inadequate response to, or intolerant to, UDCA. No dose adjustment of UDCA is required.

Concomitant medication used to treat pruritus associated with PBC is often required. Guidelines from the European Association for the Study of the Liver (33) and the American Association for the Study of Liver Diseases (34) both have the same recommendations for the management of pruritus. Treatment with cholestyramine, a bile acid sequestrant, should be initiated as first-line therapy. A dose of 4 g up to four times daily is recommended, with doses spaced at least 4 hours apart from the time of administration of UDCA. For patients that do not benefit from cholestyramine treatment or who are intolerant to therapy, rifampicin should be taken as second-line treatment, at 150–600 mg daily. As third-line treatment for patients who do not benefit from or are intolerant to rifampicin, naltrexone, an oral opiate antagonist, should be started at 25 mg daily, increasing to 50 mg. The final treatment option for pruritus is sertraline (75–100 mg daily), after which invasive physical approaches, such as extracorporeal albumin dialysis, plasmapheresis and bile duct drainage, or liver transplantation should be considered.

The draft BSG/UK-PBC guidelines (35) include similar recommendations for the management of pruritus. Cholestyramine 4–16 g per day (as tolerated) is recommended as first-line treatment taken 2–4 hours before or after UDCA;

Rifampicin 300–600 mg per day should be taken as second-line therapy, with oral naltrexone 50 mg per day or parenteral naloxone as thirdline treatment. Care should be taken with rifampicin to monitor hepatotoxicity, and naltrexone should be titrated slowly to avoid opiate withdrawal-like symptoms. Gabapentin and the selective serotonin reuptake inhibitor sertraline are also used; however, it is noted that a small trial with gabapentin failed to show benefit over placebo (37), and further evaluation is recommended.

Another common symptom of fatigue, experienced by up to 78% of patients, has been shown to improve with treatment with modafinil at a dose of 100–200 mg/day, although the EASL and AASLD do not recommend therapy for fatigue resulting from PBC (33, 34). The draft BSG/UK-PBC guidelines (35) recommending treating direct contributing factors, such as nocturnal pruritus, associated autoimmune disease and age-related conditions, modifying exacerbating processes, such as depression, autonomic dysfunction and sleep disturbance, as well as assisting with lifestyle adjustments, coping mechanisms and support. Modafinil is mentioned as a possible treatment for severe daytime somnolence.

2.5 Innovation

PBC is a rare disease with high clinical and patient unmet need with no new drugs developed in nearly 20 years. OCA is a rationally designed FXR agonist that is the first to provide a novel, innovative therapy for patients with PBC. There has been a long scientific history of in vitro and in vivo studies of the FXR axis, and OCA has finally provided an opportunity to use this mechanism of action to benefit patients with no other currently available treatment options. OCA received orphan drug designation by the EMA for the treatment of PBC in July 2010 (20). Intercept is currently pursuing a Promising Innovative Medicines (PIM) application for current and future indications for OCA.

There is a large unmet need in the area of PBC to provide a therapy for patients who have an inadequate response to or are intolerant to UDCA (up to 74% of UDCA-treated patients with PBC have a sub-optimal or absent response and are at significantly increased risk of clinical complications, the requirement of a liver transplant, and death (1, 11-14)). Patients who do not respond to or are intolerant to UDCA and show clear progression must live with the knowledge that their health is deteriorating and that their only treatment option is a liver transplant at some point in the future. Living with this knowledge, the fear of complications associated with liver transplant, and the uncertainty surrounding whether a donor liver will be available, are all likely to have a substantial detriment on mental health and quality of life. In addition, PBC progresses extremely rapidly in the latter stages of the disease, and so transplantation is sometimes not possible in time, since once patients are unwell enough to qualify for inclusion on the transplant list, it is often too late.

A recent study of a US database found that 16.5% of patients with PBC died while on the liver transplant waiting list, and 8% were removed from the list due to being too ill (7). It was also found that patients with PBC were more likely to die on the transplant waiting list than patients with other liver diseases, such as hepatitis C, alcoholic liver disease,

Company evidence submission template for obeticholic acid [ID785]

and hepatitis B (7). The organ donation and transplantation activity report for the UK states that only 51% of patients on the liver transplant list received a transplant between 1 April 2015 and 31 March 2016, while 5% died and 10% were removed from the list (38). In the economic analysis in Section 5, patients are only removed from the transplant waiting list if they die, so removing patients who are too ill is not captured in the health-related quality of life calculations. In addition, PBC patients are disadvantaged when it comes to allocating donor livers for transplant, since the current method for prioritising patients is based on the UK Model for End Stage Liver Disease (UKELD) score, which doesn't accurately predict the survival of patients with PBC (14). Even after transplant, up to 43% of patients will have a recurrence of PBC within 15 years (39) and only 8% of patients return to work (14). OCA has the potential to make a substantial and meaningful improvement in the quality and quantity of life for patients with PBC by providing an alternative or additional efficacious treatment option that will reduce the risk of, delay, or prevent the need for liver transplant. Another aspect of PBC that has a significant detrimental impact on a patient's quality of life is fatigue, which can lead to social isolation (35), since patients find it more difficult to lead a normal active life. This can affect both the patient's mental and physical health.

OCA offers a unique therapeutic modality, providing both hepatoprotective effects and potent and selective FXR-mediated effects, to patients who are currently at continued risk of hepatocellular carcinoma (HCC), fibrosis, cirrhosis and progression to liver transplantation or death. FXR activation not only reduces bile acid synthesis and promotes choleresis, but also mediates other anti-inflammatory and anti-fibrotic pathways. Recent evidence has emerged to suggest that inflammation in PBC may result in the senescence of bile duct epithelial cells, which show very slow or no division and may contribute to further injury and the irreversible loss of bile duct (40). Therefore, it is possible that the anti-inflammatory effects of OCA may also contribute to the prevention of bile duct loss.

OCA represents a step-change in the treatment of PBC. There are currently no drugs other than UDCA approved for this disease, and no drugs that are licensed or effective in this setting (for patients who do not have an adequate response to, or who are intolerant to, UDCA). Given the lack of emerging therapeutic options in the last 20 years, other treatments have been trialled for use in PBC, including budesonide and fibrates (which are contraindicated in PBC (22, 23)). However, limited efficacy has been observed in these other treatments.

OCA has been shown to be both clinically effective and to have a favourable safety and tolerability profile. In addition, it is likely that the clinical benefit of OCA is underestimated in POISE (a Phase 3 trial in OCA described in Section 4), since the trial design dictated that patients in the titration group were not up-titrated from 5 mg once daily at 6 months if they met the trial primary endpoint (ALP <1.67x ULN [equivalent to 200 U/L], total bilirubin \leq ULN [equivalent to 20 µmol/L], and ALP decrease \geq 15% from baseline). In clinical practice, the up-titration of these patients to 10 mg based on tolerability to achieve optimal response would likely produce further benefit by generating further reductions in ALP and bilirubin levels.

Health condition and position of the technology in the treatment pathway

Summary

- PBC is a rare, progressive, debilitating autoimmune liver disease that follows an unpredictable course and that, if inadequately controlled, leads to complications (including ascites, varices and portal hypertension), liver transplant, and death
 - The course of this ductopenic disease involves bile duct loss, which leads to the accumulation of toxic bile acids in the liver, resulting in inflammation, fibrosis and cirrhosis
 - Unlike inflammatory liver diseases (such as hepatitis), PBC can progress extremely rapidly in the latter stages of the disease, and so transplantation is sometimes not possible as patients may be too unwell
- There are considerable healthcare costs associated with PBC
 - In 2014/15 in England for PBC there were:
 - 707 hospital admissions
 - 963 consultant episodes
 - 3,767 bed days
 - 45 liver transplants
 - The cost of PBC increases with the stage of disease
 - Liver transplant was reported to cost £64,452 in 2014
- PBC is significantly more prevalent in women, with age of diagnosis typically between 30 and 65 years
 - In POISE (the pivotal Phase 3 trial for obeticholic acid [OCA] described in Section 4), 91% of patients were female, with a mean age of 55.8 years
 - This is of working age in the UK, and therefore PBC is likely to have an impact on societal costs
 - Patients diagnosed at a young age are more likely to have rapidly progressing disease
 - This can have a devastating effect on the wellbeing of families
- Current management of PBC focuses on reducing ALP levels
 - The only licensed treatment for PBC is ursodeoxycholic acid (UDCA)
 - However, up to 74% of patients have an incomplete response to UDCA, and there are currently no available licensed or effective treatment options for these patients
 - Patients who have an inadequate response to UDCA are at significantly increased risk of clinical complications, the requirement of liver transplant, HCC, or death

3.1 *Disease overview*

Primary biliary cholangitis (PBC), recently renamed from primary biliary cirrhosis, is a rare, progressive, autoimmune, non-viral disease of the liver that gradually destroys the interlobular bile ducts. This causes an accumulation of cytotoxic bile acids in the liver, which leads to inflammation, liver fibrosis, cirrhosis, and ultimately liver failure. Complications of PBC include portal hypertension, ascites, peripheral oedema, bleeding varices, osteoporosis, and hepatocellular carcinoma (HCC). The final stages of PBC – cirrhosis and hepatic decompensation – is terminal unless a liver transplant is performed.

3

Liver transplant is a major operation that carries a risk of rejection and potentially serious complications, and even after transplant, up to 43% of patients will have a recurrence of PBC within 15 years (41). Whilst liver failure is the usual cause of death in most patients, other causes include a bleed from the oesophagus or stomach and hepatocellular carcinoma.

The destructive cycle of PBC is initiated by an immune response that results in inflammation targeting the intrahepatic bile ducts. As the disease progresses, the liver suffers from bile duct loss (ductopenia), leading to the impairment of bile flow from the liver to the intestine, resulting in increased hepatocellular bile acid concentrations (cholestasis) (42). Bile acids are natural detergents, and abnormally elevated hepatocellular concentrations are damaging to the liver, leading to inflammation, fibrosis and liver damage, which in turn causes further bile duct loss. Eventually PBC leads to cirrhosis and hepatic decompensation.

The serologic hallmark of PBC is the presence of anti-mitochondrial antibodies (AMA), which are present in 90–95% of patients with PBC but less than 1% of healthy controls (34). The course of the disease follows a cholestatic pattern, with rising levels of alkaline phosphatase (ALP) and elevated serum bilirubin in the late stages of the disease.

The destructive cycle of PBC is shown in Figure 4.



Figure 4: The destructive cycle of PBC

Abbreviations: ALP, alkaline phosphatase; GGT, gamma-glutamyl transpeptidase; PBC, primary biliary cholangitis/cirrhosis.

On an individual basis, the course of the disease is unpredictable and its prognosis varies greatly (43), and there is currently no predictor to indicate which patients will progress slowly or rapidly, although patients with earlier age of onset and/or of male sex often have more aggressive disease that is refractory to existing treatment (1). Figure 5 shows the pattern of disease for PBC.

Company evidence submission template for obeticholic acid [ID785]



Figure 5: Pattern of ductopenia and cholestasis in PBC

Abbreviations: PBC, primary biliary cholangitis. Source: Selmi 2011 (44).

The presence of AMA and elevated ALP levels are two early characteristic markers of PBC, with concurrent hepatocellular damage (inflammation) resulting in increased levels of AST, ALT, and GGT (34, 42, 45). It is important to treat patients when their ALP levels >1.67x ULN, equivalent to 200 U/L. An increase in serum bilirubin is detected only when significant liver damage has occurred, with a sharp increase occurring in the terminal phase (45, 46). Therefore, a rise in bilirubin to as low as 0.5x ULN, is a significant change in the course of the disease (47). An overview of disease progression and the associated changes in biochemical markers is shown in Figure 6.



Figure 6: The progression of PBC and associated changes in biochemical markers

European guidelines from EASL outline the following diagnostic criteria for PBC (33):

- Biochemical evidence of cholestasis based on ALP elevation for ≥6 months; and
- Positive AMA titre (≥1:40) in serum or if AMA are absent, antibodies against AMA-M2 (anti-PDC-E2) and/or PBC-specific antibodies (anti-Sp100 and anti Gp210); or
- In the absence of PBC antibodies, liver biopsy consistent with PBC

PBC can be diagnosed providing two of the above three criteria are met, typically ALP and AMA. A liver biopsy can be used to further substantiate the diagnosis if needed.

The more recent draft BSG/UK-PBC guidelines (35) assess PBC by:

- Cholestatic liver biochemistry (elevation in serum ALP and GGT; elevation in bilirubin and fall in serum albumin are features of advanced disease and are important prognostic markers)
- AMA or other PBC-specific autoantibody at a titre of >1/40 (there is evidence to suggest that PBC-linked antinuclear antibody [ANA] may be associated with more rapidly progressive disease and disease that is less responsive to UDCA therapy)
- Diagnostic or supportive liver histology.

The presence of all three of these factors indicates definite PBC, whereas two out of the three indicates the presence of probable PBC.

In addition, serum ALP along with other biochemical parameters such as bilirubin are used to manage patients, test the efficacy of novel therapies in clinical studies, and gauge the risk of long-term adverse clinical outcomes (17, 48). Analysis from a large research group (the Global PBC study group) shows that in patients with PBC, ALP and bilirubin levels strongly correlate with death and liver transplantation, with a combination of both variables improving prognostic prediction for patients (15). The lower the ALP and bilirubin levels, the better the transplant-free survival, as shown in Figure 7.

Figure 7: Bilirubin and ALP at 1-year follow-up



Abbreviations: ALP, alkaline phosphatase; ULN, upper limit of normal. Source: Lammers et al, 2014 (15).

The estimated prevalence of PBC in the UK is approximately 3.9 per 10,000 population (46), equating to approximately 19,175 people in England (49) and making it a rare disease. The incidence of PBC is 0.58 per 10,000 population (50). Approximately 90% of people with the condition are women, and age of diagnosis is typically between 30 and

65 years (8). Whilst the cause of the disease is not known, genetic predispositions and environmental factors (exposure to chemicals) have been described (42, 51, 52).

Approximately 60–80% of patients with PBC are asymptomatic at diagnosis (53). The diagnosis of PBC in asymptomatic patients is usually established after the chance finding of an elevated ALP level during the course of an unrelated illness (43, 53).

There is an urgency to diagnose and treat patients early in the course of the disease to prevent/slow progression and to avoid/delay the latter stage complications of PBC. When treatment is delayed until PBC has progressed, survival is significantly worse than in the general population. In a Dutch prospective cohort study (13), it was found that survival for patients with early initiation of treatment (with UDCA) was comparable to the survival of the overall Dutch population (p=0.254). However, for patients with moderately advanced PBC before treatment initiation, survival was significantly worse (p<0.001). In addition, an analysis (10) combining individual data from three studies in which patients were either treated for 4 years (with UDCA), or received placebo for 2 years and UDCA for the following 2 years, showed that the probability of survival free of liver transplantation was significantly greater in the patients treated for 4 years (p<0.001; relative risk 1.92; 95% CI 1.30, 2.82).

The most common symptoms of PBC are pruritus and fatigue. The severity of cholestatic pruritus is not related to the prognosis or severity of the disease (48, 54, 55). Pruritus frequently subsides spontaneously when cirrhosis and hepatic decompensation develop, suggesting the pruritogen is synthesised in the liver (56, 57). Neither the exact pruritogen nor the underlying mechanisms for the cause of cholestatic pruritus have been completely elucidated (58). Fatigue, which can be severe, is related to autonomic dysfunction in the brainstem (59). This study also showed that fatigue was an independent predictor of mortality, particularly cardiac death.

3.2 Burden to patients, carers and society

PBC has a substantial detrimental impact on quality of life, and HRQoL impairment is correlated with the severity of the disease (60-65). In a study of 2,353 patients in the UK (54), 35% reported impairment of HRQoL compared with 6% of healthy controls (p<0.001), and 46% rated their overall health as 'fair' or 'poor' compared with 15% of healthy controls (p<0.0001).

Physical quality of life scores have been shown to be consistently lower in patients with PBC compared with the general population (64, 66-69), as measured by the SF-36, a widely used, validated, 36-item patient-reported survey of patient health (Table 6). The differences between patients with PBC and the general population were found to be statistically significant and clinically important, as measured by a minimal clinically important difference (MCID) of 5 points (70).

	PBC population		General po		
Study	Mean	SD	Mean	SD	Significance
Younossi et al (2000)	39	Not stated	50	Not stated	Not stated

Table C.							
i able 0:	31-30 PC3	Scores in t	пе гос р	opulation	and the	yenerai p	opulation

Sogolow et al (2008)	39	9.61	47	11.18	p<0.05
Montagnese et al (2010)	41	31	47	9	p<0.0001
Montagnese et al (2013)	38.61	9.61	46.79	11.18	p<0.0001
Lasker et al (2011)	40.5	10	46.79	11.18	p<0.0001

Abbreviations: PBC, primary biliary cholangitis/cirrhosis; PCS, physical component summary; SD, standard deviation.

The mental quality of life in patients with PBC was also consistently lower than the general population across trials (64, 66-69). One aspect of PBC that may contribute to the detriment in mental quality of life is the uncertainty associated with the disease. Once diagnosed, patients may remain in a fairly stable disease state for a number of years, with uncertainty as to when their health may deteriorate further and what the outcomes may be.

PBC follows an unpredictable course, and the lack of treatment options for patients who do not have an adequate response to, or are intolerant to, UDCA will result in a more rapid disease progression. Inadequate response to UDCA has been shown to be associated with an increased risk of death, need for liver transplantation and the development of HCC (1, 11-13, 15-19).

The main symptoms to have a detrimental impact on quality of life for people with PBC are chronic fatigue (54, 63, 71-76) and pruritus (76), both of which can be severe. These symptoms frequently contribute to anxiety and depression (71).

A case-control study in the US that matched 1,032 patients with PBC with 1041 controls (77) found that a significantly larger percentage of patients with PBC needed help with routine aspects of daily living than controls (13% vs 10%, respectively; p=0.008), with difficulty in performing household chores (28% vs 21%, respectively; p=0.039), having limitations in the type of professional work that they could perform (36% vs 22%, respectively; p<0.001), having to change jobs because of health (39% vs 26%, respectively; p<0.001), having limitations in the performance of housework (31% vs 19%, respectively; p<0.001) and difficulty in accomplishing everyday activities (41% vs 30%, respectively; p<0.001) being reported significantly more frequently. The same study found that patients with PBC were significantly less likely to participate in sports, physical exercise and hobbies.

PBC is also associated with considerable healthcare costs. In 2014/15, there were 707 hospital admissions in England for PBC (ICD10 K74.3), accounting for 963 consultant episodes and 3,767 bed days (2).

In most patients, the management of complications such as ascites, oesophageal varices, variceal bleeding, encephalopathy, and liver transplantation accounts for the majority of resource use in patients with PBC (78). In Norway, costs for initial hospitalisation in 2005 ranged from \in 3,059 for ascites and encephalopathy to \in 8,156 for variceal bleeding, with follow-up costs for each event of over \in 2,000 per year (79).

PBC is one of the most frequent indications for liver transplantation in Europe (5, 6). There were 621 elective liver transplants performed in the UK in 2014/2015, of which at least 7% were for PBC (3). Overall, the number of patients waiting for a liver transplant

Company evidence submission template for obeticholic acid [ID785]

has steadily increased since 2007/8 and recent data found that PBC patients on the transplant waiting list were more likely to die before transplant compared to patients with other liver diseases (7). Singh et al reported a cost of £64,452 for a liver transplant in 2014 (4).

As the majority of cases of PBC affect patients of working age, lost productivity is likely to contribute to the burden on society. The mean age of patients in the POISE trial (discussed in Section 4) is 55.8 years, which is well below the age of retirement in the UK. The population enrolled in POISE has been validated as representative of patients with PBC in the UK by clinical expert opinion.

3.3 Clinical pathway of care

Patients are commonly asymptomatic at diagnosis, and are referred to secondary care on discovering abnormal liver function and/or positive autoantibody by blood tests at a GP visit for an unrelated illness (43, 53). Occasionally patients are referred internally, most commonly from a rheumatologist. Rarely, patients are referred due to pruritus, abnormal ultrasound, or decompensation (ascites). A hepatology specialist will perform a liver screen, including tests for AMA and immunoglobulin, an ultrasound, and a FibroScan[®] to identify fibrosis. A liver biopsy can be performed to confirm diagnosis, although this varies between centres.

On diagnosis of PBC, UDCA is prescribed at 13–15 mg/kg/day. Patients are monitored at 3–4 months for tolerability, and at 6 and 12 months to gauge response and compliance to therapy. There are currently no licensed or effective therapeutic options to treat patients with PBC who have an inadequate response to, or are intolerant to, UDCA.

Patients are referred to the liver transplant list if their UKELD score ≥49 or if they have decompensated disease. UKELD score is a measure of disease severity for end-stage liver disease, and is calculated using the patient's INR, serum creatinine, serum bilirubin, and serum sodium using the following formula:

 $(5.395\times \ln(INR)) + (1.485\times \ln(creatinine)) + (3.13\times \ln(bilirubin) - (81.565\times \ln(Na)) + 435$

3.4 Life expectancy

Despite a number of studies investigating overall survival in PBC, life expectancy is difficult to determine, in part due to the intrinsic variability of the disease itself, but also due to the variability in study design and the time period over which these studies have taken place. A total of 130 deaths (113 female and 17 male) from PBC were registered in England and Wales during 2014 (80).

Life expectancy can be influenced by a number of factors, including the stage of disease at diagnosis, patient age at diagnosis, the rate of progression in the individual, and which, if any, treatments have been received (81). The estimated average time from first appearance of AMA to death is approximately 20–22 years without treatment (82, 83).

The natural history of untreated PBC is one of unpredictable progression through four phases: preclinical, asymptomatic, symptomatic (including systemic and portal hypertensive), and liver insufficiency. The duration of each phase varies but as this is a

ductopenic disease, the latter stages of the disease are non-linear and patients can experience rapid progression. Published data relating to each phase of PBC are reported in Table 7.

Disease phase	Outcome	N	Time to outcome	Reference
Pre-clinical	Time from first detection of AMA to persistently abnormal ALP	28	Median 5.6 years	Metcalf 1996 (83)
Asymptomatic	Percentage of patients developing symptoms	37 (untreated)	89% developed symptoms within 2– 4 years	Balasubramaniam 1990 (84)
		36 (untreated)	67% symptomatic at median follow-up of 12.1 years	Mahl 1994 (85)
		91 (24% UDCA treated)	36% symptomatic at a median follow-up of 5 years	Springer 1999 (86)
Symptomatic	Mean time from development of portal hypertensive symptoms to death	111 (ascites), 32 (peripheral oedema)	3.1 years	Chan 2005 (87)
Liver insufficiency	Mean time from development of	55	4.1 years (bilirubin >34 µmol/L)	Shapiro 1979 (88)
	persistently elevated bilirubin		2.1 years (bilirubin >102 μmol/L)	
			1.4 years (bilirubin >170 µmol/L)	

Table 7: Duration of different phases of PBC

Abbreviations: ALP, alkaline phosphatase; AMA, anti-mitochondrial antibodies; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

3.5 Relevant NICE guidance, pathways or commissioning guides

NICE have published guidance on the assessment and management of cirrhosis in over 16s (89). This guidance is primarily designed to address the diagnosis and management of the complications of cirrhosis, although PBC is not specifically discussed. No other relevant NICE guidance or NICE pathway is available.

There are currently no commissioning guides for the treatment of PBC.

3.6 Clinical guidelines

EASL guidelines provide current recommendations for the management of PBC in Europe (33). In addition, the British Society of Gastroenterology (BSG) and the UK-PBC have recently compiled a set of draft guidelines for PBC (35), which are now in a late stage of development and will be published shortly.

3.6.1 First-line treatment

The EASL guidelines recommend UDCA as first-line treatment for PBC and state that it is the only drug licensed to treat PBC that demonstrates any long-term efficacy. It is recommended that all patients with PBC, including those who present asymptomatically, receive UDCA at a dose of 13-15 mg/kg/day for long-term treatment, and that response to treatment should be evaluated after 1 year using either the Paris I criteria or the Barcelona criteria (see Table 8) (33).

Table 8: Biochemical treatment response criteria
--

Disease severity criteria	Responder criteria
Paris I [†] (11)	ALP ≤3x ULN and AST ≤2x ULN and total bilirubin ≤1 mg/dL (17 µmol/L)
Barcelona (16)	ALP \leq 1 x ULN or decrease in ALP >40%

Abbreviations: ALP, Alkaline phosphatase; AST, Aspartate aminotransferase; ULN, upper limit of normal.

Although not yet published, Intercept have been allowed access to the BSG/UK-PBC draft guidelines for the purpose of this appraisal. These draft guidelines also recommend UDCA for all patients at a dose of 13–15 mg/kg/day (35). This can be given as a single daily dose or divided doses if tolerability is an issue, and if tolerated, treatment should be lifelong.

There are several response criteria that have been proposed to define nonresponse/progression (some of which are described in Section 4.3.6.2); however, there is no consensus as to which of these criteria should be used. It has been shown that attaining an ALP <1.67x ULN is associated with a statistically significantly reduced risk of disease progression over the subsequent 10 years (18). In addition, in a later study it was found that combining ALP and bilirubin (ALP <1.67x ULN and bilirubin ≤1 mg/dL) was the most discriminating algorithm for predicting clinical outcomes such as varices, ascites, death, or liver transplantation (17). Therefore, the POISE study in OCA used response criteria of:

- ALP <1.67x ULN, which equates to 200 U/L, and
- Total bilirubin <ULN, which equates to 20 $\mu mol/L$ (1.17 mg/dL), and
- A 15% decrease from baseline in ALP, which was incorporated as a conservative estimate so that patients were excluded who had only a small change in ALP from slightly above 200 U/L at baseline to slightly below this level. This ensured that only subjects with a relevant clinical effect were judged to have a successful response in terms of trial endpoints.

3.6.2 Second-line treatment

There are no licensed or effective drugs approved for second-line treatment for the management of patients with an inadequate response to, or intolerance to, UDCA. Given the lack of emerging therapeutic options in the last 20 years, other treatments have been trialled for use in PBC, including budesonide and fibrates (which are contraindicated in PBC (22, 23)). However, limited efficacy has been observed in these other treatments.

The BSG/UK-PBC guidelines (35) stress that inadequate response to UDCA is associated with an increased risk of liver transplantation, development of HCC, and death (1, 11-13, 15-19) and that there are currently no licensed second-line treatments in

Company evidence submission template for obeticholic acid [ID785]

the UK. It is recommended that patients who do not respond to UDCA should participate in a clinical trial.

Further studies in second-line treatment for PBC are warranted.

3.6.3 Liver transplantation

Liver transplantation is the only treatment for patients with late-stage PBC, where UDCA has limited efficacy (33). The BSG/UK-PBC draft guidelines recommend that liver transplantation should be considered in all patients with bilirubin >50 µmol/L or evidence of decompensated liver disease. In addition, due to the variable nature of PBC, transplantation should be considered for all patients with advanced disease, as evidenced by jaundice, portal hypertension, or signs of early decompensation (e.g. ascites, encephalopathy, sarcopenia). Pruritus refractory to all medical therapy is also an indication for liver transplantation regardless of the stage of disease; however, due to limited organ availability, patients with a high UKELD score are currently prioritised, which would often not include patients with uncontrolled pruritus.

The EASL guidelines (33) recommend transplantation for patients with:

- Decompensated cirrhosis with an unacceptable quality of life, or
- Anticipated death within a year due to treatment-resistant ascites and spontaneous bacterial peritonitis, recurrent variceal bleeding, encephalopathy, or hepatocellular carcinoma

While the EASL guidelines state that liver transplant has dramatically improved survival in patients with late-stage PBC (33), it is important to note that there are risks associated with having a transplant, which can occur up to several years after the procedure. Risks associated with transplant include (90):

- Rejection of the new liver;
- Bleeding;
- The new liver not working, requiring a new transplant to be carried out immediately;
- An increased risk of infection;
- Loss of kidney function;
- Problems with blood flow to and from the liver; and
- An increased risk of certain types of cancer, particularly skin cancer.

After liver transplant it is necessary for patients to take immunosuppressants for the rest of their lives. Side effects from these include headaches, high blood pressure, tremors, an increased risk of infections, kidney failure, mood swings, muscle weakness, increased appetite and weight gain, and changes to your mental state, such as confusion, hallucinations and suicidal thoughts (91).

In addition, donated livers are scarce, and a recent study of a US database found that 16.5% of patients with PBC died while on the liver transplant waiting list, and 8% were removed from the list due to being too sick (7).

3.7 Issues relating to current clinical practice

Current clinical practice relies on the only licensed drug for PBC, UDCA. There are no licensed drugs approved for treatment of patients with an inadequate response to UDCA. Given the lack of emerging therapeutic options in the last 20 years, other treatments have been trialled for use in PBC, including budesonide and fibrates (which are contraindicated in PBC). However, limited efficacy has been observed. This leaves patients without any further treatment options if they are unable to tolerate or have an inadequate response to UDCA, which is common with PBC; up to 74% of UDCA-treated patients with PBC have a sub-optimal or absent response (1, 11-14) and are at significantly increased risk of clinical complications, the requirement of a liver transplant, HCC, or death (1, 11-13, 15-19).

3.8 Equality

People with PBC face stigma in society because of the negative connotations of the term 'cirrhosis' and the association with alcoholism and drug abuse (30). This is one of the reasons that PBC has recently undergone a name change to primary biliary cholangitis; however, this change is likely to take time to be adopted universally. In addition, PBC is also a rare disease and it is essential that patients have the same opportunities to gain access to new treatments. PBC mainly affects women, which itself presents a challenge with diagnosis since the early symptoms of PBC are often wrongly dismissed as menopausal symptoms or depression (92). This can lead to delays in treatment and corresponding worse outcomes, as well as frustration and distress to the patient.

4 Clinical effectiveness

Summary

- POISE, a 12-month randomised, double-blind, placebo-controlled Phase 3 study, is the largest PBC clinical trial to date designed with input from the UK and global PBC groups to assess the efficacy, safety and tolerability of OCA
 - Subjects who were taking UDCA before the trial (93% of subjects) continued to take UDCA throughout the trial
 - Arms were:
 - Placebo (with or without UDCA)
 - 10 mg OCA fixed dose (with or without UDCA)
 - OCA titration group (with or without UDCA)
 - Clinical expert opinion validated that the patient population in POISE is representative of the population with PBC in the UK
- Almost five times as many patients in the OCA 10 mg (47% of subjects) and OCA titration (46% of subjects) groups met the primary endpoint compared with the control arm (10% of subjects; p<0.0001) at 12 months
- Other important secondary endpoints, such as hepatocellular transaminases, markers of cholestasis, inflammation, hepatobiliary injury, fibrosis, and apoptosis, demonstrated clinically and statistically significant improvements compared with placebo that were sustained during the 12-month double-blind period. OCA demonstrates a complete biochemical improvement across the range of all the standard markers of cholestatic disease
- Subgroup analysis demonstrated that the effect of OCA on achieving the primary composite endpoint and changes in ALP and total bilirubin were independent of age at diagnosis, duration of PBC, and baseline ALP
- An ongoing long-term safety extension (LTSE) of POISE has shown continuing efficacy of OCA in controlling ALP and bilirubin levels to a degree which should halt progression of disease
- Evidence from two Phase 2 studies and their LTSEs provide further support for the efficacy of OCA observed in POISE
- OCA was generally well tolerated, with pruritus being the most commonly reported adverse
 event
 - However, it is important to consider that pruritus is a common symptom of PBC
 - 63% of patients in POISE had a history of pruritus
 - Therefore, patients and their clinicians are typically familiar with the condition and its management

4.1 Identification and selection of relevant studies

A systematic review was conducted to identify all relevant evidence for the efficacy and safety of interventions used to treat PBC. All randomised controlled trials investigating an intervention to treat PBC were included. The initial review was conducted in September 2014, followed by two updates in September 2015 and June 2016.

4.1.1 *Review question*

The PICOS (population, interventions, comparators, outcomes and study type) principle was applied to define the following review questions:

• "What is the clinical efficacy and safety of interventions used to treat primary biliary cirrhosis?"

4.1.2 Search methodology

Studies of interest were identified by searching the electronic databases shown in Table 9, with no restrictions on date of publication. Searches were conducted using the following interfaces:

- EMBASE (which covers Embase, Medline and Medline (R) In-Process)
- The Cochrane Library (which covers the Cochrane Database of Systematic Reviews, Database of Abstract of Reviews of Effects, and Cochrane Central Register of Controlled Trials).

Table 9: Databases searched and interfaces used in the systematic reviews for evidence relating to interventions for primary biliary cirrhosis[†]

Database	Interface
Embase	EMBASE
Medline	EMBASE
Medline (R) In-Process	EMBASE
Cochrane Database of Systematic Reviews	The Cochrane Library
Database of Abstracts of Reviews of Effects	The Cochrane Library
The Cochrane Central Register of Controlled Trials (CENTRAL)*	The Cochrane Library

^{†T}he term primary biliary cirrhosis was included in all searches; however, in the 2016 update search, the term 'primary biliary cholangitis' was also included due to the recent name change of the disease.

The search strategy is included in more detail in Appendix 2.

The initial systematic review identified 28 randomised controlled trials relating to interventions for primary biliary cirrhosis, three of which included OCA therapy. The first update identified a further four articles, three of which included OCA therapy; and the second update identified a further 27 articles (mostly conference abstracts), 19 of which included OCA therapy. Since the systematic literature search was performed, results from POISE have been published by Nevens et al (31). The systematic review schematic is shown in Figure 8.

Searches of the Cochrane database of systematic reviews and the database of abstracts of reviews of effects (DARE) identified 57 publications. Of these, nine were Cochrane systematic literature reviews that had previously been performed to identify RCTs for interventions in PBC. The search strategies and scopes for these reviews were closely aligned to that of this systematic review, and therefore publications were removed from

this review if they had previously been captured and reviewed by the Cochrane reviews, to avoid duplication.



Figure 8: PRISMA diagram for the systematic review of clinical evidence

4.2 List of relevant randomised controlled trials

The included articles that were relevant to OCA relate to three RCTs, one Phase 3 (POISE) (Table 10) and two Phase 2 studies (747-201 and 747-202) (described in Section 4.7.2.2). All trials had long-term safety extension phases, the results of which are summarised in Section 4.7.2.

Trial no. (acronym)	Population	Intervention	Comparator	Primary study ref(s)	Is study excluded from further discussion? If yes state rationale
747-301 (POISE); NCT014735 24	Patients diagnosed with PBC with ALP ≥1.67x ULN and/or total bilirubin >ULN but <2x ULN who fail to respond to or are intolerant to treatment with UDCA.	Oral OCA (5 mg or 10 mg) taken OD.	Placebo	Clinical study report (93)	No
LTSE to POISE	All subjects who completed the 12- month double-blind phase of POISE and who were willing to enrol in the 5-year LTSE phase of the study.	Oral OCA (5– 25 mg [†]) taken OD	N/A	Interim clinical study report (94)	No

Table 10: List of relevant Phase 3 RCTs

Abbreviations: ALP, alkaline phosphatase; LTSE, long-term safety extension; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal. [†]All subjects initiated OCA at 5 mg OD; daily dose could be up-titrated if a satisfactory response was not achieved in 5 mg increments to a total dose of 25 mg OD (one increment per 3 months permitted), depending on tolerability.

Summaries of the included publications are shown in Table 11, Table 12, Table 13 and Table 14.

Citation details	Objective
Hirschfield (2012) (95)	Presents efficacy/safety results from the open-
	label long-term extension phase
Conference poster	
Kowdley (2011) (96)	Presents efficacy/safety data of 12-week
	double-blind treatment phase
Conference abstract	
Kowdley (2011b) (97)	Presents efficacy/safety data of 12-week
	double-blind treatment phase
Conference abstract	
Kowdley (2014) (98)	Presents efficacy/safety results from the open-
	label long-term extension phase
Conference abstract	
Kowdley (2015) (99)	Presents efficacy/safety results from the open-
	label long-term extension phase
Conference abstract	

Table 11: Included articles relating to the Phase 2 study 747-201

Table	12. Included	articles	relating	to the	Phase	2 study	747-202
Iable	12. Included	articies	relating	to the	I Hase	z siuuy	1 71-202

Citation details	Objective		
Hirschfield (2011) (100)	Presents efficacy/safety results from the open-		
	label long-term extension phase		
Slide deck			
Hirschfield (2012) (95)	Presents efficacy/safety results from the open-		
	label long-term extension phase		
Conference poster			
Hirschfield (2015) (101)	Presents efficacy/safety data from the 12-week		
	double-blind treatment phase and open label		
Full publication	extension phase		
Marschall (2010) (102)	Presents efficacy/safety data of 12-week		
	double-blind treatment phase		
Conference abstract			
Mason (2010) (103)	Presents efficacy/safety data of 12-week		
	double-blind treatment phase		
Conference abstract			

Table 13: Included articles relating to the Phase 3 study POISE

Citation details	Objective
Kowdley (2014) (98)	Presents results from the open-label long-term
	extension phase
Conference abstract	
Andreone (2016) (104)	Subgroup analysis of Italian patients enrolled in POISE
Conference abstract	
Edwards (2016) (105)	Analysis of relationship between plasma total OCA exposure and changes in alkaline
Conference abstract	phosphatase and bilirubin
Malecha (2016) (106)	Analysis of baseline factors associated with pruritus
Conference abstract	
Mayne (2015) (107)	Analysis of validated methods to quantify pruritus
Conference abstract	
Mayo (2016) (108)	Analysis of strategies to mitigate pruritus
Conference abstract	Description of the state of the
Nevens (2014) (109)	double-blind treatment phase
Conference abstract	
Nevens (2016) (31)	Presents efficacy/safety data from the 1-year double-blind treatment phase and 2-year open
Full publication	label extension phase
Pares (2015) (110)	Analysis of effect of OCA on bone mineral density
Conference abstract	
Pencek (2015) (111)	Analysis to support early up-titration (<6 months) from 5 to 10 mg OCA/day
Conference abstract	
Peters (2015) (112)	Presents safety results from the open-label long-term extension phase
Conference abstract	
Peters (2016) (113)	Presents safety results from the open-label long-term extension phase
Conference abstract	

Citation details	Objective
Trauner (2015) (114)	Presents data on markers of cholestasis from the open-label long-term extension phase
Conference abstract	
Trauner (2016) (115)	Presents data on markers of cholestasis from the open-label long-term extension phase in
Conference abstract	patients treated with <10 mg OCA/day

Table 14: Included articles with integrated analyses			
Citation details	Objective		
Invernizzi (2015) (116)	Integrated analysis from the two phase II and		
	Phase III POISE RCTs in patients treated with		
Conference abstract	OCA <10 mg/day		
Jones (2015) (117)	Integrated analysis from the two phase II and		
	Phase III POISE RCTs restricted to patients		
Conference abstract	with elevated bilirubin		
Lutz (2016) (118)	Integrated analysis from the two phase II and		
	Phase III POISE RCTs according to baseline		
Conference abstract	disease severity		
Mayne (2015) (119)	Integrated analysis from the OCA 10 mg		
	treatment arm of the two phase II and Phase III		
Conference abstract	POISE RCTs according to baseline alkaline		
	phosphatase levels		
Pencek (2015b) (120)	Integrated analysis from the two phase II and		
	Phase III POISE RCTs based on age and		
Conference abstract	gender		
Pencek (2016) (121)	Integrated analysis from the two phase II and		
	Phase III POISE RCTs based on age and		
Conference abstract	gender		

ماميا مستأمام

4.3 Summary of methodology of the relevant randomised controlled trials

The clinical study programme for OCA includes two Phase 2 studies (Study 747-201 (96, 97, 122) and 747-202 (101, 123, 124)) and one pivotal Phase 3 study (Study 747-301 -POISE (93, 109)), all of which had long-term safety extensions (LTSEs). The doubleblind phases of these trials are completed, and the results from the double-blind phase of POISE are presented as the main efficacy evidence in Section 4.7.1, and of the two Phase 2 studies in Section 4.7.2 as supporting evidence. The LTSE phases of POISE and 747-201 are ongoing and the LTSE phase of 747-202 is completed. Interim/final results (as applicable) of these LTSEs are summarised as supporting evidence in Section 4.7.2.

4.3.1 Study objectives

The primary objectives of POISE were to assess the effects of OCA in subjects with PBC on serum ALP and total bilirubin (composite endpoint) and to evaluate the safety profile of OCA.

The secondary objectives were to evaluate the effects of OCA in subjects with PBC on:

 Hepatocellular injury and liver function, including histology (inflammatory, structural [portal, parenchymal], and fibrotic assessments)

Company evidence submission template for obeticholic acid [ID785]

- Disease-specific symptoms
- Biomarkers and non-invasive assessments of liver fibrosis
- Bile acids
- Other exploratory evaluations

The objectives of the LTSE phase of the study were the same as for the double-blind phase.

4.3.2 Study design

POISE was a 12-month international, multi-centre, Phase 3, randomised, double-blind, placebo-controlled, parallel group study in subjects aged ≥18 years with PBC. The study was designed to assess the efficacy, safety, and tolerability of OCA by evaluating a titration group and a fixed dose group. In the titration group, subjects initiated treatment on 5 mg OCA once daily (OD), increasing to 10 mg OCA at 6 months if they met defined criteria (see Section 4.3.2.2). Subjects in the fixed dose group received a fixed dose of 10 mg OCA OD. Subjects who were taking UDCA at the time of randomisation continued to do so at prescribed doses and frequency for the study duration, and those who were intolerant to UDCA took only study drug or placebo. All subjects were screened for eligibility during an 8-week screening period prior to entering the study to allow for collection of repeat serum chemistry samples (at least 2 weeks apart), if necessary, to confirm pre-treatment ALP and total bilirubin values.

Figure 9: Study design



Abbreviations: BL, baseline; ITT, intention-to-treat; UDCA, ursodeoxycholic acid; W, week. [†]If patients were on UDCA at screening, they were to continue on UDCA throughout the study. A small proportion of subjects in the study did not receive UDCA (7%), and therefore received OCA as monotherapy. Following completion of the 12-month double-blind phase, subjects, including those who received placebo, were given the opportunity to enrol in the open-label LTSE phase of the study. The Month 12 visit from the double-blind phase served as the Day 1 visit of the LTSE phase. To preserve blinding for the double-blind phase, subject treatment allocation in the double-blind phase was not made available until all the double-blind data were final and a 'clean' database was locked, which was more than 12 months for the first subjects who enrolled in the study.

Because subjects were unaware of previous treatment (placebo/5 mg OCA/10 mg OCA), all subjects in the LTSE started at an initial dose of 5 mg OCA. Doses could be uptitrated or down-titrated as described in Section 4.3.2.2. Subjects who received study drug (OCA or placebo) in combination with UDCA during the double-blind phase were to continue taking UDCA during the LTSE phase. However, any subject who had a clinically significant therapeutic response during the LTSE phase was able to have their UDCA withdrawn if determined clinically appropriate by the investigator. Subjects who received study drug as monotherapy in the double-blind phase were to receive OCA monotherapy in the LTSE phase.

The study design of the LTSE phase is presented in Figure 10. Subjects were contacted by study site staff 2 weeks after starting the LTSE phase to ensure compliance and record any AEs. Clinic visits for study procedures and clinical laboratory evaluations occurred every 3 months. If a subject withdrew from the study early, all of the end-of-treatment evaluations were performed at the time of withdrawal. Additionally, the subject returned for a follow-up visit 4 weeks after their final dose of OCA.



Figure 10: LTSE phase study design

Abbreviations: LTSE, long-term safety extension; M, month; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

[†] Subjects who had a clinically significant therapeutic response during the open-label LTSE phase could have their UDCA withdrawn if determined clinically appropriate by the investigator.

4.3.2.1 Randomisation and stratification

In the double-blind phase, subjects were randomised in a 1:1:1 ratio via an interactive web response system (IWRS) to the placebo group, the OCA titration group, or the OCA 10 mg fixed dose group.

Randomisation was stratified according to pre-identified criteria:

Company evidence submission template for obeticholic acid [ID785]

- a group at higher risk of developing clinical outcomes based on ALP, AST, and total bilirubin levels, i.e. Paris I criteria (ALP >3x ULN and AST >2x ULN and total bilirubin >ULN) (11); and
- the small group of subjects who could not tolerate UDCA therapy.

In addition, subjects were stratified by the presence or absence of the following biochemical response criteria and tolerance to UDCA treatment:

- ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, and intolerant to UDCA
- ALP ≤3x ULN and/or AST ≤2x ULN and/or bilirubin ≤ULN, and intolerant to UDCA
- ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, and currently taking UDCA
- ALP ≤3x ULN and/or AST ≤2x ULN and/or bilirubin ≤ULN, and currently taking UDCA

4.3.2.2 Study drugs

In the double-blind phase, subjects randomised to the fixed dose group received 10 mg OCA OD throughout the double-blind phase. Subjects randomised to the titration group received OCA 5 mg OD for the initial 6-month period. A request for titration to OCA 10 mg OD for the remainder of the double-blind phase (Months 6-12) could be made by the investigator via the IWRS for subjects who met any of the following criteria at the Month 6 assessment:

- ALP ≥1.67x ULN, and/or
- Total bilirubin >ULN, or
- <15% ALP reduction at month versus the mean pre-treatment value, and
- Provided AEs (e.g. severe pruritus) did not limit the administration of the higher dose of OCA.

Both the investigator and the subject remained unaware whether or not titration had occurred.

At the start of the LTSE phase, subjects were unaware of previous treatment (placebo/5 mg OCA/10 mg OCA) in order to preserve blinding for the double-blind phase. Therefore, all subjects in the LTSE started at an initial dose of 5 mg OCA. Doses could be up-titrated in 5 mg increments not more than once every 3 months following discussion with and approval by the sponsor, to a maximum of 25 mg, if the subject met the same criteria as for up-titration in the double-blind phase.

Based on the investigator's clinical judgement, subjects could be titrated to lower doses. As down-titration was likely to occur due to an AE, the investigator was to determine whether to gradually down-titrate in 5 mg increments or whether to down-titrate directly to the new intended dose, e.g. directly from 25 mg to 15 mg.

4.3.2.3 Selection of doses used

In the Phase 2 studies 747-201 (OCA 10 mg and 50 mg) and 747-202 (OCA 10 mg, 25 mg, and 50 mg) (see Section 4.7.2), OCA significantly reduced ALP levels in subjects with PBC, and no dose-relationship was observed (i.e. OCA 10 mg was as effective as OCA 50 mg). However, a dose-related increase in the incidence and severity of pruritus was observed across the dose range evaluated. It therefore appears that a lower dose of

Company evidence submission template for obeticholic acid [ID785]

OCA is associated with a lower incidence of pruritus without compromising effectiveness. Therefore, POISE was designed to assess the efficacy, safety, and tolerability of a lower dose of OCA (5 mg) and a 10 mg dose. This also corresponds to the proposed licensed indication, whereby patients initiate treatment with 5 mg OCA and titrate to 10 mg OCA.

4.3.2.4 Blinding

Blinding was employed in the double-blind phase. Tablets of 5 mg OCA, 10 mg OCA and placebo were visually identical, thus ensuring the double-blind nature of the study. Investigational product bottles were not labelled with either a subject randomisation number or tablet strength.

Access to randomisation codes and the corresponding treatment assignment was made available through IWRS to the appropriate Sponsor designee(s) who were responsible for reporting SAEs to the regulatory authorities. This information could be accessed only in the event of a medical emergency. No other Sponsor personnel or vendor/clinical research organisation had access to blinded subject treatment codes until all study data had been entered into the study database, validated, and the database locked.

4.3.3 Eligibility criteria

Key inclusion and exclusion criteria are shown in Table 15.

	Key inclusion criteria	Key exclusion criteria
•	Male or female aged ≥18 years Definite/probable PBC diagnosis [†] as	 History or presence of other concomitant liver disease[§]
	demonstrated by the presence of ≥2 of the following:	 Clinical complications of PBC or clinically significant hepatic decompensation[¶]
	 Elevated ALP levels for at least 6 months Positive AMA titer or if AMA negative and/or low titer (<1:80) 	 Severe pruritus or pruritus requiring systemic treatment (e.g. with BAS or rifampicin) within 2 months prior to randomisation
	PBC specific antibodies (anti- GP210 and/or anti-SP100) and/or antibodies against the major M2 components (PDC-E2, 2-oxo- glutaric acid dehydrogenase complex)	 Administration within 6 months prior to randomisation and throughout the study of azathioprine, colchicine, cyclosporine, methotrexate, mycophenolate mofetil, pentoxifylline; fenofibrate or other fibrates; budesonide and other systemic
	 Liver biopsy result consistent with PBC 	corticosteroids; potentially hepatotoxic drugs (including α-methyl-dopa, sodium valoroic acid, isopiazide, or pitrofurantoin)
•	ALP \geq 1.67x ULN and/or total bilirubin $>$ ULN but $<$ 2x ULN	 Administration within 12 months prior to
•	Taking UDCA for \geq 12 months prior to randomisation with a stable dose for \geq 3 months, or no UDCA for \geq 3 months prior to randomisation if unable to tolerate UDCA	randomisation and throughout the study of antibodies or immunotherapy directed against interleukins or other cytokines or chemokines
•	Female subjects to be post-menopausal, surgically sterile, or prepared to use ≥ 1	Previous participation in a clinical trial using OCA
	effective method of contraception during the study period and for 30 days after end of trial	 History or presence of clinically concerning cardiac arrhythmias, or prolongation of QT or QTc interval (>500 ms)
		 Pregnancy or lactating

Table 15: Key inclusion and exclusion criteria

Key inclusion criteria	Key exclusion criteria
	History of HIV infection
	 Presence of any disease or condition that interferes with the absorption, distribution, metabolism, or excretion of drugs including bile salt metabolism in the intestine. Patients with inflammatory bowel disease or who have undergone gastric bypass procedures will be excluded (gastric lap band is acceptable)
	 Medical conditions that could cause non- hepatic increases in ALP (e.g. Paget's disease) or that could diminish life expectancy to <2 years, including known cancers
	 History of alcohol[‡] or other substance abuse within 1 year prior to randomisation
	Blood or plasma donation within 30 days prior to randomisation
	 Mentally unable to complete a signed consent form

Abbreviations: AASLD, American Association for the Study of Liver Disease; ALP, alkaline phosphatase; AMA, anti-mitochondrial antibody; BAS, bile acid sequestrants; EASL, European Association for the Study of Liver; HIV, human immunodeficiency virus; MELD, Model for End Stage Liver Disease; OCA, obeticholic acid; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal. [†]Consistent with AASLD and EASL Practice Guidelines (33, 34). [§]Hepatitis C virus infection, primary sclerosing cholangitis, alcoholic liver disease, definite autoimmune liver disease, overlap hepatitis, nonalcoholic steatohepatitis, or Gilbert's syndrome. Subjects with hepatitis B virus were also excluded; however, subjects who had seroconverted could be included following consultation with the medical monitor. [¶]Includes history of liver transplantation, current placement on a liver transplant list, current MELD score ≥15 (MELD is a scoring system for assessing the severity of chronic liver disease, where the higher the score, the more severe the disease), portal hypertension with complications, cirrhosis with complications, hepatorenal syndrome (type I or II), or screening serum creatinine >2 mg/dL. [‡]Defined as consumption of more than 210 mL of alcohol per week (i.e., the equivalent of 14 4-ounce (125 mL) glasses of wine or 14 12-ounce cans/bottles of beer).

4.3.4 Location

The double-blind phase of the study was conducted in 59 sites in 13 countries in different geographical regions. Locations included the US (15 sites), Germany (10 sites), UK (9 sites), Poland (5 sites), the Netherlands (4 sites), Italy (4 sites), Australia (3 sites), Canada (2 sites), Spain (2 sites), Austria (2 sites), Belgium (1 site), France (1 site), and Sweden (1 site). The sites in the UK included 7 sites in England (London, Oxford, Newcastle-Upon-Tyne, Birmingham, Nottingham, Bristol, and Manchester) and 2 sites in Scotland (Larbert and Dundee).

The LTSE phase was conducted in the same sites as the double-blind phase of the study, with the exception of the US, where there were 14 sites in the LTSE phase compared with 15 sites in the double-blind phase.

4.3.5 Prior and concomitant therapy

Prohibited medication is listed as part of the trial exclusion criteria in Table 15.
Administration of the following medications was permitted as specified below:

- Topical or inhaled corticosteroids
- Herbal supplements or botanical preparations that are purported to affect the liver (e.g. milk thistle), provided that the dose and treatment regimen of these agents was kept constant during the double-blind phase.
- UDCA treatment for those patients who were taking UDCA at study entry
- Bile acid sequestrants (BAS) or aluminium hydroxide or smectite-containing antacids. Subjects were instructed to stagger their dosing of investigational product (and UDCA if taking) and BAS and/or antacids, ensuring at least 4 hours between doses of the BAS and/or antacids and investigational product (and UDCA if taking).
- Hormonal contraceptives

Concomitant medications were to be kept at a stable dose prior to randomisation. Investigators endeavoured to keep the doses of all concomitant medications the same during the course of the study, where medically appropriate. Subjects with other concomitant conditions that were not well controlled or whose medication needs were anticipated to change during the study were not enrolled in the study.

4.3.6 Study endpoints

4.3.6.1 Primary endpoint

The primary endpoint for the double-blind phase of the trial was the percentage of subjects achieving a composite endpoint of:

- ALP <1.67x ULN, and
- total bilirubin ≤ULN, and
- ALP decrease ≥15% from baseline

in the 10 mg OCA fixed dose group at Month 12.

Secondary analyses of the primary endpoint included the percentage of subjects achieving the primary endpoint in the OCA titration group at Month 12, the percentage of subjects reaching the endpoint at Week 2, Month 3, Month 6 and Month 9, and comparing the 10 mg OCA fixed dose group with the OCA titration group at Month 6.

The primary variable for the LTSE phase was the percentage of subjects achieving the composite endpoint based on previous treatment in the double-blind phase (i.e. 5 mg OCA, 10 mg OCA, or placebo).

Justification for the use of a composite surrogate endpoint

ALP is a key biochemical marker of PBC disease, and elevated bilirubin is indicative of end-stage disease. Both ALP and bilirubin are commonly used in an array of algorithms to predict survival and transplant outcomes in patients with PBC (11-13, 16, 125, 126). A patient-level meta-analysis study by Lammers et al demonstrated the significant prognostic value of ALP and bilirubin levels on long-term clinical outcomes in patients with PBC (15). The study demonstrated that lower levels of ALP and bilirubin strongly correlated with a longer transplant-free survival in a log linear manner. Of patients with ALP \leq 2.0x ULN, 84% survived for 10 years compared with 63% of those with levels

>2.0x ULN, and of patients with bilirubin ≤1.0x ULN, 86% survived for 10 years versus just 41% in patients with bilirubin >1.0x ULN (15). When evaluating ALP alone, attaining an ALP <1.67x ULN is associated with a statistically significantly reduced risk of disease progression over the subsequent 10 years, and is therefore a key clinical goal when evaluating the effectiveness of treatment in patients with PBC (18).

In addition, bilirubin is an independent predictor of PBC prognosis and disease progression, and was used as an endpoint to support the approval of UDCA in the EU. Importantly, changes in ALP and bilirubin occur during different stages of PBC disease progression, with ALP changes occurring during the early/asymptomatic stages whereas changes in bilirubin occur during the later/decompensated stages of the disease. Therefore, the prognostic value of these surrogate markers is significantly increased when combined and evaluated together (15). Momah et al from the Mayo clinic evaluated different biochemical thresholds versus a combination of clinical outcomes such as varices, ascites, death, or liver transplantation in a cohort of UDCA-treated patients, and concluded that combining ALP and bilirubin (ALP <1.67x ULN and bilirubin ≤1 mg/dL) was the most discriminating of the algorithms they evaluated (17).

The validity of the combined surrogate endpoint proposed by Momah et al has since been supported by a meta-analysis of patient-level data from the Global PBC study (15). The Global PBC study comprised data from 4,845 patients from 15 centres in North America and Europe, and is the largest international database of PBC patients to date. The Global PBC study demonstrated that the combined use of ALP and bilirubin levels provided greater prognostic predictability versus either component alone (15). Therefore, the use of both ALP and bilirubin for evaluating patients with PBC represents an evidence-based, clinically meaningful surrogate endpoint for use in clinical studies (15).

Consistent with the data presented above for the combined prognostic value of ALP and bilirubin, a composite endpoint of ALP <1.67x ULN (200 U/L) and bilirubin \leq ULN (20 µmol/L) was used in POISE. In addition, a minimum ALP reduction of \geq 15% from baseline was also included as part of the composite endpoint. This was incorporated as a conservative estimate so that patients were excluded who had only a small change in ALP from slightly above 200 U/L at baseline to slightly below this level. This ensured that only subjects with a relevant clinical effect were judged to have a successful response.

Due to the rarity, slow rate of progression in most patients, and chronic nature of PBC, it is a challenge to design studies based on absolute clinical long-term outcomes such as transplant-free survival and mortality. Therefore, there is a need to consider the use of surrogate endpoints to predict clinical benefit. As described above, ALP and bilirubin strongly predict the prognosis of PBC. In addition, NICE have previously accepted surrogate endpoints in the appraisals of other treatments, such as diabetes (127) and idiopathic pulmonary fibrosis (128).

4.3.6.2 Secondary endpoints for the double-blind phase of POISE

ALP response rates

Secondary efficacy endpoints relating to ALP levels were:

- Absolute and percentage change from baseline in ALP at Month 6 and Month 12
- Percentage of subjects with a decrease in ALP from baseline of ≥10%, ≥15%, ≥20%, and ≥40% at Week 2, Month 3, Month 6, Month 9 and Month 12, summarised by treatment group
- Percentage of subjects with ALP ≤ULN, summarised by treatment at all postbaseline assessments

Consistent with the log linear relationship between outcomes and ALP, the Global PBC study found that reductions in ALP were associated with increased survival with a favourable hazard ratio for all categorical percentage reductions. Statistical significance was achieved with reductions of 20–30% and >40% (15).

Biochemical treatment response criteria

The percentage of subjects meeting the Paris I, Paris II, Mayo II, Toronto II, or Rotterdam response criteria were summarised by treatment group at Week 2, Month 3, Month 6, Month 9, and Month 12. The analysis was repeated for the subgroups of subjects who met and did not meet the requirement of a responder at baseline for the endpoint analysed. The different response criteria are defined in Table 16.

Disease severity criteria	Responder criteria			
Paris I [†] (11)	ALP ≤3x ULN and AST ≤2x ULN and total bilirubin ≤ULN			
Paris II (12)	ALP ≤1.5x ULN and AST ≤1.5x ULN and total bilirubin ≤ULN			
Toronto II (18)	ALP ≤1.76x ULN			
Mayo II [†] (17)	ALP ≤1.67x ULN and total bilirubin ≤ULN			
Rotterdam (13)	Normal bilirubin (≤ULN) and normal albumin (≥LLN) after treatment when one or both parameters were abnormal before treatment			

Table 16: Biochemical treatment response criteria

Abbreviations: ALP, alkaline phosphatase; AST aspartate aminotransferase; LLN, lower limit of normal; ULN, upper limit of normal.

[†]Deviation from Paris I and Mayo II incorporates total bilirubin ≤ULN instead of ≤1 mg/dL of normal total bilirubin levels.

In addition, the number and percentage of subjects with:

- Normal bilirubin (≤ULN) and normal albumin (≥LLN),
- Moderate (bilirubin >ULN or albumin <LLN), and
- Severe (bilirubin >ULN and albumin <LLN)

were summarised by treatment group at baseline, Week 2, Month 3, Month 6, Month 9 and Month 12.

Clinical laboratory values

The absolute and percentage change from baseline in ALP, GGT, ALT, AST, total and conjugated (direct) bilirubin, albumin, and prothrombin time (PT) and international standardised ratio (INR) (both measure to determine the clotting tendency of blood) were summarised by treatment group and visit.

The majority of patients display abnormal liver function tests and typically include elevated ALP, GGT, and mild elevations of AST. During the later stages of the disease, elevations of bilirubin, PT and INR are typically observed (33, 34, 43). Changes in liver function tests relate to the stage of the disease and the severity of any underlying histological lesions (34).

Questionnaires

PBC-40

The PBC-40 is a PBC-specific, validated questionnaire that has significantly greater relevance to problems specifically associated with PBC as opposed to other frequently used health-related quality of life (HRQoL) measures (129). The absolute change from baseline in six domains (cognitive, social, emotional function, fatigue, itch, and general symptoms) were summarised by treatment using descriptive statistics at Week 2, Month 3, Month 6, Month 9, and Month 12.

Patient research questionnaire

A simple patient research questionnaire was administered at Month 12, or at termination if the subject withdrew from the study prior to this, to request feedback about the subjects' perception of the study.

Biomarkers and non-invasive assessments of liver fibrosis

The absolute change from baseline was measured for:

- Markers of hepatic fibrosis, inflammation and other disease relevant biomarkers, including CRP, tumour necrosis factor-α (TNF-α), transforming growth factor-β (TGF-β), fibroblast growth factor-19 (FGF-19), Interleukin-6 (IL-6), CK-18, autotaxin, and lysophosphatidic acid (LA), at Month 6 and Month 12
- Enhanced liver fibrosis (ELF) score and its components hyaluronic acid (HA), procollagen-3 N-terminal peptide (P3NP), and tissue inhibitor of metalloproteinase 1 (TIMP-1) at Month 6 and Month 12
- Hepatic stiffness measurements (at selected study sites) at Month 12, assessed by transient elastography (TE).

Bile acids

The absolute values and the change from baseline were summarised by treatment group for the following at Month 6 and Month 12:

- Total bile acids, total endogenous bile acids, and totals for the individual bile acids (UDCA, chenodeoxycholic acid [CDCA], deoxycholic acid [DCA], cholic acid [CA] and lithocholic acid [LCA]) and their respective conjugates
- Proportion of each of the individual bile acids relative to total bile acids

Subjects taking UDCA and subjects not taking UDCA were analysed separately. Only samples from subjects who had a confirmed fasting of approximately 8 hours or more prior to their visit were included in the analysis.

OCA PK analysis

The values at Month 6 and Month 12 for OCA (unconjugated), glyco-OCA, tauro-OCA, and total OCA were summarised by active treatment group. Only samples from subjects who had a confirmed fasting (based on the eCRF) of approximately 8 hours or more prior to their visit were included in the analysis.

The effect of BAS on OCA and total bile acid concentration, as well as the percentage change in ALP, were explored as part of the PK analysis. Relationships between plasma total OCA concentrations (unconjugated and conjugated) and FGF-19 concentrations, endogenous bile acid concentrations, ALP and liver enzyme levels, and severity of pruritus were explored, as appropriate.

Safety

Treatment-emergent adverse events (TEAEs) were defined as any adverse events (AEs) that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product. The incidence of TEAEs were recorded by treatment group.

Pruritus is considered an AE of special interest because it is both the most common symptom of PBC and the most frequently reported AE in the Phase 2 PBC studies for OCA. The 5-D pruritus questionnaire and a pruritus VAS were used at Week 2, Month 3, Month 6, Month 9 and Month 12, and results and changes from baseline were summarised at all time points.

Clinical laboratory evaluations

Physical examination, vital signs (blood pressure, heart rate and temperature), body weight and BMI, electrocardiogram, Dual-emission x-ray absorptiometry (DEXA) scan of the femoral neck and lumbar spine (to measure bone mineral density), Mayo Risk Score (MRS, a scoring system to assess survival) and MELD score (a scoring system for assessing the severity of chronic liver disease) were assessed at Week 2, Month 3, Month 6, Month 9 and Month12.

4.3.6.3 Secondary endpoints for the LTSE phase of POISE

Clinical laboratory values

The absolute and percentage change from baseline in ALP, GGT, ALT, AST, total and conjugated bilirubin, albumin, PT and INR were measured.

ALP response rates

The percentage of subjects with the following were measured:

- Decrease from baseline in ALP of \geq 15% and \geq 40%
- ALP ≤ULN
- ALP ≤2x ULN (15)
- ALP ≤1.76x ULN (18)
- ALP ≤1.67x ULN (18)

Biochemical treatment response criteria

The percentage of subjects meeting the Paris I (11), Paris II (12) and Mayo II (17) response criteria were measured. Definitions of these criteria can be found in Section 4.3.6.2.

Disease-specific symptoms

The absolute change from baseline in disease-specific symptoms were assessed using the HRQoL PBC-40 domain scores (general symptoms, itch, fatigue, cognitive function, social, emotional).

Biomarkers and non-invasive assessments of liver fibrosis

The following were measured:

- Absolute change from baseline in ELF score and its components HA, P3NP and TIMP-1
- Absolute change from baseline in hepatic stiffness (at selected study sites)
- Absolute and percentage change from baseline in other markers of hepatic fibrosis, inflammation, and other disease-relevant biomarkers, including CRP, TNF-α, TGFβ, FGF-19, IL-6, and CK-18

Antibodies

The absolute change from baseline in PBC antibodies (IgA, IgG and IgM) and other cytokines and interleukins (IL-12 and IL-23) were measured.

Pharmacokinetics and pharmacodynamics

The change from baseline in OCA and bile acids were summarised, including descriptive statistics of OCA plasma concentrations and the extent of BAS exposure. Initial evaluations of the effects of BAS on OCA, total bile acid concentrations and ALP were performed.

Company evidence submission template for obeticholic acid [ID785]

Safety

Safety endpoints included the following:

- TEAEs, serious adverse events (SAEs), severe AEs, related AEs, AEs leading to study drug withdrawal or study discontinuation, and AEs leading to death
- AEs of special interest: pruritus, hepatic disorders, dyslipidaemia, and cardiovascular disorders
- Physical examination
- Body weight and BMI
- Vital sign measurements
- 12-lead electrocardiograms
- DEXA scans
- Patient questionnaires: 5-D pruritus and Pruritus visual analogue scale (VAS)
- Clinical laboratory evaluations: haematology, serum chemistry, urine chemistry, and lipoprotein analytes
- MRS and MELD score

4.4 Statistical analysis and definition of study groups in POISE

4.4.1 Analysis populations

Intention-to-treat (ITT) population: All randomised subjects who received at least one dose of investigational product, with treatment assignment based on the randomised treatment. The ITT population was used for the summary of baseline characteristics and summaries and analyses of efficacy data in the double-blind phase.

Completer population: For the double-blind phase, the completer population consisted of all randomised subjects who received at least one dose of investigational product and participated through the duration of the double-blind phase (12 months), with treatment assignment based on the randomised treatment. For the LTSE phase, the completer population included all subjects who were exposed to OCA for at least a specified time, allowing for interim annual data cut-offs updated each year to include an additional year of completed minimum exposure time. The completer population was used to summarise specified baseline characteristics and additional analyses of efficacy data

Efficacy evaluable (EE) population: The EE population was only used in the doubleblind phase, and included all subjects in the completer population who did not have any major protocol deviations that could potentially affect the efficacy of the investigational product. The EE population was used in the double-blind phase for the summary of baseline characteristics and summaries and analyses of efficacy data, unless the exclusion of subjects from the completer population was less than 10%.

Pharmacokinetic (PK) population: All subjects who had at least one confirmed fasting analysable sample at the Month 6 or Month 12 visit, and who did not have any major protocol deviations that could potentially affect exposure levels. The PK population was used for the bile acid and OCA PK analyses.

Safety population: All subjects who received at least one dose of investigational product, with treatment assignment based on the treatment actually received. The safety population was used for the analysis of all safety data in the double-blind phase and for the analysis of all safety and efficacy data in the LTSE phase.

4.4.2 Primary hypothesis

The null hypothesis for the primary efficacy endpoint in the double-blind phase was that the response rates to the primary efficacy analysis at 12 months are equal between the placebo and OCA 10 mg fixed dose treatment groups.

4.4.3 Determination of sample size

In the double-blind phase, the planned sample size was approximately 180 subjects (60 in each group). A previous Phase 2 study found that 9% of subjects receiving placebo and 40% of subjects receiving OCA 10 mg achieved a positive response to this study's endpoint. Assuming slightly more conservative response rates of 14% and 40%, respectively, and based on the use of a two-sided test of equality of binomial proportions at the 5% level of significance, a sample size of 60 subjects per group would provide 90% power to detect a difference between the 10 mg fixed dose OCA group and the placebo group.

Due to a screening window of up to 8 weeks, some subjects had already successfully completed screening procedures, met all inclusion criteria, and had been scheduled for randomisation after the planned sample size of 180 subjects had been met. These subjects proceeded to be randomised upon approval by the Sponsor.

No formal sample size calculations were performed for the LTSE phase of the study, since all subjects who completed treatment during the double-blind phase were eligible to continue into the LTSE phase.

4.4.4 Statistical analyses

Statistical analyses of the primary and secondary outcomes in the double-blind phase, including sensitivity analyses, are summarised in Table 17.

Table 17: Statistical analyses

Endpoint	Statistical analyses
Primary endpoint	
Primary efficacy analysis	Completed using a CMH test stratified by the randomisation stratification factor. Missing values were considered a non- response. Statistical significance was declared if the p-value was ≤0.05.
Secondary analyses of the primary endpoint	The same statistical methods described for the primary endpoint were used.
Secondary endpoints	
ALP response rates	The same statistical methods described for the primary endpoint were used.
Biochemical treatment response criteria	The same statistical methods described for the primary endpoint were used for the percentage of subjects meeting the response criteria. Only descriptive statistics were provided for the subgroup of subjects who met the criteria of a responder at baseline.
Clinical laboratory values	Analyses of observed laboratory values were performed using an ANCOVA model at each visit, with absolute change from baseline as the dependent variable, including treatment group and randomisation stratification factor as fixed effects and baseline value as a covariate. The same analysis was performed using percentage change from baseline as the dependent variable. The results, as well as estimates of LS means, SE, and 95% CIs, were presented by treatment group. Estimates of the mean difference between each OCA group and placebo, the SE of the difference, and 95% CI of the difference were also presented.
Questionnaires	PBC-40 results were summarised using descriptive statistics for each domain score. The change from baseline for each domain score was analysed using the same ANCOVA model described for the analysis of clinical laboratory values. Results from the patient research questionnaire were presented as frequency counts of the responses.
Biomarkers and non-invasive assessments of liver fibrosis	Hepatic stiffness and the ELF score and its components were summarised using descriptive statistics. The absolute change from baseline was analysed using the same ANCOVA model described for the analysis of clinical laboratory values.
Bile acids	The absolute values and the change from baseline were summarised using descriptive statistics. For total UDCA and total endogenous bile acid, the change from baseline concentrations within each treatment group was compared using a paired t-test.
OCA PK analysis	Values for OCA (unconjugated), glyco-OCA, tauro-OCA, and total OCA, and the extent of BAS exposure were summarised using descriptive statistics. Initial evaluation of the effects of BAS on OCA and total bile acid concentration, as well as on ALP, was performed using correlation analysis (bile acid concentrations and ALP versus the extent of BAS exposure prior to Month 6 and Month 12 endpoints). For total bile acids, the regression analysis was performed on

Endpoint	Statistical analyses
	absolute values as well as absolute and percentage change from baseline by treatment group. For OCA, the regression analysis was performed on absolute values by active treatment group (OCA titration and OCA 10 mg fixed dose). For ALP, the regression analysis was based on the percentage change from baseline values by treatment group. For Month 12 assessments, separate summaries were provided for subjects in the OCA titration group who remained at 5 mg, and for those who titrated to 10 mg. The correlation coefficient and the estimated regression equation were included in the correlation plots.
Safety	Investigational product exposure, average daily dose, study drug compliance and TEAEs were summarised with descriptive statistics. For pruritus (AE of special interest), Kaplan-Meier (product-limit) estimates were calculated by treatment group. The quartiles, including the median time-to-event and their respective two-sided 95% CIs were presented. The OCA treatment groups compared with placebo were summarised using a stratified log-rank test controlling for the randomisation stratification factor. Kaplan-Meier estimates were plotted as a 'survival curve' for each treatment group. Results from the 5-D pruritus questionnaire were summarised using descriptive statistics. Pruritus VAS and the change from baseline VAS were summarised using descriptive statistics, and the change from baseline was also analysed using the same ANCOVA model described for the analysis for clinical laboratory values.
Clinical laboratory evaluations	Clinical laboratory evaluations were summarised using descriptive statistics.

Abbreviations: ALP, alkaline phosphatase; ALT, alanine aminotransferase; ANCOVA, analysis of covariance; AST, aspartate aminotransferase; BAS, bile acid sequestrants; CI, confidence interval; CMH, Cochran-Mantel-Haenszel; ELF, enhanced liver fibrosis; GGT, gamma-glutamyl transferase; INR, international normalised ratio; LS, least squares; OCA, obeticholic acid; PBC, primary biliary cholangitis; PT, prothrombin time; SE, standard error; TE, transient elastography; TEAE, treatment-emergent adverse event; UDCA, ursodeoxycholic acid; ULN, upper limit of normal; VAS, visual analogue scale.

4.5 *Participant flow in the relevant randomised controlled trials*

4.5.1 Patient disposition

In total, 217 subjects were randomised. Of these, 144 were randomised to receive treatment with OCA, and 73 were randomised to receive placebo. Of those randomised to OCA treatment, 71 were randomised to the titration group (with a starting dose of 5 mg OCA), and the remaining 73 were randomised to the 10 mg fixed dose OCA treatment group. In total, 216 subjects received at least one dose of study drug. Patients were then stratified by biochemical status (see Table 18). A total of 94% and 91% of randomised subjects completed 6 months and 12 months of the study, respectively. There was one subject in the OCA titration group who withdrew consent after randomisation and prior to the first dose of investigational product, leaving 70 subjects in the ITT group (Table 19).

Strata	Placebo, n (%)	OCA titration n (%)	10 mg fixed dose OCA, n (%)	Total, n (%)
Total	73 (100)	71 (100)	73 (100)	217
ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, intolerant to UDCA	3 (4)	3 (4)	3 (4)	9 (4)
ALP ≤3x ULN and/or AST ≤2x ULN and/or bilirubin ≤ULN, intolerant to UDCA	2 (3)	2 (3)	2 (3)	6 (3)
ALP >3x ULN and/or AST >2x ULN and/or bilirubin >ULN, currently taking UDCA	23 (32)	22 (31)	23 (32)	68 (31)
ALP ≤3x ULN and/or AST ≤2x ULN and/or bilirubin <uln, currently taking UDCA</uln, 	45 (62)	44 (62)	45 (62)	134 (62)

Table 18: Subject stratification[†]

Abbreviations: ALP, alkaline phosphatase; AST, aspartate aminotransferase; OCA, obeticholic acid; UDCA, ursodeoxycholic acid; ULN, upper limit of normal. [†]Columns may not add due to rounding.

A CONSORT flow diagram for POISE is presented in Figure 11 and analysis populations are presented in Table 19.

Figure 11: CONSORT diagram for POISE trial



Abbreviations: AE, adverse event; LTSE, long-term safety extension; OCA, obeticholic acid.

	Number of subjects, n (%)			
	Placebo	OCA titration	OCA 10 mg fixed dose	Total
Enrolled/randomised	73 (100)	71 (100)	73 (100)	217 (100)
ITT population§	73 (100)	70 [†] (99)	73 (100)	216 (<100)
Completer population [¶]	70 (96)	64 (90)	64 (88)	198 (91)
EE population [‡]	67 (92)	63 (89)	62 (85)	192 (88)
PK population ^{††}	0 (0)	66 (93)	60 (82)	126 (58)
Safety population§§	73 (100)	70 (99)	73 (100)	216 (<100)

Table 19: Analysis populations

Abbreviations EE, efficacy evaluable; ITT, intention-to-treat; OCA, obeticholic acid; PK, pharmacokinetics. [†]There was one subject in the OCA titration group who withdrew after randomisation, leaving 70 subjects in the ITT group; [§]All randomised subjects who received at least one dose of investigational product. Treatment assignment is based on the randomised treatment; [¶]All randomised subjects who received at least one dose of investigational product and participated through the end of the double-blind phase (12 months). Treatment assignment is based on the randomised treatment; [‡]All subjects in the completer population who did not have any major protocol deviations that could have potentially affected the efficacy of the investigational product. Treatment assignment is based on the randomised treatment; ^{††}All randomised subjects who received at least one dose of OCA who have at least one confirmed fasting sample at Month 6 and Month 12 visits (subjects must have been fasting for approximately 8 hours prior to the visit) and who did not have any major protocol deviations that could have potentially affected exposure levels; ^{§§}All subjects who received at least one dose of study drug. Treatment assignment is based on the treatment actually received.

4.5.2 Baseline characteristics and demographics

4.5.2.1 Baseline patient demographics

Key demographic and baseline characteristics are summarised in Table 20. Treatment groups were well balanced for each key demographic and baseline variable. For the overall population, mean age was 55.8 years, with a range from 29 to 86 years, and a total of 81% of subjects were <65 years of age. As expected with PBC, the study population was predominantly female (91%) and white (94%). The majority of the

Company evidence submission template for obeticholic acid [ID785]

population was European (67%), followed by North American (29%), and Australian (4%). The mean body weight and BMI were 69.8 kg and 26.0 kg/m², respectively, with 82% of subjects having a BMI <30 kg/m². Clinical expert opinion has validated that the patient population in POISE is representative of the population with PBC in the UK.

The majority (93%) of the population were taking UDCA at baseline. A small subset of patients (n=11) were unable to tolerate UDCA and received OCA as monotherapy. Efficacy results for this patient subgroup are presented in Section 4.8.4.3 but should be interpreted with caution due to low patient numbers.

	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
Age, years				
Mean (SD)	55.5 (10.0)	55.8 (10.5)	56.2 (11.0)	55.8 (10.5)
Median	55.0	54.5	56.0	55.0
Min, max	35, 78	29, 83	30, 86	29, 86
Age subgroups,	n (%)			
<65 years	60 (82)	60 (86)	56 (77)	176 (81)
≥65 years	13 (18)	10 (14)	17 (23)	40 (19)
Gender, n (%)				
Male	5 (7)	5 (7)	10 (14)	20 (9)
Female	68 (93)	65 (93)	63 (86)	196 (91)
Race/ethnicity, n	(%)			
White	66 (90)	67 (96)	70 (96)	203 (94)
Non-white	7 (10)	3 (4)	3 (4)	13 (6)
Body weight, kg				
Mean (SD)	70.2 (13.3)	68.2 (13.1)	71.0 (15.3)	69.8 (13.9)
Median	70.5	65.2	67.6	67.5
Min, max	41.0, 106.0	46.7, 101.8	50.8, 134.0	41.0, 134.0
Region, n (%)				
Europe	49 (67)	45 (64)	51 (70)	145 (67)
North America	21 (29)	20 (29)	21 (29)	62 (29)
Australia	3 (4)	5 (7)	1 (1)	9 (4)
BMI, kg/m²				
Mean (SD)	26.2 (4.4)	25.8 (4.9)	26.3 (5.1)	26.0 (4.8)
Median	25.9	24.5	25.1	25.0
Min, max	16.4, 37.6	17.7, 40.7	20.4, 49.2	16.4, 49.2
BMI subgroups, I	n (%)			
<30 kg/m ²	58 (79)	58 (83)	61 (84)	177 (82)
≥30 kg/m²	15 (21)	11 (16)	12 (16)	38 (18)

Table 20: Baseline patient demographi

Company evidence submission template for obeticholic acid [ID785]

	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
Pre-treatment live	er biopsy, n (%)			
Yes	7 (10)	13 (19)	9 (12)	29 (13)
No	66 (90)	57 (81)	64 (88)	187 (87)
UDCA use at baseline, n (%)				
Yes	68 (93)	65 (93)	67 (92)	200 (93)
No	5 (7)	5 (7)	6 (8)	16 (7)

Abbreviations: BMI, body mass index; OCA, obeticholic acid; SD, standard deviation; UDCA, ursodeoxycholic acid.

4.5.2.2 Baseline disease characteristics

PBC baseline disease characteristics are summarised in Table 21. In general, each variable was well balanced across treatment groups. Overall, the mean age at time of diagnosis was 47.3 years with a mean duration of PBC of 8.6 years, and there was a comparable percentage of subjects with a duration of PBC \leq 7.5 years versus >7.5 years. There were slightly more subjects <50 years of age at PBC diagnosis (58%) compared with \geq 50 years of age.

Patients were comparable between groups based on a history of pruritus, but the overall incidence of pruritus at baseline was slightly higher for subjects in the placebo treatment group (64%) and OCA 10 mg fixed dose group (60%) than in the OCA titration group (53%).Overall severity of pruritus at baseline was assessed by the investigator as mild (43%), moderate (15%), or severe (1%).

A total of 128 (59%) subjects reported a history of fatigue prior to entering the study, with the most recent fatigue events being mild for 34% of patients, moderate for 19% of patients, and severe for 5% of patients. The overall incidence of fatigue was slightly higher for subjects in the placebo treatment group (67%) than in the OCA titration and OCA 10 mg fixed dose groups (54% and 56%, respectively).

Disease characteristic	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
History of pruritu	s, n (%)			
Yes	47 (64)	45 (64)	45 (62)	137 (63)
No	26 (36)	25 (36)	28 (38)	79 (37)
Severity of most recent pruritus event for subjects who had history of pruritus, n (%)				
Mild	31 (66)	29 (64)	34 (76)	94 (69)
Moderate	14 (30)	13 (29)	8 (18)	35 (26)
Severe	1 (2)	2 (4)	3 (7)	6 (4)
Unknown	1 (2)	1 (2)	0 (0)	2 (1)

Table 21: Baseline PBC disease characteristics: ITT population

Disease characteristic	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
Pruritus at baseli	ne, n (%)			
Yes	47 (64)	37 (53)	44 (60)	128 (59)
Mild	32 (44)	27 (39)	33 (45)	92 (43)
Moderate	13 (18)	10 (14)	10 (14)	33 (15)
Severe	2 (3)	0 (0)	1 (1)	3 (1)
No	26 (36)	33 (47)	29 (40)	88 (41)
History of fatigue	, n (%)			
Yes	49 (67)	38 (54)	41 (56)	128 (59)
No	24 (33)	32 (46)	32 (44)	88 (41)
Overall severity of	of PBC-related fation	jue, n (%)		
Mild	28 (38)	17 (24)	29 (40)	74 (34)
Moderate	16 (22)	16 (23)	8 (11)	40 (19)
Severe	3 (4)	5 (7)	3 (4)	11 (5)
Age at PBC diagr	nosis, years			
Mean (SD)	47.3 (9.3)	47.6 (11.7)	47.1 (10.6)	47.3 (10.5)
Median	48.0	48.0	47.0	47.5
Min, Max	31, 74	25, 82	24, 78	24, 82
Age at PBC diagr	nosis subgroups, n	ı (%)		
<50 years	45 (62)	38 (54)	42 (58)	125 (58)
≥50 years	28 (38)	32 (46)	31 (42)	91 (42)
Mean duration of	PBC, years			
Mean (SD)	8.3 (5.4)	8.3 (5.8)	9.2 (6.9)	8.6 (6.0)
Median	7.4	7.2	8.5	7.8
Min, max	0.9, 21.8	0.3, 27.0	0.0, 32.3	0.0, 32.3
Duration of PBC	subgroups, n (%)			
≤7.5 years	39 (53)	36 (51)	30 (41)	105 (49)
>7.5 years	34 (47)	34 (49)	43 (59)	111 (51)

Abbreviations: OCA, obeticholic acid; ITT, intention-to-treat; PBC, primary biliary cholangitis; SD, standard deviation.

4.5.2.3 Baseline liver laboratory parameters

Key baseline laboratory variables indicative of intrahepatic cholestasis, hepatocellular injury, and synthetic hepatic function are summarised in Table 22. In general, mean

baseline values were well balanced across treatment groups. These baseline data were consistent with the protocol inclusion criterion for subjects with PBC with either ALP \geq 1.67x ULN^a or total bilirubin >ULN but <2x ULN, and the exclusion of subjects with clinically significant hepatic decompensation. Mean baseline albumin values were predominantly (81– 89%) \geq LLN. In addition, the majority (94–99%) of subjects had a baseline INR \leq 1.3, indicative of a population in an early stage of disease progression. Conjugated bilirubin is widely accepted as an early sign of hepatobiliary damage. The mean conjugated bilirubin levels at baseline were approximately 1.5–2.0x ULN, indicating evidence of hepatic dysfunction in the study population.

Mean baseline ALP values were well balanced across treatment groups, with 29% of subjects having a baseline ALP >3x ULN. Mean baseline total bilirubin values ranged from 10.3 µmol/L to 11.8 µmol/L across treatment groups, with 92% of subjects within normal range. Therefore, the majority of subjects met the inclusion criterion of ALP \geq 1.67x ULN rather than the total bilirubin criterion >ULN but <2x ULN.

As expected in this patient population with intrahepatic cholestasis, GGT was substantially elevated across all three treatment groups (approximately 9–12x ULN). Mean baseline GGT levels were slightly higher in the placebo group than in the OCA groups; however, it should be noted the placebo group had a larger degree of variability than the OCA groups.

Hepatocellular transaminases (ALT and AST) were also elevated across all treatment groups, albeit to a much smaller magnitude (approximately 2x ULN) than GGT. This was not unexpected, given that PBC is often associated with mild and less consistent elevations of ALT and AST, which are typically more indicative of liver damage rather than intrahepatic cholestasis (34).

	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)
ALP, U/L			
Mean (SD)	327.5 (115.0)	325.9 (116.2)	316.3 (103.9)
Min, max	143.8, 745.9	186.8, 811.0	207.1, 619.5
≤3x ULN, n (%)	50 (68)	51 (73)	53 (73)
>3x ULN, n (%)	23 (32)	19 (27)	20 (27)

Table 22: Baseline liver parameters: ITT population

^a ALP ULN=118.3 U/L (female) and 124.2 U/L (male); total bilirubin ULN=19.32 µmol/L (female) and 25.48 µmol/L (male).

	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)			
Total bilirubin, µmol/L						
Mean (SD)	11.8 (7.4)	10.3 (5.5)	11.3 (6.7)			
Min, max	2.3, 39.3	2.1, 36.4	1.6, 34.4			
≤ULN, n (%)	66 (90)	66 (94)	66 (90)			
>ULN, n (%)	7 (10)	4 (6)	7 (10)			
Conjugated bilirubin,	umol/L	· · ·				
Mean (SD)	5.5 (6.0)	4.5 (4.5)	4.9 (4.4)			
Min, max	1.5, 31.3	1.5, 35.2	1.5, 23.9			
≤ULN, n (%)	37 (51)	37 (53)	35 (48)			
>ULN, n (%)	36 (49)	33 (47)	38 (52)			
GGT, U/L		· · ·				
Mean (SD)	309.6 (449.4)	252.8 (167.0)	261.1 (207.4)			
Min, max	45.2, 3263.5	43.1, 727.8	48.2, 1084.0			
ALT, U/L						
Mean (SD)	56.0 (30.3)	61.6 (39.0)	56.3 (39.7)			
Min, max	17.1, 174.9	17.1, 245.5	16.3, 240.2			
AST, U/L						
Mean (SD)	48.8 (22.4)	52.3 (25.3)	50.5 (31.1)			
Min, max	20.6, 156.5	20.1, 173.4	23.3, 186.5			
Albumin, g/L						
Mean (SD)	42.8 (3.1)	43.0 (3.1)	43.7 (2.7)			
Min, max	32.5, 48.0	33.0, 50.5	36.5, 49.5			
<lln, (%)<="" n="" td=""><td>14 (19)</td><td>8 (11)</td><td>8 (11)</td></lln,>	14 (19)	8 (11)	8 (11)			
≥LLN, n (%)	59 (81)	62 (89)	65 (89)			
INR [†]						
Mean (SD)	0.98 (0.1)	1.07 (0.4)	1.02 (0.3)			
Min, max	0.7, 1.2	0.9, 3.5	0.8, 3.5			
≤1.3	72 (99)	66 (94)	69 (95)			
>1.3	0 (0)	3 (4)	2 (3)			

Abbreviations: ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; GGT, gamma-glutamyl transferase; INR, international standardised ratio; ITT, intention-to-treat; LLN, lower limit of normal; OCA, obeticholic acid; SD, standard deviation; ULN, upper limit of normal. [†]Placebo n=72, OCA titration n=69, and OCA 10 mg fixed dose n=71.

4.6 Quality assessment of the relevant randomised controlled trials

A complete quality assessment for POISE is provided in Appendix 3.

4.7 Clinical effectiveness results

4.7.1 *POISE*

4.7.1.1 Primary efficacy endpoint at 12 months

The primary efficacy endpoint (superiority of 10 mg OCA fixed dose vs placebo, measured by the percentage of subjects with ALP <1.67x ULN <u>and</u> total bilirubin \leq ULN <u>and</u> ALP decrease \geq 15% from baseline at 12 months) was met (see Table 23 and Figure 12). Almost five times as many patients (47% of subjects) in the OCA 10 mg fixed dose group achieved the primary endpoint at 12 months than in the placebo group (10% of subjects; p<0.0001). The primary efficacy analysis was repeated using the EE population and Completer populations, which were numerically comparable with the ITT population and statistically significant compared with placebo (p<0.0001).

The key secondary analysis of the primary endpoint demonstrated that 46% of subjects in the OCA titration group also achieved the primary endpoint after 12 months of OCA treatment (p<0.0001 vs placebo) (Figure 12). It is worth noting that those subjects who failed to respond to treatment with 5 mg OCA at Month 6 in the titration group had their dose increased to 10 mg OCA until Month 12. As such, there was an increase in the percentage of responders in those who titrated when comparing Month 6 data with Month 12 data. The increase in responders in this subgroup of patients is responsible for the overall increase in responders in the OCA titration group. On the other hand, patients who met the primary endpoint at Month 6 did not up-titrate, despite the fact that they may have experienced further benefit from the increased dose. Indeed, some patients who remained at 5 mg OCA were responders at Month 6 but not at Month 12. This is one of the reasons that the SmPC recommends all patients to up-titrate from 5 mg to 10 mg after 6 months, if OCA is tolerated.

	Responders (%)					
	Month 6	Month 12				
Placebo (n=73)	7%	10%				
10 mg OCA (n=73)	51%	47%				
Titration OCA (n=70)	34%	46%				
Titration subgroup [†]						
Remained at 5 mg OCA for 12 months (n=36)	67% [§]	53%				
Titrated to 10 mg OCA at Month 6 (n=33)	0%	39%				

Tabla	22.	Summar	/ of	nriman		ond	noint	analy	1606
i apie	Z3:	Summary	V OT	primary	y emicacy	/ ena	point	anary	yses.

Abbreviations: ALP, alkaline phosphatase; OCA, obeticholic acid.

[†]There was one subject who withdrew from the trial due to an AE after 8 days of study medication, and

Company evidence submission template for obeticholic acid [ID785]

therefore there were no data for this subject at Month 6 and Month 12. [§]Of the 12 subjects who did not respond but did not up-titrate, nine did not increase their dose due to adverse events, and three recorded their reason as 'other'.

Other secondary analyses of the primary endpoint included the percentage of subjects achieving the endpoint at 2 weeks, and at 3, 6, 9 and 12 months (Figure 12). Both OCA treatment groups were superior to placebo in achieving the composite endpoint at all time points across the 12-month treatment period (p<0.0001 versus placebo). At Month 6, a higher percentage of subjects in the OCA 10 mg group (51%) achieved the composite endpoint compared with the OCA titration group (34%), which was statistically significant (p=0.0358). However, this is not surprising given the dose response of OCA since at this point, all patients in the OCA titration group had been receiving 5 mg per day for 6 months compared with 10 mg per day in the OCA 10 mg fixed dose group.





Abbreviations: OCA, obeticholic acid.

* p-value for treatment group versus placebo; # p-value for the between treatment group comparison at Month 6 of OCA titration (5 mg) and OCA 10 mg. P-values obtained using Cochran-Mantel-Haenszel tests stratified by randomization strata factor. Missing values were considered a non-response.

In sensitivity analyses of the primary endpoint, using observed data only, both OCA treatment groups were superior to placebo at all time points, with statistically significant differences apparent as early as Week 2 (p<0.0001). Statistically significantly (p=0.0051) more subjects in the OCA 10 mg group achieved the endpoint at 6 months compared with those in the titration group (Figure 13). Between Month 6 and Month 12, the percentage of subjects achieving the endpoint increased from 35% to 50% for the OCA titration group and was maintained for the OCA 10 mg fixed dose group (58% at Month 6 and 55% at Month 12). As with the primary endpoint, the increase observed from Month 6 to Month 12 in the OCA titration group is attributable to the group that titrated to 10 mg following the Month 6 visit versus those that remained on OCA 5 mg.

Figure 13 Sensitivity analysis of primary endpoint using observed data only



Abbreviations: OCA, obeticholic acid.

*p<0.0001; #p=0.0051 for the between treatment group comparison at Month 6 of OCA titration (5 mg) and OCA 10 mg. p-value obtained using Cochran-Mantel-Haenszel test stratified by randomisation strata factor.

4.7.1.2 Secondary efficacy endpoints

Responders based on ALP

The percentages of subjects achieving ALP reductions of $\geq 15\%$, $\geq 20\%$, and $\geq 40\%$ at 6 months and 12 months are summarised in Figure 14. The odds ratios of achieving each respective reduction for both OCA groups vs placebo are summarised in Table 24. At both time points and for each ALP category, a statistically significantly higher percentage of OCA-treated subjects achieved the reduction in ALP compared with placebo (p<0.0001 vs placebo for both OCA treatment groups). At 12 months, 21 (30%) and 25 (34%) subjects from the OCA titration and OCA 10 mg fixed dose groups, respectively, achieved an ALP reduction from baseline $\geq 40\%$ compared with 1 (1%) subject in the placebo group.





p-values versus placebo obtained using the Cochran-Mantel-Haenszel test stratified by randomisation strata factor. ***p<0.0001.

Odds ratio	≥15% re	duction	≥20 rec	duction	≥40% reduction		
	OCA titration	OCA 10 mg	OCA titration	OCA 10 mg	OCA titration	OCA 10 mg	
Month 6	7.8 (3.7, 16.5)	12.1 (5.5, 26.5)	10.2 (4.5, 22.8)	16.1 (7.0, 36.8)	NA†	NA†	
p-value for treatment ratio (OCA vs placebo) [†]	<0.0001	<0.0001	<0.0001	<0.0001	NA†	NA†	
Month 12	8.4 (4.0, 17.9)	8.2 (3.9, 17.3)	7.8 (3.7, 16.4)	8.8 (4.2, 18.7)	34.7 (4.4, 271.3)	43.0 (5.5, 334.9)	
p-value for treatment ratio (OCA vs placebo) [‡]	<0.0001	<0.0001	<0.0001	<0.0001	0.0007	0.0003	

Table 24: Odds ratio of percentage reduction in ALP from baseline: ITT population

[↑] Not applicable (NA) since zero subjects from the placebo group achieved ≥40% reduction from Baseline at Month 6. [‡]p-values obtained using a logistic regression model with terms for treatment and the randomisation strata factor.

Responders based on the biochemical response criteria

Percentage of non-responders at baseline who responded at Month 6 and Month 12

Figure 15 and Table 25 summarise the percentage of non-responders at baseline who became responders at Month 6 and Month 12 for each of the biochemical treatment response criteria. A higher percentage of OCA-treated subjects (both in the titration group and the 10 mg fixed dose group) responded to each of the criteria at both Month 6 and Month 12 compared with placebo-treated subjects. The difference between placebo and each OCA treatment group was statistically significant at both time points for all criteria, with the exception of the Rotterdam criteria where the number of subjects who were non-responders at baseline was low.



Figure 15: Percentage of responders based on Paris I[†], Toronto II, Mayo II and Paris II biochemical treatment response criteria: ITT population

Abbreviations: ALP, alkaline phosphatase; AST, aspartate aminotransferase; OCA, obeticholic acid; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

[†] Deviation from Paris I: incorporation of total bilirubin <ULN instead of <1 mg/dL of normal total bilirubin levels; **p-value <0.01; ***p-value <0.0001 for comparing treatments using Cochran-Mantel-Haenszel General Association test were stratified by randomisation strata factor.

Endpoint, n (%)	Plac	ebo	OCA titration		OCA 10 mg fixed dose		
	Month 6	Month 12	Month 6	Month 12	Month 6	Month 12	
Paris I: ALP ≤3x ULN and AST	≤2x ULN and total I	oilirubin ≤ULN		·			
Non-responders at baseline	3	4	3	36	3	5	
Responders	6 (18)	6 (18)	20 (56)	23 (64)	18 (51)	20 (57)	
p-value	-	-	0.0013	0.0002	0.0025	0.0007	
Odds ratio vs placebo (95% Cl)	_	_	6.0 (1.9, 18.6)	9.4 (2.8, 31.0)	5.2 (1.7, 16.0)	7.2 (2.2, 23.7)	
p-value	-	-	0.0019	0.0002	0.0045	0.0011	
Paris II: ALP ≤1.5x ULN and AST ≤1.5x ULN and total bilirubin ≤ULN							
Non-responders at baseline	7	3	70		73		
Responders	3 (4)	3 (4)	13 (19)	19 (27)	19 (26)	19 (26)	
p-value	-	-	0.0055	0.0001	0.0001	0.0002	
Odds ratio vs placebo (95% Cl)	_	_	5.8 (1.5, 21.9)	9.1 (2.5, 32.6)	9.4 (2.6, 34.5)	8.5 (2.4, 30.6)	
p-value	-	-	0.0095	0.0007	0.0007	0.0010	
Mayo II: ALP ≤1.67x ULN and t	total bilirubin ≤ULN						
Non-responders at baseline	7	3	6	69	73		
Responders	8 (11)	11 (15)	23 (33)	32 (46)	38 (52)	36 (49)	
p-value	-	-	0.0010	<0.0001	<0.0001	<0.0001	
Odds ratio vs placebo (95% Cl)	_	_	4.5 (1.8, 11.2)	5.7 (2.5, 13.1)	10.7 (4.3, 26.5)	6.4 (2.8, 14.7)	
p-value	-	-	0.0013	<0.0001	<0.0001	<0.0001	

Table 25: Percentage of responders and odds ratios based on biochemical treatment response criteria: ITT population

Endpoint, n (%)	Plac	cebo	OCA titration		OCA 10 mg fixed dose				
	Month 6	Month 12	Month 6	Month 12	Month 6	Month 12			
Toronto II: ALP ≤1.76x ULN	Toronto II: ALP ≤1.76x ULN								
Non-responders at baseline	7	0	6	67	7	70			
Responders	10 (14)	11 (16)	31 (46)	34 (51)	41 (59)	42 (60)			
p-value	-	-	<0.0001	<0.0001	<0.0001	<0.0001			
Odds ratio vs placebo (95% Cl)	-	-	6.5 (2.7, 15.5)	7.2 (3.0, 17.2)	11.9 (4.9, 29.0)	11.5 (4.7, 27.7)			
p-value	-	-	<0.0001	<0.0001	<0.0001	<0.0001			
Rotterdam (normal range): tot	al bilirubin ≤ULN ar	nd albumin ≥LLN							
Non-responders at baseline	1	7	12		13				
Responders	2 (12)	1 (6)	3 (25)	2 (17)	3 (23)	3 (23)			
p-value	-	-	0.4170	0.3537	0.3412	0.0992			
Odds ratio vs placebo (95% Cl)	-	-	3.0 (0.4, 23.2)	4.3 (0.3, 58.8)	2.7 (0.4, 20.3)	6.2 (0.5, 72.0)			
p-value	-	-	0.2980	0.2704	0.3239	0.1429			

Abbreviations: ALP, alkaline phosphatase; AST, aspartate aminotransferase; CI, confidence interval; LLN, lower limit of normal' OCA, obeticholic acid; ULN upper limit of normal.

Maintenance of response in subjects who were responders at baseline

A separate analysis for those subjects that met the responder criteria at baseline for each of the pre-defined criteria was performed for each outcome, to assess the maintenance of response during the 12-month treatment period. Given that the majority (>95%) of subjects were non-responders at baseline for most endpoint criteria, data are presented only for the Paris I endpoint where approximately 50% of the subjects were responders at baseline.

In OCA-treated subjects who met the Paris I criteria of responder at baseline (ALP $\leq 3x$ ULN and AST $\leq 2x$ ULN and total bilirubin \leq ULN), the response was sustained at Month 6 and Month 12 (Figure 16).



Figure 16 Percentage responders at baseline to Paris I criteria at Month 6 and Month 12

Abbreviations: OCA, obeticholic acid. Per statistical analysis plan, only descriptive statistics were provided.

Absolute and percentage change in ALP

Within 2 weeks, there was a rapid, statistically significant reduction in ALP in both OCA treatment groups when compared with placebo, which was maintained for the duration of the study (Figure 17). Table 26 summarises the least squares (LS) mean of the absolute and percentage change from baseline in ALP at Month 6 and Month 12 for the ITT population. For both OCA treatment groups, clinically and statistically significant improvements from baseline were observed at both Month 6 and Month 12, compared with placebo (p<0.0001 for both OCA treatment groups). At Month 12, the LS mean percentage change from baseline was –33% and –39% for the OCA titration and 10 mg fixed dose groups, respectively, compared with only –5% for the placebo group. ALP is the key biomarker for PBC, and a reduction in ALP is the main objective of treatment with PBC and has been shown to correlate with reduced risks of complications, liver transplantation, and death (15, 17, 18).



Figure 17: Change in ALP from baseline over time: ITT population

Abbreviations: ALP, alkaline phosphatase; ITT, intention-to-treat; OCA, obeticholic acid; UDCA, ursodeoxycholic acid. *** p<0.0001 vs placebo.

Mean ALP, U/L (SE)	Placebo	OCA titration	OCA 10 mg fixed dose
Baseline [†]	327.5 (13.5)	325.9 (13.9)	316.3 (12.2)
Month 6			
Mean	311.1 (14.4)	239.3 (13.2)	196.1 (8.4)
LS mean change	-21.7 (13.2)	-91.2 (12.9)	–121.5 (13.2)
OCA vs placebo	-	-69.6 (11.7)	-99.9 (12.0)
p-value [‡]	-	<0.0001	<0.0001
LS mean % change	-6.8 (3.5)	-27.4 (3.4)	-36.5 (3.5)
OCA vs placebo	-	-20.7 (3.1)	-29.8 (3.2)
p-value [‡]	-	<0.0001	<0.0001
Month 12			
Mean	321.3 (17.1)	219.5 (12.5)	192.3 (7.8)
LS mean change	-14.4 (14.7)	–112.5 (14.4)	-129.9 (14.6)
OCA vs placebo	-	-98.1 (13.1)	–115.5 (13.2)
p-value [‡]	-	<0.0001	<0.0001
LS mean % change	-4.8 (3.8)	-33.0 (3.7)	-39.1 (3.8)
OCA vs placebo	-	-28.2 (3.4)	-34.4 (3.4)
p-value [‡]	_	<0.0001	<0.0001

Table 26: Change in Al	.P from basel	ine at Month 6 an	nd Month 12:	ITT population
------------------------	---------------	-------------------	--------------	----------------

Abbreviations: ALP, alkaline phosphatase; ANCOVA, analysis of covariance; ITT, intention-to-treat; LS, least squares; OCA, obeticholic acid.

[†] Baseline is defined as the mean of all available evaluations prior to treatment; [‡]p-values obtained using an ANCOVA model with baseline value as a covariate, and treatment and randomisation strata factor as fixed effects.

Company evidence submission template for obeticholic acid [ID785]

Similar data were observed for the completer and EE populations. Sensitivity analyses demonstrated comparable clinically and statistically significant improvements for both OCA treatment groups compared with placebo at all time points (p <0.0001 versus placebo).

Absolute and percentage change in total and conjugated bilirubin

For the ITT population, the baseline mean total bilirubin was below the ULN for all treatment groups. Decreases in the absolute change from baseline in total bilirubin were observed for both OCA treatment groups, compared with an increase for the placebo treatment group. This is particularly important as it clearly shows improvement in the OCA treatment groups but disease progression in the placebo group, 93% of which were taking UDCA during the trial. An increase in bilirubin levels is associated with an increased risk of liver transplantation and death, and this risk starts at levels >0.5x ULN (10 μ mol/L) (47). At 12 months, mean bilirubin levels in both OCA treatment groups are <10 μ mol/L, whereas the mean level in the placebo group is 13.2 μ mol/L. This reinforces the fact that without an additional treatment option, patients who do not respond to, or are intolerant to, UDCA are at increased risk of disease progression.

The difference in the absolute change from baseline in total bilirubin between placebo and each respective OCA treatment group was statistically significant (p < 0.01) as early as Month 3, and at all subsequent time points with the exception of the OCA titration group at Month 9 (Table 27 and Figure 18).

Baseline conjugated bilirubin was elevated across treatment groups, which is consistent with intrahepatic obstruction with hepatobiliary damage observed in PBC. Mean baseline conjugated bilirubin was 5.5, 4.5, and 4.9 μ mol/L, for the placebo, OCA titration, and OCA 10 mg fixed dose treatment groups, respectively. Both OCA treatment groups decreased and approached the ULN for conjugated bilirubin, while placebo increased from baseline at all time points, indicating ongoing cellular damage. For both OCA treatment groups, statistically significant differences (p<0.05) in both the absolute and percentage change from baseline relative to placebo were observed at Month 3 and at all subsequent time points. At Month 12, p-values were <0.0001 for each OCA treatment group versus placebo (Table 28 and Figure 18).

Total bilirubin (µmol/L)	Placebo	OCA titration	OCA 10 mg fixed dose
Baseline [†]	11.8 (0.9)	10.3 (0.7)	11.3 (0.8)
Month 6			
Mean (SE)	12.3 (0.9)	9.7 (0.7)	9.7 (0.5)
LS mean change	1.1 (0.6)	-0.4 (0.6)	-0.8 (0.6)
OCA vs placebo, treatment difference	_	-1.5 (0.6)	-1.9 (0.6)
p-value [‡]	-	0.0089	0.0011

Table 27. C	hange in t	total bilirubin	from I	hasolino -	at Month	6 and	Month	12. ITT	nonulat	ion
I able Z1. C	nange m	lolai piinupin	nom	Jasenne a		o anu	wonun	12.111	populat	1011

Total bilirubin (µmol/L)	Placebo	OCA titration	OCA 10 mg fixed dose
LS mean % change	13.8 (8.8)	-3.0 (8.6)	7.4 (8.9)
OCA vs placebo, treatment difference	_	-16.8 (8.1)	-6.5 (8.2)
p-value [‡]	-	0.0381	0.4307
Month 12			
Mean (SE)	13.2 (1.0)	9.9 (0.6)	9.7 (0.6)
LS mean change	1.9 (0.7)	-0.4 (0.7)	-1.0 (0.7)
OCA vs placebo, treatment difference	_	-2.3 (0.6)	-2.9 (0.7)
p-value [‡]	_	0.0004	<0.0001
LS mean % change (SE)	19.5 (6.8)	1.2 (6.7)	-0.2 (6.9)
OCA vs placebo, treatment difference	_	-18.3 (6.3)	-19.8 (6.3)
p-value [‡]	_	0.0039	0.0020

Abbreviations: ANCOVA, analysis of covariance; ITT, intention-to-treat; LS, least squares; OCA, obeticholic

acid; SE, standard error. [†]Baseline is defined as the mean of all available evaluations prior to treatment; [‡]p-values obtained using an ANCOVA model with baseline value as a covariate and fixed effects for treatment and randomisation strata factor.

Table 28: Change in co	onjugated bilirubin fror	n baseline at Month 6 a	and Month 12: ITT
population			
			1

Conjugated bilirubin (µmol/L)	Placebo	OCA titration	OCA 10 mg fixed dose
Baseline [†]	5.5 (0.7)	4.5 (0.5)	4.9 (0.5)
Month 6			
Mean	6.2 (0.8)	4.0 (0.4)	3.5 (0.3)
LS mean change	1.0 (0.4)	-0.4 (0.4)	-1.0 (0.4)
OCA vs placebo, treatment difference	_	-1.4 (0.3)	-1.9 (0.3)
p-value [‡]	-	<0.0001	<0.0001
LS mean % change	23.0 (6.0)	-1.0 (6.0)	-6.0 (7.0)
OCA vs placebo	-	-24.0 (6.0)	-28.0 (6.0)
p-value [‡]	_	0.0001	<0.0001

Conjugated bilirubin (µmol/L)	Placebo	OCA titration	OCA 10 mg fixed dose
Month 12			
Mean	6.9 (0.9)	4.2 (0.4)	3.8 (0.3)
LS mean change	1.9 (0.5)	-0.2 (0.5)	-0.5 (0.5)
OCA vs placebo, treatment difference	_	-2.1 (0.5)	-2.4 (0.5)
p-value [‡]	_	<0.0001	<0.0001
LS mean % change	39.0 (7.0)	12.0 (7.0)	5.0 (7.0)
OCA vs placebo, treatment difference	-	-27.0 (7.0)	-34.0 (7.0)
p-value [‡]	_	<0.0001	<0.0001

Abbreviations: ANCOVA, analysis of covariance; ITT, intention-to-treat; LS, least squares; OCA, obeticholic acid.

[†]Baseline is defined as the mean of all available evaluations prior to treatment; [‡]p-values obtained using an ANCOVA model with baseline value as a covariate and fixed effects for treatment and randomisation strata factor.

Figure 18: Summary of change in total and conjugated bilirubin



Abbreviations: ANCOVA, analysis of covariance; LS, least squares; OCA, obeticholic acid; SE, standard error. *p<0.05, **p<0.01, ***p<0.0001 vs placebo; p-values for comparing OCA treatments to placebo were obtained using an ANCOVA model with baseline value as a covariate and fixed effects for treatment and randomization strata factor.

Absolute and percentage change in GGT, ALT and AST

Results for GGT, ALT, and AST values at Month 6 and Month 12 are summarised in Table 29. As expected, GGT, ALT, and AST values were all elevated at baseline. Treatment with OCA resulted in a clinically and statistically significant improvement from baseline in all three parameters. Improvements were observed as early as 2 weeks, with the largest response generally observed at 3 months. For both OCA treatment groups and across all three parameters, the differences compared with placebo in LS mean reductions from baseline were statistically significant at all time points (Figure 19). The statistically significant reductions in these biochemical markers support the evidence from ALP and bilirubin that OCA is slowing disease progression and improving liver function.



Figure 19 GGT, ALT, and AST change from baseline over time

Abbreviations: ALT, alanine aminotransferase; ANCOVA, analysis of covariance; AST aspartate aminotransferase; GGT, gamma-glutamyl transferase; OCA, obeticholic acid. **p<0.01; ***p≤0.0001; p-values for comparing OCA treatments to placebo were obtained using an ANCOVA model with baseline value as a covariate and fixed effects for treatment and randomization strata factor.

	GGT (U/L)			ALT (U/L)			AST (U/L)			
Endpoint mean (SE)	Placebo	OCA titration	OCA 10 mg fixed dose	Placebo	OCA titration	OCA 10 mg fixed dose	Placebo	OCA titration	OCA 10 mg fixed dose	
Baseline [†]	309.6 (52.6)	252.8 (20.0)	261.1 (24.3)	56.0 (3.5)	61.6 (4.7)	56.3 (4.7)	48.8 (2.6)	52.3 (3.0)	50.5 (3.6)	
Month 6										
Mean (SE)	270.7 (41.0)	136.4 (12.3)	95.2 (11.4)	49.7 (3.0)	40.7 (3.8)	32.7 (2.2)	46.7 (2.5)	40.6 (2.7)	36.8 (2.1)	
LS mean change from baseline	-32.4 (19.3)	-132.1 (18.8)	–181.5 (19.6)	6.8 (3.0)	-18.8 (3.0)	-23.4 (3.1)	0.1 (2.2)	8.2 (2.1)	-10.2 (2.2)	
OCA vs placebo, treatment difference	_	–99.8 (17.5)	–149.2 (17.8)	_	-12.0 (2.8)	-16.5 (2.8)	_	-8.3 (2.0)	-10.3 (2.0)	
p-value [‡]	-	<0.0001	<0.0001	-	<0.0001	<0.0001	-	<0.0001	<0.0001	
LS mean % change from baseline	-9.0 (5.0)	-44.0 (4.8)	-64.1 (5.1)	-9.3 (3.8)	–31.1 (3.8)	-36.2 (3.9)	2.1 (3.5)	-15.4 (3.5)	–15.4 (3.6)	
OCA vs placebo, treatment difference	_	-35.0 (4.5)	-55.1 (4.6)	_	-21.9 (3.5)	-26.9 (3.5)	_	-17.5 (3.2)	-17.5 (3.2)	
p-value [‡]	_	<0.0001	<0.0001	_	<0.0001	<0.0001	_	<0.0001	<0.0001	

Table 29: Change in GGT, ALT, and AST from baseline at Month 6 and 12: ITT population

	GGT (U/L)			ALT (U/L)			AST (U/L)			
Endpoint mean (SE)	Placebo	OCA titration	OCA 10 mg fixed dose	Placebo	OCA titration	OCA 10 mg fixed dose	Placebo	OCA titration	OCA 10 mg fixed dose	
Month 12										
Mean	301.8 (51.1)	114.2 (12.5)	91.9 (10.2)	52.8 (3.4)	39.0 (4.2)	32.1 (2.6)	51.6 (4.7)	39.5 (3.1)	36.4 (2.4)	
LS mean change from baseline	6.7 (25.6)	-140.8 (24.7)	-176.7 (25.6)	-5.0 (3.3)	-21.3 (3.3)	–25.3 (3.4)	1.0 (4.2)	-13.0 (4.2)	-15.0 (4.3)	
OCA vs placebo, treatment difference	_	-147.5 (23.1)	–183.4 (23.3)	_	-16.3 (3.0)	-20.4 (3.1)	_	-14.1 (3.8)	-16.0 (3.9)	
p-value [‡]	-	<0.0001	<0.0001	-	<0.0001	<0.0001	-	0.0003	<0.0001	
LS mean % change from baseline	0.8 (5.7)	-50.3 (5.5)	-63.7 (5.2)	-4.7 (5.0)	-35.5 (4.9)	-41.7 (5.0)	7.7 (8.8)	-21.9 (8.7)	-23.7 (8.9)	
OCA vs placebo, treatment difference	_	-51.1 (5.1)	-64.5 (5.2)	_	-30.9 (4.5)	37.0 (4.6)	_	-29.6 (8.0)	-31.4 (8.1)	
p-value [‡]	-	<0.0001	<0.0001	-	<0.0001	<0.0001	_	0.0003	0.0001	

Abbreviations: ALT, alanine aminotransferase; ANCOVA, analysis of covariance; AST, aspartate aminotransferase; GGT, gamma glutamyl transferase; ITT, intention-to-treat; LS, least squares; OCA, obeticholic acid; SE, standard error. [†]Baseline is defined as the mean of all available evaluations prior to treatment; [‡]p-values obtained using an ANCOVA model with baseline value as a covariate and fixed

effects for treatment and randomisation strata factor.

Absolute change in disease-specific symptoms (measured by change in PBC-40 and domain scores)

The disease-specific measure for PBC showed no clinically significant differences compared with placebo for the global score or individual scores of general symptoms, fatigue, cognitive function, and emotional/social domains. However, a difference was observed in itch scores. During the initial 3 months of treatment, the largest LS mean increase in itch was observed for the OCA 10 mg group, followed by the OCA titration group (Figure 20). The LS mean difference in itch score between the OCA 10 mg group and placebo group was statistically significant at both Week 2 (p= 0.0048) and Month 3 (p<0.0001) but not at any subsequent time points.





Abbreviations: LS, least squares; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; SE, standard error; UDCA, ursodeoxycholic acid. *p=0.0048; ***p<0.0001

Other relevant biomarkers and non-invasive assessments of liver fibrosis

Enhanced liver fibrosis (ELF) score

Baseline and Month 12 ELF data were available for 84% of subjects and are summarised in Table 30. Baseline ELF scores were slightly higher in the placebo group than in the OCA treatment groups, and were indicative of the upper end of moderate fibrosis to the lower end of severe fibrosis. At Month 12, the magnitude of the increase in total ELF score was lower in the OCA treatment groups than in the placebo group, although the treatment difference vs placebo was not significant for either OCA treatment group.

ELF score [†]	Placebo		OCA titration		OCA 10 mg fixed dose		
	n	n Mean (SE)		Mean (SE)	n	Mean (SE)	
Baseline [‡]	70 10.03 (0.15)		65 9.76 (0.13)		71	9.81 (0.14)	
Month 6							
Mean	69 10.00 (0.14)		67	9.83 (0.14)	60	9.83 (0.15)	
Absolute change	67 0.09 (0.06)		63	0.13 (0.08)	59	0.14 (0.07)	
LS mean absolute change	67 0.04 (0.11)		63	0.06 (0.11)	59	0.09 (0.11)	
Treatment difference vs placebo		-		0.03 (0.10)		0.05 (0.10)	
p-value [§]	_		0.7811		0.6209		
Month 12							
Mean	66	10.12 (0.16)	61	9.73 (0.13)	64	9.77 (0.15)	
Absolute change	64	0.17 (0.08)	55	0.15 (0.08)	63	0.08 (0.07)	
LS mean absolute change	64	0.33 (0.13)	55	0.24 (0.12)	63	0.20 (0.12)	
Treatment difference vs placebo		_		-0.08 (0.11)		-0.13 (0.11)	
p-value [§]	_		0.4718		0.2324		

Table 30: ELF score using observed data: ITT population

Abbreviations: ANCOVA, analysis of covariance; ELF, enhanced liver fibrosis; ITT, intention-to-treat; LS, least squares; OCA, obeticholic acid; SE, standard error.

[†]ELF score ranges: 7.7 for a high sensitivity exclusion of fibrosis, 9.8 for a high specificity identification of fibrosis (sensitivity 69%, specificity 98% for moderate fibrosis), and 11.3 to discriminate cirrhosis (sensitivity 83%, specificity 97%); [‡]Baseline is defined as the day 0 evaluation prior to treatment; [§]p-values comparing OCA treatments to placebo were obtained using an ANCOVA model with baseline value as a covariate and fixed effects for treatment and randomisation strata factor.

Hepatic stiffness

Mean and median baseline hepatic stiffness values using FibroScan[®]Fibroscan transient elastography were generally in the advanced fibrosis range (130). After 12 months of treatment, there was a general increase in hepatic stiffness across all three treatment groups, with the lowest increase occurring in the OCA treatment groups. However, there were no statistically significant differences between the placebo group and either OCA treatment groups.

Fibroblast growth factor-19 (FGF-19)

The median difference in absolute change and percentage change from baseline between the placebo and each OCA treatment group was statistically significant at Month 6 (p<0.0001) and Month 12 (p=0.0002 for the OCA titration group and p<0.0001 OCA 10 mg fixed dose group).

Cytokeratin-18 (CK-18)

Considerable inter-subject variability was shown across all three treatment groups for baseline CK-18 levels, with the median >ULN. During the study period, there was a statistically significant reduction at 6 months in the OCA titration and OCA 10 mg fixed

dose groups (p=0.0012 and 0.0077, respectively) and at 12 months (p=0.0003 and <0.0001, respectively).

Total bile acids

Total bile acid concentrations were similar at baseline across all three treatment groups, and there were no significant differences at Month 6 or Month 12 between the OCA titration and OCA 10 mg fixed dose treatment groups.

Total endogenous bile acid

Bile acid retention and cholestasis are key factors influencing the progressive loss of liver function in PBC, therefore a reduction in total endogenous bile will have a significant clinical benefit for the patient. OCA has a unique mode of action mediated through FXR receptor activation, leading to the reduction of endogenous bile acids demonstrated below.

The mean baseline levels of total endogenous bile acids were similar in the placebo group and the OCA titration group at baseline (6.63 vs 7.06 μ mol, respectively), but was higher in the OCA 10 mg fixed dose treatment group (9.48 μ mol). At Month 6, there was an overall decrease of -1.41μ mol and -5.72μ mol in the OCA titration and OCA 10 mg fixed dose groups, respectively, and an increase of 2.24 μ mol in the placebo group. This corresponds to a 34% increase from baseline in the placebo group, but a 20% reduction in the OCA titration group and a 60% reduction in the OCA 10 mg fixed dose group.

By Month 12, the mean total endogenous bile acid levels had increased to 14.59 μ mol in the placebo treatment group, but had decreased to 3.77 μ mol in the OCA titration group and 3.86 μ mol in the OCA 10 mg fixed dose group. When compared with baseline, bile acid levels had more than doubled in the placebo group (an increase of 220%), but had decreased by 47% and 59% in the OCA titration and 10 mg fixed dose groups, respectively. This represents a statistically significant reduction by Month 12 in the OCA titration and OCA 10 mg fixed dose groups (p=0.0010 and 0.0037, respectively).

Importantly, LCA – a toxic secondary bile acid – remained stable across all treatment groups when compared with baseline.

Inflammatory indirect markers

C-reactive protein (CRP)

Clinically and statistically significant decreases in CRP values were observed at Month 6 and Month 12 for both OCA treatment groups. The median change at Month 12 was -0.5 mg/L for the OCA titration group (p=0.0005 vs placebo) and -0.6 mg/L for the OCA 10 mg fixed dose group (p=0.0022 vs placebo), compared with an increase of 0.35 mg/L for the placebo group. Consistent with other efficacy parameters, the median change at Month 6 was smaller for subjects receiving OCA 5 mg compared with those receiving OCA 10 mg.

Tumour necrosis factor-α (TNF-α)

Baseline median levels were >ULN for TNF- α across all three treatment groups and slightly higher in subjects treated with placebo vs those treated with OCA. The OCA titration group had smaller increases at Month 6 and Month 12 compared with placebo, while the OCA 10 mg fixed dose group achieved a statistically significant reduction at Month 6 and Month 12 versus placebo (p=0.0002 and 0.0077, respectively).

Transforming growth factor- β (TGF- β) and interleukin-6 (IL-6)

Both parameters were within normal limits at baseline, and no significant changes were observed for TGF- β and IL-6 at Month 6 and Month 12 versus placebo.

Drug dose, drug concentration, and relationship to response

There was higher systemic exposure to OCA in the OCA 10 mg fixed dose treatment group than in the titration group. A dose relationship was observed, with 35% and 58% of subjects in the OCA titration and 10 mg fixed dose treatment groups, respectively, achieving the primary endpoint at Month 6. The magnitude of the LS mean reduction was larger for ALP, total bilirubin, and conjugated bilirubin at Month 6 in subjects that received 10 mg OCA versus those that received 5 mg OCA.

Effect of bile acid sequestrants (BAS) on the exposure and efficacy of OCA

For the OCA 10 mg fixed dose group, the use of BAS did not affect the change from baseline for ALP, total bilirubin, and conjugated bilirubin at Month 6 or Month 12, and therefore did not alter the efficacy of OCA.

For the OCA titration group, the use of BAS appeared to modestly attenuate the effect of OCA on ALP and bilirubin from baseline to Month 6 and Month 12. This supports a dose-relationship response and may suggest that 5 mg is the minimally efficacious level.

Since BAS are used to control pruritus, the small to no effect on OCA efficacy is important, meaning that pruritus as a symptom of PBC and OCA treatment can be managed without affecting OCA activity.

4.7.1.3 Conclusion

Increases in ALP, bilirubin, GGT, ALT and AST are associated with disease progression (Figure 21). In POISE, all of these, as well as other relevant biomarkers (FGF-19, CK-18 and bile acids), inflammation biomarkers (CRP, TNF- α , TGF- β and IL-6), and assessments of liver fibrosis, indicate that OCA slows, halts or even reverses disease progression in patients with an inadequate response to, or who are intolerant to, UDCA.


Figure 21: The progression of PBC and associated changes in biochemical markers

These are important results for these patients, who currently have no available licensed or effective treatment options, and face the inevitability of disease progression to complications, the need for liver transplant, HCC, and death. Further evidence is needed to show that these reductions in biomarkers translate into a benefit in clinical outcomes, e.g. transplant-free survival. COBALT is an ongoing long-term study that seeks to evaluate the effect of OCA on clinical outcomes. It is described in Section 4.14.

4.7.2 Supporting evidence

4.7.2.1 Long-term safety extension of POISE

A total of 198 subjects completed the 12-month double-blind phase of the POISE trial, of which 193 subjects (97%) opted to continue into the LTSE phase. Of the five subjects who did not opt into the LTSE phase, four were in the placebo group in the double-blind phase of the trial, and one was in the OCA titration group for the double-blind phase. All 64 subjects who were randomised to 10 mg OCA and completed the double-blind phase chose to enter the LTSE phase of the study.

A total of 77% of subjects had a weighted average daily dose of >5 mg (i.e. up-titrated during the LTSE phase). This is in contrast to the double-blind phase, where only 48% of subjects in the titration group increased their dose to 10 mg. In the titration group in the double-blind phase, 35% of patients did not up-titrate because they had reached the criteria for response and were therefore not eligible for an increased dose (as per the protocol), and 13% did not up-titrate due to adverse events. The remaining 4% recorded their reason for not up-titrating as 'other'.

Primary efficacy endpoint

- The percentage of subjects achieving the key primary endpoint in the double-blind phase was sustained over the subsequent 12-month open-label period (52% and 44% in the OCA titration group and OCA 10 mg group, respectively).
- There was a decrease from month 12 of the double-blind phase to month 3 of the LTSE phase in the percentage of subjects achieved the primary endpoint for subjects previously treated with 10 mg OCA. This was expected due to the down-

Company evidence submission template for obeticholic acid [ID785]

titration from 10 mg OCA to 5 mg OCA, however, the response rate was improved at month 6.

- For subjects who previously received placebo in the 12-month double-blind phase but subsequently entered the LTSE phase of the study and initiated treatment with 5 mg OCA, there was a robust response in the percentage of subjects who achieved the primary endpoint between Month 12 of the double-blind phase to Month 9 of the LTSE phase (9% vs 34%, respectively). Subjects were able to uptitrate in 5 mg increments not more than once every three months if the response criteria were not met and AEs did not limit the administration of the higher dose of OCA, and results in this group of patients are similar to the results seen in the OCA titration group in the double-blind phase.
- Similar consistent, clinically relevant percentages of subjects achieving the primary endpoint were also observed for the 2-year completer population.

ALP

- Mean improvements in ALP values for subjects randomised to OCA treatment in the double-blind phase were sustained over the subsequent open-label period, notably, absolute mean ALP approached 1.67x ULN.
- There was a slight increase in ALP at Month 3 in subjects who previously received 10 mg OCA.
- ALP levels decreased in the LTSE subjects who received placebo in the doubleblind phase to levels almost equal to those in the OCA treatment groups (Figure 22).

Figure 22: ALP levels over time by treatment in the double-blind phase



Abbreviations: ALP, alkaline phosphatase; LTSE, long-term safety extension; OCA, obeticholic acid; SD, standard deviation.

Bilirubin

- Following initiation of OCA in the LTSE, mean total bilirubin levels decreased and were comparable to or below the double-blind baseline values.
- For those that received OCA during the double-blind phase, continued OCA treatment maintained total bilirubin concentrations at or below baseline.

GGT, ALT, and AST

- The clinically and statistically significant reductions from baseline GGT, ALT, and AST observed in OCA-treated subjects during the 12-month double-blind phase were sustained during the LTSE.
- For subjects who received placebo during the double-blind phase, similar statistically significant reductions from the double-blind baseline were observed for all three parameters at month 3 of the LTSE.

Summary

In summary, the efficacy associated with continued treatment with OCA was durable and consistent with the effects observed in the double-blind phase of the POISE study. For subjects originally randomised to OCA, continued treatment was associated with sustained improvements in liver chemistry. For subjects originally randomised to placebo, there were clinically significant improvements in liver chemistry following initiation of OCA treatment. Importantly, the tendency towards a deterioration in total bilirubin in placebo subjects was reversed following initiation of OCA treatment.

4.7.2.2 Phase 2 studies

There are two Phase 2 supporting studies, 747-201 and 747-202. The studies evaluated the efficacy, safety, and tolerability of OCA with UDCA (study 747-202) or without UDCA (study 747-201) vs placebo. The studies both included a 10 mg OCA treatment arm and other treatment arms at higher doses. In this section, key efficacy and safety results were summarised only for subjects who received 10 mg OCA, since this is the upper limit for the licensed indication of OCA. Both studies included a double-blind phase and an open-label long-term safety extension phase. Results from study 747-201 are summarised in Table 31 (double-blind phase) and Table 32 (interim results from the long-term extension phase), and results from study 747-202 are summarised in Table 33 (double-blind phase) and Table 34 (long-term extension phase).

Title	A study of INT-747-201 (6-ECDCA; OCA) monotherapy in patients with primary biliary cirrhosis [†]
Study design	A 3-month international, multi-centre, randomised, double-blind, placebo- controlled, multi-dose, Phase II parallel group study of OCA monotherapy in subjects with a proven or likely diagnosis of PBC.
Location	18 centres in 6 countries (UK [5 sites], USA [4 sites], Canada [3 sites], Germany [4 sites], France [1 site], Spain [1 site]).

Table 31: Stud	v 747-201	(double-blind	phase)
	, •		p

Primary	To assess the effects of OCA in subjects with proven or likely PBC on:	
objectives	Serum ALP levels	
	Safety.	
Secondary	To assess the effects of OCA in subjects with proven or likely PBC on:	
objectives	Hepatocellular injury and liver function	
	Disease-specific and general health symptoms	
	Biomarkers of hepatic inflammation and fibrosis	
	Plasma trough concentrations of OCA and its major known conjugates.	
Interventions	Eligible subjects were randomised (1:1:1) to placebo, OCA 10 mg, or OCA 50 mg	
Sample size and power calculation	A total of 71 subjects were screened, 60 were randomised, and 59 received investigational product: placebo (n=23), OCA 10 mg (n=20), and OCA 50 mg (n=16).	
	Approximately 20 subjects per group resulted in 49% power to detect an effect size of 0.6466 for the difference in the primary efficacy endpoint (change in serum ALP) between treatment arms and placebo using a Wilcoxon-Mann-Whitney rank-sum test with a 0.05, two-sided significance level.	
Key inclusion/	Proven or likely PBC	
exclusion	Aged 18–70 years (18–75 years in the UK)	
Criteria	 Both male and female subjects had to use one method of contraception unless surgically sterile (males and females) or postmenopausal (females) 	
Primary outcome	Percentage change in serum ALP from baseline to end of study	
Secondary outcomes	 Absolute changes in serum ALP levels from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up 	
	 Percentage of subjects meeting PBC responder criteria per the Paris I, Toronto I, Toronto II, Toronto III, Toronto IV, Mayo II, and Barcelona disease prognostic risk criteria at Day 85/end of treatment 	
	 Absolute and percentage change in serum AST, ALT, GGT, and conjugated (direct) bilirubin values from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up Safety 	
Other officery	 DBC 40 Ool Questionnoire: Change from Deceline to Dev 20, Dev 57 	
endpoints	PBC-40 QoL Questionnaire: Change from Baseline to Day 29, Day 57, and Day 85/end of treatment for each domain	
Statistical methods	A hierarchical testing strategy (131) was planned for the comparison of OCA treatment groups versus placebo. The statistical significance was to be evaluated in order as follows: if statistical significance at α =0.05 was observed for the OCA 10 mg group versus placebo, then the statistical significance at α = 0.05 for the OCA 50 mg versus placebo was to be performed. If no statistical significance was observed at α =0.05 at the first step, then the subsequent comparisons were not considered statistically significant, regardless of the p-value. For secondary endpoints, pairwise comparisons for placebo versus both OCA treatment groups at each on-treatment visit were performed using Wilcoxon-Mann-Whitney, chi-squared (with continuity correction) or Fisher's exact test.	
Results for OCA	l0 mg vs placebo	
Patient population	Baseline and demographic characteristics were generally well-balanced across the treatment groups.	

Primary endpoint	There was a clinically and statistically significant improvement in ALP levels from baseline to end of study with OCA 10 mg versus placebo (p<0.0001). The mean (SD) percentage change in ALP levels was -44.5% (24.4) for the OCA 10 mg group versus +0.4% (15.3) for placebo.
Secondary	Efficacy:
endpoints	• Results for the absolute change in ALP were supportive of the primary endpoint analyses at Day 15, 29, 57, 85/end of treatment, and 99/follow-up. There was a statistically significant two-fold decrease in ALP from 461.6 U/L at baseline to 228.1 U/L at Day 85/end of treatment (p<0.0001)
	 Percentages of OCA-treated subjects meeting PBC responder criteria at Day 85/end of treatment were significantly greater with OCA 10 mg than with placebo across all disease prognostic criteria apart from the Paris I criterion and the Toronto III criterion
	 Treatment with OCA 10 mg resulted in a statistically significantly greater proportion of subjects achieving ALP <1.67x ULN <u>and</u> total bilirubin ≤ULN <u>and</u> ALP decrease ≥15% from baseline when compared with placebo (44% vs 4%, respectively)
	 Improvements in other markers of liver injury were supported by decreases in GGT, ALT, and AST
	• There was a statistically significantly greater decrease in conjugated bilirubin levels between baseline and Day 85/end of treatment in the OCA 10 mg group vs the placebo group (p=0.0184)
	• There were no statistically significant changes between baseline and end of treatment in the SF-36 survey, or in the general symptoms, cognitive function, and emotional/social domains of the PBC-40 questionnaire
	• Mean levels of endogenous bile acids decreased by 10% from baseline to Day 85/end of treatment in the OCA 10 mg group compared with a 52% increase in the placebo group.
	Safety:
	• AEs were reported by 90% and 91% of subjects treated with OCA 10 mg and placebo, respectively
	The majority of AEs were mild or moderate in severity
	The most commonly reported AE across groups was pruritus
	There were no deaths
	• There was one reported SAE (rash); it occurred in the placebo group.
Conclusion	OCA monotherapy at 10 mg was highly effective in mitigating key clinical laboratory indicators of PBC disease, including significantly reducing serum ALP levels, increasing the percentage of responders, and reducing markers of liver injury (i.e. GGT, AST, and ALT levels). OCA treatment was well tolerated, with pruritus, a known side effect of PBC, being the most common adverse event with OCA.
Publications	Kowdley 2011 (96)

Abbreviations: AE, adverse event; ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; 6-ECDCA, 6alpha-ethyl-chenodeoxycholic acid; ET, early termination; GGT, gamma-glutamyl transferase; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; SAE, serious adverse event; QOL, quality of life; ULN, upper limit of normal. [†]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cholangitis. At the time of the design of the protocol of this study, the official name was primary biliary cirrhosis and, as such, this is reflected in this table.

Title	A study of INT-747-201 (6-ECDCA; OCA) monotherapy in patients with primary biliary cirrhosis [†]
Study design	A 72-month, open-label, long-term safety extension to provide further evaluation of the long-term safety and efficacy of OCA in doses up to 50 mg OD in subjects with PBC. Subjects who completed the 3-month, double-blind phase of Study 747-201 had the option to enrol in the LTSE.
Location	UK (3 sites). US (3 sites), Canada (3 sites) and Spain (1 site).
Primary objectives	Safety
Secondary	ALP levels, as an assessment of efficacy
objectives	Hepatocellular injury and liver function
	Disease-specific and general health symptoms
Interventions	OCA. The starting dose of OCA was based on the dose of OCA or placebo received in the double-blind phase, or on the timing of entry into the LTSE phase:
	Placebo subjects: Start on 10 mg OCA OD
	 Active treatment subjects: Resume the same randomised dose of OCA, provided that the OCA dose was well tolerated during the double-blind phase
	 Double-blind completed subjects: If the last dose of OCA was taken 1 month before the start of the LTSE phase, start on 10 mg OCA OD
	 Subjects with significant pruritus or other AEs: Must not have received OCA for at least 1 month before starting the LTSE phase, start on a maximum dose of 10 mg OCA OD
	OCA was to be titrated from 10 mg to 25 mg to 50 mg OD no faster than at 8- week intervals, until one of the following occurred:
	ALP was in the normal range
	The Investigator considered that an adequate therapeutic response was obtained
	AEs limited the administration of higher doses
	Exceptions were allowed, e.g. titration could have been performed more slowly and with smaller increments, and subjects could be titrated to doses above 50 mg OD if OCA 50 mg was well tolerated for ≥3 months.
Sample size and power calculation	All subjects who completed treatment during the double-blind phase were eligible to continue into the LTSE phase at selected study sites. No formal sample size calculations were performed.
Key inclusion/ exclusion criteria	Completion or participation (including early termination subjects) of treatment during the double-blind phase of the study
Primary efficacy endpoint	None
Secondary efficacy	Absolute and percentage change from baseline in ALP at each 3-month visit
outcomes	 Absolute and percentage change from baseline in AST, ALT, GGT, and total and conjugated bilirubin at each 3-month visit

Table 32: Study 747-201 (interim results from the long-term safety extension phase)

to last available visit

•

Absolute change from baseline in SF-36 scores and summary measures

	Absolute change from baseline in PBC-40 domains at 6-month visits
Other efficacy endpoints	Percentage of subjects that meet disease response criteria: Paris I, Paris II, Toronto I, Toronto III, Mayo II, and Barcelona
	• Percentage of subjects with ALP within normal range and a decrease from baseline of 10%, 15%, 20% and 40% at each 3-month visit
	• Percentage of subjects with ALP <1.0, 1.25, 1.5, 1.67, 1.76, 2.0 and 3.0x ULN who had ≥the respective value at double-blind baseline, at each 3-month visit
	 Percentage of subjects with ALP <1.67x ULN, total bilirubin ≤ULN, and ≥15% decrease in ALP who had ALP ≥1.67x ULN or total bilirubin >ULN at baseline, at each 3-month visit
	Absolute and percentage change from baseline in ELF score and its components (HA, P3NP and TIMP-1) at 6-month visits
	• Absolute change from baseline in biomarkers of hepatic inflammation and fibrosis (CRP, TNF- α , TGF- β , and IgM) at 6-month visits
	Titration effects on ALP, AST, ALT and GGT
	 Absolute change from baseline for pruritus 5-D and VAS scores at 3-month visits
Primary safety	Pruritus (AEs)
endpoints	• TEAEs
Secondary	Vital signs
endpoints	Physical examinations
	12-lead electrocardiograms
	Laboratory results
Statistical methods	As dose modifications (both up- and down-titration) were allowed, treatment groups were defined in two ways: weighted average daily dose (WADD) and dose for 80% duration.
	WADD was presented to account for the flexibility of dose adjustments, titration, and frequency as specified by the protocol. Dose for 80% duration was presented regardless of subject non-compliant investigational product interruptions, Investigator-prescribed drug holidays, or dosing frequency.
	WADD and dose for 80% duration were categorised into the following treatment dose groups in summary tables as applicable:
	● ≤10 mg
	• >10 mg to ≤25 mg
	• >25 mg
	Total OCA (combining all dose groups)
Results	
Patient population	A total of 28 subjects participated in the LTSE phase, with 23 (82%) participating for at least 1 year
Efficacy results	Long-term treatment with OCA was associated with sustained reductions in ALP that were consistent with the observed effect of OCA in the double-blind phase of the study, showing the consistency and durability of OCA treatment. The apparent further decrease in ALP in the LTSE was due to the transitioning of placebo subjects to OCA treatment, which further decreased mean ALP of the entire cohort.
	Similar to ALP, sustained mean reductions in AS1, AL1, GG1, and total bilirubin were observed at most LTSE time points. Changes in conjugated bilirubin tended to be more variable; however, values generally remained

	comparable to baseline.
Safety results	Long-term treatment with OCA was generally well tolerated. Consistent with the double-blind phase of the study and PBC, pruritus was the most common TEAE. Most pruritus events were mild to moderate in severity, with three subjects discontinuing the study due to pruritus. Other TEAEs with an exposure-adjusted incidence >10 events per 100 PYE included nausea, fatigue and arthralgia. In total seven subjects discontinued the study due to an AE or clinical laboratory value, eight subjects experienced a total of 19 serious AEs, and there was no clear dose relationship or system organ class grouping. There were no deaths in this study.
Conclusion	Long-term treatment with OCA is effective and well tolerated.
Publications	Hirschfield 2012 (95) Kowdley 2014 (98) Kowdley 2015 (99)

Abbreviations: AE, adverse event; ALP, alkaline phosphatase; ALT, alanine transaminase; AST, aspartate transaminase; CRP, C-reactive protein; 6-ECDCA, 6-alpha-ethyl-chenodeoxycholic acid; ELF, enhanced liver fibrosis; GGT, gamma-glutamyl transpeptidase; HA, hyaluronic acid; IgM, immunoglobulin M; LTSE, long-term safety extension; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; P3NP, procollagen III N-peptide; PYE, patient years of exposure; TEAE, treatment-emergent adverse event; TGF-β, transforming growth factor-beta; TIMP-1, tissue inhibitor of metalloproteinase-1; TNF-α, tumour necrosis factor-alpha; ULN, upper limit of normal; VAS, visual analogue scale; WADD, weighted average daily dose. [†]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cholangitis. At the time of the design of the protocol of this study, the official name was primary biliary cirrhosis and, as such, this is reflected in this table.

Title	A study of INT-747 (6-ECDCA; OCA) in combination with ursodeoxycholic acid (UDCA) in subjects with primary biliary cirrhosis [¶]
Study design	A 3-month, international, multi-centre, randomised, double-blind, placebo-controlled, multi-dose, Phase II parallel group study of OCA in combination with UDCA in subjects with a proven or likely diagnosis of PBC.
Location	30 centres in 8 countries (USA [11 sites], Canada [6 sites], Germany [4 sites], UK [4 sites], The Netherlands [2 sites], Austria [1 site], France [1 site], Spain [1 site])
Primary objectives	 To assess the effect of OCA in combination with UDCA in subjects with proven or likely PBC on: ALP levels Safety.
Secondary objectives	 To assess the effect of OCA in subjects with proven or likely PBC on: Hepatocellular injury and liver function Disease-specific and general health symptoms Biomarkers of hepatic inflammation and fibrosis Plasma trough concentrations of OCA and its major known conjugates.
Interventions	Eligible subjects were randomised (1:1:1:1) to placebo, OCA 10 mg, OCA 25 mg or OCA 50 mg
Sample size and power calculation	A total of 222 subjects were screened and 165 randomised: placebo (n=38), OCA 10 mg (n=38), OCA 25 mg (n=48), and OCA 50 mg (n=41). With 35 patients per group, there was 80% power to detect an effect

Table 33: Study 747-202 (double-blind phase)

	size of 0.70, which translates to approximately a 10% mean greater reduction in ALP levels between groups.
Key inclusion/	Proven or likely PBC
exclusion criteria	Aged 18–70 years (18–75 years in the UK)
	 Both male and female subjects had to use one method of contraception unless surgically sterile (males and females) or postmenopausal (females)
Primary outcome	Percentage change in serum ALP from baseline to end of study
Secondary outcomes	Changes in serum ALP levels from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up
	Responder analyses of ALP response
	• Change in serum AST, ALT, GGT, serum albumin and conjugated (direct) bilirubin values from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up
	 Changes in CRP, non-esterified fatty acid, TNF-α & β, TGF-β, bile acids, glutathione, IgM, and osteopontin from Baseline to Day 85/end of treatment
	SF-36 and PBC-40 QoL questionnaires
	 Bile acid analysis and change in FGF-19 from baseline to Day 85/end of treatment
	Safety
Statistical methods	The percentage change from baseline to end of study was described with summary statistics. The two-sided Wilcoxon-Mann-Whitney test at the 5% level of significance was used for the primary endpoint. A hierarchical testing strategy (131) was proposed to account for multiple comparisons. The order of evaluation of statistical significance was as reported for study 747-201 in Table 31.
Results for 10 mg OCA v	s placebo
Patient population	Baseline and demographic characteristics were generally well- balanced across the treatment groups.
Primary endpoint	There was a clinically and statistically significant reduction in percentage change in ALP levels from baseline with OCA 10 mg compared with placebo (p<0.0001). The mean (SD) percentage change in ALP levels was –23.7 (17.8) U/L with OCA 10 mg versus – 2.6 (12.5) U/L with placebo.
Secondary endpoints	• There was a statistically significantly (p<0.0001) greater reduction in ALP from baseline to end of treatment relative to placebo in the ITT, completer [†] , and mITT [§] populations.
	• There were statistically significantly higher proportions of subjects who met the responder criteria (10, 20 and 40% reduction in ALP from baseline) in the OCA 10 mg treatment group compared with placebo
	 Improvement in other markers of liver injury was supported by statistically significant decreases in GGT (p<0.0001), ALT (p<0.0001), and AST (p=0.0031) from baseline to EOT
	Conjugated bilirubin levels were all within normal ranges at Baseline and all subsequent time points
	• Only two scores were clinically significant in the SF-36 survey, however, there were no consistent patterns or indication of dose-response relationships in any domain
	• Overall there were no changes observed in the PBC-40

Publications	Hirschfield et al, 2015 (101)
	These data provide a proof of concept that OCA may be a viable therapy for PBC.
	OCA treatment was safe and general well tolerated.
	Further, OCA treatment resulted in significantly higher rates of response which is shown to correlate with improved transplant-free survival.
Conclusion	Treatment with OCA in subjects with PBC who have an inadequate response to UDCA resulted in clinically and statistically significant improvements from placebo as assessed by changes in serum ALP, markers of cholestasis, inflammation, and hepatobiliary injury, such as bilirubin, GGT, ALT, and AST.
	• AEs other than pruritus resulting in discontinuation that occurred in two or more OCA treated subjects were nausea, peripheral oedema, increased ALT, and rash.
	• There were no deaths. Pruritus was the most common AE leading to study withdrawal, which occurred only with OCA, not with placebo
	 AEs were reported by 89% and 84% of subjects in the OCA 10 mg and placebo groups, respectively. Most AEs were mild or moderate in severity
	 Significant reductions in the median total endogenous bile acids were observed in all OCA treatment groups from Baseline to ET compared with no change in the placebo group
	questionnaire for general symptoms, cognitive function, and emotional/social domains

Abbreviations: AE, adverse event; ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; BL, baseline; CRP, C-reactive protein; 6-ECDCA, 6alpha-ethyl-chenodeoxycholic acid; EOT, end of treatment; FGF, fibroblast growth factor; GGT, gamma-glutamyl transferase; Ig, immunoglobulin; ITT, intention-to-treat; mITT, modified intention-to-treat; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QoL, quality of life; TGF, tumour growth factor; TNF, tumour necrosis factor.

[†]The completer population (n=136) included all randomised subjects who received at least one dose of investigational product based on the treatment group assignment and participated until the end of the 3-month, double-blind treatment period (i.e. Day 85). [§]The modified intention-to-treat (mITT) population (n=161) included all randomised subjects who received at least one dose of investigational product and had at least one post-baseline ALP evaluation taken ≤7 days after their last dose of investigational product. Subjects were analysed according to the treatment group to which they were randomized. The primary efficacy analysis was based on the mITT population. [¶]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cirrhosis and, as such, this is reflected in this table.

Table 34: Study 747-202 (long-term safety extension phase)

Title	A study of INT-747 (6-ECDCA; OCA) in combination with ursodeoxycholic acid (UDCA) in subjects with primary biliary cirrhosis [†]
Study design	A multi-centre, open-label study to assess the long-term treatment effects of OCA in subjects with PBC. Any subject who continued to meet the protocol requirements, regardless of the treatment group to which they were assigned during the double-blind phase, was eligible to participate in the study, including subjects who had not completed the double-blind phase.
Location	UK (2 sites), US (6 sites), Canada (3 sites), Spain (1 site), and Austria (1 site).
Primary objectives	Safety

Secondary	ALP levels as an assessment of efficacy
objectives	Hepatocellular injury and liver function
	Disease-specific and general health symptoms
Interventions	OCA. The starting dose of OCA was based on the dose of OCA or placebo received in the double-blind phase, or on the timing of entry into the LTSE phase:
	Placebo subjects: Start on 10 mg OCA OD
	 Active treatment subjects: Resume the same randomised dose of OCA, provided that the OCA dose was well tolerated during the double-blind phase
	 Double-blind completed subjects: If the last dose of OCA was taken 1 month before the start of the LTSE phase, start on 10 mg OCA OD
	 Subjects with significant pruritus or other AEs: Must not have received OCA for at least 1 month before starting the LTSE phase, start on a maximum dose of 10 mg OCA OD
	OCA was to be titrated from 10 mg to 25 mg to 50 mg OD no faster than at 8- week intervals, until one of the following occurred:
	ALP was in the normal range
	 The Investigator considered that an adequate therapeutic response was obtained
	AEs limited the administration of higher doses
	Exceptions were allowed, e.g. titration could have been performed more slowly and with smaller increments, and subjects could be titrated to doses above 50 mg OD if OCA 50 mg was well tolerated for ≥3 months.
Sample size and power calculation	No formal sample size calculation was performed, since any subject who continued to meet the protocol requirements of the double-blind phase was eligible to participate in the LTSE phase.
Key inclusion/ exclusion criteria	 Completion or participation (including early termination subjects) of treatment during the double-blind phase of the study
Primary efficacy outcome	• None
Secondary efficacy outcomes	 Absolute and percentage change from baseline in ALP at 3-month visits Absolute and percentage change from baseline in AST, ALT, GGT, and total and conjugated bilirubin at 3-month visits
	Absolute change from double-blind baseline to study completion for scale scores and summary measures
	Absolute change from double-blind baseline for PBC-40 domains at 6-month visits and study completion
Other efficacy	 Percentage of subjects that met response criteria Paris I, Pairs II, Toronto I, Toronto III and Mayo II at 6-month visits and study completion
variables	 Percentage of subjects with ALP within normal range and with a reduction from double-blind baseline by 10%, 20% and 40% at 6-month visits and study completion
	 Percentage of subjects with ALP values <1.0, 1.25, 1.5, 1.67, 1.76, 2.0 and 3.0x ULN who had ≥ the respective ALP value at double-blind baseline, at 6- month visits and study completion
	 Percentage of subjects with ALP <1.67x ULN, total bilirubin ≤ULN and ≥15% decrease in ALP who had ≥1.67x ULN or total bilirubin <uln at="" at<="" baseline,="" li=""> </uln>

	6-month visits and study completion
	 Absolute and percentage change from double-blind baseline of ELF score and its components (HA, P3NP and TIMP-1) at 6-month visits and study completion
	 Absolute and percentage changes from double-blind baseline in biomarkers of hepatic inflammation and fibrosis (CRP, TGF-β and IgM) at 6-month visits and study completion
	• Titration effects on ALP, AST, ALT, GGT and pruritus, measured by the change and percentage change from the pre-titration to the post-titration value (measured by VAS and 5-D scores for pruritus)
Primary	Pruritus (AEs and clinically significant interventions)
safety	TEAEs, including:
variables	 Overall incidence
	 Severity
	 Relationship to investigational product
	 Action taken
	 Outcome
Secondary	Vital signs
safety	Physical examinations
variable	Flectrocardiogram
	Laboratory results
Other safety	Study duration, exposure and titration
variable of	Compliance of investigational product
interest	 Status at the end of study, reasons for withdrawal
	Other concomitant medications
	Titration effects on pruritus
Statistical	All safety and efficacy analyses were performed on the safety population. No
methods	algorithm for missing data imputation was employed. For AEs, if causality was missing, it was assumed to be probably related to investigational product. Missing severity was not imputed
Results	
-	
Key efficacy endpoints	 ALP was lower at Month 3 and the reduced levels were maintained out to 1 year of LTSE dosing, suggesting a durable effect of OCA
	 AST and ALT levels remained within normal range; there was a decrease at 3 months, which was maintained out to 1 year of LTSE dosing
	GGT levels were reduced by 52.4% at Month 3 and 60.4% at Month 12 compared with baseline
	Total and conjugated bilirubin were within normal range and did not increase over 12 months of LTSE dosing

	•
Key safety endpoints	 SAEs were experienced by five subjects, none of which were considered related to investigational product. None of the subjects discontinued due to the SAE and all but one of the SAEs resolved without sequelae
	 The most frequently reported TEAE was pruritus, which was experienced by 87% of subjects during LTSE treatment, and fatigue, insomnia and upper respiratory tract infection were each reported by 13% of subjects
Publications	Hirschfield 2015 (101)

Abbreviations: AE, adverse event; ALP, alkaline phosphatase; ALT, alanine transaminase; AST, aspartate transaminase; CRP, C-reactive protein; 6-ECDCA, 6alpha-ethyl-chenodeoxycholic acid; ELF, enhanced liver fibrosis; GGT, gamma-glutamyl transpeptidase; HA, hyaluronic acid; IgM, immunoglobulin M; LTSE, long-term safety extension; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; P3NP, procollagen III N-peptide; SAE, serious adverse event; TEAE, treatment-emergent adverse event; TGF-β, transforming growth factor-beta; TIMP-1, tissue inhibitor of metalloproteinase-1; UDCA, ursodeoxycholic acid; ULN, upper limit of normal; VAS, visual analogue scale. [†]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cirrhosis and, as such, this is reflected in this table.

4.8 Subgroup analysis

4.8.1 Methodology and rationale

Subgroups based on demographics (age, age at diagnosis, gender, race, and geographical site) and baseline characteristics (baseline BMI group, ALP category, total bilirubin level, years since diagnosis, and UDCA use) were evaluated. Analyses were performed on the percentage of subjects achieving the primary endpoint, as well as two secondary efficacy endpoints: change from baseline in ALP and total bilirubin at Month 12. The analyses were performed on the ITT population.

Age, age at diagnosis, years since diagnosis at baseline, and gender were parameters of interest based on findings from a large cross-sectional study (UK-PBC cohort) with a total of 2,353 subjects with PBC (n=2,132 females, n=221 males). This study highlighted that PBC is not a uniform disease with uniform risks and impact, but a disease with high-and low-risk patients (1). The primary efficacy endpoint, ALP and total bilirubin, were evaluated based on these parameters of interest (1). The patient population randomised to POISE was diagnosed early (median age at diagnosis was 47.5 years) and, as such, was classified as a high-risk population who are more likely to suffer a more severe form of the disease compared with patients diagnosed later in life (1).

In addition, subgroup analyses were performed to evaluate any differences in patient demographics, disease characteristics, and efficacy results in subjects in the OCA titration group who titrated from 5 mg OCA to 10 mg OCA versus those that remained on 5 mg OCA following the Month 6 visit (Section 4.8.4.2). These analyses were performed predominately to determine the efficacy of OCA at both 5 mg and 10 mg, since the licensed dose is 5 mg OCA, with titration to 10 mg if required.

Results are also presented for the small subgroup of patients that took OCA as monotherapy (Section 4.8.4.3).

4.8.2 Patient characteristics

A total of 69 (97%) subjects from the OCA titration group completed 6 months of the study. Of these, 36 (52%) remained at 5 mg for the duration of the 12-month treatment period, and 33 (48%) did not meet the primary composite endpoint but tolerated investigational product and titrated to 10 mg for the last 6 months of the 12-month period.

Demographic and baseline characteristics for subjects from the two OCA titration subgroups who completed the Month 6 visit are summarised in Table 35.

 Table 35: Demographic and baseline characteristics by OCA titration subgroups: subset of ITT population

 Characteristic
 OCA titration subgroups

Characteristic	OCA titration subgroups			
	Remained at 5 mg n=36	Titrated to 10 mg n=33		
Age, years				
Mean (SD)	55.4 (10.7)	55.6 (10.2)		
Min, max	30, 81	29, 83		
<65 years, n (%)	31 (86)	29 (88)		
≥65 years, n (%)	5 (14)	4 (12)		
Sex, n (%)				
Male	4 (11)	1 (3)		
Female	32 (89)	32 (97)		
Race/ethnicity, n (%)				
White	36 (100)	30 (91)		
Non-white	0 (0)	3 (9)		
Region, n (%)				
Europe	22 (61)	23 (70)		
North America	12 (33)	7 (21)		
Australia	2 (6)	3 (9)		
BMI, kg/m²				
Mean (SD)	25.5 (4.3)	26.0 (5.6)		
Min, max	18, 37	18, 41		
<30 kg/m², n (%)	31 (86)	26 (79)		
≥30 kg/m², n (%)	5 (14)	7 (21)		
UDCA use at baseline, n (%)				
Yes	32 (89)	32 (97)		
No	4 (11)	1 (3)		
Baseline ALP, U/L				
Mean (SD)	306.7 (121.9)	348.1 (109.1)		

Characteristic	OCA titration subgroups			
	Remained at 5 mg n=36	Titrated to 10 mg n=33		
Min, max	187, 811	212, 566		
≤3x ULN, n (%)	28 (78)	22 (67)		
>3x ULN, n (%)	8 (22)	11 (33)		
Baseline total bilirubin, µmol/L				
Mean (SD)	9.6 (6.2)	11.1 (4.6)		
Min, max	2, 36	4, 22		
≤ULN, n (%)	35 (97)	30 (91)		
>ULN, n (%)	1 (3)	3 (9)		
Baseline conjugated bilirubin, µmol/L				
Mean (SD)	4.9 (6.0)	4.2 (2.2)		
Min, max	1.5, 35.2	1.5, 9.6		
Baseline albumin, g/L				
Mean (SD)	42.8 (3.0)	43.4 (3.1)		
Min, max	34, 49	33, 51		
≥LLN, n (%)	30 (83)	31 (94)		
<lln, (%)<="" n="" td=""><td>6 (17)</td><td>2 (6)</td></lln,>	6 (17)	2 (6)		
Baseline INR				
Ν	36	32		
Mean (SD)	1.0 (0.4)	1.1 (0.5)		
Min, max	0.9, 3.3	0.9, 3.5		
≤1.3, n (%)	35 (97)	30 (91)		
>1.3, n (%)	1 (3)	2 (6)		

Abbreviations: ALP, alkaline phosphatase; BMI, body mass index; INR, International standardised ratio; ITT, intention-to-treat; LLN, lower limit of normal; OCA, obeticholic acid; SD, standard deviation; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

4.8.3 Statistical information

The primary efficacy endpoint, the absolute and percentage change in ALP and bilirubin from baseline, and TEAEs were descriptively analysed for several subgroup populations. The ITT population was used for the efficacy endpoints, and the safety population for AEs. The cut-off for these analyses was either consistent with appropriate regulatory guidelines (e.g. age, gender, geographical region) or represented clinically meaningful divisions. Subgroup outcomes were calculated only if there were greater than 5 subjects per group at baseline.

Baseline subgroups of interest were:

- Age: <65 years, ≥65 years
- Age at diagnosis: <50 years, ≥50 years
- Sex: male, female

Company evidence submission template for obeticholic acid [ID785]

- Race: white, non-white
- BMI: <30 kg/m², ≥30 kg/m²
- ALP level: ≤3x ULN, >3x ULN
- ALP level: tertile (only for the absolute and percentage change in ALP)
- Bilirubin level: >ULN, ≤ULN
- Use of UDCA: Yes, No
- Years since diagnosis: ≤7.5 years, >7.5 years
- Geographic region: Europe, North America/Australia

Additional subgroup analyses were performed on the primary efficacy endpoint, absolute and percentage change in ALP from baseline, and the incidence of treatment-emergent pruritus using the following subgroups:

 Subjects in the OCA titration group who completed the Month 6 titration visit: Subjects who remained at OCA 5 mg, Subjects who titrated from OCA 5 mg to OCA 10 mg

4.8.4 Results

4.8.4.1 Analysis of the primary endpoint

In general, results from the subgroups were consistent with the observed effect in the overall ITT population in that greater and statistically significant improvements were observed in OCA-treated subjects compared with placebo-treated subjects. In terms of the primary composite endpoint, the effect of OCA was consistent independent of age at diagnosis, duration of PBC, or years since diagnosis. While the analyses according to sex were confounded by an imbalance in sample size with more females than males, a numerically better response in OCA-treated subjects was consistently demonstrated versus those receiving placebo.

Detailed results are provided in Appendix 4; however, a summary of the key results are listed below.

- **ALP and total bilirubin:** A larger proportion of subjects with lower baseline ALP or total bilirubin achieved the primary endpoint.
- BMI: Modestly lower responses in the composite endpoint and ALP were observed in subjects with a BMI ≥30 kg/m² compared with subjects with a BMI <30 kg/m².
- **UDCA treatment:** When evaluating only those subjects who were not taking UDCA treatment, there was a statistically significantly higher response for OCA-treated subjects compared with placebo-treated subjects.
- Geographical region:
 - There was a statistically significantly greater percentage of subjects treated with OCA at the European sites who met the primary endpoint versus placebotreated subjects. There was no statistically significant difference between subjects treated with OCA versus those receiving placebo for the North America/Australia sites.
 - The apparent diminished efficacy in North America/Australia in terms of the primary endpoint was caused by a notable mean reduction from baseline in ALP in the North America/Australia placebo group. This will diminish the response in the OCA groups compared with placebo. Post hoc sensitivity analyses using the

last measurement prior to first dose instead of the average of all measurements prior to Day 0 demonstrated a statistically significantly higher percentage of subjects achieving the primary composite endpoint across the 12-month period with OCA vs placebo. Taking these limitations into account, no discernible qualitative differences were observed in any of the demographic subgroup categories for the primary composite endpoint, ALP, or total bilirubin analyses.

4.8.4.2 Subgroup analyses – 5 mg versus 10 mg OCA

Patients in the POISE trial were diagnosed with PBC relatively early in their life and, as such, present a more at-risk patient population (1). Therefore, it is likely that a proportion of subjects will not meet the definition of a responder (ALP <1.67x ULN <u>and</u> total bilirubin \leq ULN <u>and</u> ALP decrease \geq 15%) after 6 months of treatment with 5 mg OCA, and thus will require their dose to be titrated to 10 mg for the remaining 6 months of the study. Therefore, the subgroup analyses evaluated and compared subjects that remained on 5 mg OCA compared with those that titrated to 10 mg OCA.

Demographics and baseline characteristics

The majority of demographic categories were similar between subjects who remained on OCA 5 mg versus subjects who titrated to OCA 10 mg. There were some key differences, however, which are listed below:

- A lower percentage of subjects who remained at 5 mg had a BMI ≥30 kg/m² compared with those who titrated to OCA 10 mg
- In the North American sites, a greater percentage of subjects remained at 5 mg, while a larger percentage of subjects titrated to 10 mg in the European sites
- Baseline ALP and total bilirubin levels were lower in subjects who remained at 5 mg OCA vs those who titrated to 10 mg.

PBC disease characteristics

A similar percentage of subjects had a history of pruritus, and the severity of the most recent pruritus event prior to randomisation was generally similar between the two subgroups. However, there were some key differences in disease characteristics at baseline between the subgroups:

- A greater percentage of subjects who remained at 5 mg (62%) had pruritus ongoing at baseline compared with subjects who titrated to 10 mg (42%)
- Of those who had pruritus at baseline, the severity of the baseline pruritus was greater for subjects who remained at 5 mg compared with those who titrated to 10 mg (moderate pruritus was 39% and 7%, respectively)
- A two-fold greater percentage of subjects in the 5 mg subgroup had a history of fatigue, compared with those subjects who titrated to 10 mg (70% versus 36%, respectively).

Efficacy

Primary endpoint: At Month 12, a greater proportion of subjects who remained at 5 mg OCA achieved the primary endpoint versus those that titrated to 10 mg OCA (53% vs 39%, respectively). However, since those that were titrated from 5 mg OCA to 10 mg OCA at Month 6 had previously failed to meet the primary endpoint, i.e. 0% of these

Company evidence submission template for obeticholic acid [ID785]

patients had met the primary endpoint at Month 6, an increase to 39% demonstrates that a significant incremental benefit can be gained with titration of OCA. In addition, it is likely that the benefit of OCA is underestimated in the group who did not up-titrate at Month 6. The SmPC advises all patients to up-titrate from 5 mg to 10 mg, if tolerated. Therefore, in clinical practice it is likely that patients will benefit further from the higher dose of OCA.

ALP, total bilirubin, and conjugated bilirubin: Absolute and percentage reductions from baseline in ALP were highly statistically significant for both OCA titration subgroups. These results support the results for the primary endpoint, in that for some subjects 5 mg OCA is sufficient for a clinically and statistically significant response, but for those subjects whose response is suboptimal following 6 months of treatment at 5 mg OCA, additional efficacy can be achieved by titrating to 10 mg. However, the absolute and percentage reductions in ALP were greater at Month 12 for subjects who titrated to 10 mg OCA compared with those who remained on 5 mg OCA.

The absolute mean reduction from baseline in total bilirubin was statistically significant at both Month 6 and Month 12 for subjects who titrated to 10 mg OCA but not for subjects who remained at 5 mg OCA. Baseline conjugated bilirubin was slightly higher in subjects who remained at 5 mg OCA compared with subjects who titrated to 10 mg OCA. Conjugated bilirubin levels were reduced from baseline for both subgroups at Month 6 and Month 12, with the absolute reduction at Month 6 for subjects who titrated to 10 mg OCA being statistically significant (p=0.0378).

4.8.4.3 Subgroup analyses – Subjects not taking UDCA

Only 11 subjects took OCA as monotherapy (five in the OCA titration group and six in the OCA 10 mg fixed dose group) and five subjects took placebo without UDCA. Of these, two subjects (40%) achieved the primary composite endpoint at Month 12 in the OCA titration group and one (17%) in the OCA 10 mg fixed dose group (Table 36). ALP levels decreased from baseline to Month 12 in both groups, and bilirubin levels decreased slightly in the OCA titration group but increased slightly in the OCA 10 mg fixed dose group. None of the results were statistically significant for either group vs placebo, which is not surprising given the small patient numbers in this analysis.

	Placebo group (n=5)	OCA titration group (n=5)		OCA 10 mg fixed dose group (n=6)	
	Endpoint	Endpoint	p-value vs placebo	Endpoint	p-value vs placebo
Achieving the primary composite endpoint, n (%)	0 (0)	2 (40)	0.0833	1 (17)	0.3173
Change from baseline to Month 12 in ALP, LS mean (SE)	21.3 (98.4)	–59.5 (76.2)	0.4702	–175.8 (71.5)	0.0895
Change from baseline to Month 12 in total bilirubin, LS mean (SE)	-1.2 (3.0)	-0.5 (2.1)	0.8505	1.4 (1.8)	0.4521

Table 36: Results for subjects not taking UDCA

Abbreviations: LS, least squares; OCA, obeticholic acid; SE, standard error; UDCA, ursodeoxycholic acid.

4.9 Meta-analysis

There was only one relevant Phase III trial providing data for the efficacy of OCA in PBC, therefore a meta-analysis was not conducted.

4.10 Indirect and mixed treatment comparisons

Indirect and mixed treatment comparisons were not conducted. The pivotal trial provides direct evidence of the effect of OCA compared with UDCA in patients who had an inadequate response to UDCA of OCA compared with no additional treatment for people who are unable to tolerate UDCA.

4.11 Non-randomised and non-controlled evidence

There are no non-RCTs relevant to this submission.

4.12 Adverse reactions

All safety data reported in this section are derived from the pivotal Phase 3 study, POISE, the methodology for which is described in Section 4.3. Supporting safety data were obtained from two Phase 2 trials, 747-201 and 747-202. The methodologies for these trials are described in Section 4.7.2 and they are summarised in Section 4.7.2.

4.12.1 Summary of adverse events reported in POISE

POISE included 216 subjects in the safety population who had PBC and were randomised to receive OCA 5 mg, OCA 10 mg, or placebo as described in detail in Section 4.3.

A summary of the AEs, based on the number of events occurring in \geq 5% of patients in either treatment group are detailed in Table 37 and Table 38, and a summary of pruritus-related AEs is detailed in Appendix 5.

Subjects, n (%)	Placebo n=73	OCA titration n=70	OCA 10 mg n=73
Any TEAE	66 (90)	65 (93)	69 (95)
Total number of TEAEs	452	471	467
Any treatment-related AE ⁺	38 (52)	42 (60)	54 (74)
Any SEAs	3 (4)	11 (16)	8 (11)
Total number of SAEs	8	15	11
TEAEs by severity			
Mild	29 (40)	16 (23)	19 (26)
Moderate	28 (38)	27 (39)	29 (40)
Severe	9 (12)	22 (31)	21 (29)

Table 37: Overview of adverse events: safety population (POISE)

Subjects, n (%)	Placebo n=73	OCA titration n=70	OCA 10 mg n=73	
Any TEAE leading to discontinuation	2 (3)‡	5 (7) [§]	8 (11) [¶]	
Discontinuation due to pruritus	0 (0)	1 (1)	7 (10)	
Number of deaths	0 (0)	1 (1)	0 (0)	

Abbreviations: AE, adverse event; CI, confidence interval; eCRF, electronic case report form; SAE, serious adverse event; TEAE, treatment-emergent adverse event.

[†] Includes any events determined to be "possibly", "probably", and "definitely" related. [‡]One subject was discontinued from study due to withdrawal of consent; [§]No subjects withdrew who titrated to OCA 10 mg. [¶]One subject experienced a TEAE of fatigue, which was recorded as a discontinuation on the AE eCRF; however, the subject remained in the study and study drug was not changed.

Table 38: Summary of TEAEs, severity of AEs, and treatment-related AEs occurring in ≥5% of subjects in either OCA treatment group

SOC/preferred term, n (%) [†]	Placebo N=73	OCA titration N=70	OCA 10 mg N=73			
TEAEs occurring in ≥5% of subjects in either OCA treatment group [‡]						
Skin and subcutaneou	is tissue disorders					
Pruritus	28 (38)	39 (56)	50 (68)			
Rash	3 (4)	3 (4)	4 (5)			
Eczema	0	4 (6)	2 (3)			
General disorders and	l administration site cor	nditions				
Fatigue	10 (14)	11 (16)	17 (23)			
Oedema peripheral	2 (3)	2 (3)	5 (7)			
Pyrexia	1 (1)	0	5 (7)			
Infections and infestations						
Nasopharyngitis	13 (18)	17 (24)	13 (18)			
Upper respiratory tract infection	8 (11)	4 (6)	4 (5)			
Urinary tract infection	8 (11)	4 (6)	4 (5)			
Influenza	4 (5)	5 (7)	4 (5)			
Bronchitis	0	4 (6)	1 (1)			
Sinusitis	0	1 (1)	4 (5)			
Gastrointestinal disorders						
Nausea	9 (12)	4 (6)	8 (11)			
Diarrhoea	8 (11)	2 (3)	8 (11)			
Constipation	4 (5)	5 (7)	5 (7)			
Abdominal pain upper	5 (7)	5 (7)	4 (5)			

SOC/preferred term, n (%) [†]	Placebo N=73	OCA titration N=70	OCA 10 mg N=73			
Gastroesophageal reflux disease	4 (5)	2 (3)	4 (5)			
Dyspepsia	8 (11)	4 (6)	0			
Abdominal discomfort	1 (1)	5 (7)	0			
Musculoskeletal and c	connective tissue disord	lers				
Arthralgia	3 (4)	4 (6)	7 (10)			
Back pain	8 (11)	4 (6)	4 (5)			
Nervous system disor	ders					
Headache	13 (18)	12 (17)	6 (8)			
Respiratory, thoracic,	and mediastinal disorde	ers				
Cough	5 (7)	4 (6)	6 (8)			
Oropharyngeal pain	1 (1)	5 (7)	6 (8)			
Injury, poisoning, and	Injury, poisoning, and procedural complications					
Procedural pain	1 (1)	4 (6)	1 (1)			
Fractures	3 (4)	2 (3)	4 (5)			
Cardiac disorders						
Palpitations	1 (1)	2 (3)	5 (7)			
Eye disorders						
Dry eye	4 (5)	2 (3)	4 (5)			
Endocrine disorders						
Hypothyroidism	1 (1)	4 (6)	1 (1)			
Incidence of TEAE by	maximum severity, n (%	b)				
Mild	29 (40)	16 (23)	19 (26)			
Moderate	28 (38)	27 (39)	29 (40)			
Severe	9 (12)	22 (31)	21 (29)			
Treatment-related AEs	in ≥5% of subjects in a	ny OCA treatment grou	p, n (%) [§]			
Skin and subcutaneous tissue disorders						
Pruritus	27 (37)	35 (50)	48 (66)			
General disorders and	General disorders and administration site conditions					
Fatigue	8 (11)	6 (9)	6 (8)			
Gastrointestinal disor	ders					
Nausea	4 (5)	3 (4)	4 (5)			
Abbreviations: AE advers	e event: N total number of s	ubjects: n number of subject	ts experiencing an event			

Abbreviations: AE, adverse event; N, total number of subjects; n, number of subjects experiencing an event, OCA, obeticholic acid; SOC, systems organ class; TEAE, treatment-emergent adverse event.

† At each level of summation, subjects reporting >1 AE are counted only once.

§ treatment-related AEs include all events reported as "possible", "probable", or "definite" relationship to study drug.

Company evidence submission template for obeticholic acid [ID785]

[‡] a TEAE is defined as any event that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product. At each level of summation, subjects reporting >1 AE are counted only once using the highest severity.

4.12.2 Safety overview

4.12.2.1 Summary of key safety

OCA treatment was safe and generally well tolerated. Pruritus was the most common TEAE with a higher incidence reported in OCA treatment groups (OCA titration [56%] and OCA 10 mg [68%] versus the placebo [38%]). While pruritus was the most commonly reported TEAE, pruritus is a symptom of the disease, and therefore OCA may transiently exacerbate this feature of the disease, but patients will already be familiar with the adverse event. However, based on the rate of treatment discontinuations due to pruritus, treatment was better tolerated in subjects treated with OCA who initiated treatment at 5 mg and titrated up to 10 mg after 6 months based on clinical response. Additionally, the severity of pruritus was mitigated by this dosing strategy compared to starting at OCA 10 mg. There were no discontinuations due to pruritus in the placebo group, and in the majority of subjects who experienced pruritus in this group, the maximum severity of pruritus was mild or moderate. The incidence of TEAEs assessed as related, severe, serious, or leading to study discontinuation was higher in subjects treated with OCA, compared with placebo. With the exception of SAEs, these imbalances were predominantly attributed to pruritus.

Total TEAEs

The overall incidence of TEAEs was generally similar between OCA and placebo treatment groups. A total of 66 subjects (90%) from the placebo group reported 452 TEAEs, 65 subjects (93%) from the OCA titration group reported 471 TEAEs, and 69 subjects (95%) from the OCA 10 mg group reported a total of 467 TEAEs.

TEAEs occurring in ≥5% of subjects in either of the OCA groups

TEAEs that occurred with an incidence of ≥5% and were reported more frequently in either of the OCA treatment groups compared with placebo included pruritus, rash, eczema, fatigue, pyrexia, peripheral oedema, nasopharyngitis, influenza, bronchitis, sinusitis, diarrhoea, constipation, abdominal discomfort, arthralgia, cough, oropharyngeal pain, procedural pain, fractures, palpitations, and hypothyroidism.

TEAEs that occurred at an incidence of ≥5% and were reported with an incidence of >3% more frequently in subjects receiving OCA compared with placebo were limited to pruritus, fatigue, hypothyroidism, procedural pain, oropharyngeal pain, arthralgia, abdominal discomfort, sinusitis, peripheral oedema, pyrexia, palpitations, eczema, bronchitis, and nasopharyngitis.

Treatment-related AEs

As expected based on prior experience with OCA treatment in patients with PBC, the most common related TEAE was pruritus. In all treatment groups, the majority of pruritus AEs were considered related to investigational product. The incidence and number of subjects with related TEAEs of pruritus was 27 subjects (37%) in the placebo group, 35 subjects (50%) in the OCA titration group, and 48 subjects (66%) in the OCA 10 mg group. Fatigue and nausea were the only other related TEAE that occurred at an incidence \geq 5%; however, these events were balanced between placebo and OCA treatment groups.

4.12.2.2 Safety parameters of special interest

Pruritus

A higher incidence of pruritus was observed in the OCA titration and OCA 10 mg groups when compared with placebo (60%, 74%, and 37%, respectively). Most pruritus events were of mild or moderate severity and resolved during the treatment period. Starting at OCA 5 mg and titrating to 10 mg OCA was associated with improved tolerability versus starting at 10 mg OCA. In the majority of cases, subjects remained in the study despite varying pruritus severity.

Hepatic-related effects

There was no dose-dependent trend in the incidence of hepatic-related events. In agreement with the improvements in hepatic indices, hepatic safety was maintained during the course of the study in both OCA treatment arms.

Lipid-related effects

There was an early decrease in high-density lipoprotein cholesterol (HDLc), which later stabilised, and all values remained within the normal range. In addition, there was a modest and transient increase in low-density lipoprotein cholesterol (LDLc) in OCA-treated subjects; however, there were no differences between groups after 12 months of treatment and hypercholesterolemia doesn't appear to increase the risk of cardiovascular disease in PBC patients (47). There was a greater number of subjects in the 10 mg OCA groups that experienced a shift from normal HDLc at baseline to a lower level after 12 months when compared with placebo. However, there were no treatment differences for lipid-related AEs.

Cardiovascular-related TEAEs

There were no treatment differences observed for cardiovascular-related AEs or SAEs.

Deaths and other SAEs

One death occurred during the double-blind phase (cardiac failure) in a subject from the OCA titration group who had an extensive history of cardiovascular conditions including cardiac failure. In total, three subjects (4%), 11 subjects (16%), and 8 subjects (11%) experienced SAEs in the placebo, OCA titration, and OCA 10 mg groups, respectively. None of the SAEs were considered to be related to study drug.

AEs leading to study discontinuation

A greater proportion of subjects in the OCA titration and OCA 10 mg groups experienced TEAEs which lead to study discontinuation when compared with placebo (7%, 11%, and 3%, respectively). The majority of these were attributed to pruritus and occurred in the OCA 10 mg group (10%). One subject (1%) discontinued due to pruritus in the OCA titration group versus none in the placebo group.

Safety laboratory parameters

No clinically meaningful differences between treatment groups were observed for any safety laboratory parameters.

4.12.2.3 Other safety evaluations

- As may be expected, minimal changes in MRS and MELD scores were observed in either treatment group indicating overall stable disease state over the course of the 12-month treatment period.
- Generally, reductions in DEXA scans results were observed from baseline to month 12 in all treatment groups. However, the decrease in mean femoral neck Tscore from baseline to month 12 was significantly less in the OCA-treated groups compared with placebo. No significant differences from baseline or between treatment groups were seen in lumbar or femoral Z-scores.
- Overall, no clinically meaningful mean changes from Baseline to Month 12 in body weight or BMI were observed in any of the treatment groups and no clinically meaningful differences in vital signs or ECGs were noted.

4.12.3 Interim safety results from the long-term safety extension of POISE (Study 747-301)

Currently, there is an ongoing 5-year LTSE of the Phase 3 POISE study. As such, interim results are provided for the first 12 months of this study. For full study details, see Section 4.7.2.1.

Overall, continued treatment with OCA was safe and generally well tolerated with longer term treatment. During the LTSE, 10 subjects in the safety population discontinued from the study, 4 (2%) of these discontinuations were due to TEAEs, which was a comparable rate to that observed in the double-blind period.

The AE profile observed during the LTSE was consistent with that observed for OCA treatment in the double-blind period. Consistent with the double-blind period and the disease state in general, pruritus was the most common TEAE as assessed by crude and exposure adjusted incidence. Most pruritus events were mild to moderate in severity, with an increase in the occurrence of severe pruritus, as assessed by exposure adjusted incidence noted with increasing OCA dose with the highest rates observed in subjects who titrated to >10 mg OCA.

In the double blind phase, the use of a titration strategy mitigated subject discontinuations due to pruritus. The use of a titration strategy was similarly beneficial in the LTSE phase, with only one subject who enrolled in the LTSE discontinuing due to pruritus.

All SAEs occurring to date in the LTSE were considered unrelated or unlikely to be related to OCA. With the exception of one SAE that was fatal, all SAEs resolved with or without sequelae. There was no dose related trend or system organ class grouping of SAEs.

Lipid effects in the LTSE were consistent with the double-blind period. With continued OCA treatment, there was no change from baseline in LDL while the decrease in HDL was sustained.

4.13 Interpretation of clinical effectiveness and safety evidence

4.13.1 Principal (interim) findings from the clinical evidence highlighting the clinical benefits and harms of the technology

The double-blind phase of POISE assessed the efficacy, safety, and tolerability of OCA with or without UDCA (depending on tolerability) compared with placebo with or without UDCA (depending on tolerability) in patients with an adequate response to, or who are intolerant to, UDCA. Of 316 patients screened, 217 were randomised and 216 were included in the ITT population. The patient population was representative of patients with PBC in the UK, and baseline characteristics were generally well balanced across treatment arms.

The primary endpoint of POISE was met. Treatment with OCA resulted in clinically and statistically significant improvements from placebo as assessed by a composite ALP and total bilirubin endpoint (ALP <1.67x ULN [200 U/L], bilirubin \leq ULN [20 µmol/L] and \geq 15% decrease from baseline in ALP). Almost five times as many patients had response to treatment in the OCA groups 47% and 46% of patients in the 10 mg OCA fixed dose and the OCA titration groups, respectively) than in the placebo group (10% of subjects; p<0.0001). Furthermore, as secondary endpoints, the effect of OCA on several other independent biochemical response criteria incorporating ALP, AST, bilirubin and albumin (i.e. Paris I, Paris II, Toronto II, and Mayo II), all of which are shown to correlate with improved prognostic outcomes, were supportive of the potential effect of OCA in inhibiting the progression of PBC and improving clinical outcomes.

Patients treated with OCA were over 30 times more likely to achieve a reduction in ALP ≥40% over 12 months compared with placebo (odds ratios of 34.7 and 43.0 for the OCA titration and OCA 10 mg fixed dose groups, respectively). Mean ALP was statistically significantly reduced from as early as 3 months compared with placebo, which maintained stable ALP. Total and conjugated bilirubin levels increased with placebo, but decreased or remained stable for both OCA treatment groups, and the results were statistically significant at 6 and 12 months. In addition, GGT, ALT and AST were statistically significantly reduced with OCA compared with placebo, which maintained or increased levels.

Other secondary endpoints, including several biochemical markers of FGF-19, CK-18, total endogenous bile acid, CRP, and TNF- α demonstrated clinically and statistically significant improvements compared with placebo that were sustained during the 12-month period. These improvements, paralleled by the improvement in liver biochemistry due to FXR-mediated effects on bile acid homeostasis, are further supportive of a beneficial disease-modifying effect of FXR activation over at least a 12-month period.

Subgroup analyses demonstrated that the effect of OCA on achieving the primary composite endpoint and changes in ALP and total bilirubin were independent of age at diagnosis, duration of PBC, and baseline ALP. In general, the baseline and demographic subgroup analyses were consistent with the observed effect in the overall population, in that OCA-treated subjects had more favourable outcomes than subjects receiving

placebo. For subjects receiving BAS, efficacy was modestly attenuated in subjects receiving OCA 5 mg but was not affected in subjects receiving OCA 10 mg.

Based on the clinical response and adverse event profile, initiating subjects on OCA 5 mg and titrating to 10 mg appears to be an appropriate dosing strategy. For some subjects, the composite endpoint was achieved with 5 mg OCA, however, an additional incremental benefit was gained by titrating to 10 mg OCA in those that failed to achieve an optimal response within 6 months of initiating treatment, and this is therefore the recommended dosing strategy for all patients according to the summary of product characteristics (SmPC).

An ongoing long-term safety extension (LTSE) of POISE has shown continuing efficacy of OCA out to a further 12 months in terms of ALP and bilirubin levels. Evidence from two Phase 2 studies and their LTSEs provide further support for the efficacy of OCA observed in POISE.

OCA was generally well tolerated, with pruritus being the most commonly reported AE. However, it is important to consider that pruritus is a common symptom of PBC; 63% of patients in POISE had a history of pruritus. Therefore, patients and their clinicians are typically familiar with the condition and its management.

In conclusion, OCA for the treatment of adults with PBC who have failed treatment with or are intolerant to UDCA is an effective and tolerable treatment option. OCA (5 mg and 10 mg) resulted in clinically and statistically significant improvements in a range of evidence-based disease-related prognostic factors, which is expected based on predictive modelling to lead to a reduced risk of liver transplant and/or death. In addition, markers of inflammation and liver dysfunction were reduced.

4.13.2 Strengths and limitations of the clinical evidence base for the technology

4.13.2.1 Strengths

- 1. Design features of POISE
 - POISE was a high-quality, multi-centre, multi-national randomised, controlled, double-blind study
 - It is the largest clinical trial in PBC to date, including 216 patients
- 2. Representativeness of patient population and generalisability to UK clinical practice
 - The patient population in POISE was validated as representative of the population with PBC in the UK by clinical expert opinion
 - It included nine sites in the UK:
 - Seven in England (London, Oxford, Newcastle-Upon-Tyne, Birmingham, Nottingham, Bristol, and Manchester)
 - Two in Scotland (Larbert and Dundee)

- 3. Value of clinical outcomes observed with OCA
 - There is currently no licensed or effective treatment option for patients who have an inadequate response to, or are intolerant to, UDCA
 - These patients are at significantly increased risk of complications, the requirement of liver transplantation, HCC and death
 - Therefore, an effective treatment option for these patients is likely to make a substantial difference to the patients' health and quality of life, and also is likely to avoid the costly downstream events in PBC such as the management of complications and liver transplant
 - OCA is the first new and effective treatment for PBC in almost 20 years, and provides a novel, innovative mechanism of action targeting the FXR receptor
 - ALP is the key biomarker of PBC, with elevated bilirubin indicative of end-stage disease
 - Mean ALP decreased significantly with OCA (in both treatment groups), but remained stable at >1.67x ULN (200 U/L) in the placebo group
 - A patient-level meta-analysis (15) found that attaining an ALP <1.67x ULN is associated with a statistically significantly reduced risk of disease progression over the subsequent 10 years
 - Mean bilirubin decreased or remained stable with OCA (in both treatment groups), but increased in the placebo group to >0.5x ULN (20 μmol/L)
 - An increase in risk for liver transplantation or death starts at levels of bilirubin >0.5x ULN (47)
 - Since the vast majority (93%) of patients were taking UDCA in the placebo group, this shows clear progression of disease in these patients on the only currently licensed treatment option for PBC, correlating to an increased risk of complications, the requirement for liver transplantation, HCC and death
 - There is a substantial unmet need for an effective treatment option for these patients
 - It is likely that the clinical benefit of OCA is actually underestimated in the OCA titration group of POISE compared with that proposed for clinical practice
 - It is recommended that patients initiate treatment at 5 mg and up-titrate to 10 mg at 6 months if tolerated
 - However, in POISE, patients in the OCA titration group were only up-titrated from 5 mg to 10 mg OCA if they did not reach the primary endpoint criteria for response
 - Therefore, further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA

4.13.2.2 Limitations

The main limitation of POISE was the necessity to use surrogate endpoints. This is due to the rarity, slow rate of progression in most patients, and chronic nature of PBC.

However, ALP and bilirubin levels are key factors for determining patients' prognoses and are included in a wide range of PBC patient scoring systems. Elevated ALP levels

Company evidence submission template for obeticholic acid [ID785]

are used as a marker for cholestasis in patients with PBC throughout the progression of the condition and are a key component used in the diagnosis of PBC in European guidelines (33).

Both ALP and bilirubin are commonly used in an array of algorithms to predict survival and transplant outcomes in patients with PBC (11-13, 16, 125, 126). Data demonstrates that in patients with PBC who receive treatment with UDCA, a reduction in ALP is associated with a general reduction in the risk of death and an increase in transplant-free survival (16, 125, 126). For example, Kuiper et al demonstrated that the normalisation of ALP and/or bilirubin was associated with an overall increase in survival outcomes, irrespective of the degree of disease severity (13). More recently, a patient-level metaanalysis study by Lammers et al demonstrated the significant prognostic value of ALP and bilirubin levels on long-term clinical outcomes in patients with PBC (15). Generally, the study demonstrated that lower levels of ALP and bilirubin strongly correlated with a longer transplant-free survival in a log linear manner. Of patients with ALP $\leq 2.0x$ ULN, 84% survived for 10 years compared with 63% of those with levels >2.0x ULN, and of patients with bilirubin ≤1.0x ULN. 86% survived for 10 years versus just 41% in patients with bilirubin >1.0x ULN (15). When evaluating ALP alone, attaining an ALP <1.67x ULN is associated with a statistically significantly reduced risk of disease progression over the subsequent 10 years, and is therefore a key clinical goal when evaluating the effectiveness of treatment in patients with PBC (18).

In addition, bilirubin is an independent predictor of PBC prognosis and disease progression, and was used as an endpoint to support the marketing authorisation application of UDCA in the EU. Many PBC studies have shown that increased bilirubin levels are an independent predictor of a poor prognosis (13, 88, 132-136), with increased survival associated with levels <ULN. As noted in the American Association for the Study of Liver Disease (AASLD) Endpoints Conference (137) bilirubin levels are an important serum marker of survival and are also a key criterion of the Model for End Stage Liver Disease (MELD) score that is used to manage US liver transplant programs and others globally (138).

Importantly, changes in ALP and bilirubin occur during different stages of PBC disease progression, with ALP changes occurring during the early/asymptomatic stages whereas changes in bilirubin occur during the later/decompensated stages of the disease. Therefore, the prognostic value of these surrogate markers is significantly increased when combined and evaluated together (15). Momah et al from the Mayo clinic evaluated different biochemical thresholds versus a combination of clinical outcomes such as varices, ascites, death, or liver transplantation in a cohort of UDCA-treated patients, and concluded that combining ALP and bilirubin (ALP <1.67x ULN and bilirubin ≤1 mg/dL) was the most discriminating of the algorithms they evaluated (17).

The validity of the combined surrogate endpoint proposed by Momah et al has since been supported by a meta-analysis of patient-level data from the Global PBC study (15). The Global PBC study comprised data from 4,845 patients from 15 centres in North America and Europe, and is the largest international database of PBC patients to date. The Global PBC study demonstrated that the combined use of ALP and bilirubin levels provided greater prognostic predictability versus either component alone (15). Therefore, the use of both ALP and bilirubin for evaluating patients with PBC represents an evidence-based, clinically meaningful surrogate endpoint for use in clinical studies (15).

4.14 Other ongoing studies

There is a long-term phase 3b study (747-302, COBALT) currently recruiting participants, details of which are summarised in Table 39. Results of COBALT will establish whether OCA has an impact on long-term clinical outcomes and to what degree, and results are expected to validate the surrogate endpoints used in POISE.

There is also an ongoing phase 2 study that is summarised in Table 40. In addition, the long-term safety extension of the phase 2 study 747-201 (as described in Section 4.7.2.2) is ongoing, with a planned maximum exposure time of approximately 6 years and 3 months.

Title	Phase 3 study of obeticholic acid evaluating clinical outcomes in patients with primary biliary cirrhosis [§] (COBALT)				
Trial numbers	747-302, NCT02308111				
Trial design	Phase 3b, double-blind, randomised, placebo-controlled, multi- centre study evaluating the effect of obeticholic acid on clinical outcomes in patients with PBC				
Location	Up to 170 sites internationally, in Argentina, Australia, Austria, Belgium, Bulgaria, Canada, Chile, Denmark, Estonia, Finland, Former Serbia and Montenegro, France, Germany, Hungary, Israel, Italy, Lithuania, Mexico, Netherlands, New Zealand, Poland, Republic of Korea, Spain, Sweden, Switzerland, Turkey, the UK and the US				
Estimated enrolment	350				
Duration	Time to accrue approximately 121 primary endpoint events, estimated to be approximately 8 years, with an expected minimum follow-up time of approximately 6 years				
Interventions	 Subjects will be randomised 1:1 (stratified by standard treatment with UDCA [yes/no] and baseline liver function) to either: OCA 5 mg OD for 3 months and then titrating up to 10 mg OD for the remainder of the trial, based on tolerability; or Placebo OD 				
Primary outcome measures	 Composite endpoint of any of the following: Death Liver transplant MELD[†] score ≥15 Uncontrolled ascites, defined as diuretic resistant ascites requiring therapeutic paracentesis at a frequency of at least twice in a month Hepatocellular carcinoma confirmed by two complementary imaging modalities Hospitalisation (defined as a stay of ≥24 hours) for new onset or recurrence of any of the following: 				

Table 39: Details of the ongoing trial 747-302 (COBALT)

	 Variceal bleed
	 Encephalopathy, as defined by a West Haven score of ≥2
	 Spontaneous bacterial peritonitis confirmed by diagnostic paracentesis
Secondary outcome	First occurrence of each of the following:
measures	o Death
	 Liver transplant
	 MELD score >15
	 Uncontrolled ascites (defined as diuretic resistant ascites requiring therapeutic paracentesis at a frequency of at least twice in a month)
	 Hepatocellular carcinoma confirmed by two complementary imaging modalities
	 Hospitalisation (defined as a stay of ≥24 hours) for new onset or recurrence of any of the following:
	 Variceal bleed
	 Encephalopathy, as defined by a West Haven score of ≥2
	 Spontaneous bacterial peritonitis confirmed by diagnostic paracentesis
	 Changes from baseline in bilirubin, AST, ALT, ALP and GGT as markers of liver biochemistry
	 Changes from baseline in IgM, CRP, TNF-α and FGF-19 as markers of inflammation
	 Changes in CK-18, ELF test, and transient elastography as markers of liver fibrosis
Key eligibility criteria	Aged ≥8 years
	 Definite or probable PBC diagnosis, as demonstrated by the presence of ≥2 of the following diagnostic factors:
	 History of elevated ALP levels for at least 6 months prior to Day 0
	 Positive anti-mitochondrial antibody titer or if negative or in low titer (<1:80) PBC-specific antibodies
	 Liver biopsy consistent with PBC
	• Mean total bilirubin >ULN and ≤3x ULN or an ALP >5x ULN

Abbreviations: ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; CK-18, cytokeratin-18; CRP, C-reactive protein; ELF, enhance liver fibrosis; FGF-19, fibroblast growth factor-19; GGT, gamma-glutamyl transferase; IgM, immunoglobulin M; MELD, Model for End-Stage Liver Disease; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; TNF-α, tumour necrosis factor-α; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

[†]MELD is a scoring system for assessing the severity of chronic liver disease, where the higher the score, the more severe the disease. [§]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cholangitis. At the time of the design of the protocol of this study, the official name was primary biliary cirrhosis and, as such, this is reflected in this table.

In addition, a Phase 2 study is ongoing, which is summarised in Table 40.

Table 40:	Details o	of the	ongoing	trial	747-205
	Detans		ongoing	unai	141-205

Title	Phase 2 study on effects of OCA on lipoprotein metabolism in subjects with PBC	
Trial numbers	747-205, NCT01865812	
Trial design	Phase 2, open-label, single-arm, multi-centre study	
Location	6 sites in the US	
Estimated enrolment	25	
Duration	8 weeks	
Interventions	OCA 10 mg OD	
Primary outcome measures	Change from baseline in HDL metabolism, assessed by measuring HDL cholesterol concentration, HDL particle size and number	
Secondary outcome measures	Change from baseline in lipoprotein metabolism, assessed by measuring:	
	 Concentrations of total cholesterol and triglycerides 	
	 LDL and VLDL cholesterol concentrations, particle size and number 	
	 Concentrations of ApoA, ApoB, ApoE and LP(a) 	
	Change from baseline in reverse cholesterol transport, assessed by measuring:	
	 HDL capacity to accept cholesterol measured by LCAT and CETP activity 	
	 Pre-β1 HDL concentration 	
	 Macrophage cholesterol efflux 	
	Pharmacokinetic parameters of OCA and OCA conjugates	
Other outcome measures	Fasting levels of OCA and conjugates	
	Change from baseline in FGF-19	
	Change from baseline in lipoprotein X	
	Markers of inflammation including CRP, GlycA and GlycB	
Key eligibility criteria	Aged ≥8 years	
	• Definite or probable PBC diagnosis as demonstrated by the presence of ≥2 of the following diagnostic factors:	
	 History of elevated ALP levels for at least 6 months 	
	 A positive anti-mitochondrial antibody titer or if negative or in low titer (<1:80) PBC antibodies 	
	 Liver biopsy consistent with PBC 	

Abbreviations: ALP, alkaline phosphatase; Apo, apolipoprotein; CETP, cholesterol ester transfer protein; CRP, C-reactive protein; FGF-19, fibroblast growth factor-19; HDL, high density lipoprotein; LCAT, lecithincholesterol acyltransferase; LDL, low density lipoprotein; LP, lipoprotein; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; VLDL, very low density lipoprotein.

5 Cost effectiveness

Summary

- The model presented for OCA is a decision-analytic model that reports cost-effectiveness in terms of incremental cost per QALY.
- The model evaluates the economic consequences of OCA 5–10 mg (titrated dose) alone (in the case of UDCA intolerant PBC patients) and adding OCA 5–10 mg OD to 15.4 mg/kg UDCA (in the case of UDCA inadequate responders). In other words, the model examines the cost-effectiveness of OCA titration versus placebo in UDCA intolerant patients, and OCA + UDCA titration versus UDCA monotherapy in UDCA tolerant patients.
- The results presented in this submission use the list price of OCA. However, OCA will be
 offered with a PAS, which will reduce the price. Results of economic analyses using the PAS
 price are presented in the accompanying PAS template, as requested by NICE, and are more
 reflective of the true cost-effectiveness of OCA.
- The base case results using the list price of OCA gave an ICER of **Constant of IDCA** intolerant patients, and an ICER of **Constant**.16 in patients with an inadequate response to UDCA.
- Patients treated with OCA were found to have an 84% lower chance of undergoing liver transplant compared with patients treated with UDCA, which approaches the risk observed in the general population.

5.1 Published cost-effectiveness studies

5.1.1 Identification of studies

5.1.1.1 Overview and review question

A systematic review was conducted to identify all relevant PBC studies relating to cost effectiveness. The specific review question was:

• "What modelling techniques have been used previously to conduct economic evaluations for the treatment of PBC?"

5.1.1.2 Search methodology

Studies of interest were identified by simultaneously searching the electronic databases shown in Table 41 with no restrictions on date or language of publication. Searches were conducted in September 2014 using the following interfaces:

- EMBASE (which also covers Medline[®] and Medline[®] In-Process)
- The Cochrane Library (which covers the Cochrane Database of Systematic Reviews, Database of Abstracts of Reviews of Effects, Health Technology Assessment, NHS Economic Evaluation Database)
- EBSCO host (which covers EconLit, Health Economic Evaluations Database)

Table 41: Databases searched and interfaces u	sed in the cost effectiveness systematic
review	-

Database	Interface
Embase 1966 to 2014	Embase
Medline 1966 to 2014	Embase
Medline® In-Process 1966 to 2014	Embase
Cochrane Database of Systematic Reviews 1996 to 2014	The Cochrane Library
Database of Abstracts of Reviews of Effects 1994 to 2014	The Cochrane Library
Health Technology Assessment 1989 to 2014	The Cochrane Library
NHS Economic Evaluation Database 1968 to 2014	The Cochrane Library
EconLIT with Full Text 1961 to 2014	EBSCO host
Health Economic Evaluations Database 1990 to 2014	EBSCO host

Appendix 6 shows the complete search strategies used. The searches included terms for free text and keywords (Medical Subject Heading [MeSH] and Emtree terms) through the use of Boolean combination techniques. Searches were not restricted by study intervention or comparator to ensure all studies in this population were identified. The Cochrane Library and EBSCO host interfaces were searched using terms for the population only to broaden the results.

A grey literature search was performed to include additional studies that had not been identified by the search strategy. References included for the review had to meet the pre-specified inclusion/exclusion criteria shown as a PICOS table in Appendix 6.

The search identified 184 titles/abstracts. After screening, 167 references were excluded, leaving 17 references for full-text evaluation. Two further studies were identified from the review of grey literature. Following full-text evaluation, four references met the inclusion criteria. All of these four studies were deemed relevant to the review and to be used for data extraction. The PRISMA diagram for the systematic review is shown in Figure 23.

Figure 23: PRISMA diagram illustrating the flow of the cost and healthcare resource use systematic review



5.1.2 Description of identified studies

5.1.2.1 Overview of all included studies

The four relevant cost-utility analyses are summarised in Table 46. When comparing the four identified economic evaluations, the following salient points are noted:

- None of the analyses are economic evaluations of OCA.
- All identified economic evaluations incorporated liver transplant and death as key outcomes, with Boberg et al (79) and Ratcliffe et al (139) also allowing the possibility of re-transplantation in their patient pathway.
- In contrast to the other included papers, Boberg et al (79) was the only analysis identified to capture long-term outcomes. In addition to differences in liver transplant and death, their analysis also incorporated long-term changes in complications associated with PBC, e.g. ascites. The remaining papers focused on short-term horizons ranging from 27 months to 12 years, typically investigating outcomes associated with treatment strategies administered around the time when transplant may be required.

- Of the four economic evaluations included for review, three of these (Ratcliffe et al, Longworth et al, and Boberg et al) applied modelling techniques. The remaining article estimated costs and outcomes based on observed data. The three modelling studies applied varied techniques. The value of the complex discrete event simulation applied by Ratcliffe et al (139) is unclear, as the analysis for patients who did not receive transplants incorporated outcomes for patients who may not be eligible for transplantation. The approach taken by Longworth et al (140) estimated outcomes in the active treatment arm using observed data, with the comparator arm modelled using relative effect data. Boberg et al (79) simulated outcomes in the non-UDCA arm in a similar fashion; however, they simulated outcomes beyond those observed in data. This extrapolation beyond observed outcomes was conducted through the application of parametric survival analysis; however, details of curve fit and justification behind the selection of the Weibull curve are not reported.
- Only one of the identified papers reported QALYs (Longworth et al), with all other papers expressing effectiveness in real units. QoL analyses reported by Longworth et al (140) aligned with the NICE reference case for derivation of utility estimates.
- Studies typically undertook detailed costing analyses; however, the extent of reporting varied between studies. The costs included in analyses appear to be limited to direct costs. Reporting by Pasha et al (78) indicates that societal costs may have been included; however, their incorporation is unclear.
- Whilst analyses differ in their methods, those that compared UDCA or liver transplant to placebo each found the active therapy to dominate placebo, i.e. reducing costs and improving health outcomes.

5.1.2.2 Individual study summaries

BOBERG ET AL 2013 (Norway)

Reason for inclusion

Boberg et al (79) conducted a cost-consequence analysis in the treatment of PBC, comparing UDCA to a control strategy in which UDCA was not administered. Whilst the study was a cost-consequence analysis, the authors also presented results using a cost-effectiveness approach (incremental cost per life year gained); accordingly, this study was included for review.

Model overview

The authors simulated outcomes using a semi-Markov model, with patients transitioning between three health states:

- Alive (starting state)
- Alive after transplant
- Dead

UDCA was applied as the baseline treatment in the analysis. The probability of transitioning between the three health states whilst receiving UDCA were derived from registry data. Long-term survival data (up to 11.5 years) for PBC subjects treated with UDCA were available from a Norwegian registry. Weibull distributions were fitted to these data to estimate the following transitions:

- 'Alive' to 'Alive after transplant' Patients who died prior to transplant were censored from this analysis
- 'Alive' to 'Dead' patients who received transplant prior to death were censored from this analysis

The probability of transitioning to death from the post-transplantation state was taken from the Nordic Liver Transplant Registry, which estimated the relative risk of death following liver transplant compared with the general population. This relative risk was applied to baseline mortality rates observed in the general population to estimate the higher risk of death following liver transplant in this population. The probability of further surgery was derived from registry data; it is unclear from the report how further surgery was applied in the model.

The relative effect of not receiving UDCA was estimated from placebo data from a single trial (78). The separate hazard ratios for both transplant and death from the initial health state were estimated and applied to the baseline risk of progression to simulate outcomes in this patient group. The following hazard ratios were applied in the model:

- Mortality Hazard ratio of 1.54 for control relative to UDCA
- Transplantation Hazard ratio of 2.03 for control relative UDCA

Whilst these hazard ratios have been reported, the maturity of the data on which these estimates are based is not reported, nor are the associated significance levels.

In both arms, patients were at risk of experiencing major adverse events, which contributed to the cost profile of the simulated patients. Similar rates of adverse events were applied for both UDCA treatment and the non-UDCA PBC control strategy.

Costs were derived for each model health state and adverse event. Costs comprised regular physician visits, cost of major events, and treatment costs. No societal costs were considered. All costs were reported in 2005 Euros. A discount rate of 4% was used for both future costs and life-years.

Primary outcomes of the analysis were survival rates and adverse events, including variceal bleeding, ascites, encephalopathy, and liver transplantation.

It was not specified whether patients had previously been on treatment for PBC. Baseline characteristics of patients in the model were reported as shown in Table 42.
Variable	UDCA Norwegian patients (n=182)	UDCA Canadian patients (n=111)	Placebo Canadian patients [§] (n=111)	Placebo patients from Mayo study [†] (n=91)
Gender; females, n (%)	163 (90)	101 (91)	105 (95)	79 (87)
Age; years, mean (SD)	56.3 (8.9)	57.3	55.4	52.0 (9)
Body weight; kg, mean (SD)	66.3 (11.9)			
Dose UDCA; mg/kg/day, median (range)	20.2 (17–23)		0	0
Symptoms before start, n (%)			·
Pruritus, n (%)	91 (50)	87 (78)	79 (71)	
Fatigue, n (%)	99 (54)	87 (78)	83 (75)	
Jaundice, n (%)	12 (7)			
Ascites (clinical finding), n (%)	3 (2)	2 (1.8)	4 (3.6)	
Pruritus + fatigue, n (%)	65 (36)			
Encephalopathy, n (%)	3 (2)			
Asymptomatic, n (%)	56 (31)	13 (12)	14 (13)	
AMA titre, median (range) [‡]	1024 (0–2048)		79 (71)	
Bilirubin, mean (SD) (3–26 µmol/L)	19.4 (18.3)	40 (64)	31 (39)	31 (39)
ALP, mean (SD) (70–230 U/L)	980 (636)	588 (418)	549 (339)	1,252 (712)
ALT, mean (SD) (10–50 U/L)	110 (70)	110 (63)	109 (62)	
Albumin, mean (SD) (35–45 g/L)	40.0 (3.6)			33 (4.0)
INR, mean (SD) (0.8–1.2)	1.2 (0.2)			
IgM, mean (SD) (0.4–2.1 g/L)	5.23 (3.72)	5.9 (4.5)	5.9 (3.5)	
Hepatomegaly, n (%)	54 (30)			
Splenomegaly, n (%)	14 (8)			
Mayo risk score at inclusion [¶] , mean (SD)	4.38 (0.88)	4.6 (1.3)	4.4 (1.2)	5.1 (1.1)

Table 42: Characteristics of PBC patients at start of UDCA- or non-UDCA treatment

Abbreviations: ALT, amino transaminase; AMA,anti-mitochondrial antibody; INR, international normalised ratio; IgM, immunoglobulin M; SD, standard deviation; UDCA, ursodeoxycholic acid.[§]Data obtained from Pasha et al (78). [†]Data obtained from Lindor et al and Pasha et al (34, 78). [‡]Titer available in 150 patients. [¶]Based on 173 patients with complete set of variables in the Mayo risk score at inclusion.

Results

UDCA was found to dominate the control treatment. Life years gained were estimated to be superior (11.97 years vs. 10.78 years) in the UDCA arm, whilst costs were estimated to be lower in the UDCA arm (€151,403 vs. €157,741). Results by clinical outcome are reported below in Table 43.

Total number of events	First year	Second year	Third year	Fourth year	Fifth year
Variceal bleeding (n=16)	1.72	1.20	3.13*	1.94	1.82*
Ascites (de novo) (n=9)	1.15	1.20	1.88	0	1.21
Encephalopathy (de novo) (n=4)	0.57	0.60	0.63	0.65	0
Liver transplantation (n=3)	0.57	0	0.63	0	0.61
Death (n=16)	1.10	1.81	1.88	1.30	3.64

Table 43: Incidence (per 100 person-years) of major events over time among Norwegian PBC patients (n = 182) treated with UDCA during the first 5 years.

*Including one death.

LONGWORTH ET AL 2003 (UK)

Reason for inclusion

Longworth et al (140) conducted an economic evaluation investigating the mid-term (27 months) cost-utility of liver transplantation programs in England and Wales. The costeffectiveness of the liver transplant program in PBC was reported in isolation to other conditions included in the analysis and was included on this basis (i.e. separate reporting of data and results for PBC patients).

Analysis overview

The analysis estimated outcomes for two groups:

- Patients that underwent liver transplant for PBC
- Patients that did not undergo liver transplant for PBC

The same cohort of patients entered into each arm of the analysis (i.e. there were no differences in baseline patient characteristics in each arm). Outcomes for patients undergoing liver transplantation were estimated from observed data collected across six UK liver transplant centres. Both survival and quality of life data for these patients were collected over a maximum of a 27-month period. Since this aligned with the chosen time horizon of the analysis, no modelling was required in this arm. The study reported demographics for patients who were assessed for, or underwent, liver transplantation (see Table 44).

	PBC	ALD	PSC
Patients assessed for transplantation	N = 122	N = 155	N = 70
Men, n (%)	14 (11.4)	114 (73.6)	48 (68.6)
Age, median	57	51	50
Age, IQR	51-62	46-57	38-56
Listed for a liver transplant, n (%)	94 (77.0)	100 (64.5)	53 (75.7)
Patients who underwent a liver transplantation	N = 81	N = 82	N = 45
Patients who underwent a liver transplantation, men, n (%)	8 (9.9)	67 (81.7)	31 (68.9)

Table 44: Demographic details of patients assessed for liver transplantation and patients who underwent transplantation

	PBC	ALD	PSC
Patients who underwent a liver transplantation, age, median	56	50	49
Patients who underwent a liver transplantation, age, IQR	51–62	45–57	38–56
Emergency cases, n (%)	1 (1.2)	0	0
Retransplantation, n (%)	9 (11.1)	6 (7.3)	7 (15.6)
Survival to 2 years posttransplantation, n (%)	69 (85.2)	67 (81.7)	36 (80.0)
Bilirubin (µmol / L), median	97	47	116
Bilirubin (µmol / L), IQR	42–176	25–94	46–265
MELD score, N*	74	66	41
MELD score, median	8	10	10
MELD score, IQR	5-13	6-15	6-16

Abbreviations: ALD, alcoholic liver disease; MELD, model of end stage liver disease; PBC, primary biliary cholangitis/cirrhosis; PSC, primary sclerosing cholangitis; IQR, inter-quartile range, *MELD scores were not available for all patients.

Outcomes for patients not undergoing liver transplant were estimated from published prognostic models. Baseline data for patients that underwent liver transplant were entered into the prognostic model, to estimate the survival probability of these patients had they not undergone surgery.

Utility estimates for patients with PBC were taken from patients enrolled in the observational study using the EuroQol 5 Dimension questionnaire. Estimates were reported from point of listing on the transplant list through to 24 months post-transplantation. HRQoL was captured directly for all patients that underwent transplantation. In the scenario in which transplant was not administered, patient utility was set equal to the last pre-surgery utility value captured in the observed dataset.

Costs for patients undergoing liver transplant were collected prospectively in the observational dataset, these were valued using the mean unit cost for each resource observed over the six centres. Costs in the no transplant arm were estimated on the assumption that the final pre-transplant cost remained constant over the remainder of the analysis. The cost year was not specified.

Results

Transplantation was found to increase costs relative to no transplantation (\pounds 52,525 vs \pounds 37,301). This increase in costs was associated with an increase in QALYs equal to 0.54. This resulted in an incremental cost-effectiveness ratio (ICER) of \pounds 29,000 per QALY gained. This result was found to be robust under PSA, with the authors noting that cost-effectiveness would likely improve should the time horizon be extended.

PASHA ET AL 1999 (US)

Reason for inclusion

Pasha et al (78) conducted an economic evaluation comparing UDCA to a scenario in which UDCA was not administered. All PICOS criteria specified in this review were met and the analysis was included on this basis.

Analysis overview

The costs and health outcomes of the two scenarios were estimated over a period of four years. All outcomes were based on results observed in two clinical trials comparing UDCA to placebo, and no extrapolation of the outcomes was conducted. The economic analysis combined results from these two studies to estimate the following outcomes:

- Survival
- Adverse events (ascites, varices, variceal bleed, encephalopathy)
- Liver transplantation

Survival was estimated as the area under the curve from observed Kaplan-Meier data. Incidence of adverse events was estimated as the total number of events in each arm divided by the total years of follow-up.

Patient characteristics are presented below in Table 45.

	Mayo study		Canadia	an study
	UDCA	Placebo	UDCA	Placebo
Ν	89	91	111	111
Age (yr)	54 ± 9	52 ± 9	57.3	55.4
Gender (F/M)	81/8	79/12	101/10	105/6
Histological stage				
1 and 2	31	26	50	47
3 and 4	58	65	57	60
Mayo risk score	5.2 ± 1.1	5.2 ± 1.1	4.6 ± 1.3	4.4 ± 1.2

Table 45: Patient characteristics in Mayo and Canadian UDCA-PBC trials

Abbreviations: PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Only direct costs were incorporated in the analysis. Whilst a societal perspective was adopted, it is unclear from the reporting of the study which estimates were included as those incurred by the payer and those incurred by broader society. The costs utilised were 1995 wholesale prices.

Results

Treatment with UDCA was found to dominate placebo, as it increased survival and reduced the incidence of all major events (liver transplant and adverse events). A total increase in survival equal to 0.18 years was estimated in the UDCA arm relative to the placebo arm, with costs reducing from \$7,993 in the control arm to \$6,621 in the scenario in which UDCA was used.

RATCLIFFE ET AL 2001 (UK)

Reason for inclusion

The analysis presented by Ratcliffe et al (139) evaluates the cost-effectiveness strategies for the prioritisation of liver transplants in the UK. Two possible causes for liver transplantation are included in the analysis: PBC and alcoholic liver disease (ALD). The

analysis evaluates a scenario in which PBC patients are prioritised for liver transplant. The results of the analysis evaluating this priority setting were deemed relevant to the current review and were included.

Analysis overview

Ratcliffe et al developed a discrete event simulation to evaluate the cost-effectiveness of various transplant prioritisation scenarios. A single patient is evaluated at a time (i.e. a patient-level simulation approach was applied), with a total of 1,000 patients per simulation.

Patients entered the analysis and were assessed for suitability for transplantation. If they were accepted for transplant, the patient entered onto the waiting list for liver transplant. Whilst on this waiting list patients were at risk of death and complications. If patients survived to receive a liver transplant, they entered into a post-transplant state where they were retained unless requiring further transplantation. Those who were not listed for transplant entered into a separate model that simulated the care pathway for management of their ongoing liver disease. The analysis simulated outcomes over a 10-year time horizon. Costs used were in 1999 GBP. Patient characteristics and demographics were not reported.

Results

Incremental results were presented for two groups:

- Those who underwent transplant
- Those who did not receive any transplant

Since the analysis of those that did not receive transplant incorporated outcomes for those that may not be eligible for transplantation, there may be differences between the populations for which outcomes are compared. The impact this had on outcomes is unclear.

Regardless, the authors concluded that the prioritisation of PBC patients for liver transplant results in patients gaining an additional 4.01 life years, whilst those that did not receive liver transplant experienced 1.03 life years. Costs were equal to £59,610 in the transplant arm and £24,358 in the non-transplant arm. This resulted in an ICER of £11,830.

Study, Year, Country	Summary of model	Intervention/ comparator	Patient population (average age in years)	Time horizon	QALYs (intervention, comparator)	Costs (currency) (intervention, comparator)	ICER (per QALY gained)
Boberg et al 2013 (79), Norway	Markov model comprising the following health states: Alive (starting state) Alive after transplant Dead These health states were used to simulate disease progression from point of initiating therapy to death. Within each health state patients were at risk of ascites, variceal bleeding, and encephalopathy. Survival analysis was conducted to estimate mortality and liver transplant probability over time in the UDCA arm. Patients that underwent liver transplant were censored from mortality estimates. Treatment effect attributable to a scenario in which patients did not receive UDCA was incorporated through application of hazard ratios.	UDCA versus placebo	PBC population with an average age of 56	44 years (to 100 years of age)	Expressed as life years gained: UDCA - 11.97 years; Untreated population - 10.78 years	UDCA - €151,403 Untreated - €157,741	UDCA found to dominate a strategy of no treatment. This result was found to be robust when undertaking probabilistic sensitivity analysis, with 82% of iterations resulting in UDCA dominating a strategy of no UDCA.
Longworth et al 2003 (140), UK	Two interventions – transplant administered versus no transplant. Outcomes in the transplant group	Liver transplantation versus no liver transplantation	The cohort of 122 PBC patients consisted of patients who presented for a	27 months	Underwent transplant – 1.30 QALYs No transplant –	Underwent transplant – £52,525 No transplant -	Transplant was associated with an ICER equal to £28,716/ QALY

 Table 46: Summary list of published cost-effectiveness evaluations

Study, Year, Country	Summary of model	Intervention/ comparator	Patient population (average age in years)	Time horizon	QALYs (intervention, comparator)	Costs (currency) (intervention, comparator)	ICER (per QALY gained)
	were estimated from observed data. Transplant outcomes data were taken from a planned data capture over a 27-month period, i.e. no modelling was conducted for this arm. This included the capture of disease outcomes, quality of life (QoL) outcomes, and cost outcomes. The baseline characteristics of the transplant patients were assumed for the non-transplanted population. These baseline characteristics were applied in a published prognostic model, which estimated outcomes for patients in the absence of transplantation. Patients in this group were assumed to have a constant level of utility.		liver transplant in the UK between December 1995 and December 1996. The PBC group median age was 57, with patients undergoing liver transplant having a median age of 56		0.76 QALYs	£37,301	compared to no transplant. Approximately 60% of iterations are found to result in ICERs of £30,000 or less when considering results of probabilistic sensitivity analysis.
Pasha et al 1999 (78), USA	A within-trial analysis was conducted, i.e. no extrapolation beyond study outcomes was undertaken in either study arm. Two trials evaluating UDCA vs placebo informed the model outcomes. The results of these trials were combined, resulting in a total of 200 patients in the UDCA arm and 202 patients in the placebo arm.	UDCA versus placebo	Patients had an average age of 55, and 91% of subjects were female	4 years.	Outcomes were expressed as the incidence of adverse events, survival free of liver transplantation, and average life years in	Annual cost of treatment including cost of adverse events: UDCA - \$6,621 Placebo - \$7,993	UDCA was found to dominate placebo.

Study, Year, Country	Summary of model	Intervention/ comparator	Patient population (average age in years)	Time horizon	QALYs (intervention, comparator)	Costs (currency) (intervention, comparator)	ICER (per QALY gained)
	 Major outcomes in each of the UDCA and placebo arms were estimated, including: Death Liver transplantation Ascites Development of varices Development of variceal bleeding Encephalopathy 				each arm. No absolute estimates of life years in each arm were reported.		
Ratcliffe et al 2001 (139), UK	A discrete event simulation was built to simulate patient outcomes with end-stage liver disease, caused by either ALD or PBC (the analysis was conducted in a mixed pool of ALD/PBC; however, costs under a scenario in which PBC patients were prioritised was reported). Patients entered the model and underwent assessment for liver transplant. If a patient was selected for transplant they entered onto the waiting list. Providing the patient survived the waiting period they underwent surgery. Following surgery, patients were tracked until death unless requiring re- transplantation, at which point	Prioritisation of transplant versus no prioritisation of transplant	Age was not stated by the authors. 46% of the cohort had PBC.	10 years.	Outcomes were expressed in terms of life years. Prioritising transplant in the PBC group led to 4.01 life years gained. Under this scenario, patients that did not receive transplant survived for 1.03 years.	Costs in the PBC prioritisation scenario were equal to £59,610 for those that received transplant and £24,358 for those that did not.	Prioritisation of patients in the PBC group led to an ICER of £11,830.

Study, Year, Country	Summary of model	Intervention/ comparator	Patient population (average age in years)	Time horizon	QALYs (intervention, comparator)	Costs (currency) (intervention, comparator)	ICER (per QALY gained)
	they were re-assessed for suitability and entered into the same transplant pathway. Patients deemed unsuitable for transplant entered into a separate patient pathway capturing outcomes from the management of their underlying liver disease.						

Abbreviations: ALD, alcoholic liver disease; ICER, incremental cost-effectiveness ratio; PBC, primary biliary cholangitis/cirrhosis; QALY(s), quality-adjusted life year(s); QoL, quality of life; UDCA, ursodeoxycholic acid.

5.1.3 *Quality assessment of identified studies*

Quality assessments for all included studies are provided in Appendix 7.

5.1.4 Update of systematic review

The systematic review was updated by a supplementary review, which was conducted in March 2016. Full details of the search strategy and inclusion/exclusion criteria are provided in Appendix 6.

5.1.4.1 Results of updated systematic review

After screening, one publication was included. The publication in question was a poster of a systematic review of cost-effectiveness of current therapies in PBC, which identified two previous studies on the cost-effectiveness of UDCA(141). The authors concluded that a PBC model informed by recent advances in disease understanding is required to accurately estimate the economic burden of PBC.

5.2 De novo analysis

5.2.1 Patient population

The population considered for this economic model is patients with PBC who have failed to show adequate control with UDCA, including both UDCA-intolerant patients and patients who have previously had an inadequate response to UDCA. This reflects the population specified in the NICE scope and the anticipated marketing authorisation. The two main patient subgroups from POISE are considered:

- UDCA-intolerant patients
- UDCA inadequate responders

5.2.2 *Model structure*

A Markov state-transition model was developed to describe the progression of PBC over a lifetime horizon. The model comprises 10 health states with a three-month cycle length. Figure 24 shows a diagram of the model structure. The model captures two components of the natural history of PBC: the liver disease component showing the progression of PBC based on ALP and bilirubin biomarkers; and the clinical endpoint component once patients start progressing to decompensated cirrhosis, hepatocellular carcinoma (HCC), or are added on the liver transplantation waiting list.

A Markov structure was used as it is consistent with other approaches for liver disease modelling, for example, for hepatitis C (142). The main events and changes in the health of a PBC patient, NHS costs, and the risk of other clinical events (e.g. progression to decompensated cirrhosis) are captured by the Markov health states that have been selected.

The model includes three PBC health states contingent on their level of ALP and bilirubin at baseline as defined below:

- Low risk of PBC disease progression: ALP level ≤ threshold of 200 units / L (i.e. 1.67 x ULN) and normal bilirubin (i.e. total bilirubin [TB] ≤ 20 µmol / L)
- Progressive PBC: ALP > threshold and normal bilirubin
- Progressive PBC leading to liver failure: Abnormal bilirubin (TB > 20 µmol / L and rising, or compensated cirrhosis)

The three health states were classified as "low risk" (ALP ≤ threshold and normal bilirubin), "progressive PBC" (ALP > threshold and normal bilirubin) and "Progressive PBC leading to liver failure" (abnormal bilirubin and rising; includes patients with compensated cirrhosis) to reflect the degree of risk of progression for patients in each category. Patients in the first health state have stable ALP and bilirubin and a low risk of progression (15, 34). In the "progressive PBC" state, patients are at moderate risk of developing complications. Finally, the "progressive PBC leading to liver failure" health state combines both compensated cirrhotic patients and those with abnormal bilirubin. The increased risk of progression of patients with abnormal bilirubin has been documented in the literature. In a recent poster by Harms et al, it was shown that patients would either undergo liver transplant or die from liver-related causes within 19 months once total bilirubin reached 1.6 x ULN (141). From that point onward, the level of total bilirubin was shown to rise exponentially. On the other hand, the histological status of PBC patients is rarely documented in clinical practice, since monitoring is based on biochemical values, and liver biopsy is only considered in specific cases (e.g. disease staging, enrolment in randomised clinical trials, or differential diagnosis) (34, 44, 143).

Given the lack of data on histological progression among PBC patients, patients with abnormal bilirubin and those with compensated cirrhosis were combined into one health state. This was to reflect the risk for PBC patients of developing compensated cirrhosis prior to decompensated cirrhosis, whilst also reflecting the risk of progression to hepatocellular carcinoma or to be added to the liver transplant waiting list based on their bilirubin level. Only patients in the moderate and high risk categories are deemed eligible to receive OCA, as these patients have had PBC for approximately 9 years with ALP \geq 200 units / L despite treatment with UDCA, and have a moderate to high risk of progressing to more severe states of PBC.

In the OCA arm, patients can move through the different health states based on the biochemistry data available from POISE, while in the UDCA arm, patients inadequately controlled with UDCA can only progress from their initial pre-severe PBC health state to the severe PBC health state, as they have no effective treatment option to slow or prevent progression of PBC. Patients in the "high risk PBC" health state can progress to liver transplant, decompensated cirrhosis, or HCC, with their associated costs and health-related quality of life (HRQoL). It was assumed that patients in that health state would be in a worse condition than patients with compensated cirrhosis alone. In the decompensated health states, patients can either stay in that state, move to HCC, or progress to liver transplantation. Patients with HCC can either remain in that state, progress to liver transplantation, or die. Following liver transplantation, patients face a probability of dying or moving to the post-transplantation phase. The possibility of PBC recurring after transplant is included in the model, and after re-emergence, patients are at risk of needing a second liver transplantation.

Although not represented on the transition diagram, age- and gender-specific general population mortality rates are applied to each health state in the model. The risk of death is, however, highest in the last and most severe states (i.e. decompensated cirrhosis, HCC, pre-liver transplantation, liver transplant and post-liver transplantation). The excess mortality associated with these health states is depicted by the black coloured arrows in Figure 24. Excess mortality represents the disease-specific mortality associated with having decompensated cirrhosis, liver transplant or hepatocellular carcinoma. Dashed arrows represent health state transitions that can be investigated in sensitivity analysis.

A cohort of 1,000 patients entered the model to simulate the costs and outcomes associated with each of the treatment strategies that are considered. The option to apply a half-cycle correction is included, and is enabled by default.



Patients can die in each health state. The grey health state 'excess mortality' represents the disease-specific mortality associated with having decompensated cirrhosis, liver transplant or hepatocellular carcinoma. TB, total bilirubin.

Abbreviations: ALP, alkaline phosphatase; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; PBC, primary biliary cholangitis/cirrhosis; TB, total bilirubin.

5.2.2.1 Key features of the de novo analysis

Factor	Chosen values	Justification
Time horizonLifetime horizon, i.e. 50years or a maximum ageof 100 years		Consistent with the NICE reference case, which requires costs and effects to be measured over a sufficient time horizon to fully capture the relative costs and benefits.
		The average age of PBC patients included in the POISE trial is 56.2 years. A lifetime horizon (100 years old) was considered to be able to fully estimate the long-term impacts on costs and outcomes.
Were health effects measured in QALYs; if not, what was used?	Health effects were measured in QALYs	As per the NICE reference case. Life years, as well as the number of high-risk PBC, decompensated cirrhosis, HCC, liver transplants and deaths avoided, were also estimated.
Discount of 3.5% for utilities and costs	Selected by default	As per the NICE reference case.
Perspective (Payer – NHS)	Selected by default	As per the NICE reference case.

Table 47: Features of the de novo analysis

Abbreviations: NHS, National Health Service; PBC, primary biliary cholangitis/cirrhosis; QALYs, qualityadjusted life years.

5.2.3 Intervention technology and comparators

5.2.3.1 Intervention and comparators

The interventions considered in this submission depend on the population selected.

For UDCA-intolerant patients the following interventions are considered:

- OCA dose titration (as defined for the titration group in POISE)
- Placebo

For UDCA inadequate responders, the following interventions are considered:

- OCA dose titration (as defined for the titration group in POISE; 5 mg for the first six months of treatment, followed by 10 mg for the subsequent months) + UDCA
- UDCA monotherapy with placebo.

All UDCA doses are implemented in the model according to its UK marketing authorisation, which are also the same as those used in the POISE trial.

5.3 Clinical parameters and variables

5.3.1 How are clinical data incorporated into the model?

The POISE study (detailed in Sections 4.3, 4.4, 4.5, 4.6, 4.7, 4.8 and 4.12) is the principal source of evidence for the economic model, informing key clinical events and outcomes, including transitions to disease states, HRQoL, discontinuation, and adverse events.

The use of surrogate markers such as ALP and bilirubin levels is necessary to predict long-term clinical outcomes in patients with PBC (15, 34, 137, 141, 144). Biochemical response criteria are useful and generally accepted management tools used for stratification purposes and identifying patients in need of additional treatment; the FDA have accepted the validity of the use of ALP and bilirubin levels as surrogate markers for disease severity and progression (145). The COBALT phase III study on outcomes in PBC treated with OCA also included ALP and bilirubin levels as secondary outcomes. Numerous definitions of optimal biochemical response have been proposed in the literature (see Table 48). Although many of the response criteria are based on an assessment at 1 year or even 2 years of initial therapy, a recent study showed that assessment as early as 6 months can be used to identify high risk patients with poor prognosis and who are therefore in need of additional therapy (146).

Criteria	Evaluation Time	Biochemical response	Responder survival
Mayo (125)	6 months	ALP <2x ULN	Not reported
Barcelona (16)	1 year	ALP ≤1x ULN or decrease in ALP >40%	No significant difference between UDCA responders and the standardised general population (p=0.15)
Paris I (11)	1 year	ALP <3x ULN or AST <2x ULN or bilirubin ≤1 mg/dL	No significant difference between UDCA responders and the standardised general population (p=0.8)
Paris II (12)	1 year	ALP ≤1.5x ULN or AST ≤1.5x ULN or bilirubin ≤1 mg/dL	In early PBC, defined either histologically or biochemically, survival without adverse event [†] was significantly longer in responders compared with non- responders (p<0.001 and <0.05 respectively). Patients with histologically defined early disease who respond to treatment with UDCA experienced no progression of their disease over an average of 7 years.
Rotterdam (13)	1 year	Normal bilirubin (values ≤ULN) and/or normal albumin (values ≥LLN	Survival is significantly longer in UDCA responders vs non responders.
Toronto II (15, 18)	2 years	ALP ≤1.67x ULN	Not reported. Those who did not respond at 2 years had a five-fold greater risk of histological progression compared with responders (0.03).
Ehime (147, 148)	6 months	≥70% decrease of GGT	Transplant-free survival was significantly longer in UDCA responders compared with non-responders (p=0.01).
Mayo II (17)	1 year	ALP ≤1.67 and bilirubin ≤1mg/dL	Not reported

Table 48: Biochemical criteria of c	optimal response to UDCA in PB0
-------------------------------------	---------------------------------

Abbreviations: ALP, alkaline phosphatase; AST, aspartate aminotransferase; GGT, gamma-glutamyl transferase; LLN, lower limit of normal; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

[†]Adverse event defined as liver-related death, liver transplantation or referral to transplant unit, complication of cirrhosis, or histological evidence of cirrhosis development.

5.3.1.1 Overview of analyses

Patient-level data analyses of POISE were used to inform:

- Baseline characteristics
- Base distribution of patients according to ALP or bilirubin state
- Adverse event rates
- Transition probabilities (for biochemistry states)

These analyses were performed and reported in accordance with NICE Decision Support Unit (DSU) methodologies where relevant (149) which are reported in the relevant sections in this submission. All analyses were based on the ITT population of POISE (216 patients).

5.3.1.2 Mortality

The base case analysis models all-cause mortality data from POISE (no deaths were reported in the POISE trial).

5.3.2 Transition probabilities

5.3.2.1 Biochemistry transition matrices (OCA)

Patient-level data from POISE was used to follow the progression of patients across the biochemical health states, i.e. either based on an ALP threshold of 200 units / L, and Table 49 presents the biochemistry data used to populate the transition matrices over the first year for the two OCA-based regimens. Given the low number of patients who received monotherapy, the same transition matrices were used for the OCA titration regimen with or without UDCA. This is a rather conservative method as in clinical practice, OCA patients would be expected to have lower transition probabilities to the more severe disease states.

		0	-3 montl	าร			3-6 m	onths			6-9 m	onths			9-12 n	nonths	
From/To:	1	2	3	Disc.	Ν	1	2	3	Ν	1	2	3	Ν	1	2	3	Ν
ALP thresho	ld 200 u	nits / L															
1					I												
2																	
3									Ī								
Probabilities																	
1					1.0				1.0				1.0				1.0
2					1.0				1.0				1.0				1.0
3					1.0				1.0				1.0				1.0

Table 49: Biochemical transitions during the first year – OCA titration (POISE trial)

Health state 1: $ALP \le 200$ units / L and NB; 2: ALP > 200 units / L and NB; 3: (ALP > 200 units / L) and AB; Disc.: Discontinuation. Source: Ad-hoc table ALP and BILI shift (Data on file). Analyses for other threshold are presented in Appendix 12.

5.3.2.2 PBC-specific component (literature)

UDCA inadequate responders

Whilst OCA-based regimens rely on POISE patient-level data during the first year, the transition probabilities used to reflect progression whilst on UDCA alone, or no treatment, are based on data from various literature sources identified in the previously discussed SLRs and hand searching of data. These transition probabilities are specific to PBC where available. Where no data were available in the literature, transition probabilities were calibrated so that the risks of liver transplantation and/or death reflected those observed in POISE.

At EASL 2016, Harms et al presented a poster showing the change in risk of live transplant and / or death in POISE patients treated with UDCA and OCA. The authors demonstrated that after one year the risk of transplant and / or death was greater in patients treated with UDCA versus patients treated with OCA. Data published from the global and UK PBC groups and an ad-hoc analysis carried out based on POISE patient-level data using the GLOBE and UK risk score were used to calibrate six transition probabilities, i.e.:

- Transition from "ALP ≤ 200 units / Lx ULN and bilirubin ≤ 20 µmol / L" to "bilirubin > 20 µmol / L and rising, or compensated cirrhosis"
- Transition from "ALP > 200 units / L x ULN and bilirubin ≤ 20 µmol / L" to "abnormal bilirubin and rising, or compensated cirrhosis"
- Transition from "abnormal bilirubin and rising, or compensated cirrhosis" to either "decompensated cirrhosis" or "Pre-LT" (liver transplant waiting list)
- Transition from "decompensated cirrhosis" to "Pre-LT" or liver-related death

The calibration process followed the process shown in Figure 25 below. A more detailed explanation of the calibration process is provided in Appendix 10.

Figure	25:	Calibration	process	diagram
	-			

Calibration transition from DCC to pre-LT and liver-related death	Calibration transition from AB and rising, or CC to DCC or liver- related death	Calibration transition from "ALP > 1,67 x ULN and NB" to "AB and rising, or CC"	Calibration transition from "ALP ≤1,67 x ULN and NB" to "AB and rising, or CC"
---	--	--	---

Abbreviations: AB, abnormal bilirubin (total bilirubin >1.0 x ULN); ALP, alkaline phosphatase; CC, compensated cirrhosis; DCC decompensated cirrhosis; LT, liver transplantation; NB, normal bilirubin (total bilirubin \leq 1.0 x ULN); ULN, upper limit of normal.

Table 50: PBC component: Transition probabilities to "abnormal bilirubin and rising, or compensated cirrhosis" and decompensated cirrhosis

From:	То:	Probability	Time period (years)	Quarterly rate	Quarterly *probability	
ALP threshold 200 units / L						
ALP ≤ 200 units / L and NB	AB and rising, or CC	0.05	1.0	0.01	0.01	
ALP > 200 units /	AB and rising, or	0.12	1.0	0.03	0.03	

From:	To:	Probability	Time period (years)	Quarterly rate	Quarterly *probability
L and NB	CC				
To decompensated	cirrhosis (DCC)				
AB and rising, or CC	DCC	0.10	1.0	0.03	0.03

Abbreviations: AB, abnormal bilirubin; CC, compensated cirrhosis; DCC, decompensated cirrhosis; NB, normal bilirubin. * Obtained by converting quarterly rates to quarterly probabilities, using the formula p = 1 - exp(-rt), where p = probability, r = rate, and t = time.

UDCA-intolerant patients

UDCA-intolerant patients can either receive OCA monotherapy (as a titrated dose) or no treatment. Given the limited number of patients not on UDCA, there is a lack of data in the literature looking at the natural history of PBC without active treatment since the launch of UDCA. Thus, to reflect the progression of patients who did not receive any active treatment, assumptions were made based on Corpechot et al. (2000), who assessed the effect of UDCA therapy on liver fibrosis progression compared with no treatment in PBC patients(150). The study was based on a French randomised, double-blind, placebo-controlled trial in which 146 patients were randomised to receive either 13-15 mg/kg/day of UDCA or a placebo for 2 years (150). After 2 years, placebo patients crossed over to UDCA for a further 2 years, whilst patients already on UDCA remained on UDCA. Paired liver biopsies were available for 103 patients (53 UDCA and 50 placebo) during the two-year blinded period.

Corpechot et al. (2000) modelled the progression from two histologic stages (Figure 26):

- Non-fibrotic stage, defined by the presence of portal and periportal lesions without extensive fibrosis
- Fibrotic/cirrhotic stage, defined by the presence of numerous septa, bridging fibrosis or nodular cirrhotic formation.

The description provided by Corpechot et al for fibrotic and cirrhotic stages would contain compensated cirrhotic patients according to the METAVIR criteria (METAVIR stages F3 / F4 (151).

Progression between states was informed by the paired biopsy data. The authors of this study assumed that all patients in a given state and at a given time would have the same prognosis. Two transition rates were estimated to inform the model, i.e. the baseline transition rate α_0 and the regression coefficient for treatment β_t . The relationship between both parameters were expressed as follows: $\alpha_u = \alpha_0^* \exp(\beta_t x U)$ where U=0 for placebo and 1 for UDCA.

Figure 26: Two-stage Markov model (150)





Based on the simplified model parameters (detailed in Table 51), it was possible to estimate the annual probability of progressing over time in the placebo group. Probabilities for patients to transition to the fibrotic/cirrhotic stage were calibrated to estimate the number of cirrhotic patients over time without the probability of moving backwards by minimising the root mean square error (RMSE), leading to a quarterly probability of progression to severe PBC of 0.079. The same transition was assumed between "abnormal bilirubin or compensated cirrhosis" and "decompensated cirrhosis" as for UDCA non-responders.

Whilst these histologic stages do not necessarily directly reflect compensated cirrhosis and decompensated cirrhosis, they essentially represent two different fibrotic or cirrhotic stages. This is the only publication which reflects the natural history of PBC without treatment, and allows estimation of the rate of disease progression for untreated patients.

Parameter	From/to	Estimate	SE
α0	NF to F/C	0.34	0.09
βt	NF to F/C	-1.55	0.49
α0	F/C to NF	0.03	0.01
βt	F/C to NF	0.00	0

Table 51: Simplified model - progression parameters (150)

From:	То:	Probability	Time period (year)	Quarterly rate	Quarterly probability*	
ALP threshold 2	200 units / L					
ALP ≤ 200 units / L and bilirubin ≤ 20 µmol / L	AB and rising, or CC	0.25	1.0	0.07	0.07	
ALP > 200 units / L and bilirubin ≤ 20 µmol / L	AB and rising, or CC	0.25	1.0	0.07	0.07	
To decompensated cirrhosis (DCC)						
Bilirubin > 20 µmol / L and rising, or CC	DCC	0.10	1.0	0.03	0.03	

Table 52: PBC components (UDCA intolerant): Transition probabilities to severe PBC and decompensated cirrhosis

Abbreviations: AB, abnormal bilirubin; CC, compensated cirrhosis; DCC, decompensated cirrhosis. * Obtained by converting quarterly rates to quarterly probabilities, using the formula p = 1-exp(-rt), where p = probability, r = rate, and t = time.

OCA patients

For OCA, results from the long-term safety extension study showed sustained response in patients remaining on OCA for up to five years post initiation of treatment (115). For this reason, we have not assumed any additional progression to move from "ALP ≤ 200 units / L and normal bilirubin" or "ALP > 200 units / L and normal bilirubin" to "abnormal bilirubin and rising, or compensated cirrhosis" in the OCA arms after the first year. This assumption was based on the decompensation rate observed in UDCA responders presented by Harms et al in the PBC subgroup (152). The objectives of the study were to estimate the time to a decompensating cirrhotic event (i.e. the presentation of ascites, variceal bleeding or hepatic encephalopathy), as well as to identify factors predicting the occurrence of decompensating events and related outcomes (i.e. liver transplant-free survival). There were 2,938 of the 3,030 UDCA-treated patients included. The authors investigated the decompensation rate over 15 years for patients who had successfully responded to UDCA and those who had not responded, using the GLOBE score. The authors showed that the risk of developing decompensated cirrhosis was much greater in UDCA non-responders (15-year risk of ~32%) than in UDCA responders (15-year risk of ~9%) (Figure 27). It was assumed that OCA patients would follow a similar general trend to the UDCA patients in terms of their risk of developing DCC in the event of nonresponse. This is a conservative approach, as recent data shows that in patients with raised bilirubin the progression to advanced PBC is much more rapid, and with a greater guarterly probability than is shown in the base case scenario (Intercept data on file).



Figure 27: Biochemical responders vs. non-responders (152)

GLOBE-score: 0.044378 × age at start of UDCA therapy + 0.939820 × LN(bilirubin*ULN at 1 year follow-up) + 0.335648 × LN(alkaline phosphatase *ULN at 1 year follow-up) – 2.266708 × albumin*LLN at 1 year follow-up – 0.002581 × platelet count per 109/L at 1 year follow-up + 1.216865.

Table 53: PBC com	ponent: Transition	probabilities to	severe PBC	(OCA)
				/

From:	То:	Decompensation probability	Time (years)	Quarterly rate	Quarterly probability*
"ALP ≤ 200 units / L and NB" or "ALP > 200 units / L and NB"	AB and rising, or CC	0.25	1	0.07	6.80%

Abbreviations: AB, abnormal bilirubin; CC, compensated cirrhosis; ALP, alkaline phosphatase; ULN, upper limit of normal; NB, normal bilirubin. *Obtained by converting 15-year probability into quarterly rates, using the formula r = -LN(1-p)/t and quarterly rates to quarterly probabilities, using the formula p = 1-exp(-rt), where p = probability, r = rate, and t = time.

5.3.2.3 Liver disease-specific component

Transition probabilities from "abnormal bilirubin and rising, or compensated cirrhosis" to the more severe liver-specific health states were derived from the literature as shown in Table 54.

From:	To:	Probability	Time (years)	Quarterly rate	Quarterly prob.	Source		
AB and rising, or CC	HCC	0.01	1	0.00	0.35%	Assumed similar transition as between DC and HCC		
	Pre-LT	0.04	1	0.01	1.02%	Calibrated (see Appendix 10)		
DCC	Pre-LT	0.06	1	0.02	1.53%	Calibrated (see Appendix		

Table 54: Liver disease component: transition probabilities

						10)
	Death	0.17	1	0.05	3.98%	Calibrated (see Appendix 10)
	HCC	0.01	1	0.00	0.35%	Trivedi et al 2006 (153)
HCC	Pre-LT	0.04	1	0.01	1.02%	Wright et al 2006 (154), used by: STA330 (142)
	Death	0.43	1	0.14	13.11%	Wright et al 2006 (154), used by: STA330 (142)
Pre-LT	LT	0.35	1	0.11	10.21%	Kim et al 2016 (155)
	Death	0.09	1	0.02	2.33%	Kim et al 2016 (155)
LT	Death	0.21	1	0.06	5.72%	Wright et al 2006 (154), used by: STA330, Table 68 (142)
Post-LT	PBC recurrence	0.23	10	0.01	0.64%	Lindor, 2009 (34)
	Death	0.06	1	0.02	1.46%	Wright et al 2006 (154), used by: STA330, Table 68(142)
	LT	0.01	13	0.00	0.01%	Neuberger, 2003 (156)
PBC recurrence	LT	0.01	13	0.00	0.01%	Assumption

Abbreviations: AB, abnormal bilirubin; CC, compensated cirrhosis; DC, decompensated cirrhosis; HCC, hepatocellular carcinoma; LT, liver transplant; PBC, primary biliary cholangitis/cirrhosis. *Obtained by converting 15-year probability into quarterly rates, using the formula r = -LN(1-p)/t and quarterly rates to quarterly probabilities, using the formula p = 1-exp(-rt), where p = probability, r = rate, and t = time.

Based on the transition probabilities presented in this section, the model predicts the mean times to events shown in Table 55.

UDCA intolerai	nt patients	UDCA inadequate responders		
No treatment (Placebo)	OCA titration	UDCA + Placebo	OCA + UDCA titration	
0.047	0.004	0.039	0.004	
19.365	44.605	21.637	44.893	
30.635	5.395	28.363	5.107	
14.750	N/A	17.500	N/A	
	ODCA intoleral No treatment (Placebo) 0.047 19.365 30.635 14.750	ODCA intolerant patients No treatment (Placebo) OCA titration 0.047 0.004 19.365 44.605 30.635 5.395 14.750 N/A	ODCA intolerant patients ODCA inadequipments No treatment (Placebo) OCA titration UDCA + Placebo 0.047 0.004 0.039 19.365 44.605 21.637 30.635 5.395 28.363 14.750 N/A 17.500	

Mean time	to event	for liver-related	death
	Mean time	Mean time to event	Mean time to event for liver-related

*Area under the Kaplan Meier curve. [†]Area above the Kaplan Meier curve. [‡]Using the total population as done with transplant free survival.

5.3.3 Clinical expert assessment of applicability of clinical parameters

UK expert opinion was sought via a series of interviews, to provide validation of proposed methods and statistical models, and to elicit estimates of utility values for PBC-specific health states. An overview of the information provided to experts is given in Appendix 11. Each interview was conducted by at least two health economists, with

audio recordings used to ensure all relevant details were captured. These interviews were supplemented by follow-up emails to clarify any further points that were raised.

Name in document	Date held	External Attendees [†]	Topics of discussion
OCA for PBC interview 1	17/6/16	1 x UK PBC specialist physician, 1x Health Economist	Model methods, health state utilities
OCA for PBC interview 2	14/7/16	1 x UK PBC specialist physician	Model methods, health state utilities, health state costs
OCA for PBC interview 2	25/7/16	1 x UK PBC specialist physician	Model methods, health state utilities, health state costs

[†]Attendees who were neither direct employees of Intercept nor direct employees of an Interceptcommissioned vendor.

5.4 *Measurement and valuation of health effects*

5.4.1 Health-related quality-of-life data from clinical trials

Health-related quality-of-life data were not gathered in the POISE trial, so the literature was searched for health state utilities.

5.4.2 Health-related quality-of-life studies

5.4.2.1 Overview and review question

A systematic review was conducted to identify HRQoL studies from the published literature relevant to the decision problem. In particular, EQ-5D health state utility values (in line with the NICE preferred method) relating to HRQoL in PBC-specific disease states were sought. The following specific review question was explored:

• "What evidence exists reporting the quality of life in patients with PBC?"

5.4.2.2 Search methodology

Studies of interest were identified by simultaneously searching the electronic databases shown in Table 56, with no restrictions on date or language of publication. Searches were conducted in September 2014 using the following interfaces:

- EMBASE (which also covers Medline[®] and Medline[®] In-Process)
- The Cochrane Library (which covers the Cochrane Database of Systematic Reviews, Database of Abstracts of Reviews of Effects, Health Technology Assessment, NHS Economic Evaluation Database)
- EBSCO host (which covers EconLit, Health Economic Evaluations Database).

Full details of the search are provided in Appendix 8.

Table 56: Databases searched and interfaces used in the quality of life systematic review

Database	Interface
Embase 1966 to 2014	Embase
Medline 1966 to 2014	Embase

Medline® In-Process 1966 to 2014	Embase
Cochrane Database of Systematic Reviews 1996 to 2014	The Cochrane Library
Database of Abstracts of Reviews of Effects 1994 to 2014	The Cochrane Library
Health Technology Assessment 1989 to 2014	The Cochrane Library
NHS Economic Evaluation Database 1968 to 2014	The Cochrane Library
EconLIT with Full Text 1961 to 2014	EBSCO host
Health Economic Evaluations Database 1990 to 2014	EBSCO host

The searches included terms for free text and keywords (Medical Subject Heading (MeSH) and Emtree terms) through the use of Boolean combination techniques. Searches were not restricted by study intervention or comparator, to ensure studies reporting HRQoL in this population were identified. The Cochrane Library and EBSCO host interfaces were searched using terms for the population only, in order to broaden the results.

A grey literature search was performed to include additional studies that had not been identified by the search strategy. References included for the review had to meet the pre-specified inclusion/exclusion criteria shown as a PICOS table in Appendix 8.

The search identified 721 titles/abstracts. After screening, 611 were excluded, leaving 110 full texts. After the second round of screening, 107 studies were excluded, and relevant data were extracted from the remaining publications. The PRISMA diagram for the systematic review, including the March 2016 systematic review update, is shown in Figure 28.



Figure 28: PRISMA diagram illustrating the flow of the HRQoL systematic review

Five studies were found to report utility values (6, 140, 154, 157). Aberg et al evaluated the quality of life of Finnish adult patients having undergone liver transplantation between 1982 and 2007 using the 15D questionnaire, including 72 PBC patients. (6) This questionnaire is composed of 15 dimensions (mobility, vision, hearing, breathing, sleeping, eating, excretion, usual activities, mental function, discomfort, symptoms, depression, distress, vitality, and sexual activity) each assessed over five levels. PBC patients were mainly female (89%) and were an average of 52 years old at the time of transplantation and 63 years old at the time of the study. The mean time between liver disease diagnosis and transplantation was 9 years. After adjustment for age at the time of the study (covariate evaluated at age 55), patients with PBC were shown to have a similar 15D score (0.882) to other diseases and be statistically significantly lower than the general population (0.914).

Bondini et al carried out a study that aimed to compare the impact of chronic hepatitis B (CHB) on quality of life, and compared the results to patients suffering from chronic hepatitis C (CHC), PBC and healthy controls (157). The authors used three health-related quality of life questionnaires, the chronic liver disease questionnaire (CLDQ), SF-36 and the Health Utility Index (HUI Mark-2). Only 18 PBC patients were included in the study. PBC patients were shown to have a similar overall summary score to CHC patients (4.4 for CLDQ overall), and SF-36 (about 45 for MCS and 41 for PCS respectively). On the other hand, the health utility scores as estimated by HUI Mark-2

differed between CHC and PBC patients with a utility value more similar to CHB patients (0.81 for PBC, 0.55 for CHC and 0.78 for CHB patients, respectively). The authors stated that the presence of cirrhosis explained 24% of the variance in terms of the HUI Mark-2 values reported. One of the major limitations of this study is the very small sample size of the PBC group, preventing an accurate assessment of the quality of life in this population. Only HUI Mark-2 was used, although other scoring methods could have been used such as EQ-5D or converting SF-36 into SF-6D.

Longworth et al carried out a study assessing the cost-effectiveness of the liver transplantation program of England and Wales for three disease groups, including PBC (140). The quality of life of patients before and after transplantation were based on the EQ-5D questionnaire that was administered at time of listing and then every 3 months until transplantation, and then at 3, 6, 12 and 24 months after transplantation. Values and extrapolated values are shown in Figure 29 and Table 57, respectively.





Error bars depicts 95% bootstrap confidence intervals. Values of zero are included for patients who died post-transplantation

Time	Mean	95% lower Cl	95% upper Cl
At listing	0.384	0.305	0.463
3-month after listing	0.394	0.248	0.539
6-month after listing	0.500	0.266	0.731
3-month post-transplant	0.581	0.512	0.646
6-month post-transplant	0.583	0.504	0.661
12-month post-transplant	0.619	0.548	0.687
24-month post-transplant	0.623	0.546	0.693

Table 57: Utility values extrapolated from Longworth et al (2003)

Abbreviations: CI, confidence interval.

Wright et al carried out a study assessing the health benefits of antiviral therapy for mild chronic hepatitis C, including an economic evaluation (154). Some of these health states are similar to those experienced in PBC patients, including cirrhosis, HCC, post-liver transplantation and decompensated cirrhosis. A summary of the utility values observed are presented in

Table 58. All utility values are reported using the EQ-5D scale.

Health state	Utility value
Mild disease	0.77
Moderate disease	0.66
Cirrhosis	0.55
HCC	0.45
Treatment for mild disease	0.65
Treatment for moderate disease	0.55
Decompensated cirrhosis	0.45
Post-liver transplantation	0.67
SVR after mild disease	0.82
SVR after moderate disease	0.72

Table 58: Mean HRQoL for each disease stage as reported in Wright et al 2006 (154)

Abbreviations: HCC, HRQoL, health-related quality of life; hepatocellular carcinoma; SVR, sustained virologic response.

Younossi et al carried out a study which aimed to assess the utility and HRQoL of chronic liver disease patients, including chronic viral hepatitis and chronic cholestatic liver disease (CLD) (which includes both PBC and PSC patients) (64). The authors reported a utility value of 0.84 ± 0.15 for CLD patients.

5.4.2.3 Update of systematic review

An update of the systematic review was performed in March 2016. Full details of the search strategy and inclusion/exclusion criteria are provided in Appendix 8. No additional studies were identified as a result of the systematic literature review update.

5.4.2.4 Intercept clinical trial

An earlier phase II double-blind, placebo-controlled study (747-202) investigated the efficacy and safety of OCA 10, 25 and 50 mg doses compared with placebo in combination with UDCA, as described in Section 4.7.2.2 (158). The generic health-related quality of life questionnaire SF-36 was included as part of the information collected during the trial. The questionnaire was administered at baseline and at Day 85 or at the early termination (ET) visit. The SF-36 is a survey measuring eight domains (physical functioning, role limitations due to physical health, bodily pain, general health perceptions, vitality, social functioning, role limitations due to emotional problems, and mental health). It yields total scores for each of the domains as well as two component summary measures (physical and mental component summary [PCS and MCS]) (159).

Table 59: SF-36 dimensions scores at baseline	(N=165)	(747-202 CSF	2)
---	---------	--------------	----

Dimensions	Mean	SD
Physical functioning	76.9	23.9
Role physical	71.7	29.5
Bodily pain	70.4	24.6
General health	57.7	23.2
Vitality	53.2	52.7
Social functioning	80.9	24.3
Role emotional	83.2	82.0
Mental health	75.8	74.5
PCS	45.9	46.2
MCS	49.8	48.9

Abbreviations: CSR, clinical study report; MCS, mental component summary; PCS, physical component summary; SD, standard deviations;

Table 60: Included HRQoL studies

Study, Country	Population	Study type	Metric	Result
Aberg et al 2012 (6)	72 PBC patients following liver transplant Sub-group of wider population	Observational	15D instrument	Utility value for entire PBC population following liver transplant: 0.882 Result found to be consistent regardless of aetiology
Bondini et al 2007 (157)	18 PBC patients	Observational	Chronic Liver Disease Questionnaire (CLDQ) SF-36 Health Utility Index (HUI Mark-2 and Mark-3)	18 patients in PBC group led to limited analysis in this population SF-36, CLDQ and HUI values reported HUI utility value reported as 0.81 (SD=0.1) in PBC population
Longworth et al 2003 (140)	122 PBC patients assessed for transplantation and followed up for a maximum of 24 months post-surgery	Observational	EQ-5D	Full results reported in Table 57
Wright et al 2006 (154)	204 chronic mild hepatitis C patients	Observational	EQ-5D	HRQoL results reported in Table 58.
Younossi et al 2001 (64)	120 patients with chronic liver disease, with 30% of patients suffering chronic cholestatic liver disease, including PBC and PSC patients	Observational	SF-36 CLDQ HUI	The study reported an HUI utility value of 0.84 ± 0.15 for patients with cholestatic liver disease.

Abbreviations: CLDQ, chronic liver disease questionnaire; EQ-5D, EuroQOL-5 dimensions, HRQoL, health-related quality of life; HUI, health utility index; PBC, primary biliary cholangitis/cirrhosis; PSC, primary sclerosing cholangitis; SF-36, short form-36 dimensions;

155

Each paper derived utility values using different measures. Aberg et al evaluated the utility of patients with PBC using the 15D survey, and estimated the utility of patients following liver transplant (6). Bondini et al reported utility derived using the Health Utilities Index; these estimates were reported for a general PBC population (157). Longworth et al reported utility values using the EQ-5D, for patients both pre- and post-liver transplant (Table 57) (140). Finally, Younossi et al reported utility values using the HUI and the CLDQ (64). The values reported in Longworth et al were substantially lower than those reported using the alternative instruments incorporated by Aberg et al , Bondini et al and Younossi et al. There was a substantial improvement in QoL following liver transplant in the population reported by Longworth et al, with utility rising from 0.384 at listing for transplant to 0.623 two years after treatment. None of the identified papers explicitly reported utility values in a population that were in the early stages of PBC, although values reported by Bondini et al and Younossi et al may incorporate such patients (6, 140, 157).

5.4.3 Key differences

As no HRQoL data were gathered in POISE, no comparison could be made to values derived from the literature search.

5.4.4 Adverse reactions

In the POISE trial, treatment-emergent adverse events (TEAEs) were defined as any adverse event (AE) that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product; related TEAEs include all events reported having a possible, probable or definite relationship with the investigational product. Since pruritus and fatigue are two of the most common symptoms observed in PBC patients, their occurrence was expected. The most frequently reported related TEAEs in the POISE trial were included to account for the possible increase in any TEAE related to all active treatments (i.e. OCA or UDCA). In all arms, the most commonly reported TEAE was pruritus followed by fatigue and nausea. Since the number of patients not taking UDCA was limited (5 [6.8%] in the placebo, 5 [7.1%] in the OCA titration and 6 [8.2%] in the OCA 10 mg arms respectively), the incidence of TEAEs was assumed to be the same for the two indications. For UDCA-intolerant patients receiving no treatment, no adverse events were considered.

Adverse events are not explicitly modelled, and their impact is captured as part of the disease health state utility values. For the purposes of calculating adverse event cost, overall adverse event rates were used for fatigue, pruritus and nausea, and were applied to each comparator (93).

5.4.5 Health-related quality-of-life data used in cost-effectiveness analysis

In this analysis, a patient's HRQoL is assumed to be a function of:

- Their baseline characteristics (including their baseline ALP / bilirubin state)
- Risk of adverse events.

HRQoL is assumed to be constant in each of the biochemistry states, i.e. the utility for patients in a certain health state does not change over time. HRQoL worsens as patients

proceed from the biochemistry component of the model to the liver disease component of the model.

Utility values used in the cost-effectiveness analysis have been adjusted. The following health states in the liver disease component of the model had their corresponding utility values decreased by **EXE** to simulate the worsened HRQoL experienced by PBC patients in comparison to HBV/HCV patients, as an interpretation of KOL feedback by Intercept:

- Decompensated cirrhosis
- Pre-transplant: utility at listing
- Pre-transplant: 3 months after listing
- Pre-transplant: 6 months after listing
- Liver transplant: 3 months post-transplant
- Liver transplant: 6 months post-transplant
- Liver transplant: 12 months post-transplant
- Liver transplant: 24 months post-transplant.

State	Utility value	Reference in submission (section and page number)	Justification
ALP: ≤ 200 u/L and Bili: Normal	0.84	Table 60, page 155	Cholestatic disease utility reported in Younossi 2001(64)
ALP: > 200 u/L and Bili: Normal	0.84	Table 60, page 155	Cholestatic disease utility reported in Younossi 2001(64)
Bili: Abnormal and rising, or CC	0.55	Table 58, page 153	Previously reported value for compensated cirrhosis (TA330)(142)
Decompensated cirrhosis		Table 58, page 153	Previously reported value for decompensated cirrhosis (TA330); decrement of applied (142)
Hepatocellular carcinoma	0.45	Table 58, page 153	Previously reported value for HCC (TA330); KOL opinion (142)
Pre-transplant: utility at listing		Table 58, page 153	Previously reported value for pre- transplant (TA330); decrement of applied (142)
Pre-transplant: 3 months after listing		Table 58, page 153	Previously reported value for pre- transplant (TA330); decrement of applied (142)
Pre-transplant: 6 months after listing		Table 58, page 153	Previously reported value for pre- transplant (TA330); decrement of applied (142)
Liver transplant: 3 months post-transplant		Table 58, page 153	Previously reported value for post liver transplant (TA330); decrement of applied (142)

Table 61: Summary of utility values for cost-effectiveness analysis

State	Utility value	Reference in submission (section and page number)	Justification
Liver transplant: 6 months post-transplant		Table 58, page 153	Previously reported value for post liver transplant (TA330); decrement of applied (142)
Liver transplant: 12 months post-transplant		Table 58, page 153	Previously reported value for post liver transplant (TA330); decrement of applied (142)
Liver transplant: 24 months post-transplant		Table 58, page 153	Previously reported value for post liver transplant (TA330); decrement of applied (142)
Re-emergence of PBC		Table 58, page 153	Assumed equivalent to liver transplant 24-months post- transplant, without utility decrement provided according to KOL feedback. (TA330) (142)
Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; PBC, primary biliary cholangitis/cirrhosis.			

5.4.5.1 Clinical expert assessment of applicability of health state utility values

Expert opinion was sought to determine whether the use of utility values specific to HCV/HBV was appropriate within the context of PBC. All experts agreed that it is inappropriate to use HCV/HBV-specific utility values, since PBC patients are likely to have worse utility values despite being in the same health state, e.g. HRQoL for PBC patients who are awaiting liver transplant are likely to be worse than for hepatitis patients awaiting liver transplant. An exception is made in the case of hepatocellular carcinoma, where clinical experts agreed that utility values are broadly similar between HCV/HBV patients and PBC patients, as HRQoL in HCC is mainly driven by the treatment of HCC, and the underlying cause of the disease (e.g. PBC or HCV) is no longer the main factor to explain the quality of life experienced by patients in this health state. In order to address this, all utility values for HCV/HBV were decreased by to approximate the worse outcomes for PBC patients. The details of this process have been described previously in Section 5.3.3.

5.5 Cost and healthcare resource use identification, measurement and valuation

5.5.1 *Resource identification, measurement and valuation studies*

5.5.1.1 Overview and review question

A systematic review was conducted to identify all relevant PBC unit cost and resource use studies from the published literature relevant to the decision problem. The following specific review question was explored:

• "What are the costs and resource use associated with the management of PBC?"

5.5.1.2 Search methodology

Studies of interest were identified by simultaneously searching the electronic databases shown in Table 62, with no restrictions on date or language of publication. Searches were conducted in September 2014 using the following interfaces:

- EMBASE (which also covers Medline[®] and Medline[®] In-Process)
- The Cochrane Library (which covers the Cochrane Database of Systematic Reviews, Database of Abstracts of Reviews of Effects, Health Technology Assessment, NHS Economic Evaluation Database)
- EBSCO host (which covers EconLit, Health Economic Evaluations Database).

Full details of the search are provided in Appendix 9.

Table 62: Databases searched and interfaces used in the quality of life systematic review

Database	Interface
Embase 1966 to 2014	Embase
Medline 1966 to 2014	Embase
Medline® In-Process 1966 to 2014	Embase
Cochrane Database of Systematic Reviews 1996 to 2014	The Cochrane Library
Database of Abstracts of Reviews of Effects 1994 to 2014	The Cochrane Library
Health Technology Assessment 1989 to 2014	The Cochrane Library
NHS Economic Evaluation Database 1968 to 2014	The Cochrane Library
EconLIT with Full Text 1961 to 2014	EBSCO host
Health Economic Evaluations Database 1990 to 2014	EBSCO host

The searches included terms for free text and keywords (Medical Subject Heading (MeSH) and Emtree terms) through the use of Boolean combination techniques. Searches were not restricted by study intervention or comparator to ensure studies reporting costs and resource use in this population were identified. The Cochrane Library and EBSCO host interfaces were searched using terms for the population only, in order to broaden the results.

A grey literature search was also performed to include additional studies that had not been identified by the search strategy. References included for the review had to meet the pre-specified inclusion/exclusion criteria shown as a PICOS table in Appendix 9.

The search identified 192 titles/abstracts. After screening, 162 references were excluded, leaving 30 references included for full-text evaluation. No further studies were identified from the review of grey literature. A total of 11 references met the inclusion/exclusion criteria following full-text evaluation. Of these 11 references, all studies were deemed relevant to the review. The PRISMA diagram for the systematic review is shown in Figure 30.

Figure 30: PRISMA diagram illustrating the flow of the cost and healthcare resource use systematic review



Details of the 11 included studies are shown in Table 63.

Study, Year, Country	Cost and resource use data
Boberg et al (79)	 Costs were valued in Norwegian kroner (NOK), inflated to 2005 costs using the consumer price index and then converted to 2005 Euros
	 A cost of €132,903 was assigned to liver transplantation, which included donor organ harvesting and initial hospital stay
Horizon Scanning Centre (8)	 It has been identified that there were 777 hospital admissions in England due to PBC (ICD10 K74.3) in 2011 to 2012. This included 1,142 consultant episodes and 4,956 bed days
	 A pack of 100 x 250 mg tablets of ursodeoxycholic acid (Ursofalk) costs £31.88. Therefore, one year of treatment with 12– 16 mg/kg would cost between £478 and £574
Lemos et al (160)	 The annual cost of treatment with UDCA was reported as \$2,239.24 with the dose of 8 mg/kg; \$3,168.97 with the dose of 12 mg/kg; and \$4,098.71 with the dose of 15 mg/kg
Kogure et al (161)	 Kogure and colleagues acknowledge that the model for end-stage liver disease (MELD) score is useful for predicting medical expenses in living donor liver transplantation (LDLT)
	 Costs were valued in Japanese yen and then converted to US dollars using the rate of \$1.00 = ¥110.0
Longworth et al (140)	 Longworth and colleagues report the number of patients that, following transplant assessment, were listed for transplant. Of those listed, the number that went on to receive a liver transplant is also given
	 Resource use was evaluated from the point of assessment for transplantation. Resource use included the following: all subsequent inpatient stays, outpatient visits, high-cost/high-volume drugs, blood products, nutrition, physiotherapy sessions, dietician sessions, tests, treatments, and the length of the transplant operation provided at the transplant centre
	 1998–1999 GBP prices were used to value resources. These were derived by obtaining mean costs from the 6 UK liver transplant centres and weighting by number of transplantations performed
	 Based on data from the Royal College of Surgeons and Engineering and Physical Sciences Research Council an average cost per successful liver procurement is estimated at just over £7,200
	• The average cost for a patient with PBC (from point of assessment for transplant) was reported as £52,525 for a 27-month period
Gilroy et al(162)	 Resource use was measured for length of stay in ICU, period of ventilation after the liver transplantation procedure, intraoperative blood product use, surgical time and duration of hospital stay
	 Gilroy and colleagues reported that patients with a Mayo risk score greater than 7.8 used almost twice the resources of patients with a risk score less than 7.8.
Ratcliffe et al (139)	• Costs were calculated (In 1999 GBP) for liver transplants and waiting list costs in a variety of scenarios which model different

Table 63: Studies reporting resource data (original SLR and hand-searched publications)
	organ allocation policies								
	 Costs were measured in terms of net life expectancy, average net costs and overall cost-effectiveness 								
	 The ICER for the base case allocation policy was determined to be £11,557 								
	 The ICER for a transplant allocation policy based upon age (lowest age first) was £10,424 								
	 The ICER for a transplant allocation policy based upon the severity of the pre-transplant condition of the patient (with most severely ill patients given a lower priority) was £9,077 								
Pasha et al (78)	Within this analysis, only direct costs were included and were presented in 1995 US dollars								
	 \$2,543 was reported as the annual cost of UDCA 								
Kim et al (163)	 The figures below illustrate the relationship between pre-transplantation risk score and resource use following transplant, measured by days in ICU (A), hospital length of stay (B), and intraoperative blood transfusion requirements (amount of red blood cells transfused in litres ICI). 								
	These three resource components had previously been identified as contributing over 90% of the cost burden in the post- transplant phase								
	A a a a a b								



Kim et al(164)	• The majority of data reported by Kim and colleagues was not extracted on the basis that it did not relate to a PBC population. However, results of a multivariate regression analysis were reported, which captured the impact that certain variables had on non- fixed PBC costs.						
Kankaanpa et al(165)	• Kankaanpaa and colleagues report the economic costs of liver transplantations for PBC patients, in 1986 US dollars.						
Wright et al (154)	 Wright et al reported the costs of chronic hepatitis C. 						
	• Some of the costs could be applied to PBC as some hepatitis C health states are similar to PBC health states, including:						
	 Cirrhosis (compensated cirrhosis) (£1,138 per annum) 						
	 Decompensated cirrhosis (£9,120 per annum) 						
	 Hepatocellular carcinoma (£8,127 per annum) 						
Singh et al (4)	• Singh et al estimated the costs of liver transplantation in patients diagnosed with chronic hepatitis C and B in the UK						
	• The authors updated and analysed historical summary data from the original cohort study to generate updated cost estimates for liver transplantation, via conducting semi-structured interviews.						
	 The authors estimated a cost of £18,055 pre-transplantation (waiting list), £64,452 during the transplant phase, and £36,009 in two-years post-transplant. 						
	 The authors did not specify the year for which costs were calculated. 						

GBP, Great British Pounds; ICER, incremental cost-effectiveness ratio; ICU, intensive care unit; LDLT, living donor liver transplantation; MELD, model of end-stage liver disease; NOK, Norwegian Kroner; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

5.5.1.3 Update of systematic review

The systematic review was updated by a supplementary review, which was conducted in March 2016. Full details of the search strategy and inclusion/exclusion criteria are provided in Appendix 9.

Two additional conference abstracts were identified (155, 166). These publications are detailed below in Table 64.

Study, Year, Country	Cost and resource use data
Kim 2015 (155)	 Resource use for PBC and associated comorbidities including hyperlipidemia, hypothyroidism and extrahepatic autoimmune diseases were determined for South Korea
	 The nationwide total direct medical cost was \$8.5 million (2013 USD) at a cost of \$950 per patient
	 Epidemiological data were also reported
Pells 2015 (166)	• Resource savings for UDCA were reported, although it is not clear what comparators were identified.
	• The authors concluded that savings due to UDCA ranged between \$1,372 USD per annum (cost year not reported) to an extrapolated disease lifetime saving of €6,338 EUR (cost year not reported) per patient.
	 Both studies assessed cost from a direct hospital-based perspective only, and QoL and primary care costs were not considered.

Table 64: Studies reporting cost data (SLR update)

Abbreviations: EUR, Euros; PBC, primary biliary cholangitis/cirrhosis; QoL, quality of life; UDCA, ursodeoxycholic acid; USD, United States Dollars;

5.5.1.4 Appropriateness of NHS Ref costs/PbR tariffs

Unit costs for UDCA were determined using the British National Formulary (2016) and market share data, to calculate a weighted average cost per pack (167). In the base case analysis, the daily cost of UDCA is based on the observed UDCA dose as defined in the POISE trial protocol (15.4 mg/kg) (93).

NHS reference costs were used to determine the cost of outpatient appointments (average of consultant-led and non-consultant-led episodes) and costs for blood diagnostic tests.

5.5.1.5 Clinical expert assessment of applicability of cost and healthcare resource use values

The details of this process have been described previously in Section 5.3.3.

5.5.2 Intervention and comparators' costs and resource use

Results in this submission are based on the list price of OCA. However, OCA will be offered under a patient access scheme (PAS) and so the results presented in this section are for guidance only. Results of the economic analyses using the PAS price are presented in the accompanying PAS template, as requested by NICE, and are more reflective of the true cost-effectiveness of OCA.

Items	OCA (confidence interval)	Reference to section in submission	UDCA (confidence interval)	Reference to section in submission
Technology cost (list price)	£2,384.04 / 30 tablets	N/A	Weighted average of tablet formulations detailed in NHS Prescription Cost Analysis (167)	N/A
Total annual cost (based on list price)	£29,005.78	N/A	£655.33	N/A

Table 65: Unit costs associated with the technology in the economic model

Abbreviations: OCA, obeticholic acid; N/A, not applicable; UDCA, ursodeoxycholic acid.

5.5.3 Health-state costs and resource use

All costs were inflated to 2016 costs using the Hospital & Community Health Services (HCHS) index (168).

Health states	Items	Value	Source	Reference to section in submission
ALP: ≤ 200 U/L and Bili: Normal	Staff	£221.00 (1x Outpatient appointment, 1x outpatient follow- up)	KOL opinion	N/A
	Hospital costs	£27.00 (3 blood tests, 3 times per year, at a cost of £3)	NHS Schedule of Reference Costs – DAPS05	N/A
	Total	£248.00		
ALP: > 200 U/L and Bili: Normal	Staff	£345.00 (1x Outpatient appointment, 2x outpatient follow- up appointments)	KOL opinion	N/A
	Hospital costs	£27.00 (3 blood tests, 3 times per year, at a cost of £3)	NHS Schedule of Reference Costs – DAPS05	N/A
	Total	£496.00		
Bili: Abnormal and rising, or CC	Total	£6,254.00	KOL opinion: half the costs of DCC identified in Wright et al (2006)(154)	Table 63, page 161

Table 66: List of health states and a	associated costs in the economic model
---------------------------------------	--

Health states	Items	Value	Source	Reference to section in submission
DCC	Total	£12,509.00	Wright et al (2006)(154)	Table 63, page 161
HCC	Total	£11,147.00	Wright et al (2006)(154)	Table 63, page 161
Pre-transplant (end stage)	Total	£18,217.00	Singh et al (2014)(4)	Table 63, page 161
Re-emergence of PBC	Total	£248.00	Assumed identical to ALP: ≤ 200 u/L and Bili: Normal	Table 63, page 161
Liver transplant	Total	£65,029.00	Singh et al (2014)(4)	Table 63, page 161
Follow-up 1 year after liver transplantation	Total for 2 years divided by 2	£18,166.00	Singh et al (2014)(4)	Table 63, page 161
Follow-up 2 years after liver transplantation	Total for 2 years divided by 2	£18,166.00	Singh et al (2014)(4)	Table 63, page 161

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; DCC, decompensated cirrhosis; HCC, hepatocellular carcinoma; KOL, key opinion leader; PBC, primary biliary cholangitis/cirrhosis.

5.5.4 Adverse reaction unit costs and resource use

Of the three most frequent adverse events reported in POISE (pruritus, fatigue and nausea), only pruritus is treated in routine clinical practice. Therefore, only pruritus incurs a cost within the model (Table 67).

Adverse events	Items	Percentage of patients cost applies to	Value	Reference to section in submission
Pruritus	Staff (GP visit)	100%**	£54.00*	KOL input
	Cholestyramine cost / 327.10 days [†]	85%**	£105.59‡	KOL input
	Rifampicin cost / 327.10 days [†]	15%**	£191.77‡	KOL input
	Naltrexone cost / 327.10 days [†]	5%**	£228.39‡	KOL input
	Total (weighted average + staff costs)	N/A	£177.75	KOL input

Abbreviations: GP, general practitioner. *Sourced from BNF, 2016. **Sourced from KOL opinion. †Mean duration of treatment for UDCA and OCA therapies. ‡Sourced from NHS Schedule of Reference Costs.

5.5.5 Miscellaneous unit costs and resource use

The de novo cost-effectiveness analysis does not include any further miscellaneous costs or resource use.

5.6 Summary of base-case de novo analysis inputs and assumptions

5.6.1 Summary of base-case de novo analysis inputs

Patient utility data are reported in <u>Table 61</u>, and transition probabilities are reported in Table 50, Table 52 and Table 54. Costs for the intervention and comparators are reported in Table 65, and health state-specific costs are reported in Table 66. A summary of the remaining variables used in the economic analysis is provided in Table 68.

Variable	Value	CI (distribution)	Reference
Percentage of patients starting in state: ALP \leq 200 u/L and Bili: Normal	0.00%	_	Data on file
Percentage of patients starting in state: ALP > 200 u/L and Bili: Normal	76.85%	-	Data on file
Percentage of patients starting in state: Bili: abnormal and rising, or CC	23.15%	_	Data on file
Discount rate - cost	3.5%	_	NICE DSU (149)
Discount rate – outcomes	3.5%	_	NICE DSU (149)

Table 68: Summary of variables applied in the economic model

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CI, confidence interval; PBC, primary biliary cholangitis/cirrhosis.

5.6.2 Assumptions

Several assumptions were made in the model, following consultation and approval by KOLs:

- In the absence of PBC-specific inputs, hepatitis C virus (HCV) data were used as proxy for health state costs and utility values (discussed further in Sections 5.4.5 and 5.5.2)
- In the first year, patients can progress freely between the three PBC health states, as observed in the POISE OCA regimen arms. Afterwards, patients remain in the health state they are in at 12 months to reflect the sustained reduction in ALP and bilirubin demonstrated in the preliminary results from the long-term safety extension (LTSE) phase. In sensitivity analysis, OCA patients might progress from the "moderate risk PBC" state to the "high risk PBC" state with a low probability based on the literature, assuming they would follow a similar decompensation rate as UDCA GLOBE responders (152).
- UDCA patients can progress between PBC health states based on transition probabilities estimated based on the POISE patient-level data using both the

GLOBE and UK risk score, calibrated to reflect the liver transplant-free survivals (152)

- Patients suffering from decompensated cirrhosis have serious symptoms and complications from cirrhosis, such as ascites, hepatic encephalopathy or portal hypertension. This model followed previously published liver disease models that collapsed all the aforementioned states into a single state. (169, 170) The advantage of collapsing these states into a single state is to allow a patient to have several complications simultaneously, as occurs in practice
- A proportion of patients from the "high risk PBC" group is assumed to be eligible for liver transplant and added to the waiting list
- All patients in the decompensated cirrhosis health state are assumed to be candidates for liver transplantation
- Patients suffering from HCC are also candidates for liver transplantation
- Background mortality is assumed to be the same as the general population
- A probability of having a re-transplantation after recurrence of PBC has been included in the model
- Utility values reported for the following health states were decreased by reflect poorer outcomes in PBC patients versus HCV / HBV patients, according to interpretation of KOL input by Intercept:
 - o Decompensated cirrhosis
 - Pre transplant: utility at listing
 - Pre transplant: 3 months after listing
 - Pre transplant: 6 months after listing
 - Post-transplant: 3 months after transplant
 - Post-transplant: 6 months after transplant
 - Post-transplant: 12 months after transplant
 - o Post-transplant: 24 months after transplant

5.7 Base-case results

5.7.1 Base-case incremental cost effectiveness analysis results

All analyses presented in this submission are based on the list price of OCA. However, OCA will be offered under a patient access scheme (PAS) and so the results in this submission are for guidance only. It is important to note that the relative costeffectiveness and associated interpretation of the model results change significantly with the application of the proposed PAS price for OCA. Results of the economic analyses using the PAS price are presented in the accompanying PAS template, as requested by NICE, and are more reflective of the true cost-effectiveness of OCA.

Base-case results using the list price of OCA for the UDCA-intolerant population are presented in Table 69, which shows that OCA dose titration therapy is associated with incremental costs of **Control** and incremental QALYs of 6.91 compared with placebo, resulting in an ICER of **Control**. The results for the UDCA inadequate responder population are presented in Table 70. This table presents the results of the primary analysis of OCA dose titration in combination with UDCA vs UDCA monotherapy in UDCA inadequate responders. OCA dose titration + UDCA is associated with incremental costs of **Control** and incremental QALYs of 5.79, resulting in an ICER of

Table 69: Base-case results for UDCA-intolerant patients, using the list price of OCA

Technologies	Total			Incremental			ICER (<u>£</u>)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment (Placebo)	£103,233	11.30	6.61					
OCA titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Table 70: Base-case results for UDCA inadequate responders, using the list price of OCA

Technologies	Total			Incremental			ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA + Placebo	£96,977	12.35	7.85					
OCA titration + UDCA		16.75	13.64		4.40	5.79		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid.

5.7.2 Clinical outcomes from the model

Table 71 presents the model outcomes for both therapies for UDCA-intolerant patients. The results show a lower number of patients with abnormal bilirubin, cases of decompensated cirrhosis, liver transplants and liver-related deaths in the OCA-treated group.

Outcome	OCA titration	No treatment (placebo)	Incremental
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients			
Total number of cases of decompensated cirrhosis per 1,000 patients			
Total liver transplants per 1,000 patients			
Total liver related deaths per 1,000 patients			

Table 71 Summary of model outcomes for UDCA-intolerant population

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Table 72 presents the model outcomes for both therapies for UDCA inadequate responders. As with the UDCA-intolerant population, the results show a lower number of patients with abnormal bilirubin, cases of decompensated cirrhosis, liver transplants and liver-related deaths in the OCA-treated group.

Outcome	OCA titration + UDCA	UDCA + Placebo	Incremental	
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients				
Total number of cases of decompensated cirrhosis per 1,000 patients				
Total liver transplants per 1,000 patients				
Total liver related deaths per 1,000 patients				

Table 72: Summary of model outcomes for UDCA inadequate responder population

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

5.7.3 Disaggregated results of the base case incremental cost effectiveness analysis

The disaggregated results show overall superiority of OCA over its comparator in both UDCA-intolerant and UDCA inadequate responder populations.

Health state	OCA Titration	No treatment (Placebo)	Increment	Absolute increment
ALP: ≤ 200 u/L and Bili: Normal	5.981	0.000	-5.981	5.981
ALP: > 200 u/L and Bili: Normal	6.796	2.044 -4.752		4.752
Bili: Abnormal and rising, or CC	0.394	2.515	2.121	2.121
Discontinuation	0.009	0.000	-0.009	0.009
Decompensated cirrhosis	0.106	0.651	0.544	0.544
HCC	0.011	0.065	0.055	0.055
Pre transplant (end stage)	0.034	0.208	0.174	0.174
Liver transplant	0.005	0.031	0.026	0.026
Post liver transplant	0.136	0.822	0.686	0.686
Re-emergence of PBC	0.046	0.270	0.223	0.223
Total	13.520	6.606	-6.913	14.572

Table 73: Summary of QALY gain by health state for UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Health state	QALY OCA + UDCA titration	QALY UDCA + Placebo	Increment	Absolute increment
ALP: ≤ 200 u/L and Bili: Normal	5.981	0.000	-5.981	5.981
ALP: > 200 u/L and Bili: Normal	6.974	3.867	-3.107	3.107
Bili: Abnormal and rising, or CC	0.365	2.222	1.857	1.857
Discontinuation	0.009	0.004	-0.006	0.006
Decompensated cirrhosis	0.098	0.569	0.471	0.471
HCC	0.010	0.057	0.048	0.048
Pre transplant (end stage)	0.031	0.182	0.151	0.151
Liver transplant	0.005	0.027	0.023	0.023
Post liver transplant	0.125	0.701	0.577	0.577
Re-emergence of PBC	0.042	0.222	0.180	0.180
Total	13.641	7.852	-5.789	12.399

Table 74: Summary of QALY gain by health state for UDCA inadequate responders

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Health state	OCA Titration	No treatment (Placebo)	Increment	Absolute increment
ALP: ≤ 200 u/L and bili: Normal		£0.00		
ALP: > 200 u/L and bili: Normal		£1,207.08		
Bili: Abnormal and rising, or CC		£28,597.68		
Discontinuation		£0.00		
Decompensated cirrhosis		£21,284.77		
HCC		£1,620.43		
Pre transplant (end stage)		£9,890.90		
Liver transplant		£14,310.87		
Post liver transplant		£26,221.30		
Re-emergence of PBC		£99.80		
Total		£103,232.81		

Table 75: Summary of costs by health state for UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Health state	OCA + UDCA	UDCA + Placebo	Increment	Absolute increment
	titration			
ALP: \leq 200 u/L and Bili: Normal		£0.00		
ALP: > 200 u/L and Bili: Normal		£5,349.00		
Bili: Abnormal and rising, or CC		£27,926.48		
Discontinuation		£14.48		
Decompensated cirrhosis		£18,619.88		
HCC		£1,422.13		
Pre transplant (end stage)		£8,665.20		
Liver transplant		£12,526.26		
Post liver transplant		£22,371.06		
Re-emergence of PBC		£82.10		
Total		£96,976.58		

Table 76: Summary of costs by health state for UDCA inadequate responders

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

5.8 Sensitivity analyses

5.8.1 Probabilistic sensitivity analysis

5.8.1.1 Inputs

Patient characteristics (i.e. weight and baseline health state distribution) were included in the PSA alongside all other generic inputs (transition probabilities, clinical inputs and quality of life). Baseline distribution across the four PBC-specific health states were assumed to follow a Dirichlet distribution. The four variables were drawn from independent Gamma or Normal distributions contingent on whether the sample size of each category was below 340 or not, respectively. Then the required four probabilities were re-estimated dividing the sampled patient number in each category by the total sampled population. Quality of life parameters were assumed to follow a Beta distribution, costs a Gamma distribution and other continuous parameters a Normal distribution. The Gamma distribution parameters to use for the disease management costs were derived from Wright et al (154). The mean cost and standard errors reported by the authors were inflated from 2002/2003 to 2014/2015 and the parameters were estimated based on the newly inflated inputs as described below in Table 77. Where the variance was unknown, the lower and upper confidence intervals were estimated assuming a 20% variability around the mean. The PSA was performed by running 1,000 Monte Carlo simulations.

Table 77: Generic PSA inputs: UDCA-intolerant population

Parameter	Distribution and parameters	Expected value	Source/Comments
Baseline health state distribution – 200 units / L ALP threshold			
ALP ≤ 200 units / L and NB	Dirichlet	0.0%	POISE trial (93)
ALP > 200 units / L and NB	Dirichlet	76.85%	POISE trial (93)
$(ALP \le or > 200 units / L) and AB$	Dirichlet	23.15%	POISE trial (93)
Patient characteristics			
Weight (kg)	Normal: μ = 69.8, δ = 13.9	69.8	POISE trial (93)
Discontinuation probabilities			
No treatment (Placebo)	Beta: α = 3, β = 70	0.00%	POISE trial (93)
OCA titration	Beta: α = 7, β = 64	9.86%	POISE trial (93)
Adverse events			
No treatment (Placebo)			
Fatigue	Beta: α = 8, β = 65	0.00%	POISE trial (93)
Pruritus	Beta: α = 27, β = 46	0.00%	POISE trial (93)
Nausea	Beta: α = 4, β = 69	0.00%	POISE trial (93)
OCA titration			(93)
Fatigue	Beta: α = 6, β = 64	8.57%	POISE trial (93)
Pruritus	Beta: α = 35, β = 35	50.0%	POISE trial (93)
Nausea	Beta: α = 3, β = 67	4.29%	POISE trial (93)
Utility parameters			
ALP ≤ 200 units / L and NB	Beta: α=202.37, β=38.55	0.84	Younossi et al. (2001) (64)
ALP > 200 units / L and NB	Beta: α=202.37, β=38.55	0.84	Younossi et al. (2001) (64)
$(ALP \le or > 200 units / L) and AB$	Beta: α=47, β=39	0.55	Wright et al. (2006) (154)

Parameter	Distribution and parameters	Expected value	Source/Comments
Decompensated cirrhosis			Wright et al. (2006) (154) and assumed decrement of
Hepatocellular carcinoma	Beta: α=124, β=151	0.45	Wright et al. (2006) (154)
Liver transplant			Wright et al. (2006) (154) and assumed decrement of
Post-liver transplant			Wright et al. (2006) (154) and assumed decrement of
Re-emergence of PBC	Beta: α=265.98, β=131.00	0.67	KOL input
Transition probabilities			
From ALP \leq 200 units / L and NB to Bili: Abnormal and rising, or CC	Beta: α=89.44, β=1,225.60	6.80%	Assumption based on 20% variance
From ALP > 200 units / L and NB to Bili: Abnormal and rising, or CC	Beta: α=89.44, β=1,225.60	6.80%	Assumption based on 20% variance
From Bili: Abnormal and rising, or CC to decompensated cirrhosis	Beta: α=93.51, β=3,503.69	2.6%	Assumption based on 20% variance
From Bili: Abnormal and rising, or CC to HCC	Beta: α=2, β=136	0.35%	STA 330(142)
From decompensated cirrhosis to pre-liver transplant	Beta: α=94.55, β=6064.94	1.53%	Assumption based on 20% variance
From decompensated cirrhosis to death	Beta: α=92.17, β=2222.83	3.98%	Assumption based on 20% variance
From decompensated cirrhosis to HCC	Beta: α=2, β=136	0.35%	STA 330 (142)
From HCC to death	Beta: α=83.31, β=552.18	13.11%	STA 330 (142)
From pre-LT to LT	Beta: α=86.13, β=757.45	10.21%	Data on file
From pre-LT to death	Beta: α=93.78, β=3930.59	2.33%	Data on file

Parameter	Distribution and parameters	Expected value	Source/Comments
From LT to death	Beta: α=16, β=61	5.72%	STA 330(142)
From post-LT to re-emergence of PBC	Uniform: α=0.20, β=0.25	0.64%	Assumption based on 20% variance
From post-LT to LT	Beta: α=3, β=483	0.01%	Neuberger et al. (2003) (156)
From post-LT to death	Beta: α=23, β=379	1.46%	STA 330 (142)
From re-emergence of PBC to LT	Beta: α=3, β=483	0.01%	Neuberger et al. (2003) (156)
Disease management costs			
DM costs: ALP: ≤ 200 units / L and Bili: Normal	Gamma, α=96.04, β=2.58	<mark>£</mark> 248.00	KOL input
DM costs: ALP: > 200 units / L and Bili: Normal	Gamma, α=96.04, β=5.16	£496.00	KOL input
DM costs: Bili: Abnormal and rising, or CC	Gamma, α=96.04, β=65.12	£6,254.00	KOL opinion: half the costs of DCC identified in Wright et al (2006)(154)
DM costs: Decompensated cirrhosis	Gamma, α=96.04, β=130.25	£12,509.00	Wright et al (2006)(154)
DM costs: HCC	Gamma, α=96.04, β=116.07	£11,147.00	Wright et al (2006)(154)
DM costs: Pre transplant (end stage)	Gamma, α=96.04, β=189.69	£18,217.00	Singh et al (2014)(4)
DM costs: Re-emergence of PBC	Gamma, α=96.04, β=2.58	£248.00	KOL input
Liver transplant costs	Gamma, α=96.04, β=677.13	£65,029.00	Singh et al (2014)(4)
Post-liver transplant costs: Year 1	Gamma, α=96.04, β=189.16	£18,166.00	Singh et al (2014)(4)
Post-liver transplant costs: Year 2	Gamma, α=96.04, β=189.16	£18,166.00	Singh et al (2014)(4)
Adverse event costs			
Adverse event costs: pruritus	Gamma, α=96.04, β=1.85	£177.75	KOL input

Table 78: Generic PSA inputs: UDCA inadequate responder population

Parameter	Distribution and parameters	Expected value	Source/Comments
Baseline health state distribution – 200 units / L ALP threshold			
ALP ≤ 200 units / L and NB	Dirichlet	0.0%	POISE trial (93)
ALP > 200 units / L and NB	Dirichlet	76.85%	POISE trial (93)
(ALP \leq or > 200 units / L) and AB	Dirichlet	23.15%	POISE trial (93)
Patient characteristics			
Weight (kg)	Normal: μ = 69.8, δ = 13.9	69.8	POISE trial (93)
Discontinuation probabilities			
UDCA + Placebo	Beta: α = 3, β = 70	4.11%	POISE trial (93)
OCA + UDCA titration	Beta: α = 7, β = 64	9.86%	POISE trial (93)
Adverse events			
UDCA + Placebo			
Fatigue	Beta: α = 8, β = 65	10.96%	POISE trial (93)
Pruritus	Beta: α = 27, β = 46	36.99%	POISE trial (93)
Nausea	Beta: α = 4, β = 69	5.48%	POISE trial (93)
OCA + UDCA titration			
Fatigue	Beta: α = 6, β = 64	8.57%	POISE trial (93)
Pruritus	Beta: α = 35, β = 35	50.0%	POISE trial (93)
Nausea	Beta: α = 3, β = 67	4.29%	POISE trial (93)
Utility parameters			
ALP ≤ 200 units / L and NB	Beta: α=202.37, β=38.55	0.84	Younossi et al. (2001) (64)
ALP > 200 units / L and NB	Beta: α=202.37, β=38.55	0.84	Younossi et al. (2001) (64)

Parameter	Distribution and parameters	Expected value	Source/Comments
$(ALP \le or > 200 units / L) and AB$	Beta: α=47, β=39	0.55	Wright et al. (2006) (154)
Decompensated cirrhosis			Wright et al. (2006) (154) and assumed decrement of
Hepatocellular carcinoma	Beta: α=124, β=151	0.45	Wright et al. (2006) (154)
Liver transplant			Wright et al. (2006) (154) and assumed decrement of
Post-liver transplant			Wright et al. (2006) (154) and assumed decrement of
Re-emergence of PBC	Beta: α=265.98, β=131.00	0.67	KOL input
Transition probabilities			
From ALP \leq 200 units / L and NB to Bili: Abnormal and rising, or CC	Beta: α=94.80, β=7345.49	1.27%	Assumption based on 20% variance
From ALP > 200 units / L and NB to Bili: Abnormal and rising, or CC	Beta: α=92.98, β=2863.30	3.15%	Assumption based on 20% variance
From Bili: Abnormal and rising, or CC to decompensated cirrhosis	Beta: α=93.51, β=3503.69	2.6%	Assumption based on 20% variance
From Bili: Abnormal and rising, or CC to HCCd	Beta: α=2, β=136	0.35%	STA 330(142)
From decompensated cirrhosis to pre-liver transplant	Beta: α=94.55, β=6064.94	1.53%	Assumption based on 20% variance
From decompensated cirrhosis to death	Beta: α=92.17, β=2222.83	3.98%	Assumption based on 20% variance
From decompensated cirrhosis to HCC	Beta: α=2, β=136	0.35%	STA 330 (142)
From HCC to death	Beta: α=83.31, β=552.18	13.11%	STA 330 (142)
From pre-LT to LT	Beta: α=86.13, β=757.45	10.21%	Data on file
From pre-LT to death	Beta: α=93.78, β=3930.59	2.33%	Data on file
From LT to death	Beta: α=16, β=61	5.72%	STA 330(142)
From post-LT to re-emergence of PBC	Uniform: α=0.20, β=0.25	0.64%	Assumption based on 20% variance
From post-LT to LT	Beta: α=3, β=483	0.01%	Neuberger et al. (2003) (156)

Parameter	Distribution and parameters	Expected value	Source/Comments
From post-LT to death	Beta: α=23, β=379	1.46%	STA 330 (142)
From re-emergence of PBC to LT	Beta: α=3, β=483	0.01%	Neuberger et al. (2003) (156)
Disease management costs			
DM costs: ALP: ≤ 200 units / L and Bili: Normal	Gamma, α=96.04, β=2.58	£248.00	KOL input
DM costs: ALP: > 200 units / L and Bili: Normal	Gamma, α=96.04, β=5.16	£496.00	KOL input
DM costs: Bili: Abnormal and rising, or CC	Gamma, α=96.04, β=65.12	£6,254.00	KOL opinion: half the costs of DCC identified in Wright et al (2006)(154)
DM costs: Decompensated cirrhosis	Gamma, α=96.04, β=130.25	£12,509.00	Wright et al (2006)(154)
DM costs: HCC	Gamma, α=96.04, β=116.07	£11,147.00	Wright et al (2006)(154)
DM costs: Pre transplant (end stage)	Gamma, α=96.04, β=189.69	£18,217.00	Singh et al (2014)(4)
DM costs: Re-emergence of PBC	Gamma, α=96.04, β=2.58	£248.00	KOL input
Liver transplant costs	Gamma, α=96.04, β=677.13	£65,029.00	Singh et al (2014)(4)
Post-liver transplant costs: Year 1	Gamma, α=96.04, β=189.16	£18,166.00	Singh et al (2014)(4)
Post-liver transplant costs: Year 2	Gamma, α=96.04, β=189.16	£18,166.00	Singh et al (2014)(4)
Adverse event costs			
Adverse event costs: pruritus	Gamma, α=96.04, β=1.85	£177.75	KOL input

Table 79: Incremental cost effectiveness results of PSA for the UDCA intolerant population, using the list price of OCA

Technologies	Total			Incremental			ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
PSA results								
No treatment (placebo)	£103,439	11.33	6.77	-	-	-	-	-
OCA Titration		16.64	13.52		5.31	6.75		
Base case deterministic results								
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Table 80: Incremental cost effectiveness results of PSA for the UDCA inadequate responder population, using the list price of OCA

Technologies	Total			Incremental			ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
PSA results								
UDCA + placebo	£97,044	12.38	7.88	-	-	-	-	-
OCA + UDCA titration		16.75	13.65		4.37	5.77		
Base case deterministic results				•				
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA + UDCA titration		16.75	13.64		4.40	5.79		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Abbreviations: OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

intolerant population), using the OCA list price

Figure 32: Cost-effectiveness acceptability curve for OCA titration versus placebo (UDCAintolerant population), using the OCA list price

Abbreviations: OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Figure 33: Scatter plot for PSA (UDCA inadequate responders population), using the OCA list price



Abbreviations: OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Abbreviations: OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

5.8.1.2 Discussion of variation between base case and PSA results

The PSA shows that the model is linear and there is little variation between the results of the base case deterministic results and the probabilistic results. The deterministic results can reliably be used to generate scenarios.

5.8.2 Deterministic sensitivity analysis

5.8.2.1 Inputs

The generic inputs that are varied in the DSA are the discontinuation probabilities, probabilities of having an adverse event, utility weights, transition probabilities, discounting, and the probability of death for the general population.

The model also allows the user to run scenario analyses by assuming that all patients enter the model in each of the four PBC-specific health state at a time.

|--|

Parameter	Base case	Min	Max	Source
Discontinuation probabilities				
No treatment (Placebo)	0.0%	0.0%	0.0%	No variance is considered for this parameter
OCA titration	9.86%	4.12%	17.73%	POISE trial; LB and UB are limits of 95% CI estimated assuming a BETA distribution (93)
Adverse events				
No treatment (Placebo)				
Pruritus	0.0%	0.0%	0.0%	No variance is considered for this parameter
OCA titration				
Pruritus	50.0%	38.4%	61.6%	POISE trial; LB and UB are limits of 95% CI estimated assuming a BETA distribution (93)
Utility parameters				
ALP \leq 200 units / L and NB	0.84	0.79	0.88	Younossi et al. (2001) (64); LB and UB are limits of 95% CI estimated assuming a BETA distribution
ALP > 200 units / L and NB	0.84	0.79	0.88	Younossi et al. (2001) (64); LB and UB are limits of 95% CI estimated assuming a BETA distribution
(ALP \leq or > 200 units / L) and AB	0.55	0.44	0.65	Wright et al. (2006) (154); LB and UB are limits of 95% CI estimated assuming a BETA distribution
Decompensated cirrhosis				Wright et al. (2006) (154); decrement; LB and UB assumed equal to +/- 20% of the base case
Hepatocellular carcinoma	0.45	0.39	0.51	Wright et al. (2006) (154); LB and UB are limits of 95% CI estimated assuming a BETA distribution
Liver transplant				Wright et al. (2006) (154); decrement; LB and UB assumed equal to +/- 20% of the base case
Post-liver transplant				Wright et al. (2006) (154); decrement; LB and UB assumed equal to +/- 20% of the base case
Re-emergence of PBC	0.67	0.62	0.72	Wright et al. (2006) (154); LB and UB are limits of 95%

Parameter	Base case	Min	Max	Source
				CI estimated assuming a BETA distribution
Transition probabilities				
From ALP ≤ 200 units / L and NB to Bili: Abnormal and rising, or CC	6.8%	5.4%	8.2%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From ALP > 200 units / L and NB to Bili: Abnormal and rising, or CC	6.8%	5.4%	8.2%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From Bili: Abnormal and rising, or CC to decompensated cirrhosis	2.6%	2.08%	3.12%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From severe PBC to HCC	0.35%	0.04%	1.02%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to transplant	1.53%	1.23%	1.84%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to death	3.98%	3.19%	4.78%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to HCC	0.35%	0.0%	1.02%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From HCC to death	13.11%	10.49%	15.73%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From pre-LT to LT	10.21%	8.17%	12.25%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From pre-LT to death	2.33%	1.86%	2.80%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From LT to death	5.72%	3.29%	8.68%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From post-LT to re-emergence of PBC	0.64%	0.56%	0.72%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a Uniform distribution
From post-LT to LT	0.01%	0.00%	0.03%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From post-LT to death	1.46%	0.93%	2.11%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution

Parameter	Base case	Min	Мах	Source
From re-emergence of PBC to LT	0.01%	0.00%	0.03%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
Discount rates				
Discount rates – outcomes	3.5%	0.0%	6.0%	NICE - Guide to the methods of technology appraisal (171)
Discount rates – costs	3.5%	0.0%	6.0%	NICE - Guide to the methods of technology appraisal (171)
AE management costs				
Pruritus costs	£177.75	£142.20	£213.30	Calculation; LB and UB assumed equal to +/-20% of the base case
Disease management costs				
ALP ≤ 200 units / L and NB	£248.00	£198.40	£297.60	Calculation; LB and UB based on the CI from Gamma distribution
ALP > 200 units / L and NB	£496.00	£396.80	£595.20	Calculation; LB and UB based on the CI from Gamma distribution
(ALP \leq or > 200 units / L) and AB	£6,254.00	£5,003.20	£7,504.80	Calculation; LB and UB based on the CI from Gamma distribution and assuming ½ variance of DCC
Decompensated cirrhosis	£12,509.00	£10,007.20	£15,010.80	Calculation; LB and UB based on the CI from Gamma distribution
нсс	£11,147.00	£8,917.60	£13,376.40	Calculation; LB and UB based on the CI from Gamma distribution
Pre transplant (end stage)	£18,217.00	£14,573.60	£21,860.40	Calculation; LB and UB assumed equal to +/-20% of the base case
Liver transplant	£65,029.00	£198.40	£297.60	Calculation; LB and UB assumed equal to +/-20% of the base case
Post-liver transplant - Year 1	£18,166.00	£52,023.20	£78,034.80	Calculation; LB and UB assumed equal to +/-20% of the base case
Post-liver transplant - Year 2	£18,166.00	£14,532.80	£21,799.20	Calculation; LB and UB assumed equal to +/-20% of the base case

Parameter	Base case	Min	Мах	Source
Re-emergence of PBC	£248.00	£198.40	£297.60	Calculation; LB and UB based on the CI from Gamma distribution

Table 82: Generic DSA inputs: UDCA inadequate responder population

Parameter	Base case	Min	Max	Source
Discontinuation probabilities				
UDCA + Placebo	4.11%	0.87%	9.68%	POISE trial; LB and UB are limits of 95% CI estimated assuming a BETA distribution (93)
OCA + UDCA titration	9.86%	4.12%	17.73%	POISE trial; LB and UB are limits of 95% CI estimated assuming a BETA distribution (93)
Adverse events				
UDCA + Placebo				
Pruritus	36.99%	26.36%	48.29%	No variance is considered for this parameter
OCA + UDCA titration				
Pruritus	50.0%	38.4%	61.6%	POISE trial; LB and UB are limits of 95% CI estimated assuming a BETA distribution (93)
Utility parameters				
ALP ≤ 200 units / L and NB	0.84	0.79	0.88	Younossi et al. (2001) (64); LB and UB are limits of 95% CI estimated assuming a BETA distribution
ALP > 200 units / L and NB	0.84	0.79	0.88	Younossi et al. (2001) (64); LB and UB are limits of 95% CI estimated assuming a BETA distribution
$(ALP \le or > 200 units / L)$ and AB	0.55	0.44	0.65	Wright et al. (2006) (154); LB and UB are limits of 95% CI estimated assuming a BETA distribution
Decompensated cirrhosis				Wright et al. (2006) (154); decrement; LB and UB assumed equal to +/- 20% of the base case
Hepatocellular carcinoma	0.45	0.39	0.51	Wright et al. (2006) (154); LB and UB are limits of 95% CI estimated assuming a BETA distribution
Liver transplant				Wright et al. (2006) (154); decrement; LB and UB

Parameter	Base case	Min	Max	Source
				assumed equal to +/- 20% of the base case
Post-liver transplant				Wright et al. (2006) (154); decrement; LB and UB assumed equal to +/- 20% of the base case
Re-emergence of PBC	0.67	0.62	0.72	Wright et al. (2006) (154); LB and UB are limits of 95% CI estimated assuming a BETA distribution
Transition probabilities				
From ALP ≤ 200 units / L and NB to Bili: Abnormal and rising, or CC	6.8%	5.4%	8.2%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From ALP > 200 units / L and NB to Bili: Abnormal and rising, or CC	6.8%	5.4%	8.2%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From Bili: Abnormal and rising, or CC to decompensated cirrhosis	2.6%	2.08%	3.12%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From severe PBC to HCC	0.35%	0.04%	1.02%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to transplant	1.53%	1.23%	1.84%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to death	3.98%	3.19%	4.78%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From decompensated cirrhosis to HCC	0.35%	0.0%	1.02%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From HCC to death	13.11%	10.49%	15.73%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From pre-LT to LT	10.21%	8.17%	12.25%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From pre-LT to death	2.33%	1.86%	2.80%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From LT to death	5.72%	3.29%	8.68%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From post-LT to re-emergence of PBC	0.64%	0.56%	0.72%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a Uniform distribution

Parameter	Base case	Min	Мах	Source
From post-LT to LT	0.01%	0.00%	0.03%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From post-LT to death	1.46%	0.93%	2.11%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
From re-emergence of PBC to LT	0.01%	0.00%	0.03%	20% variance assumed; LB and UB are limits of 95% CI estimated assuming a BETA distribution
Discount rates				
Discount rates – outcomes	3.5%	0.0%	6.0%	NICE - Guide to the methods of technology appraisal (171)
Discount rates – costs	3.5%	0.0%	6.0%	NICE - Guide to the methods of technology appraisal (171)
AE management costs				
Pruritus costs	£177.75	£142.20	£213.30	Calculation; LB and UB assumed equal to +/-20% of the base case
Disease management costs				
ALP ≤ 200 units / L and NB	£248.00	£198.40	£297.60	Calculation; LB and UB based on the CI from Gamma distribution
ALP > 200 units / L and NB	£496.00	£396.80	£595.20	Calculation; LB and UB based on the CI from Gamma distribution
(ALP \leq or > 200 units / L) and AB	£6,254.00	£5,003.20	£7,504.80	Calculation; LB and UB based on the CI from Gamma distribution and assuming ½ variance of DCC
Decompensated cirrhosis	£12,509.00	£10,007.20	£15,010.80	Calculation; LB and UB based on the CI from Gamma distribution
НСС	£11,147.00	£8,917.60	£13,376.40	Calculation; LB and UB based on the CI from Gamma distribution
Pre transplant (end stage)	£18,217.00	£14,573.60	£21,860.40	Calculation; LB and UB assumed equal to +/-20% of the base case
Liver transplant	£65,029.00	£198.40	£297.60	Calculation; LB and UB assumed equal to +/-20% of the base case

Parameter	Base case	Min	Мах	Source
Post-liver transplant - Year 1	£18,166.00	£52,023.20	£78,034.80	Calculation; LB and UB assumed equal to +/-20% of the base case
Post-liver transplant - Year 2	£18,166.00	£14,532.80	£21,799.20	Calculation; LB and UB assumed equal to +/-20% of the base case
Re-emergence of PBC	£248.00	£198.40	£297.60	Calculation; LB and UB based on the CI from Gamma distribution

5.8.2.2 Results

All sensitivity analyses were performed at the list price of OCA (£29,005.78). Analyses using the PAS price, which are more reflective of the true cost-effectiveness of OCA, are presented in the accompanying PAS template.

Figure 35: Tornado diagram for OCA titration versus no treatment (placebo) in the UDCA intolerant population, using the list price of OCA



Figure 36: Tornado diagram for OCA + UDCA titration versus UDCA + placebo in the UDCA inadequate responder population, using the list price of OCA



5.8.3 Scenario analysis

5.8.3.1 Scenario 1: use of original HCV utility values

A scenario analysis was performed where all utility values that had been decremented from their previously reported values were set to their (HBV/HCV-specific) original values. The values changed are shown in Table 83. The results are presented in Table 84 and Table 85. OCA is still cost-effective despite using the more conservative HBV/HCV – specific utility values.

State	Base case utility value	Scenario utility value	Justification
Decompensated cirrhosis	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (142)
Pre-transplant: utility at listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (142)
Pre-transplant: 3 months after listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (142)
Pre-transplant: 6 months after listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (142)
Liver transplant: 3 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (142)
Liver transplant: 6 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (142)
Liver transplant: 12 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (142)
Liver transplant: 24 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (142)

Table 83	· Parameters	changed for	scenario	1
I able ob	. Falameters	changed for	SCENANU	

Technologies	Total			h	ncremental	I	ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment (Placebo)	£103,233	11.30	6.64	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.88		

Table 84: Scenario 1 results for UDCA intolerant patients, using the list price of OCA

Table 85: Scenario 1 results for UDCA inadequate responders, using the list price of OCA

Technologies	Total			h	Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA + Placebo	£96,977	12.35	8.11	-	-	-	-	-
OCA + UDCA titration		16.75	13.64		4.40	5.75		

5.8.3.2 Scenario 2: use of alternative transition probabilities

This scenario utilises alternative transition probabilities for pre-liver transplant to liver transplant, and pre-liver transplant to death, gathered from an ad-hoc analysis of PBC-specific data from OPTN (data on file).

Transition probability	Base case TP	Scenario TP	Justification						
Pre-LT to LT	0.35	0.44	Based on ad-hoc data analysis of OPTN PBC data						
Pre-LT to death	0.09	0.21	Based on ad-hoc data analysis of OPTN PBC data						

Table 86: Parameters changed for scenario 2

Table 87: Scenario 2 results for UDCA intolerant patients, using the list price of OCA

Technologies	Total			Incremental			ICER (<mark>£</mark>)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment (Placebo)	£95,697	10.93	6.56	-	-	-	-	-
OCA titration		16.59	13.51		5.66	6.95		

Table 88: Scenario 2 results for UDCA inadequate responders, using the list price of OCA

Technologies	Total			h	Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA + Placebo	£90,516	12.04	7.82	-	-	-	-	-
OCA + UDCA titration		16.69	13.63		4.66	5.82		

5.8.4 Summary of sensitivity analyses results

The model was generally insensitive to changes in individual inputs. The primary drivers of the cost-effectiveness results (excluding discounting rates and OCA price) were:

- UDCA-intolerant population:
 - Transition probability for ALP > 200 u/L and Bili: normal to Bili: abnormal and rising, or CC
 - o Utility value for Bili: abnormal and rising or CC
 - Utility: ALP \leq 200 u/L and Bili: normal
- UDCA inadequate responder population:
 - Transition probability for ALP > 200 u/L and Bili: normal to Bili: abnormal and rising, or CC
 - Utility value for Bili: abnormal and rising or CC
 - Utility: $ALP \leq 200 \text{ u/L}$ and Bili: normal

5.9 Subgroup analysis

No subgroups of patients were considered for this analysis.

5.10 Validation

5.10.1 Validation of de novo cost-effectiveness analysis

Internal verification of calculations was performed by the primary modeller in the first instance and checked by a second modeller involved with the model development (internal peer review). The economic model was also examined by two modellers external to the technical model development process (external peer review) (172). Verification techniques included:

- Face validity: testing that the model meets expectations based on simple calculations
- Model behaviour: testing whether varying model inputs has the expected directional effect
- Internal consistency: model outputs were compared against POISE
- Cell-by-cell checks of calculations: manual inspection of formulae
- Use of logical scenario checks and the rebuilding of important parts of the model
- A complete cross-check of inputs, sources, and supporting documentation.

The model produces outcomes at multiple time points to allow comparison against published sources.

5.11 Interpretation and conclusions of economic evidence

Results in this submission are based on the list price of OCA. However, OCA will be offered under a patient access scheme (PAS) and so the results presented in this submission are for guidance only. It is important to note that the relative cost-effectiveness and associated interpretation of the model results change significantly with the application of the proposed PAS price for OCA. Results of the economic analyses using the PAS price are presented in the accompanying PAS template, as requested by NICE, and are more reflective of the true cost-effectiveness of OCA.

The results of the economic analysis using the list price of OCA give an ICER of for UDCA-intolerant patients and an ICER of **Constant of** in UDCA inadequate responders. For both patient groups, the decrease in the number of liver transplants required for PBC patients is 84% when patients are treated with OCA.

The key limitation of this analysis is the absence of PBC-specific utility data. This was addressed by incorporating HBV/HCV utility data instead in similar health states, for example, pre-transplant stages where both hepatitis and PBC patients would be in advanced health states with comparatively poor quality of life. To reflect the generally worse outcomes and more rapid disease progression of PBC patients compared to HBV/HCV patients, utilities for disease states (excluding HCC and the re-emergence of PBC) were decremented by

The deterministic sensitivity analysis shows that the model is fairly insensitive to individual variable changes, and is robust in that no single factor can easily alter the ICER.
6 Assessment of factors relevant to the NHS and other parties

6.1 *Population: people eligible for treatment*

All patients with PBC who have had an inadequate response to UDCA or who are unable to tolerate UDCA are eligible for treatment with OCA. An estimation of the number of patients who are eligible for treatment with OCA is shown in Table 89, and is calculated from the following parameters:

- 1. Total England + Wales populations (mid-2015 data).
- 2. Prevalence of PBC in England
- 3. PBC patients diagnosed
- 4. PBC patients under treater care
- 5. PBC patients eligible for treatment

As there are currently no other treatments available for this indication, no displacement of other medicines is assumed.

	iser er parler	ne engine ie	a californi il		
	Year 1	Year 2	Year 3	Year 4	Year 5
Population, England and Wales	57,885,400	57,885,400	57,885,400	57,885,400	57,885,400
Current prevalence, %	0.04%	0.04%	0.04%	0.04%	0.04%
Prevalent population	22,691	22,691	22,691	22,691	22,691
Patients diagnosed, %	52.00%	52.00%	52.00%	52.00%	52.00%
Patients diagnosed	11,799	11,799	11,799	11,799	11,799
Patients under treater care, %	92.00%	92.00%	92.00%	92.00%	92.00%
Patients under treater care	10,855	10,855	10,855	10,855	10,855
Patients eligible for treatment, %	35.00%	35.00%	35.00%	35.00%	35.00%
Patients eligible for treatment	3,799	3,799	3,799	3,799	3,799
Discontinuation rate	30%	30%	30%	30%	30%
Patients eligible for treatment minus patients who would be anticipated to discontinue	2,660	2,660	2,660	2,660	2,660
Proportion of eligible patients treated with new medicine	9.08%	30.32%	40.94%	49.59%	58.51%
Number of patients treated in each year	241	806	1,089	1,319	1,556

Table 89: Estimation of number of patients eligible for treatment with OCA

6.2 Costs included

The following costs were taken into account within budget impact calculations:

- Technology costs
 - o OCA
 - UDCA (for UDCA tolerant patients who have had an inadequate response to UDCA alone- 92.59% of the eligible PBC patient population)

Details of unit costs for the above are presented in Section 5.5.

6.3 Resource savings

Resource savings were not included as they were best presented as part of the costutility analysis where they would be captured within the extended time horizon. It is likely that were will be some costs savings related to a decreased number of liver transplants in PBC patients. 100% of the eligible patient population were assumed to be treated with OCA. No medicines were displaced by OCA as no other medicines exist for the specified indications.

6.4 Budget impact

OCA is expected to have a positive budget impact of **sector and a sector and a sect**

- No other comparators existing for the specific indication, which means that no savings can be made by displacing cheaper medicines
- Potential cost saving aspects including decreased number of transplants not being taken into consideration (these are better considered as part of the cost-utility analysis).

	Year 1	Year 2	Year 3	Year 4	Year 5
Uptake, %					
Patients treated with OCA, n					
Medicine acquisition cost per patient per annum	£29,005.78	£29,005.78	£29,005.78	£29,005.78	£29,005.78
Supportive medicines cost per patient per annum*	£620.71	£620.71	£620.71	£620.71	£620.71
Gross additional medicines cost per patient per annum	£29,626.49	£29,626.49	£29,626.49	£29,626.49	£29,626.49
Budget impact					
Cumulative budget impact					

Table 90: Budget impact of OCA

6.5 Additional factors not included in analysis

Yearly incidence rates for PBC were not used due to the incident population being a small proportion of the population. Adverse event costs were not included in the analysis.

6.6 Limitations of the analysis

It is important to note that the budget impact analysis only considers medicine acquisition costs, and without a comparator treatment to OCA there are no savings made due to treatment displacement. Also, the short time horizon of the budget impact analysis fails to capture savings caused by avoiding more severe health states that take a longer time to occur, such as decompensated cirrhosis, HCC and liver transplants. The impact of a

significant reduction in liver transplants will have both financial and medical implications given the high demand for donated livers.

7 References

- 1. Carbone M, Mells GF, Pells G, Dawwas MF, Newton JL, Heneghan MA, et al. Sex and age are determinants of the clinical phenotype of primary biliary cirrhosis and response to ursodeoxycholic acid. Gastroenterology. 2013;144(3):560-9 e7; quiz e13-4.
- NHS. Hospital Episode Statistics, Admitted Patient Care England, 2014-15. Available at <u>http://digital.nhs.uk/searchcatalogue?productid=19420&q=primary+biliary+cirrhos</u> is&sort=Relevance&size=10&page=1#top. Accessed September 2016.
- NHS. Annual report on liver transplantation: report for 2014/2015. 2015.
- Singh J, Longworth L. Estimating the cost of liver transplantation in patients diagnosed with chronic hepatitis C and B in the UK. Value Health. 2014;17(7):A368.
- 5. Adam R, Karam V, Delvart V, O'Grady J, Mirza D, Klempnauer J, et al. Evolution of indications and results of liver transplantation in Europe. A report from the European Liver Transplant Registry (ELTR). J Hepatol. 2012;57(3):675-88.
- 6. Aberg F, Makisalo H, Hockerstedt K, Isoniemi H. Infectious complications more than 1 year after liver transplantation: a 3-decade nationwide experience. Am J Transplant. 2011;11(2):287-95.
- 7. Kaif MH, M.; Singal, A.K.; Wiesner, R. Patients with primary biliary cirrhosis have the highest wait-list mortality. Presented at Digestive Disease Week 2015.
- 8. NIHR Horizon Scanning Centre. Obeticholic acid for primary biliary cirrhosis second line (NIHR HSC ID: 6478). Available at <u>http://www.hsric.nihr.ac.uk/topics/obeticholic-acid-for-primary-biliary-cirrhosissecond-line/</u>. Accessed October 2016.
- 9. Lammers WJvB, H. R.; Hirschfield, G. M.; Pares, A.; Kumagi, T.; Caballeria, L.; Invernizzi, P.; Lleo, A. Alkaline phosphatase values are a surrogate marker in prediction of transplant free survival in patients with primary biliary cirrhosis - an international, colloborative analysis. Journal of hepatology. 2013;58(S388).
- 10. Poupon RE, Lindor KD, Cauch-Dudek K, Dickson ER, Poupon R, Heathcote EJ. Combined analysis of randomized controlled trials of ursodeoxycholic acid in primary biliary cirrhosis. Gastroenterology. 1997;113(3):884-90.
- 11. Corpechot C, Abenavoli L, Rabahi N, Chretien Y, Andreani T, Johanet C, et al. Biochemical response to ursodeoxycholic acid and long-term prognosis in primary biliary cirrhosis. Hepatology. 2008;48(3):871-7.
- 12. Corpechot C, Chazouilleres O, Poupon R. Early primary biliary cirrhosis: Biochemical response to treatment and prediction of long-term outcome. J Hepatol. 2011;55(6):1361-7.
- 13. Kuiper EM, Hansen BE, de Vries RA, den Ouden-Muller JW, van Ditzhuijsen TJ, Haagsma EB, et al. Improved prognosis of patients with primary biliary cirrhosis that have a biochemical response to ursodeoxycholic acid. Gastroenterology. 2009;136(4):1281-7.
- 14. Personal communication. UK-PBC study group.
- 15. Lammers WJ, van Buuren HR, Hirschfield GM, Janssen HL, Invernizzi P, Mason AL, et al. Levels of alkaline phosphatase and bilirubin are surrogate end points of outcomes of patients with primary biliary cirrhosis: an international follow-up study. Gastroenterology. 2014;147(6):1338-49 e5; guiz e15.
- 16. Pares A, Caballeria L, Rodes J. Excellent long-term survival in patients with primary biliary cirrhosis and biochemical response to ursodeoxycholic Acid. Gastroenterology. 2006;130(3):715-20.
- 17. Momah N, Silveira MG, Jorgensen R, Sinakos E, Lindor KD. Optimizing biochemical markers as endpoints for clinical trials in primary biliary cirrhosis. Liver Int. 2012;32(5):790-5.

Company evidence submission template for obeticholic acid [ID785]

- Kumagi T, Guindi M, Fischer SE, Arenovich T, Abdalian R, Coltescu C, et al. Baseline ductopenia and treatment response predict long-term histological progression in primary biliary cirrhosis. Am J Gastroenterol. 2010;105(10):2186-94.
- 19. Kuiper EM, Hansen BE, Adang RP, van Nieuwkerk CM, Timmer R, Drenth JP, et al. Relatively high risk for hepatocellular carcinoma in patients with primary biliary cirrhosis not responding to ursodeoxycholic acid. Eur J Gastroenterol Hepatol. 2010;22(12):1495-502.
- 20. EMA. Public summary of opinion on orphan designation: 6alpha-ethylchenodeoxycholic acid for treatment of primary biliary cirrhosis. EMA/COMP/235947/2010. Available at <u>http://www.ema.europa.eu/docs/en_GB/document_library/Orphan_designation/20</u> <u>10/08/WC500095691.pdf</u>. Accessed September 2016.
- 21. Intercept Pharmaceuticals. Draft Summary of Product Characteristics. September 2016.
- 22. Fenofibrates summary of product characteristics.
- 23. Bezafibrate summary of product chatacteristics.
- 24. Personal communication. George Mells.
- 25. Itakura J, Izumi N, Nishimura Y, Inoue K, Ueda K, Nakanishi H, et al. Prospective randomized crossover trial of combination therapy with bezafibrate and UDCA for primary biliary cirrhosis. Hepatol Res. 2004;29(4):216-22.
- 26. Iwasaki S, Ohira H, Nishiguchi S, Zeniya M, Kaneko S, Onji M, et al. The efficacy of ursodeoxycholic acid and bezafibrate combination therapy for primary biliary cirrhosis: A prospective, multicenter study. Hepatol Res. 2008;38(6):557-64.
- 27. Nakai S, Masaki T, Kurokohchi K, Deguchi A, Nishioka M. Combination therapy of bezafibrate and ursodeoxycholic acid in primary biliary cirrhosis: a preliminary study. Am J Gastroenterol. 2000;95(1):326-7.
- 28. Hosonuma K, Sato K, Yamazaki Y, Yanagisawa M, Hashizume H, Horiguchi N, et al. A prospective randomized controlled study of long-term combination therapy using ursodeoxycholic acid and bezafibrate in patients with primary biliary cirrhosis and dyslipidemia. Am J Gastroenterol. 2015;110(3):423-31.
- 29. Cheung AC, Lapointe-Shaw L, Kowgier M, Meza-Cardona J, Hirschfield GM, Janssen HL, et al. Combined ursodeoxycholic acid (UDCA) and fenofibrate in primary biliary cholangitis patients with incomplete UDCA response may improve outcomes. Aliment Pharmacol Ther. 2016;43(2):283-93.
- 30. Sogolow EDL, J.N. Stigma and liver disease. Illness, Crisis & Loss. 2010;18(3):229-55.
- 31. Nevens F, Andreone P, Mazzella G, Strasser SI, Bowlus C, Invernizzi P, et al. A placebo-controlled trial of obeticholic acid in primary biliary cholangitis. N Engl J Med. 2016;375(7):631-43.
- 32. Lefebvre P, Cariou B, Lien F, Kuipers F, Staels B. Role of bile acids and bile acid receptors in metabolic regulation. Physiol Rev. 2009;89(1):147-91.
- European Association for the Study of the Liver. EASL Clinical Practice Guidelines: Management of cholestatic liver diseases. J Hepatol. 2009;51(2):237-67.
- 34. Lindor KD, Gershwin ME, Poupon R, Kaplan M, Bergasa NV, Heathcote EJ, et al. Primary biliary cirrhosis. Hepatology. 2009;50(1):291-308.
- 35. The British Society of Gastroenterology/UK-PBC. Primary biliary cholangitis (formerly known as primary biliary cirrhosis) treatment and management guidelines. DRAFT September 2016.
- 36. Kuiper EM, van Erpecum KJ, Beuers U, Hansen BE, Thio HB, de Man RA, et al. The potent bile acid sequestrant colesevelam is not effective in cholestatic pruritus: results of a double-blind, randomized, placebo-controlled trial. Hepatology. 2010;52(4):1334-40.

- 37. Bergasa NV, McGee M, Ginsburg IH, Engler D. Gabapentin in patients with the pruritus of cholestasis: a double-blind, randomized, placebo-controlled trial. Hepatology. 2006;44(5):1317-23.
- 38. NHS Blood and Transplant. Organ donation and transplantation: Activity report 2015/16. Available at <u>http://www.odt.nhs.uk/pdf/activity-</u>report/activity_report 2015_16.pdf. Accessed September 2016.
- 39. Raczynska J, Habior A, Paczek L, Foroncewicz B, Pawelas A, Mucha K. Primary biliary cirrhosis in the era of liver transplantation. Ann Transplant. 2014;19:488-93.
- 40. Gao J, Qiao L, Wang B. Primary biliary cirrhosis is a generalized autoimmune epithelitis. Int J Mol Sci. 2015;16(3):6432-46.
- 41. Charatcharoenwitthaya P, Pimentel S, Talwalkar JA, Enders FT, Lindor KD, Krom RA, et al. Long-term survival and impact of ursodeoxycholic acid treatment for recurrent primary biliary cirrhosis after liver transplantation. Liver Transpl. 2007;13(9):1236-45.
- 42. Hirschfield GM, Gershwin ME. The immunobiology and pathophysiology of primary biliary cirrhosis. Annu Rev Pathol. 2013;8:303-30.
- 43. Kumagi T, Heathcote EJ. Primary biliary cirrhosis. Orphanet J Rare Dis. 2008;3:1.
- 44. Selmi C, Bowlus CL, Gershwin ME, Coppel RL. Primary biliary cirrhosis. Lancet. 2011;377(9777):1600-9.
- 45. Poupon R. Primary biliary cirrhosis: a 2010 update. J Hepatol. 2010;52(5):745-58.
- 46. Carey EJ, Ali AH, Lindor KD. Primary biliary cirrhosis. Lancet. 2015;386(10003):1565-75.
- 47. Bowlus CL. Obeticholic acid for the treatment of primary biliary cholangitis in adult patients: Clinical utility and patient selection. Hepat Med. 2016;8:89-95.
- 48. Beuers U, Lindor KD. A major step towards effective treatment evaluation in primary biliary cirrhosis. J Hepatol. 2011;55(6):1178-80.
- 49. Office for National Statistics. Population estimates for UK, England and Wales, Scotland and Northern Ireland: mid-2015. Available at <u>http://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/p</u> <u>opulationestimates/bulletins/annualmidyearpopulationestimates/mid2015</u>. Accessed August 2016.
- 50. Metcalf JV, Bhopal RS, Gray J, Howel D, James OF. Incidence and prevalence of primary biliary cirrhosis in the city of Newcastle upon Tyne, England. Int J Epidemiol. 1997;26(4):830-6.
- 51. Hirschfield GM, Liu X, Xu C, Lu Y, Xie G, Lu Y, et al. Primary biliary cirrhosis associated with HLA, IL12A, and IL12RB2 variants. N Engl J Med. 2009;360(24):2544-55.
- 52. Mells GF, Floyd JA, Morley KI, Cordell HJ, Franklin CS, Shin SY, et al. Genomewide association study identifies 12 new susceptibility loci for primary biliary cirrhosis. Nat Genet. 2011;43(4):329-32.
- 53. Kurtovic J, Riordan SM, Williams R. The natural history of asymptomatic primary biliary cirrhosis. QJM. 2005;98(5):331-6.
- 54. Mells GF, Pells G, Newton JL, Bathgate AJ, Burroughs AK, Heneghan MA, et al. Impact of primary biliary cirrhosis on perceived quality of life: The UK-PBC national study. Hepatology. 2013;58(1):273-83.
- 55. Beuers U, Kremer AE, Bolier R, Elferink RP. Pruritus in cholestasis: facts and fiction. Hepatology. 2014;60(1):399-407.
- 56. Jones DE. Pathogenesis of cholestatic itch: old questions, new answers, and future opportunities. Hepatology. 2012;56(4):1194-6.
- 57. Jones EA, Bergasa NV. The pruritus of cholestasis. Hepatology. 1999;29(4):1003-6.

- 58. Bolier AR, Peri S, Oude Elferink RP, Beuers U. The challenge of cholestatic pruritus. Acta Gastroenterol Belg. 2012;75(4):399-404.
- 59. Newton JL, Hudson M, Tachtatzis P, Sutcliffe K, Pairman J, Burt JA, et al. Population prevalence and symptom associations of autonomic dysfunction in primary biliary cirrhosis. Hepatology. 2007;45(6):1496-505.
- 60. Scalone L, Fagiuoli S, Ciampichini R, Gardini I, Bruno R, Pasulo L, et al. The societal burden of chronic liver diseases: results from the COME study. BMJ Open Gastroenterol. 2015;2(1):e000025.
- 61. Bjornsson E, Verbaan H, Oksanen A, Fryden A, Johansson J, Friberg S, et al. Health-related quality of life in patients with different stages of liver disease induced by hepatitis C. Scand J Gastroenterol. 2009;44(7):878-87.
- 62. Woo G, Tomlinson G, Yim C, Lilly L, Therapondos G, Wong DK, et al. Health state utilities and quality of life in patients with hepatitis B. Can J Gastroenterol. 2012;26(7):445-51.
- 63. Prince MI, James OF, Holland NP, Jones DE. Validation of a fatigue impact score in primary biliary cirrhosis: Towards a standard for clinical and trial use. J Hepatol. 2000;32(3):368-73.
- 64. Younossi ZM, Kiwi ML, Boparai N, Price LL, Guyatt G. Cholestatic liver diseases and health-related quality of life. Am J Gastroenterol. 2000;95(2):497-502.
- 65. Al-Harthy N, Kumagi T, Coltescu C, Hirschfield GM. . The specificity of fatigue in primary biliary cirrhosis: evaluation of a large clinic practice. Hepatology. 2010;52(2):562-70.
- 66. Lasker JN, Sogolow ED, Short LM, Sass DA. The impact of biopsychosocial factors on quality of life: women with primary biliary cirrhosis on waiting list and post liver transplantation. Br J Health Psychol. 2011;16(3):502-27.
- 67. Montagnese S, Nsemi LM, Cazzagon N, Facchini S, Costa L, Bergasa NV, et al. Sleep-Wake profiles in patients with primary biliary cirrhosis. Liver Int. 2013;33(2):203-9.
- 68. Montagnese S. NML, Cazzagon N., Facchini S., Costa L., Schiff S., Amodio P., & Floreani A. . Primary biliary cirrhosis: Time for sleep. . Digestive and Liver Disease. 2010;42(Supplement 4):S316.
- 69. Sogolow ED, Lasker JN, Short LM. Fatigue as a major predictor of quality of life in women with autoimmune liver disease: the case of primary biliary cirrhosis. Womens Health Issues. 2008;18(4):336-42.
- 70. Samsa G, Edelman D, Rothman ML, Williams GR, Lipscomb J, Matchar D. Determining clinically important differences in health status measures: a general approach with illustration to the Health Utilities Index Mark II. Pharmacoeconomics. 1999;15(2):141-55.
- 71. Blackburn P, Freeston M, Baker CR, Jones DE, Newton JL. The role of psychological factors in the fatigue of primary biliary cirrhosis. Liver Int. 2007;27(5):654-61.
- 72. Goldblatt J, Taylor PJ, Lipman T, Prince MI, Baragiotta A, Bassendine MF, et al. The true impact of fatigue in primary biliary cirrhosis: A population study. Gastroenterology. 2002;122(5):1235-41.
- 73. Hong SJ, J. Anxiety and depression in primary biliary cirrhosis [P0968]. Journal of Gastroenterology and Hepatology. 2013;28(Suppl. 3):23-693.
- 74. Huet PM, Deslauriers J, Tran A, Faucher C, Charbonneau J. Impact of fatigue on the quality of life of patients with primary biliary cirrhosis. Am J Gastroenterol. 2000;95(3):760-7.
- 75. Stanca CM, Bach N, Krause C, Tandon N, Freni MA, Gutierrez JA, et al. Evaluation of fatigue in U.S. patients with primary biliary cirrhosis. Am J Gastroenterol. 2005;100(5):1104-9.
- 76. Wunsch E, Post M, Gutkowski K, Marlicz W, Szymanik B, Hartleb M, et al. Critical flicker frequency fails to disclose brain dysfunction in patients with primary biliary cirrhosis. Dig Liver Dis. 2010;42(11):818-21.

Company evidence submission template for obeticholic acid [ID785]

- 77. Selmi C, Gershwin ME, Lindor KD, Worman HJ, Gold EB, Watnik M, et al. Quality of life and everyday activities in patients with primary biliary cirrhosis. Hepatology. 2007;46(6):1836-43.
- 78. Pasha T, Heathcote J, Gabriel S, Cauch-Dudek K, Jorgensen R, Therneau T, et al. Cost-effectiveness of ursodeoxycholic acid therapy in primary biliary cirrhosis. Hepatology. 1999;29(1):21-6.
- 79. Boberg KM, Wisloff T, Kjollesdal KS, Stovring H, Kristiansen IS. Cost and health consequences of treatment of primary biliary cirrhosis with ursodeoxycholic acid. Aliment Pharmacol Ther. 2013;38(7):794-803.
- 81. Mayo MJ. Natural history of primary biliary cirrhosis. Clin Liver Dis. 2008;12(2):277-88; viii.
- 82. Leuschner U, Manns MP, Eisebitt R. Ursodeoxycholic acid in the therapy for primary biliary cirrhosis: effects on progression and prognosis. Z Gastroenterol. 2005;43(9):1051-9.
- 83. Metcalf JV, Mitchison HC, Palmer JM, Jones DE, Bassendine MF, James OF. Natural history of early primary biliary cirrhosis. Lancet. 1996;348(9039):1399-402.
- 84. Balasubramaniam K, Grambsch PM, Wiesner RH, Lindor KD, Dickson ER. Diminished survival in asymptomatic primary biliary cirrhosis. A prospective study. Gastroenterology. 1990;98(6):1567-71.
- 85. Mahl TC, Shockcor W, Boyer JL. Primary biliary cirrhosis: survival of a large cohort of symptomatic and asymptomatic patients followed for 24 years. J Hepatol. 1994;20(6):707-13.
- 86. Springer J, Cauch-Dudek K, O'Rourke K, Wanless IR, Heathcote EJ. Asymptomatic primary biliary cirrhosis: a study of its natural history and prognosis. Am J Gastroenterol. 1999;94(1):47-53.
- 87. Chan CW, Carpenter JR, Rigamonti C, Gunsar F, Burroughs AK. Survival following the development of ascites and/or peripheral oedema in primary biliary cirrhosis: a staged prognostic model. Scand J Gastroenterol. 2005;40(9):1081-9.
- 88. Shapiro JM, Smith H, Schaffner F. Serum bilirubin: a prognostic factor in primary biliary cirrhosis. Gut. 1979;20(2):137-40.
- 89. National Institute for Health and Care Excellence. Cirrhosis in over 16s: assessment and management. 6 July 2016. Available at <u>https://www.nice.org.uk/guidance/ng50/resources/cirrhosis-in-over-16s-</u> assessment-and-management-1837506577093. Accessed August 2016.
- 90. NHS choices. Liver transplant. Available at <u>http://www.nhs.uk/Conditions/Liver-transplant/Pages/Introduction.aspx</u>. Accessed September 2016.
- 91. NHS choices. Liver transplant living with. Available at <u>http://www.nhs.uk/Conditions/Liver-transplant/Pages/Recovery.aspx</u>. Accessed September 2016.
- 92. Falk Hub. Dr Falk Media Update: The silent killer going undiagnosed in thousands of middle aged women. Available at <u>https://drfalkhub.com/media/dr-falk-media-update/</u>. Accessed September 2016.
- Intercept Pharmaceuticals Inc. Clinical study report: 747-301. Obeticholic acid (OCA). A Phase 3, double blind, placebo controlled trial and long term safety extension of obeticholic acid in patients with primary biliary cirrhosis. Data on File. 2015.
- 94. Intercept Pharmaceuticals Inc. Interim clinical study report: 747-301 LTSE. Obeticholic acid (OCA). A Phase 3, double-blind, placebo-controlled trial and

long-term safety extension of obeticholic acid in patients with primary biliary cirrhosis: Long-term safety extension phase. Data of File. 2015.

- 95. Hirschfield GM. Long-term (LT) therapy of a farnesoid X receptor (FXR) Agonist Obeticholic Acid (OCA) maintains biochemical response in primary biliary cirrhosis (PBC). Journal of hepatology. 2012;56(Suppl 2):S372.
- 96. Kowdley K. The first new monotherapy therapeutic PBC study in a decade? An international study evaluating the farnesoid X receptor agonist obeticholic acid in PBC. Hepatology. 2011;54(Suppl 1):416A-7A.
- 97. Kowdley KV, Jones DE, Luketic V, al. e. An international study evaluating farnesoid X receptor agonist obeticholic acid as monotherapy in PBC. Journal of hepatology. 2011;54 (Suppl 1)(S13).
- 98. Kowdley KV. Long-term treatment of primary biliary cirrhosis with the FXR agonist obeticholic acid shows durable efficacy. Journal of hepatology. 2014;60(Suppl 1):S192-S3.
- 99. Kowdley KV. Long-term safety and efficacy of obeticholic acid treatment in primary biliary cirrhosis after more than 4 years of treatment. Hepatology. 2015;62(13).
- 100. Hirschfield GM. A long term safety extension trial of the farnesoid X receptor (FXR) agonist obeticholic acid (OCA) and UDCA in primary biliary cirrhosis (PBC). Hepatology. 2011;54(Suppl. 1):429A.
- Hirschfield GM, Mason A, Luketic V, Lindor K, Gordon SC, Mayo M, et al. Efficacy of obeticholic acid in patients with primary biliary cirrhosis and inadequate response to ursodeoxycholic acid. Gastroenterology. 2015;148(4):751-61 e8.
- 102. Marschall H. The farnesoid X receptor (FXR) agonist obeticholic acid (INT-747, 6ß-ethyl chenodeoxycholic acid) in combination with ursodeoxycholic acid (UDCA) increases plasma FGF-19 concentrations but not bile acid concentration or profile in primary biliary cirrhosis (PBC). Hepatology. 2010;52:355A.
- 103. Mason A. Farnesoid-X receptor agonists: a new class of drugs for the treatment of PBC? An international study evaluationg the addition of INT-747 to ursodeoxycholic acid. Journal of hepatology. 2010;52:S2.
- 104. Andreone P. Efficacy and safety of obeticholic acid in patients with primary biliary cirrhosis: An analysis of the Italian patients from a phase 3, randomized, placebocontrolled study. Digestive and Liver Disease. 2016;48:e81.
- 105. Edwards JE. Exposure-response relationship of obeticholic acid for alkaline phosphatase and total bilirubin in patients with primary biliary cirrhosis (PBC). Gastroenterology. 2016;1:S1072-S3.
- 106. Malecha ES. Baseline factors predicting obeticholic acid induced pruritus in patients with PBC. Gastroenterology. 2016;1:S1058.
- 107. Mayne TJ. Monitoring pruritus for patients with primary biliary cirrhosis (PBC): Equivalence of visual analogue scale and multi-item questionnaires. Hepatology. 2015;62:372A.
- 108. Mayo M. Mitigation of pruritus during obeticholic acid treatment in patients with primary biliary cirrhosis: Strategies and successes. Gastroenterology. 2016;1:S1072.
- 109. Nevens F, et al. The first primary biliary cirrhosis (PBC) phase 3 trial in two decades-an international study of the FXR agonist obeticholic acid in PBC patients. Journal of hepatology. 2014;60:S525-S6.
- 110. Pares A. Fxr agonism with obeticholic acid may attenuate bone mineral density decrease in subjects with primary biliary cirrhosis. Journal of hepatology. 2015;62:22.
- 111. Pencek R. Evaluation of the posology of obeticholic acid (OCA) in patients with PBC. Hepatology. 2015;62:13.
- 112. Peters Y. Long-term safety of OCA in patients with PBC. Hepatology. 2015;62(13).

Company evidence submission template for obeticholic acid [ID785]

- 113. Peters Y. Long-term safety of obeticholic acid in patients with primary biliary cirrhosis. Digestive and Liver Disease. 2016;48:e117-e8.
- 114. Trauner M. Sustained improvement in the markers of cholestasis in an open label long term safety extension study of obeticholic acid in primary biliary cirrhosis patients. Hepatology. 2015;62(13).
- 115. Trauner M. Durable response in the markers of cholestasis through 18 months of open-label long term safety extension study of obeticholic acid in primary biliary cirrhosis. Gastroenterology. 2016;1:S1073-S4.
- 116. Invernizzi P. Integrated efficacy summary for obeticholic acid in subjects with primary biliary cirrhosis. Journal of hepatology. 2015;62:22.
- 117. Jones D. Efficacy and safety of obeticholic acid (OCA) in PBC patients with advanced disease as evidenced by abnormal bilirubin: An integrated analysis. Hepatology. 2015;62:519A-20A.
- 118. Lutz K. Analysis of the efficacy of OCA in primary biliary cirrhosis by varying patient disease severity across three randomized double-blind, placebocontrolled clinical trials. Digestive and Liver Disease. 2016;48:e91.
- 119. Mayne TJ. Obeticholic acid (OCA) has incremental efficacy in patients with higher baseline alkaline phosphatase (ALP) levels. Hepatology. 2015;62:912A-22A.
- 120. Pencek R. Integrated analysis of efficacy of obeticholic acid evaluating subgroups of patients less likely to respond to UDCA (age, age at time of PBC diagnosis, and sex). Hepatology. 2015b;62:530A.
- 121. Pencek R. Integrated analysis of efficacy of phase II and III data of obeticholic acid in PBC subpopulations based on age and sex. Digestive and Liver Disease. 2016;48:e93-4.
- 122. Intercept Pharmaceuticals Inc. Clinical study report: 747-201. Obeticholic acid (OCA). A study of INT-747 (6-ECDCA) monotherapy in patients with primary biliary cirrhosis. Data on File.
- 123. Kremer A, Shapiro D, Rizzo G, al. e. High serum autotaxin activity predisposes to severe pruritus during treatment with obeticholic acid in primary biliary cirrhosis. Journal of hepatology. 2011;54(S1).
- 124. Kowdley K, Nevens F, Andreone P. The first new trials for primary biliary cirrhosis in a decade: An international program evaluating the farnesoid X receptor agonist obeticholic acid. DDW. 2015.
- 125. Angulo P, Lindor KD, Therneau TM, Jorgensen RA, Malinchoc M, Kamath PS, et al. Utilization of the Mayo risk score in patients with primary biliary cirrhosis receiving ursodeoxycholic acid. Liver. 1999;19(2):115-21.
- 126. Meaney C, Kumagi T, Al-Harthy N. Developing end points for clinical trials in PBC: Alkaline phosphatase as a predictor of outcome. Journal of hepatology. 2011;54:S363-534.
- 127. NICE. Final appraisal determination. Liraglutide for the treatment of type 2 diabetes mellitus. September 2010. Available at <u>https://www.nice.org.uk/guidance/ta203/documents/diabetes-liraglutide-final-appraisal-determination3</u>. Accessed September 2016.
- 128. NICE. Final appraisal determination. Pirfenidone for treating idiopathic pulmonary fibrosis. April 2013. Available at <u>https://www.nice.org.uk/guidance/TA282/documents/idiopathic-pulmonary-fibrosis-pirfenidone-final-appraisal-determination-document2</u>. Accessed September 2016.
- Jacoby A, Rannard A, Buck D, Bhala N, Newton JL, James OF, et al. Development, validation, and evaluation of the PBC-40, a disease specific health related quality of life measure for primary biliary cirrhosis. Gut. 2005;54(11):1622-9.
- 130. Mueller S, Sandrin L. Liver stiffness: A novel parameter for the diagnosis of liver disease. Hepat Med. 2010;2:49-67.

- 131. Lubsen J, Kirwan BA. Combined endpoints: Can we use them? Stat Med. 2002;21(19):2959-70.
- 132. Roll J, Boyer JL, Barry D, Klatskin G. The prognostic importance of clinical and histologic features in asymptomatic and symptomatic primary biliary cirrhosis. N Engl J Med. 1983;308(1):1-7.
- Christensen E, Neuberger J, Crowe J, Altman DG, Popper H, Portmann B, et al. Beneficial effect of azathioprine and prediction of prognosis in primary biliary cirrhosis. Final results of an international trial. Gastroenterology. 1985;89(5):1084-91.
- 134. Dickson ER, Grambsch PM, Fleming TR, Fisher LD, Langworthy A. Prognosis in primary biliary cirrhosis: Model for decision making. Hepatology. 1989;10(1):1-7.
- 135. Bonnand AM, Heathcote EJ, Lindor KD, Poupon RE. Clinical significance of serum bilirubin levels under ursodeoxycholic acid therapy in patients with primary biliary cirrhosis. Hepatology. 1999;29(1):39-43.
- 136. Krzeski P, Zych W, Kraszewska E. Is serum bilirubin concentration the only valid prognostic marker in primary biliary cirrhosis? Hepatology. 1999;30(4):865-9.
- 137. Silveira MG, Brunt EM, Heathcote J, Gores GJ, Lindor KD, Mayo MJ. American Association for the Study of Liver Diseases endpoints conference: Design and endpoints for clinical trials in primary biliary cirrhosis. Hepatology. 2010;52(1):349-59.
- 138. Kamath PS, Kim WR, Advanced Liver Disease Study G. The model for end-stage liver disease (MELD). Hepatology. 2007;45(3):797-805.
- 139. Ratcliffe J, Young T, Buxton M, Eldabi T, Paul R, Burroughs A, et al. A simulation modelling approach to evaluating alternative policies for the management of the waiting list for liver transplantation. Health Care Manag Sci. 2001;4(2):117-24.
- 140. Longworth L, Young T, Buxton MJ, Ratcliffe J, Neuberger J, Burroughs A, et al. Midterm cost-effectiveness of the liver transplantation program of England and Wales for three disease groups. Liver Transpl. 2003;9(12):1295-307.
- 141. Harms MHL, W.J.; Marmon, T.; Pencek, R.; van Buuren, H.R.; Hansens, B.E.; Shapiro, D. Improvement in estimated survival after 1 year of obeticholic acid treatment [poster 352]. Journal of hepatology. 2016;64:S631-832.
- 142. NICE. Sofosbuvir for treating chronic hepatitis C. NICE technology appraisal guidance [TA330]. NICE; 2015 [updated 2015; cited 2016 12th September]; Available from: <u>https://www.nice.org.uk/guidance/ta330</u>.
- 143. Al-Harthy N, Kumagi T. Natural history and management of primary biliary cirrhosis. Hepat Med. 2012;4:61-71.
- 144. Pares A. Treatment of primary biliary cirrhosis: Is there more to offer than ursodeoxycholic acid? Clin Liver Dis. 2014;3(2):29-33.
- 145. US Food & Drug Administration. FDA approves Ocaliva for rare, chronic liver disease. 2016 [updated 2016; cited 2016 28/09]; Available from: <u>http://www.fda.gov/NewsEvents/Newsroom/PressAnnouncements/ucm503964.ht</u> m.
- 146. Zhang LN, Shi TY, Shi XH, Wang L, Yang YJ, Liu B, et al. Early biochemical response to ursodeoxycholic acid and long-term prognosis of primary biliary cirrhosis: results of a 14-year cohort study. Hepatology. 2013;58(1):264-72.
- 147. Azemoto N, Abe M, Murata Y, Hiasa Y, Hamada M, Matsuura B, et al. Early biochemical response to ursodeoxycholic acid predicts symptom development in patients with asymptomatic primary biliary cirrhosis. J Gastroenterol. 2009;44(6):630-4.
- 148. Azemoto N, Kumagi T, Abe M, Konishi I, Matsuura B, Hiasa Y, et al. Biochemical response to ursodeoxycholic acid predicts long-term outcome in Japanese patients with primary biliary cirrhosis. Hepatol Res. 2011;41(4):310-7.
- 149. Latimer N. NICE DSU Technical Support Document 14: Survival Analysis for Economic Evaluations Alongside Clinical Trials - Extrapolation with Patient-Level

Data. 2013 [updated 2013; cited 2016 12th September]; Available from: http://www.nicedsu.org.uk/Survival-analysis-TSD(2892878).htm.

- 150. Corpechot C, Carrat F, Bonnand AM, Poupon RE, Poupon R. The effect of ursodeoxycholic acid therapy on liver fibrosis progression in primary biliary cirrhosis. Hepatology. 2000;32(6):1196-9.
- 151. Rossi E, Adams LA, Bulsara M, Jeffrey GP. Assessing liver fibrosis with serum marker models. Clin Biochem Rev. 2007;28(1):3-10.
- 152. Harms MHL, W.J.; Janssen, H.L.; Corpechot, C.; Invernizzi, P.; Mayo, M.J.; Battezzati, P.M.; Floreani, A.; Pares, A.; Nevens, F.; Mason, A.L.; Kowdley, K.; Ponsioen, C.; Bruns, T.; Dalekos, G.; Thorburn, D.; Hirschfield, G.M.; LaRusso, N.F.; Lleo, A.; Cazzagon, N.; Franceschet, I.; Caballeria, L.; Zachou, K.; Poupon, R.; Cheung, A.; Trivedi, P.J.; Carbone, M.; Lindor, K.; van Buuren, H.R.; Hansen, B.E. Incidence and impact of decompensating events in primary biliary cirrhosis results of an international follow up study of 3030 patients. Abstract presented at the 66th Annual Meeting of the American Association for the Study of Liver Diseases, San Francisco, CA, USA. Novermber 13-17 2015. AASLD.
- 153. Trivedi PJ, Lammers WJ, van Buuren HR, Pares A, Floreani A, Janssen HL, et al. Stratification of hepatocellular carcinoma risk in primary biliary cirrhosis: a multicentre international study. Gut. 2016;65(2):321-9.
- 154. Wright M, Grieve R, Roberts J, Main J, Thomas HC, Investigators UKMHCT. Health benefits of antiviral therapy for mild chronic hepatitis C: randomised controlled trial and economic evaluation. Health Technol Assess. 2006;10(21):1-113, iii.
- 155. Kim WR, Lake JR, Smith JM, Skeans MA, Schladt DP, Edwards EB, et al. OPTN/SRTR 2013 Annual Data Report: liver. Am J Transplant. 2015;15 Suppl 2:1-28.
- 156. Neuberger J. Liver transplantation for primary biliary cirrhosis: indications and risk of recurrence. J Hepatol. 2003;39(2):142-8.
- 157. Bondini S, Kallman J, Dan A, Younoszai Z, Ramsey L, Nader F, et al. Healthrelated quality of life in patients with chronic hepatitis B. Liver Int. 2007;27(8):1119-25.
- 158. Intercept Pharmaceuticals Inc. Clinical study report: 747-202. Obeticholic acid (OCA). A study of INT-747 (6-ECDCA) in combination with ursodeoxycholic acid (URSO, UDCA) in subjects with primary biliary cirrhosis. Data on File.
- 159. Ware JE, Jr. SF-36 health survey update. Spine (Phila Pa 1976). 2000;25(24):3130-9.
- 160. Lemos LLPM, A.D.; Gomes, R.M.; Guerra unior, A.A>; Acurcio, F.A. Costs of primary biliary cirrhosis treatment with ursodeoxycholic acid in Brazil. Value in Health. 2013;16(7):A692.
- 161. Kogure T, Ueno Y, Kawagishi N, Kanno N, Yamagiwa Y, Fukushima K, et al. The model for end-stage liver disease score is useful for predicting economic outcomes in adult cases of living donor liver transplantation. J Gastroenterol. 2006;41(10):1005-10.
- 162. Gilroy RK, Lynch SV, Strong RW, Kerlin P, Balderson GA, Stuart KA, et al. Confirmation of the role of the Mayo Risk Score as a predictor of resource utilization after orthotopic liver transplantation for primary biliary cirrhosis. Liver Transpl. 2000;6(6):749-52.
- 163. Kim WR, Wiesner RH, Therneau TM, Poterucha JJ, Porayko MK, Evans RW, et al. Optimal timing of liver transplantation for primary biliary cirrhosis. Hepatology. 1998;28(1):33-8.
- 164. Kim WR, Therneau TM, Dickson ER, Evans RW. Preoperative predictors of resource utilization in liver transplantation. Clin Transpl. 1995:315-22.
- 165. Kankaanpaa J. Cost-effectiveness of liver transplantation. Transplant Proc. 1987;19(5):3864-6.

- 166. Pells GJ, D.; Vale, L. PTU-304 Cost-effectiveness of current therapies in primary biliary cirrhosis: a systematic review of economic evidence. Gut. 2015;64:A195.
- 167. National Health Service. Prescription Cost Analysis (PCA) Data, March 2016. Available at <u>http://www.nhsbsa.nhs.uk/PrescriptionServices/3494.aspx</u>. Accessed September 2016. 2016 [updated 2016; cited 2016 28/09]; Available from: <u>http://www.nhsbsa.nhs.uk/PrescriptionServices/3494.aspx</u>.
- 168. Personal Social Sciences Research Unit. Unit Costs of Health and Social Care 2015. 2016 [updated 2016; cited 2016 12th September]; Available from: http://www.pssru.ac.uk/project-pages/unit-costs/2015/.
- 169. Saab S, Gordon SC, Park H, Sulkowski M, Ahmed A, Younossi Z. Costeffectiveness analysis of sofosbuvir plus peginterferon/ribavirin in the treatment of chronic hepatitis C virus genotype 1 infection. Aliment Pharmacol Ther. 2014;40(6):657-75.
- 170. Shepherd J, Jones J, Hartwell D, Davidson P, Price A, Waugh N. Interferon alpha (pegylated and non-pegylated) and ribavirin for the treatment of mild chronic hepatitis C: a systematic review and economic evaluation. Health Technol Assess. 2007;11(11):1-205, iii.
- 171. Kaltenthaler E. TP, Paisley S., Squires H. NICE DSU Technical Support Document 13: Identifying and Reviewing Evidence to Inform the Conceptualisation and Population of Cost-effectivenes models. 2011 [updated 2011; cited 2016 12th September]; Available from: http://www.nicedsu.org.uk/TSD%2013%20model%20parameters.pdf.
- 172. Drummond MF, Jefferson TO. Guidelines for authors and peer reviewers of economic submissions to the BMJ. The BMJ Economic Evaluation Working Party. BMJ. 1996;313(7052):275-83.

8 Appendices

Appendix 1: SmPC

Appendix 2: Search strategy for relevant studies for efficacy and safety

Appendix 3: Quality assessments of RCTs

Appendix 4: Subgroup analyses (POISE)

Appendix 5: Safety and adverse reactions (POISE)

Appendix 6: Search strategy for cost-effectiveness studies

Appendix 7: Quality assessment of cost-effectiveness studies

Appendix 8: Search strategy for measurement and valuation of health effects

Appendix 9: Cost and healthcare resource identification, measurement and valuation studies

Appendix 10: Calibration of transition probabilities

Appendix 11: Information and questions provided to experts

Appendix 12: POISE transition probabilities for ALP 1.5x ULN, ALP 2.0x ULN

Appendix 13: Checklist of confidential information

Single technology appraisal

Obeticholic acid for primary biliary cirrhosis [ID785]

Erratum to Intercept's responses to ERG questions

21st November 2016

Erratum preface

The company would like to bring to the ERG's attention that there was an error in the original submission regarding patient numbers used in the economic model.

The effects of this change are outlined in detail in this erratum. The erratum contains full deterministic and probabilistic sensitivity analyses of the new base case, including updated versions of the scenario analyses provided in the original submission. The overall effect on the ICERs for both the UDCA-intolerant and UDCA inadequate responder patient populations is minimal, as shown in Table 1 and Table 2. The ICER for the UDCA-intolerant population has decreased by £158.00/QALY, and the ICER for the UDCA inadequate responder population has decreased by £269.00/QALY.

Table 1: Comparison of the incremental cost-effectiveness results included in the original submission versus the updated base case results submitted in this erratum document for the UDCA-intolerant population, using the list price of OCA

Technologies	Total		Incremental			ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
Base case results included in original submission								
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.91		
Updated base case results submitted in erratum								
No treatment (placebo)	£103,233	11.30	6.61	_	-	_	_	_
OCA titration		16.68	13.56		5.38	6.95		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Table 2: Comparison of the incremental cost-effectiveness results included in the original submission versus the updated base case results submitted in this erratum document for the UDCA inadequate responder population, using the list price of OCA

Technologies	Total		Incremental			ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
Base case results included in original submission								
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.79		
Updated base case results submitted in erratum								
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.78	13.68		4.43	5.83		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

The company would like to apologise for this error and the inconvenience caused. We would also like to thank the ERG for spotting the error in patient numbers and for requesting further clarification in their question B12b.

Changes to the cost-utility model

The revised cost-utility model has two changes to the originally submitted model:

- Patient-level data from the POISE CSR have been added for UDCA + no treatment (placebo) patients, and a switch to use these data has been included (as requested by NICE)
- Patient-level data for regimens containing OCA have been updated with the final data from the POISE CSR.

The addition of patient-level data for UDCA + no treatment (placebo) patients warranted minor structural changes to the model to allow the option of using patient-level data or not.

Incorporation of UDCA patient-level data

As described in the previous set of clarification responses (sent to NICE on 18/11/2016), the option to use patient-level UDCA data from POISE has been included in the model.

The majority of changes to the model were made in the 'Transition matrices' worksheet. All of the changes were made in the range L316:HL371 on this worksheet. A switch to enable or disable the use of UDCA patient-level data has been added in cell C10 on the 'Clinical inputs' worksheet. If the switch is activated, patient-level UDCA data from POISE is used to generate transition probabilities for that population.

Patient-level population numbers (added in the cells highlighted in yellow on the 'Transition matrices' worksheet) are included, along with their transition probability calculations (presented alongside). The calculations for how transition probabilities are derived are presented as formulae within the transition probability cells; for example, the patient numbers for UDCA + no treatment (placebo) patients with a recorded ALP threshold of >1.67x ULN (equating to 200 U/L) are located in the range X330:AF338. The transition probability calculations corresponding to this set of patient numbers are located in the range M330:V330. This method was used to derive all patient-level UDCA transition probability data for the first four cycles of the model.

Update of OCA patient-level data

The original submitted model contained data from an analysis of patient numbers according to an early version of the POISE patient-level dataset. Updated patient numbers have since been provided by the Intercept biostatistics team.

The model has now been updated to include the correct patient numbers. These have the effect of changing some transition probabilities used in the first four cycles of the economic model. This has had a downstream effect of changing the QALYs and costs generated for the OCA-treated patient groups.

The specific values changed were patient numbers in the transition matrices for the following treatment regimens:

- OCA titration
- OCA titration + UDCA
- OCA 10 mg (not relevant for this submission)
- OCA 10 mg + UDCA (not relevant for this submission)

Tables of all patient numbers that have been changed, their previous values, and their updated values, are presented below (Table 3, Table 4, Table 5, and Table 6). The cell references provided in the final column of the tables refer to the 'Transition matrices' worksheet of the model.

The different patient numbers have changed the corresponding transition probabilities. For example, on the 'Transition matrices' worksheet of the model, changes to patient numbers in

the range Y121:AF128 will have an impact on the transition probabilities calculated in the range N121:U128 (note that discontinuation is also applied to this specific transition matrix).

Transition		Cycle: 0–3 months		
From:	То:	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	15	16	Y121
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	31	32	Y134
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	39	40	Z146
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	13	14	Y170
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	23	25	Y171
ALP: > 1.67 ULN and Bili: Normal	ALP: \leq 1.67 ULN and Bili: Normal	22	24	Y183
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	38	39	Z183
ALP: > 1.5 ULN and Bili: Normal	ALP: \leq 1.5 ULN and Bili: Normal	12	14	Y195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	49	50	Z195
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	15	16	Y219
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	31	32	Y232
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	39	40	Z244
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	13	14	Y268
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	23	25	Y269
ALP: > 1.67 ULN and Bili: Normal	ALP: \leq 1.67 ULN and Bili: Normal	22	24	Y281
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	38	39	Z281
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	12	14	Y293
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	49	50	Z293

Table 3: OCA regimen patient numbers corrected from the original submission for the 0–3 month model cycle

Transition		Cycle: 3–6 months		
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	0	1	CJ121
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	3	4	CI124
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	2	3	CJ133
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: ≤ 2 ULN and Bili: Normal	1	2	CI136
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	32	33	CJ146
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: > 1.5 ULN and Bili: Normal	2	3	CJ148
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	30	35	CI170
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	21	22	CJ171
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	14	17	CI182
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	6	7	CJ182
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	33	35	CJ183
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	11	12	CI194
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	1	2	CJ194
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	2	4	CI195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	45	47	CJ195
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	0	1	CJ219
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	3	4	CI222
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	2	3	CJ231
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	1	2	CI234
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	32	33	CJ244
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: > 1.5 ULN and Bili: Normal	2	3	CJ246

Table 4: OCA regimen patient numbers corrected from the original submission for the 3–6 month model cycle

Transition		Cycle: 3–6 months		
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	30	35	CI268
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	21	22	CJ269
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	14	17	CI280
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	6	7	CJ280
ALP: > 1.67 ULN and Bili: Normal	35.000	33	35	CJ281
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	11	12	CI292
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	1	2	CJ292
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	2	4	CI293
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	45	47	CJ293

Table 5: OCA regimen patient numbers corrected from the original submission for the 6	6–9
month model cycle	

Transition		Cycle: 6–9 months		
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	44	46	ES121
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV121
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	7	8	ET122
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	31	33	ES133
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV133
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	19	20	ET134
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	19	20	ES145
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	9	10	ES146
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	26	27	ET146

Trar	Cycle: 6–9 months			
From	То	Previous value	Updated value	Cell reference
ALP: > 1.5 ULN and Bili: Normal	Bili: Abnormal and rising, or CC: Bili: Abnormal	0	1	EV146
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	34	38	ES170
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	3	5	ES171
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	14	17	ET171
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	16	18	ES182
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	8	12	ES183
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	23	26	ET183
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: \leq 1.5 ULN and Bili: Normal	12	13	ES194
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	3	4	ET194
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	6	7	ES195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	31	37	ET195
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	44	46	ES219
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV219
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	7	8	ET220
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	31	33	ES231
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV231
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	19	20	ET232
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	19	20	ES243
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	9	10	ES244
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	26	27	ET244
ALP: > 1.5 ULN and Bili: Normal	Bili: Abnormal and rising, or CC: Bili: Abnormal	0	1	EV244
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	34	38	ES268

Transition		Cycle: 6–9 months		
From	То	Previous value	Updated value	Cell reference
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	3	5	ES269
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	14	17	ET269
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	16	18	ES280
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	8	12	ES281
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	23	26	ET281
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	12	13	ES292
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	3	4	ET292
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	6	7	ES293
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	31	37	ET293

Table 6: OCA regimen patient numbers corrected from the original submission for th	e 9–12
month model cycle	

Transition		Cycle: 9–12 months		
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	41	44	HC121
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	29	31	HC133
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	3	4	HC134
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	18	20	HC145
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	25	26	HD146
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	38	41	HC170
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	2	3	HC171
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	12	15	HD171
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	25	28	HC182

Transition		Cycle: 9–12 months		
From	То	Previous value	Updated value	Cell reference
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	21	25	HD183
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	17	18	HC194
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	29	34	HD195
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	41	44	HC219
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	29	31	HC231
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	3	4	HC232
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	18	20	HC243
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	25	26	HD244
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	38	41	HC268
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	2	3	HC269
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	12	15	HD269
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	25	28	HC280
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	21	25	HD281
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	17	18	HC292
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	29	34	HD293

Base case results

Base case deterministic results

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.68	13.56		5.38	6.95		

Table 7: Base-case results for the UDCA-intolerant population, using the list price of OCA

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid.

Table 8: Base-case results for the UDCA inadequate responder population, using the list price of OCA

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.78	13.68		4.43	5.83		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid.

Clinical outcomes from the model

Table 9: Summary of model outcomes for the UDCA-intolerant population

Outcome	OCA titration	No treatment (placebo)	Incremental
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients			
Total number of cases of decompensated cirrhosis per 1,000 patients			
Total liver transplants per 1,000 patients			
Total liver-related deaths per 1,000 patients			

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Table 10: Summary of model outcomes for the UDCA inadequate responder population

Outcome	OCA titration + UDCA	UDCA + placebo	Incremental
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients			
Total number of cases of decompensated cirrhosis per 1,000 patients			
Total liver transplants per 1,000 patients			
Total liver-related deaths per 1,000 patients			

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Results by health state

Table 11: Summary of QALY gain by health state for the UDCA-intolerant population

Health state	OCA titration	No treatment (placebo)	Incremental
ALP: ≤200 U/L and Bili: Normal	6.322	0.000	-6.322
ALP: >200 U/L and Bili: Normal	6.515	2.044	-4.470
Bili: Abnormal and rising, or CC	0.383	2.515	2.132
Discontinuation	0.009	0.000	-0.009
Decompensated cirrhosis	0.103	0.651	0.547
НСС	0.010	0.065	0.055
Pre transplant (end stage)	0.033	0.208	0.175
Liver transplant	0.005	0.031	0.026
Post liver transplant	0.133	0.822	0.689
Re-emergence of PBC	0.045	0.270	0.225
Total	13.558	6.606	-6.952

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 12: Summary of QALY	gain by health state fe	or the UDCA inadequate re	esponder
population			

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal	6.322	0.000	-6.322
ALP: >200 U/L and Bili: Normal	6.693	3.867	-2.826
Bili: Abnormal and rising, or CC	0.354	2.222	1.868
Discontinuation	0.009	0.004	-0.006
Decompensated cirrhosis	0.095	0.569	0.474

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
HCC	0.010	0.057	0.048
Pre transplant (end stage)	0.030	0.182	0.151
Liver transplant	0.005	0.027	0.023
Post liver transplant	0.121	0.701	0.580
Re-emergence of PBC	0.040	0.222	0.181
Total	13.680	7.852	-5.828

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Health state	OCA titration	No treatment (placebo)	Incremental
ALP: ≤200 U/L and Bili: Normal	7.527	0.000	7.527
ALP: >200 U/L and Bili: Normal	7.755	2.434	5.322
Bili: Abnormal and rising, or CC	0.696	4.573	-3.877
Discontinuation	0.012	0.000	0.012
Decompensated cirrhosis	0.270	1.702	-1.431
НСС	0.023	0.145	-0.122
Pre transplant (end stage)	0.086	0.543	-0.457
Liver transplant	0.009	0.055	-0.046
Post liver transplant	0.233	1.443	-1.211
Re-emergence of PBC	0.067	0.402	-0.335
Total	16.679	11.297	5.382

Table 13: Summary of life years gained by health state for the UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

population	by nealth state for	the UDCA Inade	quate responder
Health state	OCA titration + UDCA	UDCA + placebo	Incremental

Table 14: Summary of life years gained by health	state for the UDCA inadequate responder
population	

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal	7.527	0.000	7.527
ALP: >200 U/L and Bili: Normal	7.967	4.604	3.364
Bili: Abnormal and rising, or CC	0.644	4.040	-3.396
Discontinuation	0.012	0.005	0.007
Decompensated cirrhosis	0.250	1.489	-1.239
HCC	0.021	0.128	-0.106
Pre transplant (end stage)	0.080	0.476	-0.396
Liver transplant	0.008	0.048	-0.040
Post liver transplant	0.212	1.231	-1.019

Health state	OCA titration + UDCA	UDCA + placebo	Incremental	
Re-emergence of PBC	0.060	0.331	-0.271	
Total	16.781	12.351	4.430	

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Table 15: Summary of	f costs by health s	tate for the UDCA-	intolerant populat	tion

Health state	OCA titration	No treatment (placebo)	Incremental
ALP: ≤200 U/L and Bili: Normal		£0.00	
ALP: >200 U/L and Bili: Normal		£1,207.08	
Bili: Abnormal and rising, or CC		£28,597.68	
Discontinuation		£0.00	
Decompensated cirrhosis		£21,284.77	
НСС		£1,620.43	
Pre transplant (end stage)		£9,890.90	
Liver transplant		£14,310.87	
Post liver transplant		£26,221.30	
Re-emergence of PBC		£99.80	
Total		£103,232.81	

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal		£0.00	
ALP: >200 U/L and Bili: Normal		£5,349.00	
Bili: Abnormal and rising, or CC		£27,926.48	
Discontinuation		£14.48	
Decompensated cirrhosis		£18,619.88	
HCC		£1,422.13	
Pre transplant (end stage)		£8,665.20	
Liver transplant		£12,526.26	
Post liver transplant		£22,371.06	
Re-emergence of PBC		£82.10	
Total		£96,976.58	

Table 16: Summar	v of costs b	v health state	for the LIDCA	inadequate res	nonder nonulation
Table To. Summar	y UI CUSIS D	y meanin state		mauequaleres	ponder population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Probabilistic sensitivity analysis (PSA) results

Table 17: Incremental cost-effectiveness results of PSA for the UDCA-intolerant population, using the list price of OCA

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental	
PSA results									
No treatment (placebo)	£103,500	11.32	6.60	-	-	-	-	-	
OCA titration		16.67	13.55		5.34	6.95			
Base case deterministic results									
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-	
OCA titration		16.68	13.56		5.38	6.95			

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.





The probability that OCA is cost-effective vs placebo at a willingness-to-pay threshold of \pounds 30,000 is 0.0%.

Table 18: Incremental cost-effectiveness results of PSA for the UDCA inac	lequate responder
population, using the list price of OCA	

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)		
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental		
PSA results	PSA results									
UDCA + placebo	£96,865	12.39	7.87	-	-	-	-	-		
OCA titration + UDCA		16.77	13.66		4.38	5.79				
Base case deterministic results										
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-		
OCA titration + UDCA		16.78	13.68		4.43	5.83				

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.







Deterministic sensitivity analysis





Scenario 1: use of original HCV utility values

Tahla	10. Sconario	1 results for the	LIDCA-intolerant	nonulation	using	the list	nrico d	Δ
Iaple	19. Scenario	i results for the		population	ຸບຣາກບ		price c	н

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
No treatment (placebo)	£103,233	11.30	6.91	-	-	-	-	-
OCA titration		16.68	13.61		5.38	6.70		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 20: Scenario	1 results for the l	UDCA inadequate	responder po	oulation, using t	he list
price of OCA		-		_	

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
UDCA + placebo	£96,977	12.35	8.11	-	-	-	-	-
OCA titration + UDCA		16.78	13.72		4.43	5.61		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Scenario 2: use of alternative transition probabilities

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
No treatment (placebo)	£95,697	10.93	6.43	-	-	-	-	-
OCA titration		16.62	13.53		5.69	7.10		

Table 21: Scenario 2 results for the UDCA-intolerant population, using the list price of OCA

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 22: Scenario 1 results for UDCA inadequate responders, using the list price of OCA

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£90,516	12.04	7.70	-	-	-	-	-
OCA titration + UDCA		16.73	13.65		4.69	5.95		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Technology appraisals

Patient access scheme submission template

October 2009

1 Introduction

The 2009 Pharmaceutical Price Regulation Scheme (PPRS) (www.dh.gov.uk/en/Healthcare/Medicinespharmacyandindustry/Pharmaceutic alpriceregulationscheme/2009PPRS) is a non-contractual scheme between the Department of Health and the Association of the British Pharmaceutical Industry. The purpose of the 2009 PPRS is to ensure that safe and costeffective medicines are available on reasonable terms to the NHS in England and Wales. One of the features of the 2009 PPRS is to improve patients' access to medicines at prices that better reflect their value through patient access schemes.

Patient access schemes are arrangements which may be used on an exceptional basis for the acquisition of medicines for the NHS in England and Wales. Patient access schemes propose either a discount or rebate that may be linked to the number, type or response of patients, or a change in the list price of a medicine linked to the collection of new evidence (outcomes). These schemes help to improve the cost effectiveness of a medicine and therefore allow the National Institute for Health and Care Excellence (NICE) to recommend treatments which it would otherwise not have found to be cost effective. More information on the framework for patient access schemes is provided in the 2009 PPRS

(www.dh.gov.uk/en/Healthcare/Medicinespharmacyandindustry/Pharmaceutic alpriceregulationscheme/2009PPRS.

Patient access schemes are proposed by a pharmaceutical company and agreed with the Department of Health, with input from the Patient Access Schemes Liaison Unit (PASLU) within the Centre for Health Technology Evaluation at NICE.
2 Instructions for manufacturers and sponsors

This document is the patient access scheme submission template for technology appraisals. If manufacturers and sponsors want the National Institute for Health and Care Excellence (NICE) to consider a patient access scheme as part of a technology appraisal, they should use this template. NICE can only consider a patient access scheme after formal referral from the Department of Health.

The template contains the information NICE requires to assess the impact of a patient access scheme on the clinical and cost effectiveness of a technology, in the context of a technology appraisal, and explains the way in which background information (evidence) should be presented. If you are unable to follow this format, you must state your reasons clearly. You should insert 'N/A' against sections that you do not consider relevant, and give a reason for this response.

Please refer to the following documents when completing the template:

- 'Guide to the methods of technology appraisal' https://www.nice.org.uk/process/pmg9/chapter/1-foreword
- 'Specification for manufacturer/sponsor submission of evidence' (http://www.nice.org.uk/aboutnice/howwework/devnicetech/singletechnolog yappraisalsubmissiontemplates.jsp) and
- Pharmaceutical Price Regulation Scheme 2009
 (<u>www.dh.gov.uk/en/Healthcare/Medicinespharmacyandindustry/Pharmaceu</u>
 <u>ticalpriceregulationscheme/2009PPRS</u>).

For further details on the technology appraisal process, please see NICE's 'Guide to the single technology appraisal (STA) process' and 'Guide to the multiple technology appraisal (MTA) process'

(<u>https://www.nice.org.uk/process/pmg19/chapter/1-acknowledgements</u>). The 'Specification for manufacturer/sponsor submission of evidence' provides details on disclosure of information and equality issues. Make the submission as brief and informative as possible. Only mark information as confidential when absolutely necessary. Sufficient information must be publicly available for stakeholders to comment on the full content of the technology appraisal, including details of the proposed patient access scheme. Send submissions electronically to NICE in Word or a compatible format, not as a PDF file.

Appendices may be used to include additional information that is considered relevant to the submission. Do not include information in the appendices that has been requested in the template. Appendices should be clearly referenced in the main submission.

When making a patient access scheme submission, include:

 an updated version of the checklist of confidential information, if necessary an economic model with the patient access scheme incorporated, in accordance with the 'Guide to the methods of technology appraisal' (<u>https://www.nice.org.uk/process/pmg9/chapter/1-foreword</u>)

If you are submitting the patient access scheme at the end of the appraisal process, you should update the economic model to reflect the assumptions that the Appraisal Committee considered to be most plausible. No other changes should be made to the model.

3 Details of the patient access scheme

3.1 Please give the name of the technology and the disease area to which the patient access scheme applies.

Obeticholic acid (Ocaliva®) for the treatment of primary biliary cholangitis

3.2 Please outline the rationale for developing the patient access scheme.

The Pharmaceutical Price Regulation Scheme 2014 makes provisions for manufacturers and sponsors to submit proposals for patient access schemes to the Department of Health. These schemes involve innovative pricing agreements designed to improve cost effectiveness and facilitate patient access to specific drugs or other technologies. Intercept would like to take advantage of this flexibility to offer the best possible value to the NHS, without impacting on the international pricing reference point of the UK

3.3 Please describe the type of patient access scheme, as defined by the PPRS.

Simple discount scheme - percentage discount from the UK list price

- 3.4 Please provide specific details of the patient population to which the patient access scheme applies. Does the scheme apply to the whole licensed population or only to a specific subgroup (for example, type of tumour, location of tumour)? If so:
 - How is the subgroup defined?
 - If certain criteria have been used to select patients, why have these have been chosen?
 - How are the criteria measured and why have the measures been chosen?

The patient access scheme will apply to the final NICE approved population

3.5 Please provide details of when the scheme will apply to the population specified in 3.4. Is the scheme dependent on certain

criteria, for example, degree of response, response by a certain time point, number of injections? If so:

- Why have the criteria been chosen?
- How are the criteria measured and why have the measures been chosen.

The scheme is not dependent on any other criteria

3.6 What proportion of the patient population (specified in 3.4) is expected to meet the scheme criteria (specified in 3.5)?

All patients specified in 3.4 are expected to meet the scheme criteria

3.7 Please explain in detail the financial aspects of the scheme. How will any rebates be calculated and paid?

There will be no rebate required as the discount will be applied at the point of purchase

3.8 Please provide details of how the scheme will be administered.
 Please specify whether any additional information will need to be collected, explaining when this will be done and by whom.

The discount will be shown on the original invoice from either the sole distributor or the homecare company to the purchasing organisation (e.g. the prescribing hospital unit)

3.9 Please provide a flow diagram that clearly shows how the scheme will operate. Any funding flows must be clearly demonstrated.

Provider (either hospital or homecare provider) places an order with the direct/sole distributor for either homecare delivery or delivery to the hospital unit							
Direct/sole distributor supplies requested amount of product and applies percentage discount to the original invoice							
Provider pays direct/sole distributor as per original invoice amount, list price less the application of the PAS discount							

3.10 Please provide details of the duration of the scheme.

The scheme will be in place for the life of the NICE technology appraisal guidance if the product is recommended for use

3.11 Are there any equity or equalities issues relating to the scheme, taking into account current legislation and, if applicable, any concerns identified during the course of the appraisal? If so, how have these been addressed?

No issues relating to equity or equalities have been identified

3.12 If available, please list any scheme agreement forms, patient registration forms, pharmacy claim forms/rebate forms, guides for pharmacists and physicians and patient information documents.
 Please include copies in the appendices.

None available at this time

3.13 In the exceptional case that you are submitting an outcome-based scheme, as defined by the PPRS, please also refer to appendix B.

4 Cost effectiveness

4.1 If the population to whom the scheme applies (as described in sections 3.4 and 3.5) has not been presented in the main manufacturer/sponsor submission of evidence for the technology appraisal (for example, the population is different as there has been a change in clinical outcomes or a new continuation rule), please (re-)submit the relevant sections from the 'Specification for manufacturer/sponsor submission of evidence' (particularly sections 5.5, 6.7 and 6.9). You should complete those sections both with and without the patient access scheme. You must also complete the rest of this template.

Not applicable. The population has been presented in the technology appraisal

4.2 If you are submitting the patient access scheme at the end of the technology appraisal process, you should update the economic model to reflect the assumptions that the Appraisal Committee considered to be most plausible. No other changes should be made to the model.

Not applicable. The PAS has not been submitted at the end of the technology appraisal process

4.3 Please provide details of how the patient access scheme has been incorporated into the economic model. If applicable, please also provide details of any changes made to the model to reflect the assumptions that the Appraisal Committee considered most plausible.

In the model the PAS has been implemented as a straight discount off the list price

4.4 Please provide the clinical effectiveness data resulting from the evidence synthesis and used in the economic model which includes the patient access scheme.

Not applicable. The PAS has been implemented as a straight discount off the list price

4.5 Please list any costs associated with the implementation and operation of the patient access scheme (for example, additional pharmacy time for stock management or rebate calculations). A suggested format is presented in table 1. Please give the reference source of these costs. Please refer to section 6.5 of the 'Specification for manufacturer/sponsor submission of evidence'.

As the PAS is implemented as a straight discount off the list price, there are no additional costs associated with the implementation and operation of the PAS.

4.6 Please provide details of any additional treatment-related costs incurred by implementing the patient access scheme. A suggested format is presented in table 2. The costs should be provided for the intervention both with and without the patient access scheme. Please give the reference source of these costs.

Not applicable. The PAS has been implemented as a straight discount off the list price. There are no additional treatment-related costs incurred with or without the PAS.

Summary results

Base-case analysis

- 4.7 Please present in separate tables the cost-effectiveness results as follows.¹
 - the results for the intervention without the patient access scheme
 - the results for the intervention with the patient access scheme.

A suggested format is shown below (table 3).

Variable	UDCA intolerant population		UDCA inadeque popu	uate responder Ilation
	No treatment OCA (placebo) titration		Placebo + UDCA	OCA titration + UDCA
Total costs	£103,233		£96,977	
Difference in total costs	-		-	
LYG	11.30	16.65	12.35	16.75
LYG difference	-	5.35	-	4.40
QALYs	6.61	13.52	7.85	13.64
QALY difference	-	6.91	-	5.79
ICER (cost/QALY)	_		_	

Table 1: Base case cost-effectiveness results without PAS

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

¹ For outcome-based schemes, please see section 5.2.8 in appendix B.

Variable	UDCA intolera	nt population	UDCA inadequate responder population			
	No treatment OCA (placebo) titration		Placebo + UDCA	OCA titration + UDCA		
Total costs	£103,233	£251,443	£96,977	£261,527		
Difference in total costs	—	£148,210	-	£164,551		
LYG	11.30	16.65	12.35	16.75		
LYG difference	-	5.35	-	4.40		
QALYs	6.61	13.52	7.85	13.64		
QALY difference	_	6.91	_	5.79		
ICER (cost/QALY)	_	£21,438	_	£28,425		

Table 2: Base case cost-effectiveness results with PAS

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

4.8 Please present in separate tables the incremental results as follows. ²

- the results for the intervention without the patient access scheme
- the results for the intervention with the patient access scheme.

List the interventions and comparator(s) from least to most expensive. Present the incremental cost-effectiveness ratios (ICERs) in comparison with baseline (usually standard care), and the incremental analysis ranking technologies in terms of dominance and extended dominance. A suggested format is presented in table 4.

² For outcome-based schemes, please see section 5.2.9 in appendix B.

Technologie		Total		Incremental			ICER versus	ICER
S	Costs	LYG	QALY s	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
No treatment (placebo)	£103,23 3	11.30	6.61	-	-	-	-	_
OCA titration		16.65	13.52		5.35	6.91		

Table 3: Base case incremental results for the UDCA intolerant population, without PAS

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 4: Base	case incremental	results for th	e UDCA into	olerant por	pulation.	with PAS
				• · • · • · • • •		

Technologie	Total			Incremental			ICER versus	ICER
S	Costs	LYG	QALY s	Costs	LYG	QALYs	(cost/QALY)	(cost/QALY)
No treatment (placebo)	£103,23 3	11.30	6.61	-	-	-	-	_
OCA titration	£251,44 3	16.65	13.52	£148,210	5.35	6.91	£21,438	£21,438

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 5: Base case incremental results for the UDCA inadequate responder	population,
without PAS	-

Technologie	Total			Incremental			ICER versus	ICER	
S	5	Costs	LYG	QALY s	Costs	LYG	QALYs	(cost/QALY)	(cost/QALY)
Placebo + UDCA	£96,977	12.35	7.85						
OCA titration + UDCA		16.75	13.64		4.40	5.79			

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid

Table 4: Base case increme	ntal results for the UE	DCA inadequate respond	ler population,
with PAS			

Technologie	Total			Incremental			ICER versus	ICER
S	Costs LYG QA		QALY s	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
Placebo + UDCA	£96,977	12.35	7.85	-	-	-	-	_
OCA titration + UDCA	£261,52 7	16.75	13.64	£164,551	4.40	5.79	£28,425	£28,425

Abbreviations: ICER, incremental cost-effectiveness ratio LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Sensitivity analyses

4.9 Please present deterministic sensitivity analysis results as described for the main manufacturer/sponsor submission of

evidence for the technology appraisal. Consider using tornado diagrams.

The generic inputs that are varied in the deterministic sensitivity analysis (DSA) are the discontinuation probabilities, probabilities of having an adverse event, utility weights, transition probabilities, discounting, and the probability of death for the general population.

The model also allows the user to run scenario analyses by assuming that all patients enter the model in each of the four PBC-specific health states at a time.

The results of the DSA are presented below in Figure 1, Figure 2, Figure 3, and Figure 4.





Abbreviations: ALP, alkaline phosphatase; bili, bilirubin; CC, compensated cirrhosis; DM, disease management; HCC, hepatocellular carcinoma; ICER, incremental cost-effectiveness ratio; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.



Abbreviations: ALP, alkaline phosphatase; bili, bilirubin; CC, compensated cirrhosis; DM, disease management; HCC, hepatocellular carcinoma; ICER, incremental cost-effectiveness ratio; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.





Abbreviations: ALP, alkaline phosphatase; bili, bilirubin; CC, compensated cirrhosis; DM, disease management; ICER, incremental cost-effectiveness ratio; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal;.





Abbreviations: ALP, alkaline phosphatase; bili, bilirubin; CC, compensated cirrhosis; DM, disease management; ICER, incremental cost-effectiveness ratio; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

4.10 Please present any probabilistic sensitivity analysis results, and include scatter plots and cost-effectiveness acceptability curves.

Patient characteristics (i.e. weight and baseline health state distribution) were included in the PSA alongside all other generic inputs (transition probabilities, clinical inputs and quality of life). Data including confidence intervals and distributions were sourced from the literature. Where the variance was unknown, the lower and upper confidence intervals were estimated assuming a 20% variability around the mean. The PSA was performed by running 1,000 Monte Carlo simulations.

The results without the implementation of the PAS are presented below in Table 5 and Table 6. The results with the implementation of the PAS are presented in Table 7 and Table 8.

Table 5: Incremental cost-effectiveness results of PSA for the UDCA intolerant population, without PAS

Technologies	Total			Total Incremental			ICER versus	ICER
	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
PSA results								
No treatment (placebo)	£103,439	11.33	6.77	-	-	-	-	-
OCA titration		16.64	13.52		5.31	6.75		
Base case deter	ministic resu	lts						
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Figure 5: Scatter plot for PSA for the UDCA intolerant population, without PAS



Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Figure 6: Cost-effectiveness acceptability curve for OCA titration versus placebo for the UDCA-intolerant population, without PAS



Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid.

Table 6: Increi population, wi	nental cost-effectiveness ı thout PAS	results of PSA for the UDC	A inadequate	responder
Technologies	Total	Incremental	ICER versus	ICER

Technologies	Total			Incremental			ICER versus	ICER	
	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)	
PSA results									
UDCA + placebo	£97,044	12.38	7.88	-	-	-	-	-	
OCA + UDCA titration		16.75	13.65		4.37	5.77			
Base case deter	ministic resu	lts							
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-	
OCA + UDCA titration		16.75	13.64		4.40	5.79			

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life-year; UDCA ursodeoxycholic acid.

Figure 7: Scatter plot for PSA for the UDCA inadequate responder population, without PAS



Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Figure 8: Cost-effectiveness acceptability curve for OCA + UDCA titration versus UDCA + placebo for the UDCA inadequate responder population without PAS



Abbreviations: OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Table 7: Incremental cost effectiveness results of PSA for the UDCA intolerant population, with PAS

Technologies		Total		Inc	rementa	al	ICER versus	ICER
	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
PSA results								
No treatment (placebo)	£103,607	11.32	6.59	-	-	-	_	-
OCA titration	£251,856	16.66	13.54	£148,249	5.34	6.95	£21,339	£21,339
Base case deter	ministic resu	lts						
No treatment (placebo)	£103,233	11.30	6.61	_	-	-	_	-
OCA titration	£251,443	16.65	13.52	£148,210	5.35	6.91	£21,438	£21,438

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.



Figure 9: Scatter plot for PSA for the UDCA intolerant population, with PAS

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Figure 10: Cost-effectiveness acceptability curve for OCA titration versus placebo for the UDCA intolerant population, with PAS

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid.

Table 8: Incremental cost effectiveness results of PSA for the UDCA inadequate responde	r
population, with PAS	

Technologies		Total		Inc	rementa	al	ICER versus	ICER	
	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)	
PSA results									
Placebo + UDCA	£96,840	12.38	7.87	-	-	-	_	_	
OCA titration + UDCA	£261,193	16.74	13.63	£164,353	4.36	5.76	£28,526	£28,526	
Base case deter	ministic resu	lts							
Placebo + UDCA	£96,977	12.35	7.85	-	-	-	_	_	
OCA titration + UDCA	£261,527	16.75	13.64	£164,551	4.40	5.79	£28,425	£28,425	

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.



Figure 11: Scatter plot for PSA for the UDCA inadequate responder population, with PAS

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; UDCA, QALY, quality-adjusted life year; ursodeoxycholic acid.



Figure 12: Cost-effectiveness acceptability curve for OCA + UDCA titration versus UDCA + placebo for the UDCA inadequate responder population, with PAS

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid.

4.11 Please present scenario analysis results as described for the main manufacturer/sponsor submission of evidence for the technology appraisal.

Scenario 2: use of original HCV utility values

A scenario analysis was performed where all utility values that had been decremented from their previously reported values were set to their original values, which were hepatitis B/hepatitis C-specific). The values changed are shown in Table 9. The non-PAS results are presented in Table 10 and Table 11. The results with the implementation of the PAS are presented in Table 12 and Table 13.

State	Base case utility value	Scenario utility value	Justification
Decompensated cirrhosis	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (143)
Pre-transplant: utility at listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (143)
Pre-transplant: 3 months after listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (143)
Pre-transplant: 6 months after listing	0.38	0.45	Previously reported value for decompensated cirrhosis (TA330) (143)
Liver transplant: 3 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (143)
Liver transplant: 6 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (143)
Liver transplant: 12 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (143)
Liver transplant: 24 months post-transplant	0.57	0.67	Previously reported value for decompensated cirrhosis (TA330) (143)

Table 9: Parameters changed for scenario 2

Technologie	Total			Incremental			ICER versus	ICER
S	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
No treatment (placebo)	<u>£</u> 103,233	11.30	6.64	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.88		

Table 10: Scenario 2 results for the UDCA intolerant population, without PAS

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 11: Scenario 2 results for the UDCA inadequate responder popula	ation,	without
PAS		

Technologie		Total		Inc	rement	al	ICER versus	ICER
S	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)
Placebo + UDCA	£96,977	12.35	8.11	-	-	-	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.75		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 12: Scenario 2 results for the UDCA i	intolerant population, with PAS
---	---------------------------------

Technologie		Total		Inc	rement	al	ICER versus	ICER incremental (cost/QALY)	
S	Costs	LYG	QALYs	Costs	LYG	QALYs	cost/QALY)		
No treatment (placebo)	£103,233	11.30	6.91	-	-	-	-	-	
OCA titration	£251,443	16.65	13.57	£148,210	5.35	6.66	£22,250	£22,250	

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 13: Scenario 2 r	results for the UDCA	inadequate responde	population, with PAS

Technologie	Total			Incremental			ICER versus	ICER	
S	Costs	LYG	QALYs	Costs	LYG	QALYs	baseline (cost/QALY)	(cost/QALY)	
Placebo + UDCA	£96,977	12.35	8.11	-	-	-	_	-	
OCA titration + UDCA	£261,527	16.75	13.69	£164,551	4.40	5.57	£29,524	£29,524	

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Scenario 3: use of alternative transition probabilities

This scenario utilises alternative transition probabilities for pre-liver transplant to liver transplant, and pre-liver transplant to death, gathered from an ad-hoc analysis of PBC-specific data from the Organ Procurement and Transplantation Network (OPTN) (data on file). The parameters changed are shown in Table 14. Results without the PAS are shown in Table 15 and Table

16. Results with the implementation of the PAS are shown in Table 17 and Table 18.

Transition probability	Base case TP	Scenario TP	Justification
Pre-LT to LT	0.35	0.44	Based on ad-hoc data analysis of OPTN PBC data
Pre-LT to death	0.09	0.21	Based on ad-hoc data analysis of OPTN PBC data

Table 14: Parameters changed for scenario 3

Abbreviations: LT, liver transplant; OPTN, Organ Procurement and Transplantation Network; PBC, primary biliary cholangitis/cirrhosis; TP, transition probability.

Table 15: 3 results for the UDCA intolerant population, without PAS

Technologie	Total			Incremental			ICER	ICER
5	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline (cost/QALY)	(cost/QALY)
No treatment (Placebo)	<mark>£</mark> 95,697	10.93	6.56	-	-	-	-	-
OCA titration		16.59	13.51		5.66	6.95		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table 16: Scenario 3 results for the UDCA inadequate responder population, without PAS

Technologie	Total			Incremental			ICER	
S	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline (cost/QALY)	(cost/QALY)
UDCA + Placebo	£90,516	12.04	7.82	-	-	-	-	-
OCA + UDCA titration		16.69	13.63		4.66	5.82		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Table	17: Scena	ario 3 resu	Its for the	UDCA	intolerant	populatio	n. with PAS
						pepalale	

Technologie	Total			Incremental			ICER	ICER
S	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline (cost/QALY)	(cost/QALY)
No treatment (placebo)	£95,697	10.93	6.43	-	Ι	-	_	_
OCA titration	£250,197	16.59	13.49	£154,499	5.66	7.06	£21,874	£21,874

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

Technologie	Total			Incremental			ICER	ICER
S	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline (cost/QALY)	(cost/QALY)
Placebo + UDCA	£90,516	12.04	7.70	-	-	-	_	_
OCA titration + UDCA	£260,386	16.69	13.61	£169,870	4.66	5.91	£28,724	£28,724

Table 18: Scenario 3 results for the UDCA inadequate responder population, with PAS

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-year gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life-year; UDCA, ursodeoxycholic acid.

4.12 If any of the criteria on which the patient access scheme depends are clinical variable (for example, choice of response measure, level of response, duration of treatment), sensitivity analyses around the individual criteria should be provided, so that the Appraisal Committee can determine which criteria are the most appropriate to use.

The PAS functions as a direct discount, so this section is not applicable.

Impact of patient access scheme on ICERs

4.13 For financially based schemes, please present the results showing the impact of the patient access scheme on the ICERs for the base-case and any scenario analyses. A suggested format is shown below (see table 5). If you are submitting the patient access scheme at the end of the appraisal process, you must include the scenario with the assumptions that the Appraisal Committee considered to be most plausible.

Table 15. Results 5	nowing the impe		1.3			
	UDCA intoler (OCA titratio	ant population on vs placebo)	UDCA inadequate responder population (UDCA + OCA titration vs UDCA + placebo)			
	Without PAS	With PAS	Without PAS	With PAS		
Scenario 1 (base case)		£21,438		£28,425		
Scenario 2 (alternative utilities)		£22,250		£29,524		
Scenario 3 (alternative TPs)		£21,874		£28,724		

Table 19: Results showing the impact of PAS on ICERs

Abbreviations: ICER, incremental cost-effectiveness ratio; OCA, obeticholic acid; PAS, patient access scheme, TPs, transition probabilities; UDCA, ursodeoxycholic acid.

5 Appendices

5.1 Appendix A: Additional documents

5.1.1 If available, please include copies of patient access scheme agreement forms, patient registration forms, pharmacy claim forms/rebate forms, guides for pharmacists and physicians, patient information documents.

Response

5.2 Appendix B: Details of outcome-based schemes

- 5.2.1 If you are submitting a proven value: price increase scheme, as defined in the PPRS, please provide the following information:
 - the current price of the intervention
 - the proposed higher price of the intervention, which will be supported by the collection of new evidence
 - a suggested date for when NICE should consider the additional evidence.

Response

- 5.2.2 If you are submitting an expected value: rebate scheme, as defined in the PPRS, please provide the following details:
 - the current price of the intervention (the price that will be supported by the collection of new evidence)
 - the planned lower price of the intervention in the event that the additional evidence does not support the current price
 - a suggested date for when NICE should consider the additional evidence.

Response

- 5.2.3 If you are submitting a risk-sharing scheme, as defined in the PPRS, please provide the following details:
 - the current price of the intervention (the price that will be supported by the collection of new evidence)
 - the proposed relationship between future price changes and the evidence to be collected.

Response

- 5.2.4 For outcome-based schemes, as defined in the PPRS, please provide the full details of the new information (evidence) planned to be collected, who will collect it and who will carry the cost associated with this planned data collection. Details of the new information (evidence) may include:
 - design of the new study
 - patient population of the new study
 - outcomes of the new study
 - expected duration of data collection
 - planned statistical analysis, definition of study groups and reporting (including uncertainty)
 - expected results of the new study
 - planned evidence synthesis/pooling of data (if applicable)
 - expected results of the evidence synthesis/pooling of data (if applicable).

Response

5.2.5 If you are submitting a risk-sharing scheme, please specify the period between the time points when the additional evidence will be considered.

Response

5.2.6 Please provide the clinical effectiveness data resulting from the evidence synthesis and used in the economic modelling of the

patient access scheme at the different time points when the additional evidence is to be considered.

Response

5.2.7 Please provide the other data used in the economic modelling of the patient access scheme at the different time points when the additional evidence is to be considered. These data could include cost/resource use, health-related quality of life and utilities.

Response

- 5.2.8 Please present the cost-effectiveness results as follows.
 - For proven value: price increase schemes, please summarise in separate tables:
 - the results based on current evidence and current price
 - the anticipated results based on the expected new evidence and the proposed higher price.
 - For expected value: rebate schemes, please summarise in separate tables:
 - the results based on the expected new evidence and the current price (which will be supported by the additional evidence collection)
 - the results based on the current evidence and the lower price (if the new evidence is not forthcoming).
 - For risk-sharing schemes, please summarise in separate tables:
 - the results based on current evidence and current price
 - the results based on the expected new evidence and the current price (which will be supported by the additional evidence collection)
 - the results based on the current evidence and the lower price (if the new evidence is not forthcoming)
 - the anticipated results based on the expected new evidence and the proposed higher price.

A suggested format is shown in table 3, section 4.7.

5.2.9 Please present in separate tables the incremental results for the different scenarios as described above in section 5.2.8 for the type of outcome-based scheme being submitted.

List the interventions and comparator(s) from least to most expensive. Present the incremental cost-effectiveness ratios (ICERs) in comparison with baseline (usually standard care), and the incremental analysis ranking technologies in terms of dominance and extended dominance. A suggested format is presented in table 4, section 4.8.

Single technology appraisal

Obeticholic acid for primary biliary cirrhosis [ID785]

Erratum to Intercept's responses to ERG questions

21st November 2016

Erratum preface

The company would like to bring to the ERG's attention that there was an error in the original submission regarding patient numbers used in the economic model.

The effects of this change are outlined in detail in this erratum. The erratum contains full deterministic and probabilistic sensitivity analyses of the new base case, including updated versions of the scenario analyses provided in the original submission. The overall effect on the ICERs for both the UDCA-intolerant and UDCA inadequate responder patient populations is minimal, as shown in Table 1 and Table 2. The ICER for the UDCA-intolerant population has decreased by £158.00/QALY, and the ICER for the UDCA inadequate responder population has decreased by £269.00/QALY.

Table 1: Comparison of the incremental cost-effectiveness results included in the original submission versus the updated base case results submitted in this erratum document for the UDCA-intolerant population, using the list price of OCA

Technologies	Total Incremental			ICER (cost/QALY)	ICER (cost/QALY)			
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
Base case results included in original submission								
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.65	13.52		5.35	6.91		
Updated base of	ase results s	ubmitted	in erratum					
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	-
OCA titration		16.68	13.56		5.38	6.95		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Table 2: Comparison of the incremental cost-effectiveness results included in the original submission versus the updated base case results submitted in this erratum document for the UDCA inadequate responder population, using the list price of OCA

Technologies		Total Incremental			ICER (cost/QALY)	ICER (cost/QALY)		
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
Base case resu	Base case results included in original submission							
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.79		
Updated base of	ase results s	ubmitted	in erratum					
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA		16.78	13.68		4.43	5.83		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

The company would like to apologise for this error and the inconvenience caused. We would also like to thank the ERG for spotting the error in patient numbers and for requesting further clarification in their question B12b.

Changes to the cost-utility model

The revised cost-utility model has two changes to the originally submitted model:

- Patient-level data from the POISE CSR have been added for UDCA + no treatment (placebo) patients, and a switch to use these data has been included (as requested by NICE)
- Patient-level data for regimens containing OCA have been updated with the final data from the POISE CSR.

The addition of patient-level data for UDCA + no treatment (placebo) patients warranted minor structural changes to the model to allow the option of using patient-level data or not.

Incorporation of UDCA patient-level data

As described in the previous set of clarification responses (sent to NICE on 18/11/2016), the option to use patient-level UDCA data from POISE has been included in the model.

The majority of changes to the model were made in the 'Transition matrices' worksheet. All of the changes were made in the range L316:HL371 on this worksheet. A switch to enable or disable the use of UDCA patient-level data has been added in cell C10 on the 'Clinical inputs' worksheet. If the switch is activated, patient-level UDCA data from POISE is used to generate transition probabilities for that population.

Patient-level population numbers (added in the cells highlighted in yellow on the 'Transition matrices' worksheet) are included, along with their transition probability calculations (presented alongside). The calculations for how transition probabilities are derived are presented as formulae within the transition probability cells; for example, the patient

numbers for UDCA + no treatment (placebo) patients with a recorded ALP threshold of >1.67x ULN (equating to 200 U/L) are located in the range X330:AF338. The transition probability calculations corresponding to this set of patient numbers are located in the range M330:V330. This method was used to derive all patient-level UDCA transition probability data for the first four cycles of the model.

Update of OCA patient-level data

The original submitted model contained data from an analysis of patient numbers according to an early version of the POISE patient-level dataset. Updated patient numbers have since been provided by the Intercept biostatistics team.

The model has now been updated to include the correct patient numbers. These have the effect of changing some transition probabilities used in the first four cycles of the economic model. This has had a downstream effect of changing the QALYs and costs generated for the OCA-treated patient groups.

The specific values changed were patient numbers in the transition matrices for the following treatment regimens:

- OCA titration
- OCA titration + UDCA
- OCA 10 mg (not relevant for this submission)
- OCA 10 mg + UDCA (not relevant for this submission)

Tables of all patient numbers that have been changed, their previous values, and their updated values, are presented below (Table 3, Table 4, Table 5, and Table 6). The cell references provided in the final column of the tables refer to the 'Transition matrices' worksheet of the model.

The different patient numbers have changed the corresponding transition probabilities. For example, on the 'Transition matrices' worksheet of the model, changes to patient numbers in the range Y121:AF128 will have an impact on the transition probabilities calculated in the range N121:U128 (note that discontinuation is also applied to this specific transition matrix).

Table 3: OCA regimen patient numbers corrected from	n the original submission for the 0–3
month model cycle	

Trar	Transition			Cycle: 0–3 months				
From:	То:	Previous value	Updated value	Cell reference				
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	15	16	Y121				
ALP: > 1.67 ULN and Bili: Normal	ALP: \leq 1.67 ULN and Bili: Normal	31	32	Y134				
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	39	40	Z146				
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	13	14	Y170				
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	23	25	Y171				

ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	22	24	Y183
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	38	39	Z183
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	12	14	Y195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	49	50	Z195
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	15	16	Y219
ALP: > 1.67 ULN and Bili: Normal	ALP: \leq 1.67 ULN and Bili: Normal	31	32	Y232
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	39	40	Z244
ALP: ≤ 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	13	14	Y268
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	23	25	Y269
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	22	24	Y281
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	38	39	Z281
ALP: > 1.5 ULN and Bili: Normal	ALP: \leq 1.5 ULN and Bili: Normal	12	14	Y293
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	49	50	Z293

Table 4: OCA regimen patient numbers corrected from the original submission for the	he 3–6
month model cycle	

Transition		Cycle: 3–6 months			
From	То	Previous value	Updated value	Cell reference	
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	0	1	CJ121	
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	3	4	CI124	
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	2	3	CJ133	
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	1	2	CI136	
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	32	33	CJ146	
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: > 1.5 ULN and Bili: Normal	2	3	CJ148	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	30	35	CI170	

Trar	Cycle: 3–6 months		onths	
From	То	Previous value	Updated value	Cell reference
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	21	22	CJ171
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	14	17	CI182
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	6	7	CJ182
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	33	35	CJ183
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	11	12	CI194
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	1	2	CJ194
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	2	4	CI195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	45	47	CJ195
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	0	1	CJ219
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	3	4	CI222
ALP: ≤ 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	2	3	CJ231
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: \leq 2 ULN and Bili: Normal	1	2	CI234
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	32	33	CJ244
Bili: Abnormal and rising, or CC: Bili: Abnormal	ALP: > 1.5 ULN and Bili: Normal	2	3	CJ246
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	30	35	CI268
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	21	22	CJ269
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	14	17	CI280
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	6	7	CJ280
ALP: > 1.67 ULN and Bili: Normal	35.000	33	35	CJ281
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	11	12	CI292
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	1	2	CJ292
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	2	4	CI293

Transition		Cycle: 3–6 months			
From	То	Previous Updated C value value refer			
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	45	47	CJ293	

Table 5: OCA regimen patient numbers corrected from the original submission for the 6-	-9
month model cycle	

Trai	nsition	Cycle: 6–9 mc		onths	
From	То	Previous value	Updated value	Cell reference	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	44	46	ES121	
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV121	
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	7	8	ET122	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	31	33	ES133	
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV133	
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	19	20	ET134	
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	19	20	ES145	
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	9	10	ES146	
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	26	27	ET146	
ALP: > 1.5 ULN and Bili: Normal	Bili: Abnormal and rising, or CC: Bili: Abnormal	0	1	EV146	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	34	38	ES170	
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	3	5	ES171	
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	14	17	ET171	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	16	18	ES182	
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	8	12	ES183	
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	23	26	ET183	
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	12	13	ES194	

Trar	Transition		Cycle: 6–9 months	
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	3	4	ET194
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	6	7	ES195
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	31	37	ET195
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	44	46	ES219
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV219
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	7	8	ET220
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	31	33	ES231
ALP: ≤ 2 ULN and Bili: Normal	Bili: Abnormal and rising, or CC	0	1	EV231
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	19	20	ET232
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	19	20	ES243
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	9	10	ES244
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	26	27	ET244
ALP: > 1.5 ULN and Bili: Normal	Bili: Abnormal and rising, or CC: Bili: Abnormal	0	1	EV244
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	34	38	ES268
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	3	5	ES269
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	14	17	ET269
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	16	18	ES280
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	8	12	ES281
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	23	26	ET281
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	12	13	ES292
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	3	4	ET292
ALP: > 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	6	7	ES293

Transition		Cycle: 6–9 months			
From	То	Previous Updated Ce value value refer		Cell reference	
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	31	37	ET293	

Table 6: OCA regimen patient numbers corrected from the original submission for the 9-	12
month model cycle	

Trar	Transition Cyc		ycle: 9–12 m	nonths
From	То	Previous value	Updated value	Cell reference
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	41	44	HC121
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	29	31	HC133
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	3	4	HC134
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	18	20	HC145
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	25	26	HD146
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	38	41	HC170
ALP: > 2 ULN and Bili: Normal	ALP: ≤ 2 ULN and Bili: Normal	2	3	HC171
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	12	15	HD171
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	25	28	HC182
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	21	25	HD183
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	17	18	HC194
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	29	34	HD195
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	41	44	HC219
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	29	31	HC231
ALP: > 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	3	4	HC232
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: ≤ 1.5 ULN and Bili: Normal	18	20	HC243
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	25	26	HD244

Transition		Cycle: 9–12 months			
From	То	Previous value	Updated value	Cell reference	
ALP: ≤ 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	38	41	HC268	
ALP: > 2 ULN and Bili: Normal	ALP: \leq 2 ULN and Bili: Normal	2	3	HC269	
ALP: > 2 ULN and Bili: Normal	ALP: > 2 ULN and Bili: Normal	12	15	HD269	
ALP: ≤ 1.67 ULN and Bili: Normal	ALP: ≤ 1.67 ULN and Bili: Normal	25	28	HC280	
ALP: > 1.67 ULN and Bili: Normal	ALP: > 1.67 ULN and Bili: Normal	21	25	HD281	
ALP: ≤ 1.5 ULN and Bili: Normal	ALP: \leq 1.5 ULN and Bili: Normal	17	18	HC292	
ALP: > 1.5 ULN and Bili: Normal	ALP: > 1.5 ULN and Bili: Normal	29	34	HD293	

Appendix 1: Results using the patient access scheme (PAS) price of OCA

Base case deterministic results

Table 7: Base-case results for the UDCA-intolerant population, using the PAS price of OCA

Technologies	Total			Incremental		ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	_
OCA titration	£251,671	16.68	13.56	£148,439	5.38	6.95	£21,351	£21,351

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Table 8: Base-case results for the UDCA inadequate responder population, using the PAS price of OCA

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-
OCA titration + UDCA	£261,791	16.78	13.68	£164,814	4.43	5.83	£28,281	£28,281

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid
Clinical outcomes from the model

Table 9: Summary of model outcomes for the UDCA-intolerant population

Outcome	OCA titration	No treatment (placebo)	Incremental
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients			
Total number of cases of decompensated cirrhosis per 1,000 patients			
Total liver transplants per 1,000 patients			
Total liver-related deaths per 1,000 patients			

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Table 10: Summary of model outcomes for the UDCA inadequate responder population

Outcome	OCA titration + UDCA	UDCA + placebo	Incremental
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients			
Total number of cases of decompensated cirrhosis per 1,000 patients			
Total liver transplants per 1,000 patients			
Total liver-related deaths per 1,000 patients			

Abbreviations: Bili, bilirubin; CC, compensated cirrhosis; OCA, obeticholic acid; UDCA, ursodeoxycholic acid.

Results by health state

Health state	OCA titration	OCA titration No treatment (placebo)	
ALP: ≤200 U/L and Bili: Normal	6.322	0.000	-6.322
ALP: >200 U/L and Bili: Normal	6.515	2.044	-4.470
Bili: Abnormal and rising, or CC	0.383	2.515	2.132
Discontinuation	0.009	0.000	-0.009
Decompensated cirrhosis	0.103	0.651	0.547
HCC	0.010	0.065	0.055
Pre transplant (end stage)	0.033	0.208	0.175
Liver transplant	0.005	0.031	0.026
Post liver transplant	0.133	0.822	0.689
Re-emergence of PBC	0.045	0.270	0.225
Total	13.558	6.606	-6.952

Table 11: Summary of QALY gain by health state for the UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal	6.322	0.000	-6.322
ALP: >200 U/L and Bili: Normal	6.693	3.867	-2.826
Bili: Abnormal and rising, or CC	0.354	2.222	1.868
Discontinuation	0.009	0.004	-0.006
Decompensated cirrhosis	0.095	0.569	0.474
НСС	0.010	0.057	0.048
Pre transplant (end stage)	0.030	0.182	0.151
Liver transplant	0.005	0.027	0.023
Post liver transplant	0.121	0.701	0.580
Re-emergence of PBC	0.040	0.222	0.181
Total	13.680	7.852	-5.828

Table 12: Summary of QALY gain by health state for the UDCA inadequate responder population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Health state	OCA titration	No treatment (placebo)	Incremental
ALP: ≤200 U/L and Bili: Normal	7.527	0.000	7.527
ALP: >200 U/L and Bili: Normal	7.755	2.434	5.322
Bili: Abnormal and rising, or CC	0.696	4.573	-3.877
Discontinuation	0.012	0.000	0.012
Decompensated cirrhosis	0.270	1.702	-1.431
НСС	0.023	0.145	-0.122
Pre transplant (end stage)	0.086	0.543	-0.457
Liver transplant	0.009	0.055	-0.046
Post liver transplant	0.233	1.443	-1.211
Re-emergence of PBC	0.067	0.402	-0.335
Total	16.679	11.297	5.382

Table 13: Summary of life years gained by health state for the UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Table 14: Summary of life years gained	by health state for	the UDCA inade	quate responder
population			

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal	7.527	0.000	7.527
ALP: >200 U/L and Bili: Normal	7.967	4.604	3.364
Bili: Abnormal and rising, or CC	0.644	4.040	-3.396
Discontinuation	0.012	0.005	0.007
Decompensated cirrhosis	0.250	1.489	-1.239
HCC	0.021	0.128	-0.106
Pre transplant (end stage)	0.080	0.476	-0.396
Liver transplant	0.008	0.048	-0.040
Post liver transplant	0.212	1.231	-1.019
Re-emergence of PBC	0.060	0.331	-0.271
Total	16.781	12.351	4.430

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Health state	OCA titration	OCA titration No treatment (placebo)		
ALP: ≤200 U/L and Bili: Normal		£0.00		
ALP: >200 U/L and Bili: Normal		£1,207.08		
Bili: Abnormal and rising, or CC		£28,597.68		
Discontinuation		£0.00		
Decompensated cirrhosis		£21,284.77		
HCC		£1,620.43		
Pre transplant (end stage)		£9,890.90		
Liver transplant		£14,310.87		
Post liver transplant		£26,221.30		
Re-emergence of PBC		£99.80		
Total		£103,232.81		

Table 15: Summary of costs by health state for the UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Table 16: Summary	of costs by	health state for the UDCA	inadequate responder p	opulation
-------------------	-------------	---------------------------	------------------------	-----------

Health state	OCA titration + UDCA	UDCA + placebo	Incremental
ALP: ≤200 U/L and Bili: Normal		£0.00	
ALP: >200 U/L and Bili: Normal		£5,349.00	
Bili: Abnormal and rising, or CC		£27,926.48	
Discontinuation		£14.48	
Decompensated cirrhosis		£18,619.88	
НСС		£1,422.13	
Pre transplant (end stage)		£8,665.20	
Liver transplant		£12,526.26	
Post liver transplant		£22,371.06	
Re-emergence of PBC		£82.10	
Total		£96,976.58	

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

PSA results

Table 17: Incremental cost-effectiveness results of PSA for the UDCA intolerant population, using the PAS price of OCA

Technologies	Total		Incremental		ICER (cost/QALY)	ICER (cost/QALY)		
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
PSA results								
No treatment (placebo)	£103,252	11.31	6.60	-	-	-	-	-
OCA titration	£251,440	16.67	13.56	£148,188	5.36	6.95	£21,309	£21,309
Base case deterministic results								
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	_	-
OCA titration	£251,671	16.68	13.56	£148,439	5.38	6.95	£21,351	£21,351

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.





Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Figure 2: Cost-effectiveness acceptability curve for the UDCA-intolerant population, using the OCA PAS price

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid.

The probability that OCA is cost-effective vs placebo at a willingness-to-pay threshold of £30,000 is 99.8%.

Table 18: Incremental cost-effectiveness results of PSA for the UDCA inadequate responde
population, using the PAS price of OCA

Technologies	Total Incremental		al	ICER (costs / QALY)	ICER (costs / QALY)						
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental			
PSA results											
UDCA + placebo	£96,928	12.37	7.84	-	-	-	-	-			
OCA titration + UDCA	£261,641	16.77	13.65	£164,712	4.40	5.82	£28,321	£28,321			
Base case detern	Base case deterministic results										
UDCA + placebo	£96,977	12.35	7.85	-	-	-	-	-			
OCA titration + UDCA	£261,791	16.78	13.68	£164,814	4.43	5.83	£28,281	£28,281			

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Figure 3: Scatter plot for PSA for the UDCA inadequate responder population, using the PAS price of OCA

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.



Figure 4: Cost-effectiveness acceptability curve for the UDCA inadequate responder population, using the PAS price of OCA

Abbreviations: OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid.

The probability that OCA titration + UDCA is cost-effective vs UDCA + placebo at a willingness-to-pay threshold of £30,000 is 64.0%.

Deterministic sensitivity analysis





Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; DM, disease management; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Figure 6: Tornado diagram for the UDCA inadequate responder population, using the PAS price of OCA



Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; DM, disease management; LT, liver transplant; OCA, obeticholic acid; PAS, patient access scheme; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Scenario 1: use of original HCV utility values

Table 19: Scenario 1 results for the UDCA-intolerant population, using the PAS price of OCA

Technologies		Total Incremental			ICER (cost/QALY)	ICER (cost/QALY)		
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
No treatment (placebo)	£103,233	11.30	6.91	-	-	-	-	-
OCA titration	£251,671	16.68	13.61	£148,439	5.38	6.70	£22,160	£22,160

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 20: Scenario 1 results for the UDCA inadequate responder population, using the PAS price of OCA

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
UDCA + placebo	£96,977	12.35	8.11	-	-	-	-	-
OCA titration + UDCA	£261,791	16.78	13.72	£164,814	4.43	5.61	£29,374	£29,374

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Scenario 2: use of alternative transition probabilities

Technologies	Total		Incremental			ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
No treatment (placebo)	£94,717	10.89	6.39	-	-	-	-	-
OCA titration	£250,303	16.61	13.52	£155,586	5.73	7.13	£21,824	£21,824

Table 21: Scenario 2 results for the UDCA-intolerant population, using the PAS price of OCA

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 22: Scenario 2 results for the UDCA inadequate responder population, using the PAS price of OCA

Technologies	Total		Incremental			ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
UDCA + placebo	£89,666	12.00	7.67	-	-	-	-	_
OCA titration + UDCA	£260,540	16.72	13.65	£170,874	4.72	5.98	£28,596	£28,596

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; PAS, patient access scheme; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

NICE National Institute for Health and Care Excellence Level 1A City Tower Manchester M1 4BT United Kingdom

+44 (0)300 323 0140

Single technology appraisal

Obeticholic acid for primary biliary cirrhosis [ID785]

Dear Intercept,

The Evidence Review Group, Kleijnen Systematic Reviews, and the technical team at NICE have looked at the submission received on 30 September 2016 from Intercept. In general they felt that it is well presented and clear. However, the ERG and the NICE technical team would like further clarification on the clinical and cost effectiveness data (see questions listed at end of letter).

The ERG and the technical team at NICE will be addressing these issues in their reports.

Please provide your written response to the clarification questions by **5pm** on **4 November 2016**. Your response and any supporting documents should be uploaded to NICE Docs/Appraisals <u>https://appraisals.nice.org.uk/request/20103</u>

Two versions of your written response should be submitted; one with academic/commercial-in-confidence information clearly marked and one with this information removed.

Please <u>underline</u> all confidential information, and separately highlight information that is submitted as <u>commercial in confidence</u> in turquoise, and all information submitted as <u>academic in confidence</u> in yellow.

If you present data that are not already referenced in the main body of your submission and that are academic/commercial in confidence, please complete the attached checklist for confidential information.

Please do not embed documents (PDFs or spreadsheets) in your response because this may result in them being lost or unreadable.

If you have any queries on the technical issues raised in this letter, please contact Technical Lead Any procedural questions

should be addressed to Project Manager

Yours sincerely

Eleanor Donegan

Technical Adviser – Appraisals Centre for Health Technology Evaluation



+44 (0)300 323 0140

Encl. checklist for confidential information

Section A: Clarification on effectiveness data

Literature searching (all searches)

- A1. Please confirm the host for the following resources:
 - **Priority question:** Regarding the Embase strategy reported for the 2014/2015 searches: the company submission states "Embase (platform covers Medline and Medline ® In-Process)" (table 2 Appendix 2). Was this a single search conducted simultaneously over the 3 separate databases? Or a single search of Embase (which now contains all records from Medline and Medline in Process)?
 - Regarding the Embase/Medline/Medline in Process 2014 and 2015 searches: please clarify whether "Embase" refers to Embase.com or Embase via Ovid or EBSCO. Regarding the Embase 2016 searches: please confirm that this used the Ovid platform (as stated for both the Medline and Medline in Process searches). Regarding the Cochrane Library 2014 and 2015 searches: please confirm that this used the Wiley interface.
- A2. Were the date ranges used to identify clinical evidence the same as those reported in the cost effectiveness and health-related quality of life (HRQoL) sections?

Literature searching (clinical evidence)

- A3. **Priority question:** Section 4.1 and Appendix 2 of the company submission describe a systematic review of studies of the efficacy and safety of interventions for primary biliary cirrhosis/cholangitis (PBC). Please clarify the following in relation to the selection and identification of studies.
 - The eligibility criteria for the systematic review of trials specified 'All randomised controlled trials investigating an intervention to treat PBC were included' (company submission page 44). A post-hoc decision was made to exclude all randomised controlled trials (RCTs) that did not include at least one obeticholic acid (OCA) treatment arm (monotherapy or combination) (Appendix 2). Please explain the rationale for this decision.
 - Please confirm that studies of mixed populations of PBC and patients with other conditions were excluded even if data were available for patients with PBC separately?



+44 (0)300 323 0140

- A4. **Priority question:** Please provide a list of studies excluded from the review and reasons for exclusion. This list should include as a minimum the 136 articles excluded because they were within the scope of previous Cochrane reviews. It should also include the 36 articles (25 RCTS) excluded based on having no OCA treatment arm. These numbers have been derived from the PRISMA flow chart on page 46 of the company submission.
- A5. Section 4.10 of the company submission states that 'indirect and mixed treatment comparisons were not conducted'. Was this decision made a priori or as a result of a lack of evidence? No mention is made of any searches, however page 44 of the clinical effectiveness states that "all randomised controlled trials investigating an intervention to treat PBC were included" and the strategies reported in Appendix 2 combine terms for PBC with an RCT filter. Please confirm if these searches were also screened to inform section 4.10.
- A6. Section 4.11 of the company submission states that 'There are no non-RCTs relevant to this submission.' However the systematic review specifies RCTs as an inclusion criterion for study design. Please clarify if any searches were conducted to identify non-RCTS (to inform this section or adverse event estimates in the model) and provide full search strategies.
- A7. Section 4.12 of the company submission states that all safety data reported in this section were derived from the POISE study, with supporting safety data obtained from two phase 2 trials. No searches for adverse events were reported. Please confirm whether any searches were conducted and if yes, provide full search strategies.
- A8. Please provide search dates for the grey literature searches reported on page 8 of Appendix 2.

The POISE Trial

- A9. **Priority question:** The POISE trial is based on surrogate outcomes. Tools exist to predict long-term outcomes from these surrogate outcomes. The ERG have identified a paper by Carbone et al (UK-PBC Risk Scores).¹ Please produce risk scores according to this paper for each intervention group at baseline and at 12 months follow-up, using patient level data from the POISE trial. Please calculate risk scores (5 years, 10 years and 15 years) for each patient at baseline and calculate the average for all patients in each treatment arm at baseline, and do the same for 12 months follow-up.
- A10. **Priority question:** The POISE trial was restricted to patients with a relatively early stage of disease. What is the evidence for safety and effectiveness of OCA in more advanced stages of PBC?



+44 (0)300 323 0140

- A11. The company submission lists 7 study centres in England (page 54). Please state the number of patients recruited in England.
- A12. In the long-term extension to POISE patients were permitted to titrate to lower doses. Was this permitted in the double-blind phase of the trial in any of the treatment arms?

The COBALT Trial

A13. The ongoing COBALT RCT is mentioned on page 119 of the company submission. The ERG notes that the study start date is December 2014 and that the estimated study completion date is April 2023. Are there any interim data available from this trial?

Other

A14. Page 30 of the company submission mentions a sub-optimal response to ursodeoxycholic acid (UDCA). Please define sub-optimal response, inadequate response and absence of response to UDCA. How many people with PBC in England have disease with A) sub-optimal B) inadequate and C) no response to UDCA, respectively?

Section B: Clarification on cost-effectiveness data

Literature searching (cost, healthcare resource use and health-related quality of life)

- B1. The PRISMA chart on page 124 of the company submission (figure 23) reports that 11 studies were excluded for study design/outcome and 2 for population after review of full text. Please list individual studies that were excluded at the full text stage and the reasons for exclusion.
- B2. The strategy reported for the 2014 Embase healthcare resource use identification search (Table 40, Appendix 9) appears to be a duplicate copy of the HRQoL Embase strategy (Table 33, Appendix 8) and does not reflect the resource use terminology utilised in the update search run in June 2016 (Tables 43 & 44, Appendix 9). If this has been included in error, please provide details of the correct strategy.
- B3. Pages 123, 153 and 162 state that a grey literature search was conducted to inform the sections on cost effectiveness, HRQoL and healthcare resource use identification. However, no further details or strategies are provided. Please confirm whether these statements refer to the grey literature search reported on page 8 of Appendix 2 for Clinical Effectiveness? If not, please provide full strategies for each search.



+44 (0)300 323 0140

Model structure and population

- B4. **Priority question:** Page 138 of the company submission states that patients with abnormal bilirubin and those with compensated cirrhosis were combined into one health state (see also figure 24).
 - a. Please confirm the definition of abnormal bilirubin and justify this definition.
 - Is abnormal defined as total bilirubin >20 µmol/L and rising? Or, does total bilirubin >20 µmol/L indicate that bilirubin levels are abnormal and also rising?
 - b. Please provide a clinical rationale for combining abnormal bilirubin and compensated cirrhosis into one health state. Is abnormal bilirubin assumed to be equivalent to having compensated cirrhosis?
 - c. Are the transitions from and to this health state reflective of this combination of health states (people with abnormal bilirubin **and** people with compensated cirrhosis).
 - d. Please provide the working definition of compensated cirrhosis, particularly in terms of total bilirubin?
- B5. **Priority question:** The model structure considers thresholds based on ALP and total bilirubin.
 - a. Figure 24 of the company submission states thresholds of 200 u/L for ALP and 20 µmol/L for total bilirubin. Different thresholds have been used in the economic model submitted in Excel (ALP; 1.67 ULN and TB of 1.0 x ULN). Table 49 of the company submission does not include a threshold for total bilirubin. Please clarify what threshold is used in the company's base-case results and provide all Tables, Figures and results using this threshold.
 - b. Please justify why the Metavir fibrosis score is not used (F0-F4) in the model structure.
- B6. Priority question: The liver disease component of the model assumes that patients treated with OCA (monotherapy or combined with UDCA) can move to an improved health state (e.g. to ALP ≤ 200 u/L and TB ≤ 20 µmol/L), while patients not treated with OCA cannot. Please justify this assumption.
- B7. The model seems to imply that demographics (in terms of age, gender and weight) are similar for both UDCA-intolerant patients and UCDA inadequate responders (see model settings). However, the company submission states "patients with earlier age of onset and/or male sex often have more aggressive disease that is refractory to

+44 (0)300 323 0140

existing treatment". Please explain how the model accounts for different demographic profiles for UDCA intolerant patients and UCDA inadequate responders.

- B8. In the first year of the model (liver disease component), the transition probabilities for the OCA treatment arm seem to be based on the OCA arms of the POISE RCT. Patients then remain in the health state they were in at 12 months, to reflect the sustained reduction in ALP and bilirubin demonstrated in the preliminary results from the long-term safety extension phase. Please confirm.
- B9. The model includes a health state for recurrence of PBC (PBC re-emerge). It is unclear what this health state exactly entails (e.g. is it a re-infection of liver disease; or re-emergence of specifically PBC)
 - a. Please provide a precise definition of the PBC re-emerge health state.
 - b. Please clarify whether this health state considers the re-administration of OCA and/or UDCA for the treatment of PBC? Please justify why.

Comparators

- B10. Priority question: Fibrates have been excluded as a comparator on the basis that "Fibrates are not licensed in the UK, nor are they standard of care, and they are contraindicated in PBC. They are rarely used, with only for patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC)" (table 1 of the company submission). The usage of fibrates in the UK was referenced to a personal communication with George Mells. Given that fibrates are a comparator in the scope:
 - a. Please provide details of the personal communication with George Mells (and response)?
 - b. Is there any other corroborating evidence in support of the personal communication? Please comment on the clinical trial evidence for fibrates in PBC (ideally this evidence would be sourced systematically, but it is acknowledged that this might not be feasible in the timeframe).
 - c. Please provide details of the number of patients (by country) who were excluded from the POISE trial because of use of fibrates (a medication which was prohibited in the trial).
 - d. In addition, please provide reasons for exclusion for the 99 screen failures mentioned in the CONSORT patient-flow diagram (company submission, Figure 11).
 - e. Please include fibrates in the economic model and present the results.



+44 (0)300 323 0140

B11. The company have suggested that placebo is a comparator for people who are unable to tolerate UDCA. Please confirm that this refers to 'no additional treatment' in the final scope and describe the management of disease in patients who cannot tolerate UDCA.

Treatment effectiveness and adverse events

- B12. **Priority question:** Table 49 of the company submission appears to describe the transition probabilities for the OCA comparator. The table is unclear:
 - a. Please provide the primary source(s) of the numbers used in the upper part of the table.
 - b. Please explain how probabilities are obtained from the upper part of the table.
 - c. Please provide the methods (calculations) and results.
 - d. Please explain how these numbers are used in the model and how discontinuation is handled in the model.
 - e. Please explain why discontinuation is only included in the first 3 months of the year.
 - f. Please provide results of a sensitivity analysis including discontinuation in the later months (i.e. months 3-6, 6-9, 9-12).
- B13. **Priority question:** For the comparators in the model, transition probabilities are primarily obtained from the literature.
 - a. Please justify why transition probabilities for the PBC-specific component of the model are not based on POISE for all treatment arms.
 - b. Please provide the results of a sensitivity analysis using the clinical data from POISE to inform the transition probabilities for all comparators for the PBC-specific component of the model. Provide details of the methods used.
- B14. **Priority question:** Please explain the following concerning Table 50 of the company submission:
 - a. Please provide the primary source(s) on which the transition probabilities are based.
 - b. Please detail the characteristics of patients included in this/these primary source(s).



+44 (0)300 323 0140

- c. Please provide an overview of the data extracted from this/these primary source(s) to calculate the transition probabilities.
- d. In addition to Appendix 10, please provide a detailed explanation of the methods and sources used for calibration after data extraction (i.e. provide the calculations for the quarterly transition probabilities) and an Excel sheet containing these calculations.
- B15. **Priority question:** Please explain the following concerning Table 52 of the company submission:
 - a. Please provide the primary source(s) on which the transition probabilities are based.
 - b. Please detail the characteristics of patients included in this/these primary source(s).
 - c. Please provide an overview of the data extracted from this/these primary source(s) to calculate the transition probabilities
 - d. Please explain how Table 51 of the company submission should be used to obtain the estimates in Table 52.
- B16. **Priority question:** Please provide an overview of all transition probabilities used for all comparators and all time points (if time-dependent) for the PBC-specific component, incorporating:
 - a. the estimated probability
 - b. standard error
 - c. primary source
 - d. justification for source
 - e. method of calculation if applicable.
- B17. Please provide an overview of adverse event proportions for fatigue, pruritus and nausea (stratified by treatment).

Health related quality of life

B18. **Priority question:** A recent publication (Dyson et al 2016²) suggested that....."The majority of patients with primary biliary cholangitis do not feel their QoL is impaired, although impairment is reported by a sizeable minority".



+44 (0)300 323 0140

- a. Please justify the primary sources used for quality of life estimates.
- b. Please justify the high utility values for PBC, considering the general population utility (age-dependent).
- c. Please clarify why the utilities from a hepatitis B/C population are appropriate (with or without the reduction of **1**, and justify that these health state utility values are based on the most appropriate source(s). Please explain the rationale for reducing the hepatitis values by **1**.
- B19. The impact of adverse events on health-related quality of life is not included in the cost-effectiveness model. Please provide a sensitivity analysis including the influence of adverse events on quality of life.

Resource use and costs

- B20. It is unclear how the technology cost is used to obtain the total annual technology costs (company submission Table 65). Please clarify and justify the calculation of total annual technology costs.
- B21. Costs of transplant and follow up costs are derived from the literature.
 - Please explain the relevance of using transplant-related cost assumptions based on Hepatitis C and B patients (Wright et al. 2006³ and Singh 2014⁴). Explain why NHS reference costs were not used.
 - b. Please provide the results of a sensitivity analysis using reference cost estimates for liver transplant.
- B22. Outpatient appointment costs are based on expert opinion.
 - a. Please justify why outpatient appointment costs are based on expert opinion and not NHS reference costs (Table 66).
 - Please provide a cost estimate for outpatient appointments based on NHS reference costs. Provide the results of a sensitivity analysis using this estimate.
- B23. Please justify why the health state cost for 'Abnormal/Total bilirubin> 20 μmol/L and rising, or CC' is half the 'DCC' health state cost.

Results and validation

B24. Please provide an overview of disaggregated life years gained, as provided for QALYs gained in Tables 73 and 74 of the company submission.



+44 (0)300 323 0140

- B25. Please provide the outcomes of expert opinion meetings (i.e. individual answers on all of the questions provided in Appendix 11).
- B26. No cross and external validations have been performed and no results of the validation steps described in Section 5.10.1 of the company submission are provided.
 - a. Please perform cross and external validations (for instance, by using the following tool: http://www.uk-pbc.com/resources/tools/riskcalculator/).
 - b. Please provide the results of other validation steps as described in Section 5.10.1 (e.g. comparison of model results with POISE data).

References

[1] Carbone M, Sharp SJ, Flack S, Paximadas D, Spiess K, Adgey C, et al. The UK-PBC risk scores: derivation and validation of a scoring system for long-term prediction of end-stage liver disease in primary biliary cholangitis. *Hepatology* 2016;63(3):930-50.

[2] Dyson JK, Wilkinson N, Jopson L, Mells G, Bathgate A, Heneghan MA, et al. The interrelationship of symptom severity and quality of life in 2055 patients with primary biliary cholangitis. *Aliment Pharmacol Ther* 2016:Epub 2016 Sep 19.

[3] Wright M, Grieve R, Roberts J, Main J, Thomas HC. Health benefits of antiviral therapy for mild chronic hepatitis C: randomised controlled trial and economic evaluation. *Health Technol Assess* 2006;10(21):1-113, iii.

[4] Singh J, Longworth L. Estimating the cost of liver transplantation in patients diagnosed with chronic hepatitis C and B in the Uk [Abstr PGI33]. Paper presented at ISPOR 17th Annual European Congress; 8-12 Nov 2014; Amsterdam: The Netherlands *Value Health* 2014;17(7):A368.

Single technology appraisal

Obeticholic acid for primary biliary cirrhosis [ID785]

Intercept's responses to ERG questions

04 November 2016

Section A: Clarification on effectiveness data

Literature searching (all searches)

- A1. Please confirm the host for the following resources:
 - **Priority question:** Regarding the Embase strategy reported for the 2014/2015 searches: the company submission states "Embase (platform covers Medline and Medline ® In-Process)" (table 2 Appendix 2). Was this a single search conducted simultaneously over the 3 separate databases? Or a single search of Embase (which now contains all records from Medline and Medline in Process)?

A single search of Embase, which contains all records from Medline and Medline in-Process, was performed.

Regarding the Embase/Medline/Medline in Process 2014 and 2015 searches: please clarify whether "Embase" refers to Embase.com or Embase via Ovid or EBSCO. Regarding the Embase 2016 searches: please confirm that this used the Ovid platform (as stated for both the Medline and Medline in Process searches). Regarding the Cochrane Library 2014 and 2015 searches: please confirm that this used the Wiley interface.

The Embase searches in 2014 and 2015 used Embase.com, and the Embase 2016 search used the Ovid platform.

The Cochrane library searches in 2014 and 2015 used the Wiley interface.

A2. Were the date ranges used to identify clinical evidence the same as those reported in the cost effectiveness and health-related quality of life (HRQoL) sections?

The date ranges of all searches conducted in 2014 were from database creation as reported in the cost-effectiveness and HRQoL searches (e.g. 1966 for Embase) to 2014. Subsequent update searches were from date of previous search to date of current search, e.g. the 2015 update searched from 2014 to 2015.

Literature searching (clinical evidence)

- A3. **Priority question:** Section 4.1 and Appendix 2 of the company submission describe a systematic review of studies of the efficacy and safety of interventions for primary biliary cirrhosis/cholangitis (PBC). Please clarify the following in relation to the selection and identification of studies.
 - The eligibility criteria for the systematic review of trials specified 'All randomised controlled trials investigating an intervention to treat PBC were included' (company submission page 44). A post-hoc decision was made to exclude all randomised controlled trials (RCTs) that did not include at least one obeticholic acid (OCA) treatment arm (monotherapy or combination) (Appendix 2). Please explain the rationale for this decision.

When the protocol for the systematic literature review was designed, the strategy was to keep the criteria as broad as possible. However, for the current submission, it was clear that there were no other treatments licensed or being used in the UK for patients with an inadequate response to or intolerance to UDCA. Therefore, only RCTs that included at least one OCA arm were relevant for this submission.

• Please confirm that studies of mixed populations of PBC and patients with other conditions were excluded even if data were available for patients with PBC separately?

Studies of mixed populations of PBC and patients with other conditions were excluded, even if data were available for patients with PBC separately.

A4. **Priority question:** Please provide a list of studies excluded from the review and reasons for exclusion. This list should include as a minimum the 136 articles excluded because they were within the scope of previous Cochrane reviews. It should also include the 36 articles (25 RCTS) excluded based on having no OCA treatment arm. These numbers have been derived from the PRISMA flow chart on page 46 of the company submission.

A total of 136 publications were removed at second pass because they fell within the scope of the Cochrane reviews. The relevant publications are listed in Appendix 1 at the end of this document.

The 36 publications excluded based on having no OCA treatment arm are listed in Appendix 2 at the end of this document.

A5. Section 4.10 of the company submission states that 'indirect and mixed treatment comparisons were not conducted'. Was this decision made a priori or as a result of a lack of evidence? No mention is made of any searches, however page 44 of the clinical effectiveness states that "all randomised controlled trials investigating an intervention to treat PBC were included" and the strategies reported in Appendix 2

combine terms for PBC with an RCT filter. Please confirm if these searches were also screened to inform section 4.10.

It was identified that there were no treatments licensed or being used in the UK for patients with an inadequate response or intolerance to UDCA, and therefore there was no relevant comparator to OCA other than UDCA, which has been compared head to head in POISE. Therefore, an indirect or mixed treatment comparison was not necessary. The reason for including any intervention to treat PBC in the systematic literature review, and the subsequent post-hoc decision to only include publications with an OCA treatment arm, is explained in the answer to question A3.

A6. Section 4.11 of the company submission states that 'There are no non-RCTs relevant to this submission.' However the systematic review specifies RCTs as an inclusion criterion for study design. Please clarify if any searches were conducted to identify non-RCTS (to inform this section or adverse event estimates in the model) and provide full search strategies.

No non-RCT searches were conducted, since head-to-head RCT data (more robust than non-RCT data) were available for OCA vs UDCA in POISE.

A7. Section 4.12 of the company submission states that all safety data reported in this section were derived from the POISE study, with supporting safety data obtained from two phase 2 trials. No searches for adverse events were reported. Please confirm whether any searches were conducted and if yes, provide full search strategies.

No separate searches for adverse events were conducted, since adverse events and any safety data for OCA would be included in the evidence identified in the clinical systematic literature review.

A8. Please provide search dates for the grey literature searches reported on page 8 of Appendix 2.

Grey literature searches were conducted on the same dates as the relevant database searches.

The POISE Trial

A9. **Priority question:** The POISE trial is based on surrogate outcomes. Tools exist to predict long-term outcomes from these surrogate outcomes. The ERG have identified a paper by Carbone et al (UK-PBC Risk Scores). Please produce risk scores according to this paper for each intervention group at baseline and at 12 months follow-up, using patient level data from the POISE trial. Please calculate risk scores (5 years, 10 years and 15 years) for each patient at baseline and calculate the average for all patients in each treatment arm at baseline, and do the same for 12 months follow-up.

Another recent publication by Carbone et al investigated the risk of end-stage liver disease (ESLD) in POISE patients (1). The authors aimed to assess the change in the predicted risk of ESLD with the UK-PBC algorithm in patients who initially received placebo \pm UDCA in the double-blind phase of the trial, then switched to OCA \pm UDCA during the open-label extension stage of the trial.

After 1 year of continued standard-of-care treatment, patients on placebo ±UDCA experienced an increase in their predicted risk of ESLD (using the UK-PBC model), due to worsening liver biochemistry (Table 1). After 1 year of OCA treatment, the predicted risk of ESLD was reduced at 5, 10 and 15 years.

	Baseline Placebo ± UDCA (n=73)	Month 12 DB Placebo ± UDCA (n=68)	Month 12 OLE Placebo ± UDCA (n=58)
Median (IQR) 5-year risk, %	1.9 (1.1, 3.5)	2.3 (1.1, 4.4)	1.4 (0.8, 3.3)*
Median (IQR) 10-year risk, %	6.4 (3.8, 11.3)	7.5 (3.6, 14.0)	4.7 (2.7, 10.6)*
Median (IQR) 15-year risk, %	11.5 (6.9, 19.9)	13.5 (6.6, 24.4)	8.5 (5.0, 18.8)*

Table 1: Predicted risk of ESLD in patients with PBC before and after OCA treatment

Abbreviations: DB, double blind; ESLD, end-stage liver disease; IQR, interquartile range; OCA, obeticholic acid; OLE, open label extension; PBC, primary biliary cirrhosis/cholangitis; UDCA, ursodeoxycholic acid. *p<0.05; p-value for within-group comparison using Wilcoxon Signed Rank test comparing Month 12 DB and Month 12 OLE.

Source: Carbone 2016 (1)

A10. **Priority question:** The POISE trial was restricted to patients with a relatively early stage of disease. What is the evidence for safety and effectiveness of OCA in more advanced stages of PBC?

Two analyses have been performed to assess the safety and efficacy of OCA in more advanced stages of PBC.

The safety of OCA was evaluated in a more advanced disease population using several definitions to approximate severe disease, including a clinical composite definition (based on biochemical criteria, non-invasive measures of fibrosis, biopsies, and/or medical history of decompensation) and presence of cirrhosis. Using the clinical composite definition, 72 patients (30, 22, and 20 patients in the placebo, OCA titration, and OCA 10 mg treatment arms, respectively) were identified with evidence of advanced disease.

Albeit a small sample size, based on the entirety of data, there was not an increased risk in patients with advanced disease. All clinical hepatic AEs in POISE occurred in four patients with advanced disease per the clinical composite criteria (one in the placebo arm, two in the OCA titration arm, and one in the OCA 10 mg arm), and are likely related to disease progression as part of the natural history of PBC.

A post-hoc analysis was also performed (2) to assess the efficacy and safety of OCA in the subset of patients in POISE with cirrhosis, as defined by biopsy-proven cirrhosis, transient elastography of \geq 16.9 kPa, or history of cirrhosis. It was found that 13 patients in the placebo arm, 13 patients in the OCA titration arm, and 10 patients in the OCA 10 mg arm had cirrhosis. ALP levels were significantly more reduced over 12 months in the OCA treatment

groups than the placebo groups, and bilirubin increased in the placebo group but remained stable in the OCA treatment groups (). The primary composite endpoint in POISE (ALP <1.67x ULN and bilirubin ≤ULN and ≥15% reduction in ALP from baseline) was met by 8% of subjects in the placebo arm, 54% of subjects in the OCA titration arm, and 40% of subjects in the OCA 10 mg arm after 12 months.

	Placebo ± UDCA (n=13)	OCA titration ± UDCA (n=13)	OCA 10 mg ± UDCA (n=10)
Mean baseline ALP (SD), U/L	322.5 (138.9)	352.0 (173.6)	305.7 (91.6)
LS mean change from baseline at Month 12 in ALP (SE), U/L	-24.3 (50.6)	–157.9 (48.8)**	–176.2 (56.5)**
Mean baseline total bilirubin (SD), µmol/L	13.0 (7.8)	14.2 (7.7)	16.5 (7.7)
LS mean change from baseline at Month 12 in total bilirubin (SE), µmol/L	4.7 (1.8)	0.6 (1.6)*	-0.5 (2.1)*

Table 2: Effect on ALP and total bilirubin on patients with cirrhosis in POISE

Abbreviations: ALP, alkaline phosphatase; LS, least squares; SD, standard deviation; SE, standard error; UDCA, ursodeoxycholic acid.

In terms of safety in patients with cirrhosis, there were no additional concerns compared with the overall patient population in POISE. However, pruritus was experienced by a greater proportion of patients with cirrhosis than the general population in the OCA treatment groups (69% vs 56%, respectively, in the OCA titration group, and 80% vs 68%, respectively, in the OCA 10 mg group).

The long-term clinical benefit and safety of OCA is currently being investigated in an ongoing double-blind, placebo-controlled confirmatory study (747-302; COBALT), which includes patients with more advanced stages of PBC.

A11. The company submission lists 7 study centres in England (page 54). Please state the number of patients recruited in England.

There were 16 patients recruited in England that were randomised into the trial.

A12. In the long-term extension to POISE patients were permitted to titrate to lower doses. Was this permitted in the double-blind phase of the trial in any of the treatment arms?

Patients were not permitted to titrate to lower doses of OCA in the double-blind phase of POISE. All patients initiated the LTSE phase of POISE at 5 mg once daily, and so this assessed the effect of down-titration on the patients in the OCA 10 mg fixed dose arm of the double-blind phase and patients in the titration arm who had up-titrated to 10 mg OCA for months 6–12 in the double-blind phase.

The COBALT Trial

A13. The ongoing COBALT RCT is mentioned on page 119 of the company submission. The ERG notes that the study start date is December 2014 and that the estimated

study completion date is April 2023. Are there any interim data available from this trial?

No, there are no interim data available and no interim analysis is planned since COBALT is an events-driven trial and 121 primary endpoint events need to have occurred before the trial will report and analyses will be performed.

Other

A14. Page 30 of the company submission mentions a sub-optimal response to ursodeoxycholic acid (UDCA). Please define sub-optimal response, inadequate response and absence of response to UDCA. How many people with PBC in England have disease with A) sub-optimal B) inadequate and C) no response to UDCA, respectively?

For the purpose of this submission, a suboptimal response is the same as an inadequate response. A complete lack of response to UDCA has not been documented in the literature, and is likely included in the definition of suboptimal or inadequate response. Therefore, it is not possible to split people into the three separate categories specified in the question. Nearly 40% of patients with PBC exhibit inadequate response to UDCA (3).

Section B: Clarification on cost-effectiveness data

Literature searching (cost, healthcare resource use and health-related quality of life)

B1. The PRISMA chart on page 124 of the company submission (figure 23) reports that 11 studies were excluded for study design/outcome and 2 for population after review of full text. Please list individual studies that were excluded at the full text stage and the reasons for exclusion.

The articles that were excluded for study design/outcome are listed in Appendix 3 and those excluded for population are listed in Appendix 4, both of which are at the end of this document.

B2. The strategy reported for the 2014 Embase healthcare resource use identification search (Table 40, Appendix 9) appears to be a duplicate copy of the HRQoL Embase strategy (Table 33, Appendix 8) and does not reflect the resource use terminology utilised in the update search run in June 2016 (Tables 43 & 44, Appendix 9). If this has been included in error, please provide details of the correct strategy.

This was an error. A corrected version of the appendix is included below.

Appendix 9: Cost and healthcare resource identification, measurement and valuation studies

Databases searched and service provider

- Embase 1966 to 2014
- Medline 1966 to 2014
- Medline (R) In-Process 1966 to 2014
- Cochrane Database of Systematic Reviews 1996 to 2014
- Database of Abstracts of Reviews of Effects 1994 to 2014
- Health Technology Assessment 1989 to 2014
- NHS Economic Evaluation Database 1968 to 2014
- EconLIT with Full Text 1961 to 2014
- Health Economic Evaluations Database 1990 to 2014

Dates of searches

The searches were conducted in September 2014. The update was conducted in February 2016.

Search strategy

Embase	Embase (platform covers Medline and Medline® In-Process)							
Index	Description	Search terms	Hits					
1	Terms for population	'primary biliary cirrhosis'/exp OR 'primary biliary cirrhosis'	10,402					
2	Cost- effectiveness filter	socioeconomics'/exp OR 'cost benefit analysis'/exp OR 'cost effectiveness analysis'/exp OR 'cost of illness'/exp OR 'economic evaluation'/exp OR 'cost utility'/exp OR 'cost control'/exp OR 'economic aspect'/exp OR 'financial management'/exp OR 'health care cost'/exp OR 'health care financing'/exp OR 'health economics'/exp OR 'hospital cost'/exp OR fiscal:ab,ti OR financial:ab,ti OR finance:ab,ti OR funding:ab,ti OR 'cost minimization analysis'/exp OR 'cost estimate':ab:ti OR 'cost variable':ab:ti OR 'unit cost':ab:ti	1,183,117					
3	Resource use / societal cost filter	'resource use' OR 'resource utilisation' OR 'resource utilization' OR 'productivity'/exp OR 'absenteeism'/exp OR presenteeism OR 'work disability' OR 'work capacity'/exp OR 'caregiver'/exp OR 'caregiver	108,882					

Table 3: Embase cost and resource use utilization search strategy

Embase (platform covers Medline and Medline® In-Process)						
		burden'/exp OR 'caregiver support'/exp OR 'indirect cost'				
6	Combine searches	#1 AND (#2 OR #3) NOT ('animal'/exp NOT 'human'/exp) NOT (letter:it OR editorial:it OR note:it)	114			

Table 4: Cochrane Library cost and resource use utilization search strategy

The Cochrane Library (this platform covers Cochrane Database of Systematic Reviews, Database of Abstract of Reviews of Effects, Health Technology Assessment, NHS Economic Evaluation Database)

Index	Description	Search terms	Hits
1	Terms for population	MeSH descriptor: [Liver Cirrhosis, Biliary] explode all trees	236
		"primary biliary cirrhosis"	523
2	Combine searches	#1 OR #2 in Cochrane Reviews (Reviews and Protocols), Other Reviews, Technology Assessments and Economic Evaluations	65

Table 5: EBSCO cost and resource use utilization search strategy

EBSCO h Database	EBSCO host (this platform covers EconLit with Full Text, Health Economic Evaluations Database)							
Index	Description	Search terms	Hits					
1	Terms for population	Primary biliary cirrhosis	13					

Additional searches

Table 6: Cost and healthcare resource identification systematic review update search strategy EMBASE

Embase 1980 to 2016 Week 13: accessed June 13 th 2016								
#	Searches	Results						
1	exp primary biliary cirrhosis/	8269						
2	(primary biliary adj3 (cholangitis or cirrhosis)).mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword]	11214						
3	pbc.mp.	6111						
4	exp socioeconomics/	204548						
5	exp "cost benefit analysis"/	71339						
6	exp "cost effectiveness analysis"/	114497						

Embase 1980 to 2016 Week 13: accessed June 13 th 2016			
#	Searches	Results	
7	exp "cost of illness"/	16409	
8	exp economic evaluation/	242514	
9	exp "cost utility analysis"/	6759	
10	exp "cost control"/	55630	
11	exp economic aspect/	1260488	
12	exp financial management/	342101	
13	exp "health care cost"/	232855	
14	exp health care financing/	12030	
15	exp health economics/	690164	
16	exp "hospital cost"/	29031	
17	fiscal.ti,ab.	7541	
18	financial.ti,ab.	75875	
19	finance.ti,ab.	4859	
20	funding.ti,ab.	43301	
21	exp "cost minimization analysis"/	2810	
22	cost estimate.ti,ab.	275	
23	cost variable.ti,ab.	50	
24	unit cost.ti,ab.	1114	
25	4 or 5 or 6 or 7 or 8 or 9 or 10 or 11 or 12 or 13 or 14 or 15 or 16 or 17 or 18 or 19 or 20 or 21 or 22 or 23 or 24	1310899	
26	1 or 2 or 3	13078	
27	'resource use'.ti,ab.	8401	
28	'resource utili?ation'.ti,ab.	10980	
29	exp productivity/	29789	

Embase 1980 to 2016 Week 13: accessed June 13 th 2016			
#	Searches	Results	
30	exp absenteeism/	14129	
31	presenteeism.ti,ab.	1103	
32	work disability.ti,ab.	1925	
33	exp work capacity/	10237	
34	exp caregiver burden/ or exp caregiver/	53866	
35	exp caregiver support/	1937	
36	indirect cost*.ti,ab.	6527	
37	27 or 28 or 29 or 30 or 31 or 32 or 33 or 34 or 35 or 36	128789	
38	25 or 37	1395559	
39	26 and 38	221	
40	limit 39 to yr="2014 -Current"	41	

Table 7: Cost and healthcare resource identification systematic review update search strategy-MEDLINE

Ovid MEDLINE(R) In-Process & Other Non-Indexed Citations and Ovid MEDLINE(R) 1946 to Present: accessed June 13 th 2016			
#	Searches	Results	
1	(primary biliary adj3 (cholangitis or cirrhosis)).mp. [mp=title, abstract, original title, name of substance word, subject heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier]	7343	
2	pbc.mp.	4127	
3	exp Liver Cirrhosis, Biliary/	7480	
4	exp Cost-Benefit Analysis/	66406	

5	exp "Cost of Illness"/	20694
6	exp "Cost Control"/	30556
7	exp Financial Management/	82454
8	exp Health Care Costs/	52500
9	exp Hospital Costs/	8906
10	fiscal.ti,ab.	6486
11	financial.ti,ab.	58421
12	finance.ti,ab.	4025
13	funding.ti,ab.	36742
14	cost estimate.ti,ab.	189
15	cost variable.ti,ab.	37
16	unit cost.ti,ab.	773
17	4 or 5 or 6 or 7 or 8 or 9 or 10 or 11 or 12 or 13 or 14 or 15 or 16	304288
18	'resource use'.ti,ab.	5795
19	'resource utili?ation'.ti,ab.	6434
20	exp Efficiency/	12239
21	exp Absenteeism/	7964
22	exp Presenteeism/	31

23	exp Work Capacity Evaluation/	5444
24	exp Caregivers/	25418
25	'work disability'.ti,ab.	1572
26	(caregiver adj2 (support or burden)).ti,ab.	2720
27	indirect cost*.ti,ab.	4399
28	18 or 19 or 20 or 21 or 22 or 23 or 24 or 25 or 26 or 27	67320
29	17 or 28	357642
30	1 or 2 or 3	10644
31	30 and 29	58
32	limit 31 to yr="2014 -Current"	1

Table 8: Cost and healthcare resource identification systematic review update search strategy EconLit

Econlit 1886 to February 2016: accessed March 24 th 2016			
#	Searches	Results	
1	(primary biliary adj3 (cholangitis or cirrhosis)).mp. [mp=heading words, abstract, title, country as subject]	1	
2	pbc.mp. [mp=heading words, abstract, title, country as subject]	73	
3	1 or 2	74	
4	limit 3 to yr="2014 -Current"	12	

Table 9: Cost and healthcare resource identification systematic review update search strategy EBM Reviews

EBM Reviews - Cochrane Central Register of Controlled Trials February 2016, EBM Reviews - Cochrane Database of Systematic Reviews 2005 to March 23, 2016, EBM Reviews -Database of Abstracts of Reviews of Effects 1st Quarter 2016, EBM Reviews - Health Technology Assessment 1st Quarter 2016, EBM Reviews - NHS Economic Evaluation Database 1st Quarter 2016: accessed June 13th 2016

#	Searches	Results
1	(primary biliary adj3 (cholangitis or cirrhosis)).mp. [mp=ti, ot, ab, sh, hw, kw, tx, ct]	569
2	pbc.mp.	303
3	exp Liver Cirrhosis, Biliary/	227
4	exp Cost-Benefit Analysis/	17067
5	exp "Cost of Illness"/	1227
6	exp "Cost Control"/	1226
7	exp Financial Management/	285
8	exp Health Care Costs/	6961
9	exp Hospital Costs/	1436
10	fiscal.ti,ab.	68
11	financial.ti,ab.	1948
12	finance.ti,ab.	42
13	funding.ti,ab.	2759

EBM Reviews - Cochrane Central Register of Controlled Trials February 2016, EBM Reviews - Cochrane Database of Systematic Reviews 2005 to March 23, 2016, EBM Reviews -Database of Abstracts of Reviews of Effects 1st Quarter 2016, EBM Reviews - Health Technology Assessment 1st Quarter 2016, EBM Reviews - NHS Economic Evaluation Database 1st Quarter 2016: accessed June 13th 2016

#	Searches	Results
14	cost estimate.ti,ab.	16
15	cost variable.ti,ab.	2
16	unit cost.ti,ab.	86
17	4 or 5 or 6 or 7 or 8 or 9 or 10 or 11 or 12 or 13 or 14 or 15 or 16	25365
18	'resource use'.ti,ab.	3910
19	'resource utili?ation'.ti,ab.	750
20	exp Efficiency/	309
21	exp Absenteeism/	457
22	exp Presenteeism/	0
23	exp Work Capacity Evaluation/	195
24	exp Caregivers/	1326
25	'work disability'.ti,ab.	134
26	(caregiver adj2 (support or burden)).ti,ab.	407
27	indirect cost*.ti,ab.	408
28	18 or 19 or 20 or 21 or 22 or 23 or 24 or 25 or 26 or 27	6815

EBM Reviews - Cochrane Central Register of Controlled Trials February 2016, EBM Reviews - Cochrane Database of Systematic Reviews 2005 to March 23, 2016, EBM Reviews -Database of Abstracts of Reviews of Effects 1st Quarter 2016, EBM Reviews - Health Technology Assessment 1st Quarter 2016, EBM Reviews - NHS Economic Evaluation Database 1st Quarter 2016: accessed June 13th 2016

#	Searches	Results	
29	17 or 28	30648	
30	1 or 2 or 3	668	
31	30 and 29	11	
32	limit 31 to yr="2014 -Current" [Limit not valid in DARE; records were retained]	0	

Data abstraction strategy

Identified studies were independently assessed by two reviewers in order to ascertain they met the pre-defined inclusion/exclusion criteria and any discrepancies were resolved by a third party. Relevant information was extracted into the STA template/into a pre-defined Microsoft Excel spreadsheet by a reviewer. A second reviewer checked the data extraction and any inconsistencies were resolved through discussion.

B3. Pages 123, 153 and 162 state that a grey literature search was conducted to inform the sections on cost effectiveness, HRQoL and healthcare resource use identification. However, no further details or strategies are provided. Please confirm whether these statements refer to the grey literature search reported on page 8 of Appendix 2 for Clinical Effectiveness? If not, please provide full strategies for each search.

The grey literature searches were performed on the following websites for the costeffectiveness, HRQoL, and healthcare resource use systematic literature reviews:

- American Association for the Study of Liver Diseases (AASLD) <u>http://www.aasld.org/Pages/Default.aspx</u>
- European Association for the Study of the Liver (EASL) http://www.easl.eu/
- American College of Gastroenterology (ACG) <u>http://gi.org/</u>
- Digestive Disease Week (DDW) <u>http://www.ddw.org/</u>
- United European Gastroenterology Week (UEGW) http://www.ueg.eu/

- Canadian Digestive Disease Week http://www.cag-acg.org/
- Japan Digestive Disease Week http://www.jddw.jp/english/index.html
- Liver Foundation UK http://www.liverfoundation.org.uk/
- The Foundation for Liver Research http://www.liver-research.org.uk/
- American Liver Foundation http://www.liverfoundation.org/
- Canadian Liver Foundation http://www.liver.ca/
- British Liver Trust <u>http://www.britishlivertrust.org.uk/</u>
- The British Library http://www.bl.uk
- National Institute for Health Research http://www.hta.ac.uk/
- National Institute for Health and Care Excellence (NICE) <u>http://www.nice.org.uk/</u>
- Scottish Medicines Consortium (SMC) <u>http://www.scottishmedicines.org.uk/Home</u>
- National Centre for Pharmacoeconomics (NCPE) Ireland <u>http://www.ncpe.ie/submission-processs/hta-guidelines/</u>
- All Wales Medicines Strategy Group (AWMSG) <u>http://www.wales.nhs.uk/sites3/home.cfm?orgid=371</u>
- Institute for Quality and Efficiency in Health Care (IQWiG) <u>https://www.iqwig.de/en/home.2724.html</u>
- Haute Autorité de Santé (HAS) http://www.has-sante.fr/portail/jcms/j_5/home
- Italian Medicines Agency (AIFA) http://www.agenziafarmaco.gov.it/en
- Agencia de Evaluación de Tecnologías Sanitarias (AETS) http://www.isciii.es/
- Canadian Agency for Drugs and Technologies in Health (CADTH) <u>http://www.cadth.ca/</u>
- Pharmaceutical Benefits Advisory Committee (PBAC) http://www.pbs.gov.au/info/industry/listing/participants/pbac
- Search Engine (Google) http://www.google.co.uk

Search terms, where applicable, were 'primary biliary cirrhosis', as well as 'primary biliary cholangitis' in the update searches, due to the recent name change of the disease.

Model structure and population

- B4. **Priority question:** Page 138 of the company submission states that patients with abnormal bilirubin and those with compensated cirrhosis were combined into one health state (see also figure 24).
 - a. Please confirm the definition of abnormal bilirubin and justify this definition.

In POISE, abnormal bilirubin is considered to be anything >ULN, also commonly defined as 1.2 mg/dL or 20 μ mol/L. Total bilirubin is a well-established and independent predictor of prognosis in PBC, regardless of treatment (4-6), and is incorporated into most scoring systems and prediction models of clinical outcomes for PBC and other liver diseases,
including the Mayo Risk Score for PBC (7), MELD (8), and Child-Turcotte-Pugh (9, 10). Lammers et al conducted a meta-analysis of data from 4,119 patients with PBC (11) that demonstrated that after one year of UDCA treatment, a bilirubin level >ULN was associated with a higher risk of liver transplant or death compared with a bilirubin level <ULN (hazard ratio 3.215; 95% CI: 2.903, 3.562). The stratification of risk using bilirubin <ULN or >ULN to determine prognostic risk is common.

i. Is abnormal defined as total bilirubin >20 µmol/L and rising? Or, does total bilirubin >20 µmol/L indicate that bilirubin levels are abnormal and also rising?

As described above, abnormal bilirubin is defined as total bilirubin >ULN, which corresponds to 20 µmol/L. The interplay with rising bilirubin levels is an added component that increases risk beyond abnormal bilirubin levels in the model.

b. Please provide a clinical rationale for combining abnormal bilirubin and compensated cirrhosis into one health state. Is abnormal bilirubin assumed to be equivalent to having compensated cirrhosis?

In previous chronic liver disease models (12-15), patients were followed up in terms of their fibrotic stage progression, with all patients progressing to compensated cirrhosis before being at risk of developing decompensated cirrhosis, hepatocellular carcinoma (HCC), or being eligible to be added to the liver transplant waiting list.

In PBC, this is not possible and therefore certain assumptions had to be made. Firstly, the increased risk of progression of patients with abnormal bilirubin has been documented in the literature. In a recent poster by Harms et al, it was shown that patients would either undergo liver transplant or die from liver-related causes within 19 months once total bilirubin had reached 1.6x ULN, and the total bilirubin level increased exponentially from that point (16). Secondly, the histological status of patients with PBC is rarely documented in clinical practice, since disease monitoring is based on biochemical values (e.g. ALP and bilirubin) and liver biopsy is only considered in very specific cases (e.g. enrolment in clinical trials) (17-19). However, as shown in previous economic models, it is not possible for patients to progress to more severe chronic liver health sates, especially decompensated cirrhosis, without having developed advanced fibrosis such as compensated cirrhosis.

Given the lack of data on histological progression among patients with PBC and to account for the advancement of fibrosis before patients can progress to HCC or decompensated cirrhosis, patients with abnormal and rising bilirubin were combined with those with compensated cirrhosis to reflect the severity of the disease from either a biochemical or histological level.

c. Are the transitions from and to this health state reflective of this combination of health states (people with abnormal bilirubin **and** people with compensated cirrhosis).

Since patient histological status is rarely documented, it was not possible to inform transition probabilities based on the histological status of the patients before they reached

decompensated cirrhosis or HCC. Therefore, data based only on elevated and rising bilirubin levels were used to estimate the transition probabilities to and from this health state.

d. Please provide the working definition of compensated cirrhosis, particularly in terms of total bilirubin?

Patients with compensated cirrhosis do not have symptoms related to their cirrhosis, but may have asymptomatic oesophageal or gastric varices. Patients with decompensated cirrhosis have symptomatic complications, including those related to hepatic insufficiency (jaundice) and those related to portal hypertension (ascites, variceal haemorrhage, or hepatic encephalopathy).

Patients with cirrhosis can be sub-categorised into four stages, with stages 1 and 2 classified as compensated cirrhosis and stages 3 and 4 as decompensated cirrhosis (Table 10).

Table 10: Stages of cirrhosis

	Compensate	ed cirrhosis	Decompensated cirrhosis		
Stage	1	2	3	4	
Clinical parameters	No varices, no ascites	Varices, no ascites	Ascites ± varices	Bleeding ± ascites	

Source: D'Amico 2006 (20).

The Child-Turcotte-Pugh classification system () utilised two clinical parameters (encephalopathy and ascites) and three laboratory values (bilirubin, albumin and international normalised ratio). Patients are classified as class A, B or C based on their total points. Based on this system, compensated cirrhosis is classified as class A.

Table 11, Child Turgette Dugh	alaccification	for covority	of airrhadia
Table II. Child-Turcolle-Pugr	i classification	ior sevenity	

		Points per parameter	
	1	2	3
Encephalopathy	None	Grade 1–2 (or precipitant induced)	Grade 3–4 (or chronic)
Ascites	None	Mild to moderate (diuretic responsive)	Severe (diuretic refractrory)
Bilirubin, mg/dL	<2	2–3	>3
Albumin, g/dL	>3.5	2.8–3.5	<2.8
INR	<1.7	1.7–2.3	>2.3
Total score		Class (severity of cirrhosis)	
5–6 points		Class A	
7–9 points		Class B	
10–15 points		Class C	

Abbreviations: INR, international normalised ratio.

- B5. **Priority question:** The model structure considers thresholds based on ALP and total bilirubin.
 - a. Figure 24 of the company submission states thresholds of 200 u/L for ALP and 20 µmol/L for total bilirubin. Different thresholds have been used in the economic model submitted in Excel (ALP; 1.67 ULN and TB of 1.0 x ULN). Table 49 of the company submission does not include a threshold for total bilirubin. Please clarify what threshold is used in the company's base-case results and provide all Tables, Figures and results using this threshold.

As POISE is a global trial, there were variations in the definition of ULN for ALP. The eligibility criteria state that patients must have ALP >1.67x ULN for inclusion, and the average ALP for women in the trial was 197.561. As a result, 1.67x ULN is assumed to be 200 U/L. The ULN for total bilirubin is commonly defined as 1.2 mg/dL (20 μ mol/L), and so was assumed to be 20 μ mol/L.

b. Please justify why the Metavir fibrosis score is not used (F0-F4) in the model structure.

PBC is primarily a ductopenic disease, and therefore ALP and bilirubin levels alone will give a clear outline of the risk of a serious event in PBC. Biopsy is not used in clinical practice and staging of fibrosis through biopsy has limited clinical value. The current management of PBC focuses on reducing biochemical markers, initially ALP levels, and in the more severe stages bilirubin, in order to minimise the risk of long-term progression (21).

B6. Priority question: The liver disease component of the model assumes that patients treated with OCA (monotherapy or combined with UDCA) can move to an improved health state (e.g. to ALP ≤ 200 u/L and TB ≤ 20 µmol/L), while patients not treated with OCA cannot. Please justify this assumption.

Patients who are taking UDCA can only progress to more severe PBC health states based on transition probabilities calibrated to reflect the 10-year liver transplant-free survivals estimated based on POISE patient-level data using GLOBE and UK risk scores (1, 22). This assumption was made to account for the fact that in POISE, patients had similar ALP levels over the duration of the 12-month study period but had increasing bilirubin levels, thus worsening their risk of end-stage liver disease, liver transplant, or death.

Patients who were intolerant to UDCA were assumed to not receive any additional treatment for PBC. They progress to more severe health states based on transition probabilities obtained from the literature.

B7. The model seems to imply that demographics (in terms of age, gender and weight) are similar for both UDCA-intolerant patients and UCDA inadequate responders (see model settings). However, the company submission states "patients with earlier age of onset and/or male sex often have more aggressive disease that is refractory to existing treatment". Please explain how the model accounts for different demographic profiles for UDCA intolerant patients and UCDA inadequate responders.

The demographics used in the model are indeed similar in terms of age, gender and weight for both UDCA intolerant patients and UDCA inadequate responders and were based on the POISE overall population. This was due to the fact that only 16 patients (7.4%) did not receive UDCA during the trial, which prevented the observation of any differences in terms of age, gender and weight between the two groups.

While the model considers the same demographic profiles in both populations in the model, it is possible to change it in the 'model settings' worksheet. Given the lack of data published in PBC, especially in patients with earlier age of onset and male patients, it was not feasible to include any correcting factor to show the impact of their more aggressive disease on the transition probabilities included in the model. More research in these patients is needed.

B8. In the first year of the model (liver disease component), the transition probabilities for the OCA treatment arm seem to be based on the OCA arms of the POISE RCT. Patients then remain in the health state they were in at 12 months, to reflect the sustained reduction in ALP and bilirubin demonstrated in the preliminary results from the long-term safety extension phase. Please confirm.

In the first year, patients can progress freely between the three PBC health states as observed in the POISE OCA regimen arms. Afterwards, patients remain in the health state they are in at 12 months to reflect the sustained reduction/stabilisation of ALP and bilirubin demonstrated in the preliminary results from the LTSE phase of POISE. In sensitivity analysis, OCA patients might progress from the 'moderate risk PBC' state to the 'high risk PBC' state with a low probability based on the literature, assuming they would follow a similar decompensation rate as UDCA GLOBE responders (23).

- B9. The model includes a health state for recurrence of PBC (PBC re-emerge). It is unclear what this health state exactly entails (e.g. is it a re-infection of liver disease; or re-emergence of specifically PBC)
 - a. Please provide a precise definition of the PBC re-emerge health state.

A re-emergence of PBC is defined as the re-emergence of PBC (specifically) following a successful liver transplant. PBC is an autoimmune disease, and will not be cured by liver transplantation; 43% of patients who have a liver transplant as treatment for PBC will develop PBC again within 15 years (24).

b. Please clarify whether this health state considers the re-administration of OCA and/or UDCA for the treatment of PBC? Please justify why.

Patients experiencing recurrence of PBC will not start treatment again and can only receive a second liver transplant or die from liver-related cause.

Comparators

B10. **Priority question:** Fibrates have been excluded as a comparator on the basis that "Fibrates are not licensed in the UK, nor are they standard of care, and they are

contraindicated in PBC. They are rarely used, with only **of** patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC)" (table 1 of the company submission). The usage of fibrates in the UK was referenced to a personal communication with George Mells. Given that fibrates are a comparator in the scope:

a. Please provide details of the personal communication with George Mells (and response)?

Dr Mells is a member of the UK PBC Consortium and manages the patient database. Following a request to the Consortium, Dr Mells reported that from a sample of 2,245 patients in the database, are listed as "ever having used fibrates". A subsequent reanalysis of the database identified that patients were currently taking fibrates from a cohort of 2,353 sampled from the database (a level of

b. Is there any other corroborating evidence in support of the personal communication? Please comment on the clinical trial evidence for fibrates in PBC (ideally this evidence would be sourced systematically, but it is acknowledged that this might not be feasible in the timeframe).

A major factor to note regarding the use of fibrates in PBC is that they are actually contraindicated. Bezafibrate has a contraindication for significant hepatic disease (25), and fenofibrate is contraindicated in hepatic insufficiency including biliary cirrhosis (26).

A Cochrane Collaboration review in 2012 (27) identified only six RCTs comparing bezafibrate with UDCA treatment, either in comparison or in combination with UDCA. All trials were conducted in Japan. There was some effect on ALP seen in the trials, but all suffered from poor trial quality, low patient numbers, and had a high risk of bias. The Cochrane collaboration concluded "treatment of primary biliary cirrhosis with bezafibrate can neither be supported nor refuted based on the best current evidence available ensuing from trials in Japanese patients".

In addition, there are significant safety concerns with the use of fibrates in PBC, with one study (28) reporting three deaths in 13 patients in the UDCA + fibrates arm compared with no deaths in 14 patients in the UDCA monotherapy arm. In addition, one patient developed HCC in the fibrates + UDCA arm, compared with none in the UDCA monotherapy arm (28).

c. Please provide details of the number of patients (by country) who were excluded from the POISE trial because of use of fibrates (a medication which was prohibited in the trial).

As the use of fibrates was an exclusion criterion for POISE, patients would not be entered into screening if they were currently being treated with fibrates. A patient will only sign a consent form to allow data capture after they have been considered for screening, so it was not possible under Good Clinical Practice to capture information on patients who did not progress to screening because of the use of fibrates. There were 13 of 316 patients who were excluded due to criteria relating to prohibited medicines, which included azathioprine, colchicine, cyclosporine, methotrexate, mycophenolate mofetil, pentoxifylline, fenofibrate or other fibrates, budesonide and other systemic corticosteroids, potentially hepatotoxic drugs (including α -methyl-dopa, sodium valproic acid, isoniazide, or nitrofurantoin), antibodies, and immunotherapy directed against interleukins or other cytokines or chemokines. However, which treatment led to the exclusion was not captured, and so it is not possible to say which, if any, patients were excluded due to the use of fibrates.

d. In addition, please provide reasons for exclusion for the 99 screen failures mentioned in the CONSORT patient-flow diagram (company submission, Figure 11).

Reasons for screening failures are provided in Table 12.

Fail category	Category details	Number of patients based on highest category
Failed inclusion category 2	ALP and/or bilirubin levels not documented	63
Failed inclusion category 1	At least two confirmatory symptoms of PBC not documented	5
Failed inclusion category 4	Stable UDCA dose or intolerance to UDCA not confirmed)	4
Failed inclusion category 5	Use of contraception not confirmed	1
Exclusion category 4	Patient has received a prohibited medication in previous 6 months	8
Exclusion category 9	Patient has a condition affecting the absorption of drugs (IBD)	2
Exclusion category 2	Patient suffers significant complications of PBC	2
Exclusion category 3	Patient suffers severe pruritus	2
Exclusion category 11	Patient has uncontrolled significant medical conditions	2
Other	Other exclusions	10

Table 12: Reasons for screening failures in POISE

Abbreviations: ALP, alkaline phosphatase; IBD, inflammatory bowel disease; PBC, primary biliary cirrhosis/cholangitis; UDCA, ursodeoxycholic acid.

e. Please include fibrates in the economic model and present the results.

Fibrates are not licenced or used in the UK, and are contraindicated in PBC. In addition, the available clinical evidence for fibrates in the form of acceptable quality RCTs and underlying safety concerns does not allow a meaningful comparison. Fibrates were not included as a comparator in the company submission for these reasons.

B11. The company have suggested that placebo is a comparator for people who are unable to tolerate UDCA. Please confirm that this refers to 'no additional treatment' in the final scope and describe the management of disease in patients who cannot tolerate UDCA.

Yes, this is correct.

Treatment effectiveness and adverse events

B12. **Priority question:** Table 49 of the company submission appears to describe the transition probabilities for the OCA comparator. The table is unclear:

a. Please provide the primary source(s) of the numbers used in the upper part of the table.

The primary source for the numbers in the upper part of the table is the POISE CSR. The numbers indicate the number of patients in each of the listed biochemical states at each given timepoint:

- State 1: ALP \leq 200 U/L and normal bilirubin
- State 2: ALP > 200 U/L and normal bilirubin
- State 3: abnormal bilirubin and rising, or compensated cirrhosis.

b. Please explain how probabilities are obtained from the upper part of the table.

Probabilities were generated by converting the number of patients in a given state into a percentage value, by the following formula:

$$\frac{n_x}{N} = p$$

Where n_x is the number of patients in a given state, N is the total number of patients who initiated that three-month cycle in a given state, and p is the transition probability. For example, in the 3-6 month columns, of the 25 patients starting out in state 1, 17 remained in state 1, whilst 7 transitioned to state 2, and 1 patient transitioned to state 3. To calculate the probability of a patient remaining in state 1, the following calculation would be performed:

$$\frac{17}{25} = 0.68$$

To give a quarterly probability of 0.68 to remain in state 1.

In months 0-3, discontinuation is considered. This is factored in by the following method:

$$\left(\frac{n_x}{N}\right) - \left(d\left(\frac{n_x}{N}\right)\right) = p$$

Where *d* is the rate of discontinuation, n_x is the number of patients in a given state, *N* is the total number of patients who initiated that three-month cycle in a given state, and *p* is the transition probability. For example, in the 0-3 month columns, of the 63 patients starting out in state 2, 24 patients remained in state 1, 39 patients transitioned to state 2, and no patients transitioned to state 3. To calculate the probability of a patient remaining in state 1, the following calculation would be performed:

$$\left(\frac{24}{63}\right) - \left(0.099\left(\frac{24}{N63}\right)\right) = 0.343$$

To give a quarterly probability of 0.343 to remain in state 1.

c. Please provide the methods (calculations) and results.

The original table (Table 49 in the original submission) is shown below (Table 13). A comparison table, with Excel formulae shown, shows the calculations that were performed (Table 14). Column and row numbers are shown to allow identification of the calculations that were performed. Table 13 should be used as a reference when reviewing Table 14Table 14.

	В	с	D	E	F	G	н	I	J	к	L	М	N	0	Р	Q	R	S
15			0-	3 months	S			3-6 m	onths			6-9 mc	onths			9-12 mc	onths	
16	From/To:	1	2	3	Disc.	Ν	1	2	3	Ν	1	2	3	Ν	1	2	3	Ν
17	17 ALP threshold 200 units / L																	
18	1																	
19	2																	
20	3																	
21	Probabilities																	
22	1					1				1				1				1
23	2					1				1				1				1
24	3					1				1				1				1

Table 13: Transition probabilities calculated from POISE

Abbreviations: ALP, alkaline phosphatase; Disc, discontinuation rate.

Table 14: Transition p	robabilities calculated from POISE ((calculations shown))
------------------------	--------------------------------------	----------------------	---

		0.0	(h				0.0	/ (6 0 . . .			0.40 m anth a			
		0-3 mon	ths			3-6 months			6-9 months			9-12 months					
From/ To:	1	2	3	Dis c.	Z	1	2	3	Ν	1	2	3	Ν	1	2	3	Ν
Probabi	robabilities																
1					1				1				1				1
2					1				1				1				1
3					1				1				1				1

www.nice.org.uk

d. Please explain how these numbers are used in the model and how discontinuation is handled in the model.

The POISE data are used to determine the transition probabilities for all patients in the first year of treatment. Different transition matrices were utilised for different patient populations; the full tables of transition probabilities are available in the model on the sheet 'Transition matrices' between rows 119 and 312.

Discontinuation is only considered during the first three months of the model. Patients who discontinue stay in their treatment arm, but use transition probabilities for patients treated with placebo.

e. Please explain why discontinuation is only included in the first 3 months of the year.

The exact time point for each discontinuation event within POISE is not known. With this in mind, it was assumed that patients would discontinue during the first three months of treatment. This was based on the following assumptions:

- Patients who experience adverse events would be more likely to discontinue earlier in their treatment course rather than later, as they would be more likely to experience adverse events immediately after starting treatment
- Following an initial increase in pruritus severity (as reported by the pruritus VAS from POISE), severity of pruritus decreases over the duration of treatment (Figure 1) (29). As pruritus is a strong driver for discontinuation, it can therefore be assumed that a decrease in pruritus would cause a corresponding decrease in treatment discontinuation over the course of OCA treatment.





Abbreviations: OCA, obeticholic acid; SE, standard error; VAS, visual analogue scale.

f. Please provide results of a sensitivity analysis including discontinuation in the later months (i.e. months 3-6, 6-9, 9-12).

This analysis is not possible, as the time of discontinuation cannot specifically be set beyond 0 to 3 months, and a restructure of how discontinuation is handled in the model would be required. The impact of discontinuation on the overall ICER is negligible. To illustrate the small impact of discontinuation, a sensitivity analysis is included below where the discontinuation rate has been set to 0% for all treatment arms.

Table 15: Deterministic results for UDCA-intolerant patients with discontinuation rates	set to
0% for all treatments (using the list price of OCA)	

Technologies		Total		In	crementa	al	ICER (cost/QALY)	ICER (cost/QALY)	
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental	
No treatment (placebo)	£103,233	11.30	6.76	-	-	-	-	-	
OCA Titration		17.22	14.28		5.93	7.52			

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxychoic acid.

Table 16: Deterministic results for UDCA inadequate responders with discontinuation rates set to 0% for all treatments (using the list price of OCA)

Technologies		Total		Inc	crementa	I	ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£96,977	12.35	7.98	_	-	-	-	-
OCA titration + UDCA		17.22	14.28		4.87	6.30		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

The ICER for UDCA-intolerant patients is 0.1% lower than the base case (QALY versus QALY in the base case), whereas the ICER for UDCA inadequate responder is 1% greater than the base case (QALY versus QALY in the base case).

B13. **Priority question:** For the comparators in the model, transition probabilities are primarily obtained from the literature.

a. Please justify why transition probabilities for the PBC-specific component of the model are not based on POISE for all treatment arms.

As stated in the answer to question B6, patients' progression in the placebo arm was based on transitions reflecting their liver transplant-free survival rates at 10 years, based on patientlevel data from POISE using both GLOBE and UK risk scores (1, 22). Patients in the UDCA arm were assumed to progress only to more severe health states to reflect the fact that their ALP levels remained similar to baseline throughout the 12-month treatment period and bilirubin levels increased at all time points in POISE. b. Please provide the results of a sensitivity analysis using the clinical data from POISE to inform the transition probabilities for all comparators for the PBC-specific component of the model. Provide details of the methods used.

In the base case, the risk of end-stage liver disease associated with the UDCA arm from POISE was captured, as well as the fact that patients had similar ALP but increasing bilirubin over the 12-month study period, thus worsening their risk of end-stage liver disease, liver transplant, or death. Since the transitions reflect the data from POISE, the proposed scenario is not necessary.

- B14. **Priority question:** Please explain the following concerning Table 50 of the company submission:
 - a. Please provide the primary source(s) on which the transition probabilities are based.

UDCA inadequate responders

Transition probabilities from POISE were used where possible, however transition probabilities were not available for all health states, nor were they available for the full time horizon of the model. The OCA titration regimen relies on POISE patient-level data to generate transition probabilities; transition probabilities for UDCA-based, or placebo regimens, are based on data sourced from the literature.

Calibration of transition probabilities was carried out in four steps. These were:

- Transition from "decompensated cirrhosis" to "pre-liver transplant" and "liver-related death"
- Transition from "abnormal bilirubin and rising, or compensated cirrhosis" to "decompensated cirrhosis" or "liver-related death"
- Transition from "ALP > 200 U/L and normal bilirubin" to "abnormal bilirubin and rising, or compensated cirrhosis"
- Transition from "ALP ≤ 200 U/L and normal bilirubin" to "abnormal bilirubin and rising, or compensated cirrhosis".

Step 1: Transition from "decompensated cirrhosis" to "pre-liver transplant" and "liver-related death"

The probabilities for this transition were calibrated to reflect the ten-year liver transplant-free survival (LTFS) reported by Harms et al. (2015) for PBC patients who had developed decompensated cirrhosis (23). In this study, decompensation was defined as the first instance of any of the following events, in a given patient:

- Ascites
- Variceal bleeding
- Hepatic encephalopathy

The study included 2,938 patients. 467 clinical events were reported; 136 liver transplantations (29.1%) and 331 deaths (70.9%). Survival data was presented for two patient groups:

- Patients who underwent decompensation
- Patients who did not undergo decompensation

Figure 2: Liver-transplant free survival of decompensated and non-decompensated patients with PBC



Transitions from decompensated cirrhosis to liver transplant waitlist and liver-related death were calibrated based on the 10-year liver-transplant free survival observed for decompensated PBC patients, while respecting the relative risk stating that patients have a 2.43-fold (i.e. 331/136) greater risk of dying than having a liver transplantation. In the calibration process, a simplified Markov model including 4 health states (i.e. decompensated cirrhosis, liver transplant waitlist, liver transplantation and death) was used. This is shown in the calibration spreadsheet for UDCA inadequate responders (see question B14 d).

To perform the calibration, Solver (an add-on module for Excel 2013) was use to estimate the transition probabilities from "decompensated cirrhosis" to "pre-liver transplant" which would result in a 10-year LFTS of 10% (as shown above in Figure 2). As shown in Table 17, patients who progress to liver transplant remain there to allow the calculation of the LTFS which captures time to first event (i.e. either liver transplant or death). The calibrated annual transitions from decompensated cirrhosis were as follows:

- To liver transplant waitlist: 7%
- To liver-related death: 17%

	DCC	Pre-LT (i.e. waiting list)	LT	Death
DCC	76%	7%	0.0%	17%*
Pre-LT (i.e. waiting list)	0%	52%	35%	9%
LT	0%	0%	100%	0.0%
Death	0%	0%	0%	100%

Table 17: Step 1 calibration results: Markov model – transitions from "decompensated cirrhosis"

Abbreviations: DCC: Decompensated cirrhosis; LT: liver transplantation.

Grey cell: Calculation – 1-sum(all other transitions).

* Transition to death is calculated as transition to pre-LT x 2.43 (231/331)

Source: Harms 2015 (23)

Step 2: Transitions from "abnormal bilirubin and rising, or compensated cirrhosis"

As data for transition probabilities were not available for specific histologic states of PBC patients, calibration was performed using groups of PBC patients who have abnormal bilirubin. Transitions to decompensated cirrhosis and liver-related deaths were calibrated based on the data shown in Figure 3 to replicate the 10-year liver-transplant survival observed in the GLOBE and UK PBC cohorts (the mean of the Global and UK cohorts was used). The transition probability for "abnormal bilirubin and rising" to "hepatocellular carcinoma" is 0.014, based on Trivedi et al. (2016) (30). The 10-year LTFS rate was estimated at 0.39 and 0.48, using the Global and UK PBC risk scores respectively. Lammers et al. (2014) was used to determine the numbers of patients who would have abnormal bilirubin after 12 months (31).



Figure 3: Liver-transplant free survival rates based on ALP and bilirubin thresholds

Liver transplant-free survival for Global PBC is based on all-cause mortality or liver transplant and the UKPBC is based on liver-related death or liver transplant. Courtesy of Global PBC Study Group and UK-PBC

Abbreviations: ALP, alkaline phosphatase; PBC, primary biliary cirrhosis/cholangitis; ULN, upper limit of normal.

The transition from "abnormal bilirubin and rising, or compensated cirrhosis" to "pre-liver transplant" was calibrated first. Based on data from the UNOS database, it was determined that 30-35% of new additions to the waitlist were added due to PBC, cirrhosis or HCC. (32) The transition was calibrated to ensure that 30% of patients would be on the liver transplant waitlist ("pre-liver transplant" health state) over the next 20 years, by assuming that patients who move to the "pre-liver transplant" state remain there (i.e. patients cannot leave the waitlist unless they die). Once the transition to the "pre-liver transplant" state was determined, the transition from to "abnormal bilirubin and rising, or compensated cirrhosis" to "decompensated cirrhosis" was calibrated to ensure that LTFS was 39% at 10 years.

	AB and rising, or CC	DCC	НСС	Pre-LT (i.e. waiting list)	LT	Death
AB and rising, or CC	85%	10%	1%	4%*	0%	0%
DCC	0%	76%	0%	7%	0%	17%
НСС	0%	0%	53%	4%	0%	43%
Pre-LT (i.e. waiting list)	0%	0%	0%	52%	35%	9%
LT	0%	0%	0%	0%	100%	0%
Death	0%	0%	0%	0%	0%	100%

Table 18: Step 2 calibration: Markov model – transitions from "abnormal bilirubin and rising, or compensated cirrhosis"

Abbreviations: AB, abnormal bilirubin (total bilirubin >ULN); CC, compensated cirrhosis; DCC, decompensated cirrhosis; LT, liver transplantation.

Grey cell: Calculation – 1-sum(all other transitions).

* Transition to pre-LT was calibrated first to reflect 30% of patients moving to waitlist over next 20 years (32)

Step 3: Transitions from "ALP > 200 U/L and normal bilirubin"

The transition from "ALP > 200 U/L and normal bilirubin" to "abnormal bilirubin and rising, or compensated cirrhosis" was calibrated to reflect the LTFS as estimated by the mean of the GLOBE and UK PBC risk score on the patient-level data from POISE (33) (31). The GLOBE and UK risk score both estimate LTFS over time. Whilst the GLOBE estimate includes the risk of all-cause death, including liver-related death, the UK risk score focuses on liver-related deaths only. As a result, whilst both estimates show a worsening prognosis for PBC patients on UDCA after 12 months, the magnitude of the change is lower when the UK PBC risk score is used. With wide confidence intervals around the survival rate estimates over time (see Figure 4), transitions were calibrated (again, using Solver) so that the 10-year LTFS was 78%, the mean of the UK and GLOBE estimates (87% and 69%). The calculated transition probabilities for this step of the process are shown in Table 19.

	ALP >1.67x ULN and NB	AB and rising, or CC	DCC	нсс	Pre-LT (i.e. waiting list	LT	Death
ALP >1.67x ULN and NB	88%	12%	0%	0%	0%	0%	0%
AB and rising, or CC	0%	85%	10%	1%	4%*	0%	0%
DCC	0%	0%	76%	0%	7%	0%	17%
НСС	0%	0%	0%	53%	4%	0%	43%
Pre-LT (i.e. waiting list)	0%	0%	0%	0%	52%	35%	9%
LT	0%	0%	0%	0%	0%	100%	0%
Death	0%	0%	0%	0%	0%	0%	100%

Table 19: Step 3 calibration: Markov model – transitions from "ALP >200 U/L and normal bilirubin"

Abbreviations: AB, abnormal bilirubin (total bilirubin <ULN); ALP, alkaline phosphatase; CC, compensated cirrhosis; DCC, decompensated cirrhosis; HCC, hepatocellular carcinoma; LT, liver transplantation; NB, normal bilirubin (total bilirubin ≤ULN); ULN, upper limit of normal. Grey cell: Calculation – 1-sum(all other transitions).

Figure 4: Step 3 calibration: liver transplant-free survival rates at 5 and 10 years as estimated by GLOBE, UK risk score, and the result from the calibrated model



Abbreviations: LTFS, liver transplant-free survival.

Step 4: Transition from "ALP ≤ 200 U/L and normal bilirubin"

The transition from "ALP \leq 200 U/L and normal bilirubin" to "abnormal bilirubin and rising, or compensated cirrhosis" was calibrated to reflect the liver-transplant free survival as shown in Figure 3 for patients with "ALP \leq 1.67xULN and normal bilirubin" after 12 months on treatment. The transition was calibrated (using Solver, as previously done) so that the 10-

year LTFS was 91%, the mean of the UK and Global PBC group estimates (96% and 87%). The final table of calibrated transition probabilities is shown below in Table 20.

	ALP ≤1.67x ULN and NB	ALP >1.67x ULN and NB	AB and rising, or CC	DCC	нсс	Pre-LT (i.e. waiting list)	LT	Death
ALP ≤1.67x ULN and NB	95%	0%	5%	0%	0%	0%	0%	0%
ALP >1.67x ULN and NB	0.0%	88.4%	11.6%	0%	0%	0%	0%	0%
AB and rising, or CC	0.0%	0.0%	84.9%	9.7%	1.4%	4.1%	0%	0%
DCC	0.0%	0.0%	0.0%	75.8%	0.0%	7.1%	0%	17.2%
НСС	0.0%	0.0%	0.0%	0.0%	53.0%	4.0%	0%	43.0%
Pre-LT (i.e. waiting list)	0.0%	0.0%	0.0%	0.0%	0.0%	55.3%	35%	9%
LT	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	100%	0%
Death	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	100%

Table 20: Step 4 calibration: Markov model from ALP >1.67x ULN and normal bilirubin

Abbreviations: AB, abnormal bilirubin (total bilirubin >ULN); ALP, alkaline phosphatase; CC, compensated cirrhosis; DCC, decompensated cirrhosis; HCC, hepatic cellular carcinoma; LT, liver transplantation; NB, normal bilirubin (total bilirubin ≤ULN); ULN, upper limit of normal. Grey cell: Calculation - 1-sum(all other transitions)

b. Please detail the characteristics of patients included in this/these primary source(s).

The following sources were used:

- GLOBE PBC cohort •
- UK PBC cohort •

The baseline demographics for each cohort are detailed below in Table 21 and Table 22.

Table 21: Global PBC cohor	t baseline characteristics
----------------------------	----------------------------

Characteristic	Global PBC Study Group (N=4845)
Age at entry (SD), years	54.5 (12.0)
Duration of PBC (SD), years	7.3 (3.6 – 11.5)
Female, n (%)	4348 (90)
UDCA-treated patients, n (%)	4119 (85)
Baseline ALP (IQR), x ULN	2.10 (1.31, 3.72)
Baseline ALP >3x ULN, n (%)	606 (16)
Baseline bilirubin (IQR), x ULN	0.67 (0.45 – 1.06)
Baseline bilirubin >ULN, n (%)	740 (20)
Biochemical disease stage, n (%):	
Early: bilirubin \leq ULN and albumin \geq LLN	2040 (67)
Moderately Advanced: abnormal bilirubin or albumin	730 (24)

Characteristic	Global PBC Study Group (N=4845)
Advanced: abnormal bilirubin and albumin	259 (9)

Abbreviations: ALP, alkaline phosphatase; IQR, interquartiler range; LLN, lower limit of normal; PBC, primary biliary cirrhosis/cholangitis; SD, standard deviation UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

Characteristic	Whole cohort Female		Male	P-value (female	
	(n=2353)	(n=2132)	(n=221)	vs male)	
Female, %	90.6	_	-	_	
Age at diagnosis (range), years	55 (16–86)	55 (16–86)	60 (34–81)	<.0001	
Age at study (range), years	65 (21–91)	65 (21–91)	67 (40–89)	<.0005	
Length of follow-up period (range), years	8 (0–52)	8 (0–52)	6 (0–31)	<.01	
UDCA use, % receiving	80	80	79	NS	
Duration of reported UDCA therapy (range), years	3.7 (0.1–21.1)	3.8 (0.1–21.1)	3.6 (0.2– 19.4)	NS	
UDCA response in adequately treated patients, %	79	80	72	<.05	
ALP, U/L	266 ± 262	267 ± 262 1.8	265 ± 249 1.8	NS	
ALP (range), xULN	1.8 (0.3–23.3)	(0.3–23.3)	(0.3–21.4)	NS	
ALT/AST, U/L	47.7 ± 45.5	47.4 ± 41.3	49.1 ± 43.8	NS	
ALT/AST (range), ULN	1.1 (0.2–20.3)	1.1 (0.2–20.3)	1.2 (0.3–8.3)	NS	
Bilirubin, µmol / L	11.2 ± 8.6	11.0 ± 8.6	13.8 ± 8.8	<.0001	
Bilirubin (range), xULN	0.6 (0.1–10.5)	0.6 (0.1–10.3)	0.6 (0.1–5.0)	<.0001	
Albumin, g/L	39.4 ± 8.5	39.4 ± 8.7	39.7 ± 7.3	NS	
Albumin (range), xLLN	1.0 (0.5–1.5)	1.0 (0.5–1.5)	1.0 (0.7–1.3)	NS	

Table 22: UK	PBC cohort baseline	e characteristics
--------------	---------------------	-------------------

Abbreviations: ALP, alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; LLN, lower limit of normal; NS, not significant; PBC, primary biliary cirrhosis/cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

c. Please provide an overview of the data extracted from this/these primary source(s) to calculate the transition probabilities.

The data extracted from these publications are detailed in the calibration calculations Excel workbook which will be submitted alongside this clarification letter.

d. In addition to Appendix 10, please provide a detailed explanation of the methods and sources used for calibration after data extraction (i.e. provide the calculations for the quarterly transition probabilities) and an Excel sheet containing these calculations.

A workbook containing clarifications has been submitted alongside this clarification letter.

- B15. **Priority question:** Please explain the following concerning Table 52 of the company submission:
 - a. Please provide the primary source(s) on which the transition probabilities are based.

UDCA-intolerant patients

UDCA-intolerant patients can either receive OCA monotherapy (as a titrated dose) or no treatment. Given the limited number of patients not on UDCA, there is a lack of data in the literature looking at the natural history of PBC without active treatment since the launch of UDCA. Thus, to reflect the progression of patients who did not receive any active treatment, assumptions were made based on Corpechot et al. (2000), who assessed the effect of UDCA therapy on liver fibrosis progression compared with no treatment in PBC patients (34). The study was based on a French randomised, double-blind, placebo-controlled trial in which 146 patients were randomised to receive either 13-15 mg/kg/day of UDCA or a placebo for 2 years. After 2 years, placebo patients crossed over to UDCA for a further 2 years, whilst patients already on UDCA remained on UDCA. Paired liver biopsies were available for 103 patients (53 UDCA and 50 placebo) during the two-year blinded period.

Corpechot et al. (2000) modelled the progression from two histologic stages:

- Non-fibrotic stage, defined by the presence of portal and periportal lesions without extensive fibrosis
- Fibrotic/cirrhotic stage, defined by the presence of numerous septa, bridging fibrosis or nodular cirrhotic formation.

The description provided by Corpechot et al for fibrotic and cirrhotic stages would contain compensated cirrhotic patients according to the METAVIR criteria (METAVIR stages F3 / F4) (35).

Progression between states was informed by the paired biopsy data. The authors of this study assumed that all patients in a given state and at a given time would have the same prognosis. Two transition rates were estimated to inform the model, i.e. the baseline transition rate α_0 and the regression coefficient for treatment β_t . The relationship between both parameters were expressed as follows: $\alpha_u = \alpha_0^* \exp(\beta_t x U)$ where U=0 for placebo and 1 for UDCA.

Figure 5: Two-stage Markov model



Based on the simplified model parameters (detailed in Table 23) it was possible to estimate the annual probability of progressing over time in the placebo group. Probabilities for patients to transition to the fibrotic/cirrhotic stage were calibrated to estimate the number of cirrhotic patients over time without the probability of moving backwards by minimising the root mean square error (RMSE), leading to a quarterly probability of progression to severe PBC of 0.079. The same transition was assumed between "abnormal bilirubin or compensated cirrhosis" and "decompensated cirrhosis" as for UDCA non-responders.

Whilst these histologic stages do not necessarily directly reflect compensated cirrhosis and decompensated cirrhosis, they essentially represent two different fibrotic or cirrhotic stages. This is the only publication which reflects the natural history of PBC without treatment, and allows estimation of the rate of disease progression for untreated patients.

Parameter	From/to	Estimate	SE
αΟ	NF to F/C	0.34	0.09
βt	NF to F/C	-1.55	0.49
αΟ	F/C to NF	0.03	0.01
βt	F/C to NF	0.00	0

Table 23: Simplified model - progression parameters

Abbreviations: F/C, fibrotic/cirrhotic; NF, non-fibrotic; SE, standard error. Source: Corpechot 2000 (34).

b. Please detail the characteristics of patients included in this/these primary source(s).

The Corpechot study utilised patient data from a French randomized, double-blind, placebocontrolled trial of UDCA therapy. The baseline characteristics of the patients are presented below in Table 24.

	Table 24:	Baseline	characteristics	of pa	tients in	the c	original	study	groups
--	-----------	----------	-----------------	-------	-----------	-------	----------	-------	--------

Characteristic	UDCA*	Placebo*
No. of patients	72	73
Age, years	55 ± 1	57 ± 1

Characteristic	UDCA*	Placebo*
Female sex, n (%)	68 (94)	65 (89)
Months since diagnosis (range)	39 ± 4 (1–132)	43 ± 5 (0–240)
Bilirubin – mg/dl	1.25 ± 0.11	1.24 ± 0.15
ALP, x normal	5.61 ± 0.40	4.26 ± 0.27
ALT, x normal	2.69 ± 0.19	2.25 ± 0.13
Γ-Glutamyltransferase, x normal	13.1 ± 1.1	13.2 ± 1.1
Albumin, g/dl	3.90 ± 0.05	4.01 ± 0.05
Prothrombin time, %	96.0 ± 1.0	95.4 ± 1.1
IgM, x normal	2.56 ± 0.24	2.49 ± 0.27
Total bile acids, µg/ml	10.5 ± 1.2	11.5 ± 1.9
Hepatomegaly, n (%)	17 (24)	20 (27)
Splenomegaly, n (%)	11 (15)	15 (21)
Clinically overt disease, n (%) [†]	47 (65)	44 (60)
Mayo risk score	4.87 ± 0.11	4.80 ± 0.12
Histologic stage III or IV, n (%)	36 (50)	31 (42)

Abbreviations: ALP, alkaline phosphatase; ALT, alanine aminotransferase; IgM, immunoglobulin M; SE, standard error; UDCA, ursodeoxycholic acid.

*Plus-minus values are means \pm SE. Enzyme activity and immunoglobulin levels are expressed as multiples of the upper limit of the normal value. These values were standardized by dividing the measured value by the reference value in the laboratory. To convert values for bilirubin to micromoles per litre, multiply by 17.1. To convert values for total bile acids to micromoles per litre, multiply by 2.547. †Clinically overt disease was defined by the presence of at least of the following signs or symptoms: jaundice, fatigue, pruritus, hepatomegaly, and splenomegaly.

Source: Poupon 1994 (36)

c. Please provide an overview of the data extracted from this/these primary source(s) to calculate the transition probabilities

The authors constructed a Markov model to estimate the probability of progression to extensive fibrosis and cirrhosis from a non-fibrotic or cirrhotic state. The model closely replicated the results of the trial, which found that patients on UDCA therapy had a five-fold lower risk of developing liver fibrosis or cirrhosis compared to placebo controls (34). The model was found to replicate the results of the clinical trial closely, and those of Locke *et al*, a study on the spontaneous progression of PBC (37).

	Consecutive stage							
	UD	СА	Placebo					
Initial stage	NF	F/C	NF	F/C				
NF	38 (38.01)	6 (5.99)	15 (14.98)	14 (14.02)				
F/C	4 (4.04)	64 (63.96)	1 (0.97)	20 (20.03)				

Abbreviations: NF, nonfibrotic stage; F/C, fibrotic/cirrhotic stage; UDCA, ursodeoxycholic acid.

Note: each cell gives the number of observed and predicted (in parentheses) transitions from the stage in the row to the stage in the column. For example, the number of observed transitions under UDCA from NF to F/C is 6 and that predicted by the model is 5.99.





Abbreviations: PBC, primary biliary cirrhosis/cholangitis; UDCA, ursodeoxycholic acid. Empty circle with solid line: model predicted UDCA disease course; solid circle with solid line: model predicted placebo disease course; empty circle with dashed line: observed results of the spontaneous course of the disease.

Source: Locke 1996 (37).

d. Please explain how Table 51 of the company submission should be used to obtain the estimates in Table 52.

An Excel spreadsheet containing the calculations performed to reach the results shown in Table 52 of the submission has been attached with these clarification questions. The Markov model detailed by Corpechot *et al* was recreated and calibrated (as described in B14 a).

- B16. **Priority question:** Please provide an overview of all transition probabilities used for all comparators and all time points (if time-dependent) for the PBC-specific component, incorporating:
 - a. the estimated probability
 - b. standard error
 - c. primary source
 - d. justification for source
 - e. method of calculation if applicable.

All transition probabilities used within the model, including their sources, can be found on the 'clinical inputs' page of the model, in the range E32:L61. Details of how calibration was performed can be found in the answers to questions B14 and B15.

B17. Please provide an overview of adverse event proportions for fatigue, pruritus and nausea (stratified by treatment).

The proportion of patients who experienced fatigue, pruritus and nausea, stratified by treatment, are reported below in Table 26 and Table 27. Note that adverse event incidence data were derived from the ITT population, and not for UDCA-intolerant / UDCA inadequate responder subpopulations.

Table 26: Proportions of patients who experienced adverse events, UDCA-intolerant patients

Treatment	Fatigue	Pruritus	Nausea
No treatment (placebo)	0.00%	0.00%	0.00%
OCA titration	8.57%	50.00%	4.29%

Abbreviations: OCA, obeticholic acid; UDCA, ursodeoxycholic acid. Source: POISE (29)

Table 27: Proportions of patients who experienced adverse events, UDCA inadequate responders

Treatment	Fatigue	Pruritus	Nausea
UDCA + placebo	10.96%	36.99%	5.48%
OCA titration + UDCA	8.57%	50.00%	4.29%
Abb an intigence OOA shotish aligned in LIDOA	المتعام والمحام والمعام والمعام		

Abbreviations: OCA, obeticholic acid; UDCA, ursodeoxycholic acid. Source: POISE (29).

Health related quality of life

- B18. **Priority question:** A recent publication (Dyson et al 2016) suggested that....."The majority of patients with primary biliary cholangitis do not feel their QoL is impaired, although impairment is reported by a sizeable minority".
 - a. Please justify the primary sources used for quality of life estimates.

The sources identified were selected because whilst the underlying cause of HCV / HBV differs, patients will often experience similar health states, for example decompensated cirrhosis. These data were used as there is no PBC-specific HRQoL data which uses a utility measure preferred by NICE, e.g. EQ-5D.

b. Please justify the high utility values for PBC, considering the general population utility (age-dependent).

The utility data used for the biochemistry states in the model uses data reported in Younossi *et al*, 2001 (38). The HRQoL for cholestatic liver disease (which includes PBC / PSC) was reported as 0.84. These are the only identified utility data for PBC patients.

c. Please clarify why the utilities from a hepatitis B/C population are appropriate (with or without the reduction of **100**), and justify that these health state utility values are based on the most appropriate source(s). Please explain the rationale for reducing the hepatitis values by **100**.

KOL input stated that PBC patients are likely to have lower QoL than HCV / HBV patients in comparable health states. To reflect this, a decrement was applied to all HCV / HBV – specific utility values, aside for HCC (according to KOL opinion, in HCC, the utility value is driven by the treatment for HCC rather than the disease itself).

The decrement in utility was not applied to disease states where utility values which could be considered specific to PBC were identified, i.e. the biochemistry states relating to ALP threshold (38).

B19. The impact of adverse events on health-related quality of life is not included in the cost-effectiveness model. Please provide a sensitivity analysis including the influence of adverse events on quality of life.

According to input by KOLs, the only recorded adverse event with a meaningful impact on quality of life is pruritus, and this subsides within the first three months of treatment. If HRQoL was modelled, it would have a negligible impact on the overall QALY. The impact of The difference in the rate of pruritus between UDCA and OCA regimens is also small, further decreasing the impact that this change would have on the overall QALYs calculated.

In addition, data from the Phase 2 study 747-201 showed that there were no statistically significant changes between baseline and end of treatment in quality of life, as assessed by the SF-36 and PBC-40 health questionnaires (39).

To illustrate the minor impact of the costs of adverse events on the overall results, a sensitivity analysis has been included where the cost of adverse events has been set to £0.

Figure 7: Deterministic results for UDCA-in	ntolerant patients with ad	verse event costs set to £0
(using the list price of OCA)		

Technologies	Total			Inc	crementa	l	ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
No treatment (placebo)	£103,233	11.30	6.61	-	-	-	-	_
OCA titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 28: Deterministic results for UDCA inadequate responders with adverse event costs set to £0 (using the list price of OCA)

Technologies	Total			Ind	crementa	I	ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£96,911	12.35	7.85	_	-	-	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.79		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Omitting the cost of adverse events results in an ICER of QALY in UDCA-intolerant patients. This is a decrease of 1.96% from the base case ICER for the same population (DALY). In UDCA inadequate responders, the ICER is QALY, and there is no difference between the ICERs for the sensitivity analysis and the base case result (QALY, <0.1% decrease). In summary, the impact of adverse events on model outcomes is minimal.

Resource use and costs

B20. It is unclear how the technology cost is used to obtain the total annual technology costs (company submission Table 65). Please clarify and justify the calculation of total annual technology costs.

The list price shown in table 65 is an error; the list price should read £29,005.78. The annual price is built up from the price per tablet assuming 365 days in the year $((\pounds 2,384.04/30)^*365)$.

B21. Costs of transplant and follow up costs are derived from the literature.

a. Please explain the relevance of using transplant-related cost assumptions based on Hepatitis C and B patients (Wright et al. 2006 and Singh 2014). Explain why NHS reference costs were not used.

NHS reference costs were not used, as the cost for the liver transplant health state reported in Wright et al 2006 and Singh 2014 reflect the total annual mean resource use, however it was not clear exactly how the micro-costing for the resource use was performed in terms of procedure codes.

In addition, the costs reported in the publications have been used previously in NICE TA330: Sofosbuvir for the treatment of chronic hepatitis C (15). As there is precedence for the use of these data, and KOLs have stated that the costs of treating each of the HCV health states in question (decompensated cirrhosis, hepatocellular carcinoma, liver transplant, liver transplant follow-up) would be identical to PBC, the existing costs in the model should be appropriate.

b. Please provide the results of a sensitivity analysis using reference cost estimates for liver transplant.

For the reasons discussed in the answer to question B21 a, the NHS reference cost for a liver transplant does not capture the full extent of transplantation costs. The currency code for a liver transplant (GA01C) places the cost to the NHS at £17,746.82 per transplant, approximately 27% of that estimated by Wright *et al* when inflated to 2016 values (13).

This does not include the additional costs of supporting care which is necessary for liver transplant patients, as was captured in the cost estimates generated by Wright et al. Using the NHS reference cost for liver transplants would be likely to underestimate the true cost of liver transplantation and its associated care costs.

B22. Outpatient appointment costs are based on expert opinion.

a. Please justify why outpatient appointment costs are based on expert opinion and not NHS reference costs (Table 66).

The NHS National Tariff 2015/2016 was used to estimate the cost of outpatient appointments. KOL input informed the number of each type of appointment which patients would be expected to receive, and the number of blood test patients would be expected to undergo on an annual basis.

Please provide a cost estimate for outpatient appointments based on NHS reference costs. Provide the results of a sensitivity analysis using this estimate

A sensitivity analysis has been included below, where NHS reference costs for outpatient appointments have been included. The currency code and its respective cost are presented below in Table 29.

able zer Reichener er nepateregy eurpatient appenditiont							
Outpatient attendance type	Service code	Cost					
Hepatology (total)	306	£220.08					

Table 29: Reference cost for hepatology outpatient appointment

Source: National Schedule of Reference Costs, Year 2014-15; 15 NHS trusts and NHS foundation trusts.

The usage of this new cost impacted the calculations for the health states "ALP \leq 200 U/L and normal bilirubin", and "ALP > 200 U/L and normal bilirubin", as shown below in Table 30.

Table 30: Updated annual costs for health states " ALP ≤200 U/L and normal bilirubin", and "ALP >200 U/L and normal bilirubin"

	ALP: ≤1.67x ULN and Bili: Normal	ALP: >1.67x ULN and Bili: Normal
Outpatient visit cost: first visit [†]	£220.08	£220.08
Outpatient visit cost: quantity / year*	1	1
Outpatient visit cost: follow up [†]	£0.00	£220.08
Outpatient visit cost: quantity / year*	1	2
Blood test cost [†]	£3	£3

Blood test quantity / year*	9	9	
Total annual cost	£247.08	£687.24	

Abbreviations: ALP, alkaline phosphatase; bili, bilirubin; ULN, upper limit of normal.

[†]Sourced from the National Schedule of Reference Costs, Year 2014-15; 15 NHS trusts and NHS foundation trusts. *Sourced from KOL opinion.

The following changes to the health state costs resulted in the following deterministic results:

Table 31: Deterministic results for UDCA-intolerant patients, using NHS reference costs for outpatient appointments (using the list price of OCA)

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	Incremental
No treatment (placebo)	£103,698	11.30	6.61	-	-	-	-	-
OCA Titration		16.65	13.52		5.35	6.91		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; NHS, National Health Service; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

Table 32: Deterministic results for UDCA inadequate responders, using NHS reference costs for outpatient appointments (using the list price of OCA)

Technologies	Total			Incremental			ICER (cost/QALY)	ICER (cost/QALY)
	Costs	LYG	QALYs	Costs	LYG	QALYs	versus baseline	incremental
UDCA + placebo	£97,858	12.35	7.85	_	-	_	-	-
OCA titration + UDCA		16.75	13.64		4.40	5.79		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life-years gained; NHS, National Health Service; OCA, obeticholic acid; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid.

The use of NHS reference costs had minimal impact on the ICERs; the ICER for UDCAintolerant patients was 1.7% lower compared to the base case ICER (), and the ICER for UDCA inadequate responders () was 0.19% greater in this sensitivity analysis.

B23. Please justify why the health state cost for 'Abnormal/Total bilirubin> 20 μmol/L and rising, or CC' is half the 'DCC' health state cost.

The health state for 'abnormal total bilirubin >20 μ mol/L and rising, or CC' has been assumed to half the cost of managing DCC based on expert recommendation. The rationale for this was that patients in this health state would have a higher healthcare resource used than patients with CC due to the health issues associated with rising levels of bilirubin, since this is viewed as a sign of disease progression and liver failure. This is also why liver transplantation can be considered at this stage.

Results and validation

B24. Please provide an overview of disaggregated life years gained, as provided for QALYs gained in Tables 73 and 74 of the company submission.

Table 33 and Table 34 show the disaggregated life years gained for both UDCA-intolerant patients and UDCA inadequate responders.

Health state	OCA titration	No treatment (placebo)	Increment
ALP: ≤ 200 U/L and Bili: Normal	7.120	0.000	7.120
ALP: > 200 U/L and Bili: Normal	8.091	2.434	5.657
Bili: Abnormal and rising, or CC	0.716	4.573	-3.856
Discontinuation	0.012	0.000	0.012
Decompensated cirrhosis	0.278	1.702	-1.423
HCC	0.024	0.145	-0.122
Pre transplant (end stage)	0.089	0.543	-0.454
Liver transplant	0.009	0.055	-0.046
Post liver transplant	0.240	1.443	-1.204
Re-emergence of PBC	0.069	0.402	-0.333
Total	16.648	11.297	5.351

Table 33: Summary of life years gained by health state for UDCA-intolerant population

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

Health state	OCA titration + UDCA	UDCA + placebo	Increment
ALP: ≤ 200 U/L and Bili: Normal	7.120	0.000	7.120
ALP: > 200 U/L and Bili: Normal	8.303	4.604	3.699
Bili: Abnormal and rising, or CC	0.664	4.040	-3.376
Discontinuation	0.012	0.005	0.007
Decompensated cirrhosis	0.257	1.489	-1.231
HCC	0.022	0.128	-0.106
Pre transplant (end stage)	0.082	0.476	-0.394
Liver transplant	0.008	0.048	-0.040
Post liver transplant	0.219	1.231	-1.013
Re-emergence of PBC	0.062	0.331	-0.269
Total	16.750	12.351	4.399

Table 34: Summary of life years gained by health state for UDCA inadequate responders

Abbreviations: ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; UDCA, ursodeoxycholic acid.

B25. Please provide the outcomes of expert opinion meetings (i.e. individual answers on all of the questions provided in Appendix 11).

The company are not authorised to release the details of all the conversations with external experts; however, we will happily answer any specific questions.

- B26. No cross and external validations have been performed and no results of the validation steps described in Section 5.10.1 of the company submission are provided.
 - a. Please perform cross and external validations (for instance, by using the following tool: <u>http://www.uk-pbc.com/resources/tools/riskcalculator/</u>).

It is not possible to fulfil this request; the UK-PBC risk calculator tool is designed to handle patient-level data. In order to perform this analysis, a single mean biochemistry reading would need to be calculated for the entire patient cohort, which has the potential to introduce bias. However, question A9 (Table 1) shows the results of an analysis of ESLD risk using the UK-PBC risk score algorithm. Following one year of OCA treatment, patients had a lower risk of ESLD when they crossed over onto UDCA + placebo treatment (1). External validation with other models is also not possible, since this is the first cost-effectiveness model in PBC.

b. Please provide the results of other validation steps as described in Section 5.10.1 (e.g. comparison of model results with POISE data).

The model was not validated against POISE (this was stated in error). The model outputs were instead validated against the GLOBE and UK PBC cohort survival rates, shown in Harms *et al*, 2016 (22).

The model cannot be directly validated against POISE, as POISE was one year in length, whereas the cost-utility model has an effective time horizon of 44 years.

External validation against other models cannot be performed, as this analysis is the first cost-utility model in PBC.

References

- Carbone MJ, D.; Mells, G.;, Pencek, R.; Marmon, T.; Shapiro, D. Predicted Risk of End Stage Liver Disease with Continued Standard of Care and Subsequent Addition of Obeticholic Acid in Patients with PBC. AASLD abstract 361. Hepatology. 2016;64(1(suppl)):184A.
- 2. Vierling JMH, G.M.; Jones, D.; Groszmann, R.J.; Kowdley, K.V.; Pencek, R.; Marmon, T.; MacConell, L. Efficacy of obeticholic acid treatment in patients with primary biliary cholangitis with cirrhosis. Hepatology.64(1(suppl)):187A.
- 3. Ali AH, Tabibian JH, Carey EJ, Lindor KD. Emerging drugs for the treatment of Primary Biliary Cholangitis. Expert Opin Emerg Drugs. 2016;21(1):39-56.
- 4. Shapiro JM, Smith H, Schaffner F. Serum bilirubin: a prognostic factor in primary biliary cirrhosis. Gut. 1979 Feb;20(2):137-40.
- 5. Bonnand AM, Heathcote EJ, Lindor KD, Poupon RE. Clinical significance of serum bilirubin levels under ursodeoxycholic acid therapy in patients with primary biliary cirrhosis. Hepatology. 1999 Jan;29(1):39-43.
- 6. Krzeski P, Zych W, Kraszewska E. Is serum bilirubin concentration the only valid prognostic marker in primary biliary cirrhosis? Hepatology. 1999;30(4):865-9.
- 7. Dickson ER, Grambsch PM, Fleming TR, Fisher LD, Langworthy A. Prognosis in primary biliary cirrhosis: Model for decision making. Hepatology. 1989 Jul;10(1):1-7.
- 8. Kamath PS, Wiesner RH, Malinchoc M, Kremers W, Therneau TM, Kosberg CL, et al. A model to predict survival in patients with end-stage liver disease. Hepatology. 2001 Feb;33(2):464-70.
- 9. Child CG, Turcotte JG. Surgery and portal hypertension. Major Probl Clin Surg. 1964;1:1-85.
- 10. Pugh RN, Murray-Lyon IM, Dawson JL, Pietroni MC, Williams R. Transection of the oesophagus for bleeding oesophageal varices. Br J Surg. 1973 Aug;60(8):646-9.
- 11. Lammers WJ, Hirschfield GM, Corpechot C, Nevens F, Lindor KD, Janssen HL, et al. Development and Validation of a Scoring System to Predict Outcomes of Patients With Primary Biliary Cirrhosis Receiving Ursodeoxycholic Acid Therapy. Gastroenterology. 2015 Dec;149(7):1804-12 e4.
- 12. Shepherd J, Jones J, Hartwell D, Davidson P, Price A, Waugh N. Interferon alpha (pegylated and non-pegylated) and ribavirin for the treatment of mild chronic hepatitis C: a systematic review and economic evaluation. Health Technol Assess. 2007 Mar;11(11):1-205, iii.
- 13. Wright M, Grieve R, Roberts J, Main J, Thomas HC, Investigators UKMHCT. Health benefits of antiviral therapy for mild chronic hepatitis C: randomised controlled trial and economic evaluation. Health Technol Assess. 2006 Jul;10(21):1-113, iii.
- 14. Hartwell D, Jones J, Baxter L, Shepherd J. Peginterferon alfa and ribavirin for chronic hepatitis C in patients eligible for shortened treatment, re-treatment or in HCV/HIV co-infection: a systematic review and economic evaluation. Health Technol Assess. 2011 Apr;15(17):i-xii, 1-210.
- 15. NICE. Sofosbuvir for treating chronic hepatitis C. NICE technology appraisal guidance [TA330]. NICE; 2015 [updated 2015; cited 2016 12th September]; Available from: <u>https://www.nice.org.uk/guidance/ta330</u>.
- 16. Harms MHP, A.; Mason, A.L.; Lammers, W.J.; Battezzati, P.M.; Floreani, A.; LaRusso, N.F.; Lindor, K.; Nevens, F.; Thorburn, D.; Kowdley, K.V.; Ponsioen, C.

Behavioral patterns of total serum bilirubin prior to major clinical endpoints in 3529 patients with primary biliary cholangitis. Journal of hepatology. 2016;64:S633-4.

- 17. Lindor KD, Gershwin ME, Poupon R, Kaplan M, Bergasa NV, Heathcote EJ, et al. Primary biliary cirrhosis. Hepatology. 2009 Jul;50(1):291-308.
- 18. Selmi C, Bowlus CL, Gershwin ME, Coppel RL. Primary biliary cirrhosis. Lancet. 2011 May 7;377(9777):1600-9.
- 19. Al-Harthy N, Kumagi T. Natural history and management of primary biliary cirrhosis. Hepat Med. 2012;4:61-71.
- 20. D'Amico G, Garcia-Tsao G, Pagliaro L. Natural history and prognostic indicators of survival in cirrhosis: a systematic review of 118 studies. J Hepatol. 2006 Jan;44(1):217-31.
- 21. Lammers WJvB, H. R.; Hirschfield, G. M.; Pares, A.; Kumagi, T.; Caballeria, L.; Invernizzi, P.; Lleo, A. Alkaline phosphatase values are a surrogate marker in prediction of transplant free survival in patients with primary biliary cirrhosis - an international, colloborative analysis. Journal of hepatology. 2013;58(S388).
- 22. Harms MHL, W.J.; Marmon, T.; Pencek, R.; van Buuren, H.R.; Hansens, B.E.; Shapiro, D. Improvement in estimated survival after 1 year of obeticholic acid treatment [poster 352]. Journal of hepatology. 2016;64:S631-832.
- 23. Harms MHL, W.J.; Janssen, H.L.; Corpechot, C.; Invernizzi, P.; Mayo, M.J.; Battezzati, P.M.; Floreani, A.; Pares, A.; Nevens, F.; Mason, A.L.; Kowdley, K.; Ponsioen, C.; Bruns, T.; Dalekos, G.; Thorburn, D.; Hirschfield, G.M.; LaRusso, N.F.; Lleo, A.; Cazzagon, N.; Franceschet, I.; Caballeria, L.; Zachou, K.; Poupon, R.; Cheung, A.; Trivedi, P.J.; Carbone, M.; Lindor, K.; van Buuren, H.R.; Hansen, B.E. Incidence and impact of decompensating events in primary biliary cirrhosis - results of an international follow up study of 3030 patients. Abstract presented at the 66th Annual Meeting of the American Association for the Study of Liver Diseases, San Francisco, CA, USA. Novermber 13-17 2015. AASLD; 2015. 2015.
- 24. Charatcharoenwitthaya P, Pimentel S, Talwalkar JA, Enders FT, Lindor KD, Krom RA, et al. Long-term survival and impact of ursodeoxycholic acid treatment for recurrent primary biliary cirrhosis after liver transplantation. Liver Transpl. 2007 Sep;13(9):1236-45.
- 25. Bezafibrate summary of product chatacteristics.
- 26. Fenofibrate summary of product characteristics.
- 27. Rudic JS, Poropat G, Krstic MN, Bjelakovic G, Gluud C. Bezafibrate for primary biliary cirrhosis. Cochrane Database Syst Rev. 2012 Jan 18;1:CD009145.
- 28. Hosonuma K, Sato K, Yamazaki Y, Yanagisawa M, Hashizume H, Horiguchi N, et al. A prospective randomized controlled study of long-term combination therapy using ursodeoxycholic acid and bezafibrate in patients with primary biliary cirrhosis and dyslipidemia. Am J Gastroenterol. 2015 Mar;110(3):423-31.
- 29. Intercept Pharmaceuticals Inc. Clinical study report: 747-301. Obeticholic acid (OCA). A Phase 3, double blind, placebo controlled trial and long term safety extension of obeticholic acid in patients with primary biliary cirrhosis. Data on File. 2015.
- 30. Trivedi PJ, Lammers WJ, van Buuren HR, Pares A, Floreani A, Janssen HL, et al. Stratification of hepatocellular carcinoma risk in primary biliary cirrhosis: a multicentre international study. Gut. 2016 Feb;65(2):321-9.
- 31. Lammers WJ, van Buuren HR, Hirschfield GM, Janssen HL, Invernizzi P, Mason AL, et al. Levels of alkaline phosphatase and bilirubin are surrogate end points of outcomes of patients with primary biliary cirrhosis: an international follow-up study. Gastroenterology. 2014 Dec;147(6):1338-49 e5; quiz e15.
- 32. Kim WR, Lake JR, Smith JM, Skeans MA, Schladt DP, Edwards EB, et al. OPTN/SRTR 2013 Annual Data Report: liver. Am J Transplant. 2015 Jan;15 Suppl 2:1-28.
- 33. Carbone M, Mells GF, Pells G, Dawwas MF, Newton JL, Heneghan MA, et al. Sex and age are determinants of the clinical phenotype of primary biliary cirrhosis and

response to ursodeoxycholic acid. Gastroenterology. 2013 Mar;144(3):560-9 e7; quiz e13-4.

- 34. Corpechot C, Carrat F, Bonnand AM, Poupon RE, Poupon R. The effect of ursodeoxycholic acid therapy on liver fibrosis progression in primary biliary cirrhosis. Hepatology. 2000 Dec;32(6):1196-9.
- 35. Rossi E, Adams LA, Bulsara M, Jeffrey GP. Assessing liver fibrosis with serum marker models. Clin Biochem Rev. 2007 Feb;28(1):3-10.
- 36. Poupon RE, Poupon R, Balkau B. Ursodiol for the long-term treatment of primary biliary cirrhosis. The UDCA-PBC Study Group. N Engl J Med. 1994 May 12;330(19):1342-7.
- 37. Locke GR, 3rd, Therneau TM, Ludwig J, Dickson ER, Lindor KD. Time course of histological progression in primary biliary cirrhosis. Hepatology. 1996 Jan;23(1):52-6.
- 38. Younossi ZM, Kiwi ML, Boparai N, Price LL, Guyatt G. Cholestatic liver diseases and health-related quality of life. Am J Gastroenterol. 2000 Feb;95(2):497-502.
- 39. Intercept Pharmaceuticals Inc. Clinical study report: 747-201. Obeticholic acid (OCA). A study of INT-747 (6-ECDCA) monotherapy in patients with primary biliary cirrhosis. Data on File.

Appendix 1: Publications removed at second pass due to falling within the scope of one of the nine Cochrane reviews

- 1. Christensen, E., et al., Azathioprine in primary biliary cirrhosis: late results of an international trial. Liver, 1984. 4.
- Christensen, E., et al., Beneficial effect of azathioprine and prediction of prognosis in primary biliary cirrhosis. Final results of an international trial. Gastroenterology, 1985. 89(5): p. 1084-91.
- Poupon, R., B. Balkau, and R. Poupon, Beneficial effect of ursodeoxycholic acid (UDCA) in primary biliary cirrhosis (PBC) Final results of the French Canadial trial. Hepatology, 1990. 12.
- 4. Dickson, E. and K. Lindor, Beneficial effects of ursodeoxycholic acid in an open trial of patients with primary biliary cirrhosis. Bile acids as therapeutic agents From basic science to clinical practice Falk symposium 58 1991.
- 5. Kurihara, T., et al., Bezafibrate in the treatment of primary biliary cirrhosis: comparison with ursodeoxycholic acid. Am J Gastroenterol, 2000. 95(10): p. 2990-2.
- 6. Kanda, T., et al., Bezafibrate treatment: a new medical approach for PBC patients? J Gastroenterol, 2003. 38(6): p. 573-8.
- 7. Combes, B., et al., Biliary bile acids in primary biliary cirrhosis: effect of ursodeoxycholic acid. Hepatology, 1999. 29(6): p. 1649-54.
- Heathcote, E.J., et al., The Canadian Multicenter Double-blind Randomized Controlled Trial of ursodeoxycholic acid in primary biliary cirrhosis. Hepatology, 1994. 19(5): p. 1149-56.
- 9. Kaplan, M., et al., Colchicine is effective in the treatment of primary biliary cirrhosis. Hepatology, 1985. 5.
- 10. Bodenheimer, H., Jr., F. Schaffner, and J. Pezzullo, colchicine therapy in primary biliary cirrhosis. Hepatology, 1986. 6.
- 11. Bobadilla, J., et al., Colchine and ursodiol in the treatment of primary biliary cirrhosis. Hepatology, 1994. 20.
- 12. Huet, P., et al., The combination of ursodeoxycholic acid (UDCA) and colchicine (C) for patients with primary biliary cirrhosis (PBC): effect on hepatic function and portal hypertension. Hepatology, 1996. 23.
- 13. Leuschner, M., et al., Combination of ursodeoxycholic acid (UDCA) with budesonide (BUD) is superior to UDCA-mono-therapy in primary biliary cirrhosis (PBC). Journal of hepatology, 1999. 30(57).

- 14. Gonzalez-Koch A., B.J., Antezana C., Smok G., Cumsille M.A., The combination of ursodeoxycholic acid and methotrexate for primary biliary cirrhosis is not better than ursodeoxycholic acid alone. Journal of Hepatology 1997. 27(1): p. 143-149.
- Van Steenbergen, W., et al., Combined treatment with methotrexate and ursodeoxycholic acid in non-cirrhotic primary biliary cirrhosis. Acta Clin Belg, 1996. 51(1): p. 8-18.
- Raedsch, R., et al., [Combined ursodeoxycholic acid plus colchicine--treatment of primary biliary cirrhosis: results of a placebo-controlled double-blind study]. Z Gastroenterol, 1992. 30 Suppl 1: p. 55-7.
- Raedsch, R., et al., [Combined ursodeoxycholic acid plus colchicine--treatment of primary biliary cirrhosis: results of a placebo-controlled double-blind study]. Bile acids and the hepatobiliary system From basic science to clinical practice Proceedings of Falk symposium 68th held in Basel, Switzerland, October 12 14, 1992 1993. 30 Suppl 1: p. 303-309.
- Raedsch, R., et al., Controlled study on the effects of a combined ursodeoxycholic acid plus colchicine treatment in primary biliary cirrhosis. Bile acids and the hepatobiliary system From basic science to clinical practice Proceedings of Falk symposium 68th held in Basel, Switzerland, October 12 14, 1992 1993.
- 19. Ross, A. and S. Sherlock, A controlled trial of azathioprine in primary biliary cirrhosis. Gut, 1971. 12.
- 20. Warnes, T., et al., A controlled trial of colchicine in primary biliary cirrhosis. Hepatology, 1984. 4.
- 21. Warnes, T.W., A. Smith, and F.I. Lee, A controlled trial of colchicine in primary biliary cirrhosis. J Hepatol, 1985. 5(1): p. 1-7.
- 22. Warnes, T.W., et al., A controlled trial of colchicine in primary biliary cirrhosis. Trial design and preliminary report. J Hepatol, 1987. 5(1): p. 1-7.
- 23. Wiesner, R.H., et al., A controlled trial of cyclosporine in the treatment of primary biliary cirrhosis. N Engl J Med, 1990. 322(20): p. 1419-24.
- 24. Macklon, A., M. Bassendine, and O. James, Controlled trial of D-penicillamine in primary biliary cirrhosis: incidence of side effects and relation to dose. Hepatology, 1982. 2.
- 25. Jain, S., et al., A controlled trial of D-penicillamine therapy in primary biliary cirrhosis. Lancet, 1977. 1(8016): p. 831-4.
- 26. Mitchison, H.C., et al., A controlled trial of prednisolone treatment in primary biliary cirrhosis. Three-year results. J Hepatol, 1992. 15(3): p. 336-44.
- 27. Lombard. M, et al., Cyclosporin A in primary biliary cirrhosis: results of a long-term placebo controlled trial and effect on survival. Hepatology, 1990. 12.

- 28. Lombard, M., et al., Cyclosporin A treatment in primary biliary cirrhosis: results of a long-term placebo controlled trial. Gastroenterology, 1993. 104(2): p. 519-26.
- 29. Mora, G., et al., Does treatment with ursodeoxycholic acid (UDCA) really diminish cholesterol serum levels in primary biliary cirrhosis (PBC). Hepatology, 1994. 19.
- 30. Emond, M., et al., Does ursodeoxycholic acid improve survival in patients with primary biliary cirrhosis? Comparison of outcome in the US multicenter trial to expected survival using the Mayo Clinic prognostic model. Hepatology, 1996. 24.
- 31. Neuberger, J., et al., Double blind controlled trial of d-penicillamine in patients with primary biliary cirrhosis. Gut, 1985. 26(2): p. 114-9.
- 32. Mitchison, H., et al., Double blind placebo-controlled trial of prednisolone treatment in primary biliary cirrhosis (PBC): a 3 year update. Journal of hepatology, 1989. 9.
- 33. Kaplan, M., et al., Double blind trial of methotrexate (MTX) versus colchicine (COLCH) in primary biliary cirrhosis (PBC). Hepatology, 1993. 18(176A).
- 34. Taal, B.G., S.W. Schalm, and F.W.J. Kate, Double-blind controlled study of penicillamine in primary biliary cirrhosis: Dose-dependent effects. Nederlands tijdschrift voor geneeskunde 1982. 126.
- 35. Triger, D.R., et al., D-penicillamine in primary biliary cirrhosis: two year results of a single centre, doubled-blind controlled trial. Gut, 1980. 21.
- 36. Epstein, O., et al., D-penicillamine treatment improves survival in primary biliary cirrhosis. Lancet, 1981. 1(8233): p. 1275-7.
- 37. Nakai, S., et al., The effect of bezafibrate in patients with primary biliary cirrhosis. Hepatology, 1999. 30.
- 38. Senior, J.R., C.B. O'Brien, and E.R. Dickson, Effect of oral ursodiol treatment on the predicted probability of mortality in primary biliary cirrhosis. Hepatology, 1990. 12.
- 39. Epstein, G., et al., Effect of penicillamine on immune complexes and immunoglobulins in primary biliary cirrhosis(PBC). Gut, 1978. 19.
- 40. Balan, V., et al., Effect of ursodeoxycholic acid on serum lipids of patients with primary biliary cirrhosis. Mayo Clin Proc, 1994. 69(10): p. 923-9.
- 41. Combes, B., et al., The effect of ursodeoxycholic acid on the florid duct lesion of primary biliary cirrhosis. Hepatology, 1999. 30(3): p. 602-5.
- 42. Batta, A., et al., Effect or ursodiol on biliary bile acid composition and conjugation in patients with primary biliary cirrhosis. Gastroenterology, 1990. 98.
- 43. Ikeda, T., et al., Effects of additional administration of colchicine in ursodeoxycholic acid-treated patients with primary biliary cirrhosis: a prospective randomized study. J Hepatol, 1996. 24(1): p. 88-94.

- 44. Bodenheimer, H.C., Jr., et al., Effects of penicillamine on serum immunoglobulins and immune complex-reactive material in primary biliary cirrhosis. Gastroenterology, 1985. 88(2): p. 412-7.
- 45. Huet, P., et al., Effects of ursodeoxycholic acid (UDCA) on hepatic function and portal hypertension in primary biliary cirrhosis (PBC). Hepatology, 1990. 12.
- 46. Raedsch, R., et al., Effects of ursodeoxycholic acid and ursodeoxycholic acid plus colchicine in primary biliary cirrhosis: a double-blind pilot study. Bile acids as therapeutic agents From basic science to clinical practice Falk symposium 58 1991.
- 47. Batts, K., et al., The effects of ursodeoxycholic acid on hepatic inflammation and histologic stage in patients with primary biliary cirrhosis. Hepatology, 1993. 18.
- Batts, K.P., et al., Effects of ursodeoxycholic acid on hepatic inflammation and histological stage in patients with primary biliary cirrhosis. Am J Gastroenterol, 1996. 91(11): p. 2314-7.
- 49. Lindor, K.D., et al., Effects of ursodeoxycholic acid on survival in patients with primary biliary cirrhosis. Gastroenterology, 1996. 110(5): p. 1515-8.
- Iwasaki, S., et al., The efficacy of ursodeoxycholic acid and bezafibrate combination therapy for primary biliary cirrhosis: A prospective, multicenter study. Hepatol Res, 2008. 38(6): p. 557-64.
- 51. Bodenheimer, H., Jr., F. Schaffner, and J. Pezzullo, Evaluation of colchicine therapy in primary biliary cirrhosis. Gastroenterology, 1988. 95(1): p. 124-9.
- 52. Carithers, R., et al., Extended follow-up of patients in the US Multicenter trial of ursodeoxycholic acid for primary biliary cirrhosis. Gastroenterology, 1996. 110.
- 53. Heathcote, E., et al., A four-year follow-up study of ursodeoxycholic acid therapy for primary biliary cirrhosis. Gastroenterology, 1993. 104.
- 54. Chazouilleres, O., et al., Histological course of primary biliary cirrhosis (PBC) treated with ursodeoxycholic acid (UDCA). Hepatology, 1995. 22.
- 55. Poupon, R., et al., [Hypocholesterolemic properties of ursodeoxycholic acid in patients with primary biliary cirrhosis]. Medecine & chirurgie digestives 1995. 24: p. 163 164.
- 56. Raedsch, R., et al., [Influence of ursodeoxycholic acid and urso plus colchicine on primary biliary cirrhosis: a double blind controlled pilotstudy]. Klinische Wochenschrift 1991. 69.
- Raedsch, R., et al., Influence of ursodesoxycholic acid and urso plus colchizine on the primary biliary cirrhosis: A double-blind controlled study. Klinische Wochenschrift 1991.
 69.
- Almasio, P., et al., The Italian multi-centre randomized controlled trial of ursodeoxycholic acid vs ursodeoxycholic acid plus colchine in symptomatic primary biliary cirrhosis Hepatology, 1994. 20.
- 59. Huet, P., J. Huet, and S. Hotte, Long term effect of ursodeoxycholic acid (UDCA) on hepatic function and portal hypertension in primary biliary cirrhosis (PBC). Hepatology, 1994. 20.
- Podda, M., et al., Long-term effect of the administration of ursodeocycholic acid alone or with colchicine in patients with primary biliary cirrhosis: a double-blind multicentre study. Bile acids and the hepatobiliary system From basic science to clinical practice Proceedings of Falk symposium 68th held in Basel, Switzerland, October 12 14, 1992 1993.
- 61. Pares, A., et al., Long-term effects of ursodeoxycholic acid in primary biliary cirrhosis: results of a double-blind controlled multicentric trial. UDCA-Cooperative Group from the Spanish Association for the Study of the Liver. J Hepatol, 2000. 32(4): p. 561-6.
- 62. Pares, A., Long-term treatment of primary biliary cirrhosis with ursodeoxycholic acid: results of a randomized, double-blind, placebo-controlled trial. Journal of hepatology, 1997. 26.
- 63. Hadziyannis, S., et al., Long-term treatment of primary biliary cirrhosis with ursodeoxycholic acid: the third year of a controlled trial. Bile acids as therapeutic agents From basic science to clinical practice Falk symposium 58 1991.
- 64. Pares, A., L. Caballeria, and J. Rodes, Long-term ursodeoxycholic acid treatment delays progression of mild primary biliary cirrhosis. Journal of hepatology, 2001. 34.
- 65. Taal, B.G., et al., Low therapeutic value of D-penicillamine in a short-term prospective trial in primary biliary cirrhosis. Liver, 1983. 3(6): p. 345-52.
- 66. Giaffer, M., et al., Low-dose methotrexate in treatment of primary biliary cirrhosis. Gut, 1995. 36(A30).
- 67. Hendrickse M.T., R.E., Giaffer M.H., Soomro I., Triger D.R., Underwood J.C.E., Gleeson D., Low-dose methotrexate is ineffective in primary biliary cirrhosis: Longterm results of a placebo-controlled trial. Gastroenterology, 1999. 117(2): p. 400-407.
- 68. Kaplan, M., G. Dickstein, and C. Schmid, Methotrexate (MTX) improves histology in primary biliary cirrhosis (PBC). Hepatology, 1994. 20(152A).
- 69. Combes, B., et al., Methotrexate (MTX) plus ursodeoxycholic acid (UDCA) in the treatment of primary biliary cirrhosis. Hepatology, 2005. 42(5): p. 1184-93.
- Steenbergen, W., et al., Methrotrexate alone or in combination with ursodeoxycholic acid as possible treatment in primary biliary cirrhosis. Cholestatic Liver diseases : New Strategies for Prevention and Treatment of Hepatobiliary and Cholestatic Liver Diseases Falk Symposium 75 1994.

- 71. Oka, H., et al., A multi-center double-blind controlled trial of ursodeoxycholic acid for primary biliary cirrhosis. Gastroenterol Jpn, 1990. 25(6): p. 774-80.
- 72. Poupon, R.E., et al., A multicenter, controlled trial of ursodiol for the treatment of primary biliary cirrhosis. UDCA-PBC Study Group. N Engl J Med, 1991. 324(22): p. 1548-54.
- Almasio, P.L., et al., Multicentre randomized placebo-controlled trial of ursodeoxycholic acid with or without colchicine in symptomatic primary biliary cirrhosis. Alimentary Pharmacology and Therapeutics, 2000. 14(12).
- Leuschner, M., et al., Oral budesonide and ursodeoxycholic acid for treatment of primary biliary cirrhosis: results of a prospective double-blind trial. Gastroenterology, 1999. 117(4): p. 918-25.
- 75. Hanley, D.A., et al., Parameters of calcium metabolism during a pilot study of cyclosporin A in patients with symptomatic primary biliary cirrhosis. Clin Invest Med, 1991. 14(4): p. 282-7.
- 76. Mitchison, H., et al., A pilot double blind controlled trial of prednisolone treatment in primary biliary cirrhosis. Journal of hepatology, 1986. 3(S28).
- 77. Minuk, G.Y., et al., Pilot study of cyclosporin A in patients with symptomatic primary biliary cirrhosis. Gastroenterology, 1988. 95(5): p. 1356-63.
- Mitchison, H.C., et al., A pilot, double-blind, controlled 1-year trial of prednisolone treatment in primary biliary cirrhosis: hepatic improvement but greater bone loss. Hepatology, 1989. 10(4): p. 420-9.
- 79. Kugler, C.F. and W.E. Fleig, [Placebo-controlled double-blind study of cyclosporin A in primary biliary cirrhosis]. Z Gastroenterol, 1991. 29(12): p. 663-4.
- 80. Vuoristo, M., et al., A placebo-controlled trial of primary biliary cirrhosis treatment with colchicine and ursodeoxycholic acid. Gastroenterology, 1995. 108(5): p. 1470-8.
- Combes, B., et al., Prolonged follow-up of patients in the U.S. multicenter trial of ursodeoxycholic acid for primary biliary cirrhosis. Am J Gastroenterol, 2004. 99(2): p. 264-8.
- 82. Bodenheimer, H.C., Jr., et al., A prospective clinical trial of D-penicillamine in the treatment of primary biliary cirrhosis. Hepatology, 1985. 5(6): p. 1139-42.
- Itakura, J., et al., Prospective randomized crossover trial of combination therapy with bezafibrate and UDCA for primary biliary cirrhosis. Hepatol Res, 2004. 29(4): p. 216-222.
- 84. Kaplan, M.M., et al., A prospective trial of colchicine and methotrexate in the treatment of primary biliary cirrhosis. Gastroenterology, 1999. 117(5): p. 1173-80.

- 85. Kaplan, M.M., et al., A prospective trial of colchicine for primary biliary cirrhosis. N Engl J Med, 1986. 315(23): p. 1448-54.
- 86. Matloff, D.S., et al., A prospective trial of D-penicillamine in primary biliary cirrhosis. N Engl J Med, 1982. 306(6): p. 319-26.
- 87. Hadziyannis, S. and E. Hadziyannis, A randomised controlled trial of ursodeoxycholic acid (UDCA) in primary biliary cirrhosis (PBC). Hepatology, 1988. 8.
- 88. Hoogstraten, H.J.F., et al., A Randomized controlled trial evaluating therapy with ursodeoxycholic acid in daily doses of 10 me/kg versus 20 mg/kg in primary biliary cirrhosis. European journal of gastroenterology & hepatology, 1998. 10.
- Kaplan, M.M., et al., A randomized controlled trial of colchicine plus ursodiol versus methotrexate plus ursodiol in primary biliary cirrhosis: ten-year results. Hepatology, 2004. 39(4): p. 915-23.
- 90. Hadziyannis, S.J., E.S. Hadziyannis, and A. Makris, A randomized controlled trial of Ursodeoxyccholic acid (UDCA) in primary biliary cirrhosis (PBC). European journal of clinical investigation 1989. 19.
- 91. Bodenheimer, H., F. Schaffner, and J. Pezzullo, A randomized double-blind controlled trial of colchicine in primary biliary cirrhosis. Hepatology, 1985. 5.
- Poupon, R.E., et al., A randomized trial comparing colchicine and ursodeoxycholic acid combination to ursodeoxycholic acid in primary biliary cirrhosis. UDCA-PBC Study Group. Hepatology, 1996. 24(5): p. 1098-103.
- Poupon, R., et al., A randomized trial comparing the combination ursodeoxycholic acid (UDCA) and colchicine to UDCA alone in primary biliary cirrhosis. Hepatology, 1994.
 20.
- 94. Hoofnagle, J.H., et al., Randomized trial of chlorambucil for primary biliary cirrhosis. Gastroenterology, 1986. 91(6): p. 1327-34.
- 95. Combes, B., et al., A randomized, double-blind, placebo-controlled trial of ursodeoxycholic acid in primary biliary cirrhosis. Hepatology, 1995. 22(3): p. 759-66.
- 96. Epstein, O., et al., Reduction of immune complexes and immunoglobulins induced by D-penicillamine in primary biliary cirrhosis. N Engl J Med, 1979. 300(6): p. 274-8.
- 97. Jorgensen, R., et al., Results of long-term ursodiol treatment for patients with primary biliary cirrhosis. Am J Gastroenterol, 2002. 97(10): p. 2647-50.
- 98. Poupon, R.E., et al., Serum bile acids in primary biliary cirrhosis: effect of ursodeoxycholic acid therapy. Hepatology, 1993. 17(4): p. 599-604.
- 99. Miettinen, T.A., et al., Serum cholestanol, cholesterol precursors, and plant sterols during placebo-controlled treatment of primary biliary cirrhosis with ursodeoxycholic acid or colchicine. Hepatology, 1995. 21(5): p. 1261-8.

- 100. Battezzati, P.M., et al., Ten-year combination treatment with colchicine and ursodeoxycholic acid for primary biliary cirrhosis: a double-blind, placebo-controlled trial on symptomatic patients. Aliment Pharmacol Ther, 2001. 15(9): p. 1427-34.
- 101. Akbar, S.M., et al., Therapeutic efficacy of decreased nitrite production by bezafibrate in patients with primary biliary cirrhosis. J Gastroenterol, 2005. 40(2): p. 157-63.
- Siegel, J.L., et al., Treatment with ursodeoxycholic acid is associated with weight gain in patients with primary biliary cirrhosis. Journal of Clinical Gastroenterology, 2003. 37(2): p. 183-185.
- 103. Dickson, E.R., et al., Trial of penicillamine in advanced primary biliary cirrhosis. N Engl J Med, 1985. 312(16): p. 1011-5.
- 104. Goddard, C.J.R., et al., A trial of ursodeoxycholic acid (UDCA) and colchine in primary biliary cirrhosis (PBC). Hepatology, 1994. 20.
- Schaffner, F., I. Sternlieb, and H. Sachs, A two dose level randomized double blind controlled trial of penicillamine in primary biliary cirrhosis. The first two years. Hepatology, 1982. 2.
- Turner, I.B., et al., A two year controlled trial examining the effectiveness of ursodeoxycholic acid in primary biliary cirrhosis. J Gastroenterol Hepatol, 1994. 9(2): p. 162-8.
- Leuschner, M., et al., Ursodeoxycholic acid (UDCA) and budesonide (BUD) in the treatment of primary biliary cirrhosis (PBC) A prospective double-blind trial Hepatology, 1999. 30.
- 108. Podda, M., et al., Ursodeoxycholic acid (UDCA) for symptomatic primary biliary cirrhosis (PBC): a double-blind multicenter trial. Hepatology, 1989. 10.
- 109. Lim, A., et al., Ursodeoxycholic acid (UDCA) improves hepatic excretion in primary biliary cirrhosis (PBC). Gut, 1994. 35.
- Eriksson, L.S., et al., Ursodeoxycholic acid (UDCA) in patients with primary biliary cirrhosis (PBC) Results of a two-year randomized placebo-controlled study. Scandinavian journal of gastroenterology, 1995. 30.
- 111. Calmus, Y. and R. Poupon, Ursodeoxycholic acid (UDCA) in the treatment of chronic cholestatic diseases. Biochimie, 1991. 73(10): p. 1335-8.
- 112. Lindor, K., et al., Ursodeoxycholic acid (UDCA) is beneficial therapy for patients with primary biliary cirrhosis (PBC). Hepatology, 1992. 16.
- 113. Leuschner, U., et al., Ursodeoxycholic acid (UDCA) treatment of primary biliary cirrhosis: clinical and histological results of a controlled study. Trends in bile acid research Falk symposium 52 1989: p. 355-358.

- 114. Leuschner, U., et al., Ursodeoxycholic acid and budenoside in the treatment of primary biliary cirrhosis. Falk Symposium No 108 XV International Bile Acid Meeting Bile Acids and Cholestasis 1999.
- 115. Copaci, I., L. Micu, and L. Cojocaru, Ursodeoxycholic acid and Methotrexate for primary biliary cirrhosis. Journal of hepatology, 2001. 34(59).
- 116. Leuschner, M., et al., Ursodeoxycholic acid and prednisolone versus ursodeoxycholic acid and placebo in the treatment of early stages of primary biliary cirrhosis. J Hepatol, 1996. 25(1): p. 49-57.
- 117. Lindor, K.D., et al., Ursodeoxycholic acid delays the onset of esophageal varices in primary biliary cirrhosis. Mayo Clin Proc, 1997. 72(12): p. 1137-40.
- Papatheodoridis, G.V., et al., Ursodeoxycholic acid for primary biliary cirrhosis: final results of a 12-year, prospective, randomized, controlled trial. Am J Gastroenterol, 2002. 97(8): p. 2063-70.
- 119. Battezzati, P.M., et al., Ursodeoxycholic acid for symptomatic primary biliary cirrhosis. Preliminary analysis of a double-blind multicenter trial. Italian Multicenter Group for the Study of UDCA in PBC. J Hepatol, 1993. 17(3): p. 332-8.
- Poupon, R.E., E. Eschwege, and R. Poupon, Ursodeoxycholic acid for the treatment of primary biliary cirrhosis. Interim analysis of a double-blind multicentre randomized trial. The UDCA-PBC Study Group. J Hepatol, 1990. 11(1): p. 16-21.
- 121. Batta, A., et al., Ursodeoxycholic acid improves liver function and reduces serum and urinary endogenous bile acids in primary biliary cirrhosis. Hepatology, 1988. 8.
- 122. Leuschner, U., et al., Ursodeoxycholic acid in primary biliary cirrhosis: results of a controlled double-blind trial. Gastroenterology, 1989. 97(5): p. 1268-74.
- 123. Lindor, K.D., et al., Ursodeoxycholic acid in the treatment of primary biliary cirrhosis. Gastroenterology, 1994. 106(5): p. 1284-90.
- 124. Hwang, S.J., et al., Ursodeoxycholic acid in the treatment of primary biliary cirrhosis: a short-term, randomized, double-blind controlled, cross-over study with long-term follow up. J Gastroenterol Hepatol, 1993. 8(3): p. 217-23.
- 125. Borum, M. and H. Fromm, Ursodeoxycholic acid in the treatment of primary biliary cirrhosis: first controlled data. Hepatology, 1990. 12(1): p. 172-3.
- Leuschner, U., H. Fischer, and K. Hubner, Ursodeoxycholic acid in the treatment of primary biliary cirrhosis: results of a controlled study. Zeitschrift fur Gastroenterologie Verhandlungsband 1989. 24: p. 133.
- Leuschner, U., H. Fischer, and K. Hubner, [Ursodeoxycholic acid in the treatment of primary biliary cirrhosis: results of a controlled study]. Z Gastroenterol Verh, 1989. 24: p. 133.

- 128. Combes, B., et al., Ursodeoxycholic acid therapy in patients with primary biliary cirrhosis. Hepatology, 1991. 14.
- Eriksson, L.S., et al., Ursodeoxycholic acid treatment in patients with primary biliary cirrhosis. A Swedish multicentre, double-blind, randomized controlled study. Scand J Gastroenterol, 1997. 32(2): p. 179-86.
- 130. Kisand K.E., K.A.-L., Vuoristo M., Farkkila M., Lehtola J., Inkovaara J., Kisand K.V., Miettinen T., Krohn K., Uibo R., Ursodeoxycholic acid treatment lowers the serum level of antibodies against pyruvate dehydrogenase and influences their inhibitory capacity for the enzyme complex in patients with primary biliary cirrhosis. Journal of Molecular Medicine 1996. 74(5): p. 269-274.
- 131. Eskreis, D., et al., Ursodeoxycholic acid treatment of primary biliary cirrhosis. Gastroenterology, 1988. 83.
- 132. Poupon, R., et al., Ursodeoxycholic therapy for primary biliary cirrhosis: a four year controlled study [AASLD abstract]. Hepatology, 1992. 16.
- Papatheodoridis, G., et al., Ursodeoxycholic-acid for primary biliary cirrhosis: final results of a 12-year prospective, randomised, controlled trial. Journal of hepatology, 2000. 32.
- Poupon, R.E., R. Poupon, and B. Balkau, Ursodiol for the long-term treatment of primary biliary cirrhosis. The UDCA-PBC Study Group. N Engl J Med, 1994. 330(19): p. 1342-7.
- 135. Dickson, E., et al., Ursodiol is effective therapy for patients with primary biliary cirrhosis. Bile acids and the hepatobiliary system From basic science to clinical practice Proceedings of Falk symposium 68th held in Basel, Switzerland, October 12 14, 1992 1993.
- 136. O'Brien, C., et al., Ursodiol treatment of primary biliary cirrhosis [abstract]. Gastroenterology, 1990. 98.

Appendix 2: Publications excluded based on having no OCA treatment arm

- 1. Angulo, P., et al., Comparison of three doses of ursodeoxycholic acid in the treatment of primary biliary cirrhosis: a randomized trial. J Hepatol, 1999. 30(5): p. 830-5.
- 2. Askari, F., et al., Treatment of primary biliary cirrhosis with tetrathiomolybdate: results of a double-blind trial. Transl Res, 2010. 155(3): p. 123-30.
- 3. Buuren, H. and S. Schalm, Beneficial effect of malotilate on primary biliary cirrhosis (PBC): results of a multicentre controlled trial. Journal of hepatology, 1989. 9(S28).
- 4. Crosignani, A., et al., Tauroursodeoxycholic acid for treatment of primary biliary cirrhosis. A dose-response study. Dig Dis Sci, 1996. 41(4): p. 809-15.

- 5. Crosignani, A., et al., Tauroursodeoxycholic and ursodeoxycholic acids for the treatment of primary biliary cirrhosis: a crossover study. Hepatology, 1996. 23(I44).
- 6. Enrico, R., et al., Treatment of early (I-II STAGE) primary biliary cirrhosis (PBC): High versus standard UDCA doses after 15 years follow-up. A randomized open controlled trial. Hepatology, 2011. 54(SUPPL. 1).
- Ferri, F., P. Bernocchi, and S. Fedeli, [Taurodeoxycholic acid in the treatment of primary biliary cirrhosis. A controlled study in comparison to ursodeoxycholic acid]. Clin Ter, 1993. 143(4): p. 321-6.
- 8. Hoogstraten, H., et al., Combined bile acid-immunocuppressive therapy for primary biliary cirrhosis Results of a 1-year multi-centre, placebo controlled trial Hepatology, 1996. 24.
- 9. Hopf, C., et al., Therapeutic Equivalence of Ursodeoxycholic Acid Tablets and Ursodeoxycholic Acid Capsules for the Treatment of Primary Biliary Cirrhosis. Clin Pharmacol Drug Dev, 2013. 2(3): p. 231-6.
- Invernizzi, A., et al., Comparison between tauroursodeoxycholic and ursodeoxycholic acids in patients with primary biliary cirrhosis: a cross over study Hepatology, 1996. 24.
- Larghi, A., et al., Ursodeoxycholic and tauro-ursodeoxycholic acids for the treatment of primary biliary cirrhosis: a pilot crossover study. Aliment Pharmacol Ther, 1997. 11(2): p. 409-14.
- 12. Leuschner, M., et al., The influence of sulindac on patients with primary biliary cirrhosis that responds incompletely to ursodeoxycholic acid: a pilot study. Eur J Gastroenterol Hepatol 2002. 14: p. 1369–1376.
- 13. Liberopoulos, E.N., et al., Fenofibrate in primary biliary cirrhosis: a pilot study. Open Cardiovasc Med J, 2010. 4: p. 120-6.
- 14. Lim, A., et al., Combination ursodeoxycholic acid and immunosuppressive therapy for the treatment of primary biliary cirrhosis [abstract]. Gut, 1995. 37(A27).
- 15. Mason, A.L., MULTI-CENTER, DOUBLE BLIND, RANDOMIZED CONTROLLED TRIAL OF COMBIVIR THERAPY FOR PATIENTS WITH PRIMARY BILIARY CIRRHOSIS (PBC). Canadian Digestive Diseases Week, 2008.
- 16. Mason, A.L., et al., Clinical trial: randomized controlled study of zidovudine and lamivudine for patients with primary biliary cirrhosis stabilized on ursodiol. Aliment Pharmacol Ther, 2008. 28(7): p. 886-94.
- Mason, A.L., et al., Farnesoid-x receptor agonists: A new class of drugs for the treatment of PBC? An international study evaluating the addition of obeticholic acid (INT-747) to ursodeoxycholic acid. Hepatology, 2010. 52(SUPPL. 1).

- McCormick P.A., S.F., Epstein O., Burroughs A.K., Scheuer P.J., McIntyre N., Thalidomide as therapy for primary biliary cirrhosis: A double-blind placebo controlled pilot study. Journal of Hepatology, 1994. 21(4): p. 496-499.
- 19. McMichael, J., et al., Computer-guided randomized concentration-controlled trials of tacrolimus in autoimmunity: multiple sclerosis and primary biliary cirrhosis. Ther Drug Monit, 1996. 18(4): p. 435-7.
- Mitchison, H.C., et al., The results of a randomized double blind controlled trial evaluating malotilate in primary biliary cirrhosis. A European multicentre study group. J Hepatol, 1993. 17(2): p. 227-35.
- Poupon, R.E., R. Poupon, and B. Balkau, Ursodiol for the long-term treatment of primary biliary cirrhosis. The UDCA-PBC Study Group. N Engl J Med, 1994. 330(19): p. 1342-7.
- Rautiainen, H., et al., Budesonide combined with UDCA to improve liver histology in primary biliary cirrhosis: a three-year randomized trial. Hepatology, 2005. 41(4): p. 747-52.
- 23. Ross, A. and S. Sherlock, A trial of azathiorrine in primary biliary cirrhosis. Gut, 1970. 11(12): p. 1058.
- Sato K., H.K., Yamazaki Y., Yanagisawa M., Hashizume H., Horiguchi N., Kakizaki S., Kusano M., Yamada M., Long-term prognosis of combination therapy with ursodeoxycholic acid and bezafibrate for primary biliary cirrhosis: A prospective, multicenter, randomized controlled study. Hepatology International, 2014. 8(1): p. S15.
- 25. Schlichting, J., et al., [Ursodeoxycholic acid and Sulindac in the therapy of primary biliary cirrhosis]. Zeitschrift fur Gastroenterologie 2000. 38.
- 26. Setchell, K., et al., Tauroursodeoxycholic acid (TUDCA) appears more effective than UDCA at displacing hydrophobic bile acids from the bile acid pool of patients with primary biliary cirrhosis (PBC). Hepatology, 1994. 20.
- 27. Smart, H., et al., A double blind placebo controlled study of antioxidants in primary biliary cirrhosis (PBC) [abstract]. Gut, 1990. 31.
- 28. Tang, H., et al., [Efficacy of ursodeoxycholic acid combined with Tongdan Decoction in treatment of patients with primary biliary cirrhosis.]. World Chinese Journal of Digestology 2008. 16.
- 29. Tang, H.H., et al., Efficacy of ursodeoxycholic acid combined with Tongdan Decoction in treatment of patients with primary biliary cirrhosis. World Chinese Journal of Digestology, 2008. 16(13): p. 1417-1424.
- Tong, G.D., et al., Efficacy of ursodeoxycholic acid combined with Tongdan Decoction () on immunological indices and histopathological changes in primary biliary cirrhosis patients. Chin J Integr Med, 2012. 18(1): p. 16-22.

- 31. Van Hoogstraten, H.J.F., et al., A randomized trial in primary biliary cirrhosis comparing ursodeoxycholic acid in daily doses of either 10 mg/kg or 20 mg/kg. Alimentary Pharmacology and Therapeutics, 2988. 12(10): p. 965-971.
- 32. Watson, J.P., et al., Case report: oral antioxidant therapy for the treatment of primary biliary cirrhosis: a pilot study. J Gastroenterol Hepatol, 1999. 14(10): p. 1034-40.
- Wolfhagen, F., et al., A randomized placebo-controlled trial with prednisone/azathioprine in addition to ursodeoxycholic acid in primary biliary cirrhosis. Journal of hepatology, 1994. 21.
- Wolfhagen, F.H., et al., Triple therapy with ursodeoxycholic acid, prednisone and azathioprine in primary biliary cirrhosis: a 1-year randomized, placebo-controlled study. J Hepatol, 1998. 29(5): p. 736-42.
- 35. Wu, Y., D.K. Yao, and L. Zhu, [Clinical observation on the safety and efficacy of ursodeoxycholic acid and fuzheng huayu capsule in the treatment of primary biliary cirrhosis]. Zhongguo Zhong Xi Yi Jie He Za Zhi, 2012. 32(11): p. 1477-82.
- 36. Zifroni, A. and F. Schaffner, Long-term follow-up of patients with primary biliary cirrhosis on colchicine therapy. Hepatology, 1991. 14(6): p. 990-3.

Appendix 3: Publications excluded due to study design/outcome

- 1. Angulo, P., et al., Comparison of three doses of ursodeoxycholic acid in the treatment of primary biliary cirrhosis: a randomized trial. J Hepatol, 1999. 30(5): p. 830-5.
- 2. Berg, C., et al., Pharmacon Davos: A successful mixture from practice and science. Pharmazeutische Zeitung 2001. 146(4): p. 28-44.
- 3. Biggins, S.W., How effective are 3-month and 12-month predictive models of mortality after first liver transplantation? Commentary. Nature Clinical Practice Gastroenterology and Hepatology, 2006. 3(8): p. 430-431.
- 4. Hale, M., J.L. Newton, and D.E. Jones, Fatigue in primary biliary cirrhosis. BMJ, 2012. 345: p. e7004.
- 5. Kankaanpaa, J., Cost-effectiveness of liver transplantation. Transplant Proc, 1987. 19(5): p. 3864-6.
- 6. Kaur, S. and S.J. Cotler, Part IV. Advances in liver transplantation: Overview and status. Disease-a-Month, 1999. 150(183): p. 150-183.
- L., G., Asymptomatic primary biliary cirrhosis. Journal of insurance medicine, 2004. 36(4): p. 341-342.
- Lee, K.H., et al., Liver transplant in Singapore--coming of age. Singapore Med J, 1998. 39(2): p. 49-52.

- 9. Lemos, L.L.P., et al., Costs of primary biliary cirrhosis treatment with ursodeoxycholic acid in Brazil. Value in Health, 2013. 16(7): p. A692.
- 10. Neuberger, J., Is cyclosporin or tacrolimus treatment most beneficial after liver transplantation? Commentary. Nature Clinical Practice Gastroenterology and Hepatology, 2006. 3(11): p. 608-609.
- NIHR. Horizon Screening Research, Obeticholic acid for primary biliary cirrhosis ? second line. 2013 [cited 2014 October]; Available from: http://www.hsric.nihr.ac.uk/topics/obeticholic-acid-for-primary-biliary-cirrhosis-second-line/.

Appendix 4: Publications excluded due to population

- 1. Aberg, F., et al., Cost of a quality-adjusted life year in liver transplantation: he influence of the indication and the Model for End-Stage Liver Disease score. Liver Transplantation, 2011. 17(11).
- Kogure, T., et al., The model for end-stage liver disease score is useful for predicting economic outcomes in adult cases of living donor liver transplantation. Journal of Gastroenterology (2006) 41:10 (1005-1010). Date of Publication: October 2006, 2006. 41(10): p. 1005-1010.

Question B13b - It is not clear how the POISE effectiveness data is already reflected for the UDCA and no treatment comparators. In response to question B6 the company notes that they made assumptions based on POISE, but this is different than actually using the POISE data. In addition, the ERG are concerned that different methods were used to calculate the transition probabilities for different treatments. Using the same methodology for all comparators would ensure that the differences between the treatments considered is not due to the difference in methods.

The ERG strongly request that you provide the results of scenario analyses using transition probabilities based on POISE for all comparators, using the same methodology and assumptions for the UDCA monotherapy and the no treatment comparators as was used for the OCA-based regimens (i.e. using a similar Table for these comparators as CS Table 49 for the OCA-based regimens). The updated model and detail of the adjustments and methodology (including the formulae used to derive transition probabilities) should accompany these results.

Patients who are intolerant to UDCA would not receive active treatment in clinical practice. In POISE, only 5 patients did not receive UDCA at baseline (i.e. they were intolerant and not treated with UDCA) and none responded positively. Therefore assuming that patients intolerant to UDCA would have the same response and progression as the UDCA inadequate responders (who are still gaining benefit from treatment) was not considered appropriate as it would not reflect an accurate history and risk of progression over time. We would overstate the outcomes for this population. After discussion with key opinion leaders, and because this is an extremely small patient population, we based our transition probabilities on the literature in this population (Corpechot, 2000, J Hepatology, V32, 6, 1196-1199)

For patients with inadequate response to UDCA, patients with low ALP values at baseline were more likely to achieve the 15% criteria and fall below 1.67 threshold compared to patients with higher values; this is driven by natural variation in ALP and the lower baseline value. This outcome potentially overestimates the benefit of UDCA in patients who were close to the 1.67 threshold in the POISE trial. To account for this natural variation and expected risk of progression in this population, we considered in our base case, transition probabilities that would reflect the natural history expected for this specific population, based on the 10-year liver transplant free survival rates estimated on POISE UDCA patients.

Regarding the additional analysis requested as clarification to question B13b:

• The ERG strongly request that you provide the results of scenario analyses using transition probabilities based on POISE for all comparators, using the same methodology and assumptions for the UDCA monotherapy and the no-treatment comparators as was used for the OCA-based regimens (I.e. using a similar table for these comparators as CS table 49 for the OCA-based regiments). The updated model and detail of the adjustments and methodology (including the formulae used to derive transition probabilities) should accompany these results.

Please note: the following results were generated using updated patient-level data from POISE. The previous version of the model used an incorrect patient-level dataset from an earlier version of the POISE CSR. The updated patient numbers used in the model are now reflective of the CSR.

Methodology

The ability to use patient-level UDCA data from POISE has been added into the model. While the majority of changes to the model were made in the 'Transition matrices' worksheet, some of the formulas were also updated to reflect this new feature in the sheets 'Clinical inputs' and 'UDCA Markov Model'. All additional transitions to use for the first 4 cycles were added in the range L316:HL371 on this worksheet and the transition matrices to use after the first year in the range X9:AP27. A switch to enable or disable the use of UDCA patient-level data has been added in cell C10 on the 'Clinical inputs' worksheet. If the switch is activated, patient-level UDCA data from POISE is used to generate transition probabilities for that population.

The new patient-level population numbers (added in the cells highlighted in yellow on the 'Transition matrices' worksheet) are displayed along with their transition probability calculations (presented alongside). The calculations for how transition probabilities are derived are present as formulae within the transition probability cells; for example, the patient numbers for UDCA patients with a recorded ALP threshold of > 1.67 * ULN (200 u/L) are located in the range X330:AF338. The transition probability calculations corresponding to this set of patient numbers are located in the range M330:V330. This method was used to derive all patient-level UDCA transition probability data for the first four cycles of the model.

Results

Use of the patient-level UDCA transition data produced the following results:

CONFIDENTIAL

Table 1: Deterministic results for UDCA-intolerant patients, using the list price of OCA and UDCA patient-level data

Technologies		Total			Incremental		ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment (Placebo)	£103,233	11.30	6.61					
OCA titration		16.68	13.56		5.38	6.95		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Table 2: Base-case results for UDCA inadequate responders, using the list price of OCA, and UDCA patient-level data

Technologies		Total			Incremental		ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA+Placebo	£92,218	12.72	8.30					
OCA+UDCA Titration		16.78	13.68		4.06	5.38		

Abbreviations: ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

As no UDCA is considered for the UDCA-intolerant population, the results in Table 1 are identical to the updated base case model, i.e. corrected ICER of (submitted alongside this document). Use of the patient-level UDCA transition data for the inadequate response patient population resulted in an ICER of (which is greater than the corrected base case ICER of (submitted alongside).

Question B12 section 3) Please clarify why the number of patients in Table 13 add up to 68, while the number of patients in the trial arm is 70

The answer to the difference in the numbers is patient discontinuation in the trial. 71 patients were randomized into the 5mg titration arm, 69 patients completed to month 6, and 64 patients completed the 12 months evaluation phase (defined as the Completer Population).

The table below (from our submission) illustrates the different analysis populations:

		Number of su	ıbjects, n (%)	
	Placebo	OCA titration	OCA 10 mg fixed dose	Total
Enrolled/randomised	73 (100)	71 (100)	73 (100)	217 (100)
ITT population§	73 (100)	70† (99)	73 (100)	216 (<100)
Completer population [¶]	70 (96)	64 (90)	64 (88)	198 (91)
EE population [‡]	67 (92)	63 (89)	62 (85)	192 (88)
PK population ^{††}	0 (0)	66 (93)	60 (82)	126 (58)
Safety population§§	73 (100)	70 (99)	73 (100)	216 (<100)

Table 3: Analysis populations

Abbreviations EE, efficacy evaluable; ITT, intention-to-treat; OCA, obeticholic acid; PK, pharmacokinetics. [†]There was one subject in the OCA titration group who withdrew after randomisation, leaving 70 subjects in the ITT group; [§]All randomised subjects who received at least one dose of investigational product. Treatment assignment is based on the randomised treatment; [¶]All randomised subjects who received at least one dose of investigational product and participated through the end of the double-blind phase (12 months). Treatment assignment is based on the randomised treatment; [‡]All subjects in the completer population who did not have any major protocol deviations that could have potentially affected the efficacy of the investigational product. Treatment assignment is based on the randomised treatment; ^{††}All randomised subjects who received at least one dose of OCA who have at least one confirmed fasting sample at Month 6 and Month 12 visits (subjects must have been fasting for approximately 8 hours prior to the visit) and who did not have any major protocol deviations that could have potentially affected exposure levels; ^{§§}All subjects who received at least one dose of study drug. Treatment assignment is based on the treatment actually received.

The CONSORT diagram in figure 11 of our submission shows that 7 patients out of the 71 patients in the randomized population in the 5mg titration arm discontinued.

The reasons for discontinuation were (n):

Clinical / laboratory adverse event (3), withdrew consent (2), pruritus (1), death (1).

The difference in the numbers of patients reporting in each quarter is accounted for by discontinuation from the trial. Neither the death nor the adverse events were considered to be associated with the trial treatment. Pruritus as a known symptom of PBC and an established side effect of obeticholic acid.

Question B4b – Lack of data/evidence is not a clinical rationale for combining abnormal bilirubin (TB>20umol/L) and compensated cirrhosis into one health state. Please provide a clinical justification why combining abnormal bilirubin (TB>20umol/L) and compensated cirrhosis into one health state is appropriate. Why does the company believe that abnormal bilirubin (TB>20umol/L) can be assumed to be equivalent to having compensated cirrhosis?

We have not combined the two groups into one health state because of a lack of evidence, but because the outcomes from these groups are clinically similar. PBC is a condition driven by progressive loss of bile ducts (ductopenia) leading to mechanical failure in the transport of bile from the liver (cholestasis), which then leads to the death (or senescence) of biliary epithelial cells. The condition is functionally different from other liver conditions such as HepC. As a result, fibrosis in itself is not considered to be the key clinical measure – whilst it is present in the later stages of the condition, the underlying ductopenia means that fibrosis progression is not a good measure of severity or response to treatment.

PBC therapeutic trials are challenging to conduct because of the slow course of the disease. The choice of selection criteria and primary endpoints is of critical importance. Because most patients are currently diagnosed and treated at an early stage, the use of hard endpoints, such as the occurrence of death or liver transplantation (LT), alone appears unsuitable as this can take up to 10 years from diagnosis. The use of additional endpoints, such as the occurrence of cirrhosis or of extensive fibrosis, seems undoubtedly more relevant in this condition, however more towards the end stages of the condition. The necessity to perform repeated liver biopsies remains a serious limitation to the utilization of such endpoints. *Liver biopsies are not routinely done in clinical practice and the soon to be published UK PBC guideline recommends that 'liver biopsy is not usually required in the diagnosis of PBC or for monitoring of disease progression out with clinical trial'.*

The previously referred to UK guideline recommends that "whilst there are a range of noninvasive tools to stage and monitor disease progression. There is no consensus as to what is the optimal strategy". There is a continued need for developing non-invasive methods of liver fibrosis assessment and monitoring. In this regard, transient elastography (TE) appears as one of the most promising methods. We have used TE in the POISE trial, but not all units have established this technology in clinical practice. As a result we only have a subgroup of patients who have been reviewed at baseline and at 12 months. The results have been included in the main submission.

The role of raised ALP and bilirubin is well established in the diagnosis and management of PBC (Corpechot) and the global support group (Lammers) has demonstrated that controlling the levels of both biochemical indicators improves outcomes. As a result, raised ALP and bilirubin are used to diagnose the condition and response to therapy is measured against reduction of ALP and bilirubin. The graph below demonstrates how abnormal bilirubin significantly reduces progression free survival, even where ALP levels are below 2X ULN. The Bowlus paper indicates

from a further analysis of the Global PBC study group that the risk starts to increase at 0.5xULN, i.e. even within the normal range.



Abnormal bilirubin is a lagging indicator of progressive liver deterioration in PBC. The Harms publication shows that bilirubin at 1.6x ULN will result in significant events with median time of 19 months to a major endpoint. As a result, PBC patients with abnormal bilirubin will have a similarly increased risk of an event to patients in the compensated cirrhosis state.

In summary, abnormal bilirubin is a clear indicator of progression with similar risk of events as patients in the compensated cirrhosis state, which is why, following guidance from the principal investigator in the UK PBC Consortium, we have combined the two states in the model.

Bowlus,	Hepatic Medicine: Evidence and Research 2016:8 89-95
Lammers,	Journal of hepatology. 2013;58(S388)
BSG	UK PBC draft PBC guidelines
Corpechot,	Hepatology. 2008 Sep;48(3):871-7.
Harms	Poster SAT-351 presented at EASL 2016

Remaining response to ERG clarification questions

We would like to bring to the ERG's attention that there was an error in the original submission regarding patient numbers used in the economic model.

The original submitted model contained data from an analysis of patient numbers according to an early dataset version from POISE. A QC has since been performed on the patient numbers by the Intercept bio stats team and updated data provided.

The model has now been updated to include the correct patient numbers. These have the effect of changing some transition probabilities used in the first four cycles of the economic model. This has had a downstream effect of changing the QALYs and costs generated for the OCA treated patient groups.

The effect of this change will be outlined in detail in a forthcoming erratum. The erratum will contain a full deterministic and probabilistic analysis of the new base case, including updated versions of the scenario analyses provided in the original submission. The overall effect on the ICERs for both the intolerant and inadequate response patient populations is minimal (deterministic ICERs change by <£500).

We would like to apologise for this error and the inconvenience caused. We would also like to thank the ERG for spotting the issue with the patient numbers, and requesting further clarification in their question B12b.

The model submitted with this response is the corrected version of the model and we have used this version to respond to the ERG queries in this document.

Regarding the response to clarification questions B12b:

a. Please explain how the numbers in Table 13 of the response are used in the economic model. Please provide cell references to the transition probabilities that are used in the Markov trace – for example what are the cell references for 0.343 and 0.558 (reported in Table 13)?

We apologise for the lack of clarity in our previous answer. The data shown in Table 13 of the response are the entire ITT patient population from POISE for the OCA titration arm of the trial. The actual patient numbers used in the model were lower. This is because a given patient could only be used in transition matrices if we could track their progress through the different health states based on ALP and bilirubin every 3 months, i.e. :

- 1. Patients had their ALP and bilirubin levels recorded at their previous appointment (i.e. baseline, 3 months, 6 and 9 months)
- 2. Both ALP and bilirubin levels were recorded at the subsequent visit (i.e. 3, 6, 9 and 12 months respectively).

If a patient missed an appointment, or had only ALP or bilirubin levels recorded, they could not be used to generate transition probabilities, as the biochemistry state that the patient was previously in could not be determined. Patients were removed from the analysis if they did not meet the criteria for inclusion.

The transition matrices containing transition probabilities, and the patient numbers used to generate these probabilities, are located on the worksheet 'Transition matrices' in the accompanying updated model.

An example of a range containing an OCA-specific patient number matrix (on the 'transition matrices' sheet) is **a second second**. Its corresponding transition probability matrix is in **a second**. The formulae for generating these transition probabilities are present in this range. Patient number values that have been updated from the previous base case version of the model have been highlighted in yellow.

The transition probabilities are used in each of the Markov engines for the involved comparators. These vary depending on the population selected (UDCA intolerant versus UDCA tolerant). The proportions of patients in each model state are tabulated on the 'Markov trace' worksheet. This table is then drawn visually as a Markov trace, which can be seen on the 'Clinical inputs' worksheet.

Due to the transition probabilities changing over time, the Markov trace is composed of a number of transition matrices, which are labelled in the 'Transition matrices' worksheet.

The specific cells which the Markov trace is referencing can be determined by selecting a data series on the Markov trace. In the formula bar, the range reference for that data series on the 'Markov trace' worksheet will be present.

b. How is the discontinuation in Table 13 calculated (i.e. how did the company derive the 0.099)?

The overall discontinuation rates for the trial were taken from patient-level data from POISE. All patients are assumed to discontinue within the first three months of treatment. The discontinuation rates used are shown in cells C23, C24 and C25 on the 'Clinical inputs' worksheet.

Reply to question received from NICE 12.22pm on the 18th of November 2016

In the sheet "UDCA Markov model", the information presented in the sheet "Transition matrices " range "L329:HL339" is reflected in the in the transitions displayed rows 54:61 of the sheet "transition matrices. When you switch, the data used in the UDCA markov sheet changes in the range "BK7:BN10" taking the patient-level data from the POISE UDCA arm into account

It is correct that the transitions in C10:T27 and Y10:AP27 are similar. This is due to the fact that we still allowed the patients to be able to progress to abnormal bilirubin during the first year. This can be removed by changing as follows the probabilities in cell F10:F11 as follows:

- Before: =IF('Model parameters'!\$C\$145="yes",'Model parameters'!\$C150,0)
- New: =IF(UDCAPL="No", 'Model parameters'!\$C150,0) and =IF(UDCAPL="No", 'Model parameters'!\$C155,0) respectively

We have uploaded an updated model containing this specific feature

Question:

Dear Gordon

There seems to be an error in the model submitted with regards to the requested scenario analysis or the description in the accompanying document is unclear/incorrect.

To clarify this concern the ERG state:

- The UDCA Transition matrix is located in 'Transition matrices'!C\$10:T\$27; i.e. this is the matrix referred to in cells 'UDCA Markov model'!AR7:AU10
- According to the company's accompanying document, the transition matrices for this scenario analysis are located in 'Transition matrices'!L316:HL371; I think we can narrow this down to 'Transition matrices'!L329:HL339 for the company base-case
- If we would change the 'switch' variable indicated by the company to 'yes' ('Clinical inputs'!C10), the transition matrices in 'Transition matrices'!L329:HL339 are not reflected in 'Transition matrices'!C\$10:T\$27 and do not seem to be used in the for the UDCA Markov trace (in the 'UDCA Markov model' work sheet). Moreover, the transition matrix used in the UDCA Markov trace doesn't seem to differ between cycles 1-4 whereas those in 'Transition matrices'!L329:HL339 differ for cycles 1-4 (as I would expect as this is consistent with how it is done for the OCA-regimens).

Please can you clarify the above and if needed send us an updated/corrected version of the model.

Kind Regards

Technology Appraisals Administrator - Committee A

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Thank you for agreeing to make a submission on your organisation's view of the technology and the way it should be used in the NHS.

Healthcare professionals can provide a unique perspective on the technology within the context of current clinical practice which is not typically available from the published literature.

To help you in making your submission, we have provided a template. The questions are there as prompts to guide you. It is not essential that you answer all of them.

Please do not exceed the 8-page limit.

Your name:

Name of your organisation: BASL
I am a specialist in the treatment of people with primary biliary cirrhosis.
I am also involved closely with UK-PBC, an MRC funded nationwide

I am also involved closely with UK-PBC, an MRC funded nationwide study group that has determined the natural history of PBC with careful genotyping and phenotyping. This study has emphasised the value of the serum alkaline phosphatase as a biomarker for clinical outcome. The early clinical response of the alkaline phosphatase to ursodeoxycholic acid therapy has allowed separation of patients into distinct clinical groups with eventual high and low mortality, but at an early point in the clinical course. Thus introducing a surrogate marker of efficacy in this condition, avoiding long-term mortality based studies.

I am also the (unpaid) president of the British Association for the study of the liver (BASL) an organisation that represents clinicians treating PBC.

I have no conflict of interest and do not accept any personal funding or personal support from the pharmaceutical industry.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

What is the expected place of the technology in current practice?

The long natural clinical course for patients with PBC, sometimes decades, has made trials of therapy in PBC difficult to conduct if based on mortality or the onset of liver failure. Ursodeoxycholic acid has been used predominantly to manage pruritus and the role of this agent as a disease modifier in the treatment of PBC was not accepted universally. However recent data based on both retrospective and prospective evidence in large groups has emphasised that patients can be stratified by the response to this agent (by a reduction in the alkaline phosphatase) into those with an excellent clinical outcome and those in whom the long-term prognosis is poor. The majority of patients in the UK and elsewhere are now being offered treatment with Ursodeoxycholic acid, encouraged by patient support groups. Recent data has encouraged physicians to use larger doses nearer that recommended with benefit in many patients. This agent is well tolerated. A number of other agents are offered to manage pruritus, a major issue in PBC, with variable success and without evidence that these modify the clinical course of disease or liver biochemistry. Thus, Ursodeoxycholic acid is inexpensive, effective and safe with a low incidence of side effects, but there is a distinct cohort, identified by an unchanged alkaline phosphatase with therapy, that do not respond to treatment.

There are only two prospective randomised controlled trials comparing Obeticholic acid (OBCA) with placebo in patients unresponsive to, or intolerant of Ursodeoxycholic acid and in both studies those on active therapy had a biochemical response with a fall in the alkaline phosphatase, the best early marker we have currently POISE study, NEJM, 2016 and Hirschfield et al Gastroenterology 2015). Based on the data from natural history studies it can be inferred that OBCA is a disease modifier but that cannot be regarded as certain while outcome data such as liver failure, death or a need for liver transplantation are years from accrual. The agent is not just 'another bile acid' but has broad biological activity based on FXR agonism, which might explain effects in those unresponsive to Ursodeoxycholic acid. However side effects are much more prominent than with Ursodeoxycholic acid and to an extent appear to be dose related.

There is no clear view on where such patients should be managed. However, it would be pragmatic if patients with a good long-term prognosis identified by the alkaline phosphatase and the response to Ursodeoxycholic acid were managed in secondary care (or in some cases primary care with clear guidance). Alternatively those with a poor outcome might be better managed in tertiary care centres.

Guidelines on managing PBC (via the BSG) are close to publication and are likely to be based on similar recommendations.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

The advantages and disadvantages of the technology

The side effect profile is well described in the two randomised controlled clinical trials. The earlier study identified pruritus as an issue with this agent, which appears to be dose related. There is nothing in that profile to cause particular anxiety. The role of OBCA in de novo therapy for PBC is uncertain. However, many of those using OBCA have found pruritus to be difficult to manage and this may prove to be be an issue in 'real life' clinical practice.

Any additional sources of evidence

Implementation issues

I cannot se that implementation would be problematic for hepatology services. The patients are seen in clinic regularly in any case and many are enrolled in the national study. One model for ensuring responsible use would be to adopt the ODNs delivering antiviral therapy currently for patients with hepatitis C.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Equality I cannot see that this will be an issue.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Thank you for agreeing to make a submission on your organisation's view of the technology and the way it should be used in the NHS.

Healthcare professionals can provide a unique perspective on the technology within the context of current clinical practice which is not typically available from the published literature.

To help you in making your submission, we have provided a template. The questions are there as prompts to guide you. It is not essential that you answer all of them.

Please do not exceed the 8-page limit.

About you
Your name:
Name of your organisation: British Society of Gastroenterology/Royal College of Physicians/University of Birmingham
Are you (tick all that apply):
 a specialist in the treatment of people with the condition for which NICE is considering this technology? <u>YES</u>
 a specialist in the clinical evidence base that is to support the technology (e.g. involved in clinical trials for the technology)? <u>YES</u>
 an employee of a healthcare professional organisation that represents clinicians treating the condition for which NICE is considering the technology? If so, what is your position in the organisation where appropriate (e.g. policy officer, trustee, member etc.)? <u>NO</u>
 other? (please specify) <u>Work Strand Lead for UK-PBC (User Interface;</u> <u>www.uk-pbc.com)</u>
Links with, or funding from the tobacco industry - please declare any direct or indirect links to, and receipt of funding from the tobacco industry: <u>NONE</u>

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

What is the expected place of the technology in current practice?

Primary biliary cirrhosis (PBC; now referred to as primary biliary cholangitis) is a chronic autoimmune liver disease. Diagnosis is generally straight forward and care in the UK is overwhelmingly delivered by the NHS with the involvement of Gastroenterologists, Hepatologists and Primary Care. There are only a few specialist centres and most patients are looked after by Gastroenterologists. There is relative uniformity in practice, and estimated to be perhaps 15-20,000 patients in the UK: 1 in 1000 women over the age of 40 live with PBC. The current UK-PBC flow chart for management (Figure) is shown below and a) current recommended management is with oral Ursodeoxycholic acid (UDCA) at a dose of 13-15mg/kg/day. This is in keeping with UK (BSG/UK-PBC) guidance under review, European (EASL) and North American (AASLD) professional guidance. It is believed that the majority of patients with PBC diagnosed in the UK are now offered this therapy (pending the licencing of OCA in Europe, this is the only licenced therapy for patients with PBC). Efforts continue to optimise awareness that weight based dosing is recommended. The majority of patients tolerate therapy: UDCA intolerance is recognised in 5-10% of patients; b) There is no true geographical variation in practice: it is however recognised generally that there are variations in Hepatology services across the NHS and variation as a result of the presence of dedicated liver services, which have until late remained focused in large cities and around transplant programmes. This is being addressed over time by efforts from a number of organisations spanning professional and patient groups and increasingly every 'DGH' has a Hepatologist; c) There is consensus now that risk stratification is relevant to clinical practice and that biochemical indices are appropriate tools to stratify patients to low or high risk of disease progression; there are a number of biochemical markers that have been robustly identified and validated as appropriate surrogates of PBC outcome and there is recognition that there are strengths and weaknesses associated with individual scoring approaches (some are dichotomous others are continuous in nature). Furthermore there has been a consensus in Industry practice such that second line disease modifying clinical trials have sought patient inclusion based on a dichotomous stratifier of in essence an ALP of >1.67 x ULN after 1 year of UDCA therapy (OR elevated bilirubin values consistent with later stage disease). It is however also accepted that there is a log-linear association between ALP values and future risk of deleterious outcomes i.e. dichotomous stratification is good but response to a new therapy has potential to benefit more than just those who achieve a change in a dichotomous response criteria; d) There is no consensus as regards second line therapy for patients failing UDCA. In the UK it is not widespread practice to offer unlicensed therapies such as fibrates, methotrexate, colchicine, Budesonide, ciclosporin. Only occasional patients are offered such therapy and UK-PBC data suggests this is not widespread or adopted by the majority. For example UK-PBC data (personal communication Dr G Mells, University of Cambridge) shows that in a recent updated survey of a significant proportion of the overall cohort, only 58 patients out of 2245, had ever used a fibrate. Additionally it was not possible to ascertain if this was for hyperlipidaemia or PBC. It can thus be fairly concluded that it is not UK practice to offer any specific therapy for patients who do not respond to UDCA or who are intolerant. In this setting of non-response current practice is to either enrol into formal clinical trials, or refer to an expert centre. To that end it is notable that every NHS trust is represented in UK-PBC, a marker of interest in the UK delivering evidence based care not anecdote based care. Whilst an advantage to off-label drug use has been availability without restriction, significant disadvantages have limited this approach appropriately: namely a lack of evidence of benefit from randomised trials, a reluctance to use off-label therapy and a concern about toxicity that spans liver injury (e.g. fibrates are overtly contraindicated by the manufacturer in liver disease/PBC and hepatotoxicity is clearly documented) as well as infection risk, metabolic risk and renal/haematologic toxicity. BSG and UK-PBC guidelines therefore suggest that in

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

those patients stratified to high risk based on UDCA treatment failure that second line licenced therapy should be sought if available or alternatively patients offered access to approved clinical trials.



Subgroups:

This is addressed really by the concept of risk stratification by risk scores. This is a new technology designed as second line to UDCA because the concept is clear that there are patients treated with UDCA whose response to UDCA predicts a very low risk of progressive disease: these responders do not need new therapy. The 20-30% of patients who don't respond adequately to UDCA are the group this technology has the greatest impact on. Predictors of non-response include age at diagnosis, stage of disease at presentation, bilirubin values, and degree of ductopenia (if biopsied). Male patients may also be more likely to be UDCA non-responders. It is of note that whilst risk is relevant to patients of all ages with PBC, the UK-PBC database has shown that patients under the age of 50 have a 50% chance of being a non-responder. Data from UK Transplant has also suggested that despite widespread UDCA use, the age at transplant in the UK is unchanged over the last 20 years.

As with any liver disease it is unclear whether patients in the very late stages of liver disease with overt decompensation/indication for liver transplant would benefit from addition of second line therapy. Given the side effect profile of OCA (pruritus) it is also important to note that very symptomatic patients will need close attention as tolerability in this small subgroup is relevant, and will need a personalised approach.

Technology setting:

This new technology (Obeticholic acid) is an oral therapy that can be readily delivered to patients. However it is recommended that PBC stratified as high risk is managed in expert centres. It is envisaged that this encompasses Hospital based Hepatologists (or Gastroenterologists with an interest in Hepatology) who are supported by local and regional established Hepatology networks. There is a clear track record of delivering new therapies in

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

the Hepatology space (cf. Hepatitis C therapy) and evidence that the NHS is well placed to do so in a way that captures need and reports efficacy.

Technology availability presently:

This technology is presently only available as part of approved clinical trials. To my knowledge as yet no NHS patient has received Obeticholic acid outside of approved clinical trials.

Relevant clinical guidelines:

The BSG and UK-PBC have prepared, according to BSG/NICE guidance, treatment guidelines for the management of PBC. These are presently under review by the BSG and have been widely circulated for comment across multiple stakeholders. They have been developed under the auspices of a Cholestasis Guideline Development group according to BSG guidance, and with the intent of meeting NICE guidance on treatment guidelines. They have not evaluated non-licenced therapy. The methodology used has been appropriate and is fit for purpose. It should be noted that the BSG/UK-PBC guidelines (now under BSG review following formal drafting) will have the following recommendations relevant to evaluation of this new technology:

RECOMMENDATION 11: All patients with PBC should be offered structured lifelong followup, recognising that different patients have different disease courses. For most patients secondary care will manage disease (EL 4, strong).

RECOMMENDATION 12: Ursodeoxycholic acid at 13-15mg/kg/day is recommended as the first-line pharmaco-therapy in all patients with PBC. If tolerated treatment should be lifelong. (EL 1-, strong)

RECOMMENDATION 13: Individualised risk stratification using biochemical response indices is recommended following one year of UDCA therapy. (EL 2++, Strong)

RECOMMENDATION 14: Inadequate response to UDCA is robustly associated with a greater likelihood of future liver transplantation, development of hepatocellular carcinoma and death. For those patients, clinicians should consider discussion with a liver disease specialist and the opportunity for clinical trial participation, in the absence of access to licenced second line therapies. It is of note that current clinical trials focused on disease modification have used an inclusion criteria that stratifies based on ALP and Bilirubin (e.g. an ALP value of at least 1.67 times the upper limit of the normal range or an abnormal total bilirubin) (EL 4, strong)

RECOMMENDATION 15: All patients should be evaluated for the presence of symptoms, particularly fatigue and itch. Clinicians should recognise that severity of symptoms does not correlate with stage of disease. (EL 4, strong)

RECOMMENDATION 16: Cholestyramine is first-line therapy for pruritus and should be taken separately to UDCA to avoid interaction. (EL 2+, strong)

RECOMMENDATION 17: Rifampicin is the recommended second-line therapy for pruritus. (EL 1-, strong)

The advantages and disadvantages of the technology

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Alternatives:

At present it is our opinion that there are no established alternative available therapies for patients failing UDCA therapy. This is supported by the practice of the major centres involved in UK-PBC, as well as the national level data from UK-PBC demonstrating very low uptake for non-licenced therapies. Therefore this new therapy will be an additional opportunity for patient benefit. Its use will not be overtly burdensome because it is a once daily oral therapy. However, as above, expert treaters will be appropriate, and liver disease management in the UK is appropriately engaged and set up to deliver this (e.g. successful implementation of Hepatitis C therapy across the NHS). There will be requirements on treatment to confirm benefit (a fall in ALP) as well as monitor for side effects (particularly in the rare population with decompensated liver disease), including pruritus. The management of pruritus will be according to standard practice but will need to be proactive. However the agents available for pruritus management are readily accessible and low cost (Cholestyramine and Rifampicin). The expectation is that Obeticholic acid will be delivered through established ambulatory Hepatology practices and the NHS will be well placed to safely implement this therapy without significant added cost beyond the drug itself.

Treatment rules:

The important underlying principal in developing new therapies for patients with PBC is the concept of UDCA treatment failure. It has become clear, and has been validated and extended by numerous academic efforts, that biochemical response in patients with PBC i) is associated with clinical outcomes; ii) there is a log-linear association between ALP values as relates to future clinical outcomes; iii) utilising risk stratification is an appropriate way to presently design clinical trials for second line therapy in PBC; iv) in the context of a rare and orphan disease with a slow clinical course, biochemical surrogates are reasonably likely to predict clinical outcome and benefits when evaluating new therapies; v) true clinical outcome studies of new therapies are desirable but very difficult to conduct. In this context clinical practice (see above) now recommends risk stratification using simple laboratory indices that are measured in routine clinical practice (i.e. there is no additional cost). With a variety of biochemical indices of high risk, there is no overwhelming evidence for one tool over another. Consensus has however rested on a criteria based in essence on ALP values >1.67xULN and/or bilirubin above ULN (in clinical trials often there is an upper limit added which encompasses safety parameters for trial conduct, more so than need) for trial inclusion and therefore this is likely to be the most appropriate rule for consideration of future new therapies in clinical practice. This applies to this new technology and it is the expectation that patients with ALP >1.67 x ULN and/or an elevated Bilirubin will be considered for therapy.

Effective response to therapy is based on biochemical indices as well. For the purposes of clinical trials this has been broadly defined as a fall in ALP alongside stabilisation of Bilirubin. In the context of this new agent the Phase 3 endpoint agreed after academic consultation, was that of ALP <1.67xULN, Normal Bilirubin, and fall in ALP of at least 15%. Such categorical response criteria are most valid in clinical trial settings, and it needs to be noted that there is a log-linear association between ALP values and outcome in particular, hence categorical classification of response under represents the true potential benefit to patients of therapy over prolonged use. Therefore at the present time it is clinical consensus that for new second line therapies categorical response criteria are not rigidly appropriate and broader definitions of clinical response based on overall evaluation of the patient and laboratory indices are more appropriate in the clinic e.g. ALP values fall by at least 15% persistently.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Crite	eria for Biochemical Response to UDCA	
Barcelona	ALP ≤1 x ULN or decreased by ≥40%	At 1 year
Paris I	ALP ${\leq}3$ x ULN and AST ${\leq}2$ x ULN and bilirubin ${<}1$ mg/dL	At 1 year
Rotterdam	Normalisation of abnormal bilirubin and/or albumin	At 1 year
Toronto	ALP <1.67 x ULN	At 2 years
Paris II	ALP ${\leq}1.5$ x ULN and AST ${\leq}1.5$ x ULN and bilirubin ${<}1$ mg/dL	At 1 year
Beijing	Barcelona, Paris I, or Toronto criteria	At 6 months
Biochemical + APRI	Biochemical response (Barcelona, Paris I/II, or Toronto) and APRI ≤0.54	At 1 year
Globe Score	Prognostic index: baseline age, and bilirubin, ALP, albumin, and platelet count	At 1 year
UK-PBC Risk Score	Prognostic index: baseline albumin and platelet count, plus bilirubin, ALT or AST, and ALP	At 1 year

Evidence base:

For Obeticholic acid the clinical trials have been reflective of broad clinical practice relevant to the NHS. The trials were multi-centre, international, placebo controlled and included UK centres; trials included open label extensions mirroring standard clinical practice. Care was delivered in ambulatory settings in keeping with current care offered for patients with PBC. There is ready extrapolation to UK practice. The most important outcomes from the studies were a) consistent effects in improving the surrogate marker of outcome in PBC, namely ALP; b) demonstrable evidence of stabilisation of bilirubin values (noting that bilirubin values in the normal range do still predict outcomes in large studies and that in high risk disease it is evident that you do see slow rises in bilirubin over time); c) consistent improvement in other markers of liver injury alongside ALP- ALT, AST, GGT; d) improvements in IgM concentrations; e) falls in markers of inflammation e.g. CRP; f) overall meeting of primary endpoints.

Surrogates were used in this trial and published data strongly supports that surrogates markers of liver injury (ALP, Bilirubin in particular) are reasonably likely to predict clinical benefit. It is quite clear that there are caveats to this argument that are well rehearsed, but the use and utility of surrogates needs to be recognised in the context of the disease and its current clinical course, and ongoing rare nature. It is fair to state that clinicians accept the use of these surrogates as a route to demonstrating early markers of benefit, and with the expectation that true outcome data will take considerable time to establish.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]



is a FXR agonist (bile acid derived). It is a modification of chenodeoxycholic acid. In clinical practice it is already recognised that UDCA (ursodeoxycholic acid) can precipitate pruritus in around 1% of patients, presumably because of changes in bile acid compositions. Similarly it was not a surprise that pruritus was a clear side effect of both the Phase 2 and Phase 3 trials of OCA in patients with PBC. It would appear to be dose dependent and will be an important component of patient counselling, particularly as symptoms are of importance to patients with PBC. In this regard the relevant lessons learnt from the Phase 3 clinical trial were that i) whilst pruritus was a concern the high retention of patients into the open label long term safety extension suggests high acceptance of the therapy by patients; ii) dose titration, starting at 5mg and not going above 10mg of OCA was an effective means of mitigating pruritus and significantly reducing its impact; iii) pruritus appears to settle with ongoing use; iv) standard of care interventions such as cholestyramine and rifampicin were of utility; v) dose reduction of OCA was a final approach. It must be acknowledged that the Phase 3 patients may have been biased against those with pre-existing itch to some degree, and it is therefore an expectation as OCA is used that clinician and patient education addresses potential pruritus and its management.

Hyperlipidaemia: this is a common pre-existing feature of the cholestasis associated with PBC, wherein despite hyperlipidaemia there is no evidence that patients with PBC have elevated cardiovascular risk. This is a reflection most likely of the nature of hyperlipidaemia in cholestasis wherein unusual lipoprotein particles, such as lipoprotein X, may accumulate and levels of HDL cholesterol are typically elevated. FXR agonists are predicted to have complex effects on lipids and it is not surprising that in patients with PBC given OCA there was a slight increase in LDL and an overall small fall in HDL. There were no reported excess cardiac events in the trials to date, and it is probable that the changes reported for lipid results on treatment with OCA do not change cardiac risk. It remains an area for long term monitoring but no intervention is needed. It should also be noted that for those with risk factors beyond cholesterol values, that statins are safe in patients with PBC, including those on OCA.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Advanced liver disease: the phase 3 trials included some patients with Child Pugh A cirrhosis and in those included there were no clinical concerns. Response seemed equal. It is however noted that Child Pugh B and C patients were not studied and going forward additional information will be needed to ensure safe use in these patients with high risk, late stage liver disease.

Any additional sources of evidence

Additional evidence:

I believe that the evidence needed is readily available. A relevant bibliography of studies of note would include the following studies of merit/relevance to this response:

1: Nevens F et al. A Placebo-Controlled Trial of Obeticholic Acid in Primary Biliary Cholangitis. N Engl J Med. 2016 Aug 18;375(7):631-43.

2: Lammers WJ et al. Development and Validation of a Scoring System to Predict Outcomes of Patients With Primary Biliary Cirrhosis Receiving Ursodeoxycholic Acid Therapy. Gastroenterology. 2015 Dec;149(7):1804-1812.

3: Carbone M et al. The UK-PBC risk scores: Derivation and validation of a scoring system for long-term prediction of end-stage liver disease in primary biliary cholangitis. Hepatology. 2016 Mar;63(3):930-50.

4: Trivedi PJ et al. Stratification of hepatocellular carcinoma risk in primary biliary cirrhosis: a multicentre international study. Gut. 2016 Feb;65(2):321-9.

5: Hirschfield GM et al. Efficacy of obeticholic acid in patients with primary biliary cirrhosis and inadequate response to ursodeoxycholic acid. Gastroenterology. 2015 Apr;148(4):751-61. 6: Lammers WJ et al. Levels of alkaline phosphatase and bilirubin are surrogate end points of outcomes of patients with primary biliary cirrhosis: an international follow-up study. Gastroenterology. 2014 Dec;147(6):1338-49.

7: Mells GF et al. Impact of primary biliary cirrhosis on perceived quality of life: the UK-PBC national study. Hepatology. 2013 Jul;58(1):273-83.

8: Carbone M et al. Sex and age are determinants of the clinical phenotype of primary biliary cirrhosis and response to ursodeoxycholic acid. Gastroenterology. 2013 Mar;144(3):560-569.

Implementation issues

This is an oral therapy and there would need to be very limited additional education and training to implement this therapy, all of which could readily be delivered as part of existing efforts for improved care of PBC generally by UK-PBC. Therapy is delivered from routine out patient visits and no additional facilities are envisaged. To ensure adequate reach and the opportunity to ensure rapid and timely, but safe, delivery of this therapy, it is likely desirable that regional networks are evolved, and there is the opportunity to do so using existing infrastructure used to deliver other therapies in liver disease.

Equality

None relevant but evaluation of this technology must accept and acknowledge that PBC is a rare disease and as such the conduct of clinical trials with hard clinical outcomes has, and always will, remain very challenging. Surrogates are therefore an inevitable component of the evaluation of new therapies for patients with PBC.

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Patient/carer organisation submission (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Thank you for agreeing to give us your views on this treatment that is being appraised by NICE and how it could be used in the NHS. Patients, carers and patient organisations can provide a unique perspective on conditions and their treatment that is not typically available from other sources. We are interested in hearing about:

- the experience of having the condition or caring for someone with the condition
- the experience of receiving NHS care for the condition
- the experience of having specific treatments for the condition
- the outcomes of treatment that are important to patients or carers (which might differ from those measured in clinical studies, and including health-related quality of life)
- the acceptability of different treatments and how they are given
- expectations about the risks and benefits of the treatment.

To help you give your views, we have provided a questionnaire. You do not have to answer every question — the questions are there as prompts to guide you. The length of your response should not normally exceed 10 pages.

1. About you and your organisation

Your name:

Name of your organisation: The PBC Foundation

Your position in the organisation: Head of Education and Development

Brief description of the organisation:

The PBC Foundation is a UK based international charity that provides information, help and support to those affected by PBC: patients, carers and families. The Foundation provides a free service to those it supports. The Foundation has almost 10,000 registered service users in 71 countries around the world, some of whom are already benefitting from the appraised therapy.

Links with, or funding from the tobacco industry - please declare any direct or indirect links to, and receipt of funding from the tobacco industry: None

2. Living with the condition

What is it like to live with the condition or what do carers experience when caring for someone with the condition?

Living with PBC has been described as "Living on an emotional and physical rollercoaster." As symptoms vary from week to week and day to day, patients (and carers) never know what PBC will bring on a day to day basis. Chronic fatigue (with the linked symptom of cognitive impairment), pruritus, joint/muscle/bone pain, and nausea are just some of the symptoms PBC patients face.

The fatigue can be genuinely debilitating. Intractable itch can be an indication for liver transplantation. The symptoms, just themselves, can have an enormously detrimental effect on quality of life. This often leads to social isolation which reinforces negative quality of life issues, and so begins an all too familiar spiral.

Aside from the symptoms, which are not indicative of disease progression, there are other, even more devastating aspects of living with PBC. The diagnosis of an incurable, progressive disease, particularly one with such a foreboding outline (before the widespread use of UDCA, many studies predicted life expectancy of *up to* 7 years life expectancy after diagnosis, and many patients still find this information on the internet) can lead to an insurmountable emotional burden. One in three patients with PBC experience depression. For many patients, there is an experience of the grieving process, where one deals with the loss of their perceived life and comes to terms with a shorter life with reduced quality.

We now know that early intervention combined with response to UDCA leads to normal life expectancy. This information, in itself, is of enormous benefit to patients. Once armed with this information, and assuming UDCA response, these patients live in a much improved paradigm with less fear and are better equipped to manage their symptoms on a daily basis.

However, whilst science doesn't agree on a figure, we know many, many patients who do not respond satisfactorily to UDCA. This leads to a very different and far more difficult paradigm. With patients facing life without hope, the combined challenges of emotional and physical self-management become too much, to devastating effect. Many give up work, becoming more and more inwards and reclusive: negatively affecting home and family life, also. The questions patients first ask at diagnosis- "How long will I live?"; "Will I see my children grow up?"; etc return with a much, much darker answer.

Another difficult cited by patients is a lack of knowledge and understanding within medical communities: particularly in primary care and district hospital levels. It is widely anticipated that a successful novel therapy for PBC will provide opportunity for much needed education of medics re PBC.

3. Current practice in treating the condition

Which treatment outcomes are important to patients or carers? (That is, what would patients or carers like treatment to achieve?) Which of these are most important? If possible, please explain why.

In an ideal world, patients would dearly love a cure to PBC. At present, the best we have is slower progression of liver damage for a number of patients. Patients' priorities are longevity of life, then symptom management. Many patients are holding out for a cure to the debilitating fatigue. UDCA can, and seems to, prolong life. All patients with PBC want this for themselves. To explain why would be difficult in such a small space but each of us, I am sure, understand and empathise with the hope of an equal chance of normal life expectancy.

What is your organisation's experience of currently available NHS care and of specific treatments for the condition? How acceptable are these treatments and which are preferred and why?

We know that UDCA is a safe, tolerable and successful treatment for many. The UK is renowned, unfortunately for good reason, within the international PBC community for poor treatment of PBC. In the UK, it has been proven that many patients do not have access to UDCA and many who do are not given the recommended dose. With a new medication available to those who do not respond to first line therapy, we believe very strongly that this will lead to more patients receiving first line therapy quicker, easier and in an appropriate dosage.

Liver transplantation is not a cure for PBC. It is an exchange of challenges. Even after transplant, most people with PBC experience a PBC attack upon the new liver.

We are aware of off-label use of fibrates but there is currently a dearth of peerreviewed studies that we have seen that support wide-spread use in PBC for UDCA non-responders.

4. What do patients or carers consider to be the

advantages of the treatment being appraised?

Benefits of a treatment might include its effect on:

- the course and/or outcome of the condition
- physical symptoms
- pain
- level of disability
- mental health
- quality of life (such as lifestyle and work)
- other people (for example, family, friends and employers)
- ease of use (for example, tablets rather than injection)
- where the treatment has to be used (for example, at home rather than in hospital)
- any other issues not listed above

Please list the benefits that patients or carers expect to gain from using the treatment being appraised.

Prolonged life expectancy

Improved quality of life (through improved mental health and decreased isolation) Positive impact on home and family life for carers as well as patients Less need for liver transplantation Less deaths through unavailability of liver transplantation Hope

Please explain any advantages that patients or carers think this treatment has over other NHS treatments in England.

OCA is a proposed second line therapy for PBC. As there is no current second line therapy for PBC, then this treatment is immeasurably advantageous. Whilst liver transplantation is curative for end stage liver disease (often caused by PBC) and liver failure, it is not curative for PBC.

If you know of any differences in opinion between patients or carers about the benefits of the treatment being appraised, please tell us about them.

None

5. What do patients and/or carers consider to be the

disadvantages of the treatment being appraised?

Disadvantages of a treatment might include:

- aspects of the condition that the treatment cannot help with or might make worse
- difficulties in taking or using the treatment (for example, injection rather than tablets)
- side effects (for example, type or number of problems, how often, for how long, how severe. Please describe which side effects patients might be willing to accept or tolerate and which would be difficult to accept or tolerate)
- where the treatment has to be used (for example, in hospital rather than at home)
- impact on others (for example, family, friends and employers)
- financial impact on the patient and/or their family (for example, the cost of travel to hospital or paying a carer)
- any other issues not listed above

Please list any concerns patients or carers have about current NHS treatments in England.

Both patients and carers are fearful of transplant: Fear of the operation itself Fear of being too ill but not "making the list" Fear of not passing assessment Fear of pasdsding assessment for there not to be a liver available Fear of post-operation survival quality of life

Please list any concerns patients or carers have about the treatment being appraised.

Patients are aware of complications of pruritus with the appraised medication within clinical trials, yet anecdotal evidence from those currently prescribed with the medication is that itch normalises within a short timeframe.

If you know of any differences in opinion between patients or carers about the disadvantages of the treatment being appraised, please tell us about them.

None known.

6. Patient population

Are there any groups of patients who might benefit more from the treatment than others? If so, please describe them and explain why.

Any patient for whom treatment for their PBC has been ineffective might benefit from the appraised treatment. This may include UDCA non-responders and those intolerant of UDCA. This group of patients would benefit as there is no current second line therapy available to these patients.

Are there any groups of patients who might benefit less from the

National Institute for Health and Care Excellence
treatment than others? If so, please describe them and explain why.

None known

7. Research evidence on patient or carer views of the

treatment

Is your organisation familiar with the published research literature for the treatment?

□x <u>Yes</u> □ No

If you answered 'no', please skip the rest of section 7 and move on to section 8.

Please comment on whether patients' experience of using the treatment as part of their routine NHS care reflects the experiences of patients in the clinical trials.

Do you think the clinical trials have captured outcomes that are important to patients? Are you aware of any limitations in how the treatment has been assessed in clinical trials?

If the treatment being appraised is already available in the NHS, are there any side effects that were not apparent in the clinical trials but have emerged during routine NHS care?

Are you aware of any relevant research on patient or carer views of the condition or existing treatments (for example, qualitative studies, surveys and polls)?

 \Box Yes \Box No

If yes, please provide references to the relevant studies.

8. Equality

NICE is committed to promoting equality of opportunity, eliminating unlawful discrimination and fostering good relations between people with particular protected characteristics and others. Protected characteristics are: age; being or becoming a transsexual person; being married or in a civil partnership;

National Institute for Health and Care Excellence

Appendix G – patient/carer organisation submission template

being pregnant or having a child; disability; race including colour, nationality, ethnic or national origin; religion, belief or lack of religion/belief; sex; sexual orientation.

Please let us know if you think that recommendations from this appraisal could have an adverse impact on any particular groups of people, such as:

- excluding from full consideration any people protected by the equality legislation who fall within the patient population for which the treatment is/will be licensed;
- having a different impact on people protected by the equality legislation than on the wider population, e.g. by making it more difficult in practice for a specific group to access the treatment;
- any adverse impact on people with a particular disability or disabilities.

Please let us know if you think that there are any potential equality issues that should be considered in this appraisal.

Are there groups of patients who would have difficulties using the treatment or currently available treatments? Please tell us what evidence you think would help the Committee to identify and consider such impacts.

None known

9. Other issues

Do you consider the treatment to be innovative?

□X <u>Yes</u> □ No

If yes, please explain what makes it significantly different from other treatments for the condition.

This treatment is innovative for many reasons:

Firstly, it is the first new therapy for PBC in over 20 years. Secondly, because it has a different mechanism, the dosage is fixed and not dependent on weight of patient, etc. This simplifies prescription and daily intake of medicine for those patients on this treatment. It also addresses the unmet need of a significant proportion of patients who do not respond to the only other medication available for PBC.

Are there any other issues that you would like the Appraisal Committee to consider?

10. Key messages

In no more than 5 bullet points, please summarise the key messages of your submission.

- Prolonged life expectancy
- Improved quality of live
- Reduced reliance on (and failure of) liver transplantation service
- Reduced social isolation, economic isolation, HCP burden
- Equity of service

Appendix K – clinical expert statement declaration form

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Please sign and return via NICE Docs/Appraisals.

I confirm that:

I agree with the content of the submission provided by BSG and RCP and consequently I will not be submitting a personal statement.

Name: ...

Signed:

Date:22nd February 2017.....

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Please sign and return via NICE Docs/Appraisals.

I confirm that:

• I agree with the content of the statement submitted by [The PBC Foundation] and consequently I will not be submitting a personal statement.

Name:		
Signed:		
Date:21 st December 20	16	

NATIONAL INSTITUTE FOR HEALTH AND CARE EXCELLENCE

Single Technology Appraisal (STA)

Obeticholic acid for treating primary biliary cirrhosis [ID785]

Please sign and return via NICE Docs/Appraisals.

I confirm that:

¥

• I agree with the content of the submission provided by **British Association for Study of the Liver (BASL)** and consequently I will not be submitting a personal statement.

Name:	LEORGE	MERCS.
Sianed:	1.	(70m),
Date:	22 / 02	/17.



in collaboration with:



Maastricht University

Obeticholic acid for treating primary biliary cirrhosis

Produced by	Kleijnen Systematic Reviews (KSR) Ltd. in collaboration with Erasmus University Rotterdam (EUR) and Maastricht University
Authors	Debra Fayter, Systematic Reviewer, KSR
	Bram Ramaekers, Health Economist, Maastricht UMC+
	Sabine Grimm, Health Economist, Maastricht UMC+
	Xavier Pouwels, Health Economist, Maastricht UMC+
	Nigel Armstrong, Health Economics Manager, KSR
	Steve Ryder, Health Economist, KSR
	Sonia Garcia, Systematic Reviewer, KSR
	Piet Portegijs, Systematic Reviewer, KSR
	Caro Noake, Information Specialist, KSR
	Rob Riemsma, Reviews Manager, KSR
	Manuela Joore, Health Economist, Professor of Health Technology Assessment & Decision Making, Maastricht UMC+
	Jos Kleijnen, Director, KSR; Professor of Systematic Reviews in Health Care, Maastricht University
Correspondence to	Debra Fayter, Kleijnen Systematic Reviews
	Unit 6, Escrick Business Park
	Riccall Road, Escrick
	York, UK
	YO19 6FD

Date completed 30/11/2016

Source of funding: This report was commissioned by the NIHR HTA Programme as project number STA 15/69/09.

Declared competing interests of the authors

None.

Acknowledgements

We are very grateful to the following expert who offered clinical advice:

Dr Stephen Ryder, Consultant Hepatologist at Nottingham University Hospital Trust.

The clinical expert was not directly involved in writing the report and overall responsibility for the report remains with the ERG.

Commercial in confidence (CiC) data are highlighted in blue throughout the report. Academic in confidence (AiC) data are highlighted in yellow throughout the report.

Rider on responsibility for report

The views expressed in this report are those of the authors and not necessarily those of the NIHR HTA Programme. Any errors are the responsibility of the authors.

This report should be referenced as follows:

Fayter D, Ramaekers BLT, Grimm S, Pouwels X, Armstrong N, Ryder S, Garcia S, Portegijs P, Noake C, Riemsma R, Joore MA, Kleijnen J. Obeticholic acid for treating primary biliary cirrhosis: a Single Technology Assessment. York: Kleijnen Systematic Reviews Ltd, 2016.

Contributions of authors

Debra Fayter acted as project lead and systematic reviewer on this assessment, critiqued the clinical effectiveness methods and evidence and contributed to the writing of the report. Bram Ramaekers acted as health economic project lead, critiqued the company's economic evaluation and contributed to the writing of the report. Xavier Pouwels, Sabine Grimm, Nigel Armstrong and Steve Ryder acted as health economists on this assessment, critiqued the company's economic evaluation and contributed to the writing of the report. Sonia Garcia, Rob Riemsma and Piet Portegijs acted as systematic reviewers, critiqued the clinical effectiveness methods and evidence and contributed to the writing of the report. Caro Noake critiqued the search methods in the submission and contributed to the writing of the report. Manuela Joore acted as health economist on this assessment, critiqued the company's economic evaluation, contributed to the writing of the report and provided general guidance. Jos Kleijnen critiqued the company's definition of the decision problem and their description of the underlying health problem and current service provision, contributed to the writing of the report and supervised the project.

Abbreviations

AASLD	American Association for the Study of Liver Disease
AB	Abnormal bilirubin
AE	Adverse event
ALD	Alcoholic liver disease
ALP	Alkaline phosphatase
ALT	Alanine aminotransferase
AMA	Anti-mitochondrial antibody
ANCOVA	Analysis of covariance
AST	Aspartate aminotransferase
ATC	Anatomical Therapeutic Chemical Classification System
BAS	Bile acid sequestrants
BASL	British Association for the Study of the Liver
BMI	Body mass index
BNF	British National Formulary
BSG	British Society of Gastroenterology
СА	Cholic acid
CC	Compensated cirrhosis
CDCA	Chenodeoxycholic acid
СЕ	Cost effectiveness
CEA	Cost effectiveness analysis
CEAC	Cost-effectiveness acceptability curve
СЕТР	Cholesterol ester transfer protein
СНВ	Chronic hepatitis B
СНС	Chronic hepatitis C
СНМР	Committee for Medicinal Products for Human Use
CI	Confidence interval
CK-18	Cytokeratin-18
CLD	Chronic cholestatic liver disease
CLDQ	Chronic liver disease questionnaire
СМН	Cochran-Mantel-Haenszel
CONSORT	Consolidated Standards of Reporting Trials
CRD	Centre for Reviews and Dissemination
CRP	C-reactive protein
CS	Company submission
CSR	Clinical study report
DALY	Disability-adjusted life years
DARE	Database of abstracts of reviews and effects
DC	Decompensated cirrhosis
DCA	Deoxycholic acid

DCC	Decompensated cirrhosis
df	Degrees of freedom
DSA	Deterministic sensitivity analysis
DSU	Decision Support Unit
EASL	European Association for the Study of Liver
ECDCA	Ethyl-chenodeoxycholic acid
ELF	Enhanced liver fibrosis
EMA	European Medicines Agency
EMEA	European Medicines Evaluation Agency
EO	Expert opinion leader
EOT	End of treatment
ERG	Evidence Review group
EU	European Union
FDA	Food and Drug Administration
FGF-19	Fibroblast growth factor-19
FXR	Farnesoid X receptor
GGT	Gamma-glutamyl transferase
GH	General health
GP	General practitioner
НА	Hyaluronic acid
HBV	Hepatitis B virus
HCC	Hepatocellular carcinoma
HCHS	Hospital & Community Health Services
HCV	Hepatitis C virus
HDL	High-density lipoprotein
HDLc	High-density lipoprotein cholesterol
HIV	Human immunodeficiency virus
HRQoL	Health-related quality of life
HTA	Health Technology Assessment
HUI	Health utility index
ICER	Incremental cost-effectiveness ratio
ICU	Intensive care unit
IL	Interleukin
INR	International standardised ration
IQR	Interquartile range
ITT	Intention to treat
IWRS	Interactive web response system
KM	Kaplan-Meier
KSR	Kleijnen Systematic Reviews
LA	Lysophosphatidic acid

LCA	Lithocholic acid
LCAT	Lecithin-cholesterol acyltransferase
LDL	Low density lipoprotein
LDLc	Low-density lipoprotein cholesterol
LDLT	Living donor liver transplantation
LLN	Lower limit of normal
LP	lipoprotein
LS	Least squares
LT	Liver transplantation
LTSE	Long-term safety extension
LYG	Life years gained
MCID	Minimal clinically important difference
MCS	Mental component summary
MELD	Model for End Stage Liver Disease
MH	Mental health
MESH	Medical Subject Headings
mg	Milligrams
MHRA	Medicines and Healthcare Products Regulatory Agency
MRS	Mayo risk score
MRU	Medical resource utilisation
MTC	Mixed Treatment Comparison
NA	Not applicable
NB	Normal bilirubin
NCPE	National Centre for Pharmacoeconomics
NHS	National Health Service
NICE	National Institute for Health and Care Excellence
NIHR	National Institute for Health Research
NR	Not reported
OCA	Obeticholic acid
OD	Once daily
OPTN	Organ procurement and transplantation network
OR	Odds ratio
PAS	Patient access scheme
PASLU	Patient access schemes liaison unit
PBC	Primary biliary cirrhosis (recently renamed as 'primary biliary cholangitis')
PCS	Physical component summary
PF	Physical functioning
РК	Pharmacokinetics
PSA	Probabilistic sensitivity analysis
PSC	Primary sclerosing cholangitis

PRESS	Peer Review of Electronic Search Strategies
РТ	Prothrombin time
PYE	Patient years of exposure
QALY(s)	Quality adjusted life year(s)
QoL	Quality of life
RCT	Randomised controlled trial
RE	Role emotional
REML	Restricted maximum likelihood
RMSE	Root mean square error
RP	Role physical
RR	Relative Risk; Risk Ratio
SAE	Serious adverse event
SD	Standard deviation
SE	Standard error
SF	Social functioning
SLR	Systematic literature review
SMC	Scottish Medicines Consortium
SOC	System organ class
SmPC	Summary of product characteristics
STA	Single Technology Appraisal
SVR	Sustained viral response
ТВ	Total bilirubin
ТЕ	Transient elastography
TEAE	Treatment-emergent adverse event
TGF	Transforming growth factor
TIMP	Tissue inhibitor of metalloproteinase
TNF	Tumour necrosis factor
ТР	Transition probability
UDCA	Ursodeoxycholic acid
UK	United Kingdom
ULN	Upper limit of normal
VAS	Visual analogue scale
VLDL	Very low density lipoprotein
VT	Vitality
WADD	Weighted average daily dose
WHO	World Health Organisation

Table of Contents

Abbrev	viations	3
Table of	of Tables	
Table of	of Figures	11
1. SUM	IMARY	12
1.1	Critique of the decision problem in the company's submission	12
1.2	Summary of clinical effectiveness evidence submitted by the company	12
1.3	Summary of the ERG's critique of clinical effectiveness evidence submitted	
1.4	Summary of cost effectiveness submitted evidence by the company	13
1.5	Summary of the ERG's critique of cost effectiveness evidence submitted	15
1.6	ERG commentary on the robustness of evidence submitted by the company	17
1.6.1	Strengths	17
1.6.2	2 Weaknesses and areas of uncertainty	17
1.7	Summary of exploratory and sensitivity analyses undertaken by the ERG	17
2. BAC	CKGROUND	19
2.1	Critique of company's description of underlying health problem.	19
2.2	Critique of company's overview of current service provision	20
3. CRI	TIQUE OF COMPANY'S DEFINITION OF DECISION PROBLEM	22
3.1	Population	25
3.2	Intervention	25
3.3	Comparators	
3.4	Outcomes	
3.5	Other relevant factors	29
4. CLI	NICAL EFFECTIVENESS	
4.1	Critique of the methods of review(s)	
4.1.1	Searches	
4.1.2	2 Inclusion criteria	
4.1.3	Critique of data extraction	33
4.1.4	Quality assessment	
4.1.5	5 Evidence synthesis	
4.2	Critique of trials of the technology of interest, their analysis and interpretation (standard meta-analyses of these)	and any
4.2.1	Overview of the evidence in the submission	34
4.2.2	Overview of the POISE trial	35
4.2.3	Quality assessment of POISE	44
4.2.4	POISE: Efficacy results	47
4.2.5	POISE: Safety results	49

4.3	Critique of trials identified and included in the indirect comparison and/or multiple tro- comparison	eatment 54
4.4	Critique of the indirect comparison and/or multiple treatment comparison	54
4.5	Additional work on clinical effectiveness undertaken by the ERG	54
4.6	Conclusions of the clinical effectiveness section	54
5. CO	ST EFFECTIVENESS	55
5.1	ERG comment on company's review of cost effectiveness evidence	55
5.1.	1 Searches performed for cost effectiveness section	55
5.1.2	2 Inclusion/exclusion criteria used in the study selection	57
5.1.	3 Included/excluded studies in the cost effectiveness review	57
5.1.4	4 Conclusions of the cost effectiveness review	57
5.2	Summary and critique of company's submitted economic evaluation by the ERG	58
5.2.	1 NICE reference case checklist (TABLE ONLY)	60
5.2.2	2 Model structure	
5.2.	3 Population	64
5.2.4	4 Interventions and comparators	65
5.2.:	5 Perspective, time horizon and discounting	65
5.2.	6 Treatment effectiveness and extrapolation	65
5.2.	7 Adverse events	73
5.2.3	8 Health-related quality of life	73
5.2.	9 Resources and costs	77
5.2.	10 Cost effectiveness results	80
5.2.	11 Sensitivity analyses	85
5.2.	12 Model validation and face validity check	
5.3	Exploratory and sensitivity analyses undertaken by the ERG	90
5.3.	Probabilistic sensitivity analysis (ERG base-case)	91
5.3.2	Additional exploratory and threshold analyses performed based on the ERG ba 93	ise-case
5.4	Conclusions of the cost effectiveness section	94
6. IMI	PACT ON THE ICER OF ADDITIONAL CLINICAL AND ECONOMIC ANA UNDERTAKEN BY THE ERG	LYSES 96
7. ENI	D OF LIFE	100
8. OV	ERALL CONCLUSIONS	101
8.1	Statement of principal findings	101
8.2	Strengths and limitations of the assessment	102
8.3	Suggested research priorities	102
9. REI	FERENCES	103
ADDE	NDIX 1. DETAILS OF ERC ANALVSES (FOD VALIDATION DUDDOSES)	112
AITL	NDIA 1. DETAILS OF ERGANALTSES (FOR VALIDATION FURPOSES)	112

APPENDIX 3: TORNADO DIAGRAMS OF THE DSA FOR BOTH POPULATIONS. 115

Table of Tables

Table 3.1: Decision Problem	.22
Table 3.2: Response to the omission of fibrates as a comparator in the submission	.26
Table 4.1: Eligibility criteria used in the clinical effectiveness and safety systematic review	.31
Table 4.2: Summary of the POISE trial and its extension study	.35
Table 4.3: Overview of POISE trial	.36
Table 4.4: Summary of secondary endpoints of the double-blind phase of POISE	.36
Table 4.5: POISE inclusion and exclusion criteria	. 39
Table 4.6: POISE Trial: Baseline patient demographics	.41
Table 4.7: POISE Trial: Baseline disease characteristics	.42
Table 4.8: Quality assessment of POISE	.45
Table 4.9: Summary of primary efficacy outcome in POISE	.47
Table 4.10: All-cause mortality and clinical complications: safety population (N = 216)	.48
Table 4.11: Clinical Outcomes – Transplant (MELD Score): Safety Population (N = 216)	.48
Table 4.12: Overview of adverse events: safety population (POISE)	.49
Table 4.13: Summary of TEAEs, severity of AEs, and treatment-related AEs occurring in \geq 5% participants in either OCA treatment group	o of . 50
Table 4.14: Summary of Phase 2 study characteristics (double-blind phase)	. 53
Table 5.1: Summary of the company's economic evaluation (with signposts to CS)	. 58
Table 5.2: NICE reference case checklist	.60
Table 5.3: Transitions between health states of the biomarker component – OCA titration	.66
Table 5.4: Transitions between health states of the biomarker component – UDCA alone	.67
Table 5.5: Transitions between health states of the biomarker component - non-OCA regimen	. 69
Table 5.6: Transitions between health states of the liver disease component	.71
Table 5.7: Transitions between health states of the liver disease component based on TA330	. 72
Table 5.8: Incidence of AE	.73
Table 5.9: Included HRQoL studies	. 74
Table 5.10: Health state utility values used in the company's cost effectiveness model	.75
Table 5.11: Unit costs associated with the technology in the economic model	.77
Table 5.12: List of health states and associated annual costs in the economic model	. 79
Table 5.13: List of adverse reactions and summary of costs included in the economic model	. 80
Table 5.14: Summary of QALY gain by health state for UDCA-intolerant population	. 81
Table 5.15: Summary of QALY gain by health state for UDCA inadequate responders	. 81
Table 5.16: Summary of costs by health state for UDCA-intolerant population	. 82
Table 5.17: Summary of costs by health state for UDCA inadequate responders	. 83
Table 5.18: Deterministic base-case results for UDCA-intolerant patients, using the list price for O	CA . 83
Table 5.19: Deterministic base-case results for UDCA inadequate responders, using the list price OCA	for . 84
Table 5.20: Summary of life years gained by health state for UDCA-intolerant population	. 84
Table 5.21: Summary of life years gained by health state for UDCA inadequate responders	.85
Table 5.22: Incremental cost effectiveness results of PSA for the UDCA intolerant population, us the list price of OCA	ing . 86
Table 5.23: Incremental cost effectiveness results of PSA for the UDCA inadequate respon population, using the list price of OCA	ider . 86

Table 5.24: Deterministic results for UDCA-intolerant patients, using the list price of OCA and patient-level data	UDCA 88
Table 5.25: Base-case results for UDCA inadequate responders, using the list price of OCA, and patient-level data	UDCA 88
Table 5.26: Mean time to event for liver-related death	89
Table 5.27: Summary of model outcomes for UDCA-intolerant population	89
Table 5.28: Summary of model outcomes for UDCA inadequate responder population	90
Table 5.29: ERG base-case for UDCA inadequate responders	92
Table 5.30: ERG base-case for UDCA intolerant patients	92
Table 6.1: ERG base-case - UDCA inadequate responders	96
Table 6.2: ERG base-case - UDCA intolerant patients	98
Table 6.3: ERG exploratory analysis - UDCA inadequate responders	98
Table 6.4: ERG exploratory analysis - UDCA intolerant patients	99

Table of Figures

Figure 5.1: Model structure	63
Figure 5.2: Incremental cost effectiveness planes for ERG base-case	93
Figure A2.1: Scatter plot for PSA (UDCA-intolerant population), using the OCA list price	113
Figure A2.2: Cost effectiveness acceptability curve for OCA titration versus no treatment (U intolerant population), using the OCA list price	JDCA-
Figure A2.3: Scatter plot for PSA (UDCA inadequate responders population), using the OCA lis 	st price e fined.
Figure A2.4: Cost effectiveness acceptability curve for OCA + UDCA titration versus UDCA (inadequate responders population), using the OCA list price	UDCA 114
Figure A3.1: Tornado diagram for OCA titration versus no treatment in the UDCA int population, using the list price of OCA	olerant 115
Figure A3.2: Tornado diagram for OCA + UDCA titration versus UDCA in the UDCA inad responder population, using the list price of OCA	lequate 115

1. SUMMARY

1.1 Critique of the decision problem in the company's submission

The patient population described in the final scope issued by the National Institute for Health and Care Excellence (NICE) was '*People with primary biliary cirrhosis whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid*'. For patients whose disease has an inadequate response to ursodeoxycholic acid (UDCA), obeticholic acid (OCA) was to be compared to UDCA alone or in combination with fibrates. For patients who are unable to tolerate UDCA, comparators were fibrates or no additional treatment. Outcomes included mortality, liver function based on markers of liver biochemistry, symptoms, including pruritus, fatigue and abdominal pain, time to liver transplantation, primary biliary cirrhosis (PBC)-related events, adverse effects of treatment and health-related quality of life (HRQoL).

The decision problem in the company submission (CS) differed from the scope in a number of ways. Firstly, fibrates were not considered by the company to be a comparator to OCA. Secondly, the main evidence presented (the POISE trial) considered only surrogate outcomes. No data were available from the trial on long-term clinical outcomes outlined in the scope such as mortality and liver transplantation. Finally the number of patients in POISE receiving OCA as monotherapy (i.e. without UDCA) was very small (11 patients) so results for this group of patients should be treated with some caution.

1.2 Summary of clinical effectiveness evidence submitted by the company

The company conducted a systematic review to inform the submission. The aim of the systematic review was 'to identify all relevant evidence for the efficacy and safety of interventions used to treat PBC.' A post-hoc decision was taken to include only trials with an OCA treatment arm.

The systematic review identified one main randomised controlled trial, POISE, and two supporting Phase 2 trials. The company did not pool the results of the trials.

POISE, was an international trial of 217 patients including patients from the UK. The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. Seventy-three patients received 10 mg OCA, 71 patients received 5 mg OCA rising to 10 mg during months 6 to 12 if they had an inadequate response to UDCA (the titration group).

The primary outcome of POISE was a composite one (percentage of participants with alkaline phosphatase (ALP) < 1.67 x upper limit of normal (ULN) and total bilirubin \leq ULN and ALP \geq decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events (AEs) was similar across groups. Events occurring more frequently in treatment groups included pruritus. It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

The Phase 2 trials supported the positive findings of POISE on surrogate outcomes.

1.3 Summary of the ERG's critique of clinical effectiveness evidence submitted

The systematic review conducted by the company was broadly appropriate to the scope of this submission. The CS and response to clarification provided sufficient details for the ERG to appraise the searches. A broad range of databases were searched, and additional searches of conference proceedings and other relevant resources including trials databases, HTA agencies, specialist and organisational

websites were reported. The ERG is satisfied that the direct evidence comparing OCA and UDCA has been presented. However no evidence on fibrates as a comparator was presented.

The main evidence was based on POISE, a well-conducted international randomised controlled trial of 217 patients with a 12 month follow up. The ERG agrees that this trial could not be pooled with the two Phase 2 trials identified due to differences between the trials including OCA dosage and study duration.

The ERG identified some limitations in applying the results of the POISE trial to the NICE scope.

- The number of patients intolerant to UDCA and receiving OCA as monotherapy was limited to 11 patients (five [7%] in the OCA titration group and six [8%] in the OCA 10 mg group). Five patients received OCA placebo. As such the results for this group of patients should be considered with caution due to the low numbers.
- The ERG noted that the POISE trial included mainly patients at an early stage of PBC progression. The ERG asked the company to clarify how OCA would benefit patients with more advanced PBC. The company stated that two analyses were performed in more advanced stages of PBC. The brief results of these analyses were based on a subset of 72 patients classified as having advanced disease in particular 36 patients who had cirrhosis. The ongoing COBALT trial includes more advanced patients and should provide more definitive results for those patients with more advanced disease.
- In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage of OCA. They received OCA 5mg OD for the initial six month period. The patients in POISE were only up-titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg.
- The POISE trial present evidence using surrogate outcomes only. The ERG is satisfied that the company has demonstrated some evidence of correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease is unclear.
- The definitive effects of obeticholic acid on clinical outcomes relevant to patients awaits the results of the ongoing COBALT trial

1.4 Summary of cost effectiveness submitted evidence by the company

The company conducted systematic reviews to identify relevant cost effectiveness studies, healthrelated quality of life studies, resource use and costs studies. The company did not identify any study investigating the cost effectiveness of OCA in the population of interest for the current decision problem, and hence developed a *de novo* model.

The cost effectiveness analysis described in the CS is considered by the ERG to partially meet the NICE reference case. The company developed a Markov state transition model to describe the progression of PBC over a lifetime horizon. The model is composed of two parts with a total of 10 health states. The first part of the model (referred to as the biomarker component) captures the surrogate outcomes of ALP and bilirubin biomarkers in three different health states based on the expected risk of disease progression: low risk (ALP $\leq 1.67 *$ ULN); moderate risk (ALP > 1.67 * ULN and TB $\leq 1.0 *$ ULN) and severe risk (TB > 1.0 * ULN or compensated cirrhosis; CC). The latter health state combined patients with CC and those with abnormal TB. In the second part of the model, the liver disease component, the following clinical endpoints are modelled: pre-liver transplant; decompensated cirrhosis

(DCC); hepatocellular carcinoma (HCC); liver transplant; a post-liver transplant state; potential PBC re-emergence; and death. Patients can move to the liver disease component of the model only coming from the severe risk health state of the biomarker component of the model. Regarding adverse events, only the costs in relation to pruritus were modelled, while the impact on HRQoL was not modelled. A four week cycle length was used over a time horizon of 50 years, which is effectively a life time perspective.

The economic evaluation considers PBC patients who are intolerant to or have inadequate response to treatment with UDCA. Only moderate and severe patients based on ALP and bilirubin levels, at model entry, were deemed eligible for OCA treatment in the model. This is in line with the final scope issued by NICE for this appraisal and is also in line with the study population of the pivotal POISE study.

The interventions and comparators in this model depend on the selected population. For UDCAintolerant patients, the intervention considered in this economic evaluation is OCA dose titration based on a starting dose of 5 mg taken orally, once daily, which may be increased to 10 mg once daily based on the assessment of tolerability after six months, to achieve optimal response and the comparator is no treatment. For UDCA inadequate responders, the intervention is OCA dose titration as per above in combination with UDCA; and the comparator is UDCA monotherapy. UDCA doses implemented in the model were in line with its UK marketing authorisation, which are also the same as in the POISE trial. Fibrates, listed in the scope as a comparator, were not considered as a comparator. The company argued that fibrates are not licensed in the UK, nor are they standard care, and they are contraindicated in PBC. The company also argued that they are rarely used, with only **DECE** of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC).

For the health states in the biomarker component, transition probabilities for the OCA-based regimen relied on POISE patient-level data during the first year. Given the low number of patients who received OCA monotherapy, the same transition matrices were used for the OCA titration regimen with or without UDCA. The transition probabilities used to reflect progression whilst on UDCA alone, or no treatment, are based on data from various literature sources. Mortality in the biomarker component of the model was assumed to be equal to background mortality. The literature was used to inform transitions between the health states in the liver disease component of the model, in part relying on previous assessments (i.e. TA330).

HRQoL evidence was not collected through a generic preference elicitation instrument in the POISE trial, and was therefore obtained from the literature, and based on assumptions. The company did not provide a justification for the health state utility values used for the biomarker component of the model. The company applied a decrement to the health state utility values found in the literature for the liver disease component of the model, except for the HCC health state. This decrement was based on expert opinion (further details on how the decrement was estimated were not provided by the company).

Resource use and costs included in the model were based on data from the British National Formulary (BNF), NHS reference costs, assumptions validated by expert opinion, and published sources identified in the literature review. The list price was used for OCA while drug acquisition costs for UDCA were determined from the BNF and market share data to calculate a weighted average. Health state costs were obtained from different sources: expert opinion, NHS reference costs and published studies. Finally, the costs of pruritus were based on expert opinion and the BNF.

In both populations, OCA (without and with UDCA) led to longer survival, a quality adjusted life year (QALY) gain, and higher costs than the comparators (i.e. no treatment for the UDCA-intolerant population and UDCA monotherapy for the UDCA inadequate responders). The main QALY gain of

OCA was accumulated in the low risk and moderate risk health states in the biomarker component. The same was observed for costs; more than half of the incremental cost was accrued in the low risk and moderate risk health states. The company base-case incremental cost effectiveness ratios (ICERs) (probabilistic) of OCA (without or with UDCA) versus no treatment and UDCA alone were for UDCA intolerant patients and UDCA

inadequate responders respectively. The one-way sensitivity analyses conducted by the company showed that the model results were most sensitive to the health states utility values for the health states of the biomarker component of the model and the transition probabilities between these health states. It should be noted though that the one-way sensitivity analyses were partly based on arbitrary estimates of the variance (i.e. using $\pm 20\%$ of the mean value).

1.5 Summary of the ERG's critique of cost effectiveness evidence submitted

The majority of searches in Appendices 6, 8 and 9 of the CS were well reported and easily reproducible. Searches covered a broad range of databases and grey literature resources. The ERG expressed some concerns regarding the use of study design filters. However, it is unlikely that this would have impacted on the overall recall of results. In terms of the *de novo* model, there are a number of areas of uncertainty regarding the biomarker component of the model that the ERG is concerned about. Firstly, the aggregation of two different health states (CC and abnormal TB) into one severe risk health state could be problematic, as the transition probability to the DCC state (based on the literature) may refer to CC patients only. The ERG consulted with a clinical expert who stated that the aggregation of these two health states was sensible in PBC patients because liver histology is rarely known. Secondly, the justification of no additional progression from low-risk or moderate risk to severe-risk after the first year is questionable. The company claims that this is consistent with experience with patients responding to UDCA treatment. Although the progression to DCC is higher for UDCA non-responders than for UDCA responders (which is the justification provided by the company), there is evidence for progression to DCC in both groups, so assuming no progression after 12 months might not be justified by these data. Moreover, the cited evidence stems from UDCA responders treated with UDCA, hence it might be questioned whether this would also be applicable to PBC patients who are intolerant to or have inadequate response to treatment with UDCA and are treated with OCA. Nevertheless, the clinical expert consulted by the ERG stated that this assumption was reasonable given that 'we know that patients with a response to UDCA and normal (or below 1.67x ULN ALP) have an excellent long term prognosis with no overall impact on life expectancy'. The liver disease component is similar to other assessments of liver disease treatments and seems appropriate for the current decision problem. However, the company introduced an additional health state, not typically considered in other liver disease models: the pre-liver transplant health state, without justifying why this was necessary. While the ERG acknowledges that the introduction of the pre-liver transplant health state has face validity as being a waiting list state, the ERG is concerned that it groups patients together that came from different health states (HCC, DCC, severe risk) and who therefore may experience different transition probabilities, health-related quality of life and/or costs.

The population represented in the cost effectiveness model seems to correspond to the expected licensed indication and the final scope issued by NICE for the current decision problem. However, the ERG considers that the proportion of patients entering the model in the moderate and severe risk health states should be based on data from POISE. In the company's model, the proportion of patients entering the model in the severe risk health state (23.15%) is much larger than the proportion of patients in the POISE study who are in the severe risk health state at baseline (8.42%). This could potentially bias model outcomes in favour of OCA treatment. Therefore, the ERG preferred to use the proportions obtained from the POISE study.

Regarding the estimation of treatment effectiveness in the biomarker component of the model, the main ERG concern is that the company uses the data from the POISE trial to estimate transition probabilities for the patients receiving OCA (OCA titration arm), but not for patients receiving UDCA only and no treatment. Moreover, the methods used by the company to estimate treatment effectiveness for the patients receiving UDCA only and no treatment are lacking transparency and justification and are unclear to the ERG. The ERG thinks very strong arguments must be presented to warrant favouring observational evidence over randomised evidence of comparative effectiveness (i.e. the POISE trial). Finally, the ERG noticed a discrepancy between the transition probabilities reported in the CS and those used in the economic model.

The company refers to TA330 for multiple transition probabilities. The ERG generally agrees with the approach used in TA330 for modelling liver disease. It should be noted that not all the transition probabilities of the liver disease component, used in the CS, are fully consistent with TA330. The company did not provide justifications for the sources that were used, nor for the deviations from the transition probabilities used in TA330. Therefore, the ERG prefers to use transition probabilities consistent with TA330 for the liver disease component of the model, this also entails excluding the pre liver transplant state. The only exception is the transition to HCC; here the source provided by the company was preferred as this was based on PBC patients.

Health state utility values for the low risk and moderate risk health states of the biomarker component of the model were estimated based on 35 patients who had either PBC or primary sclerosing cholangitis using the Health Utilities Index Mark 2 (HUI:2) The HUI:2 measurements were valued based on a sample of Canadian parents. This is not consistent with the NICE reference case and the representativeness of this estimate to the UK setting is uncertain. Given that a recent review reported that 34.1% of PBC patients have a 'poor' quality of life and that the current estimate is higher than the general population utility values for the age category 50-59, the ERG thinks that the utility values for these health states might be overestimated. In the absence of representative utility data specific to UK PBC patients, the ERG prefers to use the age-dependent utility values of the UK general population for these two health states in its base-case analysis. Based on expert opinion, the company applied a

utility decrement on the utility values of the liver disease component health states, except for the HCC health state. Given the lack of details, the ERG was not able to investigate the validity of this decrement. Younossi et al. 2001 report a lower utility value for chronic liver disease due to viral infection (e.g. HCV/HBV) than the utility value reported for chronic liver disease due to cholestatic disease (PBC and primary sclerosing cholangitis). This is contradictory to the company's reasoning that PBC patients have lower health state utility values than HCV/HBV patients. For these reasons, the ERG prefers to remove the **Intervent** utility decrements.

The company used expert opinion for estimating costs associated with outpatient visits. The ERG prefers to use the NHS reference costs. The company assumed, without adequate justification, that the cost associated with the severe risk health state of abnormal TB and CC are half the costs of DCC. The ERG prefers to use the cost for CC used in TA 330. Finally, the ERG regarded the estimation of liver transplant costs uncertain, and explored alternative estimates in an exploratory analysis.

The lack of transparency, regarding the justification and detail of the methods used in the CS as well as regarding the economic model submitted by the company, is an area of concern. Given this lack of transparency, also after multiple requests from the ERG, the methods used by the company are still unclear. The ERG was therefore unable to assess the validity and appropriateness of these methods. Given the lack of transparency in the submission by the company, the ERG was unable to explore the whole economic model for programming errors (although the ERG was able to replicate the Markov trace) nor to assess whether all methods used by the company conformed to best practices. Therefore,

the cost effectiveness results (presented by both the company and the ERG) should be interpreted with caution.

1.6 ERG commentary on the robustness of evidence submitted by the company

1.6.1 Strengths

The company used systematic review methods to identify the evidence on obeticholic acid for primary biliary cirrhosis. The majority of searches in the CS were well documented and easily reproducible. Searches were carried out on a broad range of resources including those recommended in the NICE 2013 guide to the methods of technology appraisal. Supplementary searches of conference proceedings and other relevant resources including trials databases, HTA agencies, specialist and organisational websites, and the checking of references lists were undertaken by the company in order to identify additional studies not retrieved by the main searches.

The main evidence was based on a well-conducted international randomised controlled trial of 217 patients with a 12 month follow up.

1.6.2 Weaknesses and areas of uncertainty

The company reported that no separate adverse events searches were undertaken as the clinical effectiveness searches were used to inform the adverse events section. However these searches contained a methodological filter intended to limit the search to RCTs. Guidance by the Centre for Reviews and Dissemination (CRD) recommends that if searches have been limited by a study design filter, additional searches should be undertaken to ensure that adverse events that are long-term, rare or unanticipated are not missed.

The main limitation of the evidence identified is that the main trial POISE is based on surrogate outcomes. The ERG is satisfied that the company has demonstrated evidence of some correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease is unclear. Furthermore, the relative efficacy of OCA and fibrates is unknown. The role of OCA as monotherapy in patients intolerant to UDCA and in those with more advanced disease is still unclear.

The lack of transparency, justification and details regarding the methods used in the CS are the main weaknesses in the cost effectiveness chapter and the model submitted by the company. Transparency is a key quality aspect of modelling. The lack of transparency hampered the validity check by the ERG. Additionally, the inability of external validation, relatively short trial follow-up using intermediate outcomes and reliance on non PBC sources in the model stress the uncertainty in the estimation of long-term outcomes.

1.7 Summary of exploratory and sensitivity analyses undertaken by the ERG

The main areas of ERG concern were not using POISE trial data directly for the non-OCA regimen, lack of transparency and justification regarding calibration methods, data and assumptions to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes, implausible high utility values in the biomarker component of the model and lack of justification for the **MRQoL** decrement in the liver disease component of the model. The ERG has incorporated various adjustments to the company base-case. The ERG base-case resulted in ICERs (probabilistic) of **Mathematical patients** respectively. The most influential adjustments/corrections made by the ERG were estimating transition probabilities using data from the POISE trial for the non-OCA regimens, and using age dependent utilities for the low and moderate risk health states. The ERG base-case ICER can be regarded as a lower bound, as transition probabilities after 12 months and the

calibration method used by the company are still considered highly uncertain, and exploratory analyses showed that alternative assumptions resulted in substantially higher ICERs (ranging from

for UDCA inadequate responders and UDCA intolerant patients

respectively).

As the methods used by the company to estimate and extrapolate treatment effectiveness are lacking transparency and justification, the ICERs (presented by both the company and the ERG) should be interpreted with caution. Given the lack of long-term results of the POISE trial, the ERG would generally agree with using a calibration approach (as adopted by the company), to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes. However, the assumptions, methods and data used should be documented in detail to avoid 'black box' criticism, and should be validated. This is particularly the case, considering that the use of alternative assumptions, methods and data resulted in substantially higher ICERs, as illustrated in the exploratory analyses by the ERG.

2. BACKGROUND

This report provides a review of the evidence submitted by Intercept Pharmaceuticals in support of obeticholic acid (trade name OCALIVA[®]) a bile acid preparation for the treatment of people with primary biliary cirrhosis (PBC) whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid. In addition to the main company submission (CS), the ERG received a submission from Professor Gideon Hirschfield on behalf of the British Society of Gastroenterology/Royal College of Physicians¹ and from Professor Graeme Alexander on behalf of the British Association for the Study of the Liver (BASL).²

The background section of this report by the Evidence Review Group (ERG) outlines and critiques the company's description of the underlying health problem and the overview of current service provision. The information is largely based on Section 3 of the CS with subsections referenced as appropriate.³

2.1 Critique of company's description of underlying health problem.

The underlying health problem is primary biliary cirrhosis (recently renamed as primary biliary cholangitis). Throughout this report we will refer to the disease as primary biliary cirrhosis as this is the terminology used in the scope issued by the National Institute for Health and Care Excellence (NICE)⁴ and in the CS.³

Section 3.1 of the company submission describes primary biliary cirrhosis as a 'rare, progressive, autoimmune, non-viral disease of the liver that gradually destroys the interlobular bile ducts. This causes an accumulation of cytotoxic bile acids in the liver which leads to inflammation, liver fibrosis, cirrhosis, and ultimately liver failure. '³ The CS further states that 'The final stages of PBC – cirrhosis and hepatic decompensation – is terminal unless a liver transplant is performed. '³ The risks of liver transplantation are outlined and the CS states that 'up to 43% of patients will have a recurrence within 15 years. '³

The CS states that 'The estimated prevalence of PBC in the UK is approximately 3.9 per 10,000 population, equating to approximately 19,175 people in England⁵ and making it a rare disease.'³ The company further state that 'Approximately 90% of people with the condition are women, and age of diagnosis is typically between 30 and 65 years⁶.'

The company notes that 'Approximately 60–80% of patients with PBC are asymptomatic at diagnosis.⁷ The diagnosis of PBC in asymptomatic patients is usually established after the chance finding of an elevated ALP level during the course of an unrelated illness^{7, 8}. '³The CS notes the need for early diagnosis and prompt treatment of patients early in the course of the disease to help prevent or to slow progression and to avoid or delay later complications of PBC. The CS notes that 'When treatment is delayed until PBC has progressed, survival is significantly worse than in the general population.'³ This statement was supported by reference to a Dutch cohort study of patients receiving ursodeoxycholic acid (UDCA).⁹

The CS notes that prognosis of the disease is unpredictable and varies from patient to patient. They state that *'there is currently no predictor to indicate which patients will progress slowly or rapidly, although patients with earlier age of onset and/or of male sex often have more aggressive disease that is refractory to existing treatment.*^{10 '3} This statement was supported by reference to an observational study of the UK-PBC Research Cohort comprising 2,353 patients.¹⁰

However the CS states that 'Analysis from a large research group (the Global PBC study group) shows that in patients with PBC, ALP and bilirubin levels strongly correlate with death and liver transplantation, with a combination of both variables improving prognostic prediction for patients¹¹'³

The supporting study was a meta-analysis of 4,845 patients diagnosed with PBC from 1959 to 2012 with a median follow-up of 7.3 years. It included both UDCA treated and non-treated patients.¹¹

The CS states that 'PBC has a substantial detrimental impact on quality of life, and HRQoL impairment is correlated with the severity of the disease. ^{'3} They cite a number of studies including a UK cohort study of 2,353 patients¹² in which '35% reported impairment of HRQoL compared with 6% of healthy controls (p<0.001), and 46% rated their overall health as 'fair' or 'poor' compared with 15% of healthy controls (p<0.0001).¹² ^{'3}

In addition to its impact on patients, the burden of the disease from the healthcare perspective is outlined. '*PBC is also associated with considerable healthcare costs. In 2014/15 there were 707 hospital admissions in England for PBC (ICD10 K74.3), accounting for 963 consultant episodes and 3,767 bed days^{7.,3} They further note that '<i>PBC is one of the most frequent indications for liver transplantation in Europe....There were 621 elective liver transplants performed in the UK in 2014/2015, of which at least 7% were for PBC.*^{13,3}

ERG comment:

- The references cited by the company were checked. The underlying health problem was considered to be appropriately described and appropriate references cited.
- The impact on patient quality of life was appropriately highlighted. The ERG notes that whilst the impact of PBC may be limited in early-diagnosed UDCA-responsive patients, among those who are unresponsive highly disabling symptoms such as fatigue and pruritus are common and patients face possible progression into cirrhosis and ultimately liver failure.

2.2 Critique of company's overview of current service provision

The company states that no NICE guidance or pathway specific to PBC is available.³

The CS states that 'Patients are commonly asymptomatic at diagnosis, and are referred to secondary care on discovering abnormal liver function and / or positive antibody by blood tests at a GP visit for an unrelated illness.^{7, 8} Occasionally patients are referred internally, most commonly from a rheumatologist. Rarely, patients are referred due to pruritus, abnormal ultrasound, or decompensation (ascites). '³

The company further state that '*The only licensed treatment for PBC is ursodeoxycholic acid* ((UDCA). '³ The company states that '*On diagnosis of PBC, UDCA is prescribed at 13-15mg/kg/day. Patients are monitored at 3-4 months for tolerability, and at 6 and 12 months to gauge response and compliance to therapy.* '³ The company cites two relevant clinical guidelines.^{14, 15} Both recommend long-term therapy at this dosage of UDCA as a first line of treatment.^{14, 15}

The company states that 'up to 74% of patients have an incomplete response to UDCA and there are currently no available licensed or effective treatment options for these patients.'³ The company states that 'There are several response criteria that have been proposed to define non-response / progression...; however, there is no consensus as to which of these criteria should be used.'³

The indication for obeticholic acid is for 'the treatment of PBC in combination with UDCA in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA' (Section 2.2.2 of the CS)³ Therefore the place of obeticholic acid in the clinical pathway for PBC would be as a second-line treatment.

The CS states that '*There are no licensed or effective drugs approved for second-line treatment for second-line treatment for the management of patients with an inadequate response to, or intolerance to, UDCA.*^{'3} They further state that 'other treatments have been trialled for use in PBC, including

budesonide and fibrates (which are contraindicated in PBC).[CS refs 22,23] However, limited efficacy has been observed in these other treatments. '³

The CS states that 'Liver transplantation is the only treatment for patients with late-stage PBC, where UDCA has limited efficacy.¹⁴' The CS outlines the challenges and risks of liver transplantation and notes that 'up to 43% of patients will have a recurrence of PBC within 15 years¹⁶. Draft guidance from the BSG note that there is no consensus on routine use of UDCA post-transplant.¹⁵'

The CS states that '*No additional tests or investigations are required for OCA treatment.*'³ Although OCA treatment is to be initiated by specialists in the treatment of PBC, the company states that no additional infrastructure will be required for treatment with OCA. However patients are required to undergo a consultation six months after treatment initiation to assess tolerability and determine if the dose should be increased to 10 mg to achieve optimal response.³

ERG comment:

- The company correctly states that there is no specific NICE guidance on PBC. The CS cites two clinical guidelines by the European Association for the Study of the Liver (EASL) published in 2009¹⁴ and the more up-to-date but as yet unpublished British Society of Gastroenterology (BSG) guidelines.¹⁵ The CS correctly states that UDCA is the only licensed treatment at first line for PBC.
- Response rates to UDCA between 20 and 70% have been identified.⁴ The company correctly highlight that determining response rates to UDCA depends on the specific criteria used to assess response. Several sets of criteria have been used across the research literature. We are advised by our clinical expert that in practice various systems may be used including simple clinical criteria.
- Clinical guidelines state that there are no second line agents when patients have failed to respond to UDCA.^{14, 15} OCA would therefore be the only agent available at second line. The company mentions other agents that have been investigated (fibrates are most relevant to this submission). The statement *'limited efficacy has been observed'* in relation to other treatments was not supported by any references.³ A further discussion of the role of fibrates as a comparator to OCA can be found in Section 3 of this report, the decision problem.

3. CRITIQUE OF COMPANY'S DEFINITION OF DECISION PROBLEM

The decision problem as presented in the CS is given in Table 3.1.

Table 3.1: Decision Problem

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
Population	People with primary biliary cirrhosis [†] whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid	As per scope	Not applicable
Intervention	OCA alone or in combination with UDCA	As per scope	OCA is taken in combination with UDCA for people whose disease has an inadequate response to UDCA, and as monotherapy in people who are unable to tolerate UDCA
Comparator(s)	 For people whose disease has an inadequate response to UDCA: UDCA alone or in combination with fibrates For people who are unable to tolerate UDCA: Fibrates No additional treatment 	 For people whose disease has an inadequate response to UDCA, the following comparators were considered: UDCA For people who are unable to tolerate UDCA, the following interventions were considered: Placebo 	Fibrates are not licensed in the UK, nor are they standard of care, and they are contraindicated in PBC ^{17, 18} . They are rarely used, with only Sector of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC).[CS24] Their efficacy is yet to be proven, with only a limited number of studies reporting results for the use of fibrates in PBC ¹⁹⁻²³ , with the following challenges: The studies were investigator-initiated and only had small patient numbers All but one study were conducted in Japanese patients In addition, there are significant safety concerns with the use of fibrates in PBC, with one study ²² reporting three deaths in

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
			13 patients in the UDCA + fibrates arm compared with no deaths in 14 patients in the UDCA monotherapy arm. In addition, one patient developed HCC in the fibrates + UDCA arm, compared with none in the UDCA monotherapy arm ²² .
Outcomes	 The outcome measures to be considered include: Mortality Liver function based on markers of liver biochemistry Symptoms, including pruritus, fatigue and abdominal pain Time to liver transplantation PBC-related events, including ascites, varices, encephalopathy and HCC Adverse effects of treatment HRQoL 	 Surrogate efficacy outcomes are included in POISE: Liver function biomarkers (ALP and bilirubin) Other biomarkers relevant to PBC (GGT, AST, ALT, FGF-19, CK-18 and bile acids) Inflammation biomarkers (CRP, TNF-α, TGF-β and IL-6) Non-invasive evaluations of fibrosis (ELF and FibroScan[®] TE) 	Due to the rare and chronic nature of PBC and the slow progression in most patients, a long-term trial is required to capture clinical outcomes such as mortality, transplant-free survival, and the incidence of complications. The primary outcome measured in POISE related to combined ALP and bilirubin levels, which have both been shown to be strongly correlated with disease prognosis ^{11, 24, 25} . Other biomarkers relevant to PBC, inflammation biomarkers, and non-invasive evaluations of fibrosis have been included to further support changes in disease progression. There is currently a long-term Phase 3b trial ongoing, COBALT, that aims to capture clinical outcomes and should support the longer-term impact of OCA on PBC already shown in POISE.
Economic analysis	The reference case stipulates that the cost effectiveness of treatments should be expressed in terms of incremental cost per quality-adjusted life year. The reference case stipulates that the time horizon for estimating clinical and cost effectiveness should be sufficiently long to	As per scope	Not applicable

	Final scope issued by NICE	Decision problem addressed in the company submission	Rationale if different from the final NICE scope
	reflect any differences in costs or outcomes between the technologies being compared. Costs will be considered from an NHS and personal social services perspective.		
Subgroups to be considered	Not applicable	Not applicable	Not applicable
Special considerations including issues related to equity or equality	Guidance will only be issued in accordance with the marketing authorisation. Where the wording of the therapeutic indication does not include specific treatment combinations, guidance will be issued only in the context of the evidence that has underpinned the marketing authorisation granted by the regulator.	Equality document	People with PBC face stigma in society because of the negative connotations of the term 'cirrhosis' and the association with alcoholism and drug abuse ²⁶ . In addition, PBC is a rare disease affecting mainly women, and it is essential that patients have the same opportunities to gain access to new treatments.

Source: Table 1 of the CS³

Footnote: [†]Note that primary biliary cirrhosis has recently undergone a name change to primary biliary cholangitis. At the time of consultation with NICE, the official name was primary biliary cirrhosis and, as such, this is reflected in the table.

ALP, alkaline phosphatase; ALT, alanine transaminase; AST, aspartate transaminase; CK-18, cytokeratin-18; CRP, C-reactive protein; ELF, enhanced liver fibrosis; FGF-19, fibroblast growth factor-19; GGT, gamma-glutamyl transpeptidase; HRQoL, health-related quality of life; HCC, hepatocellular carcinoma; IL-6, interleukin-6; NHS, National Health Service; NICE, National Institute for Health and Care Excellence; OCA, obeticholic acid; OD, once daily; PBC, primary biliary cholangitis/cirrhosis; TE, transient elastography; TGF, transforming growth factor; TNF, tumour necrosis factor; UDCA, ursodeoxycholic acid

3.1 Population

The population described in the scope issued by NICE was '*People with primary biliary cirrhosis whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid.*⁴

ERG comment:

• The CS matches the scope. However it should be noted that the main evidence submitted by the company is the POISE randomised controlled trial.²⁷ This trial has 11 patients (7%) who are unable to tolerate ursodeoxycholic acid and who are receiving obeticholic acid alone. Hence it is not representative of this patient group and results may not be reliable for this group. However based on clinical advice we have received, the percentage of patients intolerant to UDCA in practice is low, approximately 5%.

3.2 Intervention

The intervention described in the scope was 'obeticholic acid alone or in combination with ursodeoxycholic acid'.⁴

Obeticholic acid is a bile acid preparation (Anatomical Therapeutic Chemical Classification System (ATC) code A05AA04). By activating the farnesoid X receptor (FXR), OCA is expected to reduce the production of bile in the liver, thus reducing the exposure of the liver to toxic levels of bile acids. It is marketed as Ocaliva and was designated as an orphan medicinal product on 27 July 2010.

At the time of writing the submission, the company was awaiting European marketing approval for obeticholic acid. On 13 October 2016 the Committee for Medicinal Products for Human Use (CHMP) adopted a positive opinion.²⁸ The full indication was 'for the treatment of primary biliary cholangitis (also known as primary biliary cirrhosis) in combination with ursodeoxycholic acid (UDCA) in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA.' The committee further noted 'The benefits with Ocaliva are its ability to reduce alkaline phosphatase and bilirubin levels in adults with primary biliary cholangitis. This is likely to lead to clinical benefits for the patient such as delayed development of liver fibrosis, cirrhosis liver transplant and death. However, this remains to be formally demonstrated by means of the post-authorisation follow up within this conditional marketing authorisation.'²⁸

Obeticholic acid will be 'provided as a film-coated tablet containing 5 mg or 10 mg OCA. The recommended starting dose is 5 mg taken orally, once daily. Based on the assessment of tolerability after 6 months, the dose should be increased to 10 mg once daily to achieve optimal response.³.

The CS states that 'Patients should continue to take OCA for as long as the patient continues to benefit from treatment'. For patients experiencing severe intolerability due to pruritus the company advise dose reductions, dose interruptions for up to two weeks or gradual increase to achieve optimal response or discontinuation for those who continue to experience persistent intolerable pruritus (CS Table 5).³.

For those taking UDCA concomitantly with obsticholic acid, the company state that no dose adjustment of UDCA is required.³.

ERG comment:

- The CS matches the scope in that it evaluates OCA alone or in combination with UDCA.
- However, as stated above, the main efficacy trial POISE has just 11 patients (7%) who are taking obeticholic acid alone. The remaining 93% are taking it in combination with UDCA.²⁷ One of the two Phase 2 studies included in the submission as supporting evidence was a study of monotherapy including 49 patients, 20 of whom received 10 mg OCA and 23 received

placebo. The remainder received 50 mg OCA and were not relevant to the proposed dosage.²⁹ As stated above, the role of OCA monotherapy at the appropriate dosage has only been investigated in a small number of patients so results for this group of patients may not be reliable.

• In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo.²⁷ Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six months period. The patients in POISE were only up-titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response.²⁷ The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg.²⁷ The company notes that *'further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.* '³ This appears to be reasonable but is an assumption.

3.3 Comparators

For people whose disease has an inadequate response to UDCA the CS evaluates the use of UDCA alone as a comparator as per the final scope issued by NICE. For people who are unable to tolerate UDCA the CS (more specifically the POISE trial) compared obsticholic acid alone to placebo (representing no additional treatment as stated in the NICE scope). As stated above, 11 patients (7%) took obsticholic acid alone in POISE (five in the OCA titration group and six in the OCA 10 mg fixed dose group and five patients received a placebo alone). The CS presents results for this comparison but notes that these *'should be interpreted with caution due to low patient numbers.* '³

However the comparators addressed in the CS differ from those in the final scope issued by NICE in that fibrates have not been included as a comparator for people whose disease has an inadequate response to UDCA nor for those who are intolerant to UDCA. The company provide several justifications for this including: fibrates not being licensed for this indication in the UK, not being standard care, they are contraindicated in PBC, have been rarely used in the UK and efficacy is as yet unproven with a small number of limited studies with some safety concerns.³

ERG comment:

- As stated above, the role of OCA monotherapy compared to placebo (no treatment) has only been investigated in a small number of patients so results for this group may be less reliable than those taking obeticholic acid in combination with UDCA.
- The omission of fibrates from the comparators in the decision problem was investigated by the ERG in several ways. We consulted clinical experts and identified systematic reviews of fibrates as the most reliable source of up to date evidence. Our response to the omission of fibrates is given in the table below.

Assertion in the company submission	ERG comments
'Fibrates are not licensed in the UK, nor are they standard of care.' 'They are rarely used, with only of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC)' ³	Fibrates are not licensed for this indication. We are advised that fibrates are not routinely used in the UK for PBC.

Table 3.2: Response to the omission of fibrates as a comparator in the submission

Assertion in the company submission	ERG comments
<i>'Fibrates arecontraindicated in</i> <i>PBC</i> ' ³	The summary of product characteristics ^{17, 18} were checked. Some qualification of this statement is necessary. Hepatic insufficiency (including biliary cirrhosis) is mentioned as a contra-indication for fenofibrate, but the summary of product characteristics also states that patients with hepatic disease have not been studied. Significant hepatic disease is mentioned as a contraindication for bezafibrate, due to the fact that it alters the composition of bile, and that there have been isolated reports of the development of gallstones. The ERG judges that prevention of progressive disease might well outweigh the risk of gallstone disease, a risk that can be managed with cholecystectomy.
'Their efficacy is yet to be proven with only a limited number of studies reporting results for the use of fibrates in PBC ^{3, 19-23}	The ERG asked the company to 'comment on the clinical trial evidence for fibrates in PBC (ideally this evidence would be sourced systematically, but it is acknowledged that this might not be feasible in the timeframe). ³⁰ The company referred us to a Cochrane review ³¹ that concluded 'treatment of primary biliary cirrhosis with bezafibrate can neither be supported nor refuted based on the best current evidence available ensuing from trials in Japanese patients. ³² . On examination of the review this conclusion related to the effect of fibrates on mortality, liver morbidity, adverse events, pruritus, and fatigue. The Cochrane review conclusion also stated: 'Bezafibrate seems to have an effect on decreasing the activity of serum alkaline phosphatase compared with no intervention or with UDCA in patients with primary biliary cirrhosis ^{'31} The ERG identified further systematic reviews of fibrates ³³⁻³⁵ including fenofibrates. We agree that the evidence for efficacy on hard clinical outcomes is limited. However improvements in surrogate outcomes have been noted across the reviews. We are aware of at least one ongoing trial in fibrates for PBC. ^{36*}
'In addition, there are significant safety concerns with the use of fibrates in PBC'	One study (a RCT) was cited as evidence of safety concerns regarding fibrates. ²² The safety concerns raised by this study are valid. However the ERG noted that evidence for the safety of fibrates in PBC was not gathered systematically.

Source: CS³

Footnote: * This trial is due to complete data collection for the primary outcome in December 2016. One hundred patients were randomised to bezafibrate 400 mg/day or placebo, UDCA was continued. Primary outcome is complete biochemical response at 24 months and the normalisation of hepatic biochemical tests. ⁽³⁶⁾ ERG, Evidence Review Group; PBC, primary biliary cirrhosis; RCT, randomised controlled trial; UDCA, ursodeoxycholic acid; UK, United Kingdom

Overall, although fibrates may not be widely used across the UK, the ERG believes that the decision to exclude them as a comparator may not be appropriate.

3.4 Outcomes

The final scope issued by NICE specified patient outcomes including mortality, symptoms including pruritus, fatigue and abdominal pain, time to liver transplantation, PBC-related events, health-related quality of life and adverse effects of treatment.⁴ However all efficacy outcomes in the POISE trial which forms the main evidence of the submission relate to surrogate outcomes.³ These include a range of liver function biomarkers and other biomarkers related to PBC. Disease-specific quality of life based on the PBC-40 tool was collected in POISE but was not specifically mentioned as being addressed in the decision problem. Adverse events were also collected in POISE.³

The CS explains that '*due to the rare and chronic nature of PBC and the slow progression in most patients, a long-term trial is required to capture clinical outcomes such as mortality, transplant-free survival, and the incidence of complications.*'³ The POISE trial is of 12 months' duration although there is an ongoing five year long-term safety extension (LTSE). The LTSE has data from 12 months but efficacy is still based on surrogate outcomes.³ The company cite three references to provide evidence for the correlation of the primary outcome of POISE (combined alkaline phosphatise (ALP) and bilirubin levels) to disease prognosis.^{11, 24, 25}

The company also note that there is an ongoing trial of obeticholic acid (COBALT) that measures clinical outcomes. Details of the COBALT trial are provided in Table 39 of the CS.³ Briefly, this is a double-blind randomised, placebo-controlled trial of obeticholic acid on clinical outcomes in patients with PBC. The estimated enrolment is 350 across up to 170 international sites. Duration is estimated to be approximately eight years according to the time to accrue approximately 121 primary endpoint events. The primary outcome is a composite endpoint of clinical events. The start date of the trial was December 2014 and the study completion date is April 2023.³

ERG comment:

- The ERG draws the attention of the committee to the fact that the main evidence in the submission relates to surrogate outcomes. The ERG did not have the means to systematically review the validity of this correlation. Instead the ERG investigated the three references cited by the company as evidence of the correlation. The ERG observed that one of the references cited was a meta-analysis of individual patient data (4,845 patients) which concluded that *'Levels of alkaline phosphatase and bilirubin can predict outcomes (liver transplantation or death) of patients with PBC and might be used as surrogate end points in therapy trials.*^{'11} This finding was supported by a retrospective review of 73 patients treated with UDCA over 36 months.²⁴ Biochemical response predicted histological progression, assessed with paired biopsies about 10 years apart, as well.²⁵ The ERG is satisfied that the company demonstrated evidence of some correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease are unclear.
- The definitive effects of obeticholic acid on clinical outcomes relevant to patients awaits the results of the COBALT trial.³⁷
- The ERG asked the company if any interim data were available from the COBALT trial which investigates clinical outcomes. The company confirmed that *'there are no interim data available and no interim analysis is planned since COBALT is an events-driven trial and 121 primary endpoint events need to have occurred before the trial will report and analyses will be performed.'* ³²

3.5 Other relevant factors

The company highlights that PBC is a rare disease affecting mainly women as a consideration relating to equity.³ The company has offered a Patient Access Scheme (PAS) as part of this submission. Details of the scheme were not available in the submission.³

4. CLINICAL EFFECTIVENESS

4.1 Critique of the methods of review(s)

The CS states that 'A systematic review was conducted to identify all relevant evidence for the efficacy and safety of interventions used to treat PBC. '³ This systematic review is discussed in this section.

4.1.1 Searches

The following paragraphs contain summaries and critiques of all searches related to clinical effectiveness presented in the company submission. The Canadian Agency for Drugs and Technologies in Health (CADTH) evidence based checklist for the Peer Review of Electronic Search Strategies (PRESS), was used to inform this critique.³⁸ The submission was checked against the Single Technology Appraisal (STA) specification for company/sponsor submission of evidence.³⁹ The ERG has presented only the major limitations of each search strategy in the report.

Systematic literature review (CS Section 4.1.2)

The company submission stated that searches were originally undertaken in September 2014 and updated in September 2015 and June 2016. Searches were reported for Embase, MEDLINE, MEDLINE in-Process, Cochrane's CENTRAL, DARE and CDSR databases. An additional grey literature search was reported on the following resources: The trials registry, Clinical trials.gov. meeting and conference papers for the following: American Association for the study of Liver Diseases (AASLD), European Association for the Study of the Liver (EASL), American College of Gastroenterology (ACG), Digestive Disease Week (DDW), United European Gastroenterology Week (UEGW), Canadian Digestive Disease Week (CDDW) and the Japan Digestive Disease Week. As well as browsing websites for Liver Foundation UK, The Foundation for Liver Research, American Liver Foundation, Canadian Liver Foundation and the British Liver Trust. The bibliographies of selected articles were also checked to identify any further studies missed by the electronic searches or the Cochrane systematic literature reviews

The CS reported that searches were designed to retrieve RCTs for the population PBC, the study design filter was based on the BMJ Evidence Centre filter and the population terms were informed by those utilised in the Cochrane review of Ursodeoxycholic Acid in Primary Biliary Cirrhosis. The company reported that the terminology for PBC changed in 2015 when the name primary biliary cirrhosis was replaced by the term primary biliary cholangitis. This change was reflected in the 2016 update searches.

ERG comment: The database searches were clearly structured and documented. No language limits were applied. In their response to clarification the company confirmed that the date span for each resource was the same for those reported for the cost effectiveness and HRQoL searches. Searches were conducted over a broad range of bibliographic databases and grey literature resources, and a recognised RCT filter was referenced.

For the original 2014 systematic literature review (SLR) and 2015 update searches, the company conducted a single search of Embase.com on the understanding that it now contains all MEDLINE and MEDLINE In-Process content. This was replaced by separate searches of Embase and MEDLINE via the Ovid interface for the subsequent 2016 update. Whilst the ERG accepts this single approach as being adequate, the ERG considers it preferable to conduct a separate companion MEDLINE search in order to fully utilise the power of database specific study design filters developed to make the most of an individual databases subject headings. However given the searches of additional bibliographic databases and grey literature resources reported by company, it is unlikely that this omission would have impacted on the overall recall of results. The ERG noted that the same approach to the search of
Medline content via Embase.com was adopted for all of the original literature searches reported in Appendices 6, 8 and 9, therefore the same limitations will have applied.

Adverse Events

Section 4.12 of the company submission states that all safety data reported in this section were derived from the POISE study, with supporting safety data obtained from two Phase 2 trials. No searches for adverse events were reported.

ERG comment: The ERG queried this lack of searches in their points of clarification and the company confirmed '*No separate searches for adverse events were conducted, since adverse events and any safety data for OCA would be included in the evidence identified in the clinical systematic literature review*'³² The clinical effectiveness searches incorporated a methodological filter intended to limit the search to specific study designs, namely RCTs. Guidance by the Centre for Reviews and Dissemination (CRD)⁴⁰ recommends that if searches have been limited by a study design filter, additional searches should be undertaken to ensure that adverse events that are long-term, rare or unanticipated are not missed. The ERG considered that it was possible that some relevant evidence may not have been identified as a consequence of the study design limits used. The ERG was unable to undertake independent adverse events searches and review the results within the STA timeline, as this would be outside of the ERG remit.

4.1.2 Inclusion criteria

The eligibility criteria for the systematic review of the efficacy and safety of interventions used to treat PBC is given in Table 4.1. It was stated that study selection was *'performed by two independent reviewers and discrepancies were resolved by a third independent reviewer*^{'41}

	Inclusion	Exclusion
Population	Primary Biliary Cirrhosis (confirmed diagnosis as defined in international guidelines ^{42, 43}	Biliary Cirrhosis and any other liver condition Mixed populations of primary biliary cirrhosis and patients with other conditions
Interventions / comparators	Any intervention used to treat primary biliary cirrhosis	
Outcomes	 All-cause mortality All-cause mortality or liver transplantation Adverse events (serious and non-serious) Quality of Life Pruritus; number of patients with pruritus or pruritus score Fatigue: number of patients with fatigue Liver-related morbidity (number of patients who developed jaundice, portal hypertension, oesophageal varices, gastric varices, upper gastrointestinal haemorrhage, ascites, hepatic encephalopathy, hepato-renal syndrome). Biochemical markers: serum bilirubin, serum alkaline phosphatases, serum gamma- 	

Table 4.1: Eligibility criteria used in the clinical effectiveness and safety systematic review

	Inclusion	Exclusion			
	glutamyltransferase, serum aspartate aminotransferase, serum alanine aminotransferase, serum albumin, total cholesterol, plasma immunoglobulins, prothrombin index.				
	 Liver biopsy (or FibroScan[®]) findings: worsening of liver histological stage or score 				
	• Enhanced Liver Fibrosis (ELF) test				
	• Transient Elastography (TE)				
Study design	Randomised controlled trials (RCTs; including extension studies of RCTs where found)	• Meta-analysis and reviews			
		Non-randomised controlled trials			
Source: Table 2 of the CS Appendix 2 ⁴¹					
CS, company sub	mission; ELF, Enhanced Liver Fibrosis; RCT, randomised	l controlled trial; TE, Transient			

CS, company submission; ELF, Enhanced Liver Fibrosis; RCT, randomised controlled trial; TE, Transient Elastography

ERG comment:

- The population, intervention, comparators and outcomes reported in the above table from the CS reflect the scope issued by NICE.⁴ However the ERG had a number of concerns in relation to eligibility criteria for the review and the selection process used.
- The CS states that 'A systematic review was conducted to identify all relevant evidence for the efficacy and safety of interventions used to treat PBC. All randomised controlled trials investigating an intervention to treat PBC were included.' ³ However according to Appendix 2 of the submission, a post-hoc decision was made to exclude all randomised trials (RCTs) that did not include at least one obeticholic acid (OCA) treatment arm (monotherapy or combination).⁴¹ The ERG asked the company to explain the rationale for this decision.³⁰ The company responded 'When the protocol for the systematic literature review was designed, the strategy was to keep the criteria as broad as possible. However, for the current submission, it was clear that there were no other treatments licensed or being used in the UK for patients with an inadequate response to or intolerance to UDCA. Therefore, only RCTs that included at least one OCA arm were relevant for this submission.^{'41} The ERG identified from the flow chart (Figure 8 of the CS)³ that 36 publications had been excluded on this basis and asked the company to provide bibliographic details of the 36 studies. The list provided by the company in response to the letter of clarification was checked.⁴¹ and the ERG confirmed that studies including an obeticholic acid arm had been appropriately included. The decision to include only RCTs with at least one OCA arm meant that no indirect comparisons between OCA and fibrates could be made in the absence of direct evidence.
- The company identified nine Cochrane systematic literature reviews 'that had previously been performed to identify RCTs for interventions in PBC. The search strategies and scopes for these reviews were closely aligned to that of this systematic review, and therefore publications were removed from this review if they had previously been captured and reviewed by the Cochrane reviews to avoid duplication.'³ The ERG identified from the flow chart (Figure 8 of the CS) that 136 publications were rejected on the basis of being included in Cochrane reviews.³ The company provided a list of these publications in the response to clarification document.⁴¹ The list was checked and the ERG noted that no relevant direct evidence comparing obsticholic acid with one of the comparators in the NICE scope appeared to have been omitted.

- Section 4.11 of the CS states that '*There are no non-RCTs relevant to this submission*.'³ As the systematic review inclusion criteria specified RCTs only, the ERG queried whether searches had been conducted separately to determine the lack of non-RCT evidence.³⁰ The company stated that '*No non-RCT searches were conducted, since head-to-head RCT data (more robust than non-RCT data) were available for OCA vs UDCA in POISE*.'⁴¹ The statement that '*There are no non-RCTs relevant to this submission*' is not supported by a systematic review of the evidence. Therefore possibly relevant non-RCTs could have been missed in relation to effectiveness in the absence of direct evidence and in relation to safety.
- Section 4.12 of the CS states that 'all safety data reported in this section were derived from the POISE study, with supporting safety data obtained from two phase 2 trials'.³ The company confirmed that 'no separate searches for adverse events were conducted, since adverse events and any safety data for OCA would be included in the evidence identified in the clinical systematic literature review.'⁴¹ However the systematic review was limited to RCTs and, as stated above, the clinical effectiveness searches incorporated a methodological filter intended to limit the search to studies with this design. Guidance by the Centre for Reviews and Dissemination (CRD)⁴⁰ recommends that if searches have been limited by a study design filter, additional searches should be undertaken to ensure that adverse events that are long-term, rare or unanticipated are not missed. The ERG considered that it was possible that some relevant evidence on adverse events may not have been identified as a consequence of the study design limits used.
- The ERG asked the company to confirm that studies of mixed populations of PBC and patients with other conditions were excluded even if data were available for patients with PBC separately³⁰The company confirmed that '*Studies of mixed populations of PBC and patients with other conditions were excluded, even if data were available for patients with PBC separately.* '⁴¹ This exclusion criterion could have led to relevant data being omitted.
- Within the constraints of conducting the evaluation of the CS in a short timeframe, the ERG could not carry out further reviews of the evidence and search for missing studies. Overall the ERG is satisfied that the company has identified the RCTs directly comparing OCA with other treatments outlined in the scope issued by NICE.⁴ However this could have been supplemented by indirect evidence on fibrates to inform the NICE scope as no direct evidence is presented for this comparator and by searching for adverse events beyond the RCT literature.

4.1.3 Critique of data extraction

No specific details were provided in the CS regarding methods of data extraction of studies for the review of clinical effectiveness. It was unclear if more than one reviewer was involved in this process.

ERG comment: It is good practice to include details of methods of data extraction when reporting a systematic review, in order to ascertain that the review was carried out appropriately. As study selection was performed by two independent reviewers with discrepancies resolved by a third independent reviewer, it may be reasonable to assume that such procedures were followed for data extraction.

4.1.4 Quality assessment

Quality assessment of the three RCTs included in the submission was given in Appendix 3 of the CS.⁴¹ Elements assessed were randomisation procedures, allocation concealment, comparability of groups, blinding of care providers, patients and outcome assessors and drop out, selective reporting of outcomes and use of intention to treat analysis and appropriate methods for dealing with missing data.⁴¹

No specific details were provided in the CS regarding methods of quality assessment of studies for the review of effectiveness. It was unclear if more than one reviewer was involved in this process.

ERG comment: Study quality appeared to have been assessed using appropriate tools. However it is good practice to include details of methods of quality assessment when reporting a systematic review, in order to ascertain that the review was carried out appropriately. As study selection was performed by two independent reviewers with discrepancies resolved by a third independent reviewer, it may be reasonable to assume that such procedures were followed for quality assessment. The quality of the included trial, POISE, is discussed in Section 4.2.3.

4.1.5 Evidence synthesis

The CS states that '*There was only one relevant Phase III trial providing data for the efficacy of OCA in PBC, therefore a meta-analysis was not conducted.*'³

The CS also states in Section 4.10 of the submission that *'indirect and mixed treatment comparisons were not conducted*'.³

ERG comment:

- The ERG acknowledges that two Phase 2 trials were included as supporting evidence in the submission^{29, 44} but agrees that not pooling these with the main trial data from POISE²⁷ is reasonable. Both Phase 2 studies were of three months' duration only and not all patients received OCA at the licensed dose.^{29, 44} Results relevant to the licensed dose are presented in the CS for both trials.³
- The ERG queried whether the decision not to conduct indirect or mixed treatment comparisons was made a priori or as a result of a lack of evidence from the systematic review.³⁰ The company reiterated that '*there were no treatments licensed or being used in the UK for patients with an inadequate response or intolerance to UDCA, and therefore there was no relevant comparator to OCA other than UDCA, which has been compared head to head in POISE. Therefore, an indirect or mixed treatment comparison was not necessary*.²⁴¹
- The ERG draws to the attention of the committee that the decision not to perform indirect comparisons appears appropriate given the direct evidence comparing OCA and UDCA. However there is no direct evidence comparing OCA and fibrates.

4.2 Critique of trials of the technology of interest, their analysis and interpretation (and any standard meta-analyses of these)

4.2.1 Overview of the evidence in the submission

The evidence base for the clinical efficacy of obeticholic acid (OCA) for primary biliary cirrhosis, consists of one Phase 3 randomised controlled trial (RCT), POISE,⁴⁵ and two Phase 2 RCTs, ^{29, 44} as identified by the systematic literature review. POISE provides the main evidence with the two Phase 2 trials treated as supporting evidence.

The POISE trial was a 12-month, international, multicentre, placebo-controlled study with a randomised, parallel-group design in patients aged \geq 18 years with PBC who had previously failed to respond to treatment with ursodeoxycholic acid (UDCA) or were intolerant to UDCA.⁴⁵ The double-blind phase of the POISE trial lasted 12 months. A five-year extension phase of this trial was offered to participants of POISE willing to enrol.

The two Phase 2 supporting studies, 747-201 and 747-202 evaluated the efficacy, safety, and tolerability of OCA with UDCA (study 747-202)²⁹ or without UDCA (study 747-201)⁴⁴ versus placebo. Both studies included a double-blind phase and an open-label long-term safety extension phase. The double-blind phases have been completed. With regards to the intervention in both these studies, the CS states that they *included a 10mg OCA and an arm at higher doses [...] key efficacy and safety results were*

summarised only for subjects who received 10 mg OCA, since this is the upper limit for the licensed indication of OCA.³

An additional trial, COBALT (NCT02308111) is mentioned in the CS (Section 4.14 of the submission).³ This trial seeks to evaluate the effect on OCA on clinical outcomes such as transplant-free survival. The ERG can confirm that the trial is still ongoing and that no interim data has been made available.³⁷ Please see Section 3.4 of this report for a discussion relating to COBALT. In addition to this trial, another Phase 2 study (747-205) is currently ongoing in the US that includes patients from 8eightyears of age and aims to investigate the change from baseline in HDL metabolism.³

As stated in the CS, '*the results from the double-blind phase of POISE are presented as the main efficacy evidence and the two Phase 2 studies as supporting evidence*'. ³ In this regard, this report follows the same structure. More detail will be provided on the double blind phase of POISE with a briefer summary of the Phase 2 trials.

ERG comment:

- The ERG agrees that the POISE trial provides the main evidence for the comparison with UDCA or placebo and that the two Phase 2 trials represent supporting evidence.
- The ERG is satisfied that no data from the ongoing COBALT trial could have been used to inform the CS.
- The included trials did not compare OCA to fibrates as outlined in the scope.⁴

4.2.2 Overview of the POISE trial

An overview of the POISE trial and its extension study is presented in Table 4.2.

Trial no. (acronym)	Population	Intervention	Comparator
747-301 (POISE); NCT01473524	Patients diagnosed with PBC with ALP ≥1.67x ULN and/or total bilirubin >ULN but <2x ULN who fail to respond to or are intolerant to treatment with UDCA.	Oral OCA (5 mg or 10 mg) taken OD.	Placebo
LTSE to POISE	All participants who completed the 12-month double-blind phase of POISE and who were willing to enrol in the 5-year LTSE phase of the study.	Oral OCA (5– 25 mg ^a) taken OD	N/A

Table 4.2: Summary of the POISE trial and its extension study

Source: CS

Footnote: a) All patients initiated OCA at 5 mg OD; daily dose could be up-titrated if a satisfactory response was not achieved in 5 mg increments to a total dose of 25 mg OD (one increment per 3 months permitted), depending on tolerability

ALP, alkaline phosphatase; LTSE, long-term safety extension; OCA, Obeticholic acid; OD, once daily; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

The main methodological features of the double-blind phase of the POISE trial are summarised in Table 4.3.

Trial Design	Phase 3, randomised double-blind, placebo-controlled, parallel group trial. Stratified randomisation according to:		
	- higher risk of developing clinical outcomes i.e. Paris I criteria*		
	- intolerance to UDCA		
	- presence or absence of biochemical response to UDCA treatment		
Setting	59 sites in 13 countries. (7 sites in England and 2 sites in Scotland)		
Participants	217 were randomised (216 gave consent)		
Interventions	3 arms:		
	Placebo (with or without UDCA) $(n = 73)$		
	10 mg OCA, with or without UDCA ($n = 73$)		
	Titration (5mg OCA rising to 10 mg during months 6 to 12 if inadequate response to UDCA), with or without UDCA) ($n = 71$)		
Follow-up	12 months		
Primary Outcome	Percentage of participants in the 10 mg OCA fixed dose group at 12 month achieving the composite endpoint:		
	- ALP <1.67x ULN, and		
	- total bilirubin ≤ULN, and		
	- ALP decrease $\geq 15\%$ from baseline		
Source CS ³			
Footnote: * Paris I criteria defined as ALP $\leq 3x$ ULN and AST $\leq 2x$ ULN and total bilirubin ≤ 1 mg/dl			
(17 µmol/l)			
ALP, alkaline phosphatase; CS, company submission; TEAEs, Treatment-emergent adverse events; ULN,			
upper limit of normal			

Table 4.3: Overview of POISE trial

Table 4.4 details the secondary endpoints of the POISE trial and their definitions.

Study outcome	Definition
ALP response rates	Absolute and percentage change from baseline in ALP at Month 6 and Month 12
	Percentage of participants with a decrease in ALP from baseline of $\geq 10\%$, $\geq 15\%$, $\geq 20\%$, and $\geq 40\%$ at Week 2, Month 3, Month 6, Month 9 and Month 12, Percentage of participants with ALP \leq ULN, summarised by treatment at all post-baseline assessments
Biochemical treatment response criteria	 Percentage of participants meeting the Paris I, Paris II, Mayo II, Toronto II, or Rotterdam response criteria* at Week 2, Month 3, Month 6, Month 9, and Month 12. The analysis was repeated for subgroups of participants who met and did not meet the requirement of a responder at baseline for the endpoint analysed. Number and percentage of participants with: Normal bilirubin (≤ULN) and normal albumin (≥LLN), Moderate (bilirubin >ULN or albumin <lln), and<="" li=""> Severe (bilirubin >ULN and albumin <lln)< li=""> at baseline, Week 2, Month 3, Month 6, Month 9 and Month 12. </lln)<></lln),>
Clinical laboratory values	Defined as the absolute and percentage change from baseline in ALP, GGT, ALT, AST, total and conjugated (direct) bilirubin, albumin, and prothrombin time (PT) and international standardised ratio (INR) summarised by treatment group and visit.

Table 1 1. Summary	of secondary a	ndpoints of th	a daubla blind i	hasa of POISE
Table 4.4: Summary	of secondary e	napoints of th	e double-billia j	mase of POISE

Study outcome	Definition
Questionnaire PBC-40	Absolute change from baseline in six domains (cognitive, social, emotional function, fatigue, itch, and general symptoms) were summarised by treatment using descriptive statistics at Week 2, Month 3, Month 6, Month 9, and Month 12. A total score was not calculated.
Patient Research Questionnaire	'A simple patient research questionnaire was administered at Month 12, or at termination if the subject withdrew from the study prior to this, to request feedback about the subjects' perception of the study.' ³
Biomarkers and non-invasive assessments of liver fibrosis	Absolute change from baseline in the following markers: Markers of hepatic fibrosis, inflammation and other disease relevant biomarkers, including CRP, tumour necrosis factor- α (TNF- α), transforming growth factor β (TGF- β), fibroblast growth factor-19 (FGF-19), Interleukin-6 (IL-6), CK-18, autotaxin, and lysophosphatidic acid (LA), at Month 6 and Month 12 Enhanced liver fibrosis (ELF) score and its components hyaluronic acid (HA), procollagen-3 N-terminal peptide (P3NP), and tissue inhibitor of metalloproteinase 1 (TIMP-1) at Month 6 and Month 12 Hepatic stiffness measurements (at selected study sites) at Month 12, assessed by transient elastography (TE).
Bile acids	Absolute values and change from baseline by treatment group at 6 and 12 months' follow-up in the following: Total bile acids, total endogenous bile acids, and totals for the individual bile acids (UDCA, chenodeoxycholic acid [CDCA], deoxycholic acid [DCA], cholic acid [CA] and lithocholic acid [LCA]) and their respective conjugates Proportion of each of the individual bile acids relative to total bile acids
OCA Pharmacokinetics analysis	Values at Month 6 and Month 12 for OCA (unconjugated), glyco-OCA, tauro-OCA, and total OCA from participants who had a confirmed fasting of approximately 8 hours or more prior to their visit were included in the analysis. The effect of BAS on OCA and total bile acid concentration, as well as the percentage change in ALP, were explored as part of the PK analysis. Relationships between plasma total OCA concentrations (unconjugated and conjugated) and FGF-19 concentrations, endogenous bile acid concentrations, ALP and liver enzyme levels, and severity of pruritus were explored.
Safety	TEAEs defined as any adverse events (AEs) that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product. Additionally, Pruritus was measured by using the 5-D pruritus questionnaire and a VAS score at week 2, months 3, 6, 9 and 12.
Clinical laboratory evaluations	Physical examinations, vital signs, body weight and BMI, electrocardiongram, dual-emission x-ray absorptioametry (DEXA) scan of the femoral neck and lumbar spine, Mayo Risk Score to assess survival and MELD score to assess the severity of chronic liver disease were assessed at week 2, months 3, 6, 9, and 12.
Source: Section 4.3.6 Footnote: *Different re BAS, bile acids seques	of CS ³ esponse criteria studied in POISE are reported in Table 16 of the CS. strants; CS, company submission; TEAEs, Treatment-emergent adverse events

Assessment of clinical outcomes

The company stated in the clinical study report (CSR) that '*The incidence rate of clinical outcomes during the 12 month double-blind phase of Study 747-301 was expected to be low given the relatively early stages of disease in the enrolled patient population (total bilirubin <2x ULN). However, for completeness, the incidence of such events was retrospectively assessed based on the occurrence of predefined MedDRA Preferred Terms and the incidence of reaching a MELD score >15 (with a baseline MELD of <15'.⁴⁶*

The pre-determined preferred terms, which were used to define a clinical outcome included the following: death (all-cause); liver transplant; model of end stage liver disease (MELD) score ≥ 15 ; hospitalisation (as defined by a stay of 24 hours or greater) for new onset or recurrence of: variceal bleed, encephalopathy (as defined by a West Haven score of ≥ 2), spontaneous bacterial peritonitis (confirmed by diagnostic paracentesis, uncontrolled ascites (diuretic resistant ascites requiring therapeutic paracentesis at a frequency of at least twice in a month) and hepatocellular carcinoma confirmed by two complementary imaging modalities.⁴⁶

The company further stated that '*These events were not adjudicated by an independent committee as is standard for clinical outcomes trials that prospectively collect clinical outcomes.*'⁴⁶

The incidence of these events is reported in Section 4.2.4 of this report.

Table 4.5 describes the eligibility criteria for the POISE trial.

Table 4.5: POISE inclusion and exclusion criteria

Inclusion criteria	Exclusion criteria
Male or female aged ≥ 18 years	History or presence of other concomitant liver disease [§]
Definite/probable PBC diagnosis [†] as demonstrated by the	Clinical complications of PBC or clinically significant hepatic decompensation [¶]
presence of ≥ 2 of the following:	Severe pruritus or pruritus requiring systemic treatment (e.g. with BAS or rifampicin) within
• Elevated ALP levels for at least 6 months	2 months prior to randomisation
• Positive AMA titer or if AMA negative and/or low titer (<1:80) PBC specific antibodies (anti- GP210 and/or anti-SP100) and/or antibodies against the major M2 components (PDC-E2, 2- oxo-glutaric acid dehydrogenase complex)	Administration within 6 months prior to randomisation and throughout the study of azathioprine, colchicine, cyclosporine, methotrexate, mycophenolate mofetil, pentoxifylline; fenofibrate or other fibrates; budesonide and other systemic corticosteroids; potentially hepatotoxic drugs (including α -methyl-dopa, sodium valproic acid, isoniazide, or nitrofurantoin)
• Liver biopsy result consistent with PBC	Administration within 12 months prior to randomisation and throughout the study of
ALP \geq 1.67x ULN and/or total bilirubin \geq ULN but \leq 2x	antibodies or immunotherapy directed against interleukins or other cytokines or chemokines
ULN	Previous participation in a clinical trial using OCA
Taking UDCA for ≥ 12 months prior to randomisation	History or presence of clinically concerning cardiac arrhythmias, or prolongation of QT or
with a stable dose for ≥ 3 months, or no UDCA for	Q1c interval (>500 ms)
≥ 3 months prior to randomisation if unable to tolerate	Pregnancy or lactating
	History of HIV infection
Female participants to be post-menopausal, surgically sterile, or prepared to use ≥ 1 effective method of contraception during the study period and for 30 days after end of trial	Presence of any disease or condition that interferes with the absorption, distribution, metabolism, or excretion of drugs including bile salt metabolism in the intestine. Patients with inflammatory bowel disease or who have undergone gastric bypass procedures will be excluded (gastric lap band is acceptable)
	Medical conditions that could cause non-hepatic increases in ALP (e.g. Paget's disease) or that could diminish life expectancy to <2 years, including known cancers
	History of alcohol [‡] or other substance abuse within 1 year prior to randomisation
	Blood or plasma donation within 30 days prior to randomisation
	Mentally unable to complete a signed consent form

Source: Table 15 of the CS³

Footnotes: [†]Consistent with AASLD and EASL Practice Guidelines; [§]Hepatitis C virus infection, primary sclerosing cholangitis, alcoholic liver disease, definite autoimmune liver disease, overlap hepatitis, non-alcoholic steatohepatitis, or Gilbert's syndrome. Subjects with hepatitis B virus were also excluded; however, subjects who had seroconverted could be included following consultation with the medical monitor; [¶]Includes history of liver transplantation, current placement on a liver transplant list, current MELD score \geq 15 (MELD is a scoring system for assessing the severity of chronic liver disease, where the higher the score, the more severe the disease), portal

hypertension with complications, cirrhosis with complications, hepatorenal syndrome (type I or II), or screening serum creatinine >2 mg/dl; [‡]Defined as consumption of more than 210 mL of alcohol per week (i.e., the equivalent of 14 4-ounce (125 mL) glasses of wine or 14 12-ounce cans/bottles of beer). AASLD, American Association for the Study of Liver Disease; ALP, alkaline phosphatase; AMA, anti-mitochondrial antibody; BAS, bile acid sequestrants; CS, company

submission; EASL, European Association for the Study of Liver; HIV, human immunodeficiency virus; MELD, Model for End Stage Liver Disease; OCA, obeticholic acid; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

The baseline patient characteristics of POISE are reproduced in Table 4.6.

	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)	
Age, years					
Mean (SD)	55.5 (10.0)	55.8 (10.5)	56.2 (11.0)	55.8 (10.5)	
Median	55.0	54.5	56.0	55.0	
Min, max	35, 78	29, 83	30, 86	29, 86	
Age subgroups, n	l (%)				
<65 years	60 (82)	60 (86)	56 (77)	176 (81)	
≥65 years	13 (18)	10 (14)	17 (23)	40 (19)	
Gender, n (%)					
Male	5 (7)	5 (7)	10 (14)	20 (9)	
Female	68 (93)	65 (93)	63 (86)	196 (91)	
Race/ethnicity, n	(%)				
White	66 (90)	67 (96)	70 (96)	203 (94)	
Non-white	7 (10)	3 (4)	3 (4)	13 (6)	
Body weight, kg					
Mean (SD)	70.2 (13.3)	68.2 (13.1)	71.0 (15.3)	69.8 (13.9)	
Median	70.5	65.2	67.6	67.5	
Min, max	41.0, 106.0	46.7, 101.8	50.8, 134.0	41.0, 134.0	
Region, n (%)					
Europe	49 (67)	45 (64)	51 (70)	145 (67)	
North America	21 (29)	20 (29)	21 (29)	62 (29)	
Australia	3 (4)	5 (7)	1 (1)	9 (4)	
BMI, kg/m ²					
Mean (SD)	26.2 (4.4)	25.8 (4.9)	26.3 (5.1)	26.0 (4.8)	
Median	25.9	24.5	25.1	25.0	
Min, max	16.4, 37.6	17.7, 40.7	20.4, 49.2	16.4, 49.2	
BMI subgroups,	n (%)				
$<30 \text{ kg/m}^2$	58 (79)	58 (83)	61 (84)	177 (82)	
$\geq 30 \text{ kg/m}^2$	15 (21)	11 (16)	12 (16)	38 (18)	
Pre-treatment liver biopsy, n (%)					
Yes	7 (10)	13 (19)	9 (12)	29 (13)	
No	66 (90)	57 (81)	64 (88)	187 (87)	
UDCA use at bas	eline, n (%)				
Yes	68 (93)	65 (93)	67 (92)	200 (93)	
No	5 (7)	5 (7)	6 (8)	16 (7)	
Source: Table 20 in the CS. ³ BMI, body mass index; CS, company submission; OCA, obeticholic acid; SD, standard deviation; UDCA, ursodeoxycholic acid					

 Table 4.6: POISE Trial: Baseline patient demographics

The company states that 'treatment groups were well balanced for each key demographic and baseline variable. For the overall population, mean age was 55.8 years, with a range from 29 to 86 years, and a total of 81% of subjects were <65 years of age'.³ The study population was composed mainly of women (91%) of white ethnic background (94%). The majority of the population was European (67%), followed by North American (29%) and Australian (4%). The company stated that 'clinical expert opinion has validated that the patient population in POISE is representative of the population with PBC in the UK'³

The table below summarises the disease characteristics of the intention to treat (ITT) population included in the double-blind phase of POISE.

Disease characteristic	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
History of prurite	us, n (%)			
Yes	47 (64)	45 (64)	45 (62)	137 (63)
No	26 (36)	25 (36)	28 (38)	79 (37)
Severity of most 1	recent pruritus eve	nt for subjects who	had history of pr	uritus, n (%)
Mild	31 (66)	29 (64)	34 (76)	94 (69)
Moderate	14 (30)	13 (29)	8 (18)	35 (26)
Severe	1 (2)	2 (4)	3 (7)	6 (4)
Unknown	1 (2)	1 (2)	0 (0)	2 (1)
Pruritus at baseli	ne, n (%)			
Yes	47 (64)	37 (53)	44 (60)	128 (59)
Mild	32 (44)	27 (39)	33 (45)	92 (43)
Moderate	13 (18)	10 (14)	10 (14)	33 (15)
Severe	2 (3)	0 (0)	1 (1)	3 (1)
No	26 (36)	33 (47)	29 (40)	88 (41)
History of fatigue	e, n (%)			
Yes	49 (67)	38 (54)	41 (56)	128 (59)
No	24 (33)	32 (46)	32 (44)	88 (41)
Overall severity of	of PBC-related fati	gue, n (%)		
Mild	28 (38)	17 (24)	29 (40)	74 (34)
Moderate	16 (22)	16 (23)	8 (11)	40 (19)
Severe	3 (4)	5 (7)	3 (4)	11 (5)
Age at PBC diagr	iosis, years			
Mean (SD)	47.3 (9.3)	47.6 (11.7)	47.1 (10.6)	47.3 (10.5)
Median	48.0	48.0	47.0	47.5
Min, Max	31, 74	25, 82	24, 78	24, 82
Age at PBC diagr	osis subgroups, n	(%)		
<50 years	45 (62)	38 (54)	42 (58)	125 (58)
\geq 50 years	28 (38)	32 (46)	31 (42)	91 (42)

Table 4.7: POISE Trial: Baseline disease characteristics

Disease characteristic	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
Mean duration of	PBC, years			
Mean (SD)	8.3 (5.4)	8.3 (5.8)	9.2 (6.9)	8.6 (6.0)
Median	7.4	7.2	8.5	7.8
Min, max	0.9, 21.8	0.3, 27.0	0.0, 32.3	0.0, 32.3
Duration of PBC subgroups, n (%)				
\leq 7.5 years	39 (53)	36 (51)	30 (41)	105 (49)
>7.5 years	34 (47)	34 (49)	43 (59)	111 (51)
Source: Table 21 of CS ³				

The CS states that 'the mean age at time of diagnosis was 47.3 years with a mean duration of PBC of 8.6 years' and that 'There were slightly more subjects <50 years of age at PBC diagnosis (58%) compared with \geq 50 years of age.' Fifty-nine percent of patients had pruritus at baseline (43% mild, 15% moderate and 1% severe) and 59% had a history of fatigue. The CS notes that 'the majority (94–99%) of subjects had a baseline INR \leq 1.3, indicative of a population in an early stage of disease progression'.

The CS notes that 'In general, each variable was well balanced across treatment groups.' However 'the overall incidence of pruritus at baseline was slightly higher for subjects in the placebo treatment group (64% and OCA 10mg fixed dose group (60%) than in the OCA titration group (53%).....The overall incidence of fatigue was slightly higher for subjects in the placebo treatment group (67%) than in the OCA titration and OCA 10mg fixed dose groups (54% and 56%, respectively).'

ERG comments on POISE:

POISE, has several strengths and matches the NICE scope in several ways:

- It is a randomised controlled trial in a PBC population relevant to England and Wales.
- It is a multicentre international trial with 217 participants comparing OCA to UDCA alone in patients with an inadequate response to UDCA and comparing OCA to no treatment in patients intolerant to UDCA.
- Follow-up is of 12 months' duration covering an extensive range of surrogate outcomes.

There are a number of limitations in applying the results of the trial to the NICE scope:

- The number of patients intolerant to UDCA and receiving OCA as monotherapy was limited to 11 patients (five [7%] in the OCA titration group and six [8%] in the OCA 10 mg group). Five patients received OCA placebo. As such the results for this group of patients should be considered with caution due to the low numbers.
- The ERG noted that the POISE trial included patients at an early state of PBC progression as stated in Section 4.5.2.3 of the CS '*the majority (94–99%) of subjects had a baseline INR* ≤*1.3, indicative of a population in an early stage of disease progression*'.³ The ERG asked the company to clarify how OCA would benefit patients with more advanced PBC. In the response to clarification letter, the company stated that two analyses were performed in more advanced stages of PBC. Several definitions were used to approximate severity of the disease including a clinical composite definition and presence of cirrhosis. The results of these analyses are provided in Section 4.2.4. It should be noted that these are based on a subset of 72 patients

classified as having advanced disease and 36 who had cirrhosis.³²The ongoing COBALT trial includes more advanced patients and should provide more definitive results for these patients.

- Although stratified on several important variables, the POISE trial had differences between treatment groups at baseline. In particular the placebo group had a higher incidence of pruritus and fatigue which should be borne in mind particularly when interpreting results of adverse events.
- In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo.²⁷ Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only up-titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response.²⁷ The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg.²⁷ The company notes that '*further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA*.'³ This appears to be reasonable but is an assumption.
- The POISE trial presents evidence using surrogate outcomes only for clinical effectiveness. As stated in Section 3.4, the ERG is satisfied that the company has demonstrated evidence of some correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease is unclear.
- In the POISE CSR some early clinical outcomes were retrospectively gathered.⁴⁶ Although these are presented in Section 4.2.5, the time scale of the trial (12 months) does not allow conclusions to be drawn based on this early long-term evidence. The definitive effects of obeticholic acid on clinical outcomes relevant to patients awaits the results of the COBALT trial.³⁷
- As stated in the NICE scope for Obeticholic acid, the HRQoL of PBC patients is relevant for this submission.⁴ The company used a disease-specific questionnaire PBC-40 that assesses symptoms across several domains: fatigue, emotional and social, cognitive function, general symptoms and itch.⁴⁷ Brief results as presented in the CSR are given in Section 4.2.4 of the report.

4.2.3 Quality assessment of POISE

We reproduce the company's quality assessment of the POISE trial³ alongside the ERG's views on the quality of the trial.

Study question	How is the question addressed in the study?	Company Grade (yes/no/not clear/NA)	ERG Grade (yes/no/not clear/NA)
Was randomisation carried out appropriately?	Subjects were randomised via an IWRS based on a pre- defined randomisation code with stratification criteria to ensure subjects were randomised equally within each sub- group.	Yes	Yes
Was the concealment of treatment allocation adequate?	The IWRS served as an investigational product inventory and management system. All investigational product was visually identical for both OCA treatment arms and for the placebo group. Investigational product bottles were not labelled with either a subject randomisation number or tablet strength, to ensure that neither subject nor Investigator was unblinded. Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor designee(s) in the event of a medical emergence. No other Sponsor personnel or vendor/CRO had access to blinded subject treatment codes until all study data were entered into the study database, validated, and the database locked.	Yes	Yes
Were the groups similar at the outset of the study in terms of prognostic factors, for example severity of disease?	Baseline characteristics were similar between treatment arms	Yes	Yes

Table 4.8: Quality assessment of POISE

Study question	How is the question addressed in the study?	Company Grade (yes/no/not clear/NA)	ERG Grade (yes/no/not clear/NA)
Were the care providers, participants and outcome assessors blind to treatment allocation? If any of these people were not blinded, what might be the likely impact on the risk of bias (for each outcome)?	All investigational product was visually identical for both OCA treatment arms and for the placebo group. Investigational product bottles were not labelled with either a subject randomisation number or tablet strength, to ensure that neither subject nor Investigator was unblinded. Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor designee(s) in the event of a medical emergence. No other Sponsor personnel or vendor/CRO had access to blinded subject treatment codes until all study data were entered into the study database, validated, and the database locked.	Yes	Yes
Were there any unexpected imbalances in drop- outs between groups? If so, were they explained or adjusted for?	Slightly more patients withdrew from the OCA treatment groups (10% in the titration group and 12% in the 10 mg group) than the placebo group (4%). This was mainly due to pruritus.	Yes	Yes
Is there any evidence to suggest that the authors measured more outcomes than they reported?	Results are reported for all outcomes specified in the methodology.	No	No
Did the analysis include an intention-to-treat analysis? If so, was this appropriate and were appropriate methods used to account for missing data?	An ITT population was used where appropriate. Missing values were considered a non-response.	Yes	Yes
Source: CS, Appendix 3 ⁴¹ Abbreviations: ITT, intention to treat; IWRS, interactiv	ve web response system	L	

ERG comment: It can be seen that the ERG agrees with the company's assessment and finds that the trial has been well conducted with appropriate procedures for randomisation, allocation concealment, blinding and outcome assessment.

4.2.4 POISE: Efficacy results

The primary efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months) is detailed in Table 4.9 below.

	Responders (%)
	Month 12
Placebo (n=73)	10%
10 mg OCA (n=73)	47%
Titration OCA (n=70)	46%
Titration subgroup [†]	
Remained at 5 mg OCA for 12 months (n=36)	53%
Titrated to 10 mg OCA at Month 6 (n=33)	39%

'	Table 4.9:	Summary	of	primary	efficacy	outcome	in	POI	SE
									_

Source: Table 23 CS³

Footnote: [†]There was one participant who withdrew from the trial due to an AE after 8 days of study medication, and therefore there were no data for this participant at Month 6 and Month 12. §Of the 12 participants who did not respond but did not up-titrate, nine did not increase their dose due to adverse events, and three recorded their reason as 'other'

ALP, alkaline phosphatase; CS, company submission; OCA, obeticholic acid.

It can be seen from the table above that at 12 months 47% of participants in the 10mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons).

Secondary outcomes

For the two main surrogate outcomes relevant to the primary composite outcome in POISE, namely ALP and total bilirubin levels, the company reports that at 12 months, 25 (34%) and 21 (30%) of participants from the OCA 10 mg fixed dose and OCA titration groups respectively achieved an ALP reduction from baseline \geq 40% compared with 1% in the placebo group.³ For the total bilirubin outcome, decreases in the absolute change from baseline were observed for both OCA treatment groups compared with an increase for the placebo treatment group. At 12 months, mean bilirubin levels in the OCA 10 mg group were 9.7 (SE 0.6), 9.9 (SE 0.6) in the OCA titration group and 13.2 (1.0) in the placebo group.³

The CSR states that 'The disease-specific measure for PBC showed no clinically significant improvements in comparison to placebo for the global score or individual scores of general symptoms, fatigue, cognitive function, and emotional/social domains; however, a difference was observed in itch scores in the earlier treatment months.'⁴⁶ The CSR further states that 'During the initial 3 months of treatment, the largest LS mean increase in itch was observed for the OCA 10 mg group, followed by the OCA titration group.... The LS mean difference in itch score between the OCA 10 mg group and placebo group was statistically significant at both Week 2 (p = 0.0048) and Month 3 (p < 0.0001) but not at any subsequent time points.' ⁴⁶

Retrospective assessment of clinical outcomes

The results of the retrospective assessment of pre-defined clinical outcomes during the 12 month double-blind phase is given in Table 4.10.46

Outcomes ^a	No of patients	Treatment at the time of event	Preferred term Score	Time to onset from Day 0	
All cause mortality	1	OCA 5 mg	Cardiac failure ^b	257	
Clinical complications					
Oesophageal / bleeding varices	1	Placebo	Upper GI haemorrhage ^c Varices oesophageal Varices oesophageal	75 92 134	
Ascites / Diuretic- Resistant Ascites	1	OCA 5 mg	Ascites	360	
Hepatic encephalopathy	1	OCA 5 mg	Hepatic encephalopathy Hepatic encephalopathy	360 378	

Table 4.10: All-cause mortality and clinical complications: safety population (N = 216)

Source: CSR 747-301

Footnotes: a) No clinical outcome events were reported in liver-related mortality, hepatic cellular carcinoma (HCC), spontaneous bacterial peritonitis, or hepatorenal syndrome or hepatopulmonary syndromes.

b The event qualified as a clinical endpoint due to fatal outcome.

c The gastrointestinal haemorrhage was considered due to variceal bleeding.

The incidence of reaching a MELD score >15 (with a baseline MELD of <15) is given in Table 4.11.⁴⁶

No of patients	Treatment	Baseline MELD score	Score	Visit
1	Placebo	6.4	15.8	Week 2
			17.1	Month 3
			21.7	Month 6
1	Placebo	6.4	21.7	Week 2
			17.1	Month 6
			17.6	Month 9
			20.1	Month 12
1	Placebo	8.5	23.6	Month 9
1	Placebo	6.8	20.8	Month 6
1	OCA 5 mg	8.8	16.6	Month 3
1	OCA 5 mg	6.4	17.5	Week 2
1	OCA 10 mg	6.7	18.0	Unscheduled ^a
1	OCA 10 mg	11.3	16.0	EOT ^b

Table 4.11: Clinical Outcomes – Transplant (MELD Score): Safety Population (N = 216)

Source: CSR 747-30146

Footnotes: a) Unscheduled visit occurred approximately 4.5 months post-Baseline; b) End of treatment (EOT) occurred 185 days post-Baseline. Participant early terminated due to pruritus CSR, clinical study report; MELD, Model for End Stage Liver Disease

Subgroup of patients with advanced disease

A total of 72 participants (30, 22 and 20 in the placebo, OCA titration and OCA 10 mg respectively) were classified as having advanced disease using a clinical composite definition that included biochemical criteria (not specified), non-invasive measures of fibrosis, biopsies and/or medical history of decompensation and presence of cirrhosis.³² The company concluded that '*there was not an increased risk (of adverse events) in patients with advanced disease*'.³² No efficacy data were presented for these patients. A post-hoc analysis of efficacy and safety of 36 patients with cirrhosis (13 in the placebo arm, 13 in the OCA titration arm and 10 in the OCA 10 mg arm) was conducted. The company stated that '*ALP levels were significantly more reduced over 12 months in the OCA treatment groups than the placebo groups and bilirubin increased in the placebo group but remained stable in the OCA treatment group. The primary composite outcome in POISEwas met by 8% of subjects in the placebo arm, 54% in the OCA titration arm and 40% in the OCA 10 mg arm after 12 months.'³²*

ERG comment:

- The ERG notes the improvements shown in the OCA treatment groups in terms of the combined efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin ≤ ULN and ALP decrease from baseline at 12 months). This is supported by improvements in other surrogate markers. However no firm data on clinical outcomes are available. It is too early to draw conclusions on long-term clinical outcomes and the need for transplant based on the retrospective data available.
- The ERG notes that improvements in surrogate outcomes were not reflected in the disease specific quality of life tool (PBC-40) over the 12 month period.
- The improvements seen in patients with more advanced disease (briefly reported) are based on lower patient numbers as the majority of patients in POISE had earlier stage disease.
- As few patients used OCA as monotherapy it is unclear if results accurately reflect this patient group.

4.2.5 POISE: Safety results

A summary of the adverse events as described in the CS³ is detailed in Table 4.12.

Participants, n (%)	Placebo n=73	OCA titration n=70	OCA 10 mg n=73
Any TEAE	66 (90)	65 (93)	69 (95)
Total number of TEAEs	452	471	467
Any treatment-related AE ⁺	38 (52)	42 (60)	54 (74)
Any SAEs	3 (4)	11 (16)	8 (11)
Total number of SAEs	8	15	11
TEAEs by severity			
Mild	29 (40)	16 (23)	19 (26)
Moderate	28 (38)	27 (39)	29 (40)
Severe	9 (12)	22 (31)	21 (29)
Any TEAE leading to discontinuation	2 (3) [‡]	5 (7) [§]	8 (11) [¶]
Discontinuation due to pruritus	0 (0)	1 (1)	7 (10)
Number of deaths	0 (0)	1 (1)	0 (0)

 Table 4.12: Overview of adverse events: safety population (POISE)

Participants, n (%)	Placebo	OCA titration	OCA 10 mg
	n=73	n=70	n=73

Source: Table 37 CS³

Footnotes: [†]Includes any events determined to be 'possibly', 'probably', and 'definitely' related. [‡]One participant was discontinued from study due to withdrawal of consent; [§]No participants withdrew who titrated to OCA 10 mg. [¶]One participant experienced a TEAE of fatigue, which was recorded as a discontinuation on the AE eCRF; however, the participant remained in the study and study drug was not changed

AE, adverse event; CI, confidence interval; CS, company submission; eCRF, electronic case report form; SAE, serious adverse event; TEAE, treatment-emergent adverse event

Over 90% of patients across the trial experienced an adverse event. Occurrence of any AE was comparable across treatment groups. However a larger number of events were rated as severe in the OCA treatment groups (OCA 10 mg 29%, OCA titration 31% and placebo 12%). Of note, 10% of patients in the OCA 10 mg group discontinued due to pruritus.

The company stated that '*TEAEs that occurred with an incidence of* \geq 5% and were reported more frequently in either of the OCA treatment groups compared with placebo included pruritus, rash, eczema, fatigue, pyrexia, peripheral oedema, nasopharangitis, influenza, bronchitis, sinusitis, diarrhoea, constipation, abdominal discomfort, arthralgia, cough, oropharyngeal pain, procedural pain, fractures, palpitations and hypothyroidism.'³ Full details of specific adverse events are given in Table 4.13.

SOC/preferred term, n (%) [†]	Placebo N=73	OCA titration N=70	OCA 10 mg N=73				
TEAEs occurring in \geq	5% of participants in ei	ther OCA treatment gro	oup [‡]				
Skin and subcutaneous	s tissue disorders						
Pruritus	28 (38)	39 (56)	50 (68)				
Rash	3 (4)	3 (4)	4 (5)				
Eczema	0	4 (6)	2 (3)				
General disorders and	administration site con	ditions					
Fatigue	10 (14)	11 (16)	17 (23)				
Oedema peripheral	2 (3)	2 (3)	5 (7)				
Pyrexia	1 (1)	0	5 (7)				
Infections and infestat	ions						
Nasopharyngitis	13 (18)	17 (24)	13 (18)				
Upper respiratory tract infection	8 (11)	4 (6)	4 (5)				
Urinary tract infection	8 (11)	4 (6)	4 (5)				
Influenza	4 (5)	5 (7)	4 (5)				
Bronchitis	0	4 (6)	1 (1)				
Sinusitis	0	1 (1)	4 (5)				
Gastrointestinal disord	Gastrointestinal disorders						
Nausea	9 (12)	4 (6)	8 (11)				
Diarrhoea	8 (11)	2 (3)	8 (11)				
Constipation	4 (5)	5 (7)	5 (7)				

Table 4.13: Summary of TEAEs, severity of AEs, and treatment-related AEs occurring in ≥5% of participants in either OCA treatment group

SOC/preferred term, n (%) [†]	Placebo N=73	OCA titration N=70	OCA 10 mg N=73		
Abdominal pain upper	5 (7)	5 (7)	4 (5)		
Gastroesophageal reflux disease	4 (5)	2 (3)	4 (5)		
Dyspepsia	8 (11)	4 (6)	0		
Abdominal discomfort	1 (1)	5 (7)	0		
Musculoskeletal and c	onnective tissue disorde	rs			
Arthralgia	3 (4)	4 (6)	7 (10)		
Back pain	8 (11)	4 (6)	4 (5)		
Nervous system disord	lers				
Headache	13 (18)	12 (17)	6 (8)		
Respiratory, thoracic,	and mediastinal disorde	ers			
Cough	5 (7)	4 (6)	6 (8)		
Oropharyngeal pain	1 (1)	5 (7)	6 (8)		
Injury, poisoning, and	procedural complicatio	ns			
Procedural pain	1 (1)	4 (6)	1 (1)		
Fractures	3 (4)	2 (3)	4 (5)		
Cardiac disorders					
Palpitations	1 (1)	2 (3)	5 (7)		
Eye disorders					
Dry eye	4 (5)	2 (3)	4 (5)		
Endocrine disorders					
Hypothyroidism	1 (1)	4 (6)	1 (1)		
Incidence of TEAE by	maximum severity, n (%	%)			
Mild	29 (40)	16 (23)	19 (26)		
Moderate	28 (38)	27 (39)	29 (40)		
Severe	9 (12)	22 (31)	21 (29)		
Treatment-related AE	s in ≥5% of participants	in any OCA treatment	group, n (%)§		
Skin and subcutaneous	s tissue disorders				
Pruritus	27 (37)	35 (50)	48 (66)		
General disorders and	administration site con	ditions			
Fatigue	8 (11)	6 (9)	6 (8)		
Gastrointestinal disord	lers				
Nausea	4 (5)	3 (4)	4 (5)		
Source: Table 38 CS ³ . Footnotes: [†] At each level of summation, participants reporting >1 AE are counted only once; [‡] a TEAE is defined as any event that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product. At each level of summation, participants reporting >1 AE are counted only once using the highest severity; [§] treatment-related AEs include all events reported as 'possible', 'probable', or 'definite' relationship to study drug. AE, adverse event; CS, company submission; N, total number of participants; n, number of participants experiencing an event, OCA, obeticholic acid; SOC,					
systems organ class; TEAE, treatment-emergent adverse event					

Pruritus was the most commonly occurring TEAE with higher incidence in the OCA arms (OCA 10 mg [68%], OCA titration [56%] and placebo [38%]). The company stated that '*Most pruritus events were of mild or moderate severity and resolved during the treatment period. Starting at OCA 5 mg and titrating to 10 mg OCA was associated with improved tolerability versus starting at 10 mg OCA.*'³

The company also state that '*Fatigue and nausea were the only other related TEAE that occurred at an incidence* \geq *5%; however, these events were balanced between placebo and OCA treatment arms.*'³

ERG comment:

- Adverse events overall appear broadly comparable in OCA and placebo groups. However a larger number in the OCA group appear to be rated severe.
- Ten percent of patients discontinued treatment due to pruritus in the OCA 10 mg group (1% in the titration group, 0 in the placebo group). Pruritus was more common in the OCA groups than in placebo groups. The company's comments on the benefits of titration of dose in relation to pruritus appear to be appropriate.

4.2.6 Overview of the supporting Phase 2 trials (747-201 and 747-202):

The Phase 2 trials evaluated the efficacy, safety, and tolerability of OCA with UDCA (study 747-202) or without UDCA (study 747-201) vs. placebo. Both included a 10 mg OCA treatment arm and other treatment arms at higher doses. In the CS, key efficacy and safety results were summarised only for patients who received 10 mg OCA, since this is the upper limit for the licensed indication of OCA. Both studies included a double-blind phase and an open-label long-term safety extension phase. Table 4.14 includes the study design characteristics of the double-blind phase.

Trial number	Study 747-201	Study 747-202
Settings and locations	18 centres in 6 countries (UK, USA, Canada, Germany, France, Spain)	30 centres in 8 countries (USA, Canada, Germany, UK, The Netherlands, Austria, France, Spain)
Duration of trial	3 months	3 months
Trial design	Multi-centre, randomised, double-blind, placebo-controlled, multi-dose, Phase 2 parallel group study of OCA monotherapy in participants with a proven or likely diagnosis of PBC	Multi-centre, randomised, double-blind, placebo-controlled, multi-dose, Phase 2 parallel group study of OCA in with UDCA in participants with a proven or likely diagnosis of PBC.
Eligibility criteria for	Proven or likely PBC	Proven or likely PBC
participants	Aged 18–70 years (18–75 years in the UK)	Aged 18–70 years (18–75 years in the UK)
Number randomised	60 were randomised, 59 participated (23 placebo, 20 OCA 10 mg, 16 OCA 50 mg)	165 were randomised (38 to placebo, 38 to OCA 10 mg, 48 to OCA 25 mg, 41 to OCA 50 mg)
Number of arms	3 arms (placebo, OCA 10 mg and OCA 50 mg)	4 arms (placebo, OCA 10 mg, OCA 25 mg, OCA 50 mg)
Primary endpoint	Percentage change in serum ALP from baseline to end of study	Percentage change in serum ALP from baseline to end of study
Secondary endpoints	Absolute changes in serum ALP levels from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up Percentage of participants meeting PBC responder criteria as per	Changes in serum ALP levels from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up Responder analyses of ALP response
	the Paris I, Toronto I, Toronto II, Toronto III, Toronto IV, Mayo II, and Barcelona disease prognostic risk criteria at Day 85/end of treatment	Change in serum AST, ALT, GGT, serum albumin and conjugated (direct) bilirubin values from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up
	Absolute and percentage change in serum AST, ALT, GGT, and conjugated (direct) bilirubin values from baseline to Day 15, Day 29, Day 57, Day 85/end of treatment and Day 99/follow-up	Changes in CRP, non-esterified fatty acid, TNF- α & β , TGF- β , bile acids, glutathione, IgM, and osteopontin from Baseline to Day 85/end of treatment
	Safety	SF-36 and PBC-40 QoL questionnaires
		Bile acid analysis and change in FGF-19 from baseline to Day 85/end of treatment
		Safety

 Table 4.14: Summary of Phase 2 study characteristics (double-blind phase)

Source: CS³

ALP, alkaline phosphatase; AST, aspartate aminotransferase; CRP, C-reactive protein; GGT, gamma-glutamyl transferase; TGF, transforming growth factor; TNF, tumour necrosis factor, UDCA, ursodeoxycholic acid

In summary, both studies found a statistically significant effect of OCA 10 mg on ALP levels from baseline to end of study versus placebo. In study 747-201 (monotherapy) the mean (SD) percentage change in ALP levels was -44.5% (24.4) for the OCA 10 mg group versus +0.4% (15.3) for placebo. Study 747-202 reported mean (SD) percentage change in ALP levels was -44.5% (24.4) for the OCA 10 mg group versus +0.4% (15.3) for placebo. Both results were statistically significant at p<0.0001.

ERG comment: The Phase two trials lend support to the findings of POISE on the benefits of OCA on surrogate outcomes for patients with PBC. However both trials were of a short-term duration (three months) and a variety of doses were used limiting the comparability of the trials to POISE and the possibility of pooling results.

4.3 Critique of trials identified and included in the indirect comparison and/or multiple treatment comparison

Not applicable as there was no indirect comparison or multiple treatment comparison in the CS.

4.4 Critique of the indirect comparison and/or multiple treatment comparison

Not applicable as there was no indirect comparison or multiple treatment comparison in the CS.

4.5 Additional work on clinical effectiveness undertaken by the ERG

Not applicable.

4.6 Conclusions of the clinical effectiveness section

The company conducted a systematic review to identify studies comparing OCA to the comparators outlined in the NICE scope.⁴ Overall the ERG is satisfied that the relevant direct evidence comparing OCA and UDCA has been presented. However no trials comparing OCA to fibrates were identified.

One Phase 3 trial, POISE, with 217 patients was presented as the main source of evidence.²⁷ The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. However the group receiving OCA as monotherapy is underrepresented in this trial (11 patients). Additionally, the majority of patients in POISE appeared to be at an earlier stage of disease so the effects on those with more advanced disease are less clear.

The POISE trial was well-conducted. However it only examined surrogate outcomes. These were justified by the company based on previously published research. In POISE, OCA shows positive effects on surrogate endpoints, and there is some evidence that surrogate endpoints are related to relevant outcomes. However, the size of the relationship is unclear.

The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events was similar across groups. Events occurring more frequently in treatment groups were noted. The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

Two supporting Phase 2 trials were presented (including one of OCA monotherapy and one of OCA in combination with UDCA). These were not similar enough to be pooled with POISE but added support to the positive findings on surrogate outcomes. Clinical outcomes await the publication of the COBALT trial.³⁷

5. COST EFFECTIVENESS

5.1 ERG comment on company's review of cost effectiveness evidence

The following section includes searches for the cost effectiveness analysis review, measurement and evaluation of health effects as well as for cost and healthcare resource identification, measurement and valuation. Moreover, the review of cost effectiveness analysis studies is considered

5.1.1 Searches performed for cost effectiveness section

The following paragraphs contain summaries and critiques of all searches related to cost effectiveness presented in the company submission.

Objective and searches for cost effectiveness analysis review

A systematic review was conducted to identify all relevant PBC studies relating to cost effectiveness. More specifically, the review question was: what modelling techniques have been used previously to conduct economic evaluations for the treatment of PBC?

The CS reported that searches were carried out in September 2014 and updated in 2016. The original searches were not limited by date or language. Searches were carried out on a broad range of databases including those recommended in the NICE 2013 guide to the methods of technology appraisal Sections 5.2.2 and 5.2.4.⁴⁸ A supplementary grey literature search was reported in Section 5.1.1.2, but no further details were provided in the appendices.

ERG comment: The majority of searches in Appendix 6 were well reported and easily reproducible. The ERG queried the lack of further information regarding the grey literature searches in their request for clarification.³⁰ In their response the company reported that where applicable the following resources were searched using the terms 'primary biliary cirrhosis' and primary biliary cholangitis' in order to inform the cost effectiveness, HRQoL and healthcare resource use literature reviews:

- American Association for the Study of Liver Diseases (AASLD) http://www.aasld.org/Pages/Default.aspx
- European Association for the Study of the Liver (EASL) http://www.easl.eu/
- American College of Gastroenterology (ACG) http://gi.org/
- Digestive Disease Week (DDW) http://www.ddw.org/
- United European Gastroenterology Week (UEGW) http://www.ueg.eu/
- Canadian Digestive Disease Week http://www.cag-acg.org/
- Japan Digestive Disease Week http://www.jddw.jp/english/index.html
- Liver Foundation UK http://www.liverfoundation.org.uk/
- www.nice.org.uk
- The Foundation for Liver Research http://www.liver-research.org.uk/
- American Liver Foundation http://www.liverfoundation.org/
- Canadian Liver Foundation http://www.liver.ca/
- British Liver Trust http://www.britishlivertrust.org.uk/
- The British Library http://www.bl.uk
- National Institute for Health Research http://www.hta.ac.uk/
- National Institute for Health and Care Excellence (NICE) http://www.nice.org.uk/
- Scottish Medicines Consortium (SMC) http://www.scottishmedicines.org.uk/Home
- National Centre for Pharmacoeconomics (NCPE) Ireland http://www.ncpe.ie/submissionprocesss/hta-guidelines/

- All Wales Medicines Strategy Group (AWMSG) http://www.wales.nhs.uk/sites3/home.cfm?orgid=371
- Institute for Quality and Efficiency in Health Care (IQWiG) https://www.iqwig.de/en/home.2724.html
- Haute Autorité de Santé (HAS) http://www.has-sante.fr/portail/jcms/j_5/home
- Italian Medicines Agency (AIFA) http://www.agenziafarmaco.gov.it/en
- Agencia de Evaluación de Tecnologías Sanitarias (AETS) http://www.isciii.es/
- Canadian Agency for Drugs and Technologies in Health (CADTH) http://www.cadth.ca/
- Pharmaceutical Benefits Advisory Committee (PBAC)
 <u>http://www.pbs.gov.au/info/industry/listing/participants/pbac</u>
- Search Engine (Google) <u>http://www.google.co.uk</u>³²

The ERG noted a disparity in the date reported for the 2016 update searches, Section 6.2 (Appendix 6)⁴¹ stated that these were carried out in February 2016, but individual strategies carry a search date of June 2016. This error was also present in Section 9.2 (Appendix 9).⁴¹ Although not explicitly stated the cost effectiveness filter used in the Embase and MEDLINE strategies reported in Tables 20, 23 and 24 (Appendix 6) appeared to be based on the pragmatic filter developed by the Scottish Intercollegiate Guidelines Network (SIGN)⁴⁹ to identify Economic studies.⁴¹ Although some attempt had been made to translate the Embase filter in the 2016 MEDLINE update search, the MEDLINE version of the filter created by SIGN, which contains additional relevant MeSH had not been used, therefore potentially useful records may have been missed. However given the searches of additional bibliographic databases and the above list of grey literature resources, it is unlikely that this would have impacted on the overall recall of results. The ERG noted that an economics filter was included in the update of the NHS EED search. As this is an economics database the ERG believes it is not necessary to include this facet, as this may result in unnecessarily restricting the results retrieved. However given the breadth of the searches reported this is unlikely to have impacted on the overall recall of results.

Objective and searches for measurement and valuation of health effects

A systematic review was conducted to identify HRQoL studies from the published literature relevant to the decision problem. In particular, EQ-5D health state utility values (in line with the NICE preferred method) relating to HRQoL in PBC-specific disease states were sought. More specifically, the review question was: what evidence exists reporting the quality of life in patients with PBC?

Searches were carried out on a broad range of databases including those recommended in the NICE 2013 guide to the methods of technology appraisal Sections 5.2.2 and 5.2.4.⁴⁸ No date or language limits were applied. Supplementary searches of conference proceedings and other relevant resources including HTA agencies and organisational websites were reported, for further details of the grey literature searches please see the company's response to clarification above.

ERG comment: Searches were well reported and easily reproducible. The HRQoL filters contained a good combination of relevant subject heading terms and free text terms. As with the previous search the ERG noted that the HRQoL filter was included in the update of the NHS EED search, as stated above this is unlikely to have been consequential but the same limitations will apply. As with the section above a disparity was noted in the date reported for the update searches, Section 8.2 (Appendix 8) stated that these were carried out in February 2016, but individual strategies carry a search date of May 2016.

Objective and searches for cost and healthcare resource identification, measurement and valuation

A systematic review was conducted to identify all relevant PBC unit cost and resource use studies from the published literature relevant to the decision problem. More specifically, the review question was: what are the costs and resource use associated with the management of PBC?

Searches were carried out on a broad range of databases including those recommended in the NICE 2013 guide to the methods of technology appraisal Sections 5.2.2 and 5.2.4.⁴⁸ No date or language limits were applied. Supplementary searches of conference proceedings and other relevant resources including HTA agencies and organisational websites were reported, for further details of the grey literature searches please see the company's response to clarification above.

ERG comment: The ERG noted that the strategy reported for the 2014 Embase healthcare resource use identification search (Table 40, Appendix 9) appeared to be a duplicate copy of the HRQoL Embase strategy (Table 33, Appendix 8) and did not reflect the resource use terminology utilised in the update search run in June 2016 (Tables 43 & 44, Appendix 9). The company confirmed that this had been included in error and provided full details of the correct strategy, which included a resource use/societal cost filter which contained a combination of relevant subject heading and free text terms.

5.1.2 Inclusion/exclusion criteria used in the study selection

The pre-specified inclusion/exclusion criteria are shown as a PICOS table in Appendix 6 of the CS (Tables 27 and 28).⁴¹

ERG comment: The eligibility criteria used by the company seem appropriate to the ERG.

5.1.3 Included/excluded studies in the cost effectiveness review

The search identified 184 titles/abstracts. After screening, 167 references were excluded, leaving 17 references for full-text evaluation. Two further studies were identified from the review of grey literature. Following full-text evaluation, four references met the inclusion criteria.⁵⁰⁻⁵³ All of these four studies were deemed relevant to the review and to be used for data extraction.

Based on these four studies, the company noted that:

- None of the analyses are economic evaluations of OCA.
- Three studies⁵⁰⁻⁵² applied modelling techniques, and one study⁵¹ considered long-term outcomes.
- All identified economic evaluations incorporated liver transplant and death as key outcomes, (while two also allowed the possibility of re-transplantation in their patient pathway).
- Only one of the identified papers⁵⁰ reported QALYs, this study aligned with the NICE reference case for derivation of utility estimates.
- Studies typically undertook detailed costing analyses; however, the extent of reporting varied between studies. The costs included in analyses appear to be limited to direct costs.
- Whilst analyses differed in their methods, those that compared UDCA or liver transplant to placebo each found the active therapy to dominate placebo, i.e. reducing costs and improving health outcomes.

5.1.4 Conclusions of the cost effectiveness review

The CS provides an overview of the included studies but no specific conclusion regarding the cost effectiveness of OCA is formulated.

ERG comment: Since the identified studies did not consider all relevant costs and outcomes (i.e. only considered the short-term) and/or did not consider the intervention of interest, the ERG agrees that no specific conclusion from the review could be formulated.

5.2 Summary and critique of company's submitted economic evaluation by the ERG

Table 5.1: Summary	v of the com	nany's ecor	nomic evaluatio	on (with s	signnosts t	o CS)
Table 5.1. Summar	y of the con	pany secon	ionne evaluatio	m (with a	ngnpusis i	ບັບສຸ

	Approach	Source / Justification	Signpost (location in CS)
Model	A semi-Markov state transition model was constructed to evaluate the cost effectiveness of OCA compared with placebo in treating patients with PBC who have failed to show adequate control with UDCA treatment, i.e. UDCA- intolerant patients or patients who had inadequate response to UDCA.		5.2.2
States and events	 In the first year, the model comprises of the following liver disease component health states, or death: Low risk of disease progression: ALP ≤ 1.67 * ULN Progressive PBC: ALP > 1.67 * ULN and TB ≤ 1.0 * ULN Progressive PBC: TB > 1.0 * ULN or compensated cirrhosis (CC) In the following periods, patients can progress to: Pre-liver transplant decompensated cirrhosis, hepatocellular carcinoma (HCC) and from there can move on to: liver transplant, with a post-liver transplant state and potential PBC reemergence excess mortality 	A Markov structure was used as it was consistent with other approaches for liver disease modelling. Patients with abnormal bilirubin and CC were combined into one health state due to lack of data and to reflect the risk of patients developing CC prior to DCC and also the risk of progressing to HCC or liver transplant waiting list based on bilirubin levels.	5.2.2

	Approach	Source / Justification	Signpost (location in CS)
Comparators	 No treatment (in UDCA-intolerant patients) UDCA monotherapy (in patients with inadequate response to UDCA) 	Fibrates were not used as a comparator in the company's model because fibrates are not licensed in the UK, nor are they standard of care, and they are contraindicated in PBC. They are rarely used, with only December of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC). The efficacy of fibrates in treating PBC is yet to be proven.	5.2.3, 1.1
Population	 Patients who have failed to show adequate control with UDCA treatment, i.e. UDCA-intolerant patients Patients who had inadequate response to UDCA 		5.2.1
Treatment effectiveness	Film-coated tablet containing 5 mg or 10 mg OCA. The recommended starting dose is 5 mg taken orally, once daily. Based on the assessment of tolerability after 6 months, the dose should be increased to 10 mg once daily to achieve optimal response (titration dose). The POISE study is the principal source of evidence on treatment effectiveness. Effectiveness is measured by the decrease in risk of moving to the progressive PBC states in the first year.	This is the dose stated in the marketing authorisation and used in the POISE study.	5.3.1, 1.2, 5.3.2
Adverse events	The following adverse events were identified: fatigue, pruritus and nausea. The health care cost consequences of pruritus only were incorporated in the model.		5.4.4
Health related QoL	Health related quality of life data were sourced from the literature. These quality of life data were used to calculate health state utility values.	Literature	5.4

	Approach	Source / Justification	Signpost (location in CS)
Resource utilisation and costs	Treatment costs, health state costs and adverse event costs were taken into account in the economic model. Treatment costs were based on OCA list price and equalled £2,384.04 per 30 tablets, and on BNF data for UDCA. Health state costs and adverse event costs were based on NHS reference costs, assumptions validated by expert opinion, and literature.	BNF, literature, expert opinion, assumption, NHS reference cost	5.5
Discount rates	Discount of 3.5% for utilities and costs	As per NICE scope	5.2.2.1
Sub groups	No subgroups were considered. There are, however, two populations (as stated above).		5.2.1
Sensitivity analysis	Both DSA and PSA were performed as well as two scenario analyses. The model was the most sensitive to the choice of utility values for the liver disease component health states.		5.8

ALP = Alkaline phosphatase; BNF = British National Formulary; CC = compensated cirrhosis; CS = company submission; DSA = deterministic sensitivity analysis; HCC = hepatocellular carcinoma; NHS = National Health Service; NICE = National Institute for Health and Care Excellence; OCA = obeticholic acid; PBC = primary biliary cirrhosis; PSA = probabilistic sensitivity analysis; TB = total bilirubin; TTD = time to treatment discontinuation; UDCA = ursodeoxycholic acid; UK = United Kingdom; ULN = upper limit of normal

5.2.1 NICE reference case checklist (TABLE ONLY)

	Table	5.2:	NICE	reference	case	checklist
--	-------	------	------	-----------	------	-----------

Elements of the economic evaluation	Reference Case	Included in submission	Comment on whether <i>de</i> <i>novo</i> evaluation meets requirements of NICE reference case
Population	As per NICE scope	Yes	As per NICE reference case
Comparator(s)	Therapies routinely used in the NHS, including technologies regarded as current best practice	Yes	UDCA and no treatment are used as comparators. Fibrates are excluded from the analysis but the company argues that these are not routinely used in the NHS.

Elements of the economic evaluation	Reference Case	Included in submission	Comment on whether <i>de</i> <i>novo</i> evaluation meets requirements of NICE reference case
Type of economic evaluation	Cost effectiveness analysis	Yes	As per NICE reference case
Perspective on costs	NHS and PSS	Yes	As per NICE reference case
Perspective on outcomes	All health effects on individuals	Partly	Disutilities of adverse events are not considered.
Time horizon	Lifetime horizon, i.e. 50 years or a maximum age of 100 years	Yes	As per NICE reference case.
Synthesis of evidence in outcomes	Systematic review	Yes	As per NICE reference case partially. Evidence of effectiveness came primarily from the POISE trial but also from other sources and unclear calibration methods.
Measure of health effects	Quality adjusted life years (QALYs)	Yes	As per NICE reference case partially. Utility decrements applied to published utility values lacked justification.
Source of data for measurement HRQoL	Described using a standardised and validated instrument	Yes	The EQ-5D-3L and HUI utility values are derived from the literature, expert opinion and decrements are applied to some health state specific utility values.
Source of preference data for valuation of changes in HRQoL	Visual analogue scale (VAS), and time-trade off (TTO)	Partly	The UK TTO valuations have been used as a default for the EQ-5D-3L questionnaire and VAS/SG from a sample of Canadian parents for the HUI questionnaire.
Discount rate	An annual rate of 3.5% on both costs and health effects	Yes	As per NICE reference case
Equity weighting	An additional QALY has the same weight regardless of the other characteristics of the individuals receiving the health benefit	Yes	As per NICE reference case
Sensitivity analysis	Probabilistic modelling	Yes	As per NICE reference case
EQ-5D-3L = European Quality of Life-5 Dimensions, 3 Levels; HRQoL = health-related quality of life; NHS = National Health Service; HUI = health Utility index; NICE = National Institute for Health and Clinical Excellence; PSA = probabilistic sensitivity analysis; quality-adjusted life years; PSS = Personal Social Services; QALY = quality-adjusted life year; TTO = Time trade off; UDCA = ursodeoxycholic acid; UK = United Kingdom; VAS = visual analogue scale			

5.2.2 Model structure

The company developed a Markov state transition model to describe the progression of PBC over a lifetime horizon.³ The model comprises two parts with a total of 10 health states. In the first part, the model captures the biomarker component based on the surrogate outcomes of alkaline phosphatase (ALP) and bilirubin biomarkers in three different health states based on the expected risk of disease progression: low risk (ALP \leq 1.67 * upper limit of normal; ULN); moderate risk (ALP > 1.67 * ULN and total bilirubin; TB \leq 1.0 * ULN) and severe risk (TB > 1.0 * ULN or compensated cirrhosis; CC). The latter health state combined both patients with compensated cirrhosis (CC) and those with abnormal and rising TB. The company acknowledged that CC preceded decompensated cirrhosis (DCC) but that data to confirm if patients with PBC had CC were sparse. In clinical practice, the histological status of PBC patients is rarely documented, and hence the risk of patients moving from CC to DCC is unknown. Moreover, in response to a request for further clarification,⁵⁴ the company argued that progression to DCC can occur without the fibrosis that typically characterises CC:

'PBC is a condition driven by progressive loss of bile ducts (ductopenia) leading to mechanical failure in the transport of bile from the liver (cholestasis), which then leads to the death (or senescence) of biliary epithelial cells. The condition is functionally different from other liver conditions such as HepC. As a result, fibrosis in itself is not considered to be the key clinical measure – whilst it is present in the later stages of the condition, the underlying ductopenia means that fibrosis progression is not a good measure of severity or response to treatment.'

The company therefore combined CC and abnormal bilirubin in one health state to reflect that patients with abnormal TB, as those with CC, were documented to exhibit higher mortality or undergo liver transplant. CC and abnormal bilirubin were thus assumed to be equivalent health states. In the first year, patients on OCA treatment can move from low, to moderate, to severe, and vice versa in the biomarker component. Patients in the comparator arm can only progress from low or moderate risk to the severe risk health state directly in the first year.

In the second part of the model, here called the liver disease component, the following clinical endpoints are modelled: pre-liver transplant; DCC; HCC; liver transplant; a post-liver transplant state; potential PBC re-emergence; and death. Patients can move to the liver disease component of the model only coming from the severe risk health state in the biomarker component. Patients can move to pre-liver transplant or HCC directly, or they can move to DCC first and from there to pre-liver transplant or HCC. From HCC they can also move on to pre-liver transplant. After liver transplant, patients move to the post-liver transplant health state, from which PBC can re-emerge. The model structure is shown in Figure 5.1.

Costs and health-related utilities associated with each health state were calculated per cycle. Relevant adverse events identified by the company were fatigue, pruritus and nausea, but costs were only modelled in relation to pruritus, and the impact on health-related quality of life (HRQoL) was not modelled. A three months cycle length was used. The model was programmed in Excel.⁵⁵



Figure 5.1: Model structure

Patients can die in each health state. The grey health state 'excess mortality' represents the disease-specific mortality associated with having decompensated cirrhosis, liver transplant or hepatocellular carcinoma. TB, total bilirubin.

Source: CS Figure 24³

ERG comment:

The biomarker component

The biomarker component of the model is unique to the present decision problem and the condition. There are a number of areas of uncertainty that the ERG is concerned about:

- 1. The aggregation of two different health states into one (CC and abnormal TB count) to form the severe risk state could be problematic, as the transition probability (TP) to the DCC state may refer to CC patients only. The company claimed in the response to the clarification letter that all TPs leading to and coming out of this combined health state were estimated based on data on elevated and rising bilirubin levels. The ERG consulted with a clinical expert who stated that the aggregation of these two health states was reasonable in PBC patients because elevated bilirubin is deemed the best short term predictor of adverse outcomes in PBC. The clinical expert stated that a caveat was the assumption that the elevated bilirubin was not due to other causes (Gilberts disease primarily) but considered this a minor confounder with no significant impact on the model. The ERG concludes that, while it appears reasonable to group elevated and rising bilirubin and CC in one health state in PBC, there remains significant uncertainty surrounding the TP with which patients transition from the severe risk health state to the DCC health state.
- 2. Patients receiving OCA treatment and who are in the low and moderate risk biomarker component health states at the end of the first year are assumed to remain there for the remainder of their lives. The company claims that this is consistent with experience with patients responding to UDCA treatment. The ERG consulted a clinical expert who backed the company's claim, stating that patients with a response to UDCA and normal (or below 1.67x ULN ALP) have an excellent long term prognosis with no overall impact on life expectancy. The ERG remains concerned that 1) the long term prognosis for OCA responders may not be equivalent to that of UDCA responders, and that 2) even if there is only a very small proportion of patients that lose response, this may have an effect on model outcomes. The ERG therefore performed exploratory analyses relaxing this assumption.

3. The assumption that patients in the comparator arm can only transit from the low risk health state to the severe risk health state directly, without the possibility of moving to the moderate risk health state, may lack plausibility because patients in the low risk state are not treated with OCA and there should therefore be no difference between the TPs in both treatment arms. The company, in response to the clarification letter,³² however, claimed that this assumption was made to account for the fact that in POISE, patients not receiving OCA had similar ALP levels over the duration of the 12 month study period but had increasing bilirubin levels, thus worsening their risk of end-stage liver disease, liver transplant, or death. The ERG considers that the same TP should be applied to patients moving from low to moderate or severe risk for consistency between OCA and the comparator arm. When using the company's assumption of none of the patients starting in a low risk health state, this assumption has no impact on the model outcomes.

The liver disease component

The model structure for this component is similar to other assessments of liver disease treatments and the ERG is satisfied that it is appropriate for the current decision problem. According to the company, the use of a Markov structure for the liver disease component of the model was consistent with other approaches to modelling liver diseases. However, the company's model diverges from those used in other liver diseases in that an additional pre-liver transplant health state was introduced (compared to, for instance, Technology Appraisal (TA) 330,⁵⁶ in which this state does not exist). No justification was provided for the introduction of the pre-liver transplant health state and it does not appear to be typically considered in other liver disease models. While the ERG acknowledges that the introduction of the preliver transplant health state has face validity as being a waiting list state with its own costs associated with it, the ERG is concerned that it groups patients together that came from different health states (HCC, DCC, severe risk) and who therefore may experience different health-related quality of life. The ERG therefore uses the liver disease component model structure and transition probabilities from TA 330 in its base-case.⁵⁶ The ERG furthermore notes that in the model, it is possible for patients to move from the post-liver transplant state back to the liver transplant state, without going through PBC reemergence. This is inconsistent with the company's model structure in Figure 5.1 and no justification was provided for this.

5.2.3 Population

The economic evaluation considers patients with PBC who are intolerant to or have inadequate response to treatment with UDCA. Only moderate and severe patients based on ALP and TB levels, at model entry, were deemed eligible to receive OCA treatment in the model. This is in line with the final scope issued by NICE for this appraisal and is also in line with the study population of the pivotal POISE study. Patients enter the model in the moderate (76.85%) and severe risk (23.15%) health states in the biomarker component of the model, according to ALP and bilirubin levels. These proportions of patients starting the model were not justified and apparently not in line with the distribution of patients in the POISE study: the distribution there was 91.58% in the moderate and 8.42% in the severe risk health state, based on Table 49 in the CS for the OCA titration arm,³ the company's model for the OCA treatment arm³ and the second clarification excel file for the UDCA arm and OCA arms.³²

ERG comment: The population represented in the cost effectiveness model corresponds to the expected licensed indication and the final scope issued by NICE for the current decision problem. However, the ERG considers that the proportions of patients entering the model in the moderate and severe risk health states should be based on data from POISE, and, although the company claims that the used proportions are derived from POISE, it is not clear how this was done. In the company's model, the proportion of patients entering the model in the severe risk health state is much larger than the

proportion of patients entering the POISE study in the severe risk health state (23.15% in the company's model compared to 8.42%). This leads to more patients remaining in the severe risk health state and moving to more severe disease (i.e. the liver disease component in the model). This in turn would potentially bias model outcomes in favour of OCA or OCA titration treatment. The ERG therefore uses the proportions of patients starting in the moderate and severe health states derived from the POISE OCA, OCA titration and UDCA treatment arms, as described above.

5.2.4 Interventions and comparators

The interventions and comparators in this model depend on the selected population. For UDCAintolerant patients, the intervention considered in this economic evaluation is OCA dose titration based on a starting dose of 5 mg taken orally, once daily, which may be increased to 10 mg once daily based on the assessment of tolerability after six months, to achieve optimal response (as was done in the titration arm of the POISE study); and the comparator is no treatment. For UDCA inadequate responders, the intervention is OCA dose titration as per above in combination with UDCA; and the comparator is UDCA monotherapy. Only patients in the moderate and severe risk categories were deemed eligible to receive OCA. UDCA doses implemented in the model were in line with its UK marketing authorisation, which are also the same as in the POISE trial. Fibrates were not considered as a comparator in the CS because fibrates are not licensed in the UK, nor are they standard of care, and they are contraindicated in PBC. According to the company, they are rarely used, with only

of patients in the UK-PBC cohort ever having taken fibrates for any condition (not necessarily for PBC).³ See Section 3.3 for more details.

ERG comment: Despite the low patient numbers that would have been treated with fibrates and the lack of evidence on fibrates, the decision to exclude fibrates as a comparator is inconsistent with the scope and may not be appropriate. See Section 3.3 for more details. The ERG also notes that there is an ongoing study on PBC patients treated with fibrates and that treatment with fibrates may present a viable comparator in the future.

5.2.5 Perspective, time horizon and discounting

The analysis was conducted from the perspective of the payer, i.e. the NHS England and Wales, over a time horizon of 50 years. Costs and outcomes were discounted by 3.5%.

ERG comment: This is in line with the NICE reference case.

5.2.6 Treatment effectiveness and extrapolation

Data from the POISE trial and various literature sources were used to estimate TPs between the health states in the biomarker component of the model (based on surrogate markers, i.e. ALP and bilirubin levels). The literature was used to estimate and extrapolate TPs for the health states in the liver disease component of the model.

Treatment effectiveness for health states in the biomarker component

Mortality in the biomarker component of the model was assumed to be equal to background mortality (National Life Tables, United Kingdom, 2012-2014; Office for National Statistics).⁵⁷ In addition, treatment discontinuation was assumed to only occur within the first three months of treatment. The company claimed that it was not possible to model treatment discontinuation beyond the first three months. Moreover, for discontinued patients, the TPs for UDCA or no treatment were assumed (independently of initial treatment) for UDCA non-responders and UDCA intolerant patients respectively.

Obeticholic acid with and without UDCA for inadequate UDCA responders and UDCA intolerant patients, respectively

For OCA, the transition probabilities during the first year between the biomarker component health states as well as treatment discontinuation (not shown in model structure Figure in CS; Figure 5.1) were based on the POISE trial (see Table 5.3). Here the company assumed that patients would only discontinue during the first three months of treatment given that the exact time of discontinuation in the POISE trial is unknown.

Given the low number of patients who received OCA monotherapy, the same transition matrices were used for the OCA titration regimen with or without UDCA (i.e. to UDCA tolerant and intolerant patients). The company states that this is reasonable as 'in clinical practice, OCA patients would be expected to have lower TPs to the more severe disease states'.

The company assumed no progression from low-risk or moderate risk to severe-risk after the first year. This assumption was, according to the company, justified based on the low decompensation rate observed in UDCA responders presented by Harms et al. (2015) in the PBC subgroup.⁵⁸ This study (details obtained from an oral presentation) included a total of 3,030 patients, of which 2,938 were UDCA-treated and showed that the risk of developing DCC was greater in UDCA non-responders (15-year risk of ~32%) than in UDCA responders (15-year risk of ~9%) (CS Figure 27).³

		0-3 months (N=68)					
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk						
	Moderate risk						
	Severe risk						
		3-6 months (N=68)					
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk				NA		
	Moderate risk				NA		
	Severe risk				NA		
		6-9 months (N=68)					
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk				NA		
	Moderate risk				NA		
	Severe risk				NA		
		9-12 months (N=68)					
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk				NA		
	Moderate risk				NA		
	Severe risk				NA		

Table 5.3: Transitions between health states of the biomarker component – OCA titration
		12 months+ (a	12 months+ (assumption)				
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk	1.000	0.000	0.000	NA		
	Moderate risk	0.000	1.000	0.000	NA		
	Severe risk	0.000	0.000	1.000	NA		
Source: CS Table 49 ³							
Footnote: Note that low risk corresponds to ALP ≤ 200 u/lL and TB ≤ 20 µmol/l; moderate risk corresponds to							
ALP > 200 u/I	L and TB \leq 20 μ mol/	l and; severe risk	corresponds to TB >	> 20 µmol/l			
NA, not applic	cable						

UDCA alone and no treatment for inadequate UDCA responders and UDCA intolerant patient respectively

For UDCA inadequate responders, the company did not estimate TPs using POISE trial data but opted instead for calibrating TPs (Table 5.4) based on PBC specific data from the literature considering 10 year liver transplant-free survival estimated using GLOBE and UK risk scores.⁵⁹ The justification for using a calibration approach was that transition probabilities were not available from POISE for all health states nor over the full time horizon of the model. The company did not provide a detailed account of their calibration method in the CS. In the response to the clarification letter,³² the company provided more detail and an Excel spreadsheet used for the calibration, which was done in four steps: The company calibrated 1) the transition from DCC to the pre-liver transplant health state and liver-related death; 2) transitions from the severe risk health state to the liver disease component health states; 3) transitions from the moderate risk health state to the severe risk health state and other liver disease component states and; 4) transitions from the low risk health state to the severe risk health state and other liver disease component states. There is little detail on the methods, the estimates used and the sources of data.

Given the limited number of patients not on UDCA (i.e. patients receiving no treatment), there is a lack of evidence on the natural history of PBC without active treatment since the launch of UDCA. Therefore, to reflect the progression of patients who did not receive any active treatment, the company made assumptions based on Corpechot et al. 2000,⁶⁰ who assessed the effect of UDCA therapy on liver fibrosis progression compared with no treatment in PBC patients (see Table 5.4).

		UDCA alone	UDCA alone (for UDCA inadequate responders)				
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk	NA ^a	NA ^a	NA ^a	NA		
	Moderate risk	0.000	0.969	0.031	NA		
	Severe risk	0.000	0.000	1.00	NA		
		No treatment	(for UDCA-intole	erant patients)			
		То					
		Low-risk	Moderate risk	Severe risk	Discontinuation		
From	Low-risk	NA ^a	NA ^a	NA ^a	NA		
	Moderate risk	0.000	0.932	0.068	NA		
	Severe risk	0.000	0.000	1.000	NA		

 Table 5.4: Transitions between health states of the biomarker component – UDCA alone

Source: CS Tables 50, 52 and 53³

Footnote: "No patients will transit to (or start in) the low-risk health state, hence these transition probabilities are not applicable.

Note that low risk corresponds to $ALP \le 200 \text{ u/l}$ and $TB \le 20 \text{ µmol/l}$; moderate risk corresponds to ALP > 200 u/l and $TB \le 20 \text{ µmol/l}$ and; severe risk corresponds to TB > 20 µmol/l NA, not applicable (or not reported in the CS)

ERG comment: The ERG's concerns regarding the estimation of treatment effectiveness for health states in the biomarker component are: 1) the usage of unclear calibration methods based on the literature instead of the POISE trial for the non-OCA regimen; 2) not exploring different ALP and TB thresholds for defining the health states; 3) assumption of no additional progression from low-risk or moderate risk to severe-risk after the first year in patients receiving OCA as well as no additional treatment discontinuation after the first year and; 4) discrepancy between the TPs reported in CS Table 49 and those used in the economic model.⁵⁵

Regarding the first 12 months in the model, the ERG considers the rationale provided by the company for not using the data from the POISE trial to estimate treatment effectiveness for the patients receiving UDCA only or no treatment as insufficient and inconsistent. The lack of long-term data in the POISE trial is not only an issue for the UDCA arm but as much so for the OCA treatment arm. The ERG thinks that very strong arguments must be present to neglect randomised evidence of comparative effectiveness in favour of observational evidence. Randomisation renders groups comparable not only in respect of known prognostic factors, but also with regard to unknown factors; hence observed differences in outcomes can be attributed to the treatment. The company did not however provide a justification for using the literature instead of the POISE patient-level data for modelling the effectiveness of UDCA and no treatment. Therefore, the ERG requested an analysis using the data from the POISE trial for UDCA and no treatment.³⁰ The company provided this analysis after further requests (post-clarification letter response). The ERG adopted this analysis in the ERG base-case (i.e. the POISE trial). See Table 5.5 for the TPs (assuming equal TPs for the UDCA and no treatment between the biomarker health states, as is done for the OCA regimen).

Considering the TPs after 12 months, the methods used by the company to estimate treatment effectiveness for the patients receiving UDCA only and no treatment are lacking transparency and justification (i.e. the calibration process described in CS Appendix 10 and in response to clarification question B14, as well as the estimation process described in response to clarification question B15). The additional Excel file provided by the company did not provide the necessary clarity as the values of interest are hard copied into the cells without the formulae to derive these. Moreover, the exact 10 year liver transplant-free survival used to calibrate the TPs is unclear to the ERG (in response to clarification question B14, the company mentions multiple probabilities; 10%, 78% and 91% without appropriate referencing). The methods and data used are poorly documented: the principal references are abstracts to posters without any data available. Furthermore, the company does not use the POISE comparator data where it is available for the start of the calibration process: instead of calibrating the POISE patient proportions with abnormal TB after one year of follow-up (10% in the UDCA and OCA treatment arms, and 6% in the OCA titration arm) on the 10 year liver transplant free survival (LTFS) data, the company uses patient proportions from another study, without justification.¹¹ Upon checking the reference, it does not appear that these patient proportions were available from the publication, but two year follow-up data suggest that 21% of patients had abnormal TB in that study. It is therefore unclear what estimate the company derived from this study and why evidence from the POISE study was not used. Without the necessary data and detailed description of methods that would have allowed verifying the methods and data used by the company for the long-term extrapolation, the ERG attempted to verify outcomes by cross-checking the model prediction with long-term survival in PBC patients based on Lammers et al. 2015.⁶¹ The ERG notes a relatively poor match between the number of deaths predicted by the company's model with the 10 year LTFS study⁶¹18.7% of tolerant patients in the UDCA treatment arm would have died after 7.8 years in the former, while 14.8% of patients were reported to have died at 7.8 years of follow-up in the latter. The ERG agrees in principle with the approach to calibrate the TPs based on external sources in this specific case where there is lack of long-term evidence. However, according to the ERG the company has failed to provide proper justification for their calibration method, to demonstrate that it is conducted correctly and that it results in plausible TPs. Therefore, with the exception of the TPs in the biomarker component that could be based on reported POISE trial results (i.e. for the first 12 months), the calibrated TPs are maintained in the ERG base-case although the ERG considers the approach a 'black box'. To show the impact of alternative assumptions, exploratory scenario analyses are provided by the ERG (this includes using TPs from TA330 in the liver component of the model).

		0-3 months (N	N=68) - inadequat	e UDCA respor	ıders
		То			
		Low-risk	Moderate risk	Severe risk	Discontinuation
From	Low-risk				
	Moderate risk				
	Severe risk				
		0-3 months (N	N=68) - UDCA inte	olerant patients	5
		То			
		Low-risk	Moderate risk	Severe risk	Discontinuation
From	Low-risk				NA
	Moderate risk				NA
	Severe risk				NA
		3-6 months (N=68) - equal for both comparators			
		То			
		Low-risk	Moderate risk	Severe risk	Discontinuation
From	Low-risk				NA
	Moderate risk				NA
	Severe risk				NA
		6-9 months (N	N=69) - equal for h	ooth comparato	ors
		То			
		Low-risk	Moderate risk	Severe risk	Discontinuation
From	Low-risk				NA
	Moderate risk				NA
	Severe risk				NA
		9-12 months	(N=69) - equal for	both comparat	tors
		То			
		Low-risk	Moderate risk	Severe risk	Discontinuation
From	Low-risk				NA
	Moderate risk				NA

Table 5.5: Transitions between health states of the biomarker component – non-OCA regimen

	Severe risk				NA			
		12 months+ - calibration	2 months+ - inadequate UDCA responders based on calibration					
		То						
		Low-risk	Moderate risk	Severe risk	Discontinuation			
From	Low-risk				NA			
	Moderate risk				NA			
	Severe risk				NA			
		12 months+ - UDCA intolerant patients based on calibration						
		То						
		Low-risk	Moderate risk	Severe risk	Discontinuation			
From	Low-risk				NA			
	Moderate risk				NA			
	Severe risk NA							
Source: CS Table 49 and economic model submitted by the company ⁵⁵								
Footnote: No	te that low risk corre-	sponds to ALP \leq	200 u/l and TB \leq 2	0 μmol/l; modera	te risk corresponds to			
ALP > 200 u/	/l and TB $\leq 20 \ \mu mol/$	l and; severe risk	corresponds to TB >	20 μmol/l				
NA, not appli	icable		-					

The company did not explore the impact of different thresholds for ALP and TB on the estimated cost effectiveness. It is unclear to the ERG what the impact would be of using different ALP and TB thresholds to define the health states in the biomarker component of the model.

For the OCA regimen, the justification for no additional progression from low-risk or moderate risk to severe-risk after the first year is questionable. Although the progression to DCC is higher for UDCA non-responders than for UDCA responders (which is the justification provided by the company), in both groups there is progression to DCC so assuming no progression (fixed, with life long duration) might not be justified by these data. Moreover, this is observed in UDCA responders treated with UDCA, hence it might be questioned whether this would also be applicable to PBC patients who were intolerant to or had inadequate response to treatment with UDCA and are now treated with OCA. In fact, it could be argued that POISE captures patients at varying stages of their disease. The TPs shown in Table 5.5 suggest that patients continue to move between health states even at 9-12 months. This is not surprising given that patients in the POISE trial had the condition for an average of eight years at the time they were enrolled in the trial and given that their treatment would not have changed. The ERG therefore considers there to be no strong justification to assume patients stop moving between the biomarker states after one year. Nevertheless, the clinical expert consulted by the ERG stated that this assumption was reasonable given that 'we know that patients with a response to UDCA and normal (or below 1.67x ULN) ALP have an excellent long term prognosis with no overall impact on life expectancy'. For OCA, the ERG therefore considers the possibility of patients continuing to move between the biomarker states as observed in the POISE study beyond the initial 12 months in an exploratory analysis (based on the abovementioned 15-year risk of DCC of $\sim 9\%$). Similarly, the ERG also considers that the assumption of no treatment discontinuation after 12 months is questionable. Moreover, it is unclear how the company calculated/retrieved the proportion (e.g. the reported in Table 5.3) of discontinued patients and why this was only considered in the first three months of the model. This is not appropriately justified. However, the company stated that the impact of treatment discontinuation is expected to be minor (see company response to clarification question B12f).

Finally, the ERG noticed that there was a discrepancy between the TPs reported in CS Table 49 and those used in the economic model. The adjusted TPs used in the economic model correspond to CS Table 49 (i.e. Table 5.3 above).

Treatment effectiveness for health states in the liver disease component

The TPs for the liver disease component of the model were equal for all comparators and mainly based on the literature (see Table 5.6). The company argued that it used the model structure from TA330⁶² to be consistent with other approaches for modelling the liver disease component. The company also refers to TA330⁶² for multiple TPs. These TPs are derived from patients with HCV. Only the TP from DCC to HCC was based on PBC patients.⁶³ The same calibration (described above for TPs for UDCA and no treatment in the biomarker component) was used for TPs from severe risk in the biomarker component to DCC, HCC and pre-LT, and DCC to pre-LT and death. This calibration aimed to reflect the 10-year liver transplant-free survival estimated based on POISE patient-level data using GLOBE and UK risk scores. No justification was provided by the company for the sources or methods used to derive these TPs.

From:	To:	Probability	Time (years)	Quarterly probability	Source
Severe risk	DCC	0.10	1	0.0260	Calibrated (CS Appendix 10)
	НСС	0.01	1	0.0035	Assumption
	Pre-LT	0.04	1	0.0102	Calibrated (CS Appendix 10)
DCC	Pre-LT	0.06	1	0.0153	Calibrated (CS Appendix 10)
	Death	0.17	1	0.0398	Calibrated (CS Appendix 10)
	НСС	0.01	1	0.0035	Trivedi et al. 2006 ⁶³
HCC	Pre-LT	0.04	1	0.0102	Wright et al. 2006 ⁶⁴
	Death	0.43	1	0.1311	Wright et al. 2006 ⁶⁴
Pre-LT	LT	0.35	1	0.1021	Kim et al. 2016 ⁶⁵
	Death	0.09	1	0.0233	Kim et al. 2016 ⁶⁵
LT	Death	0.21	1	0.0572	Wright et al. 2006 ⁶⁴
Post-LT	PBC recurrence	0.23	10	0.0064	Lindor, 2009 ⁶⁶
	Death	0.06	1	0.0146	Wright et al. 2006 ⁶⁴
	LT	0.01	13	0.0001	Neuberger, 2003 ⁶⁷
PBC recurrence	LT	0.01	13	0.0001	Assumption
Source: CS Tab	oles 50, 52 and 54	3			

Table 5.6: Transitions between health states of the liver disease component

CC, compensated cirrhosis; CS, company submission; DCC, decompensated cirrhosis; HCC, hepatocellular carcinoma; LT, liver transplant; PBC, primary biliary cholangitis/cirrhosis; TB, total bilirubin

ERG comment: The ERG generally agrees with the approach used in TA330⁶² for modelling liver disease in patients with HCV.⁶² It should be noted that neither the model structure of the liver disease component (see Section 5.2.2) nor the TPs of the liver disease component, used in the CS, are fully

consistent with TA330.⁶² The company did not provide any justifications for the sources that were used (if not based on the calibration approach). Moreover, for the TP from severe risk and DCC, the abovementioned ERG comments regarding the calibration also apply for the TPs in the liver disease component of the model. Therefore, the ERG used TPs consistent with TA330⁶² in an exploratory analysis (Table 5.7). This also entailed excluding the pre liver transplant state as discussed in Section 5.2.2. The only exception is the transition from DCC to HCC; this was based on the source provided by the company⁶³ instead of the source that was used in TA330⁶² (that is based on a study by Cardoso et al⁶⁸ based on hepatitis C (HBC) patients) as the source provided by the company was based on PBC patients.

From:	То:	Quarterly probability CS	Quarterly probability ERG exploratory scenario	ERG source
Severe risk	DCC	0.0260	0.0104	Cardoso ⁶⁸
	НСС	0.0035	0.0035	Assumption
DCC	Pre-LT	0.0102	NA	
DCC	Pre-LT	0.0153	NA	
	Death	0.0398	0.0663 ^b	Riemsma et al. 2016 ^{56c}
	HCC	0.0035	0.0035	Trivedi et al2009 ⁶³
	LT	NA	0.0055	Siebert et al. 2005 ⁶⁹
НСС	Pre-LT	0.0102	NA	
	Death	0.1311	0.1311	Fattovich et al. 1997 ⁷⁰ Shepherd et al. 2007 ⁷¹
	LT	NA	NA ^a	
Pre-LT	LT	0.1021	NA	
	Death	0.0233	NA	
LT	Death	0.0572	0.0572	Fattovich et al. 1997 ⁷⁰ Shepherd ⁷¹
Post-LT	PBC recurrence	0.0064	0.0064	Lindor ⁶⁶
	Death	0.0146	0.0146	Fattovich et al. 1997 ⁷⁰ Shepherd et al. 2007 ⁷¹
	LT	0.0001	0.0001 ^b	Neuberger, 2003 ⁶⁷
PBC recurrence	LT	0.0001	0.0001	Assumption
Footnotes: Note that the ^a No transition was assur	discrepant transitions ned between HCC and	are printed in bold . LT in TA330 ⁶²		

Table 5.7: Transitions between health states of the liver disease component based on TA330

^bThe ERG was unable to reproduce this number from the original source / the source provided by the company.

From:	То:	Quarterly probability CS	Quarterly probability ERG exploratory scenario	ERG source	
^c Original citation in the	CS62 and ERG report of	of TA330 ⁵⁶ was: 'Cheun	g M, Foster G, Irving	W, Walker A,	
Hudson B, Verma S. A	Antiviral treatment in p	atients with advanced I	HCV cirrhosis using	sofosbuvir and	
ledipasvir/ daclatasvir, v	with or without ribaviri	n – outcomes compared	to untreated patients	and long term	
outcomes. Presented at the European Association for the Study of the Liver (EASL); Barcelona. 13-17 April					
2016; Barcelona, Spain [PowerPoint Presentation].'					
CC, compensated cirrh	osis; DCC, decompens	sated cirrhosis; HCC, l	hepatocellular carcino	oma; LT, liver	
transplant: NA, not appli	cable: PBC, primary bil	iary cholangitis/cirrhosis	TB. total bilirubin		

5.2.7 Adverse events

The most commonly reported treatment-emergent adverse events in the POISE trial were pruritus followed by fatigue and nausea (Table 5.8). The company did not explicitly model the impact of these adverse events on health-related quality of life, as it was assumed by the company that their impact is captured as part of the health state utility values. Only the adverse event cost of pruritus were considered, because, according to the company, this is treated in routine clinical practice.

	Fatigue	Pruritus	Nausea	
No treatment	0.0000	0.0000	0.0000	
UDCA	0.1096	0.3699	0.0548	
OCA Titration 0.0857 0.5000 0.0429				
Source: economic model submitted by the company ⁵⁵				

Table 5.8: Incidence of AE

ERG comment: The economic impact of pruritus only was considered in the company's cost effectiveness model. The differential impact of adverse events on health-related quality of life was not considered by the company. This is not in line with best modelling practices. In the clarification letter (question B19), the ERG requested that the company incorporate the differential impact of adverse events on health-related quality of life. However, the company did not provide these analyses and stated that the impact of this scenario would be negligible. The company did not support this statement with a scenario analysis wherein the differential impact of adverse events on health-related quality of life was not support this assumption on the cost effectiveness remains uncertain.

5.2.8 Health-related quality of life

HRQoL evidence was not collected through a generic preference elicitation instrument in the POISE trial⁴⁵ and was therefore obtained from the literature.

Results of the literature review of HRQoL evidence

The company performed a systematic literature review to identify relevant HRQoL evidence for the current decision problem. Five studies were identified after abstract and full text screening (Figure 28 of the CS).^{50, 64, 72-74} Table 5.9 provides an overview of the included studies. A description of the search strategy is provided in Section 5.1.1.

 Table 5.9: Included HRQoL studies

Study, Country	Population	Study type	Metric	Result		
Aberg et al. 2012 ⁷²	72 PBC patients following liver transplant Sub-group of wider population	Observational	15D instrument	Utility value for entire PBC population following liver transplant: 0.882 Result found to be consistent regardless of aetiology		
Bondini et al. 2007 ⁷³	18 PBC patients	Observational	Chronic Liver Disease Questionnaire (CLDQ) SF-36 Health Utility Index (HUI Mark- 2 and Mark-3)	 18 patients in PBC group led to limited analysis in this population SF-36, CLDQ and HUI values reported HUI utility value reported as 0.81 (SD=0.1) in PBC population 		
Longworth et al. 2003 ⁵⁰	122 PBC patients assessed for transplantation and followed up for a maximum of 24 months post-surgery	Observational	EQ-5D	Full results reported in Table 57 of the CS ³		
Wright et al. 2006 ⁶⁴	204 chronic mild hepatitis C patients	Observational	EQ-5D	HRQoL results reported in Table 58 of the CS ³		
Younossi et al. 2001 ⁷⁴	120 patients with chronic liver disease, with 30% of patients suffering chronic cholestatic liver disease, including PBC and PSC patients	Observational	SF-36 CLDQ HUI	The study reported an HUI utility value of 0.84 ± 0.15 for patients with cholestatic liver disease.		
Source: Table 60 of t CLDQ, chronic live cholangitis/cirrhosis;	Source: Table 60 of the CS ³ CLDQ, chronic liver disease questionnaire; EQ-5D, EuroQOL-5 dimensions, HRQoL, health-related quality of life; HUI, health utility index; PBC, primary biliary cholangitis/cirrhosis; PSC, primary sclerosing cholangitis; SE-36, short form-36 dimensions					

HRQoL evidence used in the cost effectiveness model

In the company's cost effectiveness model, health state utility values have been obtained from the literature. Younossi et al. 2001⁷⁴ report utility values for patients having chronic liver disease due to viral infection(e.g. hepatitis B/C patients (HBV/HCV), cholestatic liver disease (i.e. PBC and primary sclerosing cholangitis patients) or other causes. Thirty-five patients with cholestatic disease were included in that study which provided the utility values for the low risk and moderate risk health states from the biomarker component of the model. No justification supported the use of this source in the CS. Wright et al. 2006⁶⁴ report utility values for chronic liver disease health states in hepatitis C patients, and was used for the remaining health states in the model (consistent with TA330⁶²). Moreover, a decrement was applied to the health state utility values found in the literature for all health states of the model, except for the three PBC health states and the HCC health state (Table 5.10). This decrement was based on expert opinion. In the CS it was stated that experts agreed that *'it is inappropriate to use [hepatitis C and B] HCV/HBV-specific utility values, since PBC patients are likely to have worse utility values despite being in the same health state'.*³ No further details were provided regarding the

identification of experts and the estimation of the decrement.

State	Utility value	Primary source [#]		
Low risk ^a	0.84	Younossi et al. 200174, 75		
Moderate risk ^b	0.84	Younossi et al. 200174		
Severe risk ^c	0.55	Wright et al. 2006 ⁶⁴		
Decompensated cirrhosis		Wright et al. 2006 ⁶⁴		
Hepatocellular carcinoma	0.45	Wright et al. 2006 ⁶⁴		
Pre-transplant: utility at listing		Wright et al. 2006 ⁶⁴		
Pre-transplant: 3 months after listing		Wright et al. 2006 ⁶⁴		
Pre-transplant: 6 months after listing		Wright et al. 2006 ⁶⁴		
Liver transplant: 3 months post-transplant		Wright et al. 2006 ⁶⁴		
Liver transplant: 6 months post-transplant		Wright et al. 2006 ⁶⁴		
Liver transplant: 12 months post-transplant		Wright et al. 2006 ⁶⁴		
Liver transplant: 24 months post-transplant		Wright et al. 2006 ⁶⁴		
Re-emergence of PBC	d	Wright et al. 2006 ⁶⁴		
Source: Table 61 of the CS^3 a referred as 'ALP: < 200 µ/L and Bili: Normal' i	in the CS^3			
^b referred as 'ALP: $\geq 200 \text{ u/L}$ and Bili: Normal' i	in the CS^3			
^c referred as 'Bili: Abnormal and rising, or CC' in	n the CS^3			
^d Assumed equivalent to liver transplant 24 me	onths post-transpl	ant, without utility decrement		
provided according expert opinion feedback				
[#] Wright et al. 2006^{64} is referred as TA 330 in the CS ³				
expert				
opinion .				
ALP, alkaline phosphatase; Bili, bilirubin; CC,	compensated cirrl	nosis; CS, company submission; PBC,		
primary biliary cholangitis/cirrhosis				

 Table 5.10: Health state utility values used in the company's cost effectiveness model

Impact of adverse events on HRQoL

The most frequent adverse events (AEs) were pruritus, fatigue and nausea. Their impact on HRQoL was not explicitly modelled in the company's cost effectiveness model because, according to the company, the impact was captured in the health state utility values.

ERG comment: The ERG's main concerns around HRQoL estimates are the appropriateness of Younossi et al. 2001⁷⁴ to inform the health state utility values for the low and moderate risk health states in the biomarker component, the validity of the **states** utility decrement applied to the health state utility values found in the literature for all health states in the liver component of the model, except for the HCC health state, and the omission of including AEs impact on HRQoL.

The ERG asked for justification of the use of Younossi et al. 2001⁷⁴ to inform the utility values for the low and moderate risk health states in the biomarker component of the model. In its response to Clarification Question B18b, the company explained that this was the only study reporting utility data for PBC patients.³² However, the ERG notes that this estimate is based on 35 patients who have either PBC or primary sclerosing cholangitis and that the study does not provide information on the number of PBC patients, PBC patients' demographic characteristics, treatment and treatment response. Due to the small number of patients on which the utility estimate is based and the impossibility to assess whether patients from this study are comparable to patients included in the current assessment, uncertainty remains about the health state utility values for the low risk and moderate risk health states. Furthermore, this study used the valuation set from Torrance et al. 1996⁷⁶ in order to value the Health Utilities Index Mark 2 (HUI:2) measurements. This valuation set is based on a sample of parents from Canada. This is not consistent with the NICE reference case and the representativeness of this estimate to the UK setting is uncertain. Given that a recent review from Dyson et al. 2016⁷⁷ reported that 34.1% of PBC patients have a 'poor' quality of life and that the current estimate is higher than the general population utility values for the age category 50-59 (the starting age of the population in the model being 55.8 years)⁷⁸, the ERG thinks that the current utility values for these health states might be overestimated. In absence of utility data specific to PBC patients which is representative of the UK setting, the ERG will use the age-dependent utility values of the UK general population for these two health states in its base-case analysis.⁷⁸ An additional concern is that the low risk and moderate risk health states were valued though the HUI:2 while the remaining health states were valued through the EQ-5D. This leads to inconsistencies in the valuation of health state utility values in the model. Using UK general population utility values will remove this concern.

The company applied a utility decrement to the health state utility values found in the literature for all health states in the liver component of the model, except for the HCC health state. This decrement was based on expert opinion. Appendix 11 of the CS contains the interview guide used to elicit expert opinion but it does not contain (a summary of) answers provided by experts on the different questions.⁴¹ The ERG requested these answers, but the company replied it was not authorised to share this information (Clarification Question B25).³² The ERG was consequently not able to investigate the validity of this decrement applied on the utility values from Wright et al.2006⁶⁴ (e.g. how the experts were selected, whether the clinical experts agreed on applying a decrement on the utility values obtained from the literature and whether experts agreed on the potential magnitude of the decrement). In addition, Younossi et al. 2001⁷⁴ report lower utility value for chronic liver disease due to viral infection (e.g. HCV/HBV) than the utility value reported for chronic liver disease due to cholestatic disease (PBC and primary sclerosing cholangitis patients). This is contradictory to the company's reasoning that PBC patients have lower health state utility values than HCV/HBV patients. For these reasons, the ERG removed the **autility** decrements in the ERG base-case.

The ERG requested the company to provide a sensitivity analysis including the impact of AEs on HRQoL. The company did not meet this request but provided a sensitivity analysis in which the costs of AEs were set to £0. According to the company, this underlined the minor impact of AEs on the cost effectiveness results (Clarification Question B19).³² The ERG thinks this approach does not represent the impact of AEs on HRQoL. Given the little differences in AEs occurrence between treatments (Section 5.2.7), the little influence this implementation will have on the cost effectiveness results and time constraints, the ERG will not include AEs' impact on HRQoL in its base-case analysis.

5.2.9 Resources and costs

Resource use and costs included in the CS model were based on data from the British National Formulary (BNF) 2016,⁷⁹ NHS reference costs, assumptions validated by expert opinion, and published sources identified in the systematic literature review (SLR).

Resource identification, measurement and valuation studies

The company performed a SLR to identify relevant costs and resource use evidence for the current decision problem. After abstract and full text screening, the SLR identified a total of 13 studies reporting cost and resource use data in Section 5.5.1.2 of the CS.³ All of these studies were deemed relevant to the decision problem. In the absence of any additional sources of evidence, assumptions were made for both cost and resource inputs included in the model, where necessary, and were validated by expert clinical opinion.

ERG comment: The ERG notes small inconsistencies in the numbers of studies identified in the different phases of the SLR³ but is satisfied that relevant literature has been identified.

Intervention and comparators' costs and resource use

Drug acquisition costs for UDCA were determined from the BNF 2016⁷⁹ and market share data to calculate a weighted average cost per pack. The daily cost of UDCA was based on the observed UDCA dose as defined in the POISE trial protocol. The company used the list price of OCA in the cost-effectiveness model. The outcomes in the CS were presented only based on the list price, analyses using the proposed PAS will be submitted in a separate appendix. The dosing frequency of OCA and UDCA used in the base-case analysis was based on the dosing schedule from the POISE trial for OCA (titration dose) and the marketing authorisation of UDCA.

The annual costs of drug administration for OCA and UDCA included in the model are presented in Section 5.5.2 of the CS,³ Table 65 (reproduced below in Table 5.11). OCA and UDCA are both administered orally.

Items	OCA	UDCA
Technology cost (list price)	£2,384.04 / 30 tablets	Weighted average of tablet formulations detailed in NHS Prescription Cost Analysis
Total annual cost (based on list price, 1 tablet per day)	£29,005.78	£655.33
Source: CS, Table 65 ³		

Table 5.11: Unit costs associated with the technology in the economic model

Health state costs and resource use

Data on health state costs and resource use were obtained from different sources and all costs are detailed in Table 5.12. All costs were inflated to 2016 costs using the Hospital & Community Health Services (HCHS) index⁸⁰.

Health states	Items	Value	Source
ALP: ≤ 200 U/l and Bili: Normal	Staff	£221.00 (1x Outpatient appointment, 1x outpatient follow-up)	EO
	Hospital costs	£27.00 (3 blood tests, 3 times per year, at a cost of £3)	NHS Schedule of Reference Costs – DAPS05 ⁸¹
	Total	£248.00	
ALP: > 200 U/l and Bili: Normal	Staff	£345.00 (1x Outpatient appointment, 2x outpatient follow-up appointments)	EO
	Hospital costs	£27.00 (3 blood tests, 3 times per year, at a cost of £3)	NHS Schedule of Reference Costs – DAPS05 ⁸¹
	Total	£496.00	
Bili: Abnormal and rising, or CC	Total	£6,254.00	EO: half the costs of DCC identified in Wright et al. 2006 ⁶⁴
DCC	Total	£12,509.00	Wright et al. 2006 ⁶⁴
НСС	Total	£11,147.00	Wright et al. 2006 ⁶⁴
Pre-transplant (end stage)	Total	£18,217.00	Singh et al. 2014 ⁴²
Re-emergence of PBC	Total	£248.00	Assumed identical to ALP: $\leq 200 \text{ u/L}$ and Bili: Normal
Liver transplant	Total	£65,029.00	Singh et al. 2014 ⁴²
Follow-up 1 year after liver transplantation	Total for 2 years divided by 2	£18,166.00	Singh et al. 2014 ⁴²
Follow-up 2 years after liver transplantation	Total for 2 years divided by 2	£18,166.00	Singh et al. 2014 ⁴²

Table 5.12: List of health states and associated annual costs in the economic model

Source: CS, Table 66³

ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; DCC, decompensated cirrhosis; EO, expert opinion; HCC, hepatocellular carcinoma; PBC, primary biliary cholangitis/cirrhosis

Costs associated with the liver disease component were based on expert opinion, the literature and NHS reference costs. Costs for outpatient visits were based on expert opinion rather than on NHS reference costs.

Costs associated with the severe risk biomarker health state of abnormal TB or CC was estimated by an expert to be half of the cost of DCC (reported in Wright et al. 2006⁶⁴). The company justified this by stating that rising bilirubin levels would cause use of more health care resources.

The cost of liver transplant was derived from Singh et al. 2014⁴², which, the company claimed, was in line with TA 330⁵⁶. Furthermore, the company claimed that using NHS reference costs for an estimate

of liver transplant costs would under-estimate the cost of liver transplant, because it would not include the additional cost of supporting care.

Adverse reaction unit costs and resource use

The company considered costs associated with pruritus only and excluded costs of fatigue and nausea from their analysis. The company justified this by claiming that only pruritus was routinely treated in clinical practice. An overview of costs associated with pruritus care is shown in Table 5.13.

Adverse events	Items	Percentage of patients cost applies to	Value	Reference to section in submission
Pruritus	Staff (GP visit)	100%**	$\pounds 54.00^{*}$	EO
	Cholestyramine cost / 327.10 days [†]	85%**	£105.59 [‡]	EO
	Rifampicin cost / 327.10 days [†]	15%**	£191.77 [‡]	EO
	Naltrexone cost / 327.10 days [†]	5%**	£228.39 [‡]	EO
	Total (weighted average + staff costs)	N/A	£177.75	EO
Source: CS, Table	67 ³			1

Table 5.13: List of adverse reactions and summary of costs included in the economic model

Footnotes: * Sourced from BNF, 2016. **Sourced from EO, Mean duration of treatment for UDCA and OCA therapies; [‡] Sourced from NHS Schedule of Reference Costs

CS, company submission; EO, expert opinion; GP, general practitioner.

ERG comment: In their original model, the company had used expert opinion for estimating costs associated with outpatient visits. The company then provided a scenario analysis where NHS reference costs were used instead. The ERG was satisfied that the use of NHS reference costs had only a small effect on model outcomes. The ERG uses NHS reference costs for its base case.

The company did not provide adequate justification how the cost associated with the severe risk health state of abnormal TB and CC was estimated. The ERG is concerned that the assumption of halving the cost of DCC is arbitrary. The ERG therefore uses the cost for CC used in TA 330⁶², which was £1,561 instead of £6,254 used by the company⁵⁶ in its base case.

The company's estimate of liver transplant related costs was based on cost data in patients with hepatitis C^{42} . Costs of liver transplant in the PBC population also exist (Longworth et al. 2003)⁵⁰. If these costs based on PBC were inflated to 2015 values, costs would be similar to those used by the company (approximately £66,000 compared with £65,029 used by company). However, there is reason to believe that liver transplant costs for PBC patients should be lower than those for hepatitis C patients (Longworth et al. 2003)⁴². Furthermore, as the number of liver transplants has increased, the impact of scale economies as considered likely by Habka et al. 2015⁸² suggest that a lower inflation uplift may be more appropriate. The ERG therefore explores alternative costs of liver transplant in an exploratory analysis.

5.2.10 Cost effectiveness results

In both populations, OCA (with or without UDCA) led to longer survival, a QALY gain, and higher costs than the comparators (i.e. no treatment for the UDCA-intolerant population and UDCA alone for the UDCA inadequate responders). The company provided disaggregated results for QALYs (Tables 5.14 and 5.15) and costs by health state (Tables 5.16 and 5.17). The low risk and moderate risk health states contribute to 74% and 73% of the absolute incremental QALY gained for the UDCA-intolerant population and UDCA inadequate responders. The incremental costs per health state follow the same trend as the incremental QALYs gained per health state. The low risk and moderate risk health states account for 85% and 87% of the incremental costs for the UDCA-intolerant population and UDCA inadequate responders. Disaggregated results were not provided for life years (LYs) gained. At list price, OCA was associated with an incremental cost effectiveness ratio (ICER) of per additional QALY gained in the UDCA-intolerant patients population and an ICER of in the UDCA inadequate responder patients population (Tables 5.18 and 5.19).

Health state	OCA Titration	No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a	5.981	0.000	-5.981	5.981	41%
Moderate risk ^b	6.796	2.044	-4.752	4.752	33%
Severe risk ^c	0.394	2.515	2.121	2.121	15%
Discontinuation	0.009	0.000	-0.009	0.009	0%
Decompensated cirrhosis	0.106	0.651	0.544	0.544	4%
НСС	0.011	0.065	0.055	0.055	0%
Pre transplant (end stage)	0.034	0.208	0.174	0.174	1%
Liver transplant	0.005	0.031	0.026	0.026	0%
Post liver transplant	0.136	0.822	0.686	0.686	5%
Re-emergence of PBC	0.046	0.270	0.223	0.223	2%
Total	13.520	6.606	-6.913	14.572	100%

Table 5.14: Summary	v of (DALY	gain b	v health	state for	UDCA	-intolerant	no	nulation
Tuble Sti It Summar	, ,, ,		Sam N	y mountin	State 101		monthunt	PV	Junation

Source: Table 73 of the CS³

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS³; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

Table 5.15: Summary of	f QALY gain	by health state for	r UDCA inadequate	responders
------------------------	-------------	---------------------	-------------------	------------

Health state	OCA + UDCA titration	UDCA + No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a	5.981	0.000	-5.981	5.981	48%
Moderate risk ^b	6.974	3.867	-3.107	3.107	25%
Severe risk ^c	0.365	2.222	1.857	1.857	15%
Discontinuation	0.009	0.004	-0.006	0.006	0%
Decompensated cirrhosis	0.098	0.569	0.471	0.471	4%

Health state	OCA + UDCA titration	UDCA + No treatment	Increment	Absolute increment	Absolute increment (in %)*
НСС	0.010	0.057	0.048	0.048	0%
Pre transplant (end stage)	0.031	0.182	0.151	0.151	1%
Liver transplant	0.005	0.027	0.023	0.023	0%
Post liver transplant	0.125	0.701	0.577	0.577	5%
Re-emergence of PBC	0.042	0.222	0.180	0.180	1%
Total	13.641	7.852	-5.789	12.399	100%

Source: Table 74 of the CS³

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS³; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

Health state	OCA Titration	No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a		£0.00			
Moderate risk ^b		£1,207.08			
Severe risk ^c		£28,597.68			
Discontinuation		£0.00			
Decompensated cirrhosis		£21,284.77			
НСС		£1,620.43			
Pre transplant (end stage)		£9,890.90			
Liver transplant		£14,310.87			
Post liver transplant		£26,221.30			
Re-emergence of PBC		£99.80			
Total		£103,232.81			

Fable 5 1(. Cummerson	. of coata b	u haalth atata fau	UDCA intelement	mamulation
i adie 5.10: Summar	v of costs dy	у пеани згате тог	ОЛСА-шоегаш	DODUIALION
	,	,		population

Source: Table 75 of the CS³

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS³; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

Health state	OCA + UDCA titration	UDCA + No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a		£0.00			
Moderate risk ^b		£5,349.00			
Severe risk ^c		£27,926.48			
Discontinuation		£14.48			
Decompensated cirrhosis		£18,619.88			
НСС		£1,422.13			
Pre transplant (end stage)		£8,665.20			
Liver transplant		£12,526.26			
Post liver transplant		£22,371.06			
Re-emergence of PBC		£82.10			
Total		£96,976.58			

Table 5.17: Summary of costs by health state for UDCA inadequate responders

Source: Table 76 of the CS³

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS³; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

 Table 5.18: Deterministic base-case results for UDCA-intolerant patients, using the list price for OCA

Technologies	Total			I	ncreme	ntal	ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment	£103,233	11.30	6.61					
OCA titration		16.65	13.52		5.35	6.91		
Source: Table 69 of the CS ³								

CS, company submission; ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid

Technologies		Total		I	ncreme	ntal	ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA	£96,977	12.35	7.85					
OCA titration + UDCA		16.75	13.64		4.40	5.79		
Source: Table 70 of the CS ³								
CS, company su	ibmission; 1 DALYs qua	ICER, in ulity-adius	cremental of sted life year	cost-effec	tiveness	ratio; LYC	, life years	gained; OCA,

 Table 5.19: Deterministic base-case results for UDCA inadequate responders, using the list price for OCA

ERG comment: Upon request from the ERG, the company provided the disaggregated LYs gained per health state (Tables 5.20 and 5.21). As for the disaggregated QALYs gained and disaggregated costs per health state, the incremental LYs gained in the low risk health state are the same in both population. In addition, the low risk and the moderate risk health states contribute to more than half of the absolute increment in LYs gained, as expected when examining the incremental QALYs gained (respectively 63% and 62% of the incremental LY for the UDCA-intolerant population and UDCA inadequate responders).

The cost effectiveness results reported in the current report are potentially not representative of the true value for money of OCA (with or without UDCA) for treating PBC patients if OCA will be provided under a patient access scheme (PAS).

		·			
Health state	OCA titration	No treatment	Increment	Absolute increment [*]	Absolute increment (in %) [*]
Low risk ^a	7.120	0.000	7.120	7.120	35%
Moderate risk ^b	8.091	2.434	5.657	5.657	28%
Severe risk ^c	0.716	4.573	-3.856	3.856	19%
Discontinuation	0.012	0.000	0.012	0.012	0%
Decompensated cirrhosis	0.278	1.702	-1.423	1.423	7%
НСС	0.024	0.145	-0.122	0.122	1%
Pre transplant (end stage)	0.089	0.543	-0.454	0.454	2%
Liver transplant	0.009	0.055	-0.046	0.046	0%
Post liver transplant	0.240	1.443	-1.204	1.204	6%
Re-emergence of PBC	0.069	0.402	-0.333	0.333	2%
Total	16.648	11.297	5.351	20.227	100%

Table 5.20: Summary of life years gained by health state for UDCA-intolerant population

Health state	OCA titration	No treatment	Increment	Absolute increment*	Absolute increment (in %)*			
Source: Table 33 of the	response to the c	larification letter ³²						
Footnotes: a referred as '	ALP: $\leq 200 \text{ u/L}$	and Bili: Normal' in	the CS ³ ; ^b referr	ed as 'ALP: > 20	00 u/L and Bili:			
Normal' in the CS ³ ; ^c ref	erred as 'Bili: A	bnormal and rising, o	or CC' in the CS ²	³ ; * Calculated by	the ERG			
ALP, alkaline phosphat	tase; Bili, biliru	bin; CC, compensat	ted cirrhosis; CS	S, company sub	mission; ERG,			
Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary								
cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of								
normal								

Health state	OCA titration + UDCA	UDCA + No treatment	Increment	Absolute increment [*]	Absolute increment (in %)*
Low risk ^a	7.120	0.000	7.120	7.120	41%
Moderate risk ^b	8.303	4.604	3.699	3.699	21%
Severe risk ^c	0.664	4.040	-3.376	3.376	20%
Discontinuation	0.012	0.005	0.007	0.007	0%
Decompensated cirrhosis	0.257	1.489	-1.231	1.231	7%
НСС	0.022	0.128	-0.106	0.106	1%
Pre transplant (end stage)	0.082	0.476	-0.394	0.394	2%
Liver transplant	0.008	0.048	-0.040	0.04	0%
Post liver transplant	0.219	1.231	-1.013	1.013	6%
Re-emergence of PBC	0.062	0.331	-0.269	0.269	2%
Total	16.750	12.351	4.399	17.255	100%

Table 5.21: Summary of life years gained by health state for UDCA inadequate responders

Source: Table 34 of the response to the clarification letter³²

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS³; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

5.2.11 Sensitivity analyses

Probabilistic sensitivity analysis

The company performed a probabilistic sensitivity analysis (PSA) on the following parameters:

- Baseline distribution of patients in the 'ALP > 200 units / L and Normal Bilirubin' and '(ALP \leq or > 200 units / L) and Abnormal Bilirubin' health states
- Patients' mean weight
- Discontinuation probabilities
- Adverse event rates
- Utilities
- TPs

- Disease management costs
- Adverse event costs

To perform the PSA, the minimum and maximum values of several parameters were assumed to be +/-20% of the mean estimate. This is the case for the health state utility values decreased by the decrement, disease management costs and the following TPs: from the severe risk health state to DCC health state, from the sever risk health state to HCC, from HCC to death, from Pre-LT to LT and from Pre-LT to death. Results of the PSA for both populations (UDCA-intolerant patients and UDCA inadequate responders) compared to the deterministic results are provided in Tables 5.22 and 5.23. At list price, the probability for OCA of being cost effective at a willing-to-pay threshold of £30,000 was in both populations. Appendix 3 provides the scatter plots and cost effectiveness acceptability curves for both populations.

Table 5.22: Incremental cost effectiveness results of PSA for the UDCA intolerant populatio	n,
using the list price of OCA	

Total		Ir	ncremer	ntal	ICER (£)	ICER (£)			
Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental		
PSA results									
103,439	11.33	6.77	-	-	-	-	-		
	16.64	13.52		5.31	6.75				
nistic res	ults								
103,233	11.30	6.61	-	-	-	-	-		
	16.65	13.52		5.35	6.91				
the CS^3									
1 1 1	03,439 03,233 03,233 ne CS ³	Item Item 03,439 11.33 Item 16.64 Item 11.30 Item 16.65 Item 16.65	Iteration Dosts (£) LYG QALYs 03,439 11.33 6.77 16.64 13.52 1100000000000000000000000000000000000	Item Item Item Item Dosts (£) LYG QALYs Costs (£) 03,439 11.33 6.77 - Item 16.64 13.52 Item Distic results 03,233 11.30 6.61 - Item CS ³	Iteration Incrementation Dosts (£) LYG QALYs Costs (£) LYG 03,439 11.33 6.77 - - Interementation Interementation Interementation Interementation 03,439 11.33 6.77 - - Interementation Interementation Interementation Interementation 03,439 11.33 6.77 - - Interementation Interementation Interementation Interementation 03,439 11.33 6.77 - - Interementation Interementation 5.31 Interementation 03,233 11.30 6.61 - - Interementation Interementation 5.35 Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation Interementation </td <td>Iteration Incremental Dosts (£) LYG QALYs Costs (£) LYG QALYs 03,439 11.33 6.77 - - - 16.64 13.52 5.31 6.75 istic results 5.35 6.91 16.65 13.52 5.35 6.91 ne CS³ $4 + 4 + 5$ $5 + 4 + 4 + 5$ $5 + 4 + 4 + 5$</td> <td>Incremental Incremental ICER (£) Dosts (£) LYG QALYs Costs (£) LYG QALYs versus baseline $03,439$ 11.33 6.77 - - - 16.64 13.52 5.31 6.75 6.75 istic results $03,233$ 11.30 6.61 - - - 16.65 13.52 5.35 6.91 6.91 6.91 6.91 16.65 13.52 6.75 5.35 6.91 6.91 6.91</td>	Iteration Incremental Dosts (£) LYG QALYs Costs (£) LYG QALYs 03,439 11.33 6.77 - - - 16.64 13.52 5.31 6.75 istic results 5.35 6.91 16.65 13.52 5.35 6.91 ne CS ³ $4 + 4 + 5$ $5 + 4 + 4 + 5$ $5 + 4 + 4 + 5$	Incremental Incremental ICER (£) Dosts (£) LYG QALYs Costs (£) LYG QALYs versus baseline $03,439$ 11.33 6.77 - - - 16.64 13.52 5.31 6.75 6.75 istic results $03,233$ 11.30 6.61 - - - 16.65 13.52 5.35 6.91 6.91 6.91 6.91 16.65 13.52 6.75 5.35 6.91 6.91 6.91		

CS, company submission; ICER, incremental cost-effectiveness ratio; LYG, life years gained; PSA, probabilistic sensitivity analysis; QALY, quality-adjusted life year; OCA, obeticholic acid; UDCA, ursodeoxycholic acid

Table 5.23: Incremental cost effectiveness results of PSA for the UDCA inadequate responde	er
population, using the list price of OCA	

Technologies	Total		Incremental			ICER (£)	ICER (£)	
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	increment al
PSA results								
UDCA	£97,044	12.38	7.88	-	-	-	-	-
OCA + UDCA titration		16.75	13.65		4.37	5.77		
Base case deter	ministic re	sults						
UDCA	£96,977	12.35	7.85	-	-	-	-	-
OCA + UDCA titration		16.75	13.64		4.40	5.79		

Technologies	Total			In	crement	tal	ICER (£)	ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	versus baseline	increment al
Source: Table 80	of the CS ³							
CS, company submission; ICER, incremental cost-effectiveness ratio; LYG, life years gained; PSA, probabilistic								
sensitivity analysi	is; QALY, qı	uality-adj	usted life ye	ear; OCA, ob	eticholic	acid; UDCA	, ursodeoxyo	cholic acid

Deterministic sensitivity analyses

The influence of varying the following parameters was investigated through one-way deterministic sensitivity analyses (DSA):

- Discontinuation probabilities
- Probabilities of experiencing an AE
- Utility weights
- TPs
- Discounting
- The probability of death for the general population
- AEs costs and disease management costs (i.e. health state costs)

Tables 81 and 82 of the CS provide an overview of the variation in each model parameter used during the DSA for the UDCA-intolerant population and the UDCA inadequate responders population.³ In both populations, the most influential parameters on the cost effectiveness results were the health states utility values for the health states of the biomarker component of the model and the TPs between these health states. Tornado diagrams were provided in the CS (Appendix 2).⁴¹

In the DSA, the following parameters were varied by +/-20%:

- AEs costs and disease management costs
- Utility weights decreased by the decrements
- TPs concerning the biomarker component of the model
- The following TPs from the liver disease component of the model: DCC to LT, DCC to death, DCC to HCC, severe risk to DCC, severe risk to HCC, HCC to death, Pre-LT to LT, Pre-LT to death, LT to death, Post LT to re-emergence of PBC, Post LT to LT, Post LT to death, Re-emergence of PBC to LT, HCC to LT.

ERG comment: The variation around a substantial number of parameters is based on a +/-20% variation around the mean which does not reflect the parameter uncertainty around these parameter estimates. Furthermore, using an arbitrary variation around the mean is not in line with international modelling guidelines.⁸³ Empirical evidence should have been used to determine the variation around the estimates for the DSA.

Scenario analyses

The company performed two scenario analyses. In the first scenario analysis, the original HCV utility values (without decrease) were used for the health states of the liver disease component of the model. This resulted in ICERs of decrease) for the uDCA-intolerant patients and the UDCA inadequate responder respectively. In the second scenario analysis, the TPs from the pre-LT health state to the LT health state, and from the pre-LT health state to death were based on PBC-specific data from the Organ procurement and transplantation network (OPTN) (data on file).³ The TPs were changed in 0.44 and 0.21 instead of 0.35 and 0.09, respectively. This resulted in ICERs of

for the UDCA intolerant patients and the UDCA inadequate responders

respectively.

ERG comment: In response to the ERG requests, the company provided the results of three additional scenario analyses. The first additional scenario analysis concerns the discontinuation rate and the second concerns the implementation of the HRQoL impact of AEs in the cost effectiveness model. The third additional sensitivity analysis presented by the company included NHS reference costs for outpatients visits.³² These additional sensitivity analyses are discussed in Sections 5.2.6, 5.2.8 and 5.2.9 respectively.

Upon the ERG's request, the company provided an additional analysis in which the TPs of all comparators in the first year of the model were based on the patient-level data from POISE. This approach is considered more consistent by the ERG and results in ICERs of

for the UDCA-intolerant patients and the UDCA inadequate responders respectively (Tables 5.24 and 5.25).

Table 5.24: Deterministic results for UDCA-intolerant patients, using the list price of OCA and UDCA patient-level data

Technologies	Total			Incremental			ICER (£)	ICER (£)
	Costs (£)	LY	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
No treatment	£103,233	11.30	6.61					
OCA titration		16.68	13.56		5.38	6.95		
CS, company submission; ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; QALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid								

Table 5.25: Base-case results for UDCA inadequate responders, using the list price of OCA, and UDCA patient-level data

Technologies	Total			Incremental			ICER (£)	ICER (£)
	Costs (£)	LY	QALYs	Costs (£)	LYG	QALYs	versus baseline	incremental
UDCA	£92,218	12.72	8.30					
OCA+UDCA Titration		16.78	13.68		4.06	5.38		
CS, company submission; ICER, incremental cost-effectiveness ratio; LYG, life years gained; OCA, obeticholic acid; OALYs, quality-adjusted life years; UDCA, ursodeoxycholic acid								

5.2.12 Model validation and face validity check

The company described that the model had been internally checked by two modellers who were involved in the model development (internal peer review) and that an external peer review took place (performed by two modellers who were not involved in the development of the model). The company stated that the following verification techniques were performed during these peer reviews:

- 'Face validity: testing that the model meets expectations based on simple calculations
- Model behaviour: testing whether varying model inputs has the expected directional effect
- Internal consistency: model outputs were compared against POISE
- Cell-by-cell checks of calculations: manual inspection of formulae

- Use of logical scenario checks and the rebuilding of important parts of the model
- A complete cross-check of inputs, sources, and supporting documentation.³

The company provided the clinical outcomes of the cost effectiveness model (Tables 5.26 to 5.28), but did not compare them to any external source.

	UDCA intolera	nt patients	UDCA inadequate responders		
	No treatment	OCA titration	UDCA + No treatment	OCA + UDCA titration	
Annual rate of liver-related death	0.047	0.004	0.039	0.004	
Mean liver-related death free survival (years)*	19.365	44.605	21.637	44.893	
Years lost to liver-related death ^{\dagger}	30.635	5.395	28.363	5.107	
Median time to liver-related death (years) [‡]	14.750	N/A	17.500	N/A	
Source: Table 55 of CS^3					

Table 5.26: Mean time to event for liver-related death

Source: Table 55 of CS³

Footnotes: *Area under the Kaplan Meier curve; [†] Area above the Kaplan Meier curve; [‡] Using the total population as done with transplant free survival

Table 5.27: Summary	v of model	outcomes for	UDCA-intolerant	population
i ubic ola / i buillinui	, or mouch	outcomes for	OD OIL metre and	population

Outcome	OCA titration	No treatment	Incremental				
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients							
Total number of cases of decompensated cirrhosis per 1,000 patients							
Total liver transplants per 1,000 patients							
Total liver related deaths per 1,000 patients							
Source: Table 71 of the CS ³							
Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; OCA, obeticholic acid; UDCA,							
ursodeoxycholic acid							

Outcome	OCA titration + UDCA	UDCA	Incremental					
Total number of cases of Bili: abnormal and rising, or CC per 1,000 patients								
Total number of cases of decompensated cirrhosis per 1,000 patients								
Total liver transplants per 1,000 patients								
Total liver related deaths per 1,000 patients								
Source: Table 72 of the CS ³ Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; OCA, obeticholic acid; UDCA, ursodeoxycholic acid								

Table 5.28: Summary of model outcomes for UDCA inadequate responder population

ERG comment: The results of the validation steps were not described in the CS and were not provided in response to the clarification letter upon the ERG's request. Furthermore, the ERG requested the company to perform cross and external validation (for example, by using the UK-PBC Risk Score Calculator⁸⁴). Validation against the UK-PBC Risk Score was not performed. The company stated that it required patient-level data and calculating mean ALP or bilirubin levels per group would potentially introduce bias. The company deemed cross validation impossible in the absence of other cost utility models for PBC.³² The ERG does not agree with this, because the cost effectiveness review identified four studies focussing on PBC treatment. The ERG believes that other outcomes of the model (e.g. UDCA costs, LY for the No treatment and UDCA arms) could have been validated against these sources. If the company considered comparisons with these models inappropriate, justifications should have been provided for not performing cross validation (or why it was deemed inappropriate).

The company claimed to have validated model outputs against the GLOBE and UK PBC cohort survival rates, shown in Harms et al. 2016,⁵⁹ but did not provide the results of this validation.³² The lack of information concerning validation efforts (both in the CS and the response to the clarification letter) is a violation of good modelling practices⁸⁵ and it hampered the ERG's efforts to assess whether adequate validation efforts have been performed by the company.

Moreover, given the lack of transparency regarding the methods used by the company (e.g. concerning the calibration process described in Section 5.2.6), even after requests from the ERG (Clarification Questions B12, B14, B15 and B16³⁰), the methods used by the company are still unclear. The ERG was therefore unable to assess the validity and appropriateness of these methods.

5.3 Exploratory and sensitivity analyses undertaken by the ERG

Based on all considerations from Section 5.2, the ERG defined a new base-case. This base-case included multiple adjustments to the original base-case presented in the CS. These adjustments were subdivided into three categories (derived from Kaltenthaler 2016⁸⁶):

- Fixing errors (correcting the model were the company's submitted model was unequivocally wrong)
- Fixing violations (correcting the model where the ERG considered that the NICE reference case, scope or best practice had not been adhered to)

• Matters of judgement (amending the model were the ERG considers that reasonable alternative assumptions are preferred)

Additionally, multiple exploratory sensitivity analyses (see Section 5.3.2) were performed by the ERG to examine the potential impact of alternative assumptions on the cost effectiveness estimates.

Fixing errors

1. Fixing discrepancy between TPs reported in the CS and used in the economic model (for biomarker component of the model; see Section 5.2.6 for more details). To fix this error, the ERG used the TPs reported in the CS.

Fixing violations

- 2. Use TPs from POISE for the non-OCA regimen (for biomarker component of the model; see Section 5.2.6 for more details).
- 3. Use proportions in the starting health states as derived from the POISE trial (see Section 5.2.3 for more details).
- 4. Use NHS reference costs for outpatient visits (see Section 5.2.9 for more details).
- 5. Use health state costs of £1,561 for CC (severe risk health state in the biomarker component) (instead of £6,254, consistent with TA330 (see Section 5.2.9 for more details)
- 6. Use age-dependent utilities (from the UK general population) for the low and moderate risk health states in the biomarker component of the model (see Section 5.2.8 for more details).

Matters of judgment

7. Remove the HRQoL decrements (for health states in the liver disease component of the model; see Section 5.2.9 for more details).

5.3.1 Probabilistic sensitivity analysis (ERG base-case)

The ERG performed a PSA to obtain the ERG base-case incorporating all abovementioned adjustments. This resulted in ICERs (probabilistic) for OCA versus UDCA or no treatment of

for UDCA inadequate responders and UDCA intolerant patients respectively.

The PSA showed that OCA has a probability of being cost-effective at thresholds below £30,000 per QALY gained (Figure 5.2).

	OCA titration+UDCA			UDCA				Incremental results			
	QALYs	LYs	Costs	QALYs	LYs	Costs	ΔQALY	ΔLΥ	ΔCosts	ICER LY	ICER QALY
ERG base-case (deterministic)	12.57	16.98		8.39	13.11	£72,332	4.17	3.88			
ERG base-case (probabilistic)	12.51	16.95		8.29	13.13	£72,266	4.22	3.82			
ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA,											
ursodeoxycholic acid											

Table 5.29: ERG base-case for UDCA inadequate responders

Table 5.30: ERG base-case for UDCA intolerant patients

	OCA titration			No treatment			Incremental results				
	QALYs	LYs	Costs	QALYs	LYs	Costs	ΔQALY	ΔLΥ	ΔCosts	ICER LY	ICER QALY
ERG base-case (deterministic)	12.44	16.86		7.05	11.79	£78,461	5.38	5.07			
ERG base-case (probabilistic)	12.40	16.84		6.93	11.81	£78,541	5.47	5.03			
ERG, Evidence Review grou ursodeoxycholic acid	ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid										

Figure 5.2: Incremental cost effectiveness planes for ERG base-case

UDCA inadequate responders



UDCA intolerant patients



5.3.2 Additional exploratory and threshold analyses performed based on the ERG base-case

Four additional exploratory sensitivity analyses were performed by the ERG to examine the potential impact of alternative assumptions on the cost effectiveness estimates. These analyses were performed on the ERG base-case and investigated the impact of the following adjustments:

8. Use TPs from TA330 (for liver disease component) and accordingly exclude pre-liver disease health state and update health sate costs for liver transplant

- 9. Assume that for the non-OCA regimen, after 12 months, the TPs in the biomarker component of the model are based on the POISE trial (using the average TPs over the first 12 months).
- 10. Assume that for the OCA regimen, after 12 months, the TPs between the health states in the biomarker component of the model are 0.16% per quarter based on the progression observed in UDCA responders (instead of assuming no additional progression from low-risk or moderate risk to severe-risk after the first year).
- 11. Use alternative costs for liver transplant (£57,777 instead of £65,029).

The exploratory analysis all resulted in increased ICERs (Tables 6.3 and 6.4). The use of the TPs from TA330 resulted in ICERs of for UDCA inadequate responders and UDCA intolerant patients respectively. Assuming alternative TPs for the non-OCA regimen after 12 months, resulted in ICERs of for UDCA inadequate responders and UDCA intolerant patients respectively. Adding progression in the biomarker component of the model for the OCA regimen resulted in ICERs of for UDCA intolerant patients respectively. Finally, lowering the liver transplant costs resulted in a minor increase in ICERs to for UDCA inadequate responders and UDCA intolerant patients respectively.

5.4 Conclusions of the cost effectiveness section

The majority of searches in Appendix 6, 8 and 9 were well reported and easily reproducible. Searches covered a broad range of databases and grey literature resources. The ERG expressed some concerns regarding the use of study design filters. However, it is unlikely that this would have impacted on the overall recall of results.

Reviewing the overall evidence, the ERG confirmed that there was no existing cost effectiveness model for OCA for the current indication, and thus that development of a *de novo* model was necessary. The economic model described in the CS is considered by the ERG to partially meet the NICE reference case. The company developed a Markov state transition model to describe the progression of PBC over a lifetime horizon. The model comprises two parts with a total of 10 health states. The first part of the model (referred as the biomarker component) captures the surrogate outcomes of ALP and bilirubin biomarkers in three different health states based on the expected risk of disease progression: low risk (ALP \leq 1.67 * ULN); moderate risk (ALP > 1.67 * ULN and TB \leq 1.0 * ULN) and severe risk (TB > 1.0 * ULN or compensated cirrhosis; CC). The latter health state combined both patients with compensated cirrhosis and those with abnormal bilirubin. In the second part of the model, the liver disease component, the following clinical endpoints are modelled: pre-liver transplant; DCC; HCC; liver transplant; a post-liver transplant state; potential PBC re-emergence; and death. Patients can move to the liver disease component of the model only coming from the severe risk health state in the biomarker component of the model. Furthermore, relevant adverse events identified by the company were fatigue, pruritus and nausea, but costs were only modelled in relation to pruritus, and the differential impact on health-related HRQoL was not modelled.

The company base-case ICERs (probabilistic) of OCA (without or with UDCA) versus no treatment and UDCA monotherapy were for UDCA intolerant patients and UDCA inadequate responders respectively. The one-way sensitivity analyses conducted by the company showed that the model results were most sensitive to the health states utility values for the health states of the biomarker component of the model and the TPs between these health states. It should be noted though that the one-way sensitivity analyses were often based on arbitrary estimates of the variance (i.e. using $\pm 20\%$ of the mean value). The main areas of ERG concern were the fact that the company did not use the POISE trial data directly for the non-OCA regimen, the lack of transparency and justification regarding methods and data to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes, the implausible high utility values in the biomarker component of the model and the lack of justification for the **MR**QoL decrement for the health states of the liver disease component of the model. The ERG has incorporated various adjustments to the company base-case. The ERG base-case resulted in ICERs (probabilistic) of **MR**QoL decrement for the non-OCA regimens and using age-dependent utilities for the low and moderate risk health states. However, the ERG base-case ICER should be regarded as a lower bound as the exploratory analysis showed that alternative assumptions, for the TPs after 12 months and the non-transparent calibration method used by the company, resulted in substantially higher ICERs (ranging from

for UDCA inadequate responders and UDCA intolerant patients respectively).

As the methods used by the company to extrapolate treatment effectiveness are lacking transparency and justification, the ICERs (presented by both the company and the ERG) should be interpreted with caution. Given the lack of long-term results of the POISE trial, the ERG would generally agree with using a calibration approach (as adopted by the company), to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes. However, the assumptions, methods and data used should be validated and documented in detail to avoid 'black box' criticism and obtain credible results. This includes the assumption of no progression after 12 months when the ALP and bilirubin levels are within normal range due to treatment. Particularly, considering the impact of the usage of alternative assumptions, methods and data will potentially result in substantially higher ICERs as illustrated in the exploratory analysis by the ERG.

6. IMPACT ON THE ICER OF ADDITIONAL CLINICAL AND ECONOMIC ANALYSES UNDERTAKEN BY THE ERG

In Section 5.3 the ERG base-case was presented, which was based on various changes compared to the company base-case. Tables 6.1 and 6.2 show how each individual change impacts the ICER plus the combined effect of all changes simultaneously. The analyses numbers in Tables 6.1 and 6.2 correspond to the analyses numbers reported in Section 5.3. Also, the exploratory analyses are presented in Tables 6.3 and 6.4 (conditional on the ERG base-case). Appendix 2 contains technical details on the analyses performed by the ERG.

	(titratio	DCA on+UDCA	UDC	ĊA	Incremental results			
	QALY s	Costs	QALYs	Costs	ΔQAL Y	ΔCosts	ICER	
Company base- case (deterministic)	13.64		7.85	£96,97 7	5.79			
1. Fixing discrepancies between TPs ¹	13.68		7.85	£96,96 8	5.83			
2. Use transition probabilities from POISE for the non- OCA regimen	13.68		8.48	£90,74 0	5.20			
3. POISE trial proportion in starting health states	13.92		8.37	£92,83 4	5.55			
4. Use NHS reference costs for outpatient visits	13.68		7.85	£97,84 9	5.83			
5. Use health state costs consistent with TA330	13.68		7.85	£78,00 4	5.83			
6. Use UK age- dependent utility values	12.32		7.39	£96,96 8	4.93			
7. Remove utility decrement	13.72		8.11	£96,96 8	5.61			
ERG base-case (deterministic)	12.57		8.39	£72,33 2	4.17			
ERG base-case (probabilistic)	12.51		8.29	£72,26 6	4.22			
Footnotes: * The ERG v company: FINAL_ID7 211116 MM (ACIC); ¹	was not abl 85 Obetich These erro	e to reproduce nolic acid Cost rs were also co	these results b utility model	ased on the _list_price erratum su	e last versio v0.4_UDC ibmitted by	n of the mode A New patie the company	el sent by the ent data STC 3, 87	

Table 6.1: ERG base-case - UDCA inadequate responders

	OCA titration+UDCA		UDC	A	Incremental results		
	QALY s	Costs	QALYs	Costs	ΔQAL Y	ΔCosts	ICER
ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid							

Tuble 0.21 ERG bus										
	OCA	titration	No trea	tment	Inc	remental res	sults			
	QALY s	Costs	QALYs	Costs	ΔQAL Y	ΔCosts	ICER			
Company base- case (deterministic)	13.52		6.61	£103,23 3	6.91					
1. Fixing discrepancies between TPs ¹	13.56		6.61	£103,23 3	6.95					
2. Use transition probabilities from POISE for the non- OCA regimen	13.56		7.00	£99,799	6.56					
3. POISE trial proportion in starting health states	13.78		6.89	£100,94 2	6.89					
4. Use NHS reference costs for outpatient visits	13.56		6.61	£103,69 8	6.95					
5. Use health state costs consistent with TA330	13.56		6.61	£81,773	6.95					
6. Use UK age- dependent utility values	12.20		6.29	£103,23 3	5.92					
7. Remove utility decrement	13.61		6.91	£103,23 3	6.70					
ERG base-case (deterministic)	12.44		7.05	£78,461	5.38					
ERG base-case (probabilistic)	12.40		6.93	£78,541	5.47					
Footnotes: * The ERG	was not ab	le to reproduce	these results	based on the	last version	n of the model	sent by the			

Table 6.2: ERG base-case	- UDCA	intolerant	patients
--------------------------	--------	------------	----------

Footnotes: The ERG was not able to reproduce these results based on the last version of the model sent by the company: FINAL_ID785 Obeticholic acid Cost utility model_list_price v0.4_UDCA New patient data STC 211116 MM (ACIC); ¹ These errors were also corrected in the erratum submitted by the company⁸⁷ ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid

Table 6.3: ERG ex	ploratory analysis - Ul	DCA inadequate respo	nders

	OCA titration		No treat	ment	Incremental results		
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER
1. Use TPs and model structure from TA330	12.67		8.87	£38,094	3.80		
2. Using TPs based on the	12.79		10.20	£54,015	2.59		

	OCA	titration	No treat	ment	Inc	remental res	ults	
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER	
POISE trial after 12months for the non-OCA treatment arms								
3. Assume that TPs between biomarkers health states of the OCA arm are >0%	12.15		8.39	£72,332	3.75			
4. Use alternative costs for liver transplant	12.57		8.39	£71,083	4.17			
ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid								

	OCA	titration	No trea	tment	Inci	remental res	ults
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER
1. Use TPs and model structure from TA330	12.55		7.64	£34,009	4.91		
2. Using TPs based on the POISE trial after 12months for the non-OCA treatment arms	12.79		10.18	£45,639	2.61		
3. Assume that TPs between biomarkers health states of the OCA arm are >0%	12.02		7.05	£78,461	4.97		
4. Use alternative costs for liver transplant	12.44		7.05	£76,942	5.38		
ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid							

Table 6.4: ERG e	xploratory	v analysis -	UDCA	intolerant	natients
Table 0.4. LICO C	Apror ator y	anarysis	UDCI	monerant	patients

7. END OF LIFE

Not relevant.

8. OVERALL CONCLUSIONS

8.1 Statement of principal findings

Overall the ERG is satisfied that the relevant direct evidence comparing OCA and UDCA has been presented. However no trials comparing OCA to fibrates were identified.

One Phase 3 trial, POISE, with 217 patients was presented as the main source of evidence.²⁷ The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. However the group receiving OCA as monotherapy is underrepresented in this trial (11 patients). Additionally, the majority of patients in POISE appeared to be at an earlier stage of disease so the effects on those with more advanced disease are less clear.

The POISE trial was well-conducted. However it only examined surrogate outcomes. These were justified by the company based on previously published research. In POISE, OCA shows positive effects on surrogate endpoints, and there is some evidence that surrogate endpoints are related to relevant outcomes. However, the size of the relationship is unclear.

The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events was similar across groups. Events occurring more frequently in treatment groups were noted. The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

Two supporting Phase 2 trials were presented (including one of OCA monotherapy and one of OCA in combination with UDCA). These were not similar enough to be pooled with POISE but added support to the positive findings on surrogate outcomes. Clinical outcomes await the publication of the COBALT trial.³⁷

The main areas of ERG concern were the fact that the company did not use the POISE trial data directly for the non-OCA regimen, the lack of transparency and justification regarding methods and data to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes, the implausible high utility values in the biomarker component of the model and the lack of justification HRQoL decrement for the health states of the liver disease component of the model, except for HCC. The ERG has incorporated various adjustments to the company base-case. The ERG base-case resulted in ICERs (probabilistic) of for UDCA inadequate responders and UDCA intolerant patients respectively. The most influential adjustments/corrections made by the ERG were using TPs from the POISE trial for the non-OCA regimens and using age dependent utilities for the low and moderate risk health states. However, the ERG base-case ICER should be regarded as a lower bound as the exploratory analysis showed that alternative assumptions, for the TPs after 12 months and the non-transparent calibration method used by the company, resulted in substantially higher ICERs (ranging from and from

for UDCA inadequate responders and UDCA intolerant patients respectively).

As the methods used by the company to estimate and extrapolate treatment effectiveness are lacking transparency and justification, the ICERs (presented by both the company and the ERG) should be interpreted with caution. Given the lack of long-term results of the POISE trial, the ERG would generally agree with using a calibration approach (as adopted by the company), to extrapolate the

POISE trial data beyond the time horizon of 12 months and to final outcomes. However, the assumptions, methods and data used should be validated and documented in detail to avoid 'black box' criticism and obtain credible results. This includes the assumptions of no progression after 12 months. Particularly, considering the impact of the usage of alternative assumptions, methods and data will potentially result in substantially higher ICERs as illustrated in the exploratory analysis by the ERG.

8.2 Strengths and limitations of the assessment

The company used systematic review methods to identify the evidence on obeticholic acid for primary biliary cirrhosis. The majority of searches in the CS were well documented and easily reproducible. Searches were carried out on a broad range of resources. However no evidence was provided relating to fibrates, one of the comparators in the NICE scope. The company identified evidence comparing OCA to UDCA or placebo. The main evidence was based on a well-conducted international randomised controlled trial of 217 patients with a 12 month follow-up.

The main limitation of the evidence identified is that the main trial POISE is based on surrogate outcomes. The ERG is satisfied that the company has demonstrated some evidence of correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease is unclear. The role of OCA as monotherapy in patients intolerant to UDCA was investigated in a small number of patients so results may not be reliable. POISE included mainly patients at an early stage of disease progression so the role of OCA in more advanced disease is less clear.

The lack of transparency, justification and details regarding the methods used in the CS are the main weaknesses in the cost effectiveness chapter and the model submitted by the company. Transparency is a key quality aspect of modelling. The lack of transparency hampered the validity check by the ERG. Additionally, the inability of performing external validation, the relatively short trial follow-up using intermediate outcomes and the reliance on non PBC sources in the model stress the uncertainty in the estimation of long-term outcomes.

8.3 Suggested research priorities

The main research priority is to identify the role of OCA on clinical outcomes such as those highlighted in the NICE scope⁴ as current evidence relies on surrogate outcomes. The COBALT trial which investigates clinical outcomes is ongoing.³⁷ Duration is estimated to be approximately 8 years according to the time to accrue approximately 121 primary endpoint events. The start date of the trial was December 2014.³

There is also a need to compare OCA with other potential second line treatments including fibrates. There appear to be no ongoing trials comparing the two treatments. However we note the ongoing research into fibrates for PBC.³⁶
9. **REFERENCES**

[1] National Institute for Health and Care Excellence. *Obeticholic acid for treating primary biliary cirrhosis [ID785]. Appendix G - profession organisation submission: Professor Gideon Hirschfield, Professor of Autoimmune Liver Disease and Consultant Transplant Hepatologist, British Society of Gastroenterology/Royal College of Physicians/University of Birmingham*: NICE, 2016. 8p.

[2] National Institute for Health and Care Excellence. *Obeticholic acid for treating primary biliary cirrhosis [ID785]. Appendix G - profession organisation submission: Prof Graeme Alexander, BASL:* NICE, 2016. 4p.

[3] Intercept Pharmaceuticals Inc. *Biliary cirrhosis (primary) – obeticholic acid [ID:785]: Company evidence submission to National Institute of Health and Clinical Excellence. Single technology appraisal (STA)*, 26th October 2016 212p.

[4] National Institute for Health and Clinical Excellence. *Obeticholic acid for treating primary biliary cirrhosis: Final scope [Internet]*. London: NICE, August 2016 [accessed 1.9.16]. 3p. Available from: https://www.nice.org.uk/guidance/GID-TA10014/documents/final-scope

[5] Carey EJ, Ali AH, Lindor KD. Primary biliary cirrhosis. Lancet 2015;386(10003):1565-75.

[6] NIHR Horizon Scanning Centre. *Obeticholic acid for primary biliary cirrhosis - second line (NIHR HSC ID: 6478) [Internet]*. Birmingham: NIHR Horizon Scanning Centre, University of Birmingham, 2013 [accessed 3.10.16] Available from: <u>http://www.hsric.nihr.ac.uk/topics/obeticholic-acid-for-primary-biliary-cirrhosis-second-line/</u>

[7] Kurtovic J, Riordan SM, Williams R. The natural history of asymptomatic primary biliary cirrhosis. *QJM* 2005;98(5):331-6.

[8] Kumagi T, Heathcote EJ. Primary biliary cirrhosis. Orphanet J Rare Dis 2008;3:1.

[9] Kuiper EM, Hansen BE, de Vries RA, den Ouden-Muller JW, van Ditzhuijsen TJ, Haagsma EB, et al. Improved prognosis of patients with primary biliary cirrhosis that have a biochemical response to ursodeoxycholic acid. *Gastroenterology* 2009;136(4):1281-7.

[10] Carbone M, Mells GF, Pells G, Dawwas MF, Newton JL, Heneghan MA, et al. Sex and age are determinants of the clinical phenotype of primary biliary cirrhosis and response to ursodeoxycholic acid. *Gastroenterology* 2013;144(3):560-9.

[11] Lammers WJ, van Buuren HR, Hirschfield GM, Janssen HL, Invernizzi P, Mason AL, et al. Levels of alkaline phosphatase and bilirubin are surrogate end points of outcomes of patients with primary biliary cirrhosis: an international follow-up study. *Gastroenterology* 2014;147(6):1338-49.

[12] Mells GF, Pells G, Newton JL, Bathgate AJ, Burroughs AK, Heneghan MA, et al. Impact of primary biliary cirrhosis on perceived quality of life: the UK-PBC national study. *Hepatology* 2013;58(1):273-83.

[13] NHS Blood and Transplant. Annual report on liver transplantation: report for 2014/2015 [Internet]: NHS England, 2015 [accessed 22.9.16] Available from: http://www.odt.nhs.uk/pdf/organ_specific_report_liver_2015.pdf

[14] European Association for the Study of the Liver. EASL Clinical Practice Guidelines: management of cholestatic liver diseases. *J Hepatol* 2009;51(2):237-67.

[15] The British Society of Gastroenterology/UK-PBC. *Primary biliary cholangitis (formerly known as primary biliary cirrhosis) treatment and management guidelines [PDF provided with the company submission]* DRAFT September 2016.

[16] Charatcharoenwitthaya P, Pimentel S, Talwalkar JA, Enders FT, Lindor KD, Krom RA, et al. Long-term survival and impact of ursodeoxycholic acid treatment for recurrent primary biliary cirrhosis after liver transplantation. *Liver Transpl* 2007;13(9):1236-45.

[17] Summary of product characteristics (SPC): Fenofibrate 160 mg Film-coated Tablets [PDF provided with the company submission].

[18] Summary of product characteristics (SPC): Bezafibrate 200mg Tablets [PDF provided with the company submission].

[19] Itakura J, Izumi N, Nishimura Y, Inoue K, Ueda K, Nakanishi H, et al. Prospective randomized crossover trial of combination therapy with bezafibrate and UDCA for primary biliary cirrhosis. *Hepatol Res* 2004;29(4):216-22.

[20] Iwasaki S, Ohira H, Nishiguchi S, Zeniya M, Kaneko S, Onji M, et al. The efficacy of ursodeoxycholic acid and bezafibrate combination therapy for primary biliary cirrhosis: a prospective, multicenter study. *Hepatol Res* 2008;38(6):557-64.

[21] Nakai S, Masaki T, Kurokohchi K, Deguchi A, Nishioka M. Combination therapy of bezafibrate and ursodeoxycholic acid in primary biliary cirrhosis: a preliminary study. *Am J Gastroenterol* 2000;95(1):326-7.

[22] Hosonuma K, Sato K, Yamazaki Y, Yanagisawa M, Hashizume H, Horiguchi N, et al. A prospective randomized controlled study of long-term combination therapy using ursodeoxycholic acid and bezafibrate in patients with primary biliary cirrhosis and dyslipidemia. *Am J Gastroenterol* 2015;110(3):423-31.

[23] Cheung AC, Lapointe-Shaw L, Kowgier M, Meza-Cardona J, Hirschfield GM, Janssen HL, et al. Combined ursodeoxycholic acid (UDCA) and fenofibrate in primary biliary cholangitis patients with incomplete UDCA response may improve outcomes. *Aliment Pharmacol Ther* 2016;43(2):283-93.

[24] Momah N, Silveira MG, Jorgensen R, Sinakos E, Lindor KD. Optimizing biochemical markers as endpoints for clinical trials in primary biliary cirrhosis. *Liver Int* 2012;32(5):790-5.

[25] Kumagi T, Guindi M, Fischer SE, Arenovich T, Abdalian R, Coltescu C, et al. Baseline ductopenia and treatment response predict long-term histological progression in primary biliary cirrhosis. *Am J Gastroenterol* 2010;105(10):2186-94.

[26] Sogolow ED, Lasker JN, Short LM. Fatigue as a major predictor of quality of life in women with autoimmune liver disease: the case of primary biliary cirrhosis. *Womens Health Issues* 2008;18(4):336-42.

[27] Nevens F, Andreone P, Mazzella G, Strasser SI, Bowlus C, Invernizzi P, et al. A Placebo-controlled trial of obeticholic acid in primary biliary cholangitis. *N Engl J Med* 2016;375(7):631-643.

[28] European Medicines Agency. *CHMP summary of opinion: Ocaliva (obeticholic acid). EMA/CHMP/651446/2016 [Internet]*. London: European Medicines Agency, 2016 [accessed 15.11.16] Available from: <u>http://www.ema.europa.eu/docs/en_GB/document_library/Summary_of_opinion_-</u>Initial_authorisation/human/004093/WC500214177.pdf

[29] Kowdley K. The first new monotherapy therapeutic PBC study in a decade? An international study evaluating the farnesoid X receptor agonist obeticholic acid in PBC. *Hepatology* 2011;54(Suppl 1):416A-7A.

[30] National Institute for Health and Care Excellence. *Obeticholic acid for primary biliary cirrhosis [ID785]: Clarification letter*. London: NICE, 2016. 10p.

[31] Rudic JS, Poropat G, Krstic MN, Bjelakovic G, Gluud C. Bezafibrate for primary biliary cirrhosis. *Cochrane Database Syst Rev* 2012, Issue 1. Art. No.: CD009145. DOI: 10.1002/14651858.CD009145.pub2

[32] Intercept Pharmaceuticals Inc. *Obeticholic acid for primary biliary cirrhosis [ID:785]: response to request for clarification from the ERG*: Intercept Pharmaceuticals Inc., 2016. 62p.

[33] Grigorian AY, Mardini HE, Corpechot C, Poupon R, Levy C. Fenofibrate is effective adjunctive therapy in the treatment of primary biliary cirrhosis: a meta-analysis. *Clinics and Research in Hepatology and Gastroenterology* 2015;39(3):296-306.

[34] Zhang Y, Chen K, Dai W, Xia Y, Wang F, Shen M, et al. Combination therapy of bezafibrate and ursodeoxycholic acid for primary biliary cirrhosis: a meta-analysis. *Hepatol Res* 2015;45(1):48-58.

[35] Zhang Y, Li S, He L, Wang F, Chen K, Li J, et al. Combination therapy of fenofibrate and ursodeoxycholic acid in patients with primary biliary cirrhosis who respond incompletely to UDCA monotherapy: a meta-analysis. *Drug Design, Development and Therapy* 2015;9:2757-66.

[36] Assistance Publique - Hopitaux de Paris. Phase 3 study of bezafibrate in combination with ursodeoxycholic acid in primary biliary cirrhosis (BEZURSO). NCT01654731 In: ClinicalTrials.gov [Internet]. Bethesda (MD): National Library of Medicine (US). 2016 [accessed 9.11.16]. Available from: https://clinicaltrials.gov/ct2/show/NCT01654731?term=NCT01654731

[37] Intercept Pharmaceuticals Inc. Phase 3 study of obeticholic acid evaluating clinical outcomes in patients with primary biliary cirrhosis (COBALT). NCT02308111. In: ClinicalTrails.gov [Internet]. Bethesda (MD): National Library of Medicine (US). 2016 [accessed 15.11.16]. Available from: https://clinicaltrials.gov/ct2/show/NCT02308111

[38] Canadian Agency for Drugs and Technologies in Health. *CADTH peer review checklist for search strategies [Internet]*. Ottawa: CADTH, 2013 [accessed 17.7.13]. 3p. Available from: http://www.cadth.ca/en/resources/finding-evidence-is

[39] National Institute for Health and Care Excellence. *Single Technology Appraisal: specification for manufacturer/sponsor submission of evidence [Internet]*. London: NICE, 2012 [accessed 17.11.16]. 76p. Available from: <u>http://www.nice.org.uk/Media/Default/About/what-we-do/NICE-guidance/NICE-technology-appraisals/Specification-for-manufacturer-sponsor-submission-of-evidence-June-2012.doc</u>

[40] Centre for Reviews and Dissemination. *Systematic Reviews: CRD's guidance for undertaking reviews in health care [Internet]*. York: University of York, 2009 [accessed 23.3.11] Available from: http://www.york.ac.uk/inst/crd/SysRev/!SSL!/WebHelp/SysRev3.htm

[41] Intercept Pharmaceuticals Inc. *Biliary cirrhosis (primary) – obeticholic acid [ID:785]: Appendices to accompany company evidence submission. Single technology appraisal (STA)*, Sept 2016 95p.

[42] Singh J, Longworth L. Estimating the cost of liver transplantation in patients diagnosed with chronic hepatitis C and B in the UK [Abstr PGI33]. Paper presented at ISPOR 17th Annual European Congress; 8-12 Nov 2014; Amsterdam: The Netherlands *Value Health* 2014;17(7):A368.

[43] Adam R, Karam V, Delvart V, O'Grady J, Mirza D, Klempnauer J, et al. Evolution of indications and results of liver transplantation in Europe: a report from the European Liver Transplant Registry (ELTR). *J Hepatol* 2012;57(3):675-88.

[44] Hirschfield GM, Mason A, Luketic V, Lindor K, Gordon SC, Mayo M, et al. Efficacy of obeticholic acid in patients with primary biliary cirrhosis and inadequate response to ursodeoxycholic acid. *Gastroenterology* 2015;148(4):751-61 e8.

[45] Nevens F, Andreone P, Mazzella G, Strasser SI, Bowlus C, Invernizzi P, et al. A placebo-controlled trial of obeticholic acid in primary biliary cholangitis. *N Engl J Med* 2016;375(7):631-43.

[46] Intercept Pharmaceuticals Inc. Clinical study report: 747-301. Obeticholic acid (OCA). A Phase 3, double blind, placebo controlled trial and long term safety extension of obeticholic acid in patients with primary biliary cirrhosis. Data on File. 2015 [PDF provided with the company submission].

[47] Jacoby A, Rannard A, Buck D, Bhala N, Newton JL, James OF, et al. Development, validation, and evaluation of the PBC-40, a disease specific health related quality of life measure for primary biliary cirrhosis. *Gut* 2005;54(11):1622-9.

[48] National Institute for Health and Care Excellence. *Guide to the methods of technology appraisal 2013 [Internet]*. London: NICE, 2013 [accessed 17.11.16]. 93p. Available from: http://publications.nice.org.uk/pmg9

[49] Scottish Intercollegiate Guidelines Network (SIGN). Search Filters: Economic Studies [Internet]. Edinburgh: Healthcare Improvement Scotland, 2015 [accessed 3.11.16]. Available from: http://www.sign.ac.uk/methodology/filters.html#econ

[50] Longworth L, Young T, Buxton MJ, Ratcliffe J, Neuberger J, Burroughs A, et al. Midterm costeffectiveness of the liver transplantation program of England and Wales for three disease groups. *Liver Transpl* 2003;9(12):1295-307.

[51] Boberg KM, Wisloff T, Kjollesdal KS, Stovring H, Kristiansen IS. Cost and health consequences of treatment of primary biliary cirrhosis with ursodeoxycholic acid. *Aliment Pharmacol Ther* 2013;38(7):794-803.

[52] Ratcliffe J, Young T, Buxton M, Eldabi T, Paul R, Burroughs A, et al. A simulation modelling approach to evaluating alternative policies for the management of the waiting list for liver transplantation. *Health Care Manag Sci* 2001;4(2):117-24.

[53] Pasha T, Heathcote J, Gabriel S, Cauch-Dudek K, Jorgensen R, Therneau T, et al. Costeffectiveness of ursodeoxycholic acid therapy in primary biliary cirrhosis. *Hepatology* 1999;29(1):21-6.

[54] Intercept Pharmaceuticals Inc. *Obeticholic acid for primary biliary cirrhosis [ID:785]: Clarification responses follow up questions v0.1 MM 171116 [ACIC]:* Intercept Pharmaceuticals Inc., 2016. 6p.

[55] Intercept Pharmaceuticals Inc. *ID785 Obeticholic acid: Cost utility model_list_price v0.4_UDCA New patient data STC 211116 MM [ACIC] [Excel spreadsheet provided with the company submission]:* Intercept Pharmaceuticals Inc., 2016 [accessed 22.11.16]

[56] Riemsma R, Joore M, Ramaekers B, Leadley R, Pouwels X, Van Giessen A, et al. Sofosbuvir in combination with velpatasvir for treating chronic hepatitis C: a single technology assessment. York: Kleijnen Systematic Reviews Ltd. In: *National Institute for Health and Care Excellence, Sofosbuvir-velpatasvir for treating chronic hepatitis C [ID921]: Single technology appraisal (STA), Committee Papers*, 2016 [accessed 16.11.16]. Available from: https://www.nice.org.uk/guidance/GID-TA10064/documents/committee-papers

[57] Office for National Statistics. *National Life Tables, United Kingdom: 2012–2014 [Internet]*, 2015 [accessed 30.11.16] Available from: http://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/lifeexpectancies/bu lletins/nationallifetablesunitedkingdom/2015-09-23

[58] Harms M.H, Lammers WJ, Janssen HLA, Corpechot C, Invernizzi P, Mayo MJ, et al. Incidence and impact of decompensating events in primary biliary cirrhosis: results of an international follow up study of 3030 patients. Presented at the 66th Annual Meeting of the American Association for the Study of Liver Diseases (AASLD):13-17 Nov 2015: San Francisco, CA, USA. [PDF provided with the company submission]. 2015.

[59] Harms MHL, Lammers WJ, Marmon T, Pencek R, van Buuren HR, Hansens BE, et al. Improvement in estimated liver transplant free survival after 1 year of obeticholic acid treatment [poster 352]. Presented at The International Liver Congress; 13-17 April 2016; Barcelona: Spain. *J Hepatol* 2016;64:S634-5.

[60] Corpechot C, Carrat F, Bonnand AM, Poupon RE, Poupon R. The effect of ursodeoxycholic acid therapy on liver fibrosis progression in primary biliary cirrhosis. *Hepatology* 2000;32(6):1196-9.

[61] Lammers WJ, Hirschfield GM, Corpechot C, Nevens F, Lindor KD, Janssen HLA, et al. Development and Validation of a Scoring System to Predict Outcomes of Patients with Primary Biliary Cirrhosis Receiving Ursodeoxycholic Acid Therapy. *Gastroenterology* 2015;149(7):1804-1812e4.

[62] National Institute for Health and Care Excellence. *Sofosbuvir for treating chronic hepatitis C: NICE technology appraisal guidance TA330 [Internet]*. London: NICE, 2015 [updated 2015; accessed 12.9.16] Available from: https://www.nice.org.uk/guidance/ta330

[63] Trivedi PJ, Lammers WJ, van Buuren HR, Pares A, Floreani A, Janssen HL, et al. Stratification of hepatocellular carcinoma risk in primary biliary cirrhosis: a multicentre international study. *Gut* 2016;65(2):321-9.

[64] Wright M, Grieve R, Roberts J, Main J, Thomas HC. Health benefits of antiviral therapy for mild chronic hepatitis C: randomised controlled trial and economic evaluation. *Health Technol Assess* 2006;10(21):1-113, iii.

[65] Kim WR, Lake JR, Smith JM, Skeans MA, Schladt DP, Edwards EB, et al. OPTN/SRTR 2013 Annual Data Report: liver. *Am J Transplant* 2015;15 (Suppl 2):1-28.

[66] Lindor KD, Gershwin ME, Poupon R, Kaplan M, Bergasa NV, Heathcote EJ. Primary biliary cirrhosis. *Hepatology* 2009;50(1):291-308.

[67] Neuberger J. Liver transplantation for primary biliary cirrhosis: indications and risk of recurrence. *J Hepatol* 2003;39(2):142-8.

[68] Cardoso AC, Moucari R, Figueiredo-Mendes C, Ripault MP, Giuily N, Castelnau C, et al. Impact of peginterferon and ribavirin therapy on hepatocellular carcinoma: incidence and survival in hepatitis C patients with advanced fibrosis. *J Hepatol* 2010;52(5):652-7.

[69] Siebert U, Sroczynski G, Wasem J, Greiner W, Ravens-Sieberer U, Aidelsburger P, et al. Using competence network collaboration and decision-analytic modeling to assess the cost-effectiveness of interferon alpha-2b plus ribavirin as initial treatment of chronic hepatitis C in Germany. *Eur J Health Econ* 2005;6(2):112-23.

[70] Fattovich G, Giustina G, Degos F, Tremolada F, Diodati G, Almasio P, et al. Morbidity and mortality in compensated cirrhosis type C: a retrospective follow-up study of 384 patients. *Gastroenterology* 1997;112(2):463-72.

[71] Shepherd J, Jones J, Hartwell D, Davidson P, Price A, Waugh N. Interferon alpha (pegylated and non-pegylated) and ribavirin for the treatment of mild chronic hepatitis C: a systematic review and economic evaluation. *Health Technol Assess* 2007;11(11):1-205, iii.

[72] Aberg F, Makisalo H, Hockerstedt K, Isoniemi H. Infectious complications more than 1 year after liver transplantation: a 3-decade nationwide experience. *Am J Transplant* 2011;11(2):287-95.

[73] Bondini S, Kallman J, Dan A, Younoszai Z, Ramsey L, Nader F, et al. Health-related quality of life in patients with chronic hepatitis B. *Liver Int* 2007;27(8):1119-25.

[74] Younossi ZM, Boparai N, McCormick M, Price LL, Guyatt G. Assessment of utilities and health-related quality of life in patients with chronic liver disease. *Am J Gastroenterol* 2001;96(2):579-83.

[75] Younossi ZM, Kiwi ML, Boparai N, Price LL, Guyatt G. Cholestatic liver diseases and health-related quality of life. *Am J Gastroenterol* 2000;95(2):497-502.

[76] Torrance GW, Feeny DH, Furlong WJ, Barr RD, Zhang Y, Wang Q. Multiattribute utility function for a comprehensive health status classification system. Health Utilities Index Mark 2. *Med Care* 1996;34(7):702-22.

[77] Dyson JK, Wilkinson N, Jopson L, Mells G, Bathgate A, Heneghan MA, et al. The interrelationship of symptom severity and quality of life in 2055 patients with primary biliary cholangitis. *Aliment Pharmacol Ther* 2016;44(10):1039-50.

[78] Sullivan PW, Slejko JF, Sculpher MJ, Ghushchyan V. Catalogue of EQ-5D scores for the United Kingdom. *Med Decis Making* 2011;31(6):800-4.

[79] Joint Formulary Committee. *British National Formulary*. 72. London: BMJ Group and Pharmaceutical Press, 2016.

[80] Personal Social Services Research Unit. *Unit Costs of Health and Social Care 2015 [Internet]*. Canterbury: University of Kent, 2016 [accessed 12.9.16] Available from: http://www.pssru.ac.uk/project-pages/unit-costs/2015/.

[81] Department of Health. *NHS reference costs 2014-2015 [Internet]*. London: Department of Health, 2015 [accessed 30.11.16] Available from: https://www.gov.uk/government/publications/nhs-referencecosts-2014-to-2015 [82] Habka D, Mann D, Landes R, Soto-Gutierrez A. Future economics of liver transplantation: a 20year cost modeling forecast and the prospect of bioengineering autologous liver grafts. *PLoS One* 2015;10(7):e0131764.

[83] Briggs AH, Weinstein MC, Fenwick EA, Karnon J, Sculpher MJ, Paltiel AD. Model parameter estimation and uncertainty: a report of the ISPOR-SMDM Modeling Good Research Practices Task Force-6. *Value Health* 2012;15(6):835-42.

[84] UK-PBC. Risk Score Calculator [Internet]. Cambridge: University of Cambridge, 2015 [accessed 14.11.16]. Available from: <u>http://www.uk-pbc.com/resources/tools/riskcalculator/</u>

[85] Caro JJ, Briggs AH, Siebert U, Kuntz KM. Modeling good research practices - overview: a report of the ISPOR-SMDM Modeling Good Research Practices Task Force-1. *Med Decis Making* 2012;32(5):667-77.

[86] Kaltenthaler E, Carroll C, Hill-McManus D, Scope A, Holmes M, Rice S, et al. The use of exploratory analyses within the National Institute for Health and Care Excellence single technology appraisal process: an evaluation and qualitative analysis. *Health Technol Assess* 2016;20(26):1-48.

[87] Intercept Pharmaceuticals Inc. *Obeticholic acid for primary biliary cirrhosis [ID:785]: Erratum to Intercept's responses to ERG questions*, 21st November 2016 29p.

APPENDIX 1: DETAILS OF ERG ANALYSES (FOR VALIDATION PURPOSES)

Altered cells are printed in *italics*.

Fixing errors

 Fixing discrepancy between TPs reported in the CS and used in the economic model (for biomarker component of the model; see Section 5.2.6 for more details). To fix this error, the ERG used the TPs reported in the CS. *Transition matrices Y121,134,170,171,183,195,219,232,268,269,281,293,*

Z146,183,195,244,281,293

Fixing violations

- Use TPs from POISE for the non-OCA regimen (for biomarker component of the model; see Section 5.2.6 for more details). *Clinical inputs C10*
- 3. Use proportions in the starting health states (for biomarker component) as derived from the POISE trial (see Section 5.2.3 for more details). *Model settings C25,27, Defaults D46,*
- 4. Use NHS reference costs for outpatient visits (see Section 5.2.9 for more details). *F46Disease management costs D6:D7, D17, Defaults D286:D287*
- Use health state costs of £1,561 for CC (instead of £6,254) consistent with TA330 (for liver disease component of the model; see Section 5.2.9 for more details) Disease management costs D9, Defaults K289
- 6. Use age-dependent utilities (from the UK general population) for the low and moderate risk health states (for biomarker component of the model; see Section 5.2.8 for more details). *Quality of life inputs D6:D7, Model parameters II15:116, Defaults D346:D347, UDCA Markov model DN24:206, OCA+UDCA Titration Markov model DN24:206*

Matters of judgment

Remove the HRQoL decrement (for liver disease component of the model; see Section 5.2.9 for more details).

Quality of life inputs D13, D20:D23, Defaults D353, D356:D358

A2 APPENDIX 2: SCATTER PLOTS AND COST EFFECTIVENESS ACCEPTABILITY CURVES OF OCA (WITH OR WITHOUT UDCA) VERSUS COMPARATORS IN BOTH POPULATIONS.

Figure A2.1: Scatter plot for PSA (UDCA-intolerant population), using the OCA list price



Figure A2.1: Cost effectiveness acceptability curve for OCA titration versus no treatment (UDCA-intolerant population), using the OCA list price



Source: Figure 32 of the CS³ OCA, obeticholic acid; UDCA, ursodeoxycholic acid

Figure A2.3: Scatter plot for PSA (UDCA inadequate responders population), using the OCA list price



Source: Figure 32 of the CS³ OCA, obeticholic acid; UDCA, ursodeoxycholic acid

Figure A2.2: Cost effectiveness acceptability curve for OCA + UDCA titration versus UDCA (UDCA inadequate responders population), using the OCA list price



Source: Figure 34 of the CS³ OCA, obeticholic acid; UDCA, ursodeoxycholic acid

APPENDIX 3: TORNADO DIAGRAMS OF THE DSA FOR BOTH POPULATIONS.

Figure A3.1: Tornado diagram for OCA titration versus no treatment in the UDCA intolerant population, using the list price of OCA



Figure A3.1: Tornado diagram for OCA + UDCA titration versus UDCA in the UDCA inadequate responder population, using the list price of OCA



Source: Figure 36 of the CS³



in collaboration with:



Obeticholic acid for treating primary biliary cirrhosis Confidential Appendix

Produced by	Kleijnen Systematic Reviews (KSR) Ltd. in collaboration with Erasmus University Rotterdam (EUR) and Maastricht University							
Authors	Debra Fayter, Systematic Reviewer, KSR							
	Bram Ramaekers, Health Economist, Maastricht UMC+							
	Sabine Grimm, Health Economist, Maastricht UMC+							
	Xavier Pouwels, Health Economist, Maastricht UMC+							
	Nigel Armstrong, Health Economics Manager, KSR							
	Steve Ryder, Health Economist, KSR							
	Sonia Garcia, Systematic Reviewer, KSR							
	Piet Portegijs, Systematic Reviewer, KSR							
	Caro Noake, Information Specialist, KSR							
	Rob Riemsma, Reviews Manager, KSR							
	Manuela Joore, Health Economist, Professor of Health Technology Assessment & Decision Making, Maastricht UMC+							
	Jos Kleijnen, Director, KSR; Professor of Systematic Reviews in Health Care, Maastricht University							

Correspondence to	Debra Fayter, Kleijnen Systematic Reviews
	Unit 6, Escrick Business Park
	Riccall Road, Escrick
	York, UK
	YO19 6FD
Date completed	08/12/2016
Source of funding	This report was commissioned by the NIHR HTA Programme as project number STA 15/69/09.

In this confidential appendix, the patient access scheme (PAS) price of Obeticholic acid (OCA) has been implemented in all company's and ERG's analyses. The PAS price has been communicated by the company through the patient access scheme submission template on the 1st of December 2016.¹ The PAS price is implemented as a simple price discount of **Sector** $\frac{6}{2}$ on the list price, which lowers the pack price of OCA from £2,384.04 to **Sector** (annual prices are respectively £29,005 and **Sector**).

In order to conduct these analyses, the ERG adjusted the following cells (for the ERG base-case analyses, these adjustments were added to the ERG base-case adjustments described in Appendix 4 of the ERG report):

'Treatment cost'!D8

'Default'!C230

These analyses are performed for the company base-case and the ERG base-case analyses. Results provided in the current confidential appendix concern OCA Titration (PAS price) versus UDCA in the UDCA inadequate responder population and OCA Titration (PAS price) versus 'No treatment' in the UDCA intolerant population.

Company's cost effectiveness results

In both populations, OCA (with or without UDCA) led to longer survival, a QALY gain, and higher costs than the comparators (i.e. no treatment for the UDCA-intolerant population and UDCA alone for the UDCA inadequate responders). The company provided disaggregated results for QALYs (Tables 5.14 and 5.15 of the ERG report)² and costs by health state (Tables 1 and 2). The low risk and moderate risk health states contribute to 74% and 73% of the absolute incremental QALY gained for the UDCA-intolerant population and UDCA inadequate responders. The disaggregated results are not reproduced in this confidential appendix since the PAS price does not influence the QALY results.

In the company base-case analysis, OCA (PAS price) versus UDCA resulted in an incremental cost effectiveness ratio (ICER) of £28,425 per additional QALY gained (deterministic results) in the UDCA inadequate responder population. OCA (PAS price) versus 'No treatment' was associated with an ICER of £21,438 per additional QALY gained (deterministic results) (Tables 3 and 4).

Tables 5 and 6 present the results of the first scenario analysis performed by the company for the UDCA intolerant and UDCA inadequate populations respectively (Scenario 1). In this analysis, the PAS price of OCA is used and the **second** sensitivity analysis performed by the company, using the PAS price of OCA (Scenario 2). In this analysis, alternative transition probabilities are used from the pre-liver transplant to the liver transplant health states and from the pre-liver transplant to death health states. Results of these analyses were provided in the CS sent to the ERG on the 4th of October 2016.³

ERG's cost effectiveness results

The ERG base-case analyses results in an ICER of £44,945 per additional QALY gained for OCA versus UDCA in the UDCA inadequate responder patient population (probabilistic results). OCA versus 'No treatment' in the UDCA intolerant patient population was associated with an ICER of £31,682 per QALY gained (probabilistic results). In the UDCA inadequate responder population, OCA has a 0.0% and 0.1% probability of being cost effective at a willingness to pay of £20,000 and £30,000 per QALY, respectively. In the UDCA intolerant population, OCA has a 0% and 16.2% probability of being cost effective at a willingness to pay of £20,000 and £30,000 per QALY, respectively. Scatter plots and cost

effectiveness acceptability curves for the ERG base-case are presented in Figures 1 and 2 for the UDCA inadequate responders and in Figures 3 and 4 for the UDCA intolerant population.

Tables 9 and 10 provide the cost effectiveness results of the exploratory analyses performed by the ERG. These exploratory analyses all resulted in increased ICERs. The ERG base-case ICER should be regarded as a lower bound and it should be noted that the usage of alternative assumptions, methods and data will potentially result in substantially higher ICERs.

Health state	OCA Titration	No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a					37%
Moderate risk ^b					40%
Severe risk ^c					6%
Discontinuation					0%
Decompensated cirrhosis					5%
НСС					0%
Pre transplant (end stage)					2%
Liver transplant					3%
Post liver transplant					6%
Re-emergence of PBC					0%
Total					100%

Table 1: Summary of costs by health state for UDCA inadequate responders – company's basecase

Source: Table 76 of the CS provided by the company on the 4th of October 2016, in which the PAS price was already applied to OCA.³

Footnotes: a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS⁴; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; * Calculated by the ERG ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of

normal

Health state	OCA + UDCA titration	UDCA + No treatment	Increment	Absolute increment	Absolute increment (in %)*
Low risk ^a					35%
Moderate risk ^b					39%
Severe risk ^c					6%
Discontinuation					0%
Decompensated cirrhosis					6%
НСС					0%
Pre transplant (end stage)					3%
Liver transplant					4%
Post liver transplant					7%
Re-emergence of PBC					0%
Total					100%

 Table 2: Summary of costs by health state for UDCA-intolerant population – company's base-case

Source: Table 75 of the CS provided by the company on the 4th of October 2016, in which the PAS price was already applied to OCA.³

Footnotes: ^a referred as 'ALP: ≤ 200 u/L and Bili: Normal' in the CS⁴; ^b referred as 'ALP: > 200 u/L and Bili: Normal' in the CS³; ^c referred as 'Bili: Abnormal and rising, or CC' in the CS³; ^{*} Calculated by the ERG

ALP, alkaline phosphatase; Bili, bilirubin; CC, compensated cirrhosis; CS, company submission; ERG, Evidence Review Group; HCC, hepatocellular carcinoma; OCA, obeticholic acid; PBC, primary biliary cholangitis/cirrhosis; QALY, quality-adjusted life year; UDCA, ursodeoxycholic acid; ULN, upper limit of normal

	OCA titrat	tion+UDCA	UD	OCA	Incremental results			
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER	
Company base- case (deterministic)*	13.64	£261,527	7.85	£96,977	5.79	£164,551	£28,425	
1. Fixing discrepancies between TPs ¹	13.68	£261,774	7.85	£96,968	5.83	£164,806	£28,280	
2. Use transition probabilities from POISE for the non- OCA regimen	13.68	£261,776	8.48	£90,740	5.20	£171,036	£32,897	
3. POISE trial proportion in starting health states	13.92	£263,317	8.37	£92,834	5.55	£170,482	£30,736	
4. Use NHS reference costs for outpatient visits	13.68	£263,301	7.85	£97,849	5.83	£165,453	£28,394	
5. Use health state costs consistent with TA330	13.68	£258,741	7.85	£78,004	5.83	£180,737	£31,017	
6. Use UK age- dependent utility values	12.32	£261,776	7.39	£96,968	4.93	£164,808	£33,458	
7. Remove utility decrement	13.72	£261,776	8.11	£96,968	5.61	£164,808	£29,377	
ERG base-case (deterministic)	12.57	£262,300	8.39	£72,332	4.17	£189,968	£45,541	
ERG base-case (probabilistic)	12.53	£261,929	8.31	£72,223	4.22	£189,706	£44,945	
Footnotes: * The ERG did n	not reproduce	these results. 7	These are ba	ased on the	first versior	of the $C\overline{S}$ s	ent by the	

Table 3: ERG base-case - UDCA inadequate responders - OCA PAS price

company on the 4th of October 2016³; ¹ These errors were also corrected in the erratum submitted by the company⁵

ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid



Figure 1: Scatterplot of the ERG base-case - UDCA inadequate responders - OCA PAS price





	OCA titra	ition	No tre	atment	Incremental results		
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER
Company base- case (deterministic)*	13.52	£251,443	6.61	£103,233	6.91	£148,210	£21,438
1. Fixing discrepancies between TPs ¹	13.56	£251,671	6.61	£103,233	6.95	£148,438	£21,351
2. Use transition probabilities from POISE for the non- OCA regimen	13.56	£251,674	7.00	£99,799	6.56	£151,875	£23,152
3. POISE trial proportion in starting health states	13.78	£253,217	6.89	£100,942	6.89	£152,275	£22,111
4. Use NHS reference costs for outpatient visits	13.56	£253,159	6.61	£103,698	6.95	£149,461	£21,500
5. Use health state costs consistent with TA330	13.56	£248,395	6.61	£81,773	6.95	£166,622	£23,969
6. Use UK age- dependent utility values	12.20	£251,674	6.29	£103,233	5.92	£148,441	£25,085
7. Remove utility decrement	13.61	£251,674	6.91	£103,233	6.70	£148,441	£22,162
ERG base-case (deterministic)	12.44	£251,860	7.05	£78,461	5.38	£173,399	£32,217
ERG base-case (probabilistic)	12.39	£251,478	6.92	£78,477	5.46	£173,001	£31,682
Footnotes: * The ERG di company on the 4 th of company ⁵	d not reproduce the October 2016 ³ ; ¹	ese results. T These errors	These are b were also	ased on the corrected	first versior in the errat	tum submitt	ent by the ed by the

Table 4: ERG base-case - UDCA intolerant patients - OCA PAS price

ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid



Figure 3: Scatterplot of the ERG base-case - UDCA intolerant population - OCA PAS price





Sensitivity / scenario analyses

Technologies	Total			Inc	rement	ICER (£)			
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA versus No treatment		
No treatment	£103,23 3	11.30	6.91	-	-	-	-		
OCA titration	£251,44 3	16.65	13.57	£148,210	5.35	6.66	£22,250		
Source: Table 84	Source: Table 84 of the CS submitted by the company on the 4 th of October 2016. ³								

Table 5: Company's scenario analysis 1 results: UDCA intolerant patients

LYG: Life year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost effectiveness ratio; OCA: Obeticholic acid.

Table 6: Company's scenario analysis 1 results: UDCA inadequate responders

Technologies	Total			Inc	rement	ICER (£)	
	Costs (£)	LYG	QALY s	Costs (£)	LY G	QALYs	OCA + UDCA versus UDCA
UDCA	£96,977	12.35	8.11	-	-	-	-
OCA + UDCA titration	£261,52 7	16.75	13.69	£164,551	4.40	5.57	£29,524

Source: Table 85 of the CS submitted by the company on the 4th of October 2016.²

LYG: Life year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost effectiveness ratio; OCA: Obeticholic acid; UDCA: ursodeoxycholic acid

Table 7: Company's scenario analysis 2 results: UDCA intolerant patients

Technologies	Total			Inc	rement	ICER (£)		
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA versus No treatment	
No treatment	£95,697	10.93	6.43	-	-	-	-	
OCA titration	£250,197	16.59	13.49	£154,499	5.66	7.06	£21,874	
Source: Table 87 of the CS submitted by the company on the 4 th of October 2016. ³								
LYG: Life year ga Obeticholic acid.	LYG: Life year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost effectiveness ratio; OCA: Obeticholic acid							

Technologies	Total			Inc	rement	ICER (£)	
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA+ UDCA versus UDCA
UDCA	£90,516	12.04	7.70	-	-	-	-
OCA + UDCA titration	£260,386	16.69	13.61	£169,870	4.66	5.91	£28,724
Source: Table 88 of the CS submitted by the company on the 4 th of October 2016. ³							
LYG: Life year gat Obeticholic acid; U	ined; QALYs: JDCA: ursode	Quality-a oxycholic	djusted life y acid	years; ICER: I	ncremen	tal cost effec	ctiveness ratio; OCA:

Table 8: Company's scenario analysis 2 results: UDCA inadequate responders

Table 9: ERG exploratory analysis - UDCA inadequate responders – OCA PAS price

	OCA titra	tion	No tre	atment	Incremental results				
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER		
1. Use TPs and model structure from TA330	12.67	£259,927	8.87	£38,094	3.80	£221,832	£58,412		
2. Using TPs based on the POISE trial after 12months for the non-OCA treatment arms	12.79	£260,197	10.20	£54,015	2.59	£206,182	£79,668		
3. Assume that TPs between biomarkers health states of the OCA arm are >0%	12.15	£257,410	8.39	£72,332	3.75	£185,078	£49,294		
4. Use alternative costs for liver transplant	12.57	£262,108	8.39	£71,083	4.17	£191,025	£45,794		
ERG, Evidence Review OCA, obeticholic acid ursodeoxycholic acid	ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid								

	OCA titra	tion	No trea	atment	Incremental results			
	QALYs	Costs	QALYs	Costs	ΔQALY	ΔCosts	ICER	
1. Use TPs and model structure from TA330	12.55	£248,426	7.64	£34,009	4.91	£214,417	£43,686	
2. Using TPs based on the POISE trial after 12months for the non-OCA treatment arms	12.79	£248,486	10.18	£45,639	2.61	£202,848	£77,715	
3. Assume that TPs between biomarkers health states of the OCA arm are >0%	12.02	£247,440	7.05	£78,461	4.97	£168,979	£34,031	
4. Use alternative costs for liver transplant	12.44	£251,645	7.05	£76,942	5.38	£174,703	£32,459	
ERG, Evidence Review OCA, obeticholic acid ursodeoxycholic acid	ERG, Evidence Review group; ICER, incremental cost-effectiveness ratio; NHS, National Health Services; OCA, obeticholic acid; QALYs, quality-adjusted life years; TPs, Transition probabilities; UDCA, ursodeoxycholic acid							

Table 10: ERG exploratory analysis - UDCA intolerant patients - OCA PAS price

References

[1] Intercept Pharmaceuticals Inc. Patient access scheme submission template (Biliary cirrhosis (primary) – obeticholic acid [ID:785]): Patient Access Scheme (PAS) appendix) [document provided by the company], October 2009 [accessed 7.12.16]. 30p.

[2] Fayter D, Ramaekers B, Grimm S, Pouwels X, Armstrong N, Ryder S, et al. *Obeticholic acid for treating primary biliary cirrhosis (Evidence Review Group report commissioned by the NIHR HTA Programme as project number STA 15/69/09) [Confidential until published]*. York: Kleijnen Systematic Reviews Ltd; Maastricht University, 30th November 2016. 115p.

[3] Intercept Pharmaceuticals Inc. *Biliary cirrhosis (primary) – obeticholic acid [ID:785]: Company evidence submission to National Institute of Health and Clinical Excellence. Single technology appraisal (STA)*, 30th September 2016. 216p.

[4] Intercept Pharmaceuticals Inc. *Biliary cirrhosis (primary) – obeticholic acid [ID:785]: Company evidence submission to National Institute of Health and Clinical Excellence. Single technology appraisal (STA)*, 26th October 2016 212p.

[5] Intercept Pharmaceuticals Inc. *Obeticholic acid for primary biliary cirrhosis [ID:785]: Erratum to Intercept's responses to ERG questions*, 21st November 2016 29p.

ERG appendix corrections page

The ERG appendix contains the results of the cost-effectiveness analysis, which includes the confidential patient access scheme discount based on the model in the original company submission. The company supplied an updated model that corrected an error regarding patient numbers used in the economic model. This correction page gives the cost-effectiveness results based on the updated company erratum and replaces the corresponding tables in the ERG appendix.

Sensitivity / scenario analyses

Table 1: Company's scenario analysis 1 results: UDCA intolerant patients

Technologies	Total			Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA versus No treatment
No treatment (placebo)	£103,233	11.30	6.91	_	-	-	_
OCA titration	£251,671	16.68	13.61	£148,439	5.38	6.70	£22,160
Source: Table 35 of the Erratum submitted by the company on 20th December 2016.1							
LYG: Life-year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost-effectiveness ratio; OCA: Obeticholic acid							

Table 2: Company's scenario analysis 1 results: UDCA inadequate responders

Technologies	Total			Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA + UDCA versus UDCA
UDCA +	£96,977	12.35	8.11	-	-	_	-
placebo							
OCA titration	£261,791	16.78	13.72	£164,814	4.43	5.61	£29,374
+ UDCA							
Source: Table 36 of the Erratum submitted by the company on 20 th December 2016. ¹							
LYG: Life-year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost-effectiveness ratio; OCA: Obeticholic acid; UDCA: ursodeoxycholic acid							

¹ Intercept Pharmaceuticals Inc. *Biliary cirrhosis (primary) – obeticholic acid [ID:785]: Erratum to Intercept's responses to ERG questions. Single technology appraisal (STA)*, 21st November 2016. 18-19p.

Table 3: Company's scenario analysis 2 results: UDCA intolerant patients

Technologies	Total			Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA versus No treatment
No treatment (placebo)	£94,717	10.89	6.39	_	_	-	-
OCA titration	£250,303	16.61	13.52	£155,586	5.73	7.13	£21,824
Source: Table 37 of the Erratum submitted by the company on 20th December 2016.1							
LYG: Life-year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost-effectiveness ratio; OCA: Obeticholic acid							

Table 4: Company's scenario analysis 2 results: UDCA inadequate responders

Technologies	Total			Incremental			ICER (£)
	Costs (£)	LYG	QALYs	Costs (£)	LYG	QALYs	OCA+ UDCA versus UDCA
UDCA + placebo	£89,666	12.00	7.67	—	-	_	_
OCA titration + UDCA	£260,540	16.72	13.65	£170,874	4.72	5.98	£28,596
Source: Table 38 of the Erratum submitted by the company on 20th December 2016.1							
LYG: Life-year gained; QALYs: Quality-adjusted life years; ICER: Incremental cost-effectiveness ratio; OCA: Obeticholic acid; UDCA: ursodeoxycholic acid							



in collaboration with:

Maastricht University 2 and ERASMUS UNIVERSITEIT ROTTERDAM INSTITUTE OF HEALTH POLICY & MANAGEMENT

Obeticholic acid for treating primary biliary cirrhosis

ERRATA

This document contains errata in respect of the ERG report in response to the company's factual accuracy check.

Page nr:	Change:
12,	Percentage decrease in ALP was omitted from the primary composite outcome.
47,49,54,99	'This has been amended on all pages to 'percentage of participants with alkaline
	phosphatase (ALP) < 1.67 x upper limit of normal (ULN) and total bilirubin \leq
	ULN and ALP \geq 15% decrease from baseline at 12 months)'
20	Section 2.2, second paragraph 'antibody' replaced with 'autoantibody'.
20	In the final paragraph repeated words 'for second-line treatment' were deleted.
21	In the first full paragraph the sentence 'Draft guidance from the BSG note that
	there is no consensus on routine use of UDCA nost-transplant ' has been
	removed
26 11	
20, 11	In the description of the POISE trial one sentence has been amended for clarity.
	'The patients in POISE were only up-titrated to 10 mg OCA at six months if they
	did not reach the primary endpoint criteria for response.' One sentence has been
	added. 'The remaining patients in this arm responded to OCA treatment and so
	did not require up-titration, as per the protocol.'
28	Section 3.4 third paragraph was amended to: 'the estimated study completion
20	date is April 2023 '
35	The first full paragraph has been amended to: 'In addition to this trial another
55	Phase 2 study (747-205) is currently ongoing in the US that includes nations
	from 18 years of age and aims to investigate the change from baseline in HDI
	motion To years of age and anns to investigate the change from baseline in TIDL metabolism as the primary outcome?
26	Table 4.2 final bullet point has been amended to: 'Combinations of the presence
50	radie 4.5 million bunch point has been amended to. Combinations of the presence
	of absence of biochemical response to UDCA treatment (yes/no) and intolerance
26	10 UDCA (yes/no)
30	Table 4.5 footnote has been amended to Paris I criteria defined as $ALP \leq 5X \cup LN$
27	and AST $\leq 2x$ ULN and total billirubin \leq ULN
57	Table 4.4 Bile acids row amended to: Absolute values and change from
	baseline by treatment group at 6 and 12 months follow-up from participants
	who had a confirmed fasting of approximately 8 nours or more prior to their
10	Visit in the following:
43	ERG comments second bullet point has been amended to: 'It is a multicentre
	international trial with 21/ participants comparing OCA in combination with
	UDCA to UDCA alone in patients with an inadequate response to UDCA and
	comparing OCA to no treatment in patients intolerant to UDCA'
44	First bullet point amended: 'Although stratified on several important variables,
	the POISE trial had differences between treatment groups at baseline. In
	particular the placebo group and the OCA 10 mg fixed dose group had a slightly
	higher incidence of pruritus than the OCA titration group and the placebo group
	had a slightly higher incidence of fatigue compared with both OCA treatment
	groups, which should be borne in mind particularly when interpreting results of
	adverse events.'
54	In the first paragraph the following sentence has been amended to: 'Study 747
	202 reported mean (SD) percentage change in ALD levels was 22.7% (17.8) for
	202 reported mean (SD) percentage enange in ALF revers was $-25.7%$ (17.8) 101 the OCA 10 mg group versus -2.6% (12.5) for placeba?
	$\frac{1}{12.3} 101 \text{ praceod}.$

1. SUMMARY

1.1 Critique of the decision problem in the company's submission

The patient population described in the final scope issued by the National Institute for Health and Care Excellence (NICE) was '*People with primary biliary cirrhosis whose disease has an inadequate response to, or who are unable to tolerate, ursodeoxycholic acid*'. For patients whose disease has an inadequate response to ursodeoxycholic acid (UDCA), obeticholic acid (OCA) was to be compared to UDCA alone or in combination with fibrates. For patients who are unable to tolerate UDCA, comparators were fibrates or no additional treatment. Outcomes included mortality, liver function based on markers of liver biochemistry, symptoms, including pruritus, fatigue and abdominal pain, time to liver transplantation, primary biliary cirrhosis (PBC)-related events, adverse effects of treatment and health-related quality of life (HRQoL).

The decision problem in the company submission (CS) differed from the scope in a number of ways. Firstly, fibrates were not considered by the company to be a comparator to OCA. Secondly, the main evidence presented (the POISE trial) considered only surrogate outcomes. No data were available from the trial on long-term clinical outcomes outlined in the scope such as mortality and liver transplantation. Finally the number of patients in POISE receiving OCA as monotherapy (i.e. without UDCA) was very small (11 patients) so results for this group of patients should be treated with some caution.

1.2 Summary of clinical effectiveness evidence submitted by the company

The company conducted a systematic review to inform the submission. The aim of the systematic review was 'to identify all relevant evidence for the efficacy and safety of interventions used to treat PBC.' A post-hoc decision was taken to include only trials with an OCA treatment arm.

The systematic review identified one main randomised controlled trial, POISE, and two supporting Phase 2 trials. The company did not pool the results of the trials.

POISE, was an international trial of 217 patients including patients from the UK. The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. Seventy-three patients received 10 mg OCA, 71 patients received 5 mg OCA rising to 10 mg during months 6 to 12 if they had an inadequate response to UDCA (the titration group).

The primary outcome of POISE was a composite one (percentage of participants with alkaline phosphatase (ALP) < 1.67 x upper limit of normal (ULN) and total bilirubin \leq ULN and ALP \geq 15% decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events (AEs) was similar across groups. Events occurring more frequently in treatment groups included pruritus. It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

The Phase 2 trials supported the positive findings of POISE on surrogate outcomes.

1.3 Summary of the ERG's critique of clinical effectiveness evidence submitted

The systematic review conducted by the company was broadly appropriate to the scope of this submission. The CS and response to clarification provided sufficient details for the ERG to appraise the searches. A broad range of databases were searched, and additional searches of conference proceedings and other relevant resources including trials databases, HTA agencies, specialist and organisational

The supporting study was a meta-analysis of 4,845 patients diagnosed with PBC from 1959 to 2012 with a median follow-up of 7.3 years. It included both UDCA treated and non-treated patients.¹¹

The CS states that 'PBC has a substantial detrimental impact on quality of life, and HRQoL impairment is correlated with the severity of the disease. '³ They cite a number of studies including a UK cohort study of 2,353 patients¹² in which '35% reported impairment of HRQoL compared with 6% of healthy controls (p<0.001), and 46% rated their overall health as 'fair' or 'poor' compared with 15% of healthy controls (p<0.0001).¹² '³

In addition to its impact on patients, the burden of the disease from the healthcare perspective is outlined. '*PBC is also associated with considerable healthcare costs. In 2014/15 there were 707 hospital admissions in England for PBC (ICD10 K74.3), accounting for 963 consultant episodes and 3,767 bed days^{7.,3} They further note that '<i>PBC is one of the most frequent indications for liver transplantation in Europe....There were 621 elective liver transplants performed in the UK in 2014/2015, of which at least 7% were for PBC.*^{13,3}

ERG comment:

- The references cited by the company were checked. The underlying health problem was considered to be appropriately described and appropriate references cited.
- The impact on patient quality of life was appropriately highlighted. The ERG notes that whilst the impact of PBC may be limited in early-diagnosed UDCA-responsive patients, among those who are unresponsive highly disabling symptoms such as fatigue and pruritus are common and patients face possible progression into cirrhosis and ultimately liver failure.

2.2 Critique of company's overview of current service provision

The company states that no NICE guidance or pathway specific to PBC is available.³

The CS states that 'Patients are commonly asymptomatic at diagnosis, and are referred to secondary care on discovering abnormal liver function and / or positive autoantibody by blood tests at a GP visit for an unrelated illness.^{7, 8} Occasionally patients are referred internally, most commonly from a rheumatologist. Rarely, patients are referred due to pruritus, abnormal ultrasound, or decompensation (ascites). '³

The company further state that '*The only licensed treatment for PBC is ursodeoxycholic acid* ((UDCA). '³ The company states that '*On diagnosis of PBC, UDCA is prescribed at 13-15mg/kg/day.* Patients are monitored at 3-4 months for tolerability, and at 6 and 12 months to gauge response and compliance to therapy. '³ The company cites two relevant clinical guidelines.^{14, 15} Both recommend long-term therapy at this dosage of UDCA as a first line of treatment.^{14, 15}

The company states that 'up to 74% of patients have an incomplete response to UDCA and there are currently no available licensed or effective treatment options for these patients.'³ The company states that 'There are several response criteria that have been proposed to define non-response / progression...; however, there is no consensus as to which of these criteria should be used.'³

The indication for obeticholic acid is for 'the treatment of PBC in combination with UDCA in adults with an inadequate response to UDCA or as monotherapy in adults unable to tolerate UDCA' (Section 2.2.2 of the CS)³ Therefore the place of obeticholic acid in the clinical pathway for PBC would be as a second-line treatment.

The CS states that 'There are no licensed or effective drugs approved for second-line treatment for the management of patients with an inadequate response to, or intolerance to, UDCA.'³ They further state

that 'other treatments have been trialled for use in PBC, including budesonide and fibrates (which are contraindicated in PBC).[CS refs 22,23] However, limited efficacy has been observed in these other treatments.³

The CS states that 'Liver transplantation is the only treatment for patients with late-stage PBC, where UDCA has limited efficacy.¹⁴' The CS outlines the challenges and risks of liver transplantation and notes that 'up to 43% of patients will have a recurrence of PBC within 15 years¹⁶.'

The CS states that '*No additional tests or investigations are required for OCA treatment.*'³ Although OCA treatment is to be initiated by specialists in the treatment of PBC, the company states that no additional infrastructure will be required for treatment with OCA. However patients are required to undergo a consultation six months after treatment initiation to assess tolerability and determine if the dose should be increased to 10 mg to achieve optimal response.³

ERG comment:

- The company correctly states that there is no specific NICE guidance on PBC. The CS cites two clinical guidelines by the European Association for the Study of the Liver (EASL) published in 2009¹⁴ and the more up-to-date but as yet unpublished British Society of Gastroenterology (BSG) guidelines.¹⁵ The CS correctly states that UDCA is the only licensed treatment at first line for PBC.
- Response rates to UDCA between 20 and 70% have been identified.⁴ The company correctly highlight that determining response rates to UDCA depends on the specific criteria used to assess response. Several sets of criteria have been used across the research literature. We are advised by our clinical expert that in practice various systems may be used including simple clinical criteria.
- Clinical guidelines state that there are no second line agents when patients have failed to respond to UDCA.^{14, 15} OCA would therefore be the only agent available at second line. The company mentions other agents that have been investigated (fibrates are most relevant to this submission). The statement *'limited efficacy has been observed'* in relation to other treatments was not supported by any references.³ A further discussion of the role of fibrates as a comparator to OCA can be found in Section 3 of this report, the decision problem.

of monotherapy including 49 patients, 20 of whom received 10 mg OCA and 23 received placebo. The remainder received 50 mg OCA and were not relevant to the proposed dosage.²⁹ As stated above, the role of OCA monotherapy at the appropriate dosage has only been investigated in a small number of patients so results for this group of patients may not be reliable.

• In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo.²⁷ Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six months period. The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response.²⁷ The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg.²⁷ The remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol. The company notes that *'further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.* '³ This appears to be reasonable but is an assumption.

3.3 Comparators

For people whose disease has an inadequate response to UDCA the CS evaluates the use of UDCA alone as a comparator as per the final scope issued by NICE. For people who are unable to tolerate UDCA the CS (more specifically the POISE trial) compared obsticholic acid alone to placebo (representing no additional treatment as stated in the NICE scope). As stated above, 11 patients (7%) took obsticholic acid alone in POISE (five in the OCA titration group and six in the OCA 10 mg fixed dose group and five patients received a placebo alone). The CS presents results for this comparison but notes that these *'should be interpreted with caution due to low patient numbers.'*³

However the comparators addressed in the CS differ from those in the final scope issued by NICE in that fibrates have not been included as a comparator for people whose disease has an inadequate response to UDCA nor for those who are intolerant to UDCA. The company provide several justifications for this including: fibrates not being licensed for this indication in the UK, not being standard care, they are contraindicated in PBC, have been rarely used in the UK and efficacy is as yet unproven with a small number of limited studies with some safety concerns.³

ERG comment:

- As stated above, the role of OCA monotherapy compared to placebo (no treatment) has only been investigated in a small number of patients so results for this group may be less reliable than those taking obeticholic acid in combination with UDCA.
- The omission of fibrates from the comparators in the decision problem was investigated by the ERG in several ways. We consulted clinical experts and identified systematic reviews of fibrates as the most reliable source of up to date evidence. Our response to the omission of fibrates is given in the table below.

Assertion in the company submission	ERG comments
'Fibrates are not licensed in the UK, nor are they standard of care.' 'They are rarely used, with only for of patients in the UK-PBC cohort having ever taken fibrates for any condition (not necessarily for PBC)' ³	Fibrates are not licensed for this indication. We are advised that fibrates are not routinely used in the UK for PBC.

Table 3.1: Response to the omission of fibrates as a comparator in the submission

3.4 Outcomes

The final scope issued by NICE specified patient outcomes including mortality, symptoms including pruritus, fatigue and abdominal pain, time to liver transplantation, PBC-related events, health-related quality of life and adverse effects of treatment.⁴ However all efficacy outcomes in the POISE trial which forms the main evidence of the submission relate to surrogate outcomes.³ These include a range of liver function biomarkers and other biomarkers related to PBC. Disease-specific quality of life based on the PBC-40 tool was collected in POISE but was not specifically mentioned as being addressed in the decision problem. Adverse events were also collected in POISE.³

The CS explains that '*due to the rare and chronic nature of PBC and the slow progression in most patients, a long-term trial is required to capture clinical outcomes such as mortality, transplant-free survival, and the incidence of complications.*'³ The POISE trial is of 12 months' duration although there is an ongoing five year long-term safety extension (LTSE). The LTSE has data from 12 months but efficacy is still based on surrogate outcomes.³ The company cite three references to provide evidence for the correlation of the primary outcome of POISE (combined alkaline phosphatise (ALP) and bilirubin levels) to disease prognosis.^{11, 24, 25}

The company also note that there is an ongoing trial of obeticholic acid (COBALT) that measures clinical outcomes. Details of the COBALT trial are provided in Table 39 of the CS.³ Briefly, this is a double-blind randomised, placebo-controlled trial of obeticholic acid on clinical outcomes in patients with PBC. The estimated enrolment is 350 across up to 170 international sites. Duration is estimated to be approximately eight years according to the time to accrue approximately 121 primary endpoint events. The primary outcome is a composite endpoint of clinical events. The start date of the trial was December 2014 and the estimated study completion date is April 2023.³

ERG comment:

- The ERG draws the attention of the committee to the fact that the main evidence in the submission relates to surrogate outcomes. The ERG did not have the means to systematically review the validity of this correlation. Instead the ERG investigated the three references cited by the company as evidence of the correlation. The ERG observed that one of the references cited was a meta-analysis of individual patient data (4,845 patients) which concluded that 'Levels of alkaline phosphatase and bilirubin can predict outcomes (liver transplantation or death) of patients with PBC and might be used as surrogate end points in therapy trials. '¹¹ This finding was supported by a retrospective review of 73 patients treated with UDCA over 36 months.²⁴ Biochemical response predicted histological progression, assessed with paired biopsies about 10 years apart, as well.²⁵ The ERG is satisfied that the company demonstrated evidence of some correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease are unclear.
- The definitive effects of obeticholic acid on clinical outcomes relevant to patients awaits the results of the COBALT trial.³⁷
- The ERG asked the company if any interim data were available from the COBALT trial which investigates clinical outcomes. The company confirmed that 'there are no interim data available and no interim analysis is planned since COBALT is an events-driven trial and 121 primary endpoint events need to have occurred before the trial will report and analyses will be performed.' ³²

summarised only for subjects who received 10 mg OCA, since this is the upper limit for the licensed indication of OCA.³

An additional trial, COBALT (NCT02308111) is mentioned in the CS (Section 4.14 of the submission).³ This trial seeks to evaluate the effect on OCA on clinical outcomes such as transplant-free survival. The ERG can confirm that the trial is still ongoing and that no interim data has been made available.³⁷ Please see Section 3.4 of this report for a discussion relating to COBALT. In addition to this trial, another Phase 2 study (747-205) is currently ongoing in the US that includes patients from 18 years of age and aims to investigate the change from baseline in HDL metabolism as the primary outcome.³

As stated in the CS, '*the results from the double-blind phase of POISE are presented as the main efficacy evidence and the two Phase 2 studies as supporting evidence*'. ³ In this regard, this report follows the same structure. More detail will be provided on the double blind phase of POISE with a briefer summary of the Phase 2 trials.

ERG comment:

- The ERG agrees that the POISE trial provides the main evidence for the comparison with UDCA or placebo and that the two Phase 2 trials represent supporting evidence.
- The ERG is satisfied that no data from the ongoing COBALT trial could have been used to inform the CS.
- The included trials did not compare OCA to fibrates as outlined in the scope.⁴

4.2.2 Overview of the POISE trial

An overview of the POISE trial and its extension study is presented in Table 4.2.

Trial no. (acronym)	Population	Intervention	Comparator
747-301 (POISE); NCT01473524	Patients diagnosed with PBC with ALP ≥1.67x ULN and/or total bilirubin >ULN but <2x ULN who fail to respond to or are intolerant to treatment with UDCA.	Oral OCA (5 mg or 10 mg) taken OD.	Placebo
LTSE to POISE	All participants who completed the 12-month double-blind phase of POISE and who were willing to enrol in the 5-year LTSE phase of the study.	Oral OCA (5– 25 mg ^a) taken OD	N/A

Table 4.2: Summary of the POISE trial and its extension study

Source: CS

Footnote: a) All patients initiated OCA at 5 mg OD; daily dose could be up-titrated if a satisfactory response was not achieved in 5 mg increments to a total dose of 25 mg OD (one increment per 3 months permitted), depending on tolerability

ALP, alkaline phosphatase; LTSE, long-term safety extension; OCA, Obeticholic acid; OD, once daily; PBC, primary biliary cholangitis; UDCA, ursodeoxycholic acid; ULN, upper limit of normal.

The main methodological features of the double-blind phase of the POISE trial are summarised in Table 4.3.
Table 4.3: Overview of POISE t

Trial Design	Phase 3, randomised double-blind, placebo-controlled, parallel group trial. Stratified randomisation according to:
	- higher risk of developing clinical outcomes i.e. Paris I criteria*
	- intolerance to UDCA
	- Combinations of the presence or absence of biochemical response to UDCA treatment (yes/no) and intolerance to UDCA (yes/no)'
Setting	59 sites in 13 countries. (7 sites in England and 2 sites in Scotland)
Participants	217 were randomised (216 gave consent)
Interventions	3 arms:
	Placebo (with or without UDCA) $(n = 73)$
	10 mg OCA, with or without UDCA ($n = 73$)
	Titration (5mg OCA rising to 10 mg during months 6 to 12 if inadequate response to UDCA), with or without UDCA) ($n = 71$)
Follow-up	12 months
Primary Outcome	Percentage of participants in the 10 mg OCA fixed dose group at 12 month achieving the composite endpoint:
	- ALP <1.67x ULN, and
	- total bilirubin \leq ULN, and
	- ALP decrease $\geq 15\%$ from baseline
Source CS ³	
Footnote: * Paris I cr	iteria defined as ALP $\leq 3x$ ULN and AST $\leq 2x$ ULN and total bilirubin \leq ULN
ALP, alkaline phosp	hatase; CS, company submission; TEAEs, Treatment-emergent adverse events;
ULN, upper limit of	normal

Table 4.4 details the secondary endpoints of the POISE trial and their definitions.

Study outcome	Definition
ALP response	Absolute and percentage change from baseline in ALP at Month 6 and
rates	Month 12
	Percentage of participants with a decrease in ALP from baseline of $\geq 10\%$,
	\geq 15%, \geq 20%, and \geq 40% at Week 2, Month 3, Month 6, Month 9 and Month
	12, Percentage of participants with ALP \leq ULN, summarised by treatment at all
	post-baseline assessments

 Table 4.4: Summary of secondary endpoints of the double-blind phase of POISE

Biochemical treatment response criteria	 Percentage of participants meeting the Paris I, Paris II, Mayo II, Toronto II, or Rotterdam response criteria* at Week 2, Month 3, Month 6, Month 9, and Month 12. The analysis was repeated for subgroups of participants who met and did not meet the requirement of a responder at baseline for the endpoint analysed. Number and percentage of participants with: Normal bilirubin (≤ULN) and normal albumin (≥LLN), Moderate (bilirubin >ULN or albumin <lln), and<="" li=""> Severe (bilirubin >ULN and albumin <lln)< li=""> at baseline, Week 2, Month 3, Month 6, Month 9 and Month 12. </lln)<></lln),>
Clinical laboratory values	Defined as the absolute and percentage change from baseline in ALP, GGT, ALT, AST, total and conjugated (direct) bilirubin, albumin, and prothrombin time (PT) and international standardised ratio (INR) summarised by treatment group and visit.
Questionnaire PBC-40	Absolute change from baseline in six domains (cognitive, social, emotional function, fatigue, itch, and general symptoms) were summarised by treatment using descriptive statistics at Week 2, Month 3, Month 6, Month 9, and Month 12. A total score was not calculated.
Patient Research Questionnaire	'A simple patient research questionnaire was administered at Month 12, or at termination if the subject withdrew from the study prior to this, to request feedback about the subjects' perception of the study.' ³
Biomarkers and non-invasive assessments of liver fibrosis Bile acids	 Absolute change from baseline in the following markers: Markers of hepatic fibrosis, inflammation and other disease relevant biomarkers, including CRP, tumour necrosis factor-α (TNF-α), transforming growth factor β (TGF-β), fibroblast growth factor-19 (FGF-19), Interleukin-6 (IL-6), CK-18, autotaxin, and lysophosphatidic acid (LA), at Month 6 and Month 12 Enhanced liver fibrosis (ELF) score and its components hyaluronic acid (HA), procollagen-3 N-terminal peptide (P3NP), and tissue inhibitor of metalloproteinase 1 (TIMP-1) at Month 6 and Month 12 Hepatic stiffness measurements (at selected study sites) at Month 12, assessed by transient elastography (TE). Absolute values and change from baseline by treatment group at 6 and 12
Bile acids	Absolute values and change from baseline by treatment group at 6 and 12 months' follow-up in the following from participants who had a confirmed fasting of approximately 8 hours or more prior to their visit in the following Total bile acids, total endogenous bile acids, and totals for the individual bile acids (UDCA, chenodeoxycholic acid [CDCA], deoxycholic acid [DCA], cholic acid [CA] and lithocholic acid [LCA]) and their respective conjugates Proportion of each of the individual bile acids relative to total bile acids
OCA Pharmacokinetics analysis	Values at Month 6 and Month 12 for OCA (unconjugated), glyco-OCA, tauro-OCA, and total OCA from participants who had a confirmed fasting of

	approximately 8 hours or more prior to their visit were included in the analysis.
	The effect of BAS on OCA and total bile acid concentration, as well as the percentage change in ALP, were explored as part of the PK analysis. Relationships between plasma total OCA concentrations (unconjugated and conjugated) and FGF-19 concentrations, endogenous bile acid concentrations, ALP and liver enzyme levels, and severity of pruritus were explored.
Safety	TEAEs defined as any adverse events (AEs) that newly appeared, increased in frequency, or worsened in severity following initiation of investigational product. Additionally, Pruritus was measured by using the 5-D pruritus questionnaire and a VAS score at week 2, months 3, 6, 9 and 12.
Clinical laboratory evaluations	Physical examinations, vital signs, body weight and BMI, electrocardiongram, dual-emission x-ray absorptioametry (DEXA) scan of the femoral neck and lumbar spine, Mayo Risk Score to assess survival and MELD score to assess the severity of chronic liver disease were assessed at week 2, months 3, 6, 9, and 12.
Source: Section 4.3.6 of Footnote: *Different ro BAS, bile acids seques	of CS ³ esponse criteria studied in POISE are reported in Table 16 of the CS. strants; CS, company submission; TEAEs, Treatment-emergent adverse events

Disease characteristic	Placebo (n=73)	OCA titration (n=70)	OCA 10 mg fixed dose (n=73)	Total (n=216)
Mean duration of	f PBC, years			
Mean (SD)	8.3 (5.4)	8.3 (5.8)	9.2 (6.9)	8.6 (6.0)
Median	7.4	7.2	8.5	7.8
Min, max	0.9, 21.8	0.3, 27.0	0.0, 32.3	0.0, 32.3
Duration of PBC subgroups, n (%)				
\leq 7.5 years	39 (53)	36 (51)	30 (41)	105 (49)
>7.5 years	34 (47)	34 (49)	43 (59)	111 (51)
Source: Table 21 of CS ³				

The CS states that 'the mean age at time of diagnosis was 47.3 years with a mean duration of PBC of 8.6 years' and that 'There were slightly more subjects <50 years of age at PBC diagnosis (58%) compared with ≥ 50 years of age.' Fifty-nine percent of patients had pruritus at baseline (43% mild, 15% moderate and 1% severe) and 59% had a history of fatigue. The CS notes that 'the majority (94–99%) of subjects had a baseline INR ≤ 1.3 , indicative of a population in an early stage of disease progression'.

The CS notes that 'In general, each variable was well balanced across treatment groups.' However 'the overall incidence of pruritus at baseline was slightly higher for subjects in the placebo treatment group (64% and OCA 10mg fixed dose group (60%) than in the OCA titration group (53%).....The overall incidence of fatigue was slightly higher for subjects in the placebo treatment group (67%) than in the OCA titration and OCA 10mg fixed dose groups (54% and 56%, respectively).'

ERG comments on POISE:

POISE, has several strengths and matches the NICE scope in several ways:

- It is a randomised controlled trial in a PBC population relevant to England and Wales.
- It is a multicentre international trial with 217 participants comparing OCA in combination with UDCA to UDCA alone in patients with an inadequate response to UDCA and comparing OCA to no treatment in patients intolerant to UDCA.
- Follow-up is of 12 months' duration covering an extensive range of surrogate outcomes.

There are a number of limitations in applying the results of the trial to the NICE scope:

• The number of patients intolerant to UDCA and receiving OCA as monotherapy was limited to 11 patients (five [7%] in the OCA titration group and six [8%] in the OCA 10 mg group). Five patients received OCA placebo. As such the results for this group of patients should be considered with caution due to the low numbers.

The ERG noted that the POISE trial included patients at an early state of PBC progression as stated in Section 4.5.2.3 of the CS 'the majority (94–99%) of subjects had a baseline INR ≤ 1.3 , indicative of a population in an early stage of disease progression'.³ The ERG asked the company to clarify how OCA would benefit patients with more advanced PBC. In the response to clarification letter, the company stated that two analyses were performed in more advanced stages of PBC. Several definitions were used to approximate severity of the disease including a clinical composite definition and presence of cirrhosis. The results of these analyses are provided in Section 4.2.4. It should be noted that these are based on a subset of 72 patients classified as having advanced disease and 36 who had cirrhosis.³²The ongoing COBALT trial includes more advanced patients and should provide more definitive results for these patients.

- Although stratified on several important variables, the POISE trial had differences between treatment groups at baseline. In particular the placebo group and the OCA 10 mg fixed dose group had a slightly higher incidence of pruritus than the OCA titration group and the placebo group had a slightly higher incidence of fatigue compared with both OCA treatment groups, which should be borne in mind particularly when interpreting results of adverse events.
- In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo.²⁷ Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response.²⁷ The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg.²⁷ The remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol.The company notes that '*further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.*^{'3} This appears to be reasonable but is an assumption.
- The POISE trial presents evidence using surrogate outcomes only for clinical effectiveness. As stated in Section 3.4, the ERG is satisfied that the company has demonstrated evidence of some correlation between the surrogate outcomes of interest and longer-term clinical outcomes. However the extent of the correlation and the optimal thresholds of surrogate outcomes to predict long-term disease is unclear.
- In the POISE CSR some early clinical outcomes were retrospectively gathered.⁴⁶ Although these are presented in Section 4.2.5, the time scale of the trial (12 months) does not allow conclusions to be drawn based on this early long-term evidence. The definitive effects of obeticholic acid on clinical outcomes relevant to patients awaits the results of the COBALT trial.³⁷
- As stated in the NICE scope for Obeticholic acid, the HRQoL of PBC patients is relevant for this submission.⁴ The company used a disease-specific questionnaire PBC-40 that assesses symptoms across several domains: fatigue, emotional and social, cognitive function, general symptoms and itch.⁴⁷ Brief results as presented in the CSR are given in Section 4.2.4 of the report.

4.2.3 Quality assessment of POISE

We reproduce the company's quality assessment of the POISE trial³ alongside the ERG's views on the quality of the trial.

ERG comment: It can be seen that the ERG agrees with the company's assessment and finds that the trial has been well conducted with appropriate procedures for randomisation, allocation concealment, blinding and outcome assessment.

4.2.4 POISE: Efficacy results

The primary efficacy endpoint (percentage of participants with ALP < $1.67 \times ULN$ and total bilirubin $\leq ULN$ and ALP $\geq 15\%$ decrease from baseline at 12 months) is detailed in Table 4.9 below.

	Responders (%)		
	Month 12		
Placebo (n=73)	10%		
10 mg OCA (n=73)	47%		
Titration OCA (n=70)	46%		
Titration subgroup ^{\dagger}			
Remained at 5 mg OCA for 12 months (n=36)	53%		
Titrated to 10 mg OCA at Month 6 (n=33)	39%		
Source: Table 23 CS ³ Footnote: [†] There was one participant who withdrew from the trial due to an AE after 8 days of study medication and therefore there were no data for this participant at Month 6 and Month 12, 8Of the 12			

Table 4.5: S	ummary of	primary	efficacy	outcome	in	POISE
--------------	-----------	---------	----------	---------	----	-------

Footnote: [†]There was one participant who withdrew from the trial due to an AE after 8 days of study medication, and therefore there were no data for this participant at Month 6 and Month 12. §Of the 12 participants who did not respond but did not up-titrate, nine did not increase their dose due to adverse events, and three recorded their reason as 'other'

ALP, alkaline phosphatase; CS, company submission; OCA, obeticholic acid.

It can be seen from the table above that at 12 months 47% of participants in the 10mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons).

Secondary outcomes

For the two main surrogate outcomes relevant to the primary composite outcome in POISE, namely ALP and total bilirubin levels, the company reports that at 12 months, 25 (34%) and 21 (30%) of participants from the OCA 10 mg fixed dose and OCA titration groups respectively achieved an ALP reduction from baseline \geq 40% compared with 1% in the placebo group.³ For the total bilirubin outcome, decreases in the absolute change from baseline were observed for both OCA treatment groups compared with an increase for the placebo treatment group. At 12 months, mean bilirubin levels in the OCA 10 mg group were 9.7 (SE 0.6), 9.9 (SE 0.6) in the OCA titration group and 13.2 (1.0) in the placebo group.³

The CSR states that 'The disease-specific measure for PBC showed no clinically significant improvements in comparison to placebo for the global score or individual scores of general symptoms, fatigue, cognitive function, and emotional/social domains; however, a difference was observed in itch scores in the earlier treatment months.'⁴⁶ The CSR further states that 'During the initial 3 months of treatment, the largest LS mean increase in itch was observed for the OCA 10 mg group, followed by the OCA titration group.... The LS mean difference in itch score between the OCA 10 mg group and placebo group was statistically significant at both Week 2 (p = 0.0048) and Month 3 (p < 0.0001) but not at any subsequent time points.' ⁴⁶

Subgroup of patients with advanced disease

A total of 72 participants (30, 22 and 20 in the placebo, OCA titration and OCA 10 mg respectively) were classified as having advanced disease using a clinical composite definition that included biochemical criteria (not specified), non-invasive measures of fibrosis, biopsies and/or medical history of decompensation and presence of cirrhosis.³² The company concluded that '*there was not an increased risk (of adverse events) in patients with advanced disease*'.³² No efficacy data were presented for these patients. A post-hoc analysis of efficacy and safety of 36 patients with cirrhosis (13 in the placebo arm, 13 in the OCA titration arm and 10 in the OCA 10 mg arm) was conducted. The company stated that '*ALP levels were significantly more reduced over 12 months in the OCA treatment groups than the placebo groups and bilirubin increased in the placebo group but remained stable in the OCA treatment group. The primary composite outcome in POISEwas met by 8% of subjects in the placebo arm, 54% in the OCA 10 mg arm after 12 months.'³²*

ERG comment:

- The ERG notes the improvements shown in the OCA treatment groups in terms of the combined efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP \geq 15% decrease from baseline at 12 months). This is supported by improvements in other surrogate markers. However no firm data on clinical outcomes are available. It is too early to draw conclusions on long-term clinical outcomes and the need for transplant based on the retrospective data available.
- The ERG notes that improvements in surrogate outcomes were not reflected in the disease specific quality of life tool (PBC-40) over the 12 month period.
- The improvements seen in patients with more advanced disease (briefly reported) are based on lower patient numbers as the majority of patients in POISE had earlier stage disease.
- As few patients used OCA as monotherapy it is unclear if results accurately reflect this patient group.

4.2.5 POISE: Safety results

A summary of the adverse events as described in the CS³ is detailed in Table 4.12.

Participants, n (%)	Placebo n=73	OCA titration n=70	OCA 10 mg n=73
Any TEAE	66 (90)	65 (93)	69 (95)
Total number of TEAEs	452	471	467
Any treatment-related AE [†]	38 (52)	42 (60)	54 (74)
Any SAEs	3 (4)	11 (16)	8 (11)
Total number of SAEs	8	15	11
TEAEs by severity			
Mild	29 (40)	16 (23)	19 (26)
Moderate	28 (38)	27 (39)	29 (40)
Severe	9 (12)	22 (31)	21 (29)
Any TEAE leading to discontinuation	2 (3) [‡]	5 (7) [§]	8 (11)¶
Discontinuation due to pruritus	0 (0)	1 (1)	7 (10)
Number of deaths	0 (0)	1 (1)	0 (0)

Table 4.6: Overview of adverse events:	safety population (POISE)
--	---------------------------

In summary, both studies found a statistically significant effect of OCA 10 mg on ALP levels from baseline to end of study versus placebo. In study 747-201 (monotherapy) the mean (SD) percentage change in ALP levels was -44.5% (24.4) for the OCA 10 mg group versus +0.4% (15.3) for placebo. Study 747-202 reported mean (SD) percentage change in ALP levels was -23.7% (17.8) for the OCA 10 mg group versus -2.6% (12.5) for placebo. Both results were statistically significant at p<0.0001.

ERG comment: The Phase two trials lend support to the findings of POISE on the benefits of OCA on surrogate outcomes for patients with PBC. However both trials were of a short-term duration (three months) and a variety of doses were used limiting the comparability of the trials to POISE and the possibility of pooling results.

4.3 Critique of trials identified and included in the indirect comparison and/or multiple treatment comparison

Not applicable as there was no indirect comparison or multiple treatment comparison in the CS.

4.4 Critique of the indirect comparison and/or multiple treatment comparison

Not applicable as there was no indirect comparison or multiple treatment comparison in the CS.

4.5 Additional work on clinical effectiveness undertaken by the ERG Not applicable.

4.6 Conclusions of the clinical effectiveness section

The company conducted a systematic review to identify studies comparing OCA to the comparators outlined in the NICE scope.⁴ Overall the ERG is satisfied that the relevant direct evidence comparing OCA and UDCA has been presented. However no trials comparing OCA to fibrates were identified.

One Phase 3 trial, POISE, with 217 patients was presented as the main source of evidence.²⁷ The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. However the group receiving OCA as monotherapy is underrepresented in this trial (11 patients). Additionally, the majority of patients in POISE appeared to be at an earlier stage of disease so the effects on those with more advanced disease are less clear.

The POISE trial was well-conducted. However it only examined surrogate outcomes. These were justified by the company based on previously published research. In POISE, OCA shows positive effects on surrogate endpoints, and there is some evidence that surrogate endpoints are related to relevant outcomes. However, the size of the relationship is unclear.

The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP \geq 15% decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events was similar across groups. Events occurring more frequently in treatment groups were noted. The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

Two supporting Phase 2 trials were presented (including one of OCA monotherapy and one of OCA in combination with UDCA). These were not similar enough to be pooled with POISE but added support to the positive findings on surrogate outcomes. Clinical outcomes await the publication of the COBALT trial.³⁷

8. OVERALL CONCLUSIONS

8.1 Statement of principal findings

Overall the ERG is satisfied that the relevant direct evidence comparing OCA and UDCA has been presented. However no trials comparing OCA to fibrates were identified.

One Phase 3 trial, POISE, with 217 patients was presented as the main source of evidence.²⁷ The trial compared UDCA and combined UDCA and OCA in patients with an inadequate response to UDCA and OCA and placebo in those who were intolerant to UDCA. However the group receiving OCA as monotherapy is underrepresented in this trial (11 patients). Additionally, the majority of patients in POISE appeared to be at an earlier stage of disease so the effects on those with more advanced disease are less clear.

The POISE trial was well-conducted. However it only examined surrogate outcomes. These were justified by the company based on previously published research. In POISE, OCA shows positive effects on surrogate endpoints, and there is some evidence that surrogate endpoints are related to relevant outcomes. However, the size of the relationship is unclear.

The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and \geq 15% ALP decrease from baseline at 12 months). At 12 months 47% of participants in the OCA 10 mg group achieved the primary outcome, 46% in the titration group and 10% in the placebo group (p < 0.0001 for both comparisons). The results of other surrogate outcomes supported these findings. Incidence of adverse events was similar across groups. Events occurring more frequently in treatment groups were noted. The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability.

Two supporting Phase 2 trials were presented (including one of OCA monotherapy and one of OCA in combination with UDCA). These were not similar enough to be pooled with POISE but added support to the positive findings on surrogate outcomes. Clinical outcomes await the publication of the COBALT trial.³⁷

The main areas of ERG concern were the fact that the company did not use the POISE trial data directly for the non-OCA regimen, the lack of transparency and justification regarding methods and data to extrapolate the POISE trial data beyond the time horizon of 12 months and to final outcomes, the implausible high utility values in the biomarker component of the model and the lack of justification for HRQoL decrement for the health states of the liver disease component of the model, except HCC. The ERG has incorporated various adjustments to the company base-case. The ERG base-case resulted in ICERs (probabilistic) of £ and £ for UDCA inadequate responders and UDCA intolerant patients respectively. The most influential adjustments/corrections made by the ERG were using TPs from the POISE trial for the non-OCA regimens and using age dependent utilities for the low and moderate risk health states. However, the ERG base-case ICER should be regarded as a lower bound as the exploratory analysis showed that alternative assumptions, for the TPs after 12 months and the non-transparent calibration method used by the company, resulted in substantially higher ICERs (ranging from £ to £ and from £ to £ for UDCA inadequate responders and UDCA intolerant patients respectively).

As the methods used by the company to estimate and extrapolate treatment effectiveness are lacking transparency and justification, the ICERs (presented by both the company and the ERG) should be interpreted with caution. Given the lack of long-term results of the POISE trial, the ERG would generally agree with using a calibration approach (as adopted by the company), to extrapolate the

National Institute for Health and Care Excellence Centre for Health Technology Evaluation

Pro-forma Response

ERG report

Obeticholic acid for treating primary biliary cirrhosis [ID785]

You are asked to check the ERG report from Klejinen Systematic Reviews Ltd to ensure there are no factual inaccuracies contained within it.

If you do identify any factual inaccuracies you must inform NICE by **5pm**, **9 December 2016** using the below proforma comments table. All factual errors will be highlighted in a report and presented to the Appraisal Committee and will subsequently be published on the NICE website with the Evaluation report.

The proforma document should act as a method of detailing any inaccuracies found and how and why they should be corrected.

Issue 1 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 1.2, Page 12 First sentence in fourth paragraph of this section has an error: "The primary outcome of POISE was a composite one (percentage of participants with alkaline phosphatase (ALP) < 1.67 x upper limit of normal (ULN) and total bilirubin \leq ULN and ALP \geq decrease from baseline at 12 months)."	The sentence should be amended to: 'The primary outcome of POISE was a composite one (percentage of participants with alkaline phosphatase (ALP) < 1.67 x upper limit of normal (ULN) and total bilirubin ≤ ULN and ALP ≥15% decrease from baseline at 12 months).'	This is an error in describing the primary endpoint of POISE, the pivotal trial for OCA.	This has been corrected.

Issue 2 Error and further information required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 1.2, Page 12 Error and further information required for the last sentence of the fourth paragraph in this section: "It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability."	The sentence should be amended to: 'It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg fixed dose group-which did not titrate based on tolerability., but only 1% discontinued due to pruritus in the OCA titration group that more closely represents the OCA dosing regimen recommended for use in clinical practice.'	The percentage quoted in the original statement (10%) refers to the OCA 10 mg fixed dose group, within which there was no option to titrate the dose. In addition, Intercept believe that it is a fairer representation to also include the percentage discontinuing due to pruritus in the OCA titration group, since this more closely resembles the OCA dosing regimen recommended for use in clinical practice, and the lower rate of adverse events (including pruritus)	Not a factual error. See also issues 28 and 35.

	was one of the reasons that the	
	titration regimen was proposed for	
	use in clinical practice compared	
	with the 10 mg fixed dose regimen.	

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 1.3, Page 13 Further explanation is required in the third bullet point in this section: "In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only up-titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg."	The bullet point should be amended to: 'In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only uptitrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the There were 33 patients in the OCA titration group who actually up-titrated to 10 mg.; the remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol. The company notes that 'further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.' This appears to be reasonable but is an assumption.'	Further explanation is required for context, and to explain that 'this patient group' in the quote from the company submission refers to the entire OCA titration group and not just to the 33 patients who up- titrated. It is important to note that, in terms of efficacy, results are likely to be underestimated due to the lower dose of OCA received (compared with clinical practice) for the patients who did not up-titrate.	Not a factual error. Extra detail not needed for the executive summary.

Issue 3 Further explanation required

Issue 4 Misrepresentation

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 1.5, Page 16 There is misrepresentation in the second sentence of the final paragraph of this page: "Given this lack of transparency, also after multiple requests from the ERG, the methods used by the company are still unclear."	The sentence point should be amended to: 'Given this lack of transparency, also after multiple requests from the ERG, tThe methods used by the company are still unclear to the ERG.'	The methods and sources of data were explained both in the appendices of the company submission and in the workbook, and further explanation was provided to the ERG on request.	Not a factual inaccuracy. The methods to estimate treatment effectiveness for patients receiving UDCA only and no treatment are unclear to the ERG based on the company submission and clarification responses (see ERG comment in section 5.2.6 of the ERG report for more details). No correction needed.

Issue 5 Misquote of the company submission

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 2.2, Page 20 There is a misquote in the first sentence of the second paragraph in this section: "The CS states that ' <i>Patients are</i> <i>commonly asymptomatic at</i> <i>diagnosis, and are referred to</i> <i>secondary care on discovering</i> <i>abnormal liver function and / or</i> <i>positive antibody by blood tests</i> <i>at a GP visit for an unrelated</i> <i>illness.</i> "	The sentence should be amended to: 'The CS states that 'Patients are commonly asymptomatic at diagnosis, and are referred to secondary care on discovering abnormal liver function and / or positive auto antibody by blood tests at a GP visit for an unrelated illness.'	This is an error.	This has been corrected.

Issue 6 Typographical error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 2.2, Page 20 There is a typographical error in the first sentence of the sixth paragraph in this section: "The CS states that ' <i>There are no</i> licensed or effective drugs approved for second-line treatment for second-line treatment for the management of patients with an inadequate response to, or intolerance to, UDCA.""	The sentence should be amended to: 'The CS states that 'There are no licensed or effective drugs approved for second-line treatment for second-line treatment for the management of patients with an inadequate response to, or intolerance to, UDCA.'	This is an error.	This has been corrected.

Issue 7 Misquote of the company submission

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 2.2, Page 21 There is a misquote in seventh paragraph in this section: "The CS states that ' <i>Liver</i> <i>transplantation is the only</i> <i>treatment for patients with late-</i> <i>stage PBC, where UDCA has</i> <i>limited efficacy.</i> ' The CS outlines the challenges and risks of liver transplantation and notes that 'up to 43% of patients will have a recurrence of PBC within 15 years. Draft guidance from the	The paragraph should be amended to: 'The CS states that ' <i>Liver transplantation is the only treatment for patients with late-stage PBC, where UDCA has limited efficacy.</i> ' The CS outlines the challenges and risks of liver transplantation and notes that ' <i>up to 43% of patients will have a recurrence of PBC within 15 years. Draft guidance from the BSG note that there is no consensus on routine use of UDCA post-transplant.</i> ''	This is an error.	The sentence 'Draft guidance from the BSG note that there is no consensus on routine use of UDCA post-transplant.' Has been removed.

BSG note that there is no		
consensus on routine use of		
UDCA post-transplant.'"		

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 3.2, Page 26 Further explanation is required in the final bullet point in this section: "In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six months period. The patients in POISE were only up-titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg. The company notes that 'further benefit in terms of efficacy is likely to be seen in clinical practice in this patient	The bullet point should be amended to: 'In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six months period. The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the There were 33 patients in the OCA titration group who actually up-titrated to 10 mg-; the remaining patients in this arm responded to OCA treatment and so did not require up- titration, as per the protocol. The company notes that 'further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.' This appears to be reasonable but is an assumption.'	Further explanation is required for context, and to explain that 'this patient group' in the quote from the company submission refers to the entire OCA titration group and not just to the 33 patients who up- titrated. It is important to note that, in terms of efficacy, results are likely to be underestimated due to the lower dose of OCA received (compared with clinical practice) for the patients who did not up-titrate.	One sentence has been amended for clarity. 'The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response.' One sentence has been added. 'The remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol.'

Issue 8 Further explanation required

Issue 9 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 3.4, Page 28 There is an error in the final sentence of the third paragraph of this section: "The start date of the trial was December 2014 and the study completion date is April 2023."	The sentence should be amended to: 'The start date of the trial was December 2014 and the estimated study completion date is April 2023.'	The details of the trial that have been extracted by the ERG from clinicaltrials.gov refer to the estimated completion date.	This has been corrected.

Issue 10 Typographical error and further information required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.1, Page 35 There is a typographical error in the final sentence of the fourth paragraph of this section: "In addition to this trial, another Phase 2 study (747-205) is currently ongoing in the US that includes patients from 8eightyears of age and aims to investigate the change from baseline in HDL metabolism."	The sentence should be amended to: 'In addition to this trial, another Phase 2 study (747-205) is currently ongoing in the US that includes patients from 8eightyears 18 years of age and aims to investigate the change from baseline in HDL metabolism as the primary outcome .'	This was a typographical error in the company submission, and the sentence has been expanded to make clear that the investigation of HDL metabolism was the primary outcome, since other outcomes were measured in the trial as well as HDL metabolism as secondary outcomes.	This has been corrected.

Issue 11 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.3, Page 36 There is an error in the final bullet point in the 'Trial design' row: "presence or absence of biochemical response to UDCA treatment"	The bullet point should be amended to: 'Combinations of the presence or absence of biochemical response to UDCA treatment (yes/no) and intolerance to UDCA (yes/no) '	The stratification criteria are not reflected correctly.	This has been corrected.

Issue 12 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.3, Page 36 There is an error in the footnote: "Paris I criteria defined as ALP ≤3x ULN and AST ≤2x ULN and total bilirubin ≤1 mg/dL (17 µmol/l)"	The footnote should be amended to: 'Paris I criteria defined as ALP ≤3x ULN and AST ≤2x ULN and total bilirubin ≤ ULN 1 mg/dL (17 µmol/I) '	For the stratification criteria, total bilirubin in the response criteria was defined as ≤ULN, not ≤1 mg/dL.	This has been corrected.

Issue 13 Addition required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.4, Page 37 There is an addition required in the 'Bile acids' row: "Absolute values and change from baseline by treatment group at 6 and 12	The statement should be amended to: 'Absolute values and change from baseline by treatment group at 6 and 12 months' follow-up from participants who had a confirmed fasting of approximately 8 hours or more prior to their visit in the following:'	This has been highlighted in the 'OCA pharmacokinetics analysis' row, and so should also be included in the 'Bile acids' row for consistency.	This has been corrected.

months' follow-up in the		
following:"		

Issue 14 Addition required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.2, Page 43 There is an addition required in the second bullet point under the heading 'ERG comments on POISE': "It is a multicentre international trial with 217 participants comparing OCA to UDCA alone in patients with an inadequate response to UDCA and comparing OCA to no treatment in patients intolerant to UDCA"	The bullet point should be amended to: 'It is a multicentre international trial with 217 participants comparing OCA in combination with UDCA to UDCA alone in patients with an inadequate response to UDCA and comparing OCA to no treatment in patients intolerant to UDCA'	An addition is required to clarify that OCA was taken in combination with UDCA in patients with an inadequate response to UDCA.	This has been corrected.

Issue 15 Further explanation required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.2, Page 44 There is further explanation required in the first full bullet point on this page: "Although stratified on several important variables, the POISE trial had differences between treatment groups at baseline. In particular the placebo group had a higher	The bullet point should be amended to: 'Although stratified on several important variables, the POISE trial had differences between treatment groups at baseline. In particular the placebo group and the OCA 10 mg fixed dose group had a slightly higher incidence of pruritus than the OCA titration group and the placebo group had a slightly higher incidence of fatigue	Further explanation is required to fully reflect the results of POISE.	This has been corrected.

incidence of pruritus and fatigue	compared with both OCA treatment groups,
particularly when interpreting	when interpreting results of adverse events.'
results of adverse events."	

Issue 16 Further explanation required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.2, Page 44 There is further explanation required in the second full bullet point on this page: "In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only up- titrated to 10 mg OCA if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the 33 patients who actually up-titrated to 10 mg. The company notes that <i>'further</i> <i>benefit in terms of efficacy is</i>	The bullet point should be amended to: 'In the POISE trial 73 patients were randomised to a fixed dose of 10 mg OCA and 73 patients received the placebo. Seventy-one patients were randomised to the titration group which reflects the recommended dosage. They received OCA 5 mg OD for the initial six month period. The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response. The ERG draws to the attention of the committee that the evidence for obeticholic acid given at the recommended dosage is based on the There were 33 patients in the OCA titration group who actually up-titrated to 10 mg-; the remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol. The company notes that <i>further benefit in terms of efficacy is likely to be seen in clinical practice in this patient group due to the higher dose of OCA.</i> ' This appears to be reasonable but is an assumption.'	Further explanation is required for context, and to explain that 'this patient group' in the quote from the company submission refers to the entire OCA titration group and not just to the 33 patients who up- titrated. It is important to note that, in terms of efficacy, results are likely to be underestimated due to the lower dose of OCA received (compared with clinical practice) for the patients who did not up-titrate.	One sentence has been amended for clarity. 'The patients in POISE were only up-titrated to 10 mg OCA at six months if they did not reach the primary endpoint criteria for response.' One sentence has been added. 'The remaining patients in this arm responded to OCA treatment and so did not require up-titration, as per the protocol.'

likely to be seen in clinical		
practice in this patient group due		
to the higher dose of OCA.' This		
appears to be reasonable but is		
an assumption."		

Issue 17 Typographical error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.8, Page 45 There is a typographical error in the penultimate sentence in the second row of the table on this page: "Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor designee(s) in the event of a medical emergence."	The sentence should be amended to: 'Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor designee(s) in the event of a medical emergencey.'	This was a typographical error in the company submission.	Correct, but the mistake was in the company submission, so no correction made.

Issue 18 Typographical error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.8, Page 46 There is a typographical error in the penultimate sentence in the first row of the table on this page: "Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor	The sentence should be amended to: 'Access to randomisation codes and corresponding treatment assignment was made available to the appropriate Sponsor designee(s) in the event of a medical emergencey.'	This was a typographical error in the company submission.	Correct, but the mistake was in the company submission, so no correction made.

designee(s) in the event of a		
medical emergence."		

Issue 19 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.4, Page 47 There is an error/omission in the first sentence of this section: "The primary efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months) is detailed in Table 4.9 below."	The sentence should be amended to: 'The primary efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin ≤ ULN and ALP ≥15% decrease from baseline at 12 months) is detailed in Table 4.9 below.'	This is an error in describing the primary endpoint of POISE, the pivotal trial for OCA.	This has been corrected.

Issue 20 Error/omission

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.4, Page 47 There is an error/omission in the first sentence under the heading 'Secondary outcomes' in this section: "For the two main surrogate outcomes relevant to the primary composite outcome in POISE, namely ALP and total bilirubin levels, the company reports that at 12 months, 25 (34%) and 21 (30%) of	The sentence should be amended to: 'For the two main surrogate outcomes relevant to the primary composite outcome in POISE, namely ALP and total bilirubin levels, the company reports that at 12 months, 25 (34%) and 21 (30%) of-participants from the OCA 10 mg fixed dose and OCA titration groups respectively achieved an ALP reduction from baseline ≥40% compared with 1 (1%) participant in the placebo group.'	This is an error/omission amended for consistency.	It is clear from the text what is meant here, so no correction needed.

participants from the OCA 10 mg		
fixed dose and OCA titration		
groups respectively achieved an		
ALP reduction from baseline		
≥40% compared with 1% in the		
placebo group "		

Issue 21 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.4, Page 48 There is an error in the text above Table 4.11 in this section: "The incidence of reaching a MELD score >15 (with a baseline MELD of <15) is given in Table 4.11."	The sentence should be amended to: 'The incidence of reaching a MELD score >15 (with a baseline MELD of < 12) is given in Table 4.11.'	This is an error, and should correspond with Table 41 (page 170) of the CSR. There is an error in the text of the CSR on page 168.	This was taken from the text of the CSR which was incorrect. No correction to the report.

Issue 22 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.4, Page 49 There is an error/omission in the first sentence of the first bullet point under the heading 'ERG comment' in this section: "The ERG notes the improvements shown in the OCA treatment groups in terms of the combined efficacy endpoint (percentage of participants with ALP < 1.67 x	The sentence should be amended to: 'The ERG notes the improvements shown in the OCA treatment groups in terms of the combined efficacy endpoint (percentage of participants with ALP < 1.67 x ULN and total bilirubin ≤ ULN and ALP ≥15% decrease from baseline at 12 months).'	This is an error in describing the primary endpoint of POISE, the pivotal trial for OCA.	This has been corrected.

ULN and total bilirubin ≤ ULN and ALP decrease from baseline at 12 months) "		

Issue 23 Context

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.4, Page 49 The final bullet point under the heading 'ERG comment' in this section is out of context: "As few patients used OCA as monotherapy it is unclear if results accurately reflect this patient group."	This bullet point should be removed.	Section 4.2.4 has not discussed the results of OCA monotherapy (vs OCA in combination with UDCA), and so this statement is not relevant here.	Not a factual error.

Issue 24 Context

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.5, Page 52 The second paragraph on this page is out of context: "The company also state that ' <i>Fatigue</i> <i>and nausea were the only other</i> <i>related TEAE that occurred at an</i> <i>incidence</i> ≥5%; <i>however, these</i> <i>events were balanced between</i> <i>placebo and OCA treatment</i> <i>arms.</i> "	The paragraph should be amended to: 'The company also state that 'As expected based on prior experience with OCA treatment in patients with PBC, the most common related TEAE was pruritus. In all treatment groups, the majority of pruritus AEs were considered related to investigational product. The incidence and number of subjects with related TEAEs of pruritus was 27 subjects (37%) in the placebo group, 35 subjects (50%) in the OCA titration group, and 48 subjects (66%) in the OCA 10 mg	The previous paragraph was related to TEAEs in general, whereas this paragraph concerns treatment-related TEAEs.	Not a factual error. No correction needed.

group. Fatigue and nausea were the only other related TEAE that occurred at an	
incidence ≥5%; however, these events were balanced between placebo and OCA treatment arms.'	

Issue 25 Omission

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Table 4.14, Page 53 There is an omission in the cell relating to trial design of study 747-202: "Multi-centre, randomised, double-blind, placebo-controlled, multi-dose, Phase 2 parallel group study of OCA in with UDCA in participants with a proven or likely diagnosis of PBC."	The cell should be amended to: 'Multi-centre, randomised, double-blind, placebo-controlled, multi-dose, Phase 2 parallel group study of OCA in combination with UDCA in participants with a proven or likely diagnosis of PBC.'	This is an omission.	It is clear from the text what is meant here, so no correction needed.

Issue 26 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.2.6, Page 54 There is an error in the penultimate sentence of the first paragraph on this page: "Study 747-202 reported mean (SD) percentage change in ALP levels was -44.5% (24.4) for the OCA	The sentence should be amended to: 'Study 747-202 reported mean (SD) percentage change in ALP levels was -44.5% (24.4) - 23.7% (17.8) for the OCA 10 mg group versus +0.4% (15.3) -2.6% (12.5) for placebo.'	This is an error.	This has been corrected.

10 mg group versus +0.4% (15.3) for placebo."		

Issue 27 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.6, Page 54 There is an error/omission in the first sentence of the fourth paragraph of this section: "The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months)."	The sentence should be amended to: 'The primary outcome of POISE was a composite one (percentage of participants with ALP < 1.67 x ULN and total bilirubin ≤ ULN and ALP ≥15% decrease from baseline at 12 months).'	This is an error in describing the primary endpoint of POISE, the pivotal trial for OCA.	This has been corrected.

Issue 28 Error and further information required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 4.6, Page 54 Error and further information required for the last sentence of the fourth paragraph in this section: "The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability."	The sentence should be amended to: 'It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg fixed dose group-which did not titrate based on tolerability:, but only 1% discontinued due to pruritus in the OCA titration group that more closely represents the OCA dosing regimen recommended for use in clinical practice.'	The percentage quoted in the original statement (10%) refers to the OCA 10 mg fixed dose group, within which there was no option to titrate the dose. In addition, Intercept believe that it is a fairer representation to also include the percentage discontinuing due to pruritus in the OCA titration group, since this more closely resembles the OCA dosing regimen	Not a factual error. See also issues 2 and 35.

		recommended for use in clinical	
		practice, and the lower rate of	
		adverse events (including pruritus)	
		was one of the reasons that the	
		titration regimen was proposed for	
		use in clinical practice compared	
		with the 10 mg fixed dose regimen.	
I			

Issue 29 Misrepresentation

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 5.2.1, Page 61: There is misrepresentation in the 'Synthesis of evidence in outcomes' row of Table 5.2: "Evidence of effectiveness came primarily from the POISE trial but also from other sources and unclear calibration methods."	'Evidence of effectiveness came primarily from the POISE trial but also from other sources and unclear calibration methods.'	The methods and sources of data were explained both in the appendices of the company submission and in the workbook. The calibration was carried out using Solver in Excel to estimate the transition probability to better fit all transition-free survival estimates used as the basis for the calibration. The model was then used to compare the liver transplant-free survival estimates from POISE to the ones calculated using the model to evaluate their validity.	Not a factual inaccuracy. The methods to estimate treatment effectiveness for patients receiving UDCA only and no treatment are unclear to the ERG based on the company submission and clarification responses (see ERG comment in section 5.2.6 of the ERG report for more details). No correction needed.

Issue 30 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 5.2.3, Page 64: There is an error in the first paragraph of this section: "Patients enter the model in the moderate (76.85%) and severe risk (23.15%) health states in the biomarker component of the model, according to ALP and bilirubin levels. These proportions of patients starting the model were not justified and apparently not in line with the distribution of patients in the POISE study: the distribution there was 91.58% in the moderate and 8.42% in the severe risk health state, based on Table 49 in the CS for the OCA titration arm, the company's model for the OCA treatment arm and the second clarification excel file for the UDCA arm and OCA arms."	'Patients enter the model in the moderate (76.85%) and severe risk (23.15%) health states in the biomarker component of the model, according to ALP and bilirubin levels as well as presence of compensated cirrhosis. These proportions of patients starting the model were not justified and apparently not in line with the distribution of patients in the POISE study: the distribution there was 91.58% in the moderate and 8.42% in the severe risk health state, based on Table 49 in the CS for the OCA titration arm, the company's model for the OCA treatment arm and the second clarification excel file for the UDCA arm and OCA arms.'	The distribution of patients to the moderate (76.85%) and severe risk (23.15%) health states was intentional. The patient numbers in the model were different as only patients who had all ALP/bilirubin observations at all appointments could be used. The distribution of patients to the moderate and severe risk health states was taken directly from the POISE CSR. The ERG's suggestion (that the patient numbers used for transition probabilities used in the model should also be used for the initial patient distribution) does not take into account the patients that could not be included in the model analysis as well as patients with compensated cirrhosis.	Not a factual inaccuracy. The estimation of the proportions of patients starting in the moderate (76.85%) and severe risk (23.15%) health states is unclear to the ERG based on the company submission and clarification responses. No correction needed.

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 5.2.6, Page 67: There is misrepresentation in the third sentence onwards of the first paragraph on this page: "The company did not provide a detailed account of their calibration method in the CS. In the response to the clarification letter, the company provided more detail and an Excel spreadsheet used for the calibration, which was done in four steps: The company calibrated 1) the transition from DCC to the pre-liver transplant health state and liver-related death; 2) transitions from the severe risk health state to the liver disease component health states; 3) transitions from the moderate risk health state to the severe risk health state and other liver disease component states and; 4) transitions from the low risk health state to the severe risk health state to the severe risk health state and other liver disease component states. There is little detail on the methods, the estimates used and the sources of data."	This passage should be removed.	The methods and sources of data were explained both in the appendices of the company submission and in the workbook. The calibration was carried out using Solver in Excel to estimate the transition probability to better fit all transition-free survival estimates used as the basis for the calibration. The model was then used to compare the liver transplant-free survival estimates from POISE to the ones calculated using the model to evaluate their validity.	Not a factual inaccuracy. The methods to estimate treatment effectiveness for patients receiving UDCA only and no treatment are unclear to the ERG based on the company submission and clarification responses (see ERG comment in section 5.2.6 of the ERG report for more details). No correction needed.

Issue 32 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 5.2.12, Page 89: There is an error from the fourth sentence of the first paragraph of	'The company stated that it required patient- level data and calculating mean ALP or bilirubin levels per group would potentially	Two of the four publications stated would not have been appropriate to compare against.	Not a factual inaccuracy. This is only an addition to one sentence of the ERG report.
this page: "The company stated that it required patient-level data and calculating mean ALP or bilirubin levels per group would potentially introduce bias. The company deemed cross validation impossible in the	validation impossible in the absence of other cost utility models for PBC. The ERG does not agree with this, because the cost effectiveness review identified four studies focussing on PBC treatment, of which two may have been used for limited validation.'	Longworth et al and Ratcliffe et al both focused on liver transplantation as a comparator and therefore it would have been inappropriate to compare results to a model that examined a different intervention.	No correction needed.
absence of other cost utility models for PBC. The ERG does not agree with this, because the cost effectiveness review identified four studies focussing on PBC treatment.		For the two remaining models (Boberg et al and Pasha et al), any validation performed would only be partial in scope – only life years for the no treatment and UDCA arms could be compared.	
		Comparing UDCA costs would be of limited use as Boberg et al and Pasha et al do not elaborate on how the UDCA costs were calculated; the Boberg model took a Norwegian perspective so any comparison to the model used in the CS would have been of limited use due to the different costs of treatment between the UK and Norway.	
		The Pasha et al model did include methods for how the cost of UDCA	

	were calculated, and broadly used the same methods as the model	
	used in the company submission, albeit with different, US-specific	
	costs.	

Issue 33 Misrepresentation

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 5.2.12, Page 89: There is misrepresentation in the third paragraph on this page: "Moreover, given the lack of transparency regarding the methods used by the company (e.g. concerning the calibration process described in Section 5.2.6), even after requests from the ERG (Clarification Questions B12, B14, B15 and B16), the methods used by the company are still unclear."	This passage should be removed.	The methods and sources of data were explained both in the appendices of the company submission and in the workbook. The calibration was carried out using Solver in Excel to estimate the transition probability to better fit all transition-free survival estimates used as the basis for the calibration. The model was then used to compare the liver transplant-free survival estimates from POISE to the ones calculated using the model to evaluate their validity.	Not a factual inaccuracy. The methods to estimate treatment effectiveness for patients receiving UDCA only and no treatment are unclear to the ERG based on the company submission and clarification responses (see ERG comment in section 5.2.6 of the ERG report for more details). No correction needed.

Issue 34 Error

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 8.1, Page 99 There is an error/omission in the first sentence of the fourth	The sentence should be amended to: 'The primary outcome of POISE was a composite one (percentage of participants with ALP <	This is an error in describing the primary endpoint of POISE, the pivotal trial for OCA.	This has been corrected.

paragraph of this section: "The primary outcome of POISE was a composite one (percentage of participants with ALP < $1.67 \times$ ULN and total bilirubin \leq ULN and ALP decrease from baseline at 12 months)."	1.67 x ULN and total bilirubin ≤ ULN and ALP ≥15% decrease from baseline at 12 months).		
---	---	--	--

Issue 35 Error and further information required

Description of problem	Description of proposed amendment	Justification for amendment	ERG Response
Section 8.1, Page 99 Error and further information required for the last sentence of the fourth paragraph in this section: "The most notable was pruritus and it was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg group which did not titrate based on tolerability."	The sentence should be amended to: 'It was noted that 10% of patients discontinued treatment due to pruritus in the OCA 10 mg fixed dose group-which did not titrate based on tolerability:, but only 1% discontinued due to pruritus in the OCA titration group that more closely represents the OCA dosing regimen recommended for use in clinical practice.'	The percentage quoted in the original statement (10%) refers to the OCA 10 mg fixed dose group, within which there was no option to titrate the dose. In addition, Intercept believe that it is a fairer representation to also include the percentage discontinuing due to pruritus in the OCA titration group, since this more closely resembles the OCA dosing regimen recommended for use in clinical practice, and the lower rate of adverse events (including pruritus) was one of the reasons that the titration regimen was proposed for use in clinical practice compared with the 10 mg fixed dose regimen.	Not a factual error. See also issues 2 and 28.